# Algebra & Number Theory

Volume 7 2013 <sub>No. 2</sub>

# Algebra & Number Theory

msp.org/ant

#### EDITORS

MANAGING EDITOR Bjorn Poonen Massachusetts Institute of Technology Cambridge, USA EDITORIAL BOARD CHAIR David Eisenbud

University of California Berkeley, USA

#### BOARD OF EDITORS

Georgia Benkart	University of Wisconsin, Madison, USA	Susan Montgomery	University of Southern California, USA
Dave Benson	University of Aberdeen, Scotland	Shigefumi Mori	RIMS, Kyoto University, Japan
Richard E. Borcherds	University of California, Berkeley, USA	Raman Parimala	Emory University, USA
John H. Coates	University of Cambridge, UK	Jonathan Pila	University of Oxford, UK
J-L. Colliot-Thélène	CNRS, Université Paris-Sud, France	Victor Reiner	University of Minnesota, USA
Brian D. Conrad	University of Michigan, USA	Karl Rubin	University of California, Irvine, USA
Hélène Esnault	Freie Universität Berlin, Germany	Peter Sarnak	Princeton University, USA
Hubert Flenner	Ruhr-Universität, Germany	Joseph H. Silverman	Brown University, USA
Edward Frenkel	University of California, Berkeley, USA	Michael Singer	North Carolina State University, USA
Andrew Granville	Université de Montréal, Canada	Vasudevan Srinivas	Tata Inst. of Fund. Research, India
Joseph Gubeladze	San Francisco State University, USA	J. Toby Stafford	University of Michigan, USA
Ehud Hrushovski	Hebrew University, Israel	Bernd Sturmfels	University of California, Berkeley, USA
Craig Huneke	University of Virginia, USA	Richard Taylor	Harvard University, USA
Mikhail Kapranov	Yale University, USA	Ravi Vakil	Stanford University, USA
Yujiro Kawamata	University of Tokyo, Japan	Michel van den Bergh	Hasselt University, Belgium
János Kollár	Princeton University, USA	Marie-France Vignéras	Université Paris VII, France
Yuri Manin	Northwestern University, USA	Kei-Ichi Watanabe	Nihon University, Japan
Barry Mazur	Harvard University, USA	Efim Zelmanov	University of California, San Diego, USA
Philippe Michel	École Polytechnique Fédérale de Lausan	ne	

#### PRODUCTION

production@msp.org

Silvio Levy, Scientific Editor

See inside back cover or msp.org/ant for submission instructions.

The subscription price for 2013 is US \$200/year for the electronic version, and \$350/year (+\$40, if shipping outside the US) for print and electronic. Subscriptions, requests for back issues and changes of subscribers address should be sent to MSP.

Algebra & Number Theory (ISSN 1944-7833 electronic, 1937-0652 printed) at Mathematical Sciences Publishers, 798 Evans Hall #3840, c/o University of California, Berkeley, CA 94720-3840 is published continuously online. Periodical rate postage paid at Berkeley, CA 94704, and additional mailing offices.

ANT peer review and production are managed by EditFLOW® from Mathematical Sciences Publishers.

PUBLISHED BY mathematical sciences publishers nonprofit scientific publishing http://msp.org/ © 2013 Mathematical Sciences Publishers



# The system of representations of the Weil–Deligne group associated to an abelian variety

#### Rutger Noot

Fix a number field  $F \subset \mathbb{C}$ , an abelian variety A/F and let  $G_A$  be the Mumford–Tate group of  $A_{/\mathbb{C}}$ . After replacing F by finite extension one can assume that, for every prime number  $\ell$ , the action of the absolute Galois group  $\Gamma_F = \text{Gal}(\bar{F}/F)$  on the étale cohomology group  $H^1_t(A_{\bar{F}}, \mathbb{Q}_\ell)$  factors through a morphism  $\rho_\ell \colon \Gamma_F \to G_A(\mathbb{Q}_\ell)$ . Let v be a valuation of F and write  $\Gamma_{F_v}$  for the absolute Galois group of the completion  $F_v$ . For every  $\ell$  with  $v(\ell) = 0$ , the restriction of  $\rho_\ell$  to  $\Gamma_{F_v}$  defines a representation  $W_{F_v} \to G_{A/\mathbb{Q}_l}$  of the Weil– Deligne group.

It is conjectured that, for every  $\ell$ , this representation of  ${}^{\prime}W_{F_v}$  is defined over  $\mathbb{Q}$  as a representation with values in  $G_A$  and that the system above, for variable  $\ell$ , forms a compatible system of representations of  ${}^{\prime}W_{F_v}$  with values in  $G_A$ . A somewhat weaker version of this conjecture is proved for the valuations of F, where A has semistable reduction and for which  $\rho_{\ell}(\operatorname{Fr}_v)$  is neat.

#### Introduction

Let  $F_v$  be a finite extension of the field  $\mathbb{Q}_p$  (for some prime number p) and let X be a proper and smooth variety over  $F_v$ . The Galois group  $\Gamma_{F_v} = \text{Gal}(\bar{F}_v/F_v)$  acts on the étale cohomology groups  $\text{H}^i(X_{\bar{F}_v}, \mathbb{Q}_\ell)$  for each prime number  $\ell$  and each i. It is a major problem in arithmetic geometry to determine to what extent the properties of these representations are independent of  $\ell$ . To obtain such independence results, one has to consider the restrictions of the representations above to the Weil group  $W_{F_v}$  of  $F_v$ . This is the subgroup formed by the elements of  $\Gamma_{F_v}$ , which induce an integral power of the Frobenius automorphism on the residue field of  $F_v$ .

In what follows it will always be assumed that  $\ell \neq p$ ; the case where  $\ell = p$  will be analysed in a later paper. Let us first assume that X has good reduction at v, that is, that X extends to a proper and smooth scheme over the ring of integers of  $F_v$ . This assumption implies that the inertia subgroup of  $\Gamma_{F_v}$  acts trivially on the

MSC2000: primary 11G10; secondary 14K15, 14F20.

Keywords: abelian variety, compatible system of Galois representations, Weil-Deligne group.

étale cohomology groups of X. Moreover, it follows from Deligne's work [1974a] on the Weil conjectures that the character of the representation of  $W_{F_v}$  on each  $H^i(X_{\bar{F}_v}, \mathbb{Q}_\ell)$  has values in  $\mathbb{Q}$  and is independent of  $\ell \neq p$ . In view of the triviality of the action of inertia, this amounts to a statement on the action of the subgroup of  $\Gamma_{F_v}$  generated by a Frobenius element. We will summarise this statement by saying that the representations of  $W_{F_v}$  on the  $H^i(X_{\bar{F}_v}, \mathbb{Q}_\ell)$  (for fixed *i* and variable  $\ell$ ) are *defined over*  $\mathbb{Q}$  and that they form a *compatible system* of representations of  $W_{F_v}$ .

If X only has potentially good reduction, the inertia group no longer acts trivially but its action on a given étale cohomology group of X factors through a finite quotient. It is still conjectured that the system of  $H^i(X_{\bar{F}_v}, \mathbb{Q}_\ell)$  (as always for fixed *i* and variable  $\ell$ ) forms a compatible system of representations of  $W_{F_v}$  which are defined over  $\mathbb{Q}$ , see for example [Serre 1994, 12.13?].

Serre actually states his conjecture in more generality because it applies to motives instead of varieties. The category of (pure) motives can be seen as an intermediate between varieties and their cohomology. It is a  $\mathbb{Q}$ -linear tannakian category, that is, an abelian category in which the morphisms between two objects form a  $\mathbb{Q}$ -vector space and which is equipped with "tensor products". All reasonable cohomology functors on the category of varieties factor through the category of motives so that any motive has cohomology groups in the same way as varieties do. We also refer to the cohomology groups of a motive M as the *realisations* of M. There are different constructions of motives, depending on how the morphisms are defined, but in any case the category of varieties.

In this paper we will consider the category of motives for absolute Hodge cycles developed in [Deligne and Milne 1982] where the morphisms are defined by absolute Hodge classes. This theory is particularly efficient for dealing with abelian varieties because the motivic Galois group of an abelian variety *A* coincides with its Mumford–Tate group, defined by the Hodge structure on  $H^1_B(A_{/\mathbb{C}}, \mathbb{Q})$ .

The motivic version of the conjecture can be stated in terms of motivic Galois groups in the following way. In any good category of motives it is possible to associate a motivic Galois group to any subcategory. This group is a linear proalgebraic group over  $\mathbb{Q}$ . Its defining property is the fact that any category of motives is equivalent to the category of representations of its motivic Galois group. The motivic Galois group  $G_M$  of any object M is defined as the group associated to the tannakian category generated by M and the Tate motive. The group  $G_M$  is related to the étale cohomology of M by the fact that, for each prime number  $\ell$ , the  $\ell$ -adic Galois representation associated to M factors through  $G_M(\mathbb{Q}_\ell)$ . This implies that the corresponding representations of the Weil group factor through  $G_{M/\mathbb{Q}_\ell}$ . Serre's  $\ell$ -independence conjecture for objects in the  $\otimes$ -category generated by M is then equivalent to the statement that the representations  $W_{F_v} \to G_{M/\mathbb{Q}_\ell}$  of the Weil group form a compatible system *defined over*  $\mathbb{Q}$ , *with values in*  $G_M$ ; see 2.3 for the precise definition.

For abelian varieties of CM type, the conjecture follows from the theory of complex multiplication developed by Shimura and Taniyama [1961], which yields a considerably more precise result. Indeed, let  $F \subset \mathbb{C}$  be a number field, A/Fan abelian variety of CM type and  $G_A$  the motivic Galois group of A. Serre [1968] constructs a commutative algebraic group  $S_{F,\mathfrak{m}}$  and a canonical system of representations  $\varphi_{\ell} \colon \Gamma_F \to S_{F,\mathfrak{m}}(\mathbb{Q}_{\ell})$ . By the theory of complex multiplication, the system of  $\ell$ -adic representations associated to A is the image of the system  $(\varphi_{\ell})$  by a  $\mathbb{Q}$ -rational morphism  $S_{F,\mathfrak{m}} \to G_A$ . Moreover, Deligne [1982] gives an explicit description of the motivic Galois group of the category of abelian varieties potentially of CM type in terms of the Taniyama group.

More generally, if  $F \subset \mathbb{C}$  is a number field, A/F an abelian variety and v a valuation of F where A has good reduction, the  $\ell$ -independence conjecture was proved in [Noot 2009], by a method and under additional hypotheses similar to those of the present paper. The case where A has ordinary reduction at v has been treated in [Noot 1995] by a completely different method. In the latter case, a stronger statement can be proved and it turns out that there is an element in  $G_A(\mathbb{Q})$  that is conjugate to the  $\ell$ -adic image of Frobenius for all  $\ell \neq p$ . As noted in the introduction of [Noot 1995], this is not the case in general. The reader may consult the introduction to [Noot 2009] for a more detailed discussion.

One may ask if these results can be generalised without any assumptions on the reduction of X. Before discussing the properties of the Galois representations provided by the étale cohomology groups of a variety X, consider any  $\ell$ -adic representation  $\rho_{\ell}$  of  $\Gamma_{F_v}$  for  $\ell \neq p$ . By a theorem of Grothendieck (see [Serre and Tate 1968, Appendix; Deligne 1973, §2, §8]), the action of a sufficiently small open subgroup of the inertia group can be described using a single endomorphism  $N_{\ell}$ , the monodromy operator. The restriction of  $\rho_{\ell}$  to the Weil group  $W_{F_v}$  can then be encoded by giving  $N_{\ell}$  together with a representation  $\rho'_{\ell}$  of  $W_{F_v}$ , which is trivial on an open subgroup of the inertia group. We will refer to such a pair  $(\rho'_{\ell}, N_{\ell})$  as a *representation of the Weil–Deligne group*  $'W_{F_v}$  of  $F_v$ . Where the *p*-adic étale cohomology is concerned, Fontaine's theory associates a representation of the Weil–Deligne group to any semistable *p*-adic representation as well, but we will not go into the details here.

It is conjectured that, applying the construction above to the system of Galois representations provided by the  $H^i(X_{\bar{F}_v}, \mathbb{Q}_\ell)$ , for fixed *i* and variable  $\ell$ , one obtains a compatible system of representations of  $W_{F_v}$  which are defined over  $\mathbb{Q}$ ; see for example [Fontaine 1994, 2.4.3, conjecture  $C_{WD}$ ] for a statement encompassing the *p*-adic representation. The conjecture on the  $\ell$ -independence of the representation of the Weil–Deligne group hinges on the monodromy-weight conjecture; see

[Illusie 1994, §3]. This elusive conjecture is somewhat more accessible under the hypothesis that *X* has semistable reduction, a hypothesis which implies in particular that the representations  $\rho'_{\ell}$  are trivial on the inertia subgroup of  $W_{F_v}$ . The action of inertia on  $H^i(X_{\bar{F}_v}, \mathbb{Q}_{\ell})$  is then determined by the monodromy operator  $N_{\ell}$ . Even if *X* has semistable reduction however, the monodromy-weight conjecture has so far only been proved under far more restrictive hypotheses. Apart from the cases where *X* is a curve or an abelian variety, the main achievement is due to [Rapoport and Zink 1982], which treats the case where *X* has dimension 2. We refer to [Ochiai 1999; Ito 2004] and the work of Scholze for some recent and very recent progress on the problems. It should finally be pointed out that the discussion above has an analogue in equal characteristics, which has proved much more accessible; see for example [Deligne 1980].

This paper aims to study the motivic version of Fontaine's  $C_{WD}$  conjecture. Under some additional hypotheses, described below, we will prove the compatibility conjecture for the system  ${}^{\prime}W_{F_v} \rightarrow G_{A/\mathbb{Q}_{\ell}}$  associated to an abelian variety A defined over a number field  $F \subset \mathbb{C}$ . Here v is a fixed valuation of F and  $\ell$  runs through the set of primes with  $v(\ell) = 0$ .

The hypotheses are twofold. First of all, we need to assume that the number field F is sufficiently big. We do not only need to ensure that A has semistable reduction, but also that the Mumford–Tate group  $G_A$  is connected and even that the Frobenius element at the given place of F is *weakly neat*; see Definition 3.5. Secondly, in certain cases we will need to work in a group that is slightly larger than the Mumford–Tate group; see Section 3.3. The Mumford–Tate group coincides with the identity component of this larger group. Enlarging the group obviously weakens the notion of conjugacy. The precise result is Theorem 3.6 and all definitions used in the statement are given in Section 3.

The strategy of the proof is inspired by the previous paper [Noot 2009], which treats the good reduction case. The idea is first to prove the statement for *tractable* abelian varieties (called *accomodantes* [ibid.]). In Section 4, we recall the notion of a tractable abelian variety as well as some related constructions stemming from [ibid.]. Tractable abelian varieties have many endomorphisms and the theorem is proved by combining more or less classical results concerning abelian varieties and 1-motives with the, equally classical, work of Springer and Steinberg on conjugacy classes in linear algebraic groups. The necessary results on representations of the Weil–Deligne group associated to a 1-motive follow from the theory sketched in [Fontaine 1994] by adding the action of an endomorphism algebra throughout. We make extensive use of [Raynaud 1994], which allows the reduction to the case of strict 1-motives. This is discussed, together with the relevant prerequisites, in Sections 1 and 2. Using the results of these sections, proof of the main theorem for a tractable abelian variety is given in Sections 5 and 6.

Once we have proved the main theorem for tractable abelian varieties, the general case can be deduced using the theory, developed in [Noot 2006], of lifting Galois representations along isogenies between the Mumford–Tate groups. This final step of the proof is carried out in Section 7. In order to construct the liftings of the Galois representations, one needs to extend the base field. At first, this leads to a proof of the main theorem over an uncontrollable extension of the base field. The results of Sections 1 and 2 are used again to deduce the theorem over the original base field. The condition that the Frobenius element is weakly neat is essential in this step.

In his thesis, Laskar [2011] generalises the results of this paper, as well as those of [Noot 2009], to a larger class of varieties. He proves the main theorem of [ibid.] for the absolute Hodge motive of any variety X with good reduction belonging to the tannakian category generated by the motives of abelian varieties. Under somewhat more restrictive conditions, Laskar also generalises the results of the present paper, treating the case of curves, K3 surfaces and a Fermat hypersurfaces with semistable reduction.

Another direction for generalisation is the case of 1-motives. In this context, an analogue of the theory of Mumford–Tate liftings remains to be developed. The analogue of our results in the case where the base field is a function field in characteristic p seems inaccessible with the techniques used in this paper. Indeed, it would be necessary to develop a substitute for the theory of absolute Hodge motives and, most importantly, the concept of Mumford–Tate liftings.

#### 1. 1-motives with *L*-action

In this section, we indicate how the theory of 1-motives developed in [Raynaud 1994; Deligne 1974b, §10] works out for 1-motives with a given endomorphism field. Most of the statements are easy generalisations of those given in [Fontaine 1994] for the case where  $L = \mathbb{Q}$ . The results of this section will be applied to the study of the monodromy of an abelian variety with a given endomorphism algebra. Where these preliminary results are concerned, very little additional effort is required to deal with the more general case of 1-motives.

We also review the construction of the Weil–Deligne group of a local field and recall how to associate a representation of the Weil–Deligne group to a local Galois representation.

**1.1.** *Generalities on* 1*-motives.* In all of this section, *F* is a finite extension of  $\mathbb{Q}_p$  and  $\ell$ ,  $\ell'$  are prime numbers.

Let *M* be a 1-motive over *F*. By definition *M* is a complex  $u: Y \to G$  where *Y* is a *K*-group scheme which is locally isomorphic, in the étale topology, to  $\mathbb{Z}^r$  and *G* is a semiabelian variety over *K*. In this complex, *Y* is placed in degree -1 and

*G* in degree 0. The semiabelian variety *G* is an extension of an abelian variety *A* of dimension *g* by a torus *T* of dimension  $r^*$ . Let  $d = d(M) = r + r^* + 2g$ .

One defines the  $\ell$ -adic realisations of M following [Deligne 1974b, 10.1], taking projective limits. This differs by a trivial manipulation from Raynaud's definition [1994, 3.1] where an inductive limit is used. To be precise, for an integer n, we define  $\text{Tors}_n(M)$  as the H<sup>-1</sup> of the complex

$$Y \to Y \oplus G(\bar{F}) \to G(\bar{F}),$$
$$x \mapsto (-nx, -u(x)),$$
$$(x, y) \mapsto u(x) - ny$$

situated in degrees -2, -1, 0. Here and in what follows,  $\overline{F}$  is an algebraic closure of F. For any  $\ell$  we put

$$T_{\ell}(M) = \lim \operatorname{Tors}_{\ell^n}(M)$$
 and  $V_{\ell}(M) = T_{\ell}(M) \otimes_{\mathbb{Z}_{\ell}} \mathbb{Q}_{\ell}.$ 

Thus,  $T_{\ell}(M)$  is a free  $\mathbb{Z}_{\ell}$ -module of rank d and  $V_{\ell}(M)$  is a  $\mathbb{Q}_{\ell}$ -vector space of dimension d. For each  $\ell$ , the absolute Galois group  $\Gamma_F = \operatorname{Aut}_F(\bar{F}) = \operatorname{Gal}(\bar{F}/F)$  acts naturally on  $V_{\ell}$ .

We fix a number field  $L \subset \operatorname{End}^0(M) = \operatorname{End}(M) \otimes_{\mathbb{Z}} \mathbb{Q}$ . The endomorphism ring End(*M*) can be interpreted either as the ring of endomorphisms of the complex  $Y \to G$  or as the ring of endomorphisms of its image in the derived category  $D^b$ (fppf); see [Raynaud 1994, 2.3]. The first interpretation of End(*M*) shows that *L* acts on  $Y \otimes \mathbb{Q}$  and that *L* embeds into End<sup>0</sup>(*G*). It follows (for example) from [Milne 1986, 3.9] that any morphism of the torus *T* to an abelian variety is trivial. This implies that we have an embedding  $L \subset \operatorname{End}^0(T)$  so *L* also acts on  $Y^* \otimes \mathbb{Q}$ where  $Y^* = \operatorname{Hom}(T, \mathbb{G}_m)$ . Finally, by passing to the quotient A = G/T, we obtain an embedding  $L \subset \operatorname{End}^0(A)$ .

By functoriality, *L* acts on  $V_{\ell} = V_{\ell}(M)$ , making it into an  $L \otimes_{\mathbb{Q}} \mathbb{Q}_{\ell}$ -module. To ease notation, we will write  $L_{\ell} = L \otimes_{\mathbb{Q}} \mathbb{Q}_{\ell}$  from now on. The weight filtration of *M* (see [Raynaud 1994, 2.2]) induces an increasing  $L_{\ell}$ -linear filtration of  $V_{\ell}$  such that the nonzero components of the associated graded are

$$Gr_{-2}(V_{\ell}) \cong (Y^{\star})^{\vee} \otimes_{\mathbb{Z}} \mathbb{Q}_{\ell}(1),$$
  

$$Gr_{-1}(V_{\ell}) \cong V_{\ell}(A) = T_{\ell}(A) \otimes_{\mathbb{Z}_{\ell}} \mathbb{Q}_{\ell}$$
  

$$Gr_{0}(V_{\ell}) \cong Y \otimes_{\mathbb{Z}} \mathbb{Q}_{\ell}.$$

The action of  $\Gamma_F$  respects the weight filtration and commutes with the  $L_\ell$ -action so  $\Gamma_F$  acts  $L_\ell$ -linearly on  $\operatorname{Gr}_{\bullet}(V_\ell)$ . The isomorphisms above are  $L_\ell$ -linear and  $\Gamma_F$ -equivariant. One has  $L_\ell \cong \bigoplus_{\lambda} L_{\lambda}$ , where  $\lambda$  runs through the primes of L lying over  $\ell$ . This decomposition gives rise to corresponding decompositions of each  $L_{\ell}$ -module occurring above as a direct sum of  $L_{\lambda}$ -modules of the same rank.

The next aim is to establish a common  $\mathbb{Q}$ -structure V on the modules  $V_{\ell}$ , endowed with *L*-action and weight filtration. We first describe the associated graded. For  $V^0 = Y \otimes_{\mathbb{Z}} \mathbb{Q}$ , there is a system of canonical  $L_{\ell}$ -linear isomorphisms  $\operatorname{Gr}_0(V_{\ell}) \cong V^0 \otimes \mathbb{Q}_{\ell}$ . Next, put  $V^{-2} = (Y^* \otimes_{\mathbb{Z}} \mathbb{Q})^{\vee}$  and fix isomorphisms  $\mathbb{Q}_{\ell}(1) \cong \mathbb{Q}_{\ell}$ for each  $\ell$ . This gives rise to a system of isomorphisms  $\operatorname{Gr}_{-2}(V_{\ell}) \cong V^{-2} \otimes \mathbb{Q}_{\ell}$ , depending only on the identifications  $\mathbb{Q}_{\ell}(1) \cong \mathbb{Q}_{\ell}$ .

We finally consider  $Gr_{-1}$ . By the theorem of the primitive element,  $L = \mathbb{Q}(\alpha)$ , so it follows from [Mumford 1970, §19, Theorem 4] that there exists an *L*-vector space  $V^{-1}$  endowed with  $L_{\ell}$ -linear isomorphisms

$$\operatorname{Gr}_{-1}(V_{\ell}) \cong V^{-1} \otimes_{\mathbb{Q}} \mathbb{Q}_{\ell}$$

for every  $\ell$ ; see [Noot 2006, proof of 6.13]. As a vector space is determined up to isomorphism by its dimension,  $V^{-1}$  is unique up to *L*-linear isomorphisms and each isomorphism  $\operatorname{Gr}_{-1}(V_{\ell}) \cong V^{-1} \otimes_{\mathbb{Q}} \mathbb{Q}_{\ell}$  is unique up to  $L_{\ell}$ -linear automorphisms of  $\operatorname{Gr}_{-1}(V_{\ell})$ .

Define

$$V = V^{-2} \oplus V^{-1} \oplus V^0,$$

endowed with the natural *L*-action and the increasing filtration defined by the grading. As  $L_{\ell}$  is a semisimple algebra,  $V_{\ell}$  is  $L_{\ell}$ -isomorphic to its associated graded, so the preceding discussion gives rise to a noncanonical  $L_{\ell}$ -linear isomorphism  $V_{\ell} \cong V \otimes_{\mathbb{Q}} \mathbb{Q}_{\ell}$  for every  $\ell$ . We proved the following lemma.

**Lemma 1.2.** There exists an *L*-vector space *V* endowed with an increasing filtration and, for each  $\ell$ , an  $L_{\ell}$ -linear isomorphism  $V_{\ell} \cong V \otimes_{\mathbb{Q}} \mathbb{Q}_{\ell}$  compatible with the filtrations. In particular, each  $V_{\ell}$  is a free  $L_{\ell}$ -module whose rank is independent of  $\ell$  and the weight filtration is a filtration by free  $L_{\ell}$ -submodules of ranks independent of  $\ell$ .

**1.3.** *The group H*. We fix an *L*-vector space *V* together with a system of isomorphisms as in the lemma. Let  $H = \operatorname{Res}_{L/\mathbb{Q}} \operatorname{GL}_{/L}(V)$  be the linear algebraic group over  $\mathbb{Q}$  of *L*-linear automorphisms of *V* and let  $\mathfrak{h}$  be its Lie algebra. This means that  $\mathfrak{h} = \mathfrak{gl}_L(V)$  is the  $\mathbb{Q}$ -Lie algebra of *L*-linear endomorphisms of *V*. Obviously, *H* acts on  $\mathfrak{h}$  through the adjoint representation. By means of the identifications above,  $H_{/\mathbb{Q}_{\ell}}$  and  $\mathfrak{h} \otimes_{\mathbb{Q}} \mathbb{Q}_{\ell}$  act on  $V_{\ell}$ . This identifies  $H_{/\mathbb{Q}_{\ell}}$  with the group of  $L_{\ell}$ -linear automorphisms of  $V_{\ell}$  and  $\mathfrak{h} \otimes \mathbb{Q}_{\ell}$  with its Lie algebra. These identifications are determined up to inner automorphisms for the groups and up to the adjoint action of  $H_{/\mathbb{Q}_{\ell}}$  where the Lie algebras are concerned. Finally note that  $L = \operatorname{End}_H(V)$  and that  $L_{\ell} = \operatorname{End}_{H/\mathbb{Q}_{\ell}}(V_{\ell})$ .

**1.4.** *The monodromy operator.* From now on we assume that the 1-motive *M* is *strict* in the sense of [Raynaud 1994, Définition 4.2.3], which means that the semiabelian variety *G* has potentially good reduction. In this case, [ibid., 4.3] defines the *geometric monodromy*, a canonical additive map  $\mu: Y \otimes Y^* \to \mathbb{Q}$ . Giving  $\mu$  is equivalent to giving the induced map

$$N: V^{0} = Y \otimes \mathbb{Q} \to V^{-2} = (Y^{\star} \otimes \mathbb{Q})^{\vee}.$$
(1.4\*)

By functoriality this map is *L*-linear, so we can interpret *N* as an element of the Lie algebra  $\mathfrak{h}^{ss} \subset \text{End}(V)$ . As an endomorphism of *V*, it is nilpotent of echelon 2. In what follows, the notation *N* will be reserved for this element of  $\mathfrak{h}^{ss}$ .

The map N defines a morphism  $\operatorname{Gr}_0(V_\ell) \to \operatorname{Gr}_2(V_\ell)(-1)$  and thus a morphism  $N_\ell \colon V_\ell \to V_\ell(-1)$ . As  $N_\ell$  is  $L_\ell$ -linear,  $N_\ell \in \mathfrak{h}^{ss} \otimes \mathbb{Q}_\ell(-1)$  for each  $\ell$ . Recall that the identification  $V_\ell \cong V \otimes \mathbb{Q}_\ell$  depends on the identification  $\mathbb{Q}_\ell(1) \cong \mathbb{Q}_\ell$  fixed in Section 1.1. Using the same identification, we identify  $\mathfrak{h}^{ss} \otimes \mathbb{Q}_\ell(-1)$  with  $\mathfrak{h}^{ss} \otimes \mathbb{Q}_\ell$  and under these isomorphisms the images of  $N \otimes 1$  and  $N_\ell$  in  $\mathfrak{h}^{ss} \otimes \mathbb{Q}_\ell$  coincide.

The discussion above only depends on the isomorphisms  $\operatorname{Gr}_i(V_\ell) \cong V^i \otimes \mathbb{Q}_\ell$  for i = -2, 0, which in turn depend only on the choice of an identification  $\mathbb{Q}_\ell(1) \cong \mathbb{Q}_\ell$ . In what follows we may thus change the splitting of the weight filtration and the identification  $\operatorname{Gr}_{-1}(V_\ell) \cong V^{-1} \otimes \mathbb{Q}_\ell$  without affecting the properties above. We have established the following proposition.

**Proposition 1.5.** Let notation and assumptions be as above, in particular the motive M is assumed to be strict and  $N_{\ell} \colon V_{\ell} \to V_{\ell}(-1)$  is the  $\ell$ -adic monodromy operator. For each  $\ell$ , fix an identification of  $H_{/\mathbb{Q}_{\ell}}$  with the group of  $L_{\ell}$ -linear automorphisms of  $V_{\ell}$  and an isomorphism  $\mathbb{Q}_{\ell} \cong \mathbb{Q}_{\ell}(1)$ . Using these identifications we consider  $N_{\ell}$  as an element of  $\mathfrak{h} \otimes \mathbb{Q}_{\ell}$ .

- For every algebraically closed field  $\Omega \supset \mathbb{Q}_{\ell}$  and every  $\sigma \in \operatorname{Aut}(\Omega)$ , the image of  $N_{\ell}$  in  $(\mathfrak{h} \otimes_{\mathbb{Q}} \mathbb{Q}_{\ell}) \otimes_{\mathbb{Q}_{\ell}} \Omega = \mathfrak{h} \otimes \Omega$  is conjugate to  $\sigma(N_{\ell})$  under the adjoint action of  $H(\Omega)$ .
- If Ω is an algebraically closed field containing both Q<sub>ℓ</sub> and Q<sub>ℓ'</sub> then the images in h ⊗ Ω of N<sub>ℓ</sub> and N<sub>ℓ'</sub> are H(Ω)-conjugate.

**1.6.** *The action of inertia.* The  $\Gamma_F$ -action on each  $V_\ell$  is  $L_\ell$ -linear so the realisations of *M* give rise to a system of representations

$$\rho_{\ell} \colon \Gamma_F \to H(\mathbb{Q}_{\ell})$$

using the identifications from Section 1.1 and the group H from Section 1.3.

Following [Deligne 1973, §2] for the basic notation, we discuss the action of the inertia group  $I_F \subset \Gamma_F$ . Let v be the valuation of F with value group  $\mathbb{Z}$  and let k be the (finite) residue field. We will write  $\bar{v}$  for the valuation on  $\bar{F}$  extending v and write  $\bar{k}$  for the residue field of  $\bar{F}$ .

Let

$$\widehat{\mathbb{Z}}_{\neq p} = \lim_{p \nmid n} \mathbb{Z}/n\mathbb{Z} = \prod_{\ell \neq p} \mathbb{Z}_{\ell}$$

be the *p*-primary part of  $\widehat{\mathbb{Z}}$  and let  $\mathbb{A}_{\neq p} = \widehat{\mathbb{Z}}_{\neq p} \otimes_{\mathbb{Z}} \mathbb{Q}$ . The prime-to-*p* part of  $\mathbb{Q}/\mathbb{Z}$  is  $\mathbb{A}_{\neq p}/\widehat{\mathbb{Z}}_{\neq p} = (\mathbb{Q}/\mathbb{Z})_{\neq p}$ . For every *n* with  $p \nmid n$ , we identify  $(\frac{1}{n}\mathbb{Z}/\mathbb{Z})(1)$  with the group of *n*-th roots of unity in  $\bar{k}$ . This identifies  $(\mathbb{Q}/\mathbb{Z})_{\neq p}(1)$  with the multiplicative group  $\bar{k}^{\times}$ . There is a natural morphism  $t: I_F \to \widehat{\mathbb{Z}}_{\neq p}(1)$  such that  $\sigma(x)x^{-1} = [t(\sigma)\bar{v}(x)]$  for all  $\sigma \in I_F$  and  $x \in \bar{F}^{\times}$ . Here  $[t(\sigma)\bar{v}(x)]$  is the image in  $\bar{k}^{\times} \cong (\mathbb{Q}/\mathbb{Z})_{\neq p}(1)$  of  $t(\sigma)\bar{v}(x) \in \mathbb{A}_{\neq p}(1)$ . For  $\ell \neq p$ , we write  $t_{\ell}: I_F \to \mathbb{Z}_{\ell}(1)$  for the composite of *t* with the projection  $\widehat{\mathbb{Z}}_{\neq p}(1) \to \mathbb{Z}_{\ell}(1)$ .

In the case where *Y* and *G* have good reduction, it follows from [Raynaud 1994, Proposition 4.6.1] that if  $\ell \neq p$  then for each  $\sigma \in I_F$  one has

$$\rho_{\ell}(\sigma) = \exp(N_{\ell} \otimes t_{\ell}(\sigma)), \qquad (1.6*)$$

where  $N_{\ell} \in \mathfrak{h}^{ss} \otimes \mathbb{Q}_{\ell}(-1)$  is the  $\ell$ -adic monodromy operator defined in Section 1.4. For an arbitrary strict 1-motive, the equality above holds for all  $\sigma$  in a sufficiently small open subgroup of  $I_F$ . We finally note that (1.6\*) characterises the operator  $N_{\ell}$  as a map  $N_{\ell}: V_{\ell} \to V_{\ell}(-1)$ . This will play an important role in Section 2.2, in particular in the formula (2.2\*).

**1.7.** *Characteristic polynomials.* Write q = |k| and let  $\varphi$  be the arithmetic Frobenius automorphism  $\varphi : x \mapsto x^q$  of  $\bar{k}$  over k. The *Weil group* of F is the subgroup of  $\Gamma_F$  consisting of the elements  $\psi$  inducing an integral power of  $\varphi^{\alpha(\psi)}$  of  $\varphi$ . The map  $\alpha : W_F \to \mathbb{Z}$  thus defined is a group homomorphism and its kernel is the inertia group  $I_F \subset \Gamma_F$ . We endow the Weil group with the topology determined by the condition that  $I_F \subset W_F$  is an open subgroup carrying the topology inherited from its topology as a Galois group.

For a 1-motive M/F with L-action as before, k = -2, -1 or 0 and  $\psi \in W_F$ , let

$$P_{L_{\ell},\psi}^{(k)} \in L_{\ell}[T]$$

be the characteristic polynomial of  $\rho_{\ell}(\psi)$  acting as an  $L_{\ell}$ -linear endomorphism on the free  $L_{\ell}$ -module  $\operatorname{Gr}_k(V_{\ell})$ . Let  $P_{L_{\ell},\psi}$  be the characteristic polynomial of  $\rho_{\ell}(\psi)$ acting  $L_{\ell}$ -linearly on  $V_{\ell}$ . Obviously, one has

$$P_{L_{\ell},\psi} = \prod_{k=-2}^{0} P_{L_{\ell},\psi}^{(k)}$$

**Proposition 1.8.** Let notation and hypotheses be as above; in particular M is assumed to be strict. Let  $\ell$  run though the primes different from p. Then for each k = -2, -1, 0 and any  $\psi \in W_F$ , we have  $P_{L_\ell,\psi}^{(k)} \in L[T]$ . For fixed k and  $\psi$ , the

polynomial  $P_{L_{\ell},\psi}^{(k)}$  is independent of  $\ell$  and all its complex roots have absolute value  $q^{-\alpha(\psi)k/2}$ . The polynomial  $P_{L_{\ell},\psi}$  belongs to L[T] and is independent of  $\ell$ .

*Proof.* It is sufficient to prove the statements concerning the  $P_{L_{\ell},\psi}^{(k)}$ . For k = -2, 0, these follow from the  $\Gamma_F$ -equivariant isomorphisms  $\operatorname{Gr}_{-2}(V_{\ell}) \cong (Y^*)^{\vee} \otimes \mathbb{Q}_{\ell}(1)$  and  $\operatorname{Gr}_0(V_{\ell}) \cong Y \otimes \mathbb{Q}_{\ell}$  and the fact that  $\Gamma_F$  acts on Y and on  $Y^*$  through finite quotients.

For k = -1 we have  $\operatorname{Gr}_{-1}(V_{\ell}) \cong V_{\ell}A$ , where *A* is an abelian variety with  $L \subset \operatorname{End}^{0}(A)$ . The statement about the absolute values of the roots of  $P_{L_{\ell},\psi}^{(-1)}$  therefore follows from the corollary to Theorem 3 in [Serre and Tate 1968]; see also [Raynaud 1994, 4.7.4]. Under the assumption that *A* has good reduction and that  $\psi$  is a Frobenius element, a proof of the claims that  $P_{L_{\ell},\psi}^{(-1)} \in L[T]$  and that this polynomial is independent of  $\ell$  is sketched in [Noot 2009, 2.1]. Taking into account [Serre and Tate 1968, Theorem 2], the argument remains valid when *A* only has potentially good reduction and  $\psi \in I_F$ . For the case where  $\psi$  reduces to a nontrivial power of the Frobenius element, one replaces the use of [Serre and Tate 1968, Theorem 2] by the corollary to Theorem 3 in the same paper.

**1.9.** *Remark.* The action of any  $\psi \in W_F$  on  $\operatorname{Gr}_k(V_\ell)$  is semisimple for any k. For k = -2, 0 this results from the fact that  $\Gamma_F$  acts on  $\operatorname{Gr}_0(V_\ell)$  and on  $\operatorname{Gr}_{-2}(V_\ell)(-1)$  through a finite quotient. For k = -1 it follows from the fact that  $\operatorname{Gr}_{-1}(V_\ell) \cong V_\ell A$ , where A is an abelian variety over F with potentially good reduction. Combining this statement with the Proposition 1.8, it follows that each  $\psi \in W_F$  with  $\alpha(\psi) \neq 0$  acts semisimply on  $V_\ell$ .

**1.10.** *Frobenius weights.* As before, *M* is a strict 1-motive over *F* with *L*-action. We fix an arithmetic Frobenius element  $\Phi \in \Gamma_F$ , that is, a lifting of the Frobenius automorphism  $\varphi$  of  $\bar{k}$ ; see Section 1.7. The operator  $N_{\ell} : V_{\ell} \to V_{\ell}(-1)$  defined in Section 1.4 is  $\Gamma_F$ -equivariant, which implies that  $\operatorname{Ad}(\rho_{\ell}(\Phi))(N_{\ell}) = qN_{\ell}$ .

As noted in Section 1.9, the image  $\rho_{\ell}(\Phi)$  is semisimple and by Proposition 1.8 its eigenvalues are algebraic integers and, for any eigenvalue, all complex absolute values coincide and are equal to q,  $q^{1/2}$  or 1. For k = -2, -1, 0, let  $V_{\ell}^k \otimes \overline{\mathbb{Q}}_{\ell}$ be the sum of the eigenspaces associated to the eigenvalues with absolute value  $q^{-k/2}$ . This defines a splitting  $V_{\ell} = V_{\ell}^{-2} \oplus V_{\ell}^{-1} \oplus V_{\ell}^{0}$  of the weight filtration. The *Frobenius weight cocharacter*  $w_{\ell} : \mathbb{G}_{m/\mathbb{Q}_{\ell}} \to H_{/\mathbb{Q}_{\ell}}$  is the morphism making  $\mathbb{G}_{m/\mathbb{Q}_{\ell}}$ act on  $V_{\ell}^k$  through the (k + 1)-st power map. The reader should take note of the shift in filtration, which is introduced to simplify matters later on. Through the adjoint representation,  $w_{\ell}(t)$  acts on the line in  $\mathfrak{h}^{ss} \otimes \mathbb{Q}_{\ell}(-1)$  generated by  $N_{\ell}$  as multiplication by  $t^{-2}$ .

If M/F is any, not necessarily strict, 1-motive with L-action, then the complex absolute values of any eigenvalue of  $\rho_{\ell}(\Phi)$  are still equal to q,  $q^{1/2}$  or 1. This

follows from the existence [Raynaud 1994, 4.2.2] of a strict 1-motive M' with *L*-action endowed with a system of canonical isomorphisms  $V_{\ell}(M) \cong V_{\ell}(M')$ . The Frobenius weight cocharacter  $w_{\ell}$  of M can therefore be defined exactly as before. It corresponds to the cocharacter associated to M' by transport via the isomorphism  $V_{\ell}(M) \cong V_{\ell}(M')$  above. In general,  $w_{\ell}$  does not split the weight filtration.

Finally note that, for M strict, the identification  $V_{\ell} \cong V \otimes \mathbb{Q}_{\ell}$  can be modified, without affecting its previously established properties, to ensure that the grading on  $V_{\ell}$  defined by the Frobenius weights corresponds to the grading on V.

#### 2. The representations of the Weil–Deligne group associated to a 1-motive

**2.1.** *The Weil–Deligne group.* In addition to the conventions in Section 1.1, we will from now on assume that  $\ell$ ,  $\ell' \neq p$ . As in Section 1.7,  $W_F$  is the Weil group of *F*. We briefly summarise some of the notions introduced in [Deligne 1973, §8]; see also [Fontaine 1994].

Letting  $\psi \in W_F$  operate on the additive group  $\mathbb{G}_{a/\mathbb{Q}}$  as multiplication by  $q^{\alpha(\psi)}$ , one defines an action of the constant topological group scheme  $W_F$  on  $\mathbb{G}_{a/\mathbb{Q}}$ . The Weil–Deligne group of F is the semidirect product

$$W_F = W_F \ltimes \mathbb{G}_a$$

defined by this action, viewed as a group scheme over  $\mathbb{Q}$ .

Fix an identification  $\mathbb{Q}_{\ell} \cong \mathbb{Q}_{\ell}(1)$  as in Section 1.1, an arithmetic Frobenius element  $\Phi \in W_F$  as in Section 1.10 and consider the map  $t_{\ell}$  from Section 1.6 as a morphism

$$I_F \to \mathbb{Q}_\ell(1) \cong \mathbb{Q}_\ell = \mathbb{G}_a(\mathbb{Q}_\ell).$$

We define a system of  $\ell$ -adic representations of  $W_F$  with values in  $W_F(\mathbb{Q}_\ell)$  by

$$\psi \mapsto \left(\psi, t_{\ell}(\Phi^{-\alpha(\psi)}\psi)\right) \in (W_F \ltimes \mathbb{G}_a)(\mathbb{Q}_{\ell}).$$

For a field *E* of characteristic 0 and a linear algebraic group  $G_{/E}$  over *E*, giving an algebraic representation  $(W_F)_{/E} \rightarrow G_{/E}$  is equivalent to giving a pair  $(\rho', N)$ where  $\rho': W_F \rightarrow G_{/E}(E)$  is a linear representation that is trivial on some open subgroup of  $I_F$  and  $N \in \text{Lie}(G_{/E})$  is a nilpotent element satisfying the condition that

$$\operatorname{Ad}(\rho'(\psi))N = q^{\alpha(\psi)}N \tag{2.1*}$$

for all  $\psi \in W_F$ . The representation of  $(W_F)_{/E}$  corresponding to the pair  $(\rho', N)$  is given by  $(\psi, x) \mapsto \rho'(\psi) \exp(Nx)$ .

**2.2.**  ${}^{\prime}W_F$  and  $\ell$ -adic Galois representations. Let  $H_{/\mathbb{Q}_{\ell}}$  be a  $\mathbb{Q}_{\ell}$ -linear algebraic group and  $\rho_{\ell} \colon W_F \to H_{/\mathbb{Q}_{\ell}}(\mathbb{Q}_{\ell})$  a continuous representation. By Grothendieck's

 $\ell$ -adic monodromy theorem (see [Deligne 1973, 8.2]), there exists a nilpotent element  $N'_{\ell} \in \mathfrak{h}^{ss}(-1) = \text{Lie}(H_{/\mathbb{Q}_{\ell}})^{ss}(-1)$  such that

$$\rho_{\ell}(\psi) = \exp(N'_{\ell} t_{\ell}(\psi)) \tag{2.2*}$$

for all  $\psi$  in a sufficiently small open subgroup of  $I_F$ ; see (1.6\*). One can therefore associate to  $\rho_\ell$  a representation  $(\rho'_\ell, N'_\ell)$  of  $W_F$  with values in  $H_{/\mathbb{Q}_\ell}$  as follows.

Using the identification  $\mathbb{Q}_{\ell} \cong \mathbb{Q}_{\ell}(1)$  to interpret  $N'_{\ell}$  as an element of  $\mathfrak{h}^{ss}$ , one defines

$$\rho_{\ell}'(\psi) = \rho_{\ell}(\psi) \exp\left(-N_{\ell}' t_{\ell}(\Phi^{-\alpha(\psi)}\psi)\right).$$

Composing the corresponding algebraic representation of  ${}^{\prime}W_F$  with the natural representation  $W_F \rightarrow {}^{\prime}W_F(\mathbb{Q}_\ell)$  defined above, one recovers  $\rho_\ell$ .

According to [Deligne 1973, 8.11], the geometric conjugacy class of  $(\rho'_{\ell}, N'_{\ell})$  is independent of the choices of  $\Phi$  and of the identification  $\mathbb{Q}_{\ell} \cong \mathbb{Q}_{\ell}(1)$  made in this construction.

**2.3.** Compatible systems of representations of  ${}^{\prime}W_{F}$ . Let H be a reductive algebraic group over  $\mathbb{Q}$ . For a fixed  $\ell$ , we say that a representation  ${}^{\prime}W_{F/\mathbb{Q}_{\ell}} \to H_{\mathbb{Q}_{\ell}}$  is *defined over*  $\mathbb{Q}$  (*as a representation with values in* H) if for every algebraically closed field  $\Omega \supset \mathbb{Q}_{\ell}$ , the base extension  ${}^{\prime}W_{F/\Omega} \to H_{/\Omega}$  is conjugate under  $H(\Omega)$  to all its images under  $\operatorname{Aut}_{\mathbb{Q}}(\Omega)$ . In terms of the pair  $(\rho_{\ell}', N_{\ell}')$ , let

$$\rho'_{\ell} \otimes_{\mathbb{Q}_{\ell}} \Omega \colon W_F \to H_{/\Omega}(\Omega)$$

be the extension of scalars and let  $N'_{\ell} \otimes_{\mathbb{Q}_{\ell}} 1 \in (\mathfrak{h} \otimes_{\mathbb{Q}} \mathbb{Q}_{\ell}) \otimes_{\mathbb{Q}_{\ell}} \Omega = \mathfrak{h} \otimes \Omega$  be the image of  $N'_{\ell}$ . Then the condition above is equivalent to the condition that for every  $\sigma \in \operatorname{Aut}_{\mathbb{Q}}(\Omega)$  there is an element  $g \in H(\Omega)$  such that

$$\sigma(\rho_{\ell}' \otimes_{\mathbb{Q}_{\ell}} \Omega) = g(\rho_{\ell}' \otimes_{\mathbb{Q}_{\ell}} \Omega) g^{-1} \quad \text{and} \quad \sigma(N_{\ell}' \otimes_{\mathbb{Q}_{\ell}} 1) = \operatorname{Ad}(g)(N_{\ell}' \otimes_{\mathbb{Q}_{\ell}} 1). \quad (2.3*)$$

We say that a family of representations  ${}^{'}W_{F/\mathbb{Q}_{\ell}} \rightarrow H_{/\mathbb{Q}_{\ell}}$  is a *compatible system* of representations of  ${}^{'}W_{F}$  (with values in H) if for every pair  $(\ell, \ell')$  and every algebraically closed field  $\Omega$  containing  $\mathbb{Q}_{\ell}$  and  $\mathbb{Q}_{\ell'}$ , the base extensions to  $\Omega$  of the  $\ell$ -adic and  $\ell'$ -adic representations of  ${}^{'}W_{F}$  are  $H(\Omega)$ -conjugate. In terms of the pairs  $(\rho'_{\ell}, N'_{\ell})$  and  $(\rho'_{\ell'}, N'_{\ell'})$ , this means that there is a  $g \in H(\Omega)$  such that

$$\rho_{\ell}^{\prime} \otimes_{\mathbb{Q}_{\ell}} \Omega = g(\rho_{\ell^{\prime}}^{\prime} \otimes_{\mathbb{Q}_{\ell^{\prime}}} \Omega) g^{-1} \quad \text{and} \quad N_{\ell}^{\prime} \otimes_{\mathbb{Q}_{\ell}} 1 = \operatorname{Ad}(g)(N_{\ell^{\prime}}^{\prime} \otimes_{\mathbb{Q}_{\ell^{\prime}}} 1) \in \mathfrak{h} \otimes \Omega.$$
 (2.3<sup>†</sup>)

The action of  $H(\Omega)$  by conjugation factors through  $H(\Omega) \to H^{ad}(\Omega)$  so  $H(\Omega)$ conjugacy may be replaced by  $H^{ad}(\Omega)$ -conjugacy everywhere.

**2.4.** Application to 1-motives. We apply the discussion above to the system of  $\ell$ -adic representations  $V_{\ell}(M)$  associated to a 1-motive M with L action. Let M be as in Section 1.1 and, as was the case from Section 1.4 onward, continue to

assume *M* to be strict. The numbers  $r = \operatorname{rank}(Y)$ ,  $r^* = \dim(T)$  and  $g = \dim(A)$ are as in Section 1.1 and we fix an *L*-vector space *V* and a system of identifications  $V_{\ell} \cong V \otimes \mathbb{Q}_{\ell}$  as in Lemma 1.2. Let the algebraic group  $H = \operatorname{Res}_{L/\mathbb{Q}} \operatorname{GL}_{/L}(V)$ be as in Section 1.3. For every  $\ell$ , the group  $H_{/\mathbb{Q}_{\ell}}$  identifies with the group of  $L_{\ell} = L \otimes \mathbb{Q}_{\ell}$ -linear endomorphisms of  $V_{\ell}$ . The action of  $\Gamma_F$  on  $V_{\ell}(M)$  is  $L_{\ell}$ -linear, so it provides us with an  $\ell$ -adic representation of ' $W_F$  with values in  $H_{/\mathbb{Q}_{\ell}}$ , that is, a system of pairs  $(\rho'_{\ell}, N'_{\ell})$ , where  $N'_{\ell} \in \mathfrak{h} \otimes \mathbb{Q}_{\ell} = \operatorname{Lie}(H) \otimes \mathbb{Q}_{\ell}$  and  $\rho'_{\ell} \colon W_F \to H_{/\mathbb{Q}_{\ell}}$ .

**Lemma 2.5.** Let M/F be any 1-motive with L action. The Frobenius weight cocharacter  $w_{\ell}$  commutes with the representation  $\rho'_{\ell}$ .

*Proof.* By construction,  $\rho_{\ell}(\Phi) = \rho'_{\ell}(\Phi)$  and the same equality holds for all powers of  $\Phi$ . In the construction of the Frobenius weight cocharacter, one may replace the Frobenius element  $\Phi$ , and hence q, by any strictly positive power without modifying  $w_{\ell}$ . This means that it is sufficient to prove that there is a strictly positive power  $\Phi^n$  such that  $\rho'_{\ell}(\Phi^n)$  lies in the centre of the image of  $\rho'_{\ell}$ . This is obvious since  $\rho'_{\ell}$  factors through an extension of the group generated by  $\Phi$  by a finite quotient of  $I_F$ .

**Proposition 2.6.** Assume that we are in the situation of Section 2.4, so in particular M is strict. Each  $\ell$ -adic representation

$$W_{F/\mathbb{Q}_{\ell}} \to H_{/\mathbb{Q}_{\ell}}$$

is defined over  $\mathbb{Q}$  and these representations form a compatible system of representations of  $W_F$  with values in H.

*Proof.* We first show that each representation is defined over  $\mathbb{Q}$ .

The operator  $N'_{\ell}$  is determined by the fact that it satisfies (2.2\*) for all  $\psi$  in a sufficiently small open subgroup of the inertia group  $I_F$ . The equality (1.6\*) implies that the monodromy operator  $N_{\ell}$  has the same property so we conclude that  $N'_{\ell} = N_{\ell}$ .

It follows from Proposition 1.5 that for every  $\Omega \supset \mathbb{Q}_{\ell}$  and  $\sigma \in \operatorname{Aut}(\Omega)$  as in Section 2.3,  $N_{\ell} \in \mathfrak{h} \otimes \mathbb{Q}_{\ell}$  is  $H(\Omega)$ -conjugate to  $\sigma(N_{\ell})$ . Let  $g \in H(\Omega)$  be such that  $\sigma(N_{\ell}) = \operatorname{Ad}(g)(N_{\ell})$ . It is sufficient to show that  $\rho'_{\ell} \otimes \Omega$  and  $g^{-1}\sigma(\rho'_{\ell} \otimes \Omega)g$  are conjugate under the stabiliser of  $N_{\ell}$  in  $H(\Omega)$ . By elementary representation theory (see [Deligne 1973, Proposition 8.9]), it suffices to show that the representation  $\rho'_{\ell}$ is semisimple and that, for every  $\psi \in W_F$ , the  $L_{\ell}$ -linear characteristic polynomials of  $\rho'_{\ell}(\psi)$  and of  $g^{-1}\sigma(\rho'_{\ell} \otimes \Omega)g$  acting on each  $\operatorname{Gr}_i^{\operatorname{mon}}(V_{\ell})$  coincide. In fact, it is sufficient to prove that the traces coincide. Here  $\operatorname{Gr}_i^{\operatorname{mon}}(V_{\ell})$  is the associated graded of  $V_{\ell}$  for the monodromy filtration defined by  $N_{\ell}$ .

We first treat the semisimplicity. The restriction  $\rho'_{\ell}|_{I_F}$  is semisimple because it factors through a finite quotient of  $I_F$ . As the action of  $W_F$  on each  $\operatorname{Gr}_i(V_{\ell})$ is semisimple, the semisimplicity of  $\rho'_{\ell}$  results from Section 1.9 applied to any  $\psi \in W_F$  with  $\alpha(\psi) \neq 0$  and the fact that  $W_F$  is an extension of the group generated by the Frobenius element to  $\Phi$  by the inertia group  $I_F$ .

To establish the putative equality of the characteristic polynomials we will prove that the  $L_{\ell}$ -linear characteristic polynomials of the  $\rho'_{\ell}(\psi)$  acting on the  $\operatorname{Gr}_{i}^{\mathrm{mon}}(V_{\ell})$ lie in L[T]. By Proposition 1.8 the corresponding statement is true for the action of  $W_{F}$  on the  $\operatorname{Gr}_{i}(V_{\ell})$ , the associated graded for the weight filtration. We finish the argument by passing to the graded for the monodromy filtration.

This is accomplished by considering the filtration

 $W_{-2}^{\mathrm{mon}}V_{\ell} = \mathrm{im}(N_{\ell}) \subset W_{-2}V_{\ell} \subset W_{-1}V_{\ell} \subset W_{-1}^{\mathrm{mon}}V_{\ell} = \mathrm{ker}(N_{\ell}) \subset W_{0}V_{\ell} = V_{\ell}.$ 

The isomorphism  $\operatorname{Gr}_0(V_\ell) \cong V^0 \otimes \mathbb{Q}_\ell$  is  $\Gamma_F$ -equivariant and the action of  $\Gamma_F$  on  $V^0 \otimes \mathbb{Q}_\ell$  comes from its *L*-linear action on  $V^0$ . Similarly, the action of  $\Gamma_F$  on  $\operatorname{Gr}_{-2}(V_\ell) \cong V^{-2} \otimes \mathbb{Q}_\ell(1)$  comes from its *L*-linear action on  $V^{-2}$  and the cyclotomic action on  $\mathbb{Q}_\ell(1)$ . Finally,  $N_\ell$  comes from the *L*-linear map  $N : V^0 \to V^{-2}$  so by (2.1\*), the kernel of *N* in  $V^0$  and the image of *N* in  $V^{-2}$  are  $W_F$ -invariant *L*-linear subspaces. The representation induced by  $\rho'_\ell$  on each of the spaces

$$\operatorname{Gr}_{-2}^{\operatorname{mon}} V_{\ell} = \operatorname{im}(N_{\ell}) \subset \operatorname{Gr}_{-2} V_{\ell}, \qquad W_{-2}V_{\ell}/\operatorname{im}(N_{\ell}) \cong \operatorname{Gr}_{-2} V_{\ell}/\operatorname{Gr}_{-2}^{\operatorname{mon}} V_{\ell},$$
  
ker $(N_{\ell})/W_{-1}V_{\ell} \subset \operatorname{Gr}_{0} V_{\ell}, \quad \text{and} \quad \operatorname{Gr}_{0}^{\operatorname{mon}} V_{\ell} = V_{\ell}/\operatorname{ker}(N_{\ell}),$ 

therefore, is a base extension of a representation of  $W_F$  on an *L*-vector space. This proves the claim for  $\operatorname{Gr}_{-2}^{\text{mon}} V_{\ell}$  and  $\operatorname{Gr}_{0}^{\text{mon}} V_{\ell}$ . For  $\operatorname{Gr}_{-1}^{\text{mon}} V_{\ell}$  the claim follows similarly by considering the graded for the filtration

$$W_{-2}V_{\ell}/\operatorname{im}(N_{\ell}) \subset W_{-1}V_{\ell}/\operatorname{im}(N_{\ell}) \subset \operatorname{ker}(N_{\ell})/\operatorname{im}(N_{\ell}) = \operatorname{Gr}_{-1}^{\operatorname{mon}}V_{\ell}$$

The fact that the representations  ${}^{\prime}W_{F/\mathbb{Q}_{\ell}} \rightarrow H_{/\mathbb{Q}_{\ell}}$  form a compatible system is proved by an analogous argument. One now has to prove that, for i = -2, -1, 0and for each  $\psi \in W_F$ , the  $L_{\ell}$ -linear characteristic polynomials of  $\rho'_{\ell}(\psi)$  acting on  $\operatorname{Gr}_i^{\operatorname{mon}}(V_{\ell})$  are independent of  $\ell$ . Again by Proposition 1.8, this is true for the characteristic polynomials on the  $\operatorname{Gr}_i(V_{\ell})$ . The  $\ell$  independence of the characteristic polynomials on the  $\operatorname{Gr}_i^{\operatorname{mon}}(V_{\ell})$  follows from this by considering the combined filtration and adapting the argument above.

**Corollary 2.7.** Let M be any 1-motive over F with L-action. Then for each  $\ell$ , the  $\ell$ -adic realisation  $V_{\ell}(M)$  is a free  $L_{\ell}$ -module. For  $\ell \neq p$ , the representations  ${}^{'}W_{F/\mathbb{Q}_{\ell}} \rightarrow H_{/\mathbb{Q}_{\ell}}$  are defined over  $\mathbb{Q}$  and form a compatible system of representations of  ${}^{'}W_{F}$  with values in H.

*Proof.* By Lemma 1.2, each  $V_{\ell}(M)$  is a free  $L_{\ell}$ -module. This implies that  $H_{/\mathbb{Q}_{\ell}}$  identifies with the group of  $L_{\ell}$ -linear endomorphisms of  $V_{\ell}(M)$  and that any two such identifications differ by an inner automorphism of  $H_{/\mathbb{Q}_{\ell}}$ . It is therefore sufficient to prove the second statement for one system of such identifications.

By [Raynaud 1994, 4.2.2] there are a strict 1-motive M' over F and a system of canonical isomorphisms  $V_{\ell}(M) \cong V_{\ell}(M')$  (for every  $\ell$ ). Using these identifications and the remark above, the corollary follows from Proposition 2.6.

**Corollary 2.8.** With the notation and hypotheses of Corollary 2.7,  $\ker(\rho'_{\ell}) \subset W_F$  is independent of  $\ell$ .

#### 3. Application to abelian varieties and statement of the main theorem

We turn our attention to an abelian variety A over a number field  $F \subset \mathbb{C}$ . If F is sufficiently big, each  $\ell$ -adic representation associated to A factors through  $\rho_{\ell} \colon \Gamma_F \to G_A(\mathbb{Q}_{\ell})$ , where  $G_A$  is the Mumford–Tate group of  $A_{/\mathbb{C}}$ . For a fixed valuation v of F, the construction sketched in Section 2.2 gives rise to a system of  $\ell$ -adic representations  $W_{F_v/\mathbb{Q}_{\ell}} \to G_{A/\mathbb{Q}_{\ell}}$  of the Weil–Deligne group of  $F_v$ . It is hoped that these representations are defined over  $\mathbb{Q}$  and that they form a compatible system of representations with values in  $G_A$ .

The statement of the main theorem is somewhat weaker; loosely speaking, it states that, after a finite extension of F, the representations of  ${}^{\prime}W_{F_v}$  form a compatible system when  $G_A$  is replaced by a larger group of which  $G_A$  is the identity component. As the construction will show, only certain factors of  $G^{der}$  of type D are affected by this modification. In order to formulate the precise statement we need a number of constructions from the previous paper [Noot 2009].

**3.1.** *Notation.* From now on,  $F \subset \mathbb{C}$  is a number field and A/F an abelian variety. Let  $\overline{F}$  be the algebraic closure of F in  $\mathbb{C}$  and  $\Gamma_F = \text{Gal}(\overline{F}/F)$  the absolute Galois group. We fix a valuation  $\overline{v}$  of  $\overline{F}$  and let v be its restriction to F. Let p be the residue characteristic of v and  $F_v$  the completion of F at v. It is a finite extension of  $\mathbb{Q}_p$ . Let  $\ell, \ell' \neq p$  prime numbers.

**3.2.** *Abelian varieties.* Betti cohomology defines a fibre functor  $H_B = H_B^1$  on the category of absolute Hodge motives generated by the motive of A and the Tate motive  $\mathbb{Q}(1)$ . The Mumford–Tate  $G_A$  of A is the group of  $\otimes$ -automorphisms of this fibre functor; see [Noot 2009, 1.2] for a more detailed explanation. We will assume throughout that  $G_A$  is connected, this condition holds after replacing F by a finite extension and it implies that  $G_A$  is the smallest linear algebraic  $\mathbb{Q}$ -group such that the Hodge structure on  $H_B^1(A(\mathbb{C}), \mathbb{Q})$  is defined by a morphism  $S = \operatorname{Res}_{\mathbb{C}/\mathbb{R}} \mathbb{G}_m \to G_{A/\mathbb{R}}$ ; see [ibid., 1.2]. Let  $\mathfrak{g}_A$  be the Lie algebra of  $G_A$ .

For every  $\ell$ , the fibre functor  $H_{\acute{e}t,\ell}$  defined by the  $\ell$ -adic étale cohomology is canonically isomorphic to  $H_B \otimes \mathbb{Q}_{\ell}$ . The representation of  $\Gamma_F$  on the  $\ell$ -adic étale cohomology makes  $\Gamma_F$  act on the functor  $H_{\acute{e}t,\ell}$  and this gives rise to a morphism

$$\rho_{\ell} \colon \Gamma_F \to G_A(\mathbb{Q}_{\ell}).$$

Let  $\Phi_v$  an arithmetic Frobenius element, belonging to the decomposition group  $\Gamma_{F_v} \cong D_{\bar{v}} \subset \Gamma_F$ . This gives rise to local data as in Section 2.1. We consider the restriction of the  $\rho_\ell$  to the Weil group  $W_v = W_{F_v}$ . As explained in Section 2.2, it defines a representation of the Weil–Deligne group  $'W_v = 'W_{F_v}$ , that is, a pair  $(\rho'_\ell, N'_\ell)$  with  $\rho'_\ell \colon W_v \to G_A(\mathbb{Q}_\ell)$  and  $N'_\ell \in \mathfrak{g}^{ss}_A \otimes \mathbb{Q}_\ell$ . By Corollary 2.8, there is an open subgroup J of the inertia group  $I_{\bar{v}}$  such that each  $\rho'_\ell$  is trivial on J.

To apply the results above on 1-motives, it is convenient to work with Tate modules instead of étale cohomology groups. The  $\ell$ -adic Galois representation  $V_{\ell}A = T_{\ell}A \otimes_{\mathbb{Z}_{\ell}} \mathbb{Q}_{\ell}$  is dual to  $H^1_{\acute{e}t}(A_{/\bar{F}}, \mathbb{Q}_{\ell})$ . Identification of the fibre functor defined by  $V_{\ell}$  to the dual of the one defined by  $H_{\acute{e}t,\ell}$  endows  $V_{\ell}$  with the structure of a representation of  $G_{A/\mathbb{Q}_{\ell}}$ . The action of  $\Gamma_F$  on  $V_{\ell}A$  is given by the same morphism  $\rho_{\ell} \colon \Gamma_F \to G_A(\mathbb{Q}_{\ell})$  as before. The corresponding representation  $(\rho'_{\ell}, N'_{\ell})$  of the Weil–Deligne group ' $W_{F_v}$  is also unchanged.

**3.3.** The group  $G^{\natural ad}$ . In [Noot 2009, 1.5] one finds the construction of a group Aut'(G)/ $\overline{\mathbb{Q}}$  of automorphisms of  $G_{A/\overline{\mathbb{Q}}}$ . In this paper we will write  $G^{\natural ad}$  for this "natural extension" of  $G^{ad}$ . We briefly sketch its construction.

The derived group  $G_{A/\overline{\mathbb{Q}}}^{\text{der}}$  is the almost direct product of almost simple subgroups  $G_i \subset G_{A/\overline{\mathbb{Q}}}$ , for *i* in some index set *I*. Let  $J \subset I$  be the set of indices *i* such that  $G_i \cong \text{SO}(2k_i)/\overline{\mathbb{Q}}$  for some  $k_i \ge 4$  and for each  $i \in J$  put  $G'_i = O(2k_i) \supset G_i$ . We define

$$G^{\natural \operatorname{ad}} = \prod_{i \in J} G_i^{\prime \operatorname{ad}} \times \prod_{i \in I \setminus J} G_i^{\operatorname{ad}} \supset G_{A/\overline{\mathbb{Q}}}^{\operatorname{ad}}.$$

As this group operates trivially on the centre of  $G_{A/\overline{\mathbb{Q}}}^{\text{der}}$ , we can define an action of  $G^{\natural \, \text{ad}}$  on  $G_{A/\overline{\mathbb{Q}}}$  extending the adjoint action on  $G_{A/\overline{\mathbb{Q}}}^{\text{der}}$  and with  $G^{\natural \, \text{ad}}$  acting trivially on the centre of  $G_{A/\overline{\mathbb{Q}}}$ . Through the adjoint representation, the group  $G^{\natural \, \text{ad}}$  also acts on the Lie algebra  $\mathfrak{g} \otimes \overline{\mathbb{Q}}$ .

**3.4.** *Compatible systems revisited.* We introduce a variant of the notion, introduced in Section 2.3, of a compatible system defined over  $\mathbb{Q}$  of representations of  ${}^{\prime}W_{F_v}$  with values in  $G_A$ . This time we allow conjugation by the group  $G^{\natural ad}$  so the condition is weaker than G-conjugacy if  $G_{A/\mathbb{Q}}^{der}$  has factors of the form SO(2*k*).

Let  $v, W_v$  and  $\Phi_v$  be as in Sections 3.1 and 3.2. For a fixed  $\ell$ , let  $(\rho'_{\ell}, N'_{\ell})$  define a representation of  $W_{v/\mathbb{Q}_{\ell}}$  with values in  $G_{A/\mathbb{Q}_{\ell}}$ . We say that this representation is *defined over*  $\mathbb{Q}$  *modulo*  $G_A^{\natural \, \mathrm{ad}}$  if, for every algebraically closed field  $\Omega \supset \mathbb{Q}_{\ell}$  and every  $\sigma \in \operatorname{Aut}_{\mathbb{Q}}(\Omega)$ , there is a  $g \in G_A^{\natural \, \mathrm{ad}}(\Omega)$  such that

$$\sigma(\rho'_{\ell} \otimes_{\mathbb{Q}_{\ell}} \Omega) = g(\rho'_{\ell} \otimes_{\mathbb{Q}_{\ell}} \Omega)g^{-1} \text{ and } \sigma(N'_{\ell} \otimes_{\mathbb{Q}_{\ell}} 1) = \mathrm{Ad}(g)(N'_{\ell} \otimes_{\mathbb{Q}_{\ell}} 1).$$

We say that a system of representations  $(\rho'_{\ell}, N'_{\ell})$  of  $W_{\nu/\mathbb{Q}_{\ell}}$  is a *compatible* system of representations of  $W_{\nu}$  modulo  $G_{A}^{\natural ad}$  if for every pair  $(\ell, \ell')$  and every

algebraically closed field  $\Omega \supset \mathbb{Q}_{\ell}, \mathbb{Q}_{\ell'}$ , there is a  $g \in G_A^{\natural \operatorname{ad}}(\Omega)$  such that

$$\rho_{\ell}' \otimes_{\mathbb{Q}_{\ell}} \Omega = g(\rho_{\ell'}' \otimes_{\mathbb{Q}_{\ell'}} \Omega)g^{-1} \quad \text{and} \quad N_{\ell}' \otimes_{\mathbb{Q}_{\ell}} 1 = \operatorname{Ad}(g)(N_{\ell'}' \otimes_{\mathbb{Q}_{\ell'}} 1) \in \mathfrak{g} \otimes \Omega.$$

**Definition 3.5.** Let  $\Omega$  be an algebraically closed field, *G* a linear algebraic group over a subfield of  $\Omega$  and *V* a representation of *G*. A semisimple  $g \in G(\Omega)$  is *neat* if the Zariski closure of the subgroup of  $G(\Omega)$  generated by *g* is connected. A semisimple element  $g \in G(\Omega)$  is *weakly neat* (*with respect to V*) if 1 is the only root of unity among the quotients  $\lambda \mu^{-1}$  of eigenvalues  $\lambda$  and  $\mu$  of *g*.

For weak neatness, we will suppress the reference to V if it is clear which representation is being considered. For elements of the Mumford–Tate group of an abelian variety, we always consider the representation defined by the Tate module. A neat element is weakly neat with respect to any representation.

**Theorem 3.6.** Assume that A has semistable reduction at v and that, for some  $\ell$ , the image  $\rho'_{\ell}(\Phi_v)$  is weakly neat. Then the representations  $(\rho'_{\ell}, N'_{\ell})$  of ' $W_v$  corresponding to A are defined over  $\mathbb{Q}$  modulo the action  $G_A^{\natural ad}$ . For  $\ell \neq p$ , these representations form a compatible system of representations of ' $W_v$  modulo the action of  $G_A^{\natural ad}$ .

#### 3.7. Remarks.

**3.7.1.** The condition that *A* has semistable reduction at *v* implies that  $\rho'_{\ell}$  is trivial on  $I_{\bar{v}}$ . In particular, the condition that  $\rho'_{\ell}(\Phi_v)$  is weakly neat does not depend on the choice of the Frobenius element  $\Phi_v$ . Also note that  $\rho'_{\ell}(\Phi_v) = \rho_{\ell}(\Phi_v)$  so that the condition can also be checked on  $\rho_{\ell}(\Phi_v)$ . By the main theorem, or in a more elementary fashion by Proposition 1.8, the condition that  $\rho'_{\ell}(\Phi_v)$  is weakly neat is independent of  $\ell$ .

**3.7.2.** In general, A only has potentially semistable reduction at v. In this case,  $\rho'_{\ell}|_{I_{\bar{v}}}$  has finite image so, for  $\sigma \in I_{\bar{v}}$ , all eigenvalues of  $\rho'_{\ell}(\sigma)$  are roots of unity. The elements of  $\rho'_{\ell}(I_{\bar{v}})$  are not neat and the methods of this paper do not seem to permit one to prove that the  $\rho'_{\ell}$  form a compatible system in this case. As the monodromy operators are unchanged by a finite base extension, one may reduce to the case of stable reduction to prove that the  $N'_{\ell}$  do form a compatible system defined over  $\mathbb{Q}$ .

**3.7.3.** To check the condition of weak neatness, one has to determine the characteristic polynomial of  $\rho_{\ell}(\Phi_v)$  for at least one  $\ell$ , which is not always feasible in practice. However, the condition always holds for a power of  $\Phi_v$ , that is, after replacing *F* by a finite extension *F'* and *v* by a valuation of *F'* lying over *v*. Also note that if, for some  $\ell \neq p$ , the elements of  $\rho_{\ell}(\Gamma_F)$  are congruent to 1 modulo  $\ell$  (or congruent to 1 modulo 4 if  $\ell = 2$ ), then  $\rho_{\ell}(\Phi_v)$  is necessarily weakly neat.

On the other hand, it is easy to construct abelian varieties, even with good reduction, that do not satisfy the condition of weak neatness. For this, one may

choose a  $p^{\alpha}$ -Weil number  $\alpha$  having two conjugates differing by a nontrivial root of unity. There exists an abelian variety  $A_0$  over a finite field of characteristic psuch that the characteristic polynomial of Frobenius is a power of the minimum polynomial of  $\alpha$ . Any lifting of  $A_0$  over a number field provides a counterexample to the neatness condition of the theorem.

**3.7.4.** If the abelian variety A has good reduction at v then the monodromy  $N'_{\ell}$  is trivial and the theorem reduces to the main result, [Noot 2009, Théorème 1.8].

**3.7.5.** We finally refer to [ibid., Remark 1.9(4)] for a note on the density of the set of places v of good reduction where  $\rho_{\ell}(\Phi_v)$  is weakly neat. Density statements of this type are not useful in the present context as the number of places where A does not have good reduction is finite.

**3.8.**  $G_A$ , monodromy and Frobenius weights. As pointed out in Section 3.2, the system  $(\rho_\ell)$  is determined by the Galois representations on the Tate modules of A. From now on, we systematically adopt this point of view.

In the proof of Corollary 2.7, we applied [Raynaud 1994, 4.2] to the motive M in order to reduce to a strict motive M'. Applying the same argument to the abelian variety  $A_v = A_{/F_v}$ , one again obtains a strict 1-motive  $M'/F_v$  endowed with a system of canonical  $\Gamma_{F_v}$ -equivariant isomorphisms

$$V_{\ell}(A_v) = T_{\ell}(A_v) \otimes_{\mathbb{Z}_{\ell}} \mathbb{Q}_{\ell} \cong V_{\ell}(M').$$

Let  $M' = [Y \to G]$ , let  $Y^*$  be the character group of the toric part of G and write r and  $r^*$  for the ranks of Y and  $Y^*$ . Let g be the dimension of the quotient of G by its maximal torus.

In Section 1.10 we defined the Frobenius weight cocharacter of a local Galois representation associated to a 1-motive. Applying this construction to the restrictions  $\rho_{\ell}|_{D_v}$  we obtain the Frobenius weight cocharacter

$$w_\ell \colon \mathbb{G}_m \to \mathrm{GL}(V_\ell(A)).$$

**Lemma 3.9.** Under the conditions above we have  $r = r^*$  and the monodromy operator  $N: Y \otimes \mathbb{Q} \to Y^* \otimes \mathbb{Q}$  associated to M' is an isomorphism. For each  $\ell$ , the map  $N'_{\ell}$  defines an isomorphism from the t-eigenspace of  $w_{\ell}$  acting on  $V_{\ell}(A)$  onto the  $t^{-1}$ -eigenspace.

*Proof.* The arguments used in [Raynaud 1994, 4.2] show that  $Y = \Lambda$ , where  $\Lambda$  is the  $\mathbb{Z}$ -module in the diagram (\*\*) of [loc. cit.], so *r* is equal to the rank of  $\Lambda$ . This reference also implies that  $r^*$  is equal to the rank of  $\Lambda$ , so  $r = r^*$ . Still by [ibid., 4.2], the intersection of  $\Lambda$  with the rigid analytic generic fibre of *G* is trivial. With the notation of [ibid., 4.3], this means that for any  $y \in Y$ , there exists  $y^* \in Y^*$  with

 $\mu_o(y \otimes y^*) > 0$ . It follows that *N* induces an isomorphism  $Y \otimes \mathbb{Q} \cong Y^* \otimes \mathbb{Q}$ . All this is classical; see for example [Grothendieck et al. 1972, Exposé IX, Théorème 10.4].

The monodromy filtration on  $V_{\ell}(M')$  coincides with the weight filtration so the last statement immediately follows from the previous ones.

**Lemma 3.10.** The Frobenius weight cocharacter  $w_{\ell}$  factors through  $G_{A/\mathbb{Q}_{\ell}}$ . In fact, this cocharacter factors through a torus  $T_v \subset G_{A/\mathbb{Q}_{\ell}}$  containing  $\rho_{\ell}(\Phi_v)$ .

*Proof.* The Mumford–Tate group  $G_A$  contains the group  $\mathbb{G}_{m/\mathbb{Q}}$  of scalar multiplications of  $\mathrm{H}^1_{\mathrm{B}}(A(\mathbb{C}), \mathbb{Q})$ . Let  $w'_{\ell} = t \cdot w_{\ell}$ , let  $T'_v \subset G_{A/\mathbb{Q}_{\ell}}$  be the identity component of the Zariski closure of the subgroup of  $G_{A/\mathbb{Q}_{\ell}}$  generated by  $\rho_{\ell}(\Phi_v)$  and put  $T_v = \mathbb{G}_m T'_v$ . We will prove that  $w_{\ell}$  factors through  $T_v$  by showing that  $w'_{\ell}$  factors through  $T'_v$ .

The last statement follows from the argument used in [Serre 2000], §4 of the first letter. The proof comes down to the fact that the eigenvalues of  $w'_{\ell}(t)$  satisfy all the multiplicative relations satisfied by the archimedean absolute values of the eigenvalues of  $\rho_{\ell}(\Phi_v)$ .

#### 4. Generalities on tractable abelian varieties

**4.1.** *Tractable abelian varieties.* The notion of "variété abélienne accommodante" was introduced in [Noot 2009, 2.3]; in this paper we call such a variety a *tractable* abelian variety. Let us recall the relevant ideas.

First of all, we define the notion of an admissible representation of a reductive group. Heuristically, the admissible representations are the representations encountered when studying Shimura data of abelian type that admit an embedding into the Siegel Shimura datum. We refer to [Deligne 1979, 1.3] for this classification. To be precise, let *K* be a field of characteristic 0 and let  $\Omega \supset K$  be an algebraically closed extension. Assume that  $G^s$  is a linear algebraic group over *K* such that  $G^s_{/\Omega}$  is almost simple of type *A*, *B*, *C* or *D*. Let  $V^s$  be a faithful *K*-linear representation of  $G^s$ . We say that  $V^s$  is an *admissible* representation of  $G^s$  in the following cases:

- $G_{/\Omega}^s$  is of type  $A_n$  and  $V^s \otimes_K \Omega$  is a multiple of the direct sum of the representations of highest weights  $\varpi_1$  and  $\varpi_n$  if  $n \ge 2$  and a multiple of the representation of highest weight  $\varpi_1$  if n = 1.
- $G_{/\Omega}^s$  is of type  $B_n$  and  $V^s \otimes_K \Omega$  is a multiple of the representation of highest weight  $\overline{\omega}_n$ .
- $G_{/\Omega}^s$  is of type  $C_n$  and  $V^s \otimes_K \Omega$  is a multiple of the representation of highest weight  $\varpi_1$ .
- $G_{/\Omega}^s$  is of type  $D_n$  and  $V^s \otimes_K \Omega$  is a multiple of the representation of highest weight  $\overline{\omega}_1$ .

•  $G_{/\Omega}^s$  is of type  $D_n$  and  $V^s \otimes_K \Omega$  is a multiple of the direct sum of the representations of highest weights  $\varpi_{n-1}$  and  $\varpi_n$ .

In the first three cases, we will say that the pair  $(G^s, V^s)$  is of type  $A_n$ ,  $B_n$  or  $C_n$ , in the last two cases we say that  $(G^s, V^s)$  is of type  $D_n^{\mathbb{H}}$  or of type  $D_n^{\mathbb{R}}$ , respectively.

Returning to abelian varieties, we let A be an abelian variety over  $\mathbb{C}$  and write  $V = H^1_{\mathsf{R}}(A(\mathbb{C}), \mathbb{Q})$ . We say that A is *strictly tractable* if

- there exists a totally real number field *K* and an almost simple linear algebraic group  $G^s$  over *K* such that  $G_A^{der} = \operatorname{Res}_{K/\mathbb{Q}} G^s$ ;
- as a representation of  $G_A^{\text{der}}$ , the cohomology group V is the restriction of scalars of an admissible representation  $V^s$  of  $G^s$ ;
- if (G<sup>s</sup>, V<sup>s</sup>) is of type D<sup>ℝ</sup><sub>n</sub> then every character space in V ⊗ Q
   for the action of the centre of G<sub>A/Q</sub> is an admissible representation of a factor of G<sup>der</sup><sub>A/Q</sub>; and
- the conditions above do not hold for any proper abelian subvariety of A.

The type of a strictly tractable abelian variety is the type of the pair  $(G^s, V^s)$ .

We will say that A is *tractable* if A is isogenous to a product  $\prod_{i=1}^{m} A_i$  of strictly tractable abelian varieties  $A_i$  and  $G_A^{\text{der}} \cong \prod_{i=1}^{m} G_{A_i}^{\text{der}}$ . If  $F \subset \mathbb{C}$  is a subfield, an abelian variety A/F is (strictly) tractable if  $A_{/\mathbb{C}}$  is and if  $G_A$  is connected.

**4.2.** *Remark.* The concept of tractability is an auxiliary notion used in the proof of the main theorem. It does not seem to be of independent interest, though it is conceivable that the method of the present paper can be used in other contexts.

Heuristically, the fact that an abelian variety A is strictly tractable means that the representation of  $G_A^{\text{der}}$  on  $V = H_B^1(A(\mathbb{C}), \mathbb{Q})$  decomposes over  $\overline{\mathbb{Q}}$  as a direct sum of irreducible representations of the almost simple factors of  $G_{A/\overline{\mathbb{Q}}}^{\text{der}}$ . In particular, any abelian variety  $A/\mathbb{C}$  of dimension g with  $G_A^{\text{der}} = \operatorname{Sp}_{2g}$  is tractable. This means that, in the moduli space of g-dimensional abelian varieties, the points corresponding to nontractable varieties belong to a countable union of closed subvarieties and thus the general abelian variety is tractable.

If the simple factors of  $V \otimes \overline{\mathbb{Q}}$ , as a representation of  $G_{A/\overline{\mathbb{Q}}}^{\text{der}}$ , are tensor products of irreducible representations of the almost simple factors  $G_{A/\overline{\mathbb{Q}}}^{\text{der}}$ , then A is not tractable. The simplest such example was given by Mumford [1969]. The generic members of the families constructed there are not tractable.

More generally, any simple abelian variety  $A/\mathbb{C}$ , with  $L = \text{End}^0(A)$ , whose Mumford–Tate group coincides with the group of *L*-linear symplectic similitudes of  $H^1_B(A(\mathbb{C}), \mathbb{Q})$  is tractable. The converse is not true, as can be seen for example in the case where the Mumford–Tate group is of type  $D_n$ . Nevertheless, being tractable still means that the endomorphism algebra is big compared to the Mumford–Tate group. This is the key to the proof of the main theorem for tractable abelian varieties.

By definition, abelian varieties of CM type are not tractable. As pointed out in the introduction, the system of  $\ell$ -adic representations associated to an abelian variety of CM type is described by the theory of complex multiplication and the main theorem is true in the CM case.

**4.3.** The algebra  $L \subset \operatorname{End}^{0}(A)$ . In the proof of the main Theorem 3.6, we will adapt the ideas used in [Noot 2009]. We will in particular make use of the algebra  $L \subset \operatorname{End}(A) \otimes \mathbb{Q}$  constructed in the beginning of the proof of [ibid., Théorème 2.4]. For this construction, first decompose  $V \otimes \overline{\mathbb{Q}} = \bigoplus_{i=1}^{n} V_i$ , where the  $V_i$  are the isotypic components of the representation of  $G_{A/\overline{\mathbb{Q}}}$  on  $V \otimes \overline{\mathbb{Q}}$ . In other words, each  $V_i$  is a multiple of an irreducible representation of  $G_{A/\overline{\mathbb{Q}}}$  and  $\operatorname{Hom}_{G_{A/\overline{\mathbb{Q}}}}(V_i, V_j) = 0$  for  $i \neq j$ . This decomposition defines a subalgebra  $\overline{\mathbb{Q}}^n \subset \operatorname{End}_{G_{A/\overline{\mathbb{Q}}}}(V \otimes \overline{\mathbb{Q}})$ , with the *i*-th factor  $\overline{\mathbb{Q}}$  acting on the factor  $V_i$  by scalar multiplication. Taking  $\Gamma_{\mathbb{Q}}$ -invariants, this inclusion descends to

$$L \subset \operatorname{End}_{G_{A/\overline{\mathbb{Q}}}}(V \otimes \overline{\mathbb{Q}}) = \operatorname{End}^{0}(A_{/\mathbb{C}}) = \operatorname{End}^{0}(A),$$

where *L* is a finite, semisimple, commutative  $\mathbb{Q}$ -algebra. The last equality follows from the fact that  $G_A$  is connected and is justified in [Noot 2009, proof of 2.4].

There is a canonical isomorphism  $L \otimes \overline{\mathbb{Q}} \cong \prod_{\iota: L \to \overline{\mathbb{Q}}} \overline{\mathbb{Q}}$ ; in fact L is defined as the algebra of  $\Gamma_{\mathbb{Q}}$ -invariants in the product on the right hand side. The direct factors of  $V \otimes \overline{\mathbb{Q}}$  are indexed by the morphisms  $\iota: L \to \overline{\mathbb{Q}}$ , with the  $\iota$ -factor of  $\prod_{\iota} \overline{\mathbb{Q}}$  acting on  $V_{\iota}$  by scalar multiplications and acting trivially on the other  $V_{\kappa}$ . The decomposition of  $L \otimes \overline{\mathbb{Q}}$  thus gives rise to a decomposition  $V \otimes \overline{\mathbb{Q}} = \prod_{\iota} V_{\iota}$ . There is a similar decomposition  $V \otimes \Omega = \prod_{\iota} V_{\iota}$  for any algebraically closed field  $\Omega$  of characteristic 0, with the product taken over all morphisms  $\iota: L \to \Omega$ .

**Lemma 4.4.** Assume that  $A/\mathbb{C}$  is a strictly tractable abelian variety. Unless A is of type  $D_n^{\mathbb{R}}$ , the algebra above  $L \subset \text{End}(A) \otimes \mathbb{Q}$  is a field. If  $A/\mathbb{C}$  of type  $D_n^{\mathbb{R}}$  then the algebra L is either a field or it is isomorphic to  $L' \times L'$  for a field L'.

*Proof.* As *L* is a semisimple algebra, it decomposes as a product of fields. This decomposition gives rise to a corresponding decomposition of *A* and unless the pair  $(G^s, V^s)$  associated to *A* is of type  $A_n$  or  $D_n^{\mathbb{R}}$ , each factor still satisfies the first three conditions of the definition of a strictly tractable abelian variety. If there is more than one factor, this violates the minimality condition.

If the pair  $(G^s, V^s)$  associated to A is of type  $A_n$ , then the argument used in [Noot 2006, 5.1] shows that the complex conjugation acts on the Dynkin diagram of  $G^s$  by the main involution. This implies that a direct factor of  $V \otimes \overline{\mathbb{Q}}$ , which is a representation of highest weight  $\overline{\varpi}_1$  of some factor of  $G_{A/\overline{\mathbb{Q}}}^{\text{der}}$  belongs to the same  $\Gamma_{\overline{\mathbb{Q}}}$ -orbit as the representation of highest weight  $\overline{\varpi}_n$  of the same factor. It follows that for any decomposition of A as above, each factor still satisfies the first three conditions of the definition. The minimality condition again implies that there is only one factor.

If  $(G^s, V^s)$  is of type  $D_n^{\mathbb{R}}$ , then a direct factor L' of L may define a factor A' of A such that the associated pair  $(G^s, V^s)$  is a half spin representation. In that case, A has a factor A'' for which  $(G^s, V'^s)$  is the other half spin representation. By minimality, one must have  $A \sim A' \oplus A''$ . In this case, the set of vertices  $\varpi_{n-1}$  and the set of vertices  $\varpi_n$  of the Dynkin diagram of  $G_A^{der}$  form two separate orbits for the  $\Gamma_{\mathbb{Q}}$  action. Since these orbits are isomorphic as  $\Gamma_{\mathbb{Q}}$ -sets, it follows that  $L = L' \oplus L'$ .

**4.5.** The group *H*. For the rest of this section we will assume, in addition to the hypotheses of Section 3.1, that A/F is tractable. Let  $G_A^{\natural ad}$  be the linear algebraic group over  $\overline{\mathbb{Q}}$  introduced in Section 3.3.

To prove the theorem for A, we will use the results on 1-motives obtained in Section 2 so it is convenient to study the Galois representations defined by the Tate-modules; see Section 3.2. Let  $V = H_1(A(\mathbb{C}), \mathbb{Q})$  and for each prime number  $\ell$  put  $V_{\ell} = T_{\ell}(A) \otimes_{\mathbb{Z}_{\ell}} \mathbb{Q}_{\ell}$ , where  $T_{\ell}(A)$  is the  $\ell$ -adic Tate module of A. As in Section 3.2, there is a natural representation of  $G_A$  on V and there are canonical isomorphisms  $V_{\ell} \cong V \otimes \mathbb{Q}_{\ell}$ . The action of  $\Gamma_F$  on the  $V_{\ell}$  is given by the representations  $\rho_{\ell} \colon \Gamma_F \to G_A(\mathbb{Q}_{\ell})$ .

We closely follow the proof of [Noot 2009, Théorème 2.4]. Let the endomorphism algebra  $L \subset \operatorname{End}^0(A)$  be as in Section 4.3. It is a product of number fields  $L_i$  for  $i = 1, \ldots, s$ . This decomposition gives rise to a decomposition up to isogeny  $A \sim \prod_{i=1}^{s} A_i$  and to a corresponding decomposition  $V = \bigoplus_{i=1}^{s} V_i$ , where  $V_i = \operatorname{H}_1(A_i(\mathbb{C}), \mathbb{Q})$ . For each *i* one has  $L_i \subset \operatorname{End}^0(A_i)$  and this action endows  $V_i$ with the structure of  $L_i$ -vector space. It follows from Lemma 4.4 that each factor  $A_i$  is either strictly tractable or that it is of type  $D_n^{\mathbb{R}}$ . If  $A_i$  is not strictly tractable, it follows from the definition and again from Lemma 4.4 that there is another factor  $A_j$  such that  $L_i \cong L_j$  and the product  $A_i \times A_j$  is strictly tractable. In this case we change definitions and put  $A_i = A_i \times A_j$  and  $L_i = L_i \times L_j$ . We suppress the factors  $A_j$  and  $L_j$  and modify the value of *s* accordingly. After these modifications, we have decompositions  $L = \prod_{i=1}^{s} L_i$  and  $A \sim \prod_{i=1}^{s} A_i$ , where all factors  $A_i$  are strictly tractable and each  $L_i$  is either a field of a product  $L'_i \times L'_i$  of fields.

Let  $d_i = \dim_{L_i} V_i$  and let

$$H = \prod_{i=1}^{s} \operatorname{Res}_{L_{i}/\mathbb{Q}} \operatorname{GL}_{/L_{i}}(V_{i}) = \prod_{i=1}^{s} H_{i} \cong \prod_{i=1}^{s} \operatorname{Res}_{L_{i}/\mathbb{Q}} \operatorname{GL}_{d_{i}/L_{i}}$$

be the centraliser of L in GL(V). In the case where  $L_i \cong L'_i \times L'_i$ , the factor  $H_i$  is

$$\left(\operatorname{Res}_{L'_i/\mathbb{Q}}\operatorname{GL}_{d_i/L'_i}\right)^2$$
.

It is a linear algebraic group over  $\mathbb{Q}$ . The action of  $G_A$  on V commutes with the action of  $\operatorname{End}^0(A)$ , so  $G_A \subset H$ . The decomposition  $H = \prod_{i=1}^{s} H_i$  corresponds to the decompositions of L, of A and of V. In particular,  $H_i$  is the only factor of H acting nontrivially on  $V_i$ . Writing  $G_{A_i}$  for the Mumford–Tate group of  $A_i$  one has  $G_{A_i} \subset H_i$ .

Let  $\Omega$  be an algebraically closed extension of  $\mathbb{Q}$ . For each  $\mathbb{Q}$ -algebra homomorphism  $\iota: L \to \Omega$  there is a unique index  $i = i(\iota)$  such that  $\iota$  factors through  $L \to L_i \to \Omega$ , where  $L \to L_i$  is the projection and  $L_i \to \Omega$  a ring homomorphism. This final map is an embedding if  $L_i$  is a field and an embedding of one of the factors of  $L_i$  if  $A_i$  is of type  $D_n^{\mathbb{R}}$  and  $L_i$  is a product  $L'_i \times L'_i$ . Let  $d_{\iota} = d_i$ . Extending the base field to  $\Omega$  one obtains

$$H_{/\Omega} \cong \prod_{\iota: L \to \Omega} \operatorname{GL}_{d_{\iota}/\Omega}.$$

The group  $G_{A/\Omega}$  embeds into this product and for  $\iota: L \to \Omega$  we let  $G_{\iota}$  be its image in the factor  $GL_{d_{\iota}/\Omega}$  corresponding to  $\iota$ .

This gives rise to similar decompositions of the Lie algebras. For  $\mathfrak{h} = \text{Lie}(H)$  we have  $\mathfrak{h} \otimes_{\mathbb{Q}} \Omega = \bigoplus \mathfrak{h}_{\iota} = \bigoplus \text{End}(V_{\iota}) \cong \bigoplus \mathfrak{gl}_{d_{\iota}/\Omega}$ . The inclusion  $G_A \subset H$  induces

$$\mathfrak{g}_A \otimes \Omega = \operatorname{Lie}(G_A) \otimes \Omega \hookrightarrow \bigoplus_{\iota \colon L \to \Omega} \mathfrak{g}_\iota \subset \bigoplus_\iota \operatorname{End}(V_\iota) = \mathfrak{h} \otimes \Omega,$$
 (4.5\*)

where  $\mathfrak{g}_{\iota} \subset \mathfrak{h}_{\iota} = \operatorname{End}(V_{\iota} \otimes \Omega)$  is the Lie algebra of  $G_{\iota}$ . For both the group  $H_{/\Omega} \cong \prod \operatorname{GL}_{d_{\iota}/\Omega}$  and the Lie algebra  $\mathfrak{h} \otimes_{\mathbb{Q}} \Omega \cong \prod \mathfrak{gl}_{d_{\iota}/\Omega}$ , the obvious action of  $\sigma \in \operatorname{Aut}(\Omega)$  on the left hand side translates on the right hand side to

$$\sigma: (x_{\iota})_{\iota: L \to \Omega} \mapsto (\sigma(x_{\sigma^{-1}\iota}))_{\iota: L \to \Omega}.$$

$$(4.5^{\dagger})$$

**4.6.** *Monodromy and Frobenius weights.* The  $\ell$ -adic realisations of A decompose in the same way as the Betti realisation. Define  $V_{i,\ell} = T_{\ell}(A_i) \otimes \mathbb{Q}_{\ell}$  for every  $\ell$ . The decomposition of A gives rise to  $\Gamma_{F_v}$ -equivariant  $L_{\ell} = L \otimes \mathbb{Q}_{\ell}$ -linear decompositions  $V_{\ell} = \bigoplus_{i=1}^{s} V_{i,\ell}$  and we have canonical  $L_i$ -linear isomorphisms  $V_{i,\ell} \cong V_i \otimes_{\mathbb{Q}} \mathbb{Q}_{\ell}$ .

Assume that *A* has semistable reduction at *v* and let further notation be as in Section 3.1. Let *q* be the order of the residue field of  $F_v$ . To study the monodromy we follow Section 2.2, so we fix identifications  $\mathbb{Q}_{\ell}(1) \cong \mathbb{Q}_{\ell}$  and define the monodromy operators  $N'_{\ell}$  by (2.2\*). One has  $N'_{\ell} \in \mathfrak{g}^{ss}_A \otimes \mathbb{Q}_{\ell}$  by Section 3.2. Similarly, for each  $A_i$  we have the monodromy operator  $N'_{i,\ell} \in \mathfrak{g}^{ss}_{A_i} \otimes \mathbb{Q}_{\ell}$ . Under the embedding  $\mathfrak{g}_A \subset \bigoplus_{i=1}^s \mathfrak{g}_{A_i}$  the element  $N'_{\ell}$  maps to  $(N'_{i,\ell})_{i=1,\dots,s}$ .

For A and each one of the  $A_i$ , let the Frobenius weight cocharacters

$$w_{\ell} \colon \mathbb{G}_{m/\overline{\mathbb{Q}}_{\ell}} \to G_{A/\overline{\mathbb{Q}}_{\ell}} \quad \text{and} \quad w_{i,\ell} \colon \mathbb{G}_{m/\overline{\mathbb{Q}}_{\ell}} \to G_{A_i/\overline{\mathbb{Q}}_{\ell}}$$

be as in Section 3.8. These cocharacters factor through the Mumford–Tate groups by Lemma 3.10. As in the case of the monodromy operators, the cocharacter

$$(w_{i,\ell})_{i=1,\dots,s} \colon \mathbb{G}_{m/\overline{\mathbb{Q}}_{\ell}} \to \prod_{i=1,\dots,s} G_{A_i/\overline{\mathbb{Q}}_{\ell}}$$

is the composite of  $w_{\ell}$  with the inclusion  $G_A \subset \prod G_{A_i}$ . It follows from Lemma 3.9 that  $w_{\ell}$  and the  $w_{i,\ell}$  split the monodromy filtrations on  $V_{\ell}$  and on the  $V_{i,\ell}$ , respectively.

## **5.** Strictly tractable abelian varieties of types $A_n$ , $C_n$ and $D_n^{\mathbb{H}}$

**5.1.** We keep the notation of Section 3.1. In this section, we will assume that *A* is a strictly tractable abelian variety and that its Mumford–Tate group  $G_A$  is of type  $A_n$ ,  $C_n$  or  $D_n^{\mathbb{H}}$ . According to Lemma 4.4, the algebra *L* then is a field. In particular, where the group *H* introduced in Section 4.5 is concerned, we have  $d_i = d_1 = d$  for all  $\iota: L \to \Omega$ . As in Section 4.6, consider the monodromy operator  $N'_{\ell}$  and the weight cocharacter  $w_{\ell}$ . It follows from Corollary 2.7 that the *H*-conjugacy class of  $(N'_{\ell}, w_{\ell})$  is defined over  $\mathbb{Q}$  and that it is independent of  $\ell$ , in accordance with the terminology developed in Section 2.3.

For any algebraically closed extension  $\Omega$  of  $\mathbb{Q}$ , the Lie algebra  $\mathfrak{g}^{ss} \otimes \Omega$  is a direct sum of simple Lie algebras. The groups  $G_A^{ad}(\Omega)$  and  $G_A^{\natural ad}(\Omega)$  act on each direct factor of this Lie algebra factor through a unique simple factor. Fix a factor  $\mathfrak{g}_l^{ss}$  of  $\mathfrak{g}^{ss} \otimes \Omega$  and consider the corresponding factors  $G_l^{der}$  of  $G_{A/\Omega}^{der}$  and  $G_l^{\natural ad}$  of  $G_{A/\Omega}^{\natural ad}$ . Under the sequence of embeddings (4.5\*),  $\mathfrak{g}_l$  embeds into a simple factor  $\mathfrak{h}_l \cong \mathfrak{gl}_d$ of  $\mathfrak{h} \otimes \Omega$ . If  $G_A$  is of type  $A_n$ , then  $\mathfrak{g}_l^{ss}$  and  $G_l^{der}$  act on the direct factor  $V_l$  of  $V \otimes \Omega$ either as a multiple of the standard representation or as its dual. If  $G_A$  is of type  $C_n$ or  $D_n^{\mathbb{H}}$  in the classification,  $\mathfrak{g}_l^{ss}$  and  $G_l^{der}$  act on  $V_l$  as a multiple of the symplectic or the orthogonal representation respectively.

Let  $\Omega \supset \mathbb{Q}_{\ell}$  be an algebraically closed field and let  $\sigma \in \operatorname{Aut}(\Omega)$ . As we saw above,  $(N'_{\ell}, w_{\ell})$  and  $\sigma(N'_{\ell}, w_{\ell}) \in \mathfrak{h} \otimes \Omega$  are conjugate under the adjoint action of  $H(\Omega)$ . Writing  $N'_{\ell} = (N'_{\ell,l})_{l} \in \prod \mathfrak{h}_{l}$  as in (4.5\*) and  $w_{\ell} = (w_{\ell,l})_{l}$  it follows from the formula (4.5†) that the projections  $(N'_{\ell,l}, w_{\ell,l})$  and  $\sigma(N'_{\ell,\sigma^{-1}l}, w_{\ell,\sigma^{-1}l})$  of these pairs are  $H_{l}(\Omega)$ -conjugate.

In the case where  $G_A$  is of type  $A_n$ , it trivially follows that  $N'_{\ell,\iota}$  and  $\sigma(N'_{\ell,\sigma^{-1}\iota})$  are conjugate under the action of  $G_{\iota}^{ad}(\Omega) = G_{\iota}^{\natural ad}(\Omega)$  on  $\mathfrak{g}_{\iota}$ . In the cases where  $G_A$  is of type  $C_n$  or  $D_n$ , it follows from [Springer and Steinberg 1970, IV §2], in particular from 2.14, that  $N'_{\ell,\iota}$  and  $\sigma(N'_{\ell,\sigma^{-1}\iota})$  are conjugate under the action of  $G_{\iota}^{\natural ad}(\Omega)$  on  $\mathfrak{g}_{\iota}$ . See also [Humphreys 1995, 7.11] for a summary of the results concerning the nilpotent conjugacy classes in the classical Lie algebras.

Similarly, if  $\Omega$  is an algebraically closed field containing  $\mathbb{Q}_{\ell}$  and  $\mathbb{Q}_{\ell'}$  then the images of  $N'_{\ell}$  and  $N'_{\ell'}$  in  $\mathfrak{h}^{ss} \otimes \Omega$  are conjugate under  $H(\Omega)$ . The argument above

implies that, for each  $\iota$ , the operators  $N'_{\ell,\iota}$  and  $N'_{\ell',\iota}$  are conjugate under the action of  $G^{\natural ad}_{\iota}(\Omega)$  on  $\mathfrak{g}_{\iota}$ .

Next consider the weight cocharacter  $w_{\ell}$ . For each  $\iota$ , let  $w_{\ell,\iota}$  be the projection of  $w_{\ell}$  to  $G_{\iota}$ . Recall that by Lemma 3.9, the monodromy operator  $N'_{\ell,\iota}$  induces an isomorphism of the *t*-eigenspace of  $w_{\ell,\iota}$  in  $V_{\iota}$  onto the  $t^{-1}$ -eigenspace. Going through the arguments of [Springer and Steinberg 1970, IV], with  $N'_{\ell,\iota}$  playing the role of X, one deduces that there exists a basis of  $V_{\iota}$  satisfying the conditions of [ibid., IV, 2.19(b)] such that  $w_{\ell,\iota}$  is the *inverse* of the cocharacter  $\lambda$  defined in [ibid., IV, 2.22]. In particular,  $w_{\ell,\iota}$  factors through the derived group  $G_{\iota}^{der}$ . This fact can also quite easily be shown directly.

If  $\Omega$  is an algebraically closed field and  $\sigma \in \operatorname{Aut}(\Omega)$  then, for  $X = \sigma(N'_{\ell,\sigma^{-1}\iota})$ , there is a basis of  $V_{\iota}$  as in [ibid., IV, 2.19(b)] such that  $\sigma(w_{\ell,\sigma^{-1}\iota})$  coincides with  $\sigma(\lambda^{-1})$ . We know that  $N'_{\ell,\iota}$  and  $\sigma(N'_{\ell,\sigma^{-1}\iota})$  are conjugate under  $G_{\iota}^{\natural ad}(\Omega)$ . Moreover, any two bases of V that satisfy the conditions of [ibid., IV, 2.19(b)] are conjugate under the centraliser  $Z_{\ell,\iota}^{\natural ad}$  of  $N'_{\ell,\iota}$  in  $G_{\iota}^{\natural ad}$ . It follows that  $(N'_{\ell,\iota}, w_{\ell,\iota})$  and  $\sigma(N'_{\ell,\sigma^{-1}\iota}, w_{\ell,\sigma^{-1}\iota})$  are  $G_{\iota}^{\natural ad}(\Omega)$ -conjugate.

If  $\ell$  and  $\ell'$  are two prime numbers and if  $\Omega$  is an algebraically closed field containing both  $\mathbb{Q}_{\ell}$  and  $\mathbb{Q}_{\ell'}$ , then the same argument, applied to  $(N'_{\ell,\iota}, w_{\ell,\iota})$  and  $(N'_{\ell',\iota}, w_{\ell',\iota})$ , proves that these two pairs are  $G_{\iota}^{\sharp ad}(\Omega)$ -conjugate.

**Proposition 5.2.** Let notation and assumptions be as above. In particular, A/F is a strictly tractable abelian variety of type  $A_n$ ,  $B_n$  or  $D_n^{\mathbb{H}}$ .

- For every  $\ell$ , every algebraically closed field  $\Omega \supset \mathbb{Q}_{\ell}$  and every  $\sigma \in \operatorname{Aut}(\Omega)$ , the image of  $(N'_{\ell}, w_{\ell})$  in  $\mathfrak{g} \otimes \Omega \times X(G_{A/\Omega})$  is conjugate to  $\sigma(N'_{\ell}, w_{\ell})$  under the adjoint action of  $G_{A}^{\natural ad}(\Omega)$ .
- If  $\Omega$  is an algebraically closed field containing both  $\mathbb{Q}_{\ell}$  and  $\mathbb{Q}_{\ell'}$ , then the images in  $\mathfrak{g} \otimes \Omega \times X(G_{A/\Omega})$  of  $(N'_{\ell}, w_{\ell})$  and  $(N'_{\ell'}, w_{\ell'})$  are  $G_A^{\natural \, \mathrm{ad}}(\Omega)$ -conjugate.

*Proof.* As *A* is tractable, the group  $G_{A/\Omega}^{der}$  is the product of its almost simple factors. If  $G_A$  is of type  $C_n$  or  $D_n^{\mathbb{H}}$ , then  $G_{A/\Omega}^{der}$  is the product of the  $G_{\iota}^{der}$ . In the case where  $G_A$  is of type  $A_n$ , we saw in the proof of Lemma 4.4 that the complex conjugation acts nontrivially on each component of the Dynkin diagram. If  $n \ge 2$ , it follows from [Noot 2006, 5.1] that *L* is a CM field and hence that the complex conjugation defines an nontrivial involution  $\iota \mapsto \iota'$  on the set of embeddings  $L \to \Omega$ . In this case,  $G_{A/\Omega}^{der}$  is a product of groups  $\Delta_{\{\iota,\iota'\}}$ , where each  $\Delta_{\{\iota,\iota'\}} \subset G_{\iota}^{der} \times G_{\iota'}^{der}$  is the graph of an isomorphism  $G_{\iota}^{der} \cong G_{\iota'}^{der}$ . Identifying  $\Delta_{\{\iota,\iota'\}}$  with  $G_{\iota}^{der}$  through the projection on the first factor, the representation of  $\Delta_{\{\iota,\iota'\}}$  on  $V_{\iota}$  is a multiple of the representation with highest weight  $\varpi_1$  and  $V_{\iota'}$  is its dual, a multiple of the representation with highest weight  $\varpi_n$ . The case where n = 1 is left to the reader.

The  $N'_{\ell,\iota}$  belong to  $\mathfrak{g}_{\iota}^{ss}$  and, as we pointed out in Section 5.1, the  $w_{\ell,\iota}$  factor through  $G_{\iota}^{der}$ . The proposition follows from the fact, proved in Section 5.1, that

 $(N'_{\ell,\iota}, w_{\ell,\iota})$  and  $\sigma(N'_{\ell,\iota}, w_{\ell,\iota})$  are  $G_{\iota}^{\natural \, \mathrm{ad}}(\Omega)$ -conjugate for each  $\iota$ , combined with the formula (4.5<sup>†</sup>).

**5.3.** The centraliser of  $(N'_{\ell}, w_{\ell})$  in  $G_{A/\overline{\mathbb{Q}}_{\ell}}^{\natural}$ . We return to the construction 3.3 of the group  $G_A^{\natural \, \mathrm{ad}}$ . In the case considered here, this group is the adjoint group of a  $\overline{\mathbb{Q}}$ -group  $G_A^{\natural}$  containing the derived Mumford–Tate group  $G_{A/\overline{\mathbb{Q}}}^{\mathrm{der}}$ . In fact, one has to take  $G_A^{\natural} = G_{A/\overline{\mathbb{Q}}}^{\mathrm{der}}$  if  $G_A$  is of type  $A_n$  or  $C_n$  and in this case we put  $G_{\ell}^{\natural} = G_{\ell}^{\mathrm{der}}$  for each embedding  $\iota: L \to \overline{\mathbb{Q}}$ . If  $G_A$  is of type  $D_n^{\mathbb{H}}$ , then

$$G_{A/\overline{\mathbb{Q}}}^{\mathrm{der}} = \prod G_{\iota}^{\mathrm{der}},$$

where each  $G_{\iota}^{\text{der}} \cong \text{SO}_{2n}$ . We put  $G_{\iota}^{\natural} = O_{2n}$  and  $G_{A}^{\natural} = \prod G_{\iota}^{\natural}$ . In all cases, it is clear from the construction Section 3.3 that  $G_{A/\overline{\mathbb{Q}}}^{\text{der}} \subset G_{A}^{\natural}$  is the identity component and that the group  $G_{A}^{\natural \text{ad}}$  defined in that construction is indeed the adjoint group of  $G_{A}^{\natural}$ . Working with this group  $G_{A}^{\natural}$ , we may apply [Springer and Steinberg 1970, IV].

In what follows, the centralisers  $C_{\ell}^{\natural} \subset G_{A/\overline{\mathbb{Q}}_{\ell}}^{\natural}$  and  $C_{\ell} \subset G_{A/\overline{\mathbb{Q}}_{\ell}}$  of the pair  $(N'_{\ell}, w_{\ell})$ will play an important role. The group  $C_{\ell}$  is the subgroup of  $G_{A/\overline{\mathbb{Q}}_{\ell}}$  generated by  $C_{\ell}^{\natural} \cap G_{A/\overline{\mathbb{Q}}_{\ell}}^{der}$  and the centre of  $G_{A/\overline{\mathbb{Q}}_{\ell}}$ . The embedding

$$G_{A/\overline{\mathbb{Q}}_{\ell}}^{\natural} o \prod_{\iota} G_{\iota/\overline{\mathbb{Q}}_{\ell}}^{\natural}$$

gives rise to a similar embedding  $C_{\ell/\overline{\mathbb{Q}}_{\ell}}^{\natural} \to \prod_{\iota} C_{\ell,\iota}^{\natural}$ , where each  $C_{\ell,\iota}^{\natural}$  is the centraliser of  $(N_{\ell,\iota}', w_{\ell,\iota})$  in  $G^{\natural}_{\iota}$ . We first determine these groups  $C^{\natural}_{\ell,\iota}$ .

By Lemma 3.9,  $N'_{\ell}$  induces an isomorphism from the *t*-eigenspace of  $w_{\ell}$  onto the  $t^{-1}$ -eigenspace of  $w_{\ell}$ . As we saw in Section 5.1, this implies that, taking  $G = G^{\natural}_{\iota}$  and  $X = N'_{\ell,\iota}$  in [Springer and Steinberg 1970, IV §2], the proper choice of a basis of  $V_{\iota}$  ensures that the cocharacter  $w_{\ell,\iota}$  is the inverse of the cocharacter  $\lambda$  defined in that paper, IV 2.22. The group  $C^{\natural}_{\ell,\iota}$  is therefore equal to the group *C* of [ibid., IV 2.23(iii)]. Note that, contrary to what is affirmed in that statement, this group is not necessarily connected. Indeed, there may be two connected components; see [ibid., 2.25 and 2.26].

To give an explicit description of  $C_{\ell,\iota}^{\natural}$ , let the 1-motive M' and the dimensions  $r = r^*$  and g be as in Section 3.8. In this case,  $r = r^*$  is equal to the dimension of both the t and the  $t^{-1}$ -eigenspaces of  $w_{\ell}$ . Since  $N'_{\ell,\iota}$  is nilpotent of echelon at most 2, it is easily deduced from [ibid., IV 1.8, 2.25] that the group  $C_{\ell,\iota}^{\natural}$  is isomorphic to the  $\overline{\mathbb{Q}}_{\ell}$ -group

- $SL_r \times SL_{2g}$  if  $G_A$  is of type  $A_n$ ,
- $O_r \times Sp_{2g}$  if  $G_A$  is of type  $C_n$  or
- $\operatorname{Sp}_r \times \operatorname{O}_{2g}$  if  $G_A$  is of type  $D_n$ .

Each factor  $C_{\ell,\iota}^{\natural}$  is therefore a product  $C_{\ell,\iota,0}^{\natural} \times C_{\ell,\iota,-1}^{\natural}$  and this decomposition is determined by the cocharacter  $w_{\ell,\iota}$ . In fact, for each integer k let  $V_{\ell}^{k-1}$  and  $V_{\ell,\iota}^{k-1}$  be the  $t^k$ -eigenspaces of  $w_{\ell}$  and  $w_{\ell,\iota}$  in  $V_{\ell} = V \otimes \mathbb{Q}_{\ell}$  and  $V_{\iota} \otimes \overline{\mathbb{Q}}_{\ell}$ , respectively. This seemingly confusing notation is consistent with Section 1.10. As  $C_{\ell,\iota}^{\natural}$  commutes with  $w_{\ell,\iota}$ , it respects the grading  $V_{\ell,\iota} = \bigoplus_{k=-2,-1,0} V_{\ell,\iota}^k$  and it follows from [ibid., IV 1.8, 2.25] that for k = 0, -1, the group  $C_{\ell,\iota,k}^{\natural}$  is the image of  $C_{\ell}^{\natural}$  in GL( $V_{\ell,\iota}^k$ ). For a group of type  $A_n$  or  $C_n$ , this embedding is given by the standard or symplectic representation, respectively, and for a group of type  $B_n$  or  $D_n$  it is defined by the orthogonal representation. The monodromy operator  $N_{\ell,\iota}$  defines an isomorphism of the representations of  $C_{\ell,\iota,0}^{\natural}$  on  $V_{\ell,\iota}^0$  and on  $V_{\ell,\iota}^{-2}$ .

The decomposition above can be defined on the level of the group  $C_{\ell}$  by taking  $C_{\ell,k}$  equal to the image of  $C_{\ell}$  in  $GL(V_{\ell}^k)$ , for k = 0, -1. Finally, let  $C_{\ell,\iota,k}$  be the image of  $C_{\ell/\overline{\mathbb{Q}}_{\ell}}$  in  $GL(V_{\ell,\iota}^k)$ , so that each  $C_{\ell,\iota,k}^{der} \subset C_{\ell,\iota,k}^{\natural}$  is the identity component, with equality for all factors other than those isomorphic to an  $SO_{2n}$ .

**5.4.** The representations  $\rho'_{\ell}$ . We now turn to the representations  $\rho'_{\ell}$  of the Weil group  $W_v = W_{F_v}$ . For general  $\psi \in W_v$ , the image  $\rho'_{\ell}(\psi)$  does not belong to  $C_{\ell}(\mathbb{Q}_{\ell})$  so in order to apply the arguments of [Noot 2009, §2], we replace  $C_{\ell}$  by the group  $\widetilde{C}_{\ell} \subset G_{A/\overline{\mathbb{Q}}_{\ell}}$  generated by  $C_{\ell}$  and the image of  $w_{\ell}$ . For  $\psi \in W_v$  one has

$$\operatorname{Ad}\left(\rho_{\ell}'(\psi)\right)(N_{\ell}') = q^{\alpha(\psi)}N_{\ell}' = \operatorname{Ad}\left(w_{\ell}(q^{\alpha(\psi)/2})\right)(N_{\ell}').$$

On the other had, it follows from Lemma 2.5 that  $\rho'_{\ell}(\psi)$  and  $w_{\ell}$  commute. This implies that  $\rho'_{\ell}(\psi)w_{\ell}(q^{-\alpha(\psi)/2})$  lies in  $C_{\ell}(\overline{\mathbb{Q}}_{\ell})$  and hence that  $\rho'_{\ell}(\psi) \in \widetilde{C}_{\ell}(\mathbb{Q}_{\ell})$ . The action of  $\widetilde{C}_{\ell}$  on  $V_{\ell}$  respects the grading  $V_{\ell} = \bigoplus_{k=-2,-1,0} V_{\ell}^{k}$  so, for k = 0, -1, it makes sense to define  $\widetilde{C}_{\ell,k}$  as the image of  $\widetilde{C}_{\ell}$  in  $GL(V_{\ell,\ell}^{k})$ . The adjoint action of  $C_{\ell}^{\natural}$  on  $C_{\ell}$  extends to an action of  $C_{\ell}^{\natural}$  on  $\widetilde{C}_{\ell}$ , with the former group acting trivially on the image of  $w_{\ell}$ .

Recall that, according to Lemma 4.4, the algebra L is a field in the cases considered here. We write  $\mathbb{A}_{(L)}^d = \operatorname{Res}_{L/\mathbb{Q}} \mathbb{A}_{/L}^d$ . The discussion above shows that taking the  $L_{\ell}$ -linear characteristic polynomials of the elements of  $\widetilde{C}_{\ell}$  acting on the  $w_{\ell}$ -eigenspace  $V_{\ell}^{-1}$ , one defines a map

$$P'_{L}: \widetilde{C}_{\ell} \to \widetilde{C}_{\ell,-1} \to \mathbb{A}^{2g}_{(L)/\overline{\mathbb{Q}}_{\ell}}.$$
(5.4\*)

We will also write  $P'_L$  for the map  $\widetilde{C}_{\ell,-1} \to \mathbb{A}^{2g}_{(L)/\overline{\mathbb{Q}}_{\ell}}$ . As in [Noot 2009], the maps  $P'_L$  factor through the quotients of  $\widetilde{C}_{\ell}$  and  $\widetilde{C}_{\ell,-1}$  by the adjoint  $C^{\natural}_{\ell}$ -action.

It follows from Proposition 1.8 and from Section 3.8 that for any  $\psi \in W_v$ , the characteristic polynomial of  $\rho'_{\ell}(\psi)$  acting as an  $L_{\ell}$ -linear automorphism on  $V_{\ell}^{-1}$  has coefficients in L and is independent of  $\ell$ . This proves the following lemma.

**Lemma 5.5.** Under the hypotheses above, the image of  $\rho'_{\ell}(\psi) \in \widetilde{C}_{\ell}(\mathbb{Q}_{\ell})$  under the map  $P'_{L}$  defined in (5.4\*) lies in  $\mathbb{A}^{2g}_{(L)}(\mathbb{Q})$  and is independent of  $\ell$ .

**5.6.** *Remark.* The statement of the lemma also holds for the image of  $\rho'_{\ell}(\psi)$  under the map  $P''_{L}: \widetilde{C}_{\ell} \to \mathbb{A}^{r}(L)$  defined by taking the characteristic polynomial on  $V^{0}_{\ell}$ . This observation is of little interest for weakly neat elements.

**Proposition 5.7.** The main theorem, Theorem 3.6, holds if A is strictly tractable and its Mumford–Tate group  $G_A$  is of type  $A_n$ ,  $C_n$  or  $D_n^{\mathbb{H}}$ .

*Proof.* As the assumptions of Theorem 3.6 are now in force, A has semistable reduction at v and the image  $\rho'_{\ell}(\Phi_v)$  of Frobenius is weakly neat. This implies that the restriction of  $\rho'_{\ell}$  to  $I_{\bar{b}}$  is trivial and that  $\rho'_{\ell}(\Phi_v)$  acts on  $V^0_{\ell}$  as multiplication by  $\varepsilon = \pm 1$  and on  $V^{-2}_{\ell}$  as multiplication by  $q\varepsilon$ .

As in Section 5.1, let  $\Omega \supset \mathbb{Q}_{\ell}$  be an algebraically closed field and let  $\sigma \in \operatorname{Aut}(\Omega)$ . We have to show that the pairs  $(N'_{\ell}, \rho'_{\ell}(\Phi_v))$  and  $(\sigma(N'_{\ell}), \sigma(\rho'_{\ell}(\Phi_v)))$  are conjugate under the action of  $G^{\sharp}_{A}(\Omega)$ . By Proposition 5.2, there is a  $g \in G^{\sharp}_{A}(\Omega)$  such that

$$(N'_{\ell}, w_{\ell}) = \operatorname{Ad}(g)(\sigma(N'_{\ell}), \sigma(w_{\ell})).$$

This implies that  $\rho'_{\ell}(\Phi_v)$  and  $g\sigma(\rho'_{\ell}(\Phi_v))g^{-1}$  belong to  $\widetilde{C}_{\ell}(\Omega)$ . As  $C^{\natural}_{\ell}$  centralises  $(N'_{\ell}, w_{\ell})$ , it is enough to show that  $\rho'_{\ell}(\Phi_v)$  and  $g\sigma(\rho'_{\ell}(\Phi_v))g^{-1}$  are conjugate under  $C^{\natural}_{\ell}(\Omega)$ . By Section 1.9 and the proof of Proposition 2.6, the element  $\rho'_{\ell}(\Phi_v)$  is semisimple. Moreover

$$P'_{L}(g\sigma(\rho'_{\ell}(\Phi_{v}))g^{-1}) = \sigma(P'_{L}(\rho'_{\ell}(\Phi_{v}))) = P'_{L}(\rho'_{\ell}(\Phi_{v})),$$

where the former equality is elementary and the latter one follows from Lemma 5.5. The projections of these elements to  $\widetilde{C}_{\ell,-1}(\Omega)$  are semisimple and weakly neat and their projections to  $C_{\ell,0}(\Omega)$  lie in the centre of this group. The required statement therefore follows from Lemma 5.8.

For the  $\ell$ -independence, let  $\ell$  and  $\ell'$  be two prime numbers and let  $\Omega$  be an algebraically closed field containing  $\mathbb{Q}_{\ell}$  and  $\mathbb{Q}_{\ell'}$ . It follows from Proposition 5.2 that there exists  $g \in G_A^{\natural}(\Omega)$  such that  $(N'_{\ell}, w_{\ell}) = \operatorname{Ad}(g)(N'_{\ell'}, w_{\ell'})$ . Exactly as before we combine the Lemmas 5.5 and 5.8 to show that  $\rho'_{\ell}(\Phi_v)$  and  $\rho'_{\ell'}(\Phi_v)$  are conjugate under  $C_{\ell}^{\natural}(\Omega)$ .

**Lemma 5.8.** Let  $\widetilde{C}_{\ell}$ ,  $C_{\ell}^{\natural}$  and  $P'_{L}$  be as above and let  $g_1, g_2 \in \widetilde{C}_{\ell}(\Omega)$  be semisimple elements whose projections to  $C_{\ell,-1}^{\natural}(\Omega)$  are weakly neat and whose projections to  $\widetilde{C}_{\ell,0}(\Omega)$  act on  $V_{\ell}^{0}$  by the same scalar multiplication. If  $P'_{L}(g_1) = P'_{L}(g_2)$  then  $g_1$  and  $g_2$  are conjugate under  $C_{\ell}^{\natural}(\Omega)$ .

*Proof.* This essentially results from [Noot 2009, Lemmas 2.5 and 2.6]. First note that the variety  $\operatorname{Conj}'(\widetilde{C}_{\ell})$  considered in [ibid., Lemma 2.5] is the variety of semisimple  $C_{\ell}^{\natural}$ -conjugacy classes in  $\widetilde{C}_{\ell}$ . Similarly, for each  $\iota$  and k, the variety  $\operatorname{Conj}'(\widetilde{C}_{\ell,\iota,k})$  is the variety of semisimple  $C_{\ell,\iota,k}^{\natural}$ -conjugacy classes in  $\widetilde{C}_{\ell,\iota,k}$ . It follows that  $\operatorname{Conj}'(\widetilde{C}_{\ell})(\Omega)$  and  $\operatorname{Conj}'(\widetilde{C}_{\ell,\iota,k})(\Omega)$  are the sets of semisimple  $C_{\ell}^{\natural}(\Omega)$  and  $C_{\ell,\iota,k}^{\natural}(\Omega)$ -conjugacy classes in  $\widetilde{C}_{\ell}(\Omega)$  and in  $\widetilde{C}_{\ell,\iota,k}(\Omega)$ , respectively.

It thus follows from [ibid., Lemma 2.5] and its proof that the map

$$\widetilde{C}_{\ell}(\Omega) \to \widetilde{C}_{\ell,0}(\Omega) \times \widetilde{C}_{\ell,-1}(\Omega) \to \prod_{\iota} \widetilde{C}_{\ell,\iota,0}(\Omega) \times \prod_{\iota} \widetilde{C}_{\ell,\iota,-1}(\Omega)$$

induces an injection on weakly neat  $C_{\ell}^{\natural}(\Omega)$ -conjugacy classes. As the images of  $g_1$  and  $g_2$  in  $\widetilde{C}_{\ell,0}(\Omega)$  are equal and since these images are  $C_{\ell}^{\natural}(\Omega)$ -invariant, it is sufficient to show that the images of  $g_1$  and  $g_2$  in each Conj' $(\widetilde{C}_{\ell,\ell,-1})(\Omega)$  coincide.

To this end, note that each map

$$P'_L: \operatorname{Conj}'(\widetilde{C}_{\ell,\iota,-1})(\Omega) \to \mathbb{A}^{2g}(\Omega)$$

is injective. This is the statement of [ibid., Lemma 2.6] for the group  $C_{\ell,t,-1}^{der}$ , with  $\Omega$  instead of  $\overline{\mathbb{Q}}$ . Since  $C_{\ell,t,-1}^{der}$  is of type  $A_n$ ,  $C_n$  or  $D_n^{\mathbb{H}}$ , it follows from the remark in the beginning of the proof of [ibid., Lemma 2.6] that the lemma in question is valid in this setting.

## 6. Strictly tractable abelian varieties of types $B_n$ and $D_n^{\mathbb{R}}$

**6.1.** The monodromy in a Mumford-Tate group of type  $B_n$ . With the notation of Section 3.1, we turn to the case where A/F is a strictly tractable abelian variety with Mumford-Tate group  $G_A$  of type  $B_n$ . We will adapt the arguments of Section 5 to this case. The endomorphism algebra L and the group H are defined as in Sections 4.3 and 4.5. As in the previous cases, it follows from Lemma 4.4 that L is a field. Moreover we have  $G_A^{bad} = G_A^{ad}$ . For each prime number  $\ell$ , the monodromy operator  $N_\ell$  and the Frobenius weight cocharacter  $w_\ell$  are defined as before. Each  $w_\ell$  acts on  $V_\ell = V \otimes \mathbb{Q}_\ell$  with at most three eigenvalues. It presents a single weight if and only if A has good reduction, which is the case if and only if  $N_\ell' = 0$ .

Recall the decomposition  $L \otimes \overline{\mathbb{Q}} = \bigoplus_{\iota} \overline{\mathbb{Q}}$  from Section 4.3, where the direct sum is indexed by the maps  $\iota: L \to \overline{\mathbb{Q}}$ . As in Sections 4.3 and 4.5, we consider the resulting decompositions  $V \otimes \overline{\mathbb{Q}} = \bigoplus_{\iota} V_{\iota}$  and  $\mathfrak{h} \otimes \overline{\mathbb{Q}} = \bigoplus_{\iota} \mathfrak{h}_{\iota}$  as well as the embedding  $\mathfrak{g} \otimes \overline{\mathbb{Q}} \hookrightarrow \bigoplus_{\iota} \mathfrak{g}_{\iota}$ . We write  $G_{\iota}$  for the image of  $G_{A/\overline{\mathbb{Q}}}$  in  $GL(V_{\iota})$ . The derived group  $G_{A/\overline{\mathbb{Q}}}^{\text{der}}$  identifies with the product  $\prod_{\iota} G_{\iota}^{\text{der}}$ . Each  $G_{\iota}^{\text{der}}$  is a spin group of type  $B_n$  and its representation on  $V_{\iota}$  is a multiple of the irreducible representation  $V_{\iota}^{\text{irr}}$  of highest weight  $\varpi_n$  and hence of dimension  $d = 2^n$ .

As before, let  $(N'_{\ell,l})_{\iota}$  be the image of  $N'_{\ell}$  in  $\bigoplus_{\iota} \mathfrak{g}_{\iota} \otimes \overline{\mathbb{Q}}_{\ell}$ . Of course,  $N'_{\ell}$  lies in  $\mathfrak{g}^{ss} \otimes \mathbb{Q}_{\ell}$  and  $N'_{\ell,\iota} \in \mathfrak{g}^{ss}_{\iota} \otimes \overline{\mathbb{Q}}_{\ell}$ . The operator  $N'_{\ell}$  is  $L_{\ell}$ -linear so it belongs to  $\mathfrak{h} \otimes \mathbb{Q}_{\ell}$  and it follows from the Corollary 2.7 that its  $H(\overline{\mathbb{Q}}_{\ell})$ -orbit is defined over  $\mathbb{Q}$ . This implies the rank of the projection  $N'_{\ell,\iota}$  is independent of  $\iota$ . If  $N'_{\ell,\iota} = 0$  for some  $\iota$ , then all  $N'_{\ell,\iota}$  are trivial and then the abelian variety A has potentially good reduction. This is the situation treated in [Noot 2009]. In what follows, we will assume that this is not the case. We investigate the possible ranks of the  $N'_{\ell,\iota}$  and the corresponding forms of the cocharacters  $w_{\ell,\iota}$ .

Let  $T_i$  be a maximal torus of  $G_i$  and let  $\widetilde{T}_i = \mathbb{G}_m^d$  be a maximal torus of  $GL(V_i^{\text{irr}})$  containing it. According to [ibid., 2.6.4] we have  $d = 2^n$  and we can assume that  $T_i$  is the image of the application  $\mathbb{G}_m^{n+1} \to \widetilde{T}_i$  given by

$$(\lambda_0, \lambda_1, \ldots, \lambda_n) \mapsto (\lambda_0 \lambda_1^{\varepsilon_1} \cdots \lambda_n^{\varepsilon_n})_{(\varepsilon_1, \ldots, \varepsilon_n) = (\pm 1, \ldots, \pm 1)},$$

where the factors of  $\widetilde{T}_i$  are indexed by *n*-tuples of signs  $(\varepsilon_1, \ldots, \varepsilon_n)$ .

For each  $\iota$ , the Frobenius weight cocharacter  $w_{\ell}$  defines a cocharacter  $w_{\ell,\iota}$  of  $G_{\iota/\overline{\mathbb{Q}}_{\ell}}$ . The monodromy operator  $N'_{\ell,\iota}$  defines an isomorphism between the t and  $t^{-1}$  eigenspaces of  $w_{\ell,\iota}$  acting on  $V_{\iota}$ , so these eigenspaces have the same dimension and it follows that  $w_{\ell,\iota}$  factors through  $G^{\text{der}}_{\iota/\overline{\mathbb{Q}}_{\ell}}$ . Up to conjugation by an element of  $G^{\text{der}}_{\iota}(\overline{\mathbb{Q}}_{\ell})$ , we can assume that  $w_{\ell,\iota}$  factors through  $T_{\iota/\overline{\mathbb{Q}}_{\ell}}$ . It then lifts to a quasicocharacter  $\tilde{w}_{\ell,\iota}$  with values in  $\mathbb{G}_m^{n+1}$ . The filtration on  $V^{\text{irr}}_{\iota}$  defined by  $w_{\ell,\iota}$  has at most three weights and it follows that  $\tilde{w}_{\ell,\iota}$  projects nontrivially to at most two factors of  $\mathbb{G}_m^n$ . Moreover, if it projects nontrivially to two factors, then the two projections must coincide.

The filtration by Frobenius weights coincides with the monodromy filtration and it follows that

- if  $\tilde{w}_{\ell,\iota}$  is trivial, then  $N'_{\ell,\iota} = 0$ ;
- if  $\tilde{w}_{\ell,\iota}$  projects nontrivially to exactly one factor of  $\mathbb{G}_m^n$ , then  $N'_{\ell,\iota}$  is of rank  $2^{n-1}$  (as an endomorphism of  $V_{\iota}^{irr}$ ); and
- if  $\tilde{w}_{\ell,\iota}$  projects nontrivially to exactly two factors of  $\mathbb{G}_m^n$ , then  $N'_{\ell,\iota}$  is of rank  $2^{n-2}$ .

The first possibility is excluded by the hypothesis that  $N'_{\ell} \neq 0$ .

As for the other cases, we consider the adjoint group  $G_{\iota}^{ad}$  of  $G_{\iota}^{der}$ , which is isomorphic to the special orthogonal group  $SO_{2n+1/\overline{\mathbb{Q}}_{\ell}}$ . Let  $W_{\iota}$  be the orthogonal representation of this group, that is, the representation with highest weight  $\overline{\omega}_1$ , and let  $w_{\ell,\iota}^{ad}$  be the projection of the cocharacter  $w_{\ell,\iota}$  to  $G_{\iota}^{ad}$ .

If we are in the second case then  $w_{\ell,t}^{\text{ad}}$  acts on  $W_t$  with eigenvalues t, 1 and  $t^{-1}$  and the eigenspaces for t and for  $t^{-1}$  are 1-dimensional. The relation

$$\operatorname{Ad}(w_{\ell,\iota}(t))(N'_{\ell,\iota}) = t^2 N'_{\ell,\iota}$$

implies that the same relation holds with  $w_{\ell,\iota}^{ad}$  instead of  $w_{\ell,\iota}$ . It follows that the Jordan normal form of the image of  $N'_{\ell,\iota}$  in the orthogonal representation of  $\mathfrak{g}_{\iota}^{ss}$  has one block of size 2 and that all other blocks are of size 1. This is impossible according to [Springer and Steinberg 1970, IV 2.14; Humphreys 1995, 7.11] so the second possibility is excluded.

We study of the conjugacy class of  $(N'_{\ell}, w_{\ell})$  in the third, and only possible, case. The argument above shows that the image of  $N'_{\ell,\ell}$  in the orthogonal representation  $W_{\iota}$  of  $\mathfrak{g}_{\iota}^{ss}$  has two blocks of size 2 and that all other blocks are of size 1. Considering the orthogonal representation  $W_{\iota}$ , we also see that if we take  $X = N'_{\ell,\iota}$  in [Springer and Steinberg 1970, IV 2.19(b)], then we can assume that  $w^{ad}_{\ell,\iota}$  is the inverse of the cocharacter  $\lambda$  of [ibid., IV 2.22]. This remains true after passing to any algebraically closed field  $\Omega \supset \overline{\mathbb{Q}}_{\ell}$  and also after replacing the data  $(N'_{\ell}, w_{\ell})$  by  $\sigma(N'_{\ell}, w_{\ell})$ , where  $\sigma$  is an automorphism of  $\Omega$ . Applying [ibid., IV, §2] in the same way as in Section 5.1, we prove that for any such  $\Omega$  and  $\sigma$ , the pairs  $(N'_{\ell,\iota}, w^{ad}_{\ell,\iota})$ and  $\sigma(N'_{\ell,\iota}, w^{ad}_{\ell,\iota})$  are conjugate under the  $G^{der}_{\iota}(\Omega)$ -action. It follows that this is also the case for  $(N'_{\ell,\iota}, w_{\ell,\iota})$  and  $\sigma(N'_{\ell,\iota}, w_{\ell,\iota})$ .

One shows by the same argument that if  $\ell'$  is a second prime number and if  $\Omega$  is an algebraically closed field containing  $\mathbb{Q}_{\ell}$  and  $\mathbb{Q}_{\ell'}$ , then  $(N'_{\ell}, w_{\ell})$  and  $(N'_{\ell'}, w_{\ell'})$ are  $G_A(\Omega)$ -conjugate. This proves Proposition 5.2 in the case where A/F is a strictly tractable abelian variety of type  $B_n$ .

**6.2.** The Frobenius elements in Mumford–Tate groups of type  $B_n$ . To prove the conjugacy of the Frobenius elements, we adapt the argument used from Section 5.3 to Lemma 5.8. On the one hand, notation is simplified because  $G_A^{\natural ad} = G_A^{ad}$ , but on the other hand, they are complicated by the fact that we need to consider the orthogonal groups  $G_\iota^{ad}$  in order to apply [Springer and Steinberg 1970]. As in Section 5.3, consider the centraliser  $C_\ell \subset G_{A/\mathbb{Q}_\ell}$  of  $(N'_\ell, w_\ell)$  and note that  $\rho'_\ell(\Phi_v) \in \widetilde{C}_\ell(\mathbb{Q}_\ell)$ , where  $\widetilde{C}_\ell$  is the subgroup of  $G_{A/\mathbb{Q}_\ell}$  generated by  $C_\ell$  and the image of  $w_\ell$ . For any fixed embedding  $\iota: L \to \overline{\mathbb{Q}}$ , let  $G_\iota$  be the image of  $G_{A/\overline{\mathbb{Q}}_\ell}$  in GL( $V_\iota$ ). The centraliser  $C_{\ell,\iota} \subset G_{\iota/\overline{\mathbb{Q}}_\ell}$  of  $(N'_{\ell,\iota}, w_{\ell,\iota})$  can be described by projecting it to the adjoint group.

The action of  $G_{\iota}$  on itself by conjugation factors through the adjoint group  $G_{\iota}^{ad} \cong SO_{2n+1}$ . Recall that  $W_{\iota}$  is the orthogonal representation of this group. In view of [ibid., IV §2] and the dimension count carried out in Section 6.1, the centraliser  $C_{\ell,\iota}^{\flat} \subset G_{\iota/\overline{\mathbb{Q}}_{\ell}}^{ad}$  of the pair  $(N_{\ell,\iota}, w_{\ell,\iota}^{ad})$  satisfies

$$C_{\ell,\iota}^{\flat} \cong \operatorname{SL}_{2/\overline{\mathbb{Q}}_{\ell}} \times \operatorname{SO}_{2n-3/\overline{\mathbb{Q}}_{\ell}}.$$
(6.2\*)

Consider the tori  $T_{\iota} \subset \widetilde{T}_{\iota}$  defined in Section 6.1. Up to conjugation, we can assume that  $w_{\ell,\iota}$  factors through  $T_{\iota}$  and that its lift  $\widetilde{w}_{\ell,\iota}$  along  $\mathbb{G}_m^{n+1} \to \widetilde{T}_{\iota}$  is given by

$$\tilde{w}_{\ell,\iota} \colon \mathbb{G}_m \to \mathbb{G}_m^{n+1}, \quad t \mapsto (1, t^{1/2}, t^{1/2}, 1, \dots, 1).$$

The image  $\Delta'' \subset T_{\iota} \subset \widetilde{T}_{\iota}$  of the map  $t \mapsto (1, t^{1/2}, t^{-1/2}, 1, ..., 1)$  then projects to a maximal torus of the factor SL<sub>2</sub> of  $C_{\ell,\iota}^{\flat} \subset G_{\iota/\mathbb{Q}_{\ell}}^{ad}$ . The product  $\Delta' = \mathbb{G}_{m}^{n-2}$  of the last n-2 factors projects to a maximal torus of the factor SO<sub>2n-3</sub>.

The group  $C_{\ell,\iota} \subset G_{\iota}$  is the inverse image of  $C_{\ell,\iota}^{\flat}$ . Its derived group therefore admits an isogeny

$$C_{\ell,\iota}'' \times C_{\ell,\iota}' = \operatorname{SL}_{2/\overline{\mathbb{Q}}_{\ell}} \times C_{\ell,\iota}' \to C_{\ell,\iota}^{\operatorname{der}},$$

where  $C'_{\ell,\iota}$  is a spin group of type  $B_{n-2}$ .

The map from  $\Delta''$  to  $G_{\ell}^{\text{der}}$  factors through  $C_{\ell,\ell}'' \times 1 = \text{SL}_2$  in the product above and the image of  $\Delta''$  in  $C_{\ell,\ell}''$  is a maximal torus. Similarly,  $\Delta'$  maps through  $1 \times C_{\ell}'$ and defines a maximal torus in  $C_{\ell}'$ . Recall from Section 6.1 that  $V_{\ell}$  is a multiple of the spin representation  $V_{\ell}^{\text{irr}}$  of  $G_{\ell}$ . Considering the characters occurring in the representation of  $\Delta'' \times \Delta'$  on  $V_{\ell}^{\text{irr}}$ , one concludes that, as a representation of  $\text{SL}_{2/\overline{\mathbb{Q}}_{\ell}} \times C_{\ell}'$ , the space  $V_{\ell}^{\text{irr}}$  is a tensor product  $V_{\ell}'' \otimes V_{\ell}'$ . Here  $V_{\ell}''$  is a multiple of the direct sum of the standard representation and two copies of the trivial representation of  $\text{SL}_2$  and  $V_{\ell}'$  is the spin representation of  $C_{\ell,\ell}'$ . As representations of  $C_{\ell,\ell}$ , the *t*and  $t^{-1}$ -eigenspaces  $V_{\ell,\ell}^0$  and  $V_{\ell,\ell}^{-2} \subset V_{\ell,\ell}$  of  $w_{\ell,\ell}$  are both isomorphic to a multiple of  $V_{\ell}'$ , so the representation of  $C_{\ell,\ell}'' \times C_{\ell,\ell}'$  on  $V_{\ell,\ell}^0$  identifies  $C_{\ell,\ell}'$  with its image in  $\text{GL}(V_{\ell,\ell}^0)$ .

These observations imply that the isogeny above is in fact an isomorphism  $C''_{\ell,\iota} \times C'_{\ell,\iota} = \operatorname{SL}_{2/\overline{\mathbb{Q}}_{\ell}} \times C'_{\ell,\iota} \cong C^{\operatorname{der}}_{\ell,\iota}$  and hence

$$C_{\ell/\overline{\mathbb{Q}}_{\ell}}^{\mathrm{der}} \cong \prod_{\iota: \ L \to \overline{\mathbb{Q}}_{\ell}} (C_{\ell,\iota}'' \times C_{\ell,\iota}').$$

It also follows that  $C_{\ell}^{\text{der}}$  itself decomposes as a product  $C_{\ell}'' \times C_{\ell}'$  of algebraic groups over  $\mathbb{Q}_{\ell}$ . The group  $C_{\ell}$  is generated by  $C_{\ell}^{\text{der}}$  and the centre of  $G_{A/\mathbb{Q}_{\ell}}$  and  $\widetilde{C}_{\ell}$  is generated by  $C_{\ell}$  and the image of  $w_{\ell}$ . We will show that  $\rho'_{\ell}(\Phi_v)$  lies in  $\widetilde{C}_{\ell}'' \subset \widetilde{C}_{\ell}$ , the subgroup generated by  $C_{\ell}''$  and the centre of  $\widetilde{C}_{\ell}$ .

Indeed, as in Section 5.4, we consider the action of  $\rho'_{\ell}(\Phi_v)$  on the different  $w_{\ell}$ -eigenspaces in  $V_{\ell}$ . It was pointed out in the beginning of the proof of Proposition 5.7 that, since  $\rho'_{\ell}(\Phi_v)$  is weakly neat, it acts on  $V^0_{\ell}$  as multiplication by  $\varepsilon = \pm 1$  and on  $V^{-2}_{\ell}$  as multiplication by  $\varepsilon q$ . This means that  $\rho'_{\ell}(\Phi_v) \in \widetilde{C}''_{\ell}(\mathbb{Q}_{\ell})$ , as claimed.

The group  $C_{\ell,\iota}''$ , which is isomorphic to SL<sub>2</sub>, acts trivially on  $V_{\ell,\iota}^0$  and on  $V_{\ell,\iota}^{-2}$  and  $V_{\ell,\iota}^{-1}$  is a multiple of the standard representation. The centre of  $C_{\ell}$  acts on each  $V_{\iota}$  through a fixed character. Through  $w_{\ell}$ , the group  $\mathbb{G}_m$  acts on  $V_{\ell}^{-2}$ , on  $V_{\ell}^{-1}$  and on  $V_{\ell}^0$  as multiplication by  $t^{-1}$ , by 1 and by t, respectively. For the group  $\widetilde{C}_{\ell}''$  defined above, this discussion implies that the map

$$\widetilde{C}_{\ell}^{\prime\prime} \to \operatorname{GL}(V_{\ell}^{-2}) \times \operatorname{GL}(V_{\ell}^{-1}) \times \operatorname{GL}(V_{\ell}^{0})$$

is injective. As in the proof of Lemma 5.5, the image of  $\rho'_{\ell}(\Phi_v)$  under the map  $P'_L$ , defined in (5.4\*) by taking the *L*-linear characteristic polynomial on  $V_{\ell}^{-1}$ , lies in  $\mathbb{A}^{2g}_{(L)}(\mathbb{Q})$  and is independent of  $\ell$ . We already know that the images of  $\rho'_{\ell}(\Phi_v)$  in  $GL(V_{\ell}^{-2})$  and in  $GL(V_{\ell}^0)$  are rational scalars, independent of  $\ell$ . In the case of an abelian variety of type  $B_n$ , the main theorem now follows using a variant of Lemma 5.8, again using [Noot 2009, 2.5, 2.6]. Note that, as before, the statement of [ibid., 2.6] is valid for any  $\Omega$ , instead of just  $\mathbb{Q}$ , because the group  $\widetilde{C}''_{\ell}$  is of type  $A_1$ .

**6.3.** Abelian varieties of type  $D_n^{\mathbb{R}}$ . The case where the abelian variety *A* is strictly tractable with Mumford–Tate group of type  $D_n^{\mathbb{R}}$  can be treated by analogous arguments. We will just indicate the points where the discussion of Sections 6.1 and 6.2 needs to be modified.

First of all, the quotient of  $G_A$  one has to consider in order apply [Springer and Steinberg 1970] is not the adjoint group, but the intermediate quotient of  $G_{A/\overline{\mathbb{Q}}_{\ell}}$  for which the simple factors  $G_{\ell}^{\flat}$  are groups of the form  $SO_{2n}$ . Also, A is not necessarily simple in this case. If it is not, then the endomorphism algebra L is of the form  $L = L' \times L'$ , where L' is a number field; see Lemma 4.4.

We now follow the proof of [Noot 2009, Théorème 2.4] for this type. For each  $\iota: L \to \overline{\mathbb{Q}}$ , let  $V_{\iota}$  be the direct factor of  $V \otimes \overline{\mathbb{Q}}$  on which L acts through  $\iota$ . The group  $G_{A/\overline{\mathbb{Q}}}^{\text{der}}$  acts on  $V_{\iota}$  through a single direct factor  $G_{\iota}^{\text{der}}$ , but for  $n \ge 4$  this factor does not act faithfully on  $V_{\iota}$ . In fact  $V_{\iota}$  is a multiple of a semispin representation  $V_{\iota}^{\text{irr}}$  of  $G_{\iota}$ , with highest weight  $\overline{\varpi}_{n-1}$  say. For  $\iota = \iota^+$ , there is a  $\iota^-: L \to \overline{\mathbb{Q}}$  such that  $V_{\iota^-}$  is a multiple of the other semispin representation  $V_{\iota^-}^{\text{irr}}$  of  $G_{\iota}$ , with highest weight  $\overline{\varpi}_n$ . The representation of  $G_{A/\overline{\mathbb{Q}}}$  on  $V_{\iota^+} \oplus V_{\iota^-}$  restricts to a faithful representation of  $G_{\iota}^{\text{der}}$ . We redefine  $G_{\iota}$  as the image of  $G_{A/\overline{\mathbb{Q}}}^{\text{der}}$  in  $GL(V_{\iota^+}) \times GL(V_{\iota^-})$ .

If *L* is a field, then it is a CM field and the involution  $\iota^+ \mapsto \iota^-$  on the set of maps  $L \to \overline{\mathbb{Q}}$  defined by this construction is given by the composition with the complex conjugation on *L*. If  $L = L' \times L'$  is a product of two fields, then  $\iota^-$  is the composite of  $\iota^+$  with the involution exchanging the factors. Using the spin representation  $V_{\iota^+}^{\text{irr}} \oplus V_{\iota^-}^{\text{irr}}$  instead of  $V_{\iota}^{\text{irr}}$ , the arguments of Section 6.1 and hence the proof of Proposition 5.2 carry over to this case.

Where the discussion of Section 6.2 is concerned, the analogue of (6.2\*) states that the centraliser  $C_{\ell,\iota}^{\flat}$  of  $(N_{\ell,\iota}, w_{\ell,\iota}^{ad})$  in  $G_{\iota}^{\flat}$  is given by

$$C_{\ell,\iota}^{\flat} \cong \mathrm{SL}_{2,\overline{\mathbb{Q}}_{\ell}} \times \mathrm{SO}_{2n-4}$$

Once again,  $C_{\ell,\iota} \subset G_{\iota}$  is the inverse image of  $C_{\ell,\iota}^{\flat}$  and there is an isogeny

$$C_{\ell,\iota}'' \times C_{\ell,\iota}' \cong \operatorname{SL}_{2/\mathbb{Q}_{\ell}} \times C_{\ell,\iota}' \to C_{\ell,\iota}^{\operatorname{der}}$$

Here the group  $C'_{\ell,\iota}$  is a spin group of type  $D_{n-2}$ . Similarly to the previous case, one shows that  $V_{\iota^+}^{irr} \oplus V_{\iota^-}^{irr}$  is of the form  $V''_{\iota} \otimes V'_{\iota}$ , where  $V''_{\iota}$  is a multiple of the direct sum of the standard representation and two copies of the trivial representation of SL<sub>2</sub> and  $V'_{\iota}$  is the spin representation of  $C'_{\iota}$ . We prove once again that

$$C_{\ell,\iota}^{\mathrm{der}} = C_{\ell,\iota}'' \times C_{\ell,\iota}'$$

and we define  $\widetilde{C}_{\ell}$  and  $\widetilde{C}''_{\ell}$  as before. As in Section 6.2 one has  $\rho'_{\ell}(\Phi_v) \in \widetilde{C}''_{\ell}(\mathbb{Q}_{\ell})$ . For each pair  $\iota^+$ ,  $\iota^-$  as above, the image of  $C''_{\ell}$  in  $C''_{\ell,\iota^+} \times C''_{\ell,\iota^-}$  is the graph of an isomorphism. Each  $C''_{\ell}$  is again a group of type  $A_1$ , acting on the 1-eigenspace  $V_{l^+}^{-1} \oplus V_{l^-}^{-1}$  for  $w_\ell$  as a multiple of the standard representation. The argument can be completed as in Section 6.2.

#### 7. The proof of the main theorem, Theorem 3.6

**Proposition 7.1.** Assume that we are in the situation of Theorem 3.6 and that the variety A is tractable. Then there is a finite extension F' of F such that Theorem 3.6 holds for  $A_{/F'}$ .

*Proof.* The Proposition 5.7 and the results of Section 6 prove the proposition in the case where A is strictly tractable.

If *A* is tractable then there exists a finite extension  $F' \supset F$ , strictly tractable abelian varieties  $A_1, \ldots, A_m/F'$  and an isogeny  $A_{/F'} \sim \prod_{i=1}^m A_i$  such that the inclusion  $f: G_A \to \prod_{i=1}^m G_{A_i}$  induces an isomorphism  $G_A^{\text{der}} \cong \prod_{i=1}^m G_{A_i}^{\text{der}}$ . In that case there are isomorphisms

$$\mathfrak{g}_A^{\mathrm{ss}} \cong \bigoplus_{i=1}^m \mathfrak{g}_{A_i}^{\mathrm{ss}} \quad \text{and} \quad G_A^{\natural} \cong \prod_{i=1}^m G_{A_i}^{\natural}.$$

For the induced map

$$f_{\ell} \colon G_A(\mathbb{Q}_{\ell}) \to \prod_{i=1}^m G_{A_i}(\mathbb{Q}_{\ell}),$$

one has  $f_{\ell} \circ \rho_{A,\ell} = (\rho_{A_i,\ell})_{i=1,...,m}$ , so the tangent map to  $f_{\ell}$  sends the monodromy operator  $N'_{\ell} \in \mathfrak{g}^{ss}_A \otimes \mathbb{Q}_{\ell}$  to the *m*-tuple in  $\bigoplus_{i=1}^m \mathfrak{g}^{ss}_{A_i} \otimes \mathbb{Q}_{\ell}$  of the monodromy operators associated to the  $A_i$ . This obviously implies that  $f_{\ell} \circ \rho'_{A,\ell} = (\rho'_{A_i,\ell})_{i=1,...,m}$ . The statement for  $A_{/F'}$  therefore results immediately from the corresponding statements for the  $A_i$ .

**7.2.** *Preliminaries to the proof of Theorem 3.6.* We use the method of the proof of [Noot 2009, Théorème 1.8] in Section 3 of that paper. After fixing the notation we will indicate an omission, pointed out by Abhijit Laskar, in [Noot 2009] and explain how to complete the argument.

In what follows, the notation and the hypotheses of Section 3.1 and of Theorem 3.6 are in force, so A has semistable reduction at v and the image  $\rho_{\ell}(\Phi_v)$  of the arithmetic Frobenius is weakly neat. However, since the argument involves auxiliary abelian varieties, we write  $(\rho'_{A,\ell}, N'_{A,\ell})$  for the representation of 'W<sub>v</sub> associated to  $A_{/F_v}$ .

By [Noot 2006, §2 and Corollary 3.2] of that paper, there is a tractable abelian variety  $B_{/\bar{F}}$  such that  $B_{/\mathbb{C}}$  provides a weak Mumford–Tate lift of  $A_{/\mathbb{C}}$ . Following [Noot 2009, §3], this implies that there exists an abelian variety of CM-type  $C_{/\bar{F}}$  such that  $A_{/\bar{F}}$  belongs to the category of absolute Hodge motives generated by  $B_{/\bar{F}}$  and  $C_{/\bar{F}}$ . This fact determines a morphism of Mumford–Tate groups  $\pi : G_{B \times C} \to G_A$
but, contrary to what is stated in [ibid., §3], not a morphism  $G_B \times G_C \rightarrow G_A$ . In fact,  $G_{B \times C}$  is a closed subgroup of the product  $G_B \times G_C$  and the inclusion identifies the derived groups. This means that the diagram considered in the proof of [ibid., Théorème 1.8] has to be replaced by the diagram

which commutes for a sufficiently large finite extension F' of F. None of the above depends on v but this will not play any role in what follows.

**7.3.** Addendum to the proof of [Noot 2009, Théorème 1.8]. Recall that the statement of the theorem in question is essentially the special case of the main theorem of this paper where *A* has good reduction. In [Noot 2009] it is formulated in terms of the variety of geometric conjugacy classes of the Mumford–Tate group. We have to prove that there exists a conjugacy class  $\operatorname{Cl}_A \operatorname{Fr}_v \in \operatorname{Conj}'(G_A)(\mathbb{Q})$  containing the image of  $\rho_{A,\ell}(\Phi_v^{-1})$  of any  $\ell$  with  $v(\ell) = 0$ . Here  $\operatorname{Conj}'(G_A)(\mathbb{Q})$  containing the adjoint action of  $G_A^{\ddagger ad}$ . We refer to [ibid., 1.5] for the construction of a natural model  $\operatorname{Conj}'(G_A)$  over  $\mathbb{Q}$ . Assume for the moment that [ibid., Theorem 1.8] holds for  $B \times C/F'$ , where F' is a sufficiently big finite extension of F and v' an extension of the valuation v to F'. We then obtain a conjugacy class  $\operatorname{Cl}_{B\times C}\operatorname{Fr}_{v'} \in \operatorname{Conj}'(G_{B\times C})(\mathbb{Q})$  and its image  $\operatorname{Cl}_A \operatorname{Fr}_{v'} \in \operatorname{Conj}'(G_A)(\mathbb{Q})$  fulfils the statement [ibid., Theorem 1.8] for  $A_{/F'}$ . The proof of Theorem 1.8 there then applies and it follows that the theorem also holds for A/F.

It remains to construct  $\operatorname{Cl}_{B\times C}\operatorname{Fr}_{v'}$ . As  $G_C$  is a torus,  $\operatorname{Conj}'(G_{B\times C})/\overline{\mathbb{Q}}$  and  $\operatorname{Conj}'(G_B \times G_C)/\overline{\mathbb{Q}}$  are the quotients of  $G_{B\times C}/\overline{\mathbb{Q}}$  and of  $G_{B}/\overline{\mathbb{Q}} \times G_C/\overline{\mathbb{Q}}$  for the adjoint action by the same group, denoted  $\operatorname{Aut}'(G_B)$  in [Noot 2009, 1.5] and  $G_B^{\natural ad}$  in Section 3.3 of this paper. If  $T_{B\times C} \subset T_B \times G_C$  denote maximal tori of  $G_{B\times C}$  and of  $G_B \times G_C$ , then  $\operatorname{Conj}'(G_{B\times C})$  and  $\operatorname{Conj}'(G_B \times G_C)$  are also quotients of these tori by the finite group  $\widetilde{W}$  of [ibid., 1.6]. The group  $\widetilde{W}$  is an extension of a finite group of outer automorphisms by the Weyl group of  $G_B^{der}$ . We claim that the closed immersion  $T_{B\times C} \subset T_B \times G_C$  induces a closed immersion on the quotients for the  $\widetilde{W}$ -action.

To justify the claim, assume that  $R \to S$  is a surjective morphism of  $\mathbb{Q}$ -algebras with  $\widetilde{W}$  action. Let  $b \in S^{\widetilde{W}}$  and assume that  $a \in R$  maps to b. The average of the elements of the  $\widetilde{W}$ -orbit of a then is an element of  $R^{\widetilde{W}}$  mapping to b. It follows that  $R^{\widetilde{W}} \to S^{\widetilde{W}}$  is also surjective.

#### Rutger Noot

As B is tractable, Thórème 2.4 of [Noot 2009] provides a conjugacy class

$$(\operatorname{Cl}_B\operatorname{Fr}_{v'},\operatorname{Cl}_C\operatorname{Fr}_{v'})\in\operatorname{Conj}'(G_B\times G_C)(\mathbb{Q})$$

containing  $(\rho_{B,\ell}(\Phi_{v'}^{-1}), \rho_{C,\ell}(\Phi_{v'}^{-1}))$  for any  $\ell \neq p$ . As  $(\rho_{B,\ell}, \rho_{C,\ell})$  factors through  $G_{B\times C}$  for all  $\ell$ , it follows that  $(\operatorname{Cl}_B \operatorname{Fr}_{v'}, \operatorname{Cl}_C \operatorname{Fr}_{v'}) \in \operatorname{Conj}'(G_{B\times C})(\mathbb{Q})$ . It is obviously the class  $\operatorname{Cl}_{B\times C} \operatorname{Fr}_{v'}$  we had to construct.

*Proof of Theorem 3.6.* We take up the thread of the proof of Theorem 3.6 by considering the diagram (7.2\*). In this diagram, the map  $G_{B\times C} \hookrightarrow G_B \times G_C$  induces an isomorphism on the derived groups and it follows that  $G_{B\times C}^{\natural ad} = G_B^{\natural ad}$  and that both the subgroup  $G_{B\times C/\overline{\mathbb{Q}}} \subset (G_B \times G_C)/\overline{\mathbb{Q}}$  and the Lie subalgebra  $\mathfrak{g}_{B\times C} \otimes \overline{\mathbb{Q}} \subset (\mathfrak{g}_B \oplus \mathfrak{g}_C) \otimes \overline{\mathbb{Q}}$  are stable under the adjoint action of  $G_B^{\natural ad}$ . Taking F' big enough and fixing an extension v' of the valuation, we can assume, by Proposition 7.1, that the conclusion of the main theorem holds for B. By [Noot 2009, Corollaire 2.2] we can also assume that it is valid for C. This implies that the theorem is true for  $(B \times C)/F'$ .

Consider the statement of Theorem 3.6 for the representation of  ${}^{\prime}W_{F'_{v'}}$  associated to  $A_{F'}$ . The monodromy operators are unaffected by passing from A to  $A_{F'}$ , whereas  $\Phi_v$ , and hence the  $\rho'_{A,\ell}(\Phi_v)$ , are replaced by their f-th powers, where f is the residue degree of the extension  $F'_{v'}/F_v$ . This exponent is independent of  $\ell$ .

The variety *C* has potentially good reduction at v', so for every prime number  $\ell$ , the monodromy operator  $N'_{A,\ell} \in \mathfrak{g}_A \otimes \mathbb{Q}_\ell$  is the image of

$$(N'_{B,\ell}, 0) \in \mathfrak{g}_{B \times C} \otimes \mathbb{Q}_{\ell} \subset (\mathfrak{g}_B \oplus \mathfrak{g}_C) \otimes \mathbb{Q}_{\ell}$$

under the tangent map to  $\pi$ . Here  $N'_{B,\ell}$  is the monodromy operator associated to  $B_{/F_v}$ . We have made use of the fact, expressed by the diagram (7.2\*), that the product of the  $\ell$ -adic Galois representations associated to B and C factors through  $G_{B\times C}(\mathbb{Q}_{\ell})$ .

Similarly,

$$\rho_{A,\ell}'(\Phi_{v'}) = \pi(\rho_{B,\ell}'(\Phi_{v'}), \rho_{C,\ell}'(\Phi_{v'})),$$

which makes sense since  $(\rho'_{B,\ell}(\Phi_{v'}), \rho'_{C,\ell}(\Phi_{v'})) \in G_{B \times C}(\mathbb{Q}_{\ell}) \subset (G_B \times G_C)(\mathbb{Q}_{\ell})$ . As the theorem holds for  $(B \times C)_{/F'}$ , it follows that the theorem is true for  $A_{/F'}$ .

Now return to the original field F. Let  $\Omega \supset \mathbb{Q}_{\ell}$  be an algebraically closed field and  $\sigma \in \operatorname{Aut}(\Omega)$ . By what we just proved, the images of  $(N'_{A,\ell}, \rho'_{\ell}(\Phi_v^f))$  and  $\sigma(N'_{A,\ell}, \rho'_{\ell}(\Phi_v^f))$  in  $\mathfrak{g}_A \otimes \Omega \times G_A(\Omega)$  are conjugate by an element  $g \in G_A^{\natural}(\Omega)$ . Thus  $N'_{A,\ell} = \operatorname{Ad}(g)(\sigma(N'_{A,\ell}))$  and we will show that  $\rho'_{\ell}(\Phi_v) = g\sigma(\rho'_{\ell}(\Phi_v))g^{-1}$  as well. Indeed, applying [Raynaud 1994, 4.2] as in Section 3.8, we obtain a strict 1-motive  $M'/F_v$  and a system of  $\Gamma_{F_v}$ -equivariant isomorphisms  $V_{\ell}(A_{/F_v}) \cong V_{\ell}(M')$ . By Proposition 1.8, the characteristic polynomials of  $\rho'_{\ell}(\Phi_v)$  and  $\sigma(\rho'_{\ell}(\Phi_v))$  acting on  $V_{\ell}(A)$  coincide. This common polynomial is also the characteristic polynomial of  $g\sigma(\rho'_{\ell}(\Phi_v))g^{-1}$ . As we already know that  $\rho'_{\ell}(\Phi_v^f) = g\sigma(\rho'_{\ell}(\Phi_v^f))g^{-1}$ , the equality  $\rho'_{\ell}(\Phi_v) = g\sigma(\rho'_{\ell}(\Phi_v))g^{-1}$  follows from Lemma 7.4 below.

Similarly, let  $\Omega$  be an algebraically closed field containing  $\mathbb{Q}_{\ell}$  and  $\mathbb{Q}_{\ell'}$ . We know that the images of the pairs  $(N'_{A,\ell}, \rho'_{\ell}(\Phi^f_v))$  and  $(N'_{A,\ell'}, \rho'_{\ell'}(\Phi^f_v))$  are conjugate by some  $g \in G^{\sharp}_{A}(\Omega)$ . Again by Proposition 1.8, the characteristic polynomials of  $\rho'_{\ell}(\Phi_v)$  and  $\rho'_{\ell'}(\Phi_v)$ ) coincide so Lemma 7.4 implies that  $\rho'_{\ell}(\Phi_v) = \rho'_{\ell'}(\Phi_v)$ ). This proves the theorem for A.

**Lemma 7.4.** Assume that  $\Omega$  is an algebraically closed field, d > 0 is an integer and that  $x, y \in GL_d(\Omega)$  are two semisimple and weakly neat elements. Assume that  $x^f = y^f$  for some integer f and that x and y have the same characteristic polynomial. Then x = y.

Proof. This is a variant of [Noot 2009, Proposition 3.2].

For any semisimple element  $z \in GL_d(\Omega)$ , let  $T_z \subset GL_d$  be the torus acting by scalar multiplication on each eigenspace of z. Up to conjugation, z is a point of the diagonal torus  $\mathbb{G}_m^d \subset GL_d$  and, writing  $t_1, \ldots, t_d$  for the coordinates on  $\mathbb{G}_m^d$  and  $z = (z_1, \ldots, z_d)$ , one then has

$$T_{z} = \{(t_{1}, \dots, t_{d}) \in \mathbb{G}_{m}^{d} \mid t_{i} = t_{j} \text{ if } z_{i} = z_{j}\}.$$
(7.4\*)

Note that for every positive integer *n* one has  $T_{z^n} \subset T_z$  and that this inclusion is an equality if *z* is weakly neat.

With this notation we prove the lemma. As *x* and *y* are weakly neat and satisfy  $x^f = y^f$ , we get  $T_x = T_{x^f} = T_{y^f} = T_y$ . This implies in particular that  $y \in T_x(\Omega)$ . We can assume that *x* lies in the diagonal torus  $\mathbb{G}_m^d \subset \operatorname{GL}_d$  and we write  $x = (x_1, \ldots, x_d) \in \mathbb{G}_m^d(\Omega)$ . The fact that *x* and *y* have the same characteristic polynomial implies that there is a permutation  $\sigma \in S_d$  of the factors of the product  $\mathbb{G}_m^d$  such that  $y = \sigma(x)$ . We have  $x^f = y^f = \sigma(x^f)$  and, considering the equations (7.4\*) for  $T_{x^f}$ , it follows that  $\sigma|_{T_{x^f}} = \operatorname{id}$ . Since  $T_{x^f} = T_x$  we conclude that  $x = \sigma(x) = y$  as claimed.

#### References

- [Deligne 1974b] P. Deligne, "Théorie de Hodge, III", *Inst. Hautes Études Sci. Publ. Math.* **44** (1974), 5–77. MR 58 #16653b Zbl 0237.14003
- [Deligne 1979] P. Deligne, "Variétés de Shimura: Interprétation modulaire, et techniques de construction de modèles canoniques", pp. 247–289 in *Automorphic forms, representations and L-functions*

<sup>[</sup>Deligne 1973] P. Deligne, "Les constantes des équations fonctionnelles des fonctions *L*", pp. 501– 597 in *Modular functions of one variable, II* (Antwerp, 1972), edited by P. Deligne and W. Kuyk, Lecture Notes in Math. **349**, Springer, Berlin, 1973. MR 50 #2128 Zbl 0271.14011

<sup>[</sup>Deligne 1974a] P. Deligne, "La conjecture de Weil, I", *Inst. Hautes Études Sci. Publ. Math.* **43** (1974), 273–307. MR 49 #5013 Zbl 0287.14001

(Corvallis, OR, 1977), vol. 2, edited by A. Borel and W. Casselman, Proc. Sympos. Pure Math. 23, Amer. Math. Soc., Providence, R.I., 1979. MR 81i:10032 Zbl 0437.14012

- [Deligne 1980] P. Deligne, "La conjecture de Weil, II", *Inst. Hautes Études Sci. Publ. Math.* **52** (1980), 137–252. MR 83c:14017 Zbl 0456.14014
- [Deligne 1982] P. Deligne, "Motifs et groupes de Taniyama", pp. 261–279 in *Hodge cycles, motives, and Shimura varieties*, edited by P. Deligne et al., Lecture Notes in Mathematics **900**, Springer, Berlin, 1982. Zbl 0499.14001
- [Deligne and Milne 1982] P. Deligne and J. S. Milne, "Tannakian categories", pp. 101–228 in *Hodge cycles, motives, and Shimura varieties*, edited by P. Deligne et al., Lecture Notes in Mathematics **900**, Springer, Berlin, 1982. Zbl 0477.14004
- [Fontaine 1994] J.-M. Fontaine, "Représentations *l*-adiques potentiellement semi-stables", pp. 321–347 in *Périodes p-adiques* (Bures-sur-Yvette, 1988), Astérisque **223**, Société Mathématique de France, Paris, 1994. MR 95k:14031 Zbl 0873.14020
- [Grothendieck et al. 1972] A. Grothendieck, M. Raynaud, and D. S. Rim, *Groupes de monodromie en géométrie algébrique, I (SGA 7)*, Lecture Notes in Mathematics **288**, Springer, Berlin, 1972. MR 50 #7134
- [Humphreys 1995] J. E. Humphreys, *Conjugacy classes in semisimple algebraic groups*, Mathematical Surveys and Monographs **43**, Amer. Math. Society, Providence, RI, 1995. MR 97i:20057 Zbl 0834.20048
- [Illusie 1994] L. Illusie, "Autour du théorème de monodromie locale", pp. 9–57 in *Périodes p-adiques* (Bures-sur-Yvette, 1988), Astérisque 223, Société Mathématique de France, Paris, 1994. MR 95k:14032 Zbl 0837.14013
- [Ito 2004] T. Ito, "Weight-monodromy conjecture for certain threefolds in mixed characteristic", *Int. Math. Res. Not.* **2004**:2 (2004), 69–87. MR 2005b:14033 Zbl 1090.14004
- [Laskar 2011] A. Laskar,  $\ell$ -independence for a system of motivic representations, thesis, Université de Strasbourg, 2011, available at http://tinyurl.com/bpg6gen.
- [Milne 1986] J. S. Milne, "Abelian varieties", pp. 103–150 in *Arithmetic geometry* (Storrs, CT, 1984), edited by G. Cornell and J. H. Silverman, Springer, New York, 1986. MR 861974 Zbl 0604.14028
- [Mumford 1969] D. Mumford, "A note of Shimura's paper "Discontinuous groups and abelian varieties"", *Math. Ann.* **181** (1969), 345–351. MR 40 #1400 Zbl 0169.23301
- [Mumford 1970] D. Mumford, *Abelian varieties*, Oxford University Press, 1970. MR 44 #219 Zbl 0223.14022
- [Noot 1995] R. Noot, "Classe de conjugaison du Frobenius des variétés abéliennes à réduction ordinaire", Ann. Inst. Fourier (Grenoble) 45:5 (1995), 1239–1248. MR 96m:14064 Zbl 0834.14026
- [Noot 2006] R. Noot, "Lifting systems of Galois representations associated to abelian varieties", *J. Ramanujan Math. Soc.* **21**:4 (2006), 299–342. MR 2008a:11067 Zbl 1174.11050
- [Noot 2009] R. Noot, "Classe de conjugaison du Frobenius d'une variété abélienne sur un corps de nombres", *J. Lond. Math. Soc.* (2) **79**:1 (2009), 53–71. MR 2010e:14041 Zbl 1177.14084
- [Ochiai 1999] T. Ochiai, "*l*-independence of the trace of monodromy", *Math. Ann.* **315**:2 (1999), 321–340. MR 2000i:14024 Zbl 0980.14014
- [Rapoport and Zink 1982] M. Rapoport and T. Zink, "Über die lokale Zetafunktion von Shimuravarietäten. Monodromiefiltration und verschwindende Zyklen in ungleicher Charakteristik", *Invent. Math.* **68**:1 (1982), 21–101. MR 84i:14016 Zbl 0498.14010
- [Raynaud 1994] M. Raynaud, "1-motifs et monodromie géométrique", pp. 295–319 in *Périodes p-adiques* (Bures-sur-Yvette, 1988), Astérisque 223, Société Mathématique de France, Paris, 1994.
   MR 95h:14020 Zbl 0830.14001

- [Serre 1968] J.-P. Serre, *Abelian l-adic representations and elliptic curves*, W. A. Benjamin, New York, 1968. MR 41 #8422 Zbl 0186.25701
- [Serre 1994] J.-P. Serre, "Propriétés conjecturales des groupes de Galois motiviques et des représentations *l*-adiques", pp. 377–400 in *Motives* (Seattle, WA, 1991), edited by U. Jansen et al., Proc. Sympos. Pure Math. **55**, Amer. Math. Soc., Providence, RI, 1994. MR 95m:11059 Zbl 0812.14002
- [Serre 2000] J.-P. Serre, "Letters to Ribet, 1-1-1981 and 29-1-1981", pp. 1–20 in *Œuvres: Collected papers*, vol. IV, Springer, Berlin, 2000.
- [Serre and Tate 1968] J.-P. Serre and J. Tate, "Good reduction of abelian varieties", *Ann. of Math.* (2) **88** (1968), 492–517. MR 38 #4488 Zbl 0172.46101
- [Shimura and Taniyama 1961] G. Shimura and Y. Taniyama, *Complex multiplication of abelian* varieties and its applications to number theory, Publ. Math. Soc. Japan 6, Math. Soc. Japan, Tokyo, 1961. MR 23 #A2419 Zbl 0112.03502
- [Springer and Steinberg 1970] T. A. Springer and R. Steinberg, "Conjugacy classes", pp. 167–266 in *Seminar on algebraic groups and related finite groups* (Princeton, NJ, 1968/69), Lecture Notes in Mathematics **131**, Springer, Berlin, 1970. MR 42 #3091 Zbl 0249.20024

http://www-irma.u-strasbg.fr/~noot/

Communicated by Hélène Esnault Received 2009-08-10 Revised 2012-01-17 Accepted 2012-03-25 noot@math.unistra.fr Institut de Recherche Mathématique Avancée, UMR 7501, Université de Strasbourg and CNRS, 7 rue René Descartes, 67084 Strasbourg, France





# Fourier–Jacobi coefficients of Eisenstein series on unitary groups

## Bei Zhang

This paper studies the Fourier–Jacobi expansions of Eisenstein series on U(3, 1). I relate the Fourier–Jacobi coefficients of the Eisenstein series with special values of *L*-functions. This relationship can be applied to verify the existence of certain Eisenstein series on U(3, 1) that do not vanish modulo *p*. This is a crucial step towards one divisibility of the main conjecture for  $GL_2 \times K^{\times}$  using the method of Eisenstein congruences.

1.	Introduction	283
2.	Eisenstein series on $U(3, 1)$	288
3.	Theta functions	294
4.	Fourier-Jacobi coefficients of Eisenstein series	303
5.	Applications to Eisenstein series nonvanishing modulo p	324
Appendix A. Proof of the Siegel–Weil formula for $(U(2, 2), U(1))$		327
App	pendix B. Integral theta functions	332
Acknowledgments		336
References		336

### 1. Introduction

*Eisenstein congruence and Iwasawa main conjecture.* Eisenstein series have been intensively used in constructions of *p*-adic *L*-functions and in the Iwasawa main conjectures. Ribet [1976] used the congruences between Eisenstein series and cusp forms to prove the converse of the Herbrand theorem. This idea was extended to congruences between *p*-adic families of modular forms, which was successfully used, first by Mazur and Wiles [1984] to prove the main conjecture for real abelian fields, then by Wiles [1990] for all totally real fields. Subsequently, Skinner and Urban used this technique to study the main conjectures for the motives attached to modular forms; see [Urban 2001; Urban 2006; Skinner and Urban 2012].

MSC2010: primary 11F55; secondary 11F30, 11F27, 11R23.

*Keywords:* Iwasawa main conjecture, unitary groups, Eisenstein series, Fourier–Jacobi expansion, doubling method, nonvanishing modulo *p*.

In the ongoing joint project with Hsieh, we want to apply Eisenstein congruences to the following main conjecture for elliptic curves. Take an imaginary quadratic field K in which p splits. Fix an embedding  $i_{\infty} : \overline{\mathbb{Q}} \to \mathbb{C}$  and an isomorphism  $i : \mathbb{C} \cong \mathbb{C}_p$ . Let

$$\rho_E : \operatorname{Gal}(\overline{\mathbb{Q}}/\mathbb{Q}) \to \operatorname{GL}(H^1_{\text{et}}(E, \mathbb{Q}_p))$$

be the *p*-adic Galois representation associated to an elliptic curve *E* over  $\mathbb{Q}$  with good ordinary reduction at *p*. Let  $\eta : \mathbb{A}_K^{\times}/K^{\times} \to \mathbb{C}^{\times}$  be a Hecke character with the infinity type (k, 0), where *k* is an integer and k > 1. Put  $\mathbb{O} = \mathbb{Z}_p[\operatorname{Im} \eta]$ , the ring of values of  $\eta$ . Let  $K_{\max}$  be the unique  $\mathbb{Z}_p^2$ -extension of *K*, and let  $\Gamma = \operatorname{Gal}(K_{\max}/K)$ . Define  $\Lambda = \mathbb{O}[[\Gamma]]$  the Iwasawa algebra of  $\Gamma$ . Given  $\Psi$  a  $\Lambda$ -valued character of  $\operatorname{Gal}(\overline{\mathbb{Q}}/K)$  that interpolates  $\eta$ , there exists a unique element  $L_p(\rho_E \otimes \Psi)$  in  $\Lambda$ interpolating  $((2\pi i)^{2k}/\Omega_K^{2k})L_K(0, E \otimes \eta)$ , where  $\Omega_K$  is the CM-period associated to *K*. (About the precise normalization of  $L_K(E \otimes \eta, 0)$ , one may check [Hsieh 2011b, Definition 1]. Especially, since the weight of the CM form associated to  $\eta$  is greater than 2, the period appearing in the denominator of the normalization of  $L_K(E \otimes \eta, 0)$  is not the period of *E*, but the CM period attached to  $\eta$  and *K*.) Let  $\mathfrak{p}$  and  $\overline{\mathfrak{p}}$  be the two primes in *K* above *p*. Let  $K_S$  be the maximal unramified extension of *K* outside *S*, where

 $S = \{\mathfrak{p}, \overline{\mathfrak{p}}\} \cup \{v \text{ finite } | \eta_v \text{ is ramified, or } E \text{ has a bad reduction at } v\} \cup \{\infty\}.$ 

Use  $M^*$  to denote the Pontryagin dual of a  $\Lambda$ -module M. Define the nonprimitive  $\Lambda$ -adic Selmer group to  $\rho_E \otimes \Psi$  by

$$\operatorname{Sel}_{K}(\rho_{E}\otimes\Psi) = \operatorname{ker}\{H^{1}(\operatorname{Gal}(K_{S}/K), T\otimes\Lambda^{*}) \to H^{1}(I_{\bar{\mathfrak{p}}}, T_{\bar{\mathfrak{p}}}\otimes\Lambda^{*})\},\$$

where  $T = H_{\text{et}}^1(E, \mathbb{Z}_p)$ ,  $I_{\bar{\mathfrak{p}}}$  is the inertia group at  $\bar{\mathfrak{p}}$ ; see [Hsieh 2011b]. The following conjecture is formulated in [Greenberg 1994]:

**Conjecture 1.1.** Sel<sub>*K*</sub>( $\rho_E \otimes \Psi$ ) is cotorsion over  $\Lambda$ , and for any height 1 prime *P*,

$$\operatorname{ord}_{P} L_{p}(\rho_{E} \otimes \Psi) = l_{P}(\operatorname{Sel}_{K}(\rho_{E} \otimes \Psi)),$$

where  $l_P(\operatorname{Sel}_K(\rho_E \otimes \Psi)) = \operatorname{length}_{\Lambda_P}(\operatorname{Sel}_K(\rho_E \otimes \Psi)^* \otimes_{\Lambda} \Lambda_P).$ 

**Remark 1.2.** Though they have similar formulations, the conjecture above is different from the main conjecture of elliptic curves studied in [Skinner and Urban 2012], because the specializations of  $\Psi$  are different. Our  $\Psi$  interpolates Hecke characters over *K* with the infinity type (*k*, 0) for an integer *k* > 1; the infinity types in [ibid.] are different. In addition, we consider different  $L_p(\rho_E \otimes \Psi)$  and  $\text{Sel}_K(\rho_E \otimes \Psi)$ .

Nonvanishing modulo p of Eisenstein series. Both for Ribet's original argument and for various cases of Iwasawa main conjectures, we crucially need to guarantee the Eisenstein series (or the p-adic Eisenstein series) used in the proofs does not vanish modulo p, for the naive reason that this is sufficient to deduce that the congruent cusp form is nontrivial. A more technical reason is that it is necessary to show that the constant term of this Eisenstein series divides the Eisenstein ideal, which is the first step towards one divisibility of the main conjecture.

For main conjectures of different cuspidal representations, Eisenstein series on different reductive groups are constructed so that the *p*-adic *L*-functions in the main conjectures interpolate the constant terms of the Eisenstein series. There is no consistent way to argue nonvanishing modulo *p* of an Eisenstein series on a general reductive group. This question has been one of the obstacles of Iwasawa theory for *L*-functions of higher degrees. Urban [2006] argued that an Eisenstein–Klingen series on GSp(4) is nontrivial modulo *p* in the following way: Find an algebraic linear combination of its Fourier coefficients (this is essentially a period integral of the Eisenstein series) that turns out to be a special *L*-value, and use Vatsal's result [2003] that this *L*-value does not vanish modulo *p*. A similar argument is used as well by Skinner and Urban [2012] to show an Eisenstein series on U(2, 2) does not vanish modulo *p*.

To solve the main conjecture in our ongoing project, we need to construct an Eisenstein series on U(3, 1). To attack the question about nonvanishing modulo p of this Eisenstein series, the method is a bit different from other cases. Because U(3, 1) is nonquasisplit, the Eisenstein series has a Fourier–Jacobi expansion where the coefficients are theta functions.

*Results on Fourier–Jacobi coefficients.* From now on, let *K* be an imaginary quadratic extension of a totally real field *F*. Assume the degree of *F* over  $\mathbb{Q}$  is *r*. Let *P* be the minimal parabolic subgroup of U(3, 1) with U(2) × *K*<sup>×</sup> as the Levi part. Let  $\Pi$  be an automorphic representation of U(2)(*F*) \ U(2)( $\mathbb{A}_F$ ), which corresponds to a holomorphic weight 2 cuspidal eigenform on GL<sub>2</sub>(*F*). Take a Hecke character  $\eta$  of *K*<sup>×</sup> such that  $\eta_{\infty}(z_{\infty}) = |z_{\infty}|^k / z_{\infty}^k$ . The desirable Eisenstein series  $E_k(\cdot, \Pi, \eta)$  of weight *k* on U(3, 1) is defined by pulling back a Siegel–Eisenstein series on U(3, 3). This pullback idea was due to Shimura and was applied in the constructions of Eisenstein series in many cases; see [Urban 2006; Skinner and Urban 2012; Hsieh 2011b].

Let  $\mathcal{M}_k(U(3, 1))$  be the space of weight *k* automorphic forms on U(3, 1). A linear functional associated to a Bruhat–Schwartz function  $\phi$  on  $\mathbb{A}^2$  (where  $\mathbb{A}$  is the adèle of *F*) can be defined by

$$l_{\phi}: \mathcal{M}_{k}(\mathrm{U}(3,1)) \to \mathbb{C}, \quad \mathbf{F} \mapsto \frac{\langle \mathbf{F}_{\psi}, \theta_{\phi} \rangle \cdot \theta_{\phi}^{\mathrm{II}}(1)}{\langle \theta_{\phi}, \theta_{\phi} \rangle}, \tag{1}$$

where  $\theta_{\phi}^{\Pi}$  is the theta lifting from the cuspidal representation  $\Pi$  on U(2) to U(1) defined by  $\theta_{\phi}$ , and  $\mathbf{F}_{\psi}$  is the Fourier–Jacobi coefficient of  $\mathbf{F}$  attached to the additive character  $\psi$ . When defining the theta lifting, a Hecke character  $\chi$  will be introduced. So in the following Theorem 1.3,  $\chi$  is implicit in the left side of the equation, but appears in the right. For the precise definition of the Fourier–Jacobi coefficient of  $\mathbf{F}$ , one can refer to (25), and change  $E(\cdot, f, s)$  in that equation to  $\mathbf{F}$ . Let  $T_{\mathbb{A}}(\psi)$  be the space of adelic theta functions on  $N(F)U(2)(F) \setminus N(\mathbb{A})U(2)(\mathbb{A})$  with a well-defined inner product  $\langle \cdot, \cdot \rangle$ . Then  $\mathbf{F}_{\psi}$  and  $\theta_{\phi}$  are both elements inside.

In the following theorem, *S* denotes the set of ramified places (the ramified places include places where any data used in the computation is ramified, for example, the characters  $\eta$  and  $\chi$  are ramified, the field extension K/F is ramified, the representation  $\Pi$  is ramified, and so on). We use  $L^S$  to denote the partial *L*-function omitting the local *L*-factors at primes in *S*.

**Theorem 1.3.** For  $E_k(\cdot, \Pi, \eta)$ ,

$$\begin{split} l_{\phi}(E_{k}(\cdot,\,\Pi,\,\eta)) \\ &= C \cdot \frac{(2\pi i)^{(2k-4)r} L^{S}(\eta\chi^{-1},\,\frac{1}{2}(k-2)) L^{S}(\eta^{-1}\chi,\,\frac{1}{2}(k-4)) L^{S}(\Pi,\,\chi,\,1)}{\Omega_{K}^{2k}}, \end{split}$$

where *C* is a nonzero constant which can be explicitly calculated,  $r = [F : \mathbb{Q}]$  and  $\Omega_K$  is the *CM*-period associated to *K*.

The nature of  $l_{\phi}(E_k(\cdot, \Pi, \eta))$  depends on the normalization of  $E_k(\cdot, \Pi, \eta)$ . However, when we choose  $E_k(\cdot, \Pi, \eta)$  to be rational, we can then show that  $l_{\phi}(E_k(\cdot, \Pi, \eta)) \in \overline{\mathbb{Q}}$ . So *C* must be an algebraic number.

In the paper, the value of  $l_{\phi}(E_k(\cdot, \Pi, \eta))$  can simply be obtained by detailed computations of  $\langle E_k(\cdot, \Pi, \eta), \theta_{\phi} \rangle$ , from which one can prove Theorem 4.12. The proof uses computations of Fourier–Jacobi coefficients of the Siegel–Eisenstein series on U(3, 3), whose definition can be found in (21); pullback formulas; the theta liftings for unitary groups; and the Siegel–Weil formula. By unfolding integrals step by step, the questions are translated to computations of Rallis inner product type. Then by studying the integral structure of the space of theta functions  $T_{\mathbb{A}}(\psi)$ , we can prove:

**Proposition 1.4.** For the *p*-integral holomorphic Eisenstein series  $E_k$ ,  $l_{\phi}(E_k)$  is a *p*-integer.

**Conjecture 1.5.**  $E_k(\cdot, \Pi, \eta)$  *does not vanish* modulo *p*.

By the nonvanishing modulo *p* of Hecke *L*-values (see [Hida 2004a]), we see that this conjecture is reduced to the question "Does there exist a Hecke character  $\chi$  such that  $L^{S}(\Pi, \chi, 1)$  does not vanish modulo *p*?" There have been many results

of this flavor by Vatsal, Hida, Sun and Brakočević. I will consider this question in the near future.

The method introduced in this paper for calculating Fourier–Jacobi coefficients of Eisenstein series on U(3, 1) can be generalized well to other nonquasisplit unitary groups. I will briefly explain the generalizations in Remark 5.7 at the end of this paper. For example, this method gives an alternative way to calculate the Fourier–Jacobi coefficients of the Eisenstein series on U(2, 1) other than the one given in [Murase and Sugano 2002], and recovers the proof of nonvanishing modulo p of this Eisenstein series (compare to the discussion in [Mainardi 2004], which uses results in [Murase and Sugano 2002]).

The generalized computational results for Fourier–Jacobi coefficients of certain Eisenstein series on arbitrary unitary groups might inspire a new argument on the nonvanishing modulo p of the Eisenstein series  $E_k(\cdot, \Pi, \eta)$  on U(3, 1). Briefly, one can construct an Eisenstein series on U(4, 2) of the type explained in Remark 5.7. Thus the linear functional on this Eisenstein series will be the product of central special L-values, whose p-adic properties are much easier to study. Especially, we can first obtain nonvanishing modulo p of this Eisenstein series. Then by studying the relations between Fourier–Jacobi coefficients of  $E_k(\cdot, \Pi, \eta)$  on U(3, 1) and this Eisenstein series on U(4, 2), we can furthermore argue the nonvanishing modulo pof  $E_k(\cdot, \Pi, \eta)$ . I hope to address this question soon in another paper.

Structure of the paper and some notation. Section 2 is about the theory of Eisenstein series on unitary groups. Two types of Eisenstein series are defined: one is on U(3, 1) ( $U(I_{\zeta})$ ), and the other is the Siegel type on U(3, 3). Section 2C discusses the pullback formula. Theorem 2.6 gives the precise form, which is actually an adelic counterpart of "pullback of Eisenstein series" in [Shimura 1997].

Section 3 recalls the theory of Weil representations and theta functions. First, Section 3A introduces the Schrödinger representation  $\rho_{\psi}$  of the Heisenberg group and the Weil representation  $\omega_{\psi}$  of the metaplectic group. For this paper, we mainly need the Weil representation restricted to the dual reductive pair of unitary groups. The needed results are summarized in Section 3B, which also explains the relation between  $\omega_{\psi}|_{U(2)}$  and  $\omega_{\psi}|_{U(2,2)}$ . Section 3C briefly recalls the theta lifting that appears in (1) and the Siegel–Weil formula, which is used to calculate the theta lifting.

Section 4 is about computations of Fourier–Jacobi coefficients of the two types of Eisenstein series used in the paper. The very important result that helps give a nice expression of the Fourier–Jacobi coefficients of  $E_k(\cdot, \Pi, \eta)$  is summarized in Theorem 4.9.

Section 5, gives a strategic answer to the question of how to apply the results in Section 4 to show  $E_k(\cdot, \Pi, \eta)$  on U(3, 1) does not vanish modulo p.

Unfortunately, the Siegel–Weil formula for the reductive pair (U(2), U(1)) that I use in this paper is not to be found in the literature — I give a proof in Appendix A. Appendix B gives a very brief discussion about integral theta functions, which is used in Section 5.

Let *G* be an algebraic group defined over the totally real number field *F*. In this paper, we use  $G(\mathbb{A})$  and G(F) to denote the groups of adelic points and *F*-points of *G*, respectively. We use [*G*] to denote the quotient  $G(F) \setminus G(\mathbb{A})$ . Given an arbitrary number field *L* (for example,  $\mathbb{Q}$ , *F*, *K*), for each place *v*, we choose the additive Haar measure on  $L_v$  so that, if  $L_v \simeq \mathbb{R}$ , the measure is the usual Lebesgue measure, if  $L_v \simeq \mathbb{C}$ , the measure is 2 dx dy ( $z = x + iy \in \mathbb{C}$ ), and if *v* is a finite place, the measure gives the volume of the integer ring of  $L_v$  to be  $D_{L_v}^{-1/2}$  ( $D_{L_v}$  is the absolute discriminant of  $L_v$ ). The product of these local measures gives a measure on  $\mathbb{A}_L$ , and thus induces a measure on  $L \setminus \mathbb{A}_L$ . At each finite place *v*, the multiplicative measure on  $\mathbb{A}_{L}^{\times}$ . Notice that this multiplicative measure and the measure chosen in [Skinner and Urban 2012, 8.2.1] differ by a constant. The Haar measures on local and adelic points of algebraic groups should be clear from the definitions above and from the context.

#### 2. Eisenstein series on U(3, 1)

**2A.** Unitary groups. Let K be a totally imaginary quadratic extension of a totally real number field F. Consider an n-dimensional K-vector space V with an  $\epsilon$ -Hermitian form  $\sigma$  for  $\epsilon = \pm 1$ . Without loss of generality, from now on we fix  $\epsilon = -1$ . The unitary group associated to  $(V, \sigma)$  can be defined as follows:

$$U(\sigma) = \{g \in GL(V) \mid \sigma(xg, yg) = (x, y) \text{ for all } x, y \in V\}.$$

To obtain a good matrix representation of  $U(\sigma)$ , let us fix a Witt decomposition of V, with V = J + Z + J', where J and J' are maximal totally  $\sigma$ -isotropic subspaces, and Z is anisotropic or empty, so that dim  $J = \dim J' = r$  and dim Z = t = n - 2r. Under a suitable basis of V consistent with the Witt decomposition,  $\sigma$  can be expressed by the matrix

$$I_{\zeta} = \begin{pmatrix} 0 & 0 & I_r \\ 0 & \zeta & 0 \\ -I_r & 0 & 0 \end{pmatrix}$$

with  $\zeta = -\zeta^* \in GL_t(K)$ . (From now on, we use  $x^*$  to denote  $\overline{x}^t$  for a matrix *x*.) Then:

$$U(\sigma) = U(I_{\zeta}) = \{g \in \operatorname{GL}_n(K) \mid gI_{\zeta}g^* = I_{\zeta}\}.$$

Accordingly, the adelic unitary group  $U(I_{\zeta})(\mathbb{A})$  and local groups  $U(I_{\zeta})_v$  can be defined. Given a totally  $\sigma$ -isotropic subspace J (which may not be maximal), put  $P_J = \{g \in U(I_{\zeta}) \mid Jg = J\}$ . Then  $P_J$  is the parabolic subgroup of  $U(I_{\zeta})$  associated to J.

The Hermitian domain associated to  $U(I_{\zeta})$  is defined as

$$\mathfrak{Z} = \mathfrak{Z}(r,\zeta) = \left\{ \begin{pmatrix} x \\ y \end{pmatrix} \in \mathbb{C}_r^{r+t} \mid x \in \mathbb{C}_r^r, \ y \in \mathbb{C}_r^t, \ i(x^* - x) > iy^*\zeta^{-1}y \right\}$$
(2)

Let

$$g_{\infty} = \begin{pmatrix} a & b & c \\ d & e & f \\ h & l & p \end{pmatrix} \in U(I_{\zeta})_{\infty},$$

with  $a, p \in \mathbb{C}_r^r$  and  $e \in \mathbb{C}_t^t$ . The action of  $g_{\infty}$  on  $\mathfrak{Z}(r, \zeta)$  is

$$g_{\infty}\begin{pmatrix}x\\y\end{pmatrix} = \begin{pmatrix}(ax+by+c)(hx+ly+p)^{-1}\\(dx+ey+f)(hx+ly+p)^{-1}\end{pmatrix}.$$

The automorphy factor is

$$j\left(g_{\infty}, \begin{pmatrix} x \\ y \end{pmatrix}\right) = \det(hx + ly + p).$$
(3)

**Proposition 2.1.** (1) Pick the origin i of  $\mathfrak{Z}$  and put  $C = \{g_{\infty} \in U(I_{\zeta})_{\infty} | g_{\infty}i = i\}$ . Then C is a maximal compact subgroup of  $U(I_{\zeta})_{\infty}$ .

(2) Let  $P = P_J$ , where J is a maximal totally isotropic subspace of V. Then  $U(I_{\zeta})_{\infty} = P_{\infty}C$ .

Assume dim V = 4 so that J and J' are two isotropic lines, and Z is a 2dimensional anisotropic space; then  $U(I_{\zeta}) \subset GL_4(K)$  is a degree 4 unitary group. From now on, whenever we write  $U(I_{\zeta})$ , I mean particularly this unitary group. To  $U(I_{\zeta})$ , there is only one nontrivial parabolic subgroup up to conjugation that is  $P_J$ . We can simply denote it by P. It consists of such elements

$$p = \begin{pmatrix} a & * & * \\ & u & * \\ & & (a^*)^{-1} \end{pmatrix},$$

where  $a \in K^{\times}$  and

$$u \in U(\zeta) = \{ u \mid u\zeta u^* = \zeta \}.$$
(4)

**2B.** *Eisenstein series.* Following the notation above, *P* is the parabolic subgroup of  $U(I_{\zeta})$ . It has the Levi decomposition  $P = M \cdot N$ , where M is the Levi part which is isomorphic to  $U(\zeta) \times \mathbb{G}_m/K$ , and N is the unipotent radical, consisting of elements like

$$\begin{pmatrix} 1 & * & * \\ & I_2 & * \\ & & 1 \end{pmatrix}.$$

Given a cuspidal representation  $\Pi$  of U( $\zeta$ ) and a Hecke character  $\eta$  of K, we get a cuspidal representation  $\Pi \otimes \eta$  on the Levi part M. Then one has the induced representation

$$I_P^{U(I_{\zeta})}(\Pi \otimes \eta, s) := \operatorname{Ind}_{P(\mathbb{A})}^{U(I_{\zeta})(\mathbb{A})} \delta_P^{\frac{1}{2}+s} \cdot \Pi \otimes \eta,$$

where  $\delta_P$  is the modulus character on *P*. If we denote by  $V_{\Pi}$  the representation space of  $\Pi$ , then the representation space for  $I_P^{U(I_{\zeta})}(\Pi \otimes \eta, s)$  is the set of smooth functions  $\tilde{f}_s : U(I_{\zeta})(\mathbb{A}) \to V_{\Pi}$  such that

- (1)  $\tilde{f}_s(pg) = \delta_P(p)^{\frac{1}{2}+s} \cdot \Pi \otimes \eta(p)(\tilde{f}_s(g)), p \in P(\mathbb{A}),$
- (2)  $\tilde{f}_s$  is right **K**-finite with **K** a maximal open compact subgroup of  $U(I_{\zeta})(\mathbb{A})$ .

The action of  $U(I_{\zeta})(\mathbb{A})$  on  $I_P^{U(3,1)}(\Pi \otimes \eta, s)$  is by right translation.

If we further assume that  $\Pi$  is an irreducible submodule of  $\mathcal{A}(M)$ , where  $\mathcal{A}(M)$ denotes the set of automorphic forms on M, we can realize  $I_p^{U(I_{\zeta})}(\Pi \otimes \eta, s)$  as  $\mathbb{C}$ -valued functions rather than functions valued in  $V_{\Pi}$ . For  $\tilde{f}_s \in I_p^{U(I_{\zeta})}(\Pi \otimes \eta, s)$ , let  $f_s(g) = (\tilde{f}_s(g))(1)$ . Then  $f_s(g)$  satisfies  $f_s(nmg) = \tilde{f}_s(g)(m)$ . Define the Eisenstein series

$$E(g, f, s) = \sum_{\gamma \in P(F) \setminus U(I_{\zeta})(F)} f_s(\gamma g).$$
(5)

It can be proved that the sum converges absolutely when  $\text{Re}(s) \gg 0$ , and can be continued to a meromorphic function on  $\mathbb{C}$ .

Aside from the Eisenstein series on  $U(I_{\zeta})$ , we also want to define an Eisenstein series on U(3, 3), which we are going to use in next section. Following definitions of unitary groups in the previous section,

$$U(3,3) = \left\{ g \in \mathrm{GL}_{6}(K) \; \middle| \; g \begin{pmatrix} 0 & I_{3} \\ -I_{3} & 0 \end{pmatrix} g^{*} = \begin{pmatrix} 0 & I_{3} \\ -I_{3} & 0 \end{pmatrix} \right\}.$$
(6)

The Siegel parabolic subgroup  $\mathcal{P}$  of U(3, 3) is the one that fixes the maximal totally isotropic space of dimension 3 in a 6-dimensional Hermitian vector space. The Levi part of  $\mathcal{P}$  is isomorphic to  $GL_3(K)$ . Take the one-dimensional representation  $\eta(\det \cdot)$  on the Levi part. Then it induces the representation on U(3, 3) as

$$I_{\mathcal{P}}^{U(3,3)}(\eta,s) := \operatorname{Ind}_{\mathcal{P}(\mathbb{A})}^{U(3,3)(\mathbb{A})} \delta_{\mathcal{P}}^{\frac{1}{2}+s} \cdot \eta(\det \cdot).$$

Pick a section  $\mathbb{f}_s$  in the induced representation. The Siegel–Eisenstein series is defined by

$$\mathbb{E}(g, \mathbb{F}, s) = \sum_{\gamma \in \mathcal{P}(F) \setminus U(3,3)(F)} \mathbb{F}_s(\gamma g) \quad \text{for } g \in U(3,3)(\mathbb{A}).$$
(7)

It satisfies analytic properties similar to  $E(\cdot, f, s)$ .

**2C.** *Pullback formulas.* In this section, the Eisenstein series E(g, f, s) on  $U(I_{\zeta})$  is constructed using the pullback of a Siegel Eisenstein series on U(3, 3).

The unitary group  $U(\zeta)$  is closely related to the quaternion algebra, about which, let me quote two lemmas from [Shimura 1997].

**Lemma 2.2.** *V* is a 2-dimensional *K*-vector space with a nondegenerate Hermitian pairing described by  $\zeta$ . Then *V* is anisotropic if and only if det  $\zeta$  is represented by -1 in  $K^{\times}/N_{K/F}(K^{\times})$ .

**Lemma 2.3.** Let  $(V, \zeta)$  be anisotropic.

- (1)  $D := \{ \alpha \in \text{End}(V) \mid \sigma(\alpha^{t}x, y) = \sigma(x, \alpha y) \text{ for all } x, y \in V \}$ , where  $\iota$  is the main involution of End(V) such that  $\begin{pmatrix} a & b \\ c & d \end{pmatrix}^{\iota} = \begin{pmatrix} d & -b \\ -c & a \end{pmatrix}$ . Then D is a definite quaternion algebra over F.
- (2) U( $\zeta$ ) = { $s\alpha \mid s \in K^{\times}, \alpha \in D^{\times}, ss^{\rho} \det \alpha = 1$ }.

So, locally:

- (1) When  $K_v$  splits, namely  $K_v = F_v \times F_v$ , we have  $U(\zeta)_v \simeq GL_2(F_v)$ .
- (2) When  $K_v$  is a field and  $d(\sigma_v)$  is represented by 1, we have  $U(\zeta)_v = U(1, 1)_v$ .
- (3) When K<sub>v</sub> is a field and d(σ<sub>v</sub>) is represented by −1 (including archimedean places), we have U(ζ)<sub>v</sub> ⊂ K<sup>×</sup><sub>v</sub> · D<sup>×</sup><sub>v</sub>, and D<sub>v</sub> is a quaternion algebra. In this case, U(ζ)<sub>v</sub> is compact.

The following theorem about the Jacquet–Langlands correspondence relates automorphic forms on  $D^{\times}$  to automorphic forms on  $GL_2(F)$ .

**Theorem 2.4** [Gelbart 1975]. *S* is the set of places of *F* where *D* is ramified. To each irreducible unitary admissible representation  $\pi = \bigotimes_{v} \pi_{v}$  of  $D^{\times}_{\mathbb{A}}$ , let  $\pi'$  denote the representation of  $GL_{2}(\mathbb{A})$  whose *v*-th component is equivalent to  $\pi_{v}$  if  $v \notin S$ , and special or supercuspidal if  $v \in S$ .

- (1) The map  $\pi \to \pi'$  restricted to the collection of (greater than one-dimensional) cusp forms on  $D^{\times}$  is one-to-one onto the collection of all (equivalence classes of) cusp forms on  $GL_2(F)$  such that  $\pi'_v$  is square-integrable for each  $v \in S$ .
- (2) If we require further that  $\pi \subset L^2(D^{\times}(F) \setminus D^{\times}(\mathbb{A}))$ , then it implies that  $\pi'_v$  is one-dimensional for  $v \in S$ .

For an irreducible representation  $\pi$  on the definite quaternion algebra D,  $\pi_{\infty}$  as a finite-dimensional representation of  $D_{\infty}^{\times}$  is equivalent to  $|\det|^r \otimes \rho_n$ , where  $\rho_n$  is the *n*-th symmetric tensor product. It corresponds to the discrete series

 $\sigma(\mu_1, \mu_2)$  of  $GL_2(\mathbb{R})$  with  $\mu_1(x) = |x|^{r+n+\frac{1}{2}}$  and  $\mu_2(x) = |x|^{r-\frac{1}{2}} \operatorname{sgn}(x)^n$ .

Globally,  $\pi$  corresponds to a cuspidal newform f that is also a Hecke eigenform of weight k = n + 2.

Take a representation  $\Pi$  of  $U(\zeta)$  in this way: First pick an irreducible representation  $\lambda \cdot \pi$  on  $\mathbb{A}_K^{\times} \cdot D_{\mathbb{A}}^{\times}$  satisfying  $\lambda|_{\mathbb{A}^{\times}} = \chi_{\pi}$ , where  $\chi_{\pi}$  is the central character of  $\pi$ . Then  $\Pi = (\lambda \cdot \pi)|_{U(\zeta)(\mathbb{A})}$  gives an irreducible representation on  $U(\zeta)$ . For our later application, the cusp form f that  $\pi$  corresponds to comes from an elliptic curve. So f has weight k = 2. Then  $\Pi_{\infty}$  must be one-dimensional. In this case, the pullback formula will have a simple form. So in this paper, we always assume  $\Pi$  satisfies this condition.

Define an embedding

$$\mathfrak{e}: U(I_{\zeta}) \times U(\zeta) \to U(3,3), \quad (g,u) \mapsto \mathfrak{e}(g,u) = A_{\zeta} \left(\begin{smallmatrix} g \\ u \end{smallmatrix}\right) A_{\zeta}^{-1}, \tag{8}$$

where

$$A_{\zeta} = \begin{pmatrix} 1 & & \\ \zeta^{-1} & -\zeta^{-1} \\ & 1 & \\ \frac{1}{2} & \frac{1}{2} \end{pmatrix}.$$

Pay attention that  $U(\zeta)$  has no nontrivial parabolic subgroup. If  $\mathcal{P}$  is the Siegel parabolic subgroup of U(3, 3) and *P* is the only nontrivial parabolic subgroup of  $U(I_{\zeta})$ , we have the following lemma:

**Lemma 2.5** [Shimura 1997].  $\mathcal{P}(F) \setminus U(3, 3)(F) = \mathfrak{e}(P(F) \setminus U(I_{\zeta})(F), U(\zeta)(F)).$ 

**Theorem 2.6.** Take  $\beta \in V_{\Pi} \subset L^2(U(\zeta)(F) \setminus U(\zeta)(\mathbb{A}))$ . Then the integral

$$\int_{U(\zeta)(F)\setminus U(\zeta)(\mathbb{A})} \mathbb{E}(\mathfrak{e}(g,u),\mathbb{F},s)\beta(u)\eta^{-1}(\det u)\,du$$

gives an Eisenstein series E(g, f, s) on  $U(I_{\zeta})$  associated to  $f_s \in I_p^{U(I_{\zeta})}(\Pi \otimes \eta, s)$ , and

$$f_s(g) = \int_{U(\zeta)(\mathbb{A})} \mathbb{f}_s(\mathfrak{e}(g, u))\beta(u)\eta^{-1}(\det u) \, du.$$

*Here we understand*  $\beta(u)$  *as*  $(\Pi(u)\beta)(1)$ *.* 

*Proof.* General results about the pullback formula at infinity are essentially discussed in [Shimura 1997]. Here I give a proof using the adelic language. Using Lemma 2.5,

$$\mathbb{E}(\mathfrak{e}(g, u), \mathbb{f}, s) = \sum_{\gamma \in \mathscr{P}(F) \setminus U(3,3)(F)} \mathbb{f}_s(\gamma \mathfrak{e}(g, u))$$
$$= \sum_{\gamma_1 \in P(F) \setminus U(I_{\zeta})(F)} \sum_{\gamma_2 \in U(\zeta)(F)} \mathbb{f}_s(\mathfrak{e}(\gamma_1 g, \gamma_2 u)),$$

unfold the integral:

$$\begin{split} \int_{U(\zeta)(F)\setminus U(\zeta)(\mathbb{A})} \mathbb{E}(\mathfrak{e}(g,u),\mathbb{F},s)\beta(u)\eta^{-1}(\det u)du \\ &= \int_{U(\zeta)(F)\setminus U(\zeta)(\mathbb{A})} \sum_{\gamma_1\in P(F)\setminus U(I_{\zeta})(F)} \sum_{\gamma_2\in U(\zeta)(F)} \mathbb{F}_s(\mathfrak{e}(\gamma_1g,\gamma_2u))du \\ &= \int_{U(\zeta)(\mathbb{A})} \sum_{\gamma_1\in P(F)\setminus U(I_{\zeta})(F)} \mathbb{F}_s(\mathfrak{e}(\gamma_1g,u))\beta(u)\eta^{-1}(\det u)du \\ &= \sum_{\gamma_1\in P(F)\setminus U(3,1)(F)} \int_{U(\zeta)(\mathbb{A})} \mathbb{F}_s(\mathfrak{e}(\gamma_1g,u))\beta(u)\eta^{-1}(\det u)du. \quad (*) \end{split}$$

In the last step, we suppose *s* is in a proper range so that there are no convergence problems. Then we can change the order of the integral and the summation. Formally, (\*) looks like the Eisenstein series E(g, f, s) on  $U(I_{\zeta})$  defined by (5) with

$$f_s = \int_{U(\zeta)(\mathbb{A})} \mathbb{f}_s(\mathfrak{e}(g, u))\beta(u)\eta^{-1}(\det u)\,du.$$

We are left to show that  $f_s \in I_P^{U(I_{\zeta})}(\Pi \otimes \eta, s)$ . Take

$$p = \begin{pmatrix} a & * & * \\ v & * \\ & (a^*)^{-1} \end{pmatrix} \in P(\mathbb{A}),$$

and also note that

$$\mathfrak{e}\left(\begin{pmatrix} a & * & * \\ v & * \\ & (a^*)^{-1} \end{pmatrix}, v\right) = \begin{pmatrix} a & * & * & * \\ \zeta^{-1}v\zeta & * & * \\ & (a^*)^{-1} & * \\ & & v \end{pmatrix} \in \mathcal{P}(\mathbb{A})$$

by the embedding formula (8). So

$$\begin{split} f_{s}(pg) &= \int_{U(\zeta)(\mathbb{A})} \mathbb{f}_{s}(\mathfrak{e}(pg, u))\beta(u)\eta^{-1}(\det u) \, du \\ &= \int_{U(\zeta)(\mathbb{A})} \mathbb{f}_{s}(\mathfrak{e}(pg, vu))\beta(vu)\eta^{-1}(\det vu) \, du \\ &= \int_{U(\zeta)(\mathbb{A})} \mathbb{f}_{s}\left(\mathfrak{e}\left(\begin{pmatrix} a & * & * \\ v & * \\ (a^{*})^{-1} \end{pmatrix}, v\right) \cdot \mathfrak{e}(g, u)\right)\beta(vu)\eta^{-1}(\det vu) \, du \\ &= \int_{U(\zeta)(\mathbb{A})} \eta(a \det v)|a \det v|^{\frac{1}{2}+s} \mathbb{f}_{s}(\mathfrak{e}(g, u))\beta(vu)\eta^{-1}(\det vu) \, du \\ &= \Pi \otimes \eta(p) \, f_{s}(g)|a|^{\frac{1}{2}+s}. \end{split}$$

This means that  $f_s \in I_P^{U(I_{\zeta})}(\Pi \otimes \eta, s)$ .

#### 3. Theta functions

**3A.** *Weil representations.* Suppose *V* is a finite-dimensional vector space over a field  $\mathbb{F}$ . When  $\mathbb{F}$  is a nonarchimedean local field, use S(V) to denote the space of Bruhat–Schwartz functions (locally constant compactly supported functions) on *V*. If  $F = \mathbb{R}$  or  $\mathbb{C}$ , we first take S(V) to be  $L^2(V)$  (but later, we may add holomorphic conditions when needed).

Following the notation in previous sections, F is a totally real number field. Let W be a finite-dimensional symplectic vector space over  $F_v$  with a nondegenerate alternating form  $\langle \cdot, \cdot \rangle$ . The Heisenberg group H(W) associated to W is a nontrivial central extension of W by  $F_v$  and is defined to be a group of pairs  $\{(w, t) | w \in W, t \in F_v\}$  with the law of multiplication

$$(w_1, t_1)(w_2, t_2) = \left(w_1 + w_2, t_1 + t_2 + \frac{1}{2} \langle w_1, w_2 \rangle\right).$$

Fix an additive character  $\psi$  of  $F_v$  and a complete polarization of W as  $W = X \oplus Y$ where X and Y are maximal totally isotropic subspaces of W. Define the Schrödinger representation  $\rho_{\psi}$  of H(W) on S(X) as follows:

$$\begin{split} \rho_{\psi}(x)f(z) &= f(x+z) & \text{for } x \in X, \\ \rho_{\psi}(y)f(z) &= \psi(\langle z, y \rangle)f(z) & \text{for } y \in Y, \\ \rho_{\psi}(t)f(z) &= \psi(t)f(z) & \text{for } t \in F_{v}. \end{split}$$

**Theorem 3.1** (Stone, von Neumann). H(W) has a unique irreducible smooth representation on which  $F_v$  operates via the character  $\psi$ .

294

One may have seen other constructions of smooth irreducible representations of H(W) on which the center acts by the character  $\psi$ . However, by Theorem 3.1, they are isomorphic to one another. In this paper, I only use  $(\rho_{\psi}, S(X))$  defined above, which is so-called the Schrödinger model.

The symplectic group Sp(W) has an action on H(W) as  $g \cdot (w, t) = (gw, t)$ . By the uniqueness theorem of Stone and von Neumann, there is an operator  $\omega_{\psi}(g)$  on S(X) that it is unique up to scalar and satisfies

$$\rho_{\psi}(gw, t)\omega_{\psi}(g) = \omega_{\psi}(g)\rho_{\psi}(w, t) \quad \text{for all } (w, t) \in H(W).$$
(10)

Define the metaplectic group

$$\operatorname{Sp}_{\psi}(W) = \{(g, \omega_{\psi}(g)) \text{ such that } (10) \text{ holds}\},\$$

which fits in the following exact sequence:

$$1 \to \mathbb{C}^{\times} \to \widetilde{\operatorname{Sp}}_{\psi}(W) \xrightarrow{\operatorname{proj}} \operatorname{Sp}(W) \to 1.$$

The Weil representation of  $\widetilde{Sp}_{\psi}(W)$  is the one obtained by projecting to the second factor  $(g, \omega_{\psi}(g)) \rightarrow \omega_{\psi}(g)$ . Under the Schrödinger model  $(\rho_{\psi}, S(X))$ , the Weil representation can be explicitly written down:

$$\omega_{\psi} \begin{pmatrix} A \\ (A^{t})^{-1} \end{pmatrix} \phi(x) = |\det A|^{\frac{1}{2}} \phi(xA),$$
$$\omega_{\psi} \begin{pmatrix} I & B \\ I \end{pmatrix} \phi(x) = \psi \left(\frac{xBx^{t}}{2}\right) \phi(x),$$
$$\omega_{\psi} \begin{pmatrix} I \\ -I \end{pmatrix} \phi(x) = \gamma \hat{\phi}(x),$$

where  $\phi \in S(X)$ ,  $\hat{\phi}$  is the Fourier transform

$$\hat{\phi}(x) = \int_{F_v^n} \phi(y) \psi\left(\sum_{i=1}^n x_i y_i\right) dy,$$

the Haar measure dy is chosen so that  $\hat{\phi} = \phi(-x)$ , and  $\gamma$  is an 8-th root of unity determined by  $\psi$ .

The discussion above in the local case can mostly be generalized to the global case. The global Schrödinger representation  $\rho_{\psi}$  of  $H(W)(\mathbb{A})$  on  $S(X(\mathbb{A}))$  can be defined, where

$$S(X(\mathbb{A})) = \{ \bigotimes_v \phi_v \mid \phi_\infty \in L^2(X_\infty), \phi_v \in S(X_v), \text{ and for a.a. } v, \hat{\phi}_v = \phi_v \}.$$

Also we have the global Weil representation  $\omega_{\psi}$  of  $\widetilde{Sp}(W)(\mathbb{A})$  on  $S(X(\mathbb{A}))$ , and for each place v of F, one has the following commutative diagram:



#### **3B.** Dual reductive pair.

*Definition.* A dual reductive pair is a pair of subgroups (G, G') of the symplectic group Sp(W) such that

- (1) G is the centralizer of G' in Sp(W) and vice versa, and
- (2) the actions of G and G' are completely reducible on W. (An action is called completely reducible if every invariant subspace has an invariant complement.)

Obviously this definition can be applied locally or globally.

In this paper, the types of reductive pairs used are the following: *K* is again an imaginary quadratic extension of *F*,  $(V_1, (\cdot, \cdot)_1)$  is a skew-Hermitian space over *K*, and  $(V_2, (\cdot, \cdot)_2)$  is a Hermitian space over *K*. Take  $W = V_1 \otimes V_2$ , and on *W* define an alternating form  $\frac{1}{2} \operatorname{tr}_{K/F}(\cdot, \cdot)_1 \otimes \overline{(\cdot, \cdot)_2}$ . Then the unitary groups  $U(V_1)$  and  $U(V_2)$  form a dual reductive pair in Sp(*W*). We have the embedding

$$e: U(V_1) \times U(V_2) \to \operatorname{Sp}(W), \quad (g_1, g_2) \cdot (v_1 \otimes v_2) \mapsto = v_1 g_1 \otimes g_2^{-1} v_2.$$
(11)

In this paper, we use e to denote the embeddings of unitary groups into symplectic groups, and use e to denote the embeddings of the same type of groups, such as the embedding of one unitary group into another unitary group, or the embedding of one symplectic group into another symplectic group.

*Splittings.* For a fixed Weil representation  $g \to \omega(g)$  (for example, the Weil representation  $\omega_{\psi}$  defined through the Schrödinger model), we can define  $c(g_1, g_2)$  so that

$$\omega(g_1)\omega(g_2) = c(g_1, g_2)\omega(g_1g_2).$$

Then  $c : \operatorname{Sp}(W) \times \operatorname{Sp}(W) \to \mathbb{C}^{\times}$  is a 2-cocycle whose class is in the cohomology group  $H^2(\operatorname{Sp}(W), \mathbb{C}^{\times})$ . When the additive character  $\psi$  of F and one maximal isotropic subspace of W are fixed, c is determined using the Leray invariant. The following proposition claims that under certain condition, c could be a coboundary. Returning to the dual reductive pair  $(U(V_1), U(V_2)))$  of  $\operatorname{Sp}(W)$ , we have this:

**Proposition 3.2** [Harris et al. 1996].  $\widetilde{Sp}(W)$  splits over  $U(V_i)$  compatibly with respect to rational points for i = 1, 2. In particular, there is a splitting homomorphism

 $s_i: U(V_i)(\mathbb{A}) \to \widetilde{\mathrm{Sp}}(W)(\mathbb{A})$  such that we have the commutative diagram



where  $e_i$  is the embedding of  $U(V_i)$  into Sp(W), which is the restriction of e defined in (11). Further,  $s_i(U(V_i)(F)) \subset \widetilde{Sp}(W)(F)$ .

If dim<sub>*K*</sub>  $V_2 = m$ , we choose a character  $\chi_{V_2}$  of  $K^{\times}$  such that  $\chi_{V_2}|_{F^{\times}} = \epsilon_{K/F}^m$ , where  $\epsilon_{K/F}$  is the quadratic character associated to the extension K/F. This choice determines a lifting  $s_1 : U(V_1) \to \widetilde{Sp}(W)$ , which can be explicitly formulated. Notice that two choices  $\chi$  and  $\chi'$  of  $\chi_{V_2}$  differ by a character  $\mu$  of  $K^{\times}$ , namely  $\chi = \mu \chi'$  and  $\mu|_{F^{\times}} = 1$ . Then  $\mu$  defines a character  $\mu'$  on  $K^1$  by  $\mu'(x/\bar{x}) = \mu(x)$ , with  $x \in K^{\times}$ , and we have:

**Lemma 3.3** [Harris et al. 1996].  $s_{1,\chi} = (\mu' \circ \det) \cdot s_{1,\chi'}$ 

*Doubling method.* From the symplectic space  $W = V_1 \otimes V_2$ , we can create a new symplectic space  $\mathbb{W}$ , which is essentially two copies of W, in this way: Take  $W^- = V_1^- \otimes V_2$ . As a vector space,  $V_1^-$  is the same as  $V_1$ , but the skew-Hermitian form defined on it is  $-(\cdot, \cdot)_1$ . To  $\mathbb{W} = W \oplus W^-$ , we have one dual reductive paier  $(U(V_1 \oplus V_1^-), U(V_2))$ . We have the following commutative diagram:

$$U(V_{1} \oplus V_{1}^{-}) \xrightarrow{e_{1}} \operatorname{Sp}(W \oplus W^{-}) = \operatorname{Sp}(\mathbb{W})$$

$$\uparrow^{\mathfrak{c}} \qquad \uparrow^{\mathfrak{c}}$$

$$U(V_{1}) \times U(V_{1}) \xrightarrow{e_{1} \times e_{1}} \operatorname{Sp}(W) \times \operatorname{Sp}(W)$$

$$\uparrow^{\Delta} \qquad \uparrow^{\Delta}$$

$$U(V_{1}) \xrightarrow{e_{1}} \operatorname{Sp}(W)$$

By Proposition 3.2,  $\widetilde{Sp}(\mathbb{W})$  splits over each of the unitary groups in the diagram above. We want to determine the compatibility among the splittings. The Weil representation of  $\widetilde{Sp}(\mathbb{W})$  determines an isomorphism

$$\widetilde{\mathrm{Sp}}(\mathbb{W}) \simeq \mathrm{Sp}(\mathbb{W}) \times \mathbb{C}^{\times}$$

Group multiplications on the right hand side are described by the cocycle  $c(g_1, g_2)$ :

$$(g_1, c_1)(g_2, c_2) = (g_1g_2, c(g_1, g_2)c_1c_2).$$

The inverse image of  $\mathfrak{e}(\operatorname{Sp}(W) \times 1)$  in  $\widetilde{\operatorname{Sp}}(W \oplus W^{-})$  is isomorphic to  $\widetilde{\operatorname{Sp}}(W)$ . We choose a lift  $\tilde{\mathfrak{e}}$  of  $\mathfrak{e}$  so that

$$\widetilde{\mathfrak{e}}:\widetilde{\mathrm{Sp}}(W)\times\widetilde{\mathrm{Sp}}(W)\to\widetilde{\mathrm{Sp}}(W\oplus W^-)$$

restricted to the  $\mathbb{C}^{\times}$ -component is

$$\mathbb{C}^{\times} \times \mathbb{C}^{\times} \to \mathbb{C}^{\times}, \quad (c_1, c_2) \mapsto c_1 \bar{c}_2.$$

Since  $(U(V_1 \oplus V_1^-), U(V_2))$  is a dual reductive pair of Sp(W), and dim<sub>K</sub>  $V_2 = m$ , choose  $\chi$  such that  $\chi|_{\mathbb{A}^{\times}} = \epsilon_{K/F}^m$ . We can obtain an explicit homomorphism

$$s_{\chi}: U(V_1 \oplus V_1^-) \to \widetilde{\mathrm{Sp}}(\mathbb{W}),$$

so that

$$U(V_1 \oplus V_1^-) \xrightarrow{s_{\chi}} \widetilde{Sp}(\mathbb{W})$$

$$\uparrow^{\mathfrak{e}} \qquad \uparrow^{\tilde{\mathfrak{e}}}$$

$$U(V_1) \times U(V_1) \xrightarrow{s_{1,\chi} \times s_{1,\chi,-}} \widetilde{Sp}(W) \times \widetilde{Sp}(W)$$

Lemma 3.4 [Harris et al. 1996]. In the commutative diagram above,

$$s_{\chi}|_{U(V_1)\times 1} = s_{1,\chi}$$
 and  $s_{1,\chi,-} = (\chi^{-1} \circ \det) \cdot s_{1,\chi}$ .

*Weil representations on dual reductive pairs.* Let  $W = X \oplus Y$  be the complete polarization. Then  $\mathbb{W}$  naturally has the polarization  $\mathbb{W} = (X \oplus X) \oplus (Y \oplus Y)$ . Now take the Weil representation  $\omega_{\psi}$  of  $\widetilde{Sp}(\mathbb{W})$  constructed from the standard Schrödinger model associated to this polarization. Fix a pair of characters  $\chi_n$  and  $\chi_m$  of  $K^{\times}$  with  $\chi_n|_{\mathbb{A}^{\times}} = \epsilon_{K/F}^n$  and  $\chi_m|_{\mathbb{A}^{\times}} = \epsilon_{K/F}^m$ , where  $n = \dim_K V_1$  and  $m = \dim_K V_2$ . These characters determine the splitting homomorphisms

$$s_{\chi_m} : U(V_1 \oplus V_1^-) \to \widetilde{\operatorname{Sp}}(\mathbb{W}),$$
  

$$s_{1,\chi_m} : U(V_1) \to \widetilde{\operatorname{Sp}}(W),$$
  

$$s_{2,\chi_m} : U(V_2) \to \widetilde{\operatorname{Sp}}(W).$$

Define the representation of  $U(V_1 \oplus V_1^-)$  to be  $\omega_{\chi_m} = \omega_{\psi} \circ s_{\chi_m}$ ; the representations  $\omega_{1,\chi_m}$  and  $\omega_{2,\chi_n}$  of  $U(V_1)$  and  $U(V_2)$  can be defined similarly.

There is another polarization of  $\mathbb{W}$  that we are also interested in. The skew-Hermitian space  $V_1 \oplus V_1^-$  has the decomposition  $V_1 \oplus V_1^- = V_d \oplus V^d$  where  $V_d = \{(x, x) \mid x \in V_1\}$  and  $V^d = \{(x, -x) \mid x \in V_1\}$ . They are maximal isotropic subspaces. Thus  $\mathbb{W} = V_d \otimes V_2 \oplus V^d \otimes V_2 = \mathbb{W}_d \oplus \mathbb{W}^d$  is a complete polarization. We abuse the notation and still use  $\omega_{\psi}$  to denote the Weil representation defined from this polarization, since one can easily distinguish representations attached to two polarizations from the context. First, let us write down the Weil representation

 $\omega_{\chi_m}$  of  $U(V_1 \oplus V_1^-)$  on  $S(\mathbb{W}_d)$ , which we are going to use in later calculations. By the Witt decomposition  $V_1 \oplus V_1^- = V_d \oplus V^d$ ,  $\begin{pmatrix} A \\ (A^*)^{-1} \end{pmatrix} \in U(V_1 \oplus V_1^-)$  for  $A \in GL(V_d)$ , and  $\begin{pmatrix} 1 & B \\ 1 & 1 \end{pmatrix} \in U(V_1 \oplus V_1^-)$  if  $B = B^*$ . Given  $\phi(x) \in S(V_d \otimes V_2)$ , we have

$$\omega_{\chi_m} \begin{pmatrix} A \\ (A^*)^{-1} \end{pmatrix} \phi(x) = \chi_m(\det A) |\det A|_K^{m/2} \phi(A^t x),$$
  

$$\omega_{\chi_m} \begin{pmatrix} 1 & B \\ 1 \end{pmatrix} \phi(x) = \psi(x^* B x) \phi(x),$$
  

$$\omega_{\chi_m} \begin{pmatrix} 1 \\ -1 \end{pmatrix} \phi(x) = \hat{\phi}(x).$$
(12)

**Proposition 3.5.** (1) Under the homomorphism  $\tilde{\mathfrak{e}} : \widetilde{\mathrm{Sp}}(W) \times \widetilde{\mathrm{Sp}}(W) \to \widetilde{\mathrm{Sp}}(W)$ , we have  $\omega_{\psi} \circ \tilde{\mathfrak{e}} = \omega_{\psi} \otimes \check{\omega}_{\psi}$ , where  $\check{\omega}_{\psi}$  is the contragredient of  $\omega_{\psi}$ .

- (2) As the representation of  $U(V_1) \times U(V_1)$ , we have  $\omega_{\chi_m} \circ \mathfrak{e} \simeq \omega_{1,\chi_m} \otimes (\chi_m \circ \check{\omega}_{1,\chi_m})$ .
- (3) The representation  $\omega_{\chi_m} \circ \mathfrak{e} \circ \Delta$  of  $U(V_1)$  is isomorphic to the twist by  $\chi_m$  of the linear action of  $U(V_1)$  on  $S(\mathbb{W}_d)$ , that is, for  $\phi \in S(\mathbb{W}_d)$  and  $x \in \mathbb{W}_d$ ,

$$\omega_{\chi_m}(\mathfrak{e}(g,g))\phi(x) = \chi_m(\det g) \cdot \phi(xg).$$

Using two polarizations of  $\mathbb{W}$ , two Weil representations of  $U(V_1 \oplus V_1^-)$  are defined above. An isometry between the two representation spaces  $S((X \oplus X)(\mathbb{A}))$  and  $S(\mathbb{W}_d(\mathbb{A}))$  can be given so that it intertwines the two representations on the spaces. Let

$$\delta_{\psi}: S((X \oplus X)(\mathbb{A})) \to S(\mathbb{W}_d(\mathbb{A}))$$

be the intertwining isometry. Identify  $\mathbb{W}_d$  with W via the map  $(w, -w) \to w$  and write  $w \in W$  as w = (x, y) with respect to the polarization  $W = X \oplus Y$ . Then for  $\phi \in S((X \oplus X)(\mathbb{A}))$ , one has

$$\delta_{\psi}(\phi)(w) = \int_{X(\mathbb{A})} \psi(2\langle u, y \rangle) \phi(u+x, u-x) \, du, \tag{13}$$

where  $\langle \cdot, \cdot \rangle$  is the alternating pair on *W*. The map  $\delta_{\psi}$  intertwines the action of  $\widetilde{Sp}(\mathbb{W})$  on those two spaces. In particular, if  $\phi = \phi_1 \otimes \overline{\phi}_2$  for  $\phi_1, \phi_2 \in S(X(\mathbb{A}))$ ,

$$\delta_{\psi}(\phi_1 \otimes \phi_2)(0) = (\phi_1, \phi_2),$$

where  $(\cdot, \cdot)$  is the Hermitian inner product in the Hilbert space  $L^2(X(\mathbb{A}))$ , and

$$\omega_{\chi_m}(i(g_1,g_2))\delta_{\psi}(\phi_1\otimes\overline{\phi}_2)=\delta_{\psi}(\omega_{1,\chi_m}(g_1)\phi_1\otimes\chi_m(\det g_2)\check{\omega}_{1,\chi_m}(g_2)\overline{\phi}_2).$$

**3C.** *Theta functions and applications.* In this section, I will recall some facts about theta functions that are used in calculations of Fourier–Jacobi coefficients of Eisenstein series. More discussions about arithmetic properties of theta functions will be left to Appendix B.

*Introduction.* Recall W has a complete polarization  $W = X \oplus Y$ . On  $S(X(\mathbb{A}))$ , there is a distribution  $\theta$  defined by

$$\theta(\phi) = \sum_{l \in X(F)} \phi(l).$$

It can be proved that for each  $\phi \in S(X(\mathbb{A}))$ , the sum on the right converges absolutely. Let the Jacobi group J(W) be the semidirect product of H(W) and Sp(W). Put  $\widetilde{J}(W)(\mathbb{A}) = H(W)(\mathbb{A}) \cdot \widetilde{Sp}(W)(\mathbb{A})$ . Use  $\widetilde{g} \cdot \phi$  to represent the Weil representation of  $\widetilde{g} \in \widetilde{Sp}(W)(\mathbb{A})$  on  $\phi$ . So for each  $\phi \in S(X(\mathbb{A}))$ , the theta function  $\theta_{\phi}$  on  $\widetilde{J}(W)(\mathbb{A})$  is defined as

$$\theta_{\phi}((w,t)\tilde{g}) = \sum_{l \in X(F)} \rho_{\psi}(w,t)(\tilde{g} \cdot \phi)(l).$$

It is J(W)(F)-invariant.

When  $\chi_m$  is chosen, the splitting  $s_{1,\chi_m} : U(V_1) \to \widetilde{Sp}(W)$  is fixed as stated in Proposition 3.2. Let

$$\omega(g_1, g_2)\phi(x) = \omega_{1,\chi_m}(g_1)\phi(g_2^{-1}x)$$
(14)

for  $g_1 \in U(V_1)$  and  $g_2 \in U(V_2)$ . Thus  $\theta_{\phi}$  can also be regarded as a function on the dual reductive pair  $(U(V_1), U(V_2))$ , and

$$\theta_{\phi}(g_1, g_2) = \theta(\omega(g_1, g_2)\phi). \tag{15}$$

Theta liftings. From now on, assume that  $\dim_K V_1 = 2$ ,  $\dim_K V_2 = 1$  and assume  $(V_1, (\cdot, \cdot)_1)$  is an anisotropic skew-Hermitian space. This is exactly the situation that one will see in next sections.

Let  $\Pi$  be a cuspidal representation contained in  $L^2(U(V_1)(F) \setminus U(V_1)(\mathbb{A}))$ . For convenience, given a reductive group G over F, we use [G] to denote the quotient  $G(F) \setminus G(\mathbb{A})$ . For each smooth  $\beta \in V_{\Pi}$ , where  $V_{\Pi}$  is the representation space of  $\Pi$ , the function

$$\theta_{\phi}^{\beta}(g_2) = \int_{[U(V_1)]} \theta_{\phi}(g_1, g_2) \beta(g_1) \, dg_1$$

is well-defined, where  $\theta_{\phi}(g_1, g_2)$  is given in (15). Actually,  $\theta_{\phi}^{\beta}$  determines a slowly increasing function on  $[U(V_2)]$ . It is expected that  $\theta_{\phi}^{\beta}(g_2)$  is nonzero and generates an irreducible automorphic representation on  $U(V_2)$ . So  $\theta_{\phi}^{\beta}$  defines a theta lifting from  $U(V_1)$  to  $U(V_2)$ .

Fourier-Jacobi coefficients of Eisenstein series on unitary groups

By (11), 
$$e(g_1, g_2) \cdot (v_1 \otimes v_2) = v_1 g_1 \otimes g_2^{-1} v_2$$
. So  
 $\theta_{\phi}^{\beta}(g_2) = \chi_{\Pi}(g_2) \int_{[U(V_1)]} \theta_{\phi}(g_1, 1) \beta(g_1) dg_1,$  (16)

where  $\chi_{\Pi}$  is the central character of  $\Pi$ . Whether  $\theta_{\phi}^{\beta}$  is zero or not only depends on the scalar  $\int_{[U(V_1)]} \theta_{\phi}(g_1, 1)\beta(g_1) dg_1$ . In fact,  $|\int_{[U(V_1)]} \theta_{\phi}(g_1, 1)\beta(g_1) dg_1|^2$  can be transformed into the Rallis inner product using the Siegel–Weil formula. I am going to introduce the Siegel–Weil formula in the following and show its application in the calculation of  $|\int_{[U(V_1)]} \theta_{\phi}(g_1, 1)\beta(g_1)dg_1|^2$ .

*The Siegel–Weil formula*. The Siegel–Weil formula on unitary groups relates the value of a Siegel–Eisenstein series to the integral of a theta function. We temporarily return to the general case. Consider the reductive pair (U(n, n), U(V)) with  $\dim_K V = m$ . Use  $S(V^n(\mathbb{A}))$  to denote the space of the Weil representation  $\omega_{\psi}$ . Here we take  $S(V_{\infty}^n)$  to be the subspace of  $L^2(V_{\infty}^n)$  called the Fock space, that is,

$$S(V_{\infty}^{n}) = \left\{ \phi : V_{\infty}^{n} \to \mathbb{C} \mid \phi(v_{1}, \dots, v_{n})\psi\left(i\sum_{i=1}^{n}|v_{i}|^{2}\right) \text{ is antiholomorphic as} \\ \text{a function of } v_{i} \text{ for } v_{i} \in V_{\infty} \right\}.$$

Fix  $\chi_m$  and  $\chi_{2n}$  such that the Weil representation on the reductive pair can be defined. For  $\phi \in S(V^n(\mathbb{A}))$ ,  $g \in U(n, n)(\mathbb{A})$  and  $u \in U(V)(\mathbb{A})$ , define such a theta integral:

$$I(g,\phi) = \int_{[U(V)]} \theta_{\phi}(g,u) \, du.$$

On the other hand, let  $\mathcal{P}'$  be the Siegel parabolic subgroup of U(n, n) and  $\mathcal{H}'$  be the maximal open compact subgroup. So  $U(n, n)(\mathbb{A}) = \mathcal{P}'(\mathbb{A})\mathcal{H}'(\mathbb{A})$ . For g = pk with  $p = \begin{pmatrix} A & * \\ (A^*)^{-1} \end{pmatrix} \in \mathcal{P}'(\mathbb{A})$  and  $k \in \mathcal{H}'(\mathbb{A})$ , put  $|a(g)|_K = |\det A|_K$ . Given  $\phi \in S(V^n(\mathbb{A}))$ , let

$$f_{\phi,s}(g) = |a(g)|_K^{s-s_0} \omega(g, 1)\phi(0), \tag{17}$$

where  $s_0 = (m - n)/2$ . Then  $f_{\phi,s} \in I_{\mathcal{P}'}^{U(n,n)}(\chi_m, s)$ . The Siegel-Eisenstein series can be defined as

$$E(g, f_{\phi,s}) = \sum_{\gamma \in \mathcal{P}'(F) \setminus U(n,n)(F)} f_{\phi,s}(\gamma g).$$

It has been proved in many cases that  $E(g, f_{\phi,s_0}) = I(g, \phi)$  if  $E(g, f_{\phi,s})$  is holomorphic at  $s = s_0$  and  $I(g, \phi)$  is absolutely convergent. The case we are interested in is when n = 2 and m = 1. By a result of Weil [1965],  $I(g, \phi)$  is absolutely convergent if V is anisotropic. So it is automatically satisfied if dim<sub>K</sub> V = 1.

**Theorem 3.6.** When n = 2 and m = 1,  $E(g, f_{\phi,s})$  is holomorphic at  $s = -\frac{1}{2}$  for all  $\phi \in S(V^n(\mathbb{A}))$ , and

$$E(g, f_{\phi,s})|_{s=-\frac{1}{2}} = 2I(g, \phi).$$

For the reductive pair  $(\text{Sp}_n, O(V))$  when *V* is an anisotropic *F*-vector space, the Siegel–Weil formula has been proved by Kudla and Rallis [1988]. The proof of Theorem 3.6 is very similar to theirs, and is found in Appendix A. Let us return to the theta lifting question. With the Siegel–Weil formula, we are ready to calculate  $|\int_{[U(V_1)]} \theta_{\phi}(g, 1)\beta(g) dg|^2$ . According to (16), we have

$$\left|\int_{[U(V_1)]} \theta_{\phi}(g, 1)\beta(g) \, dg\right|^2 = (\theta_{\phi}^{\beta}, \theta_{\phi}^{\beta}), \tag{18}$$

where  $(\theta_{\phi}^{\beta}, \theta_{\phi}^{\beta}) = \int_{[U(1)]} \theta_{\phi}^{\beta}(u) \overline{\theta_{\phi}^{\beta}}(u) du$ .

Proposition 3.7. We have

$$(\theta_{\phi}^{\beta}, \theta_{\phi}^{\beta}) = \frac{1}{2} \int_{U(V_1)(\mathbb{A})} f_{\delta_{\psi}(\phi \otimes \overline{\phi}), s}(\mathfrak{e}(g, 1)) \langle \Pi(g)\beta, \beta \rangle \, dg|_{s=-\frac{1}{2}},$$

where  $\langle \Pi(g)\beta,\beta\rangle = \int_{[U(V_1)]} \beta(g'g)\overline{\beta(g')} dg'$  is the matrix coefficient of the representation  $\Pi$ .

Proof. First,

$$\begin{aligned} (\theta_{\phi}^{\beta}, \theta_{\phi}^{\beta}) &= \int_{[U(1)]} \int_{[U(V_1)]} \theta_{\phi}(g, u) \beta(g) \int_{[U(V_1)]} \overline{\theta_{\phi}(g', u) \beta(g')} \, dg \, dg' \, du \\ &= \int_{[U(V_1) \times U(V_1)]} \beta(g) \overline{\beta(g')} \Big( \int_{[U(1)]} \theta_{\phi}(g, u) \overline{\theta_{\phi}(g', u)} \, du \Big) \, dg \, dg'. \end{aligned}$$

Notice that

$$\theta_{\phi}(g, u)\overline{\theta_{\phi}(g', u)} = \chi_1^{-1}(\det g')\theta_{\delta_{\psi}(\phi \otimes \overline{\phi})}(\mathfrak{e}(g, g'), u),$$

where  $\delta_{\psi}$  is defined by (13), and  $\mathfrak{e}(g, g') \in U(V_1 \oplus V_1^-)(\mathbb{A}) \simeq U(2, 2)(\mathbb{A})$ . So  $\theta_{\delta_{\psi}(\phi \times \overline{\phi})}$  is a theta function of the reductive pair (U(2, 2), U(1)). Applying Theorem 3.6, we have

$$\int_{[U(1)]} \theta_{\phi}(g, u) \overline{\theta_{\phi}(g', u)} \, du = \chi_1^{-1}(\det g') \int_{[U(1)]} \theta_{\delta_{\psi}(\phi \otimes \overline{\phi})}(\mathfrak{e}(g, g'), u) \, du$$
$$= \frac{1}{2} \chi_1^{-1}(\det g') E(\mathfrak{e}(g, g'), f_{\delta_{\psi}(\phi \otimes \overline{\phi}), s})|_{s=-\frac{1}{2}}.$$

Unfolding the Eisenstein series, we have

$$\begin{split} \int_{[U(V_1)\times U(V_1)]} \beta(g)\overline{\beta(g')}\chi_1^{-1}(\det g')E(\mathfrak{e}(g,g'),f_{\delta_{\psi}(\phi\times\bar{\phi}),s})\,dg\,dg'\\ = \int_{U(V_1(\mathbb{A}))} f_{\delta_{\psi}(\phi\times\bar{\phi}),s}(i(g,1))\langle\Pi(g)\beta,\beta\rangle\,dg. \quad \Box \end{split}$$

#### 4. Fourier-Jacobi coefficients of Eisenstein series

In Sections 2A and 2B, we first define the unitary group  $U(I_{\zeta})$  for

$$I_{\zeta} = \begin{pmatrix} 0 & 0 & 1 \\ 0 & \zeta & 0 \\ -1 & 0 & 0 \end{pmatrix}$$

with  $\zeta \in GL_2(K)$  such that  $\zeta = -\zeta^*$  and det  $\zeta \notin N_{K/F}(K^{\times})$ , and then define an Eisenstein series E(g, f, s) on  $U(I_{\zeta})$ . In this section, we are going to use the pullback formula introduced in Section 2C (refer to Theorem 2.6) to calculate the Fourier–Jacobi coefficients of E(g, f, s).

**4A.** Fourier–Jacobi coefficients of Siegel–Eisenstein series on U(3, 3). Recall that

$$U(3,3) = \left\{ g \in \operatorname{GL}_6(K) \mid g \begin{pmatrix} I_3 \\ -I_3 \end{pmatrix} g^* = \begin{pmatrix} I_3 \\ -I_3 \end{pmatrix} \right\}.$$

In Section 2B, we define a Siegel–Eisenstein series as follows: Take  $\mathbb{f} \in I_{\mathcal{P}}^{U(3,3)}(\eta, s)$  and define

$$\mathbb{E}(g,\mathbb{F},s) = \sum_{\gamma \in \mathcal{P}(F) \setminus U(3,3)(F)} \mathbb{F}_s(\gamma g).$$

In this section, we will define the Fourier–Jacobi coefficients of this Eisenstein series, and show that if the holomorphic section  $\mathbb{F}$  is chosen properly, the Fourier–Jacobi coefficient is a product of a theta function and a Siegel–Eisenstein series on U(2, 2).

Let  $H = \{(x, y, t) \mid x, y \in K^2, t \in F\} \subset U(3, 3)$ , where

$$(x, y, t) = \begin{pmatrix} 1 & x & t + \frac{1}{2}(xy^* - yx^*) & y \\ I_2 & y^* & 0_2 \\ & 1 & \\ & -x^* & I_2 \end{pmatrix}$$

Notice that

 $(x_1, y_1, t_1) \cdot (x_2, y_2, t_2) = \left(x_1 + x_2, y_1 + y_2, t_1 + t_2 + \frac{1}{2}(x_1 y_2^* + y_2 x_1^*) - \frac{1}{2}(x_2 y_1^* - y_1 x_2^*)\right).$ 

So if we take  $\mathbb{W}_d = \{(x, 0, 0)\}$  and  $\mathbb{W}^d = \{(0, y, 0)\}$ , then  $\mathbb{W} = \mathbb{W}_d \oplus \mathbb{W}^d$  is a symplectic space with the alternating pair

$$\langle (x_1, y_1), (x_2, y_2) \rangle = x_1 y_2^* + y_2 x_1^* - x_2 y_1^* - y_1 x_2^*,$$

and  $\mathbb{W}_d$  and  $\mathbb{W}^d$  are two isotropic subspaces. By the definition of Heisenberg groups in Section 3A, *H* is a Heisenberg group associated to  $\mathbb{W}$ , and

$$T = \{(0, 0, t), t \in F\}$$

is the center of *H*. There is another subgroup of U(3, 3):

$$U(\mathbb{W}_d \oplus \mathbb{W}^d) = \left\{ \begin{pmatrix} 1 & B \\ C & D \end{pmatrix} \middle| \begin{pmatrix} A & B \\ C & D \end{pmatrix} \in U(2, 2) \right\} \simeq U(2, 2).$$
(19)

The action of  $U(W_d \oplus W^d)$  on *H* is

$$\begin{pmatrix} A & B \\ C & D \end{pmatrix} \cdot (x, y, t) = \begin{pmatrix} A & B \\ C & D \end{pmatrix}^{-1} (x, y, t) \begin{pmatrix} A & B \\ C & D \end{pmatrix} = (xA + yC, xB + yD, t).$$

Denote by  $\psi_{\mathbb{Q}}$  the additive character of  $\mathbb{Q} \setminus \mathbb{A}_{\mathbb{Q}}$  with

$$\psi_{\mathbb{Q}}(x_{\infty}) = \exp(2\pi\sqrt{-1}x_{\infty}) \quad \text{for } x_{\infty} \in \mathbb{R}.$$

For  $m \in F$ , define an additive character  $\psi_m$  (later in this paper, we may simply denote it by  $\psi$ ) of  $F \setminus A$  by

$$\psi_m(x) = \psi_{\mathbb{A}}(\operatorname{Tr}_{F/\mathbb{Q}}(mx)) \quad \text{for } x \in \mathbb{A}.$$
 (20)

Define the Fourier–Jacobi coefficient of the Eisenstein series  $\mathbb{E}(\cdot, \mathbb{F}, s)$  associated to  $\psi$  as follows:

$$\mathbb{E}_{\psi}(g,\mathbb{F},s) = \int_{[T]} \mathbb{E}(tg,\mathbb{F},s)\psi(-t)\,dt \quad \text{for } g \in U(3,3)(\mathbb{A}).$$
(21)

So E admits the Fourier-Jacobi expansion

$$\mathbb{E}(g,\mathbb{F},s) = \sum_{m\in F} \mathbb{E}_{\psi_m}(g,\mathbb{F},s).$$

We can regard  $\mathbb{E}_{\psi}$  as a function on the semidirect product of *H* and U(2, 2), denoted by  $J_H$ . Thus  $\mathbb{E}_{\psi}(hg, \mathbb{F}, s) \in C_{\psi}^{\infty}([J_H])$ , where the subindex  $\psi$  means that the left action of *T* on the functions is given by  $\psi$ .

Lemma 4.1. (1) Let

$$\xi_0 = \begin{pmatrix} 0 & 1 \\ I_2 & 0 \\ -1 & 0 & I_2 \end{pmatrix}.$$

Then  $U(3,3) = \mathcal{P}J_H \sqcup \mathcal{P}\xi_0 J_H$ , where  $\mathcal{P}$  is the Siegel parabolic subgroup of U(3,3).

(2)  $\mathcal{P} \setminus \mathcal{P}\xi_0 J_H = \xi_0 \cdot \mathbb{W}^d \setminus H \cdot \mathcal{P}' \setminus U(2, 2) = w_3 \cdot \mathbb{W}_d \setminus H \cdot w_2 \cdot \mathcal{P}' \setminus U(2, 2)$ , where  $\mathcal{P}'$  is the Siegel parabolic subgroup of U(2, 2) and  $w_n = \begin{pmatrix} I_n \\ -I_n \end{pmatrix} \in U(n, n)$ .

Applying this lemma to  $\mathbb{E}_{\psi}(hg, \mathbb{F}, s)$ , we have

$$\mathbb{E}_{\psi}(hg, \mathbb{f}, s) = \int_{[T]} \psi(-t) \sum_{\gamma \in \mathcal{P}(F) \setminus U(3,3)(F)} \mathbb{f}_{s}(\gamma thg) dt$$
$$= \int_{[T]} \psi(-t) \left( \sum_{\gamma \in \mathcal{P}(F) \setminus \mathcal{P}(F) J_{H}(F)} + \sum_{\gamma \in \mathcal{P} \setminus \mathcal{P}(F) \xi_{0} J_{H}(F)} \mathbb{f}_{s}(\gamma thg) \right) dt$$
$$= \int_{[T]} \psi(-t) \sum_{\gamma \in w_{3} \cdot \mathbb{W}_{d}(F) \setminus H(F) \cdot w_{2} \cdot \mathcal{P}'(F) \setminus U(2,2)(F)} \mathbb{f}_{s}(\gamma thg) dt,$$
(22)

because  $\int_{[T]} \psi(-t) \sum_{\gamma \in \mathcal{P}(F) \setminus \mathcal{P}(F) J_H(F)} \mathbb{f}_s(\gamma thg) dt = 0.$ 

Pick a character  $\chi$  of  $K^{\times}$  such that  $\chi|_{\mathbb{A}^{\times}} = \epsilon_{K/F}$  and  $\chi_{\infty}(z) = |z|/z$  for  $z \in K_{\infty}^{\times}$ . We defined a Weil representation  $\omega_{\chi}$  of U(2, 2) on  $S(\mathbb{W}_d)$  by (12). Here what we actually use is a bit different from the conventional one. The difference is that instead of taking  $\omega_{\chi}(w_2)\phi(x) = \hat{\phi}(x)$  we take

$$\omega_{\chi}\left(\begin{pmatrix}1\\-1\\1\end{pmatrix}\right)\phi(x) = \hat{\phi}(x).$$

Fix such an  $\omega_{\chi}$ ; we are going to prove that if  $\mathbb{f}_s$  is chosen properly, then

$$\int_{T(\mathbb{A})} \mathbb{f}_s(w_3 t(0, y, 0) w_2 g') \psi(-t) dt = c_{\mathbb{F}, \psi} \omega_{\chi}(g') \phi_{\mathbb{F}}(y) R_s(g'), \qquad (23)$$

where  $R_s(g') \in I_{\mathcal{P}'}^{U(2,2)}(\eta\chi^{-1}, s)$ ,  $c_{\mathbb{F},\psi}$  is a nonzero constant and  $\phi_{\mathbb{F}} \in S(\mathbb{W}_d(\mathbb{A}))$ . Since the integral (23) can be decomposed into the product of local integrals, we can do the calculations place by place. The results are stated in Theorem 4.9 at the end of this section. Take a subgroup  $\mathcal{H}(\mathbb{A}) = \prod \mathcal{H}_v$  of U(3, 3)( $\mathbb{A}$ ), such that  $\mathcal{H}_\infty$  is a maximal compact subgroup of U(3, 3)<sub> $\infty$ </sub>, and  $\mathcal{H}_v$  is a maximal open compact subgroup of U(3, 3)<sub>v</sub>. Let  $\mathbb{F} = \otimes' \mathbb{F}_v$ , with  $\mathbb{F}_\infty$  being right  $\mathcal{H}_\infty$ -finite and  $\mathbb{F}_v$  being spherical for almost all finite places v. By spherical, I mean that  $\mathbb{F}_v$  is  $\mathcal{H}_v$ -invariant. In the following, the computations are purely local, so we omit the subindex v.

*Case:*  $\mathbb{f}_s$  *is spherical and*  $\chi$  *is unramified.* In this case, we make these assumptions: **Assumption 4.2.** (1). Assume  $\mathbb{f}$  is normalized, so  $\mathbb{f}(1) = 1$ .

(2). Assume  $\mathbb{W}_d \cap \mathcal{H} = \mathbb{O}_K^2$ . The dual of  $\mathbb{W}_d \cap \mathcal{H}$  with respect to  $\psi(xy^* + yx^*)$  is defined as

$$(\mathbb{W}_d \cap \mathscr{K})^{\vee} = \{(0, y, 0) \in \mathbb{W}^d \mid \psi(xy^* + yx^*) = 1 \text{ for all } (x, 0, 0) \in \mathbb{W}_d \cap \mathscr{K}\}$$
  
We assume  $(\mathbb{W}_d \cap \mathscr{K})^{\vee}$  is also  $\mathbb{O}_K^2$  and  $(\mathbb{W}_d \cap \mathscr{K})^{\vee} = \mathbb{W}^d \cap \mathscr{K}.$ 

**Lemma 4.3.**  $\int_T \mathbb{f}_s(w_3(0, y, t)w_2)\psi(-t) dt = \Phi_{\mathbb{O}_K^2}(y)$ , where  $\Phi_{\mathbb{O}_K^2}$  is the characteristic function supported on  $\mathbb{O}_K^2$ .

Proof. Take

$$\boldsymbol{x} = \begin{pmatrix} 1 & & \\ -x^* & I_2 & \\ & 1 & x \\ & & I_2 \end{pmatrix} \in \mathcal{P}$$

and notice that  $\mathbf{x}w_3 = w_3(x, 0, 0)$ . So,

$$\mathbf{x}w_3t(0, y, 0)w_2 = w_3(0, y, t + xy^* + yx^*)(x, 0, 0)w_2.$$

Then

$$\begin{split} \int_{T} \mathbb{f}_{s}(w_{3}(0, y, t)w_{2})\psi(-t) dt \\ &= \int_{T} \mathbb{f}_{s}(xw_{3}(0, y, t)w_{2})\psi(-t) dt \\ &= \int_{T} \mathbb{f}_{s}(w_{3}(0, y, t+xy^{*}+yx^{*})(x, 0, 0)w_{2})\psi(-t) dt \\ &= \psi(xy^{*}+yx^{*}) \int_{T} \mathbb{f}_{s}(w_{3}(0, y, t)(x, 0, 0)w_{2})\psi(-t) dt. \end{split}$$

For  $(0, y, 0) \in \mathbb{W}^d \cap \mathcal{K}$ ,

$$\int_T \mathbb{f}_s(w_3(0, y, t)w_2)\psi(-t) \, dt = \int_T \mathbb{f}_s(w_3t)\psi(-t) \, dt = 1.$$

If  $(0, y, 0) \notin \mathbb{W}^d \cap \mathcal{H}$ , there must exist  $(x, 0, 0) \in \mathbb{W}_d \cap \mathcal{H}$ , such that  $\psi(xy^* + yx^*) \neq 1$ . So,

$$\begin{split} \int_{T} \mathbb{f}_{s}(w_{3}(0, y, t)w_{2})\psi(-t) \, dt \\ &= \int_{T} \mathbb{f}_{s} \left( \left( \int_{-x^{*}}^{1} I_{2} \int_{T}^{x} w_{3}(0, y, t)w_{2} \right) \psi(-t) \, dt \\ &= \psi(xy^{*} + yx^{*}) \int_{T} \mathbb{f}_{s}(w_{3}(0, y, t)(x, 0, 0)w_{2})\psi(-t) \, dt \\ &= \psi(xy^{*} + yx^{*}) \int_{T} \mathbb{f}_{s}(w_{3}(0, y, t))\psi(-t) \, dt. \end{split}$$

Then  $\int_T \mathbb{f}_s(w_3(0, y, t)w_2)\psi(-t)dt = 0$ , if  $(0, y, 0) \notin \mathbb{W}^d \cap \mathcal{K}$ .

We have the Iwasawa decomposition  $U(2, 2) = \mathcal{P}'\mathcal{K}'$ , where  $\mathcal{P}'$  is the Siegel parabolic subgroup of U(2, 2),  $\mathcal{K}'$  is the maximal open compact subgroup and

$$\begin{aligned} \mathscr{H}' &= \mathscr{H} \cap U(2, 2). \text{ Take } g' = p'k' = \binom{a \ b \ (a^*)^{-1}}{k'} k' \in U(2, 2), \text{ where } k' \in \mathscr{H}'. \text{ Then} \\ &\int_{T} \mathbb{f}_{s} \left( w_{3}(0, y, t) w_{2} \begin{pmatrix} a \ b \ (a^*)^{-1} \end{pmatrix} k' \right) \psi(-t) dt \\ &= \int_{T} \mathbb{f}_{s} \left( w_{3}(0, y, t) \begin{pmatrix} (a^*)^{-1} \\ -b \ a \end{pmatrix} \right) \psi(-t) dt \\ &= \int_{T} \mathbb{f}_{s} \left( \binom{a \ b \ (a^*)^{-1}}{(a^*)^{-1}} w_{3}(-yb, ya, t) \right) \psi(-t) dt \\ &= \eta(\det a) |\det a|^{s+\frac{3}{2}} \int_{T} \mathbb{f}_{s} (w_{3}(0, ya, t + yba^*y^*)) \psi(-t) dt \\ &= \eta(\det a) |\det a|^{s+\frac{3}{2}} \psi(yba^*y^*) \Phi_{\mathbb{O}_{K}^{2}}(ya) \\ &= \chi(\det a) |\det a|^{\frac{1}{2}} \Phi_{\mathbb{O}_{K}^{2}}(ya) \psi(yba^*y^*) \eta \chi^{-1}(\det a) |\det a|^{s+1} \\ &= \omega_{\chi}(p') \Phi_{\mathbb{O}_{K}^{2}}(y) R_{s}(g'). \end{aligned}$$

Notice that  $\omega_{\chi}(k')\Phi_{\mathbb{O}_{K}^{2}}=\Phi_{\mathbb{O}_{K}^{2}}$ . So

$$\int_{T} \mathbb{f}_{s}(w_{3}(0, y, t)w_{2}g')\psi(-t) dt = \omega_{\chi}(g')\Phi_{\mathbb{O}_{K}^{2}}(y)R_{s}(g')$$

with normalized spherical  $R_s \in I^{U(2,2)}_{\mathcal{P}'}(\eta\chi^{-1}, s)$ .

Archimedean places. Let  $U(n) = \{u \in GL_n(\mathbb{C}) \mid uu^* = I_n\}$  be the unitary group of degree n at an archimedean place. Take such an embedding:

$$\mathfrak{e}: U(3) \times U(3) \hookrightarrow U(3, 3),$$
$$(u, v) \mapsto \mathfrak{e}(u, v) = \begin{pmatrix} -\frac{i}{2}I_3 & \frac{i}{2}I_3 \\ \frac{1}{2}I_3 & \frac{1}{2}I_3 \end{pmatrix} \begin{pmatrix} u \\ v \end{pmatrix} \begin{pmatrix} iI_3 & I_3 \\ -iI_3 & I_3 \end{pmatrix}.$$

The Hermitian domain of a unitary group is defined in (2). Choose an initial point in the Hermitian domain of U(3, 3) to be  $i = iI_3$ , where we fix  $i = \sqrt{-1}$  once and for all. From Proposition 2.1, one choice of a maximal compact subgroup at an archimedean place for U(3, 3) is  $\mathcal{H} = \{g \in U(3, 3) \mid gi = i\}$ . Notice that the image of the embedding  $\mathfrak{e}$  defined above is exactly  $\mathcal{H}$ . So  $\mathcal{H} \simeq U(3) \times U(3)$ . Recall that U(2, 2), which is isomorphic to U( $\mathbb{W}_d \oplus \mathbb{W}^d$ ), is naturally embedded into U(3, 3) (refer to (19)). So we can choose compatibly the maximal compact subgroup  $\mathcal{H}'$  of U(2, 2) by  $\mathcal{H}' = \mathcal{H} \cap U(2, 2)$ , and obviously  $\mathcal{H}' \simeq U(2) \times U(2)$ . The initial point  $i' = iI_2$  of the Hermitian domain of U(2, 2) is invariant under the action of elements in  $\mathcal{H}'$ .

At archimedean places, take a weight (0, k) section  $\mathbb{f}_k$  such that

$$\mathbb{f}_k(g) = j(g, \boldsymbol{i})^{-k},$$

where *j* is the automorphy factor defined in (3). Assume  $\eta$  satisfies  $\eta(z) = \overline{z}^{l'}/z^l$ ; then l' + l = k, and  $\mathbb{f}_k \in I(\eta, l - 3/2)$ . Recall that  $[F : \mathbb{Q}] = r$ . Then for  $m \in F$ , let  $m_{\infty} = (m_1, m_2, \dots, m_r) \in F_{\infty} \simeq \mathbb{R}^r$ . Given  $z = (z_1, z_2, \dots, z_r) \in F_{\infty}$ ,  $\psi(z) = \exp(2\pi i m_1 z_1) \exp(2\pi i m_2 z_2) \cdots \exp(2\pi i m_r z_r)$ .

**Proposition 4.4.** *Take*  $f_k$  *and*  $\psi$  *as above. We have* 

$$\int_{T} \psi(-t) \mathbb{f}_{k}(w_{3}(0, y, t)w_{2}g') dt = c_{\psi}\omega_{\chi}(g')\phi_{\mathbb{f}_{k}}(y)R_{k-1}(g'), \qquad (24)$$

where  $g' \in U(2, 2)$ ,  $\phi_{\mathbb{F}_k}(y) = \psi(iyy^*)$ ,  $\chi$  satisfies  $\chi(z) = |z|/z$  for  $z \in K_{\infty}^{\times}$ , and

$$c_{\psi} = \frac{(-2\pi i)^{kr} (m_1 m_2 \cdots m_r)^{k-1} e^{-2\pi (m_1 + m_2 + \cdots + m_r)}}{[(k-1)!]^r}$$

The function  $R_{k-1}$  satisfies  $R_{k-1}(g') = j(g', i')^{-k+1}$ . As an induced representation on U(2, 2), it is in  $I(\eta\chi^{-1}, l-\frac{3}{2})$  and is a weight (0, k-1) section.

*Proof.* For simplicity, we prove this proposition under the assumption that r = 1, from which the general result should be easily derived. So regard *m* as a real number first.

Step 1. Let  $g' = I_4$ . We have

$$\begin{split} \int_{\mathbb{R}} \psi(-t) \mathbb{f}_k(w_3(0, y, t)w_2) \, dt &= (-1)^k \int_{\mathbb{R}} \psi(-t)(t+i+iyy^*)^{-k} \, dt \\ &= (-1)^k \psi(iyy^*+i) \int_{\mathbb{R}+i+iyy^*} \psi(-t)t^{-k} \, dt \\ &= \frac{(-2\pi i)^k m^{k-1} e^{-2\pi m}}{(k-1)!} \, \psi(iyy^*). \end{split}$$

This calculation hints that we should take  $\phi_{\mathbb{F}_k}$  to be  $\psi(iyy^*)$ .

Step 2. Replace g' by

$$p'g' = \begin{pmatrix} a \\ (a^*)^{-1} \end{pmatrix} \begin{pmatrix} I_2 & b \\ I_2 \end{pmatrix} g'.$$

If we assume that

$$\int_{\mathbb{R}} \psi(-t) \mathbb{I}_{k}(w_{3}(0, y, t)w_{2}g') dt = \frac{(-2\pi i)^{k}m^{k-1}e^{-2\pi m}}{(k-1)!} \psi(iyy^{*})R_{k-1}(g'),$$

then

$$\begin{split} \int_{\mathbb{R}} \psi(-t) \mathbb{f}_{k}(w_{3}(0, y, t)w_{2}p'g') \, dt \\ &= (\overline{\det a})^{k} \psi(yaba^{*}y^{*}) \int_{\mathbb{R}} \psi(-t) \mathbb{f}_{k}(w_{3}(0, ya, t)w_{2}g') \, dt \\ &= \frac{(-2\pi i)^{k}m^{k-1}e^{-2\pi m}}{(k-1)!} (\overline{\det a})^{k} \psi(yaba^{*}y^{*}) \psi(iyaa^{*}y^{*}) R_{k-1}(g') \\ &= \frac{(-2\pi i)^{k}m^{k-1}e^{-2\pi m}}{(k-1)!} \omega_{\chi}(p'g') \phi_{\mathbb{F}_{k}}(y) R_{k-1}(p'g'). \end{split}$$

Step 3. Take  $g' = w_2 n' = w_2 \begin{pmatrix} I_2 & b \\ I_2 \end{pmatrix}$ . Note that we take

$$\omega_{\chi}\left(\begin{pmatrix}1\\-1\\1\end{pmatrix}\right)\phi(x) = \hat{\phi}(x),$$

so one easily checks that at archimedean places,  $\omega_{\chi}(w_2)\phi(x) = -\hat{\phi}(x)$ . Then,

$$\begin{split} \int_{\mathbb{R}} \psi(-t) \mathbb{f}_{k}(w_{3}(0, y, t)w_{2}w_{2}n' dt \\ &= (-1)^{k} \int_{\mathbb{R}} \psi(-t) \det \begin{pmatrix} t+i & -y \\ -y^{*} & b+i \end{pmatrix}^{-k} dt \\ &= (-1)^{k} \int_{\mathbb{R}} \psi(-t) \det(b+i)^{-k} (t+i-y(b+i)^{-1}y^{*})^{-k} dt \\ &= \frac{(-2\pi i)^{k} m^{k-1} e^{-2\pi m}}{(k-1)!} \det(b+i)^{-k} \psi(-y(b+i)^{-1}y^{*}) \\ &= \frac{(-2\pi i)^{k} m^{k-1} e^{-2\pi m}}{(k-1)!} \omega_{\chi}(w_{2}n') \phi_{\mathbb{f}_{k}}(y) R_{k-1}(w_{2}n'). \end{split}$$

Here we use that the Fourier transform of  $\psi(y(i+b)y^*)$  is det $(1-ib)^{-1}\psi(y(-i-b)^{-1}y^*)$ .

Step 4. Let  $g' = w_2 n'_1 w_2 n'_2$ , where  $n'_1 = \begin{pmatrix} I_2 & b_1 \\ I_2 \end{pmatrix}$  and  $n'_2 = \begin{pmatrix} I_2 & b_2 \\ I_2 \end{pmatrix}$ . Both  $b_1$  and  $b_2$  are  $2 \times 2$  Hermitian matrices. We have a decomposition

$$\begin{split} w_2 n'_1 w_2 n'_2 &= w_2 \begin{pmatrix} I_2 & b_1 \\ I_2 \end{pmatrix} \begin{pmatrix} \frac{1}{\sqrt{b_2^2 + 1}} & \frac{-b_2}{\sqrt{b_2^2 + 1}} \\ \sqrt{b_2^2 + 1} \end{pmatrix} \mathfrak{e} \left( \frac{-b_2 + i}{\sqrt{b_2^2 + 1}}, \frac{-b_2 - i}{\sqrt{b_2^2 + 1}} \right) \\ &= \begin{pmatrix} \sqrt{b_2^2 + 1} \\ 1/\sqrt{b_2^2 + 1} \end{pmatrix} w_2 \begin{pmatrix} I_2 & -b_2 + \sqrt{b_2^2 + 1} b_1 \sqrt{b_2^2 + 1} \\ I_2 \end{pmatrix} \\ &\cdot \mathfrak{e} \left( \frac{-b_2 + i}{\sqrt{b_2^2 + 1}}, \frac{-b_2 - i}{\sqrt{b_2^2 + 1}} \right) \end{split}$$

From Steps 2 and 3, one can see that when  $g' \in \mathcal{P}' w_2 \mathcal{N}' \subset U(2, 2)$ , where  $\mathcal{N}'$  is the unipotent radical of  $\mathcal{P}'$ , the equality (24) holds. Then let us prove that when taking  $g' = w_2 n'_1 w_2 n'_2$ , the equality still holds. We have

$$\mathbb{f}_{k}(w_{3}(0, y, t)w_{2}w_{2}n_{1}w_{2}n_{2}) = \mathbb{f}_{k}(w_{3}(0, y, t)w_{2}p'w_{2}n')\det\left(\frac{b_{2}+i}{\sqrt{b_{2}^{2}+1}}\right)^{-k},$$

where

$$p' = \begin{pmatrix} \sqrt{b_2^2 + 1} \\ 1/\sqrt{b_2^2 + 1} \end{pmatrix} \text{ and } n' = \begin{pmatrix} I_2 & -b_2 + \sqrt{b_2^2 + 1} \\ I_2 \end{pmatrix}.$$

So

$$\begin{split} &\int_{\mathbb{R}} \psi(-t) \mathbb{f}_{k}(w_{3}(0, y, t)w_{2}w_{2}n_{1}w_{2}n_{2}) dt \\ &= \det\left(\frac{b_{2}+i}{\sqrt{b_{2}^{2}+1}}\right)^{-k} \int_{\mathbb{R}} \psi(-t) \mathbb{f}_{k}(w_{3}(0, y, t)w_{2}p'w_{2}n') dt \\ &= \frac{(-2\pi i)^{k}m^{k-1}e^{-2\pi m}}{(k-1)!} \det\left(\frac{b_{2}+i}{\sqrt{b_{2}^{2}+1}}\right)^{-k} \omega_{\chi}(p'w_{2}n')\phi_{\mathbb{F}_{k}}(y)R_{k-1}(p'w_{2}n') \\ &= \frac{(-2\pi i)^{k}m^{k-1}e^{-2\pi m}}{(k-1)!} \det\left(\frac{b_{2}+i}{\sqrt{b_{2}^{2}+1}}\right)^{-1} \omega_{\chi}(p'w_{2}n')\phi_{\mathbb{F}_{k}}(y)R_{k-1}(w_{2}n_{1}w_{2}n_{2}). \end{split}$$

Note that

$$\omega_{\chi}(p'w_{2}n')\phi_{\mathbb{f}_{k}}(y) = -\frac{1}{\det(\sqrt{b_{2}^{2}+1})}\frac{1}{\det(\frac{b_{2}i+1}{b_{2}^{2}+1}-b_{1}i)}\psi\Big(y\Big(\frac{b_{2}-i}{b_{2}^{2}+1}-b_{1}\Big)^{-1}y^{*}\Big).$$

Also,

$$\omega_{\chi}(w_{2}n'_{1}w_{2}n'_{2})\phi_{\mathbb{F}_{k}}(y) = \frac{1}{\det(1-b_{2}i)}\frac{1}{\det((1-b_{2}i)^{-1}-b_{1}i)}\psi(y((b_{2}+i)^{-1}-b_{1})^{-1}y^{*}).$$

By comparison,

$$\int_{\mathbb{R}} \psi(-t) \mathbb{f}_{k}(w_{3}(0, y, t)w_{2}g') dt = \frac{(-2\pi i)^{k}m^{k-1}e^{-2\pi m}}{(k-1)!} \omega_{\chi}(g')\phi_{\mathbb{f}_{k}}(y)R_{k-1}(g')$$
for  $g' = w_{2}n'_{1}w_{2}n'_{2}$ .

Step 5. It is known that elements in U(2, 2) can be generated by  $w_2$  and  $p' \in \mathcal{P}'$ , where  $\mathcal{P}'$  is the Siegel parabolic subgroup of U(2, 2). The following lemma shows how these elements generate U(2, 2).

Lemma 4.5. 
$$U(2, 2) = \mathfrak{P}' \cup \mathfrak{P}' w_2 \mathcal{N}' \cup \mathfrak{P}' w_2 \mathcal{N}' w_2 \mathcal{N}'.$$

The proof is straightforward, so let me skip it. In Steps 1 to 4, we have verified that

$$\int_{\mathbb{R}} \psi(-t) \mathbb{f}_{k}(w_{3}(0, y, t)w_{2}g') dt = \frac{(-2\pi i)^{k} m^{k-1} e^{-2\pi m}}{(k-1)!} \omega_{\chi}(g') \phi_{\mathbb{f}_{k}}(y) R_{k-1}(g')$$

for g' in each of the subsets of U(2, 2).

For each archimedean place, the computations are exactly as above as long as we change *m* to the corresponding  $m_i$ . Putting them together, we can prove the proposition.

 $\mathbb{f}_s$  is not spherical. Denote the set of places where  $\mathbb{f}_s$  is not the spherical element in the induced representation space by *S*. When  $\mathbb{f}_s$  is not spherical, we make such a choice:  $\mathbb{f}_s \in I(\eta, s)$  and  $\mathbb{f}_s$  is supported in the big cell  $\mathcal{P}w_3\mathcal{P}$  such that  $\mathbb{f}_s(w_3n(b)) = \Phi(b)$ , where  $n(b) = \binom{I_3 \ b}{I_3}$ , and *b* is a 3 × 3 Hermitian matrix. Let  $b = \binom{t}{y^* \ b'}$ , where  $t \in F$  and *b'* is a 2 × 2 Hermitian matrix. Assume  $\Phi(b) = \phi'(t)\phi(y)\Phi'(b')$ such that  $\phi, \phi', \Phi'$  are all Bruhat–Schwartz functions. Further we make these assumptions concerning  $\mathbb{f}_s$  and the additive character  $\psi$ :

- **Assumption 4.6.** (1) The set *S* includes all the places where  $\eta$  or  $\chi$  is ramified. In another words, when  $\chi$  or  $\eta$  is ramified, we should take  $\mathbb{F}_s$  supported on the big cell, and as long as such an  $\mathbb{F}_s$  satisfies the following two assumptions, Theorem 4.9 can be derived. However, for the sake of later computations in Remark 4.14, we need to further assume (28).
- (2) For  $b' \in \operatorname{supp} \Phi'$  and  $y \in \operatorname{supp} \phi$ , we have  $\psi(yb'y^*) = 1$ .
- (3) When  $t \in \operatorname{supp} \phi'$ , we have  $\psi(t) = 1$ .

**Remark 4.7.** The assumptions above are in general quite weak. In practice, we may first determine  $\Phi'$  and  $\phi$ . Because they are both compactly supported, as a set of *F*,  $\{yb'y^* \mid y \in \text{supp } \Phi', b' \in \text{supp } \phi\}$  must be compact. So there exists an additive character  $\psi$  (in fact, infinitely many such characters) that is constant of value 1 on this set. Moreover, we can even choose a universal  $\psi$  that is independent of the conductors of  $\eta$  and  $\chi$ . Then based on Assumption 4.6(3), one can determine the function  $\phi'$ . Then in this way, we can determine  $\Phi$ , and in turn  $\mathbb{f}_s$ .

**Lemma 4.8.** Suppose  $R_s$  is in  $I_{\mathcal{P}'}^{U(2,2)}(\eta\chi^{-1}, s)$  and is supported on the big cell  $\mathcal{P}'w_2\mathcal{P}'$ , such that

$$R_s\left(w_2\begin{pmatrix}I_2&b'\\&I_2\end{pmatrix}\right) = \Phi'(b').$$

Let  $\phi_{\mathbb{f}_s}(y) = \hat{\phi}(-y)$ . Then

$$\int_T \psi(-t) \mathbb{f}_s(w_3(0, y, t) w_2 g') dt = \left(\int_T \psi(-t) \phi'(t) dt\right) \omega_{\chi}(g') \phi_{\mathbb{f}_s}(y) R_s(g').$$

*Proof.* In order that  $w_3(0, y, t)w_2g' \in \mathcal{P}w_3\mathcal{P}$ , it must be that g' is in  $\mathcal{P}'w_2\mathcal{P}'$ , which implies that  $R_s$  is supported on  $\mathcal{P}'w_2\mathcal{P}'$ . Let  $g' = w_2n'(b') = w_2\binom{I_2 \ b'}{I_2}$  for  $n'(b') \in \mathcal{N}'$ ; then

$$\begin{split} \int_{T} \psi(-t) \mathbb{f}_{k}(w_{3}(0, y, t)w_{2}w_{2}n'(b')) \, dt \\ &= \int_{T} \psi(-t) \Phi\left( \begin{pmatrix} t & y \\ y^{*} & b' \end{pmatrix} \right) dt \\ &= \left( \int_{T} \psi(-t) \phi'(t) \, dt \right) \phi(y) \Phi'(b') \\ &= \left( \int_{T} \psi(-t) \phi'(t) \, dt \right) \omega_{\chi}(w_{2}n'(b')) \phi_{\mathbb{f}_{s}}(y) R_{s}(w_{2}n'(b')). \quad \Box \end{split}$$

Let me summarize the local computations of the three cases above in the following theorem. Let *S* be a finite set of local places, such that  $f_{k,v}$  is not spherical if and only if  $v \in S$ .

**Theorem 4.9.** For a Hecke character  $\eta$  of K and  $\eta_{\infty}(z) = \overline{z}^{k-l}/z^{l}$ , take

$$\mathbb{f}_k \in I^{U(3,3)(\mathbb{A})}_{\mathcal{P}(\mathbb{A})}\left(\eta, l-\frac{3}{2}\right).$$

Assume that  $\mathbb{f}_{k,\infty}(g) = j(g, \mathbf{i})^{-k}$ , and  $\mathbb{f}_{k,v}$  is supported on the big cell when  $v \in S$ . Choose a Hecke character  $\chi$  of K so that  $\chi_{\infty}(z) = |z|/z$ . Fix an additive character  $\psi$  of F as in (20). When  $\mathbb{f}_k$ ,  $\chi$  and  $\psi$  satisfy Assumptions 4.2 and 4.6, we have

$$\mathbb{E}_{\psi}(hg') = c_{\mathbb{F}_k,\psi}\theta_{\phi_{\mathbb{F}_k}}(hw_2^{-1}g')\mathscr{E}(g',R_{k-1}) \quad for \ h \in H(\mathbb{A}), \ g' \in U(2,2)(\mathbb{A}).$$

where

$$c_{\mathbb{f}_k,\psi} = c_{\psi} \prod_{v \in S} \operatorname{vol}(\operatorname{supp} \phi'_v),$$

 $c_{\psi}$  is given in Proposition 4.4,  $\phi_{\mathbb{F}_k} \in S(\mathbb{W}^d(\mathbb{A}))$ , and  $R_{k-1} \in I^{U(2,2)(\mathbb{A})}_{\mathcal{P}'(\mathbb{A})}(\eta\chi^{-1}, l-\frac{3}{2})$ . Specifically,  $R_{k-1,\infty}(g') = j(g', i')^{-k+1}$  and  $R_{k-1,v}$  is normalized and spherical when  $v \notin S$ . Otherwise,  $R_{k-1,v}$  is supported on the big cell. The Siegel–Eisenstein series  $\mathscr{E}$  is associated to  $R_{k-1}$ .

Proof. Recall (22):

$$\mathbb{E}_{\psi}(hg',\mathbb{F}_k) = \int_{[T]} \psi(-t') \sum_{\substack{\gamma \in w_3 \cdot \mathbb{W}_d(F) \setminus H(F) \cdot w_2 \\ \cdot \mathscr{P}'(F) \setminus U(2,2)(F)}} \mathbb{F}_k(\gamma t' hg') dt'.$$

Let  $h = (x, y, t) \in H(\mathbb{A})$ ; then

$$\begin{split} \mathbb{f}_k(w_3hw_2g')) &= \mathbb{f}_k\Big(w_3(x,0,0)(0,\,y,\,t-\frac{1}{2}(xy^*+yx^*))w_2g'\Big) \\ &= \mathbb{f}_k\Big(w_3(0,\,y,\,t-\frac{1}{2}(xy^*+yx^*))w_2g'\Big). \end{split}$$
So we get

$$\begin{split} \mathbb{E}_{\psi}(hg', \mathbb{f}_{k}) &= \int_{T(\mathbb{A})} \psi(-t') \sum_{\substack{y_{0} \in \mathbb{W}^{d}(F) \\ g_{0} \in \mathscr{P}'(F) \setminus U(2,2)(F)}} \mathbb{f}_{k}(w_{3}(0, y_{0}, t')w_{2}g_{0}hg') dt' \\ &= \int_{T(\mathbb{A})} \psi(-t') \sum_{\substack{y_{0} \in \mathbb{W}^{d}(F) \\ g_{0} \in \mathscr{P}'(F) \setminus U(2,2)(F)}} \mathbb{f}_{k}(w_{3}(0, y_{0}, t')((x, y)g_{0}^{-1}w_{2}^{-1}, t)w_{2}g_{0}g') dt' \\ &= c_{\psi} \prod_{v \in S} \int_{T_{v}} \psi_{v}(-t')\mathbb{f}_{k,v}(w_{3}(0, 0, t')) dt' \\ &\quad \cdot \sum_{\substack{y_{0} \in \mathbb{W}^{d}(F) \\ g_{0} \in \mathscr{P}'(F) \setminus U(2,2)(F)}} \omega_{\chi}(((x, y)g_{0}^{-1}w_{2}^{-1}, t)g_{0}g) \phi_{\mathbb{f}_{k}}(y_{0})R_{k-1}(g_{0}g'). \end{split}$$

Notice that  $\int_{T_v} \psi_v(-t') \mathbb{I}_{k,v}(w_3(0,0,t')) dt' = \operatorname{vol}(\operatorname{supp} \phi'_v)$  for  $v \in S$ . So

$$\mathbb{E}_{\psi}(hg', \mathbb{f}_k) = c_{\mathbb{f}_k, \psi} \theta_{\phi_{\mathbb{f}_k}}(((x, y)w_2^{-1}, t)g') \mathscr{E}(g', R_{k-1})$$
$$= c_{\mathbb{f}_k, \psi} \theta_{\phi_{\mathbb{f}_k}}(hw_2^{-1}g') \mathscr{E}(g', R_{k-1}).$$

**4B.** *Fourier–Jacobi coefficients of Eisenstein series on* U(3, 1). In this section, I will define the Fourier–Jacobi coefficients of the Eisenstein series E(g, f, s)on U( $I_{\zeta}$ ). If this E(g, f, s) is from the pullback of a Siegel–Eisenstein series  $\mathbb{E}(\cdot, \mathbb{F}, s)$ , then by applying the pullback formula (Theorem 2.6) and results about the Fourier–Jacobi coefficients of  $\mathbb{E}(\cdot, \mathbb{F}, s)$  (Theorem 4.9), we will get formulas for the Fourier–Jacobi coefficients of E(g, f, s).

*Definitions.* Let *P* be the only nontrivial parabolic subgroup *P* of  $U(I_{\zeta})$ ; then the unipotent radical of *P* is

$$N = \left\{ \begin{pmatrix} 1 & x & t + x\zeta x^*/2 \\ I_2 & \zeta x^* \\ & 1 \end{pmatrix} \middle| t \in F \right\}.$$

From another point of view, N can be regarded as a Heisenberg group attached to a 4-dimensional symplectic space W of F. Using conventional notation, denote

$$\begin{pmatrix} 1 & x & t + x\zeta x^*/2 \\ I_2 & \zeta x^* \\ & 1 \end{pmatrix} \quad \text{by } (x, t);$$

then

$$(x_1, t_1)(x_2, t_2) = (x_1 + x_2, t_1 + t_2 + \langle x_1, x_2 \rangle/2),$$

Bei Zhang

where  $\langle \cdot, \cdot \rangle$  represents the alternating pair on W, and

$$\langle x_1, x_2 \rangle = x_1 \zeta x_2^* - x_2 \zeta x_1^*.$$

The degree two unitary group  $U(\zeta) = \{u \mid u\zeta u^* = \zeta\}$  is a subgroup of  $U(I_{\zeta})$ , and has an action on N by  $u \cdot (x, t) = u^{-1}(x, t)u = (xu, t)$ .

Given the additive character  $\psi$  as in (20), define the Fourier–Jacobi coefficient of E(g, f, s) as

$$E_{\psi}(nu, f, s) = \int_{[T]} \psi(-t) E(tnu, f, s) dt$$
 (25)

for  $u \in U(\zeta)(\mathbb{A})$  and  $n \in N(\mathbb{A})$ , where *T* is the center of *N*. So  $E_{\psi} \in C_{\psi}^{\infty}([NU(\zeta)])$ .

*Pullback of*  $\mathbb{E}_{\psi}$ . Recall in (8), we define an embedding  $\mathfrak{e}: U(I_{\zeta}) \times U(\zeta) \hookrightarrow U(3, 3)$ . See that

$$\mathfrak{e}|_{U(\zeta)\times U(\zeta)}: U(\zeta)\times U(\zeta) \hookrightarrow U(2,2) = U(\mathbb{W}_d \oplus \mathbb{W}^d) \subset U(3,3).$$

Also for  $(x, t) \in N$ , we have  $\mathfrak{e}((x, t), I_2) = (x\zeta/2, x, t)$ . This implies  $\mathfrak{e}(N, I_2) \subset H$ . Specifically, the center of N is mapped to the center of H under  $\mathfrak{e}$ , which explains why I use T to denote both centers of H and N.

From Theorem 2.6, we know that

$$E(g, f, s) = \int_{[U(\zeta)]} \mathbb{E}(\mathfrak{e}(g, u), \mathbb{f}, s)\beta(u)\eta^{-1}(\det u) \, du \tag{26}$$

for  $f_s(g) = \int_{U(\zeta)(\mathbb{A})} \mathbb{f}_s(\mathfrak{e}(g, u))\beta(u)\eta^{-1}(\det u) \, du \in I_p^{U(I_\zeta)}(\Pi \otimes \eta, s)$ , where  $\beta \in V_{\Pi}$ . So it is reasonable to infer that  $E_{\psi}$  is also the pullback of  $\mathbb{E}_{\psi}$ . The following proposition is easy to verify.

**Proposition 4.10.** If E(g, f, s) is defined by (26), then

$$E_{\psi}(nu, f, s) = \int_{[U(\zeta)]} \mathbb{E}_{\psi}(\mathfrak{e}(nu, u'), \mathbb{f}, s)\beta(u')\eta^{-1}(\det u') \, du'.$$

Let us study the relation between  $N \cdot U(\zeta)$  and  $H \cdot U(2, 2)$  induced by the embedding  $\mathfrak{e}$ . *N* is the Heisenberg group associated to the symplectic group *W*, and  $U(W) = U(\zeta)$ , while  $H \subset U(3, 3)$  is the Heisenberg group associated to  $\mathbb{W} = \mathbb{W}_d + \mathbb{W}^d$ , and  $U(\mathbb{W}) = U(2, 2)$ . Notice that

$$W \stackrel{\mathfrak{c}}{\hookrightarrow} \mathbb{W}, \quad x \mapsto \left(\frac{x\zeta}{z}, x\right).$$

At the same time, we can define another embedding of  $W^-$ :

$$W^- \hookrightarrow \mathbb{W}, \quad x \mapsto \left(-\frac{x\zeta}{z}, x\right).$$

So, the two embeddings combine to give

$$W + W^{-} \xrightarrow{\sim} W = W_{d} + W^{d},$$
$$(x, -x) \mapsto (x\zeta, 0),$$
$$(x, x) \mapsto (0, x).$$

Since *W* has the polarization W = X + Y, the Weil representation of U(2, 2) can be realized on S(X + X) or on  $S(\mathbb{W}_d)$ . Recall that we define an intertwining isometry operator  $\delta_{\psi}$  between the two representations by (13). It can be applied here with a little revision. Take  $\phi \in S(X + X)$ , and let  $\delta_{\psi}'(\phi)(w) = \delta_{\psi}(\phi)(w\zeta^{-1})$  be the corresponding function in  $S(\mathbb{W}_d)$ . If we take  $\phi = \phi_1 \otimes \overline{\phi}_2$  for  $\phi_1, \phi_2 \in S(X)$ , then

 $\omega_{\chi}(\mathfrak{e}(u_1, u_2))\delta'_{\psi}(\phi_1 \otimes \overline{\phi}_2) = \delta'_{\psi}(\omega_{\chi}(u_1)\phi_1 \otimes \chi(\det u_2)\check{\omega}_{\chi}(u_2)\overline{\phi}_2).$ 

It is straightforward to verify that

$$\theta_{\delta'_{u}}(\phi_1 \otimes \overline{\phi}_2)(\mathfrak{e}(nu_1, u_2)) = \chi(\det u_2)\theta_{\phi_1}(nu_1)\theta_{\phi_2}(u_2),$$

where  $\bar{\theta}_{\phi_2}(u_2) = \sum_{x \in X(F)} \check{\omega}_{\chi}(u_2) \overline{\phi_2}(x)$ . Then Proposition 4.10 and Theorem 4.9 give this:

**Corollary 4.11.** Assume that  $\phi_{\mathbb{F}_k} = \delta'_{\psi}(\phi_1 \otimes \overline{\phi}_2)$ , and  $E(g, f_k)$  is defined as the pullback of  $\mathbb{E}(\cdot, \mathbb{F}_k)$ , where  $\mathbb{F}_k$  satisfies the conditions of Theorem 4.9. Then

$$E_{\psi}(nu, f_k) = c_{\mathbb{F}_k, \psi} \theta_{\phi_1}(nu) \int_{[U(\zeta)]} \beta(u') \chi \eta^{-1}(\det u') \mathscr{E}(\mathfrak{e}(u, u'), R_{k-1}) \overline{\theta}_{\phi_2}(u') du'.$$

Inner product of  $E_{\psi}$  with  $\theta_{\varphi}$ . As we mentioned, both  $E_{\psi}$  and  $\theta_{\varphi}$  for  $\varphi \in S(X)$  are functions in  $C_{\psi}^{\infty}([NU(\zeta)])$ . Let  $L_{\psi}^{2}([NU(\zeta)])$  be the completion of  $C_{\psi}^{\infty}([NU(\zeta)])$  with respect to the inner product

$$\langle \theta, \theta' \rangle = \int_{[NU(\zeta)]} \theta(r) \overline{\theta'}(r) \, dr.$$

Treat  $E_{\psi}$  and  $\theta_{\varphi}$  as elements in  $L^2_{\psi}([NU(\zeta)])$ . Let us calculate  $\langle E_{\psi}, \theta_{\varphi} \rangle$ .

**Theorem 4.12.** In the setting above, we have

$$\begin{split} \langle E_{\psi}(nu, f_k), \theta_{\varphi}(nu) \rangle \\ &= c_{\mathbb{F}_k, \psi}(\phi_1, \varphi) \int_{U(\zeta)(\mathbb{A})} R_{k-1}(\mathfrak{e}(u, 1)) \, du \int_{[U(\zeta)]} \beta(u) \overline{\theta}_{\phi_2}(u) \, du, \\ \prod_{v \notin S} \int_{U(\zeta)_v} R_{k-1, v}(\mathfrak{e}(u, 1)) \, du = \frac{L^S(\eta \chi^{-1}, l - \frac{1}{2}) L^S(\eta \chi^{-1}, l - \frac{3}{2})}{L^S(\eta_F \epsilon_{K/F}, 2l - 1) L^S(\eta_F, 2l - 2)}, \end{split}$$

where S is the set of places introduced in Theorem 4.9, and  $L^S$  is the partial L-function skipping the factors at  $v \in S$ .

(I will discuss the case  $v \in S$  in Remark 4.14.)

*Proof.* We prove the theorem in three steps:

Step 1.  $\langle E_{\psi}(nu, f_k), \theta_{\varphi}(nu) \rangle = c_{\mathbb{f}_k, \psi}(\phi_1, \varphi) I(E_{\psi})$ , where

$$I(E_{\psi}) = \int_{[U(\zeta) \times U(\zeta)]} \beta(u') \chi \eta^{-1}(\det u') \mathscr{E}(\mathfrak{e}(u, u'), R_{k-1}) \overline{\theta}_{\phi_2}(u') \, du \, du'.$$

Step 2.  $I(E_{\psi}) = \int_{U(\zeta)(\mathbb{A})} R_{k-1}(\mathfrak{e}(u,1)) \, du \cdot \int_{[U(\zeta)]} \beta(u) \overline{\theta}_{\phi_2}(u) \, du.$ 

Step 3. We show

$$\int_{U(\zeta)_v} R_{k-1,v}(\mathfrak{e}(u,1)) \, du = \frac{L_v(\eta\chi^{-1}, l-\frac{1}{2})L_v(\eta\chi^{-1}, l-\frac{3}{2})}{L_v(\eta_F \epsilon_{K/F}, 2l-1)L_v(\eta_F, 2l-2)}$$

for a finite place  $v \notin S$ .

The equation in Step 1 is straightforward from Corollary 4.11, because first for  $\theta_{\phi_1}, \theta_{\varphi} \in L^2_{\psi}([NU(\zeta)])$ , we have  $\langle \theta_{\phi_1}, \theta_{\varphi} \rangle = (\phi_1, \varphi)$ ; then

To get the expression of  $I(E_{\psi})$  in Step 2, let us unfold  $\mathscr{E}(\cdot, R_{k-1})$ :

$$\mathscr{C}(\mathfrak{e}(u, u'), R_{k-1}) = \sum_{\gamma \in \mathscr{P}'(F) \setminus U(2,2)(F)} R_{k-1}(\gamma \mathfrak{e}(u, u'))$$
$$= \sum_{\gamma = \mathfrak{e}(u_1, u_2) \in \Delta(U(\zeta))(F) \setminus U(\zeta)(F) \times U(\zeta)(F)} R_{k-1}(\gamma \mathfrak{e}(u, u')),$$

where  $\Delta(U(\zeta))$  is the image of the diagonal embedding of  $U(\zeta)$  to  $U(\zeta) \times U(\zeta)$ . Then,

$$I(E_{\psi}) = \int_{\Delta(U(\zeta))(F)\setminus U(\zeta)(\mathbb{A})\times U(\zeta)(\mathbb{A})} \beta(u')\chi\eta^{-1}(\det u')\overline{\theta}_{\phi_{2}}(u')R_{k-1}(\mathfrak{e}(u,u'))\,du\,du'$$
  
$$\stackrel{u=u'v}{=} \int_{U(\zeta)(\mathbb{A})\times [U(\zeta)]} \beta(u')\chi\eta^{-1}(\det u')\overline{\theta}_{\phi_{2}}(u')R_{k-1}(\mathfrak{e}(u'v,u'))\,dv\,du'$$
  
$$= \int_{U(\zeta)(\mathbb{A})} R_{k-1}(\mathfrak{e}(u,1))\,du \cdot \int_{[U(\zeta)]} \beta(u)\overline{\theta}_{\phi_{2}}(u)\,du.$$

Now we are left with the calculation of  $\int_{U(\zeta)(\mathbb{A})} R_{k-1}(\mathfrak{e}(u, 1)) du$ . It can be written as the product of local integrals:

$$\int_{U(\zeta)(\mathbb{A})} R_{k-1}(\mathfrak{e}(u,1)) \, du = \prod_{v} \int_{U(\zeta)(F_v)} R_{k-1,v}(\mathfrak{e}(u,1)) \, du.$$

In the following computations, we drop the subscript v if this does not cause confusion.

First,  $v = \infty$ . Now  $R_{k-1,\infty}$  is defined by the automorphy factor, namely  $R_{k-1,\infty}(g') = j(g', i')^{-k+1}$ , and  $U(\zeta)(F_{\infty})$  is isomorphic to copies of the compact group  $U(2)(\mathbb{R})$ . By Proposition 4.4,  $R_{k-1}(\mathfrak{e}(u, 1)) = 1$ . So,

$$\int_{U(\zeta)(F_{\infty})} R_{k-1,\infty}(\mathfrak{e}(u,1)) \, du = \operatorname{vol}(U(\zeta)(F_{\infty})).$$

Second, v splits in F. Then  $U(\zeta)(F_v) \simeq GL_2(F_v)$ . Assume  $R_{k-1,v}$  is normalized and spherical. Now, K = F + F,  $\eta(a, b) = \eta_1(a)\eta_2(b)$  for  $(a, b) \in K$ , and similarly for  $\chi$ . Since  $\chi|_{\mathbb{A}^{\times}} = \epsilon_{K/F}$ , we have  $\chi_1\chi_2 = 1$ . Let us take the Godement section representation of  $R_{k-1}$ :

$$L(\eta_F, 2l-1)L(\eta_F, 2l-2)R_{k-1}(g)$$
  
=  $\eta_1 \chi_1^{-1}(\det g) |\det g|^{l-\frac{1}{2}} \int_{\mathrm{GL}_2(F)} \Phi_{M_{2\times 4}(\mathbb{O}_F)}((0, X)g)\eta_1\eta_2(\det X) |\det X|^{2l-1} dX,$ 

where  $\Phi_{M_{2\times 4}(\mathbb{O}_F)}$  is the characteristic function of  $M_{2\times 4}(\mathbb{O}_F)$ . Then

$$L(\eta_F, 2l-1)L(\eta_F, 2l-2)R_{k-1}(\mathfrak{e}(u, 1))$$
  
=  $\eta_1 \chi_1^{-1}(\det u) |\det u|^{l-\frac{1}{2}} \int_{\operatorname{GL}_2(F)} \Phi_{M_{2\times 4}(\mathbb{O}_F)}(X, Xu) \eta_1 \eta_2(\det X) |\det X|^{2l-1} dX.$ 

So,

$$\begin{split} L(\eta_F, 2l-1)L(\eta_F, 2l-2) & \int_{U(\zeta)} R_{k-1}(\mathfrak{e}(u, 1)) \, du \\ &= L(\eta_F, 2l-1)L(\eta_F, 2l-2) \int_{\operatorname{GL}_2(F)} R_{k-1}(\mathfrak{e}(u, 1)) \, du \\ &= \int_{\operatorname{GL}_2(F) \times \operatorname{GL}_2(F)} \eta_1 \chi_1^{-1}(\det u) |\det u|^{l-\frac{1}{2}} \\ & \Phi_{M_{2\times 4}(\mathbb{O}_F)}(X, Xu) \eta_1 \eta_2(\det X) |\det X|^{2l-1} \, dX \, du \\ Y &= \int_{\operatorname{GL}_2} \eta_1 \chi_1^{-1}(\det Y) |\det Y|^{l-\frac{1}{2}} \Phi_{\mathbb{O}_F^2}(Y) \, dY \\ & \cdot \int_{\operatorname{GL}_2} \eta_2 \chi_1(\det X) |\det X|^{l-\frac{1}{2}} \Phi_{\mathbb{O}_F^2}(X) \, dX \end{split}$$

Bei Zhang

which is equal to

$$L(\eta \chi^{-1}, l-\frac{1}{2})L(\eta \chi^{-1}, l-\frac{3}{2}).$$

Then we have

$$\int_{\mathrm{GL}_2(F)} R_{k-1}(\mathfrak{e}(u,1)) \, du = \frac{L(\eta \chi^{-1}, l-\frac{1}{2})L(\eta \chi^{-1}, l-\frac{3}{2})}{L(\eta F, 2l-1)L(\eta F, 2l-2)}$$

First,  $R_{k-1,v}$  is normalized and spherical, and  $U(\zeta)(F_v)$  is quasisplit over  $F_v$ . So  $U(\zeta) \simeq U(1, 1)$ . The embedding  $\mathfrak{e}$  of  $U(1, 1) \times U(1, 1)$  to U(2, 2) is inherited from the global definition of  $\mathfrak{e}|_{U(\zeta) \times U(\zeta)}$ .

Take the local Iwasawa decomposition U(1, 1) = PK = NMK, where N, Mand K are the unipotent radical, Levi part and maximal open compact subgroup, respectively. Then  $K = U(1, 1)(\mathbb{O}_F)$ , and  $\mathfrak{e}(K, K) \subset \mathcal{H}'$ , which is the maximal compact subgroup of U(2, 2). Let u = nmk; then  $du = \delta_P^{-1}(m) dn dm dk$ , where  $\delta_P$  is the modular character on P. If  $m = (\bar{\alpha}^{-1}{}_{\alpha})$ , then  $\delta_P(m)^{-1} = |\alpha|_K$ . Let

$$w_{13} = \begin{pmatrix} 1 & 1 \\ -1 & 1 \end{pmatrix},$$

then

$$\int_{U(\zeta)} R_{k-1}(\mathfrak{e}(u,1)) \, du = \int_{NM} R_{k-1}(\mathfrak{e}(nm,1)) \delta_P^{-1}(m) \, dn \, dm$$
$$= \int_M \int_F R_{k-1} \left( w_{13} \begin{pmatrix} 1 & 1 \\ & 1 \end{pmatrix} w_{13}^{-1} \mathfrak{e}(m,1) \right) \delta_P^{-1}(m) \, dx \, dm.$$

Consider such a function on U(2, 2):

$$R'_{k-1}(g) = \int_F R_{k-1} \left( w_{13} \begin{pmatrix} 1 & x \\ & 1 \\ & & 1 \end{pmatrix} g \right) dx.$$

We have

$$\int_{U(\zeta)} R_{k-1}(\mathfrak{e}(u,1)) \, du = \int_M R'_{k-1}(w_{13}^{-1}\mathfrak{e}(m,1))\delta_P^{-1}(m) \, dm.$$
(27)

Recall that  $U(2, 2) = U(\mathbb{W}) = U(\mathbb{W}_d + \mathbb{W}^d)$ . If we have

$$\mathbb{W}_d + \mathbb{W}^d = Ke_1 + Ke_2 + Kf_1 + Kf_2,$$

then a parabolic subgroup P' fixing both  $\mathbb{W}_d$  and  $Ke_2$  can be defined. An element  $p' \in P'$  looks like

$$\begin{pmatrix} a & & d \\ b & c & e & f \\ & a' & b' \\ & & & c' \end{pmatrix}.$$

**Lemma 4.13.**  $R'_{k-1}$  is in the space of the induced representations from P', so that

$$R'_{k-1}(p'gk') = \eta \chi^{-1}(a'c)|a'|_{K}^{l-\frac{3}{2}}|c|_{K}^{l-\frac{1}{2}}R'_{k-1}(g)$$

for  $p' \in P'$ ,  $k' \in \mathcal{K}'$  and  $g \in U(2, 2)$ .

This lemma can be proved by direct calculations, which I will skip here. Applying the lemma, we see  $R'_{k-1}$  has a Godement section representation as follows:

$$\frac{L(\eta_F \epsilon_{K/F}, 2l-1)}{L(\eta_F \epsilon_{K/F}, 2l-2)} R'_{k-1}(g) = c \int_{\mathrm{GL}_2(K)} \Phi_{M_{2\times 4}(\mathbb{O}_K)}((0, X)g) \eta^{-1} \chi(\det X) |\det X|_K^{-l+\frac{3}{2}} dX \cdot \int_{K^{\times}} \Phi_{\mathbb{O}_K^4}((0, 0, 0, Z)g) \eta \chi^{-1}(|Z|_K) |Z|_K^{2l-2} dZ,$$

where the normalizing constant c satisfies

$$1 = c \int_{\mathrm{GL}_{2}(K)} \Phi_{M_{2\times2}(\mathbb{O}_{K})}(X) \eta^{-1} \chi(\det X) |\det X|_{K}^{-l+\frac{3}{2}} dX \\ \cdot \int_{K^{\times}} \Phi_{\mathbb{O}_{K}}(Z) \eta \chi^{-1}(|Z|_{K}) |Z|_{K}^{2l-2} dZ,$$
$$R'_{k-1}(1) = \int_{F} R_{k-1} \left( w_{13} \begin{pmatrix} 1 & 1 & \\ & 1 & \\ & & 1 \end{pmatrix} \right) dx = \frac{L(\eta_{F} \epsilon_{K/F}, 2l-2)}{L(\eta_{F} \epsilon_{K/F}, 2l-1)}.$$
Let  $g = w_{13}^{-1} \mathfrak{e}(m, 1) = w_{13}^{-1} \mathfrak{e}(\left(\bar{\alpha}^{-1} \alpha\right), 1).$  It can be verified that

$$R'_{k-1}(w_{13}^{-1}\mathfrak{e}(m,1)) = \frac{\eta\chi^{-1}(\alpha)|\alpha|_{K}^{l-\frac{3}{2}}}{L(\eta_{F}\epsilon_{K/F},2l-1)L(\eta_{F},2l-2)} \int_{K^{\times}} \Phi_{\mathbb{O}_{K}}(Z)\Phi_{\mathbb{O}_{K}}(Z\alpha)\eta\chi^{-1}(|Z|_{K})|Z|_{K}^{2l-2} dZ.$$

Substituting the expression of  $R'_{k-1}(w_{13}\mathfrak{e}(m, 1))$  in (27), we have

$$\begin{split} \int_{U(\zeta)} R_{k-1}(\mathfrak{e}(u,1)) \, du \\ &= \int_{M} R'_{k-1}(w_{13}^{-1}\mathfrak{e}(m,1))\delta_{P}^{-1}(m) \, dm \\ &= \frac{1}{L(\eta_{F}\epsilon_{K/F}, 2l-1)L(\eta_{F}, 2l-2)} \\ &\quad \cdot \int_{K^{\times} \times K^{\times}} \eta \chi^{-1}(\alpha) |\alpha|_{K}^{l-\frac{1}{2}} \Phi_{\mathbb{O}_{K}}(Z) \Phi_{\mathbb{O}_{K}}(Z\alpha) \eta \chi^{-1}(|Z|_{K}) |Z|_{K}^{2l-2} \, dZ \, d\alpha \\ &= \frac{L(\eta \chi^{-1}, l-\frac{1}{2})L(\eta \chi^{-1}, l-\frac{3}{2})}{L(\eta_{F}\epsilon_{K/F}, 2l-1)L(\eta_{F}, 2l-2)}. \end{split}$$

Bei Zhang

**Remark 4.14.** In Theorem 4.12,  $\int_{U(\zeta)(F_v)} R_{k-1,v}(\mathfrak{e}(u, 1)) du$  was not explicitly calculated when  $v \in S$ , because we need more assumptions and things become more technical. Let me put it here. When  $v \in S$ ,  $R_{k-1,v}$  is supported in the big cell associated to a characteristic function  $\Phi'$  on  $\operatorname{Her}_2(F_v)$ . (One can refer to the section about nonspherical  $\mathbb{F}_k$ , especially Lemma 4.8.)

Skip the subindex v now. Fix an integral ideal  $\mathfrak{c}$  of  $\mathbb{O}_K$  so that  $\eta \chi^{-1}(1 + \mathfrak{c}) = 1$ . Then pick a totally imaginary element  $\delta \in \mathbb{O}_K$  satisfying  $1 + \delta \in \mathbb{O}_K^{\times}$ . Assume that

$$\operatorname{supp} \Phi' = \operatorname{Her}_2(F) \cap \,\delta\zeta^{-1}(2I_2 + 2\mathfrak{c}\operatorname{GL}_2(\mathbb{O}_K)).$$
(28)

Define  $D_1(\mathfrak{c})$ , a subset of  $U(\zeta)$ , by

$$D_1(\mathfrak{c}) = \left\{ u \in U(\zeta) \mid u = \begin{pmatrix} a & b \\ c & d \end{pmatrix}, a, d \in -\frac{1+\delta}{1-\delta} + \delta\mathfrak{c}, b, c \in \delta\mathfrak{c} \right\}.$$

**Lemma 4.15.** Let  $u \in U(\zeta)$ . Then

 $\mathfrak{e}(u, 1) \in \operatorname{supp} R_{k-1} \Longleftrightarrow u \in D_1(\mathfrak{c}).$ 

*Proof.* First notice that for a matrix in U(2, 2),

$$\begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \operatorname{supp} R_{k-1} \Longleftrightarrow c^{-1}d \in \operatorname{supp} \Phi'.$$

By the definition of the embedding e in (8),

$$\mathbf{e}(u, 1) = \begin{pmatrix} (\zeta^{-1}(u+I_2)\zeta)/2 & \zeta^{-1}(u-I_2) \\ ((u-I_2)\zeta)/4 & (u+I_2)/2 \end{pmatrix}$$

So

$$\mathfrak{e}(u, 1) \in \operatorname{supp} R_{k-1} \iff 2\zeta^{-1}(u - I_2)^{-1}(u + I_2) \in \operatorname{supp} \Phi'$$
  
 $\iff u \in D_1(\mathfrak{c}).$ 

Then we get

$$R_{k-1}(\mathfrak{e}(u,1)) = \begin{cases} \eta \chi^{-1}(\delta) & \text{if } u \in D_1(\mathfrak{c}), \\ 0 & \text{otherwise.} \end{cases}$$

The integral

$$\int_{U(\zeta)} R_{k-1}(\mathfrak{e}(u,1)) \, du = \eta \chi^{-1}(\delta) \operatorname{vol}(D_1(\mathfrak{c})).$$

As we see from Theorem 4.12,  $\langle E_{\psi}, \theta_{\varphi} \rangle$  equals the product of an explicit constant and the integral  $\int_{[U(\zeta)]} \beta(u) \overline{\theta}_{\phi_2}(u) du$ . Using the discussion of Section 3C, I will show that this integral can be interpreted by the theta lifting from U( $\zeta$ ) to U(1).

Recall that *N* is the unipotent radical of the parabolic subgroup *P* of  $U(I_{\zeta})$ . We take *N* as the Heisenberg group associated to the 4-dimensional symplectic space *W* over *F*. On *W*, the alternating pairing is defined as  $\langle x_1, x_2 \rangle = x_1 \zeta x_2^* - x_2 \zeta x_1^*$ ,

while, as a *K*-vector space,  $W = V_1 \otimes V_2$ , where  $V_1$  and  $V_2$  are Hermitian vector spaces with dimensions 2 and 1 over *K*, respectively. The skew-Hermitian form on  $V_1$  is  $(x, y)_1 = x\zeta y^*$ . The Hermitian form on  $V_2$  is defined as  $(x, y)_2 = x\overline{y}$ . So, we immediately have the reductive pair  $(U(V_1), U(V_2))$  in Sp(*W*) with  $U(V_1) \simeq U(\zeta)$ , and  $U(V_2) \simeq U(1)$ .

Consider the theta lifting

$$\theta_{\phi_2}^{\beta}(u) = \int_{U(\zeta)} \theta_{\phi_2}(g, u) \beta(g) \, dg$$

from U( $\zeta$ ) to U(1). In Section 3C, I give the definition of a theta lifting and explain the way to compute  $|\theta_{\phi_2}^{\beta}(1)|^2$  using the Siegel–Weil formula. If we assume that the representation  $\Pi$  on U( $\zeta$ ) is self-dual, then  $V_{\Pi} \simeq V_{\check{\Pi}}$ , and

$$\left|\int_{[U(\zeta)]} \beta(u)\overline{\theta}_{\phi_2}(u) \, du\right|^2 = \left|\theta_{\phi_2}^\beta(1)\right|^2$$

Proposition 3.7 and (18) together imply:

**Corollary 4.16.** 
$$|\theta_{\phi_2}^{\beta}(1)|^2 = \frac{1}{2} \int_{U(\zeta)(\mathbb{A})} f_{\delta_{\psi}(\phi_2 \otimes \bar{\phi}_2),s}(\mathfrak{e}(g,1)) \langle \Pi(g)\beta,\beta \rangle dg|_{s=-\frac{1}{2}}.$$

**Remark 4.17.** The integral in the corollary above should be nonzero if  $\phi_2$  is chosen properly, because the theta lifting of  $\Pi$  should define a nonzero representation space of U(1). This is a special case in [Li 1992]. The nonzero result is crucial for our application.

**Remark 4.18.** If  $\phi_{2,v}$  is a standard characteristic function, then  $\delta_{\psi}(\phi_2 \otimes \overline{\phi}_2)$  is also a standard characteristic function, and  $f_{\delta_{\psi}(\phi_2 \otimes \overline{\phi}_2), s, v}$  is normalized spherical.

**Proposition 4.19.** If  $\Pi_v$  is an unramified representation with a spherical vector  $\beta$ ,  $\mathfrak{f}_{s,v}$  is the unique U(2, 2)( $\mathbb{O}_{F_v}$ )-invariant section in  $I^{U(2,2)}_{\mathcal{P}'}(\chi, s)_v$  and  $\mathfrak{f}_{s,v}(1) = 1$ , then

$$\int_{U(\zeta)_v} \mathfrak{f}_{s,v}(\mathfrak{e}(u,1)) \langle \Pi(u)\beta,\beta \rangle_v \, du = \frac{L(\Pi_v,\chi_v,s+\frac{1}{2})}{L_v(\epsilon_{K/F},2s+2)L_v(1,2s+1),}$$

where  $L_v(1, \cdot)$  is the Zeta function of the local field  $F_v$ .

**Remark 4.20.** This type of integral was considered by Piatetski-Shapiro and Rallis in many cases. Similar calculations have been done in [Li 1992]. In the (U(1), U(1)) case, Yang [1997] had explicit formulas.

*Proof.* Let me calculate the integral above in the case when  $U(\zeta)_v \simeq GL_2(F_v)$ . The computations at other types of unramified places are skipped here. Now we omit the subindex v.

Denote the matrix coefficient of  $\Pi$  by  $w_{\Pi}$ . Let  $\chi|_{F \times F} = (\chi_1, \chi_2)$ . Using the Godement section of  $\mathfrak{f}_s$ , we have

$$\frac{1}{L(\epsilon_{K/F}, 2s+2)L(\epsilon_{K/F}, 2s+1)} \int_{U(\zeta)} f_{s}(\mathfrak{e}(u,1)) \langle \Pi(u)\beta, \beta \rangle du$$

$$= \int_{GL_{2}(F)} \chi_{1}(\det u) |\det u|^{s+1}$$

$$\cdot \int_{GL_{2}(F)} \Phi_{M_{2\times4}(\mathbb{O}_{F})}((0, Z)\mathfrak{e}(u, 1))\epsilon_{K/F}(\det Z) |\det Z|^{2s+2}w_{\Pi}(u) du dZ$$

$$= \int_{GL_{2}(F)\times GL_{2}(F)} \Phi_{M_{2\times4}(\mathbb{O}_{F})}(Z, Zu)\chi_{1}(\det u) |\det u|^{s+1}$$

$$\cdot \epsilon_{K/F}(\det Z) |\det Z|^{2s+2}w_{\Pi}(u) du dZ$$

$$Y = Zu \int_{GL_{2}(F)\times GL_{2}(F)} \Phi_{M_{2\times4}(\mathbb{O}_{F})}(Y, Z)\chi_{1}(\det Y) |\det Y|^{s+1}$$

$$\cdot \chi_{2}(\det Z) |\det Z|^{s+1}w_{\Pi}(Z^{-1}Y) dY dZ. \quad (29)$$

Let  $GL_2(F) = BK$ , where *B* consists of upper triangular matrices and  $K = GL_2(\mathbb{O}_F)$ . The matrix coefficient  $w_{\Pi}$  is a zonal spherical function and satisfies

$$\int_{K} w_{\Pi}(XkY) \, dk = w_{\Pi}(X)w_{\Pi}(Y) \quad \text{for } X, Y \in \mathrm{GL}_{2}(F).$$

Lemma 4.21. The expression in (29) equals the product of

$$\int_{\mathrm{GL}_2(F)} \Phi_{M_2(\mathbb{O}_F)}(Y) \chi_1(Y) |\det Y|^{s+1} w_{\Pi}(Y) \, dY$$

and

$$\int_{\operatorname{GL}_2(F)} \Phi_{M_2(\mathbb{O}_F)}(X) \chi_2(X) |\det X|^{s+1} w_{\Pi}(X) \, dX.$$

Proof. First, it is obvious that

$$(29) = \int_{\mathrm{GL}_2(F) \times \mathrm{GL}_2(F) \times K} \Phi_{M_{2\times 4}(\mathbb{O}_F)}(kY, Z) \chi_1(\det kY) |\det kY|^{s+1} \\ \cdot \chi_2(\det Z) |\det Z|^{s+1} w_{\Pi}(Z^{-1}Y) \, dY \, dZ \, dk.$$
(30)

Then, substitute kY by Y':

$$(30) = \int_{\mathrm{GL}_{2}(F)\times\mathrm{GL}_{2}(F)} \Phi_{M_{2\times4}(\mathbb{O}_{F})}(Y', Z)\chi_{1}(\det Y')|\det Y'|^{s+1}\chi_{2}(\det Z)|\det Z|^{s+1} \\ \cdot \int_{K} w_{\Pi}(Z^{-1}k^{-1}Y')\,dk\,dY'\,dZ$$
$$= \int_{\mathrm{GL}_{2}(F)} \Phi_{M_{2}(\mathbb{O}_{F})}(Y)\chi_{1}(Y)|\det Y|^{s+1}w_{\Pi}(Y)\,dY$$
$$\cdot \int_{\mathrm{GL}_{2}(F)} \Phi_{M_{2}(\mathbb{O}_{F})}(Z)\chi_{2}(Z)|\det Z|^{s+1}w_{\Pi}(Z^{-1})\,dZ.$$

So this proves Lemma 4.21.

Let

$$Z(\Phi_{M_2(\mathbb{O}_F)}, \chi_1 \otimes w_{\Pi}, s+1) = \int_{\mathrm{GL}_2(F)} \Phi_{M_2(\mathbb{O}_F)}(Y) \chi_1(Y) |\det Y|^{s+1} w_{\Pi}(Y) \, dY.$$

Zeta integrals such as this are discussed in [Godement and Jacquet 1972].

**Lemma 4.22.** *If*  $\Pi = \pi(\mu_1, \mu_2)$ *, then* 

$$Z(\Phi_{M_2(\mathbb{O}_F)}, \chi_1 \otimes w_{\Pi}, s+1) = L(\chi_1 \mu_1, s+\frac{1}{2})L(\chi_2 \mu_2, s+\frac{1}{2}).$$

*Proof.* Let Y = pk, for  $p = \begin{pmatrix} a_1 & b \\ a_2 \end{pmatrix} \in B$  and  $k \in K$ . Then  $dY = (1/|a_1|) db da_1 da_2$ .  $w_{\Pi}(Y) = w_{\Pi}(p) = \mu_1(a_1)\mu_2(a_2)(|a_1|/|a_2|)^{\frac{1}{2}}$ . So,

$$Z(\Phi_{M_2(\mathbb{O}_F)}, \chi_1 \otimes w_{\Pi}, s+1) = \int_{\mathbb{O}_F \times \mathbb{O}_F} \chi_1 \mu_1(a_1) |a_1|^{s+\frac{1}{2}} \chi_1 \mu_2(a_2) |a_2|^{s+\frac{1}{2}} da_1 da_2$$
$$= L(\chi_1 \mu_1, s+\frac{1}{2}) L(\chi_2 \mu_2, s+\frac{1}{2}).$$

By Lemmas 4.21 and 4.22, we have

$$\int_{U(\zeta)} \mathfrak{f}_{s}(\mathfrak{e}(u,1)) \langle \Pi(u)\beta,\beta \rangle \, du = \frac{L(\Pi,\chi,s+\frac{1}{2})}{L(\epsilon_{K/F},2s+2)L(\epsilon_{K/F},2s+1)}$$

And notice that in this case,  $\epsilon_{K/F}(\varpi) = 1$  for the prime element  $\varpi$  in *F*. This proves Proposition 4.19 when  $U(\zeta) \simeq GL_2(F)$ .

Next, let me explain the local integrals of  $|\theta_{\phi_2}^{\beta}(1)|^2$  at archimedean places and ramified finite places. At archimedean places,  $f_{\delta_{\psi}(\phi_2 \otimes \overline{\phi}_2),s,\infty}$  is  $U(\zeta)_{\infty} \times U(\zeta)_{\infty}$ -invariant, and  $\Pi_{\infty}$  is one-dimensional. Apparently,

$$\int_{U(\zeta)_{\infty}} f_{\delta_{\psi}(\phi_2 \otimes \bar{\phi}_2), s, \infty}(\mathfrak{e}(g, 1)) \langle \Pi(g)\beta, \beta \rangle_{\infty} \, dg = \operatorname{vol}(U(\zeta)_{\infty}) \neq 0.$$

As for finite places, first we are always allowed to take the standard characteristic function  $\phi_{2,v}$  when  $\Pi_v$  is unramified. This means if  $\Pi_v$  is unramified, we can guarantee  $f_{\delta_{\psi}(\phi_2 \otimes \overline{\phi}_2),s,v}$  is spherical.

If  $\Pi_v$  is ramified at a finite place v, let  $D_v \subset U(\zeta)_v$  be a compact open subgroup fixing  $\Pi_v$ . Especially, when  $\Pi_v = I(\mu_1, \mu_2)$  is a ramified principal series associated to the local characters  $\mu_1$  and  $\mu_2$ , we assume that  $\mu_1$  is ramified and  $\mu_2$  is unramified. Then  $D_v = \{g \in U(\zeta)_v \mid g = \begin{pmatrix} a & b \\ c & d \end{pmatrix}, a \in 1 + \mathfrak{m}, b, d \in \mathbb{O}_{K_v}, c \in \mathfrak{m}\}$ , where  $\mathfrak{m}$  is the conductor of  $\mu_1$ . We can choose  $\phi_{2,v}$  so that  $f_{\delta_{\psi}(\phi_2 \otimes \overline{\phi}_2), s, v}(\mathfrak{e}(g, 1)) = \Phi_{D_v}(g)$ , where  $\Phi_{D_v}$  is the characteristic function with the support in  $D_v$ . For instance, if vsplits,  $U(\zeta)_v \simeq \operatorname{GL}_2(F_v)$ . Then  $\mu_1$  and  $\mu_2$  are both characters of  $F_v$ . We can take  $\phi_2$  such that

$$\phi_2(x) = \begin{cases} 1 & \text{if } x \in \left(\begin{smallmatrix} 1 \\ 0 \end{smallmatrix}\right) + \mathfrak{m} \mathbb{O}_{F_v}^2, \\ 0 & \text{otherwise.} \end{cases}$$

In this way,

$$\int_{U(\zeta)_v} f_{\delta_{\psi}(\phi_2 \otimes \bar{\phi}_2), s, v}(\mathfrak{e}(g, 1)) \langle \Pi(g)\beta, \beta \rangle_v \, dg = \operatorname{vol}(D_v) \neq 0.$$

#### 5. Applications to Eisenstein series nonvanishing modulo p

Section 4 calculated the Fourier–Jacobi coefficients of two Eisenstein series. One is the holomorphic Siegel–Eisenstein series  $\mathbb{E}_k(\cdot, \eta)$  on U(3, 3). The other is the holomorphic Eisenstein series  $E_k(\cdot, \Pi, \eta)$  on U( $I_{\zeta}$ ) that is from the pullback of  $\mathbb{E}_k(\cdot, \eta)$ . From now on, we fix a prime p, so that every prime in F above p is unramified for both Eisenstein series. In this section, we will briefly discuss how to apply the computations of Section 4 to look for an Eisenstein series on U(3, 1) that does not vanish modulo p.

- **Remark 5.1.** (1) Recall that  $\mathbb{E}_k(\cdot, \eta)$  is defined from the section  $\mathbb{F}_k$  that is spherical outside *S* (for the definition of *S*, one may refer to Theorem 4.9). In the application, *S* is usually taken to be the set of places where the given data (for instance the number fields *F* and *K*, the characters  $\eta$  and  $\chi$ , the representation  $\Pi$  and so on) are ramified. Notice that we can assume that all the data at *p* are unramified. So  $p \notin S$ . In other words,  $\mathbb{F}_k$  is spherical at *p*.
- (2)  $\mathbb{E}_k(\cdot, \eta)$  can be normalized to become a *p*-integral Eisenstein series  $\mathbb{E}^{\text{int}}$ . About this claim, one can refer to (3.3.5.3) and Remark (3.3.5.5) in [Harris et al. 2006]. The normalizing factor is given in (3.3.5.1). In fact, the situation here is simpler than [ibid.], because  $\mathbb{E}_k(\cdot, \eta)$  is unramified at *p*.

(3) The pullback of  $\mathbb{E}^{int}$  denoted by  $\mathbb{E}_{pb}^{int}$  is an Eisenstein series on  $U(I_{\zeta})$ . It only differs by a constant with  $E(\cdot, \Pi, \eta)$ . Define

$$E^{\text{int}} = \frac{\mathbb{E}_{\text{pb}}^{\text{int}}}{\Omega_K^k},\tag{31}$$

where  $\Omega_K$  is the CM-period of K and it is well-defined up to  $\overline{\mathbb{Z}}_p^{\times}$ .

**Lemma 5.2.** E<sup>int</sup> is a p-integral holomorphic Eisenstein series.

About  $\Omega_K$  and this lemma, one can refer to [Hsieh 2011b, Section 7.2].

As mentioned in the introduction, one of the motivations of this paper is to provide a possible way to argue nonvanishing modulo p of the Eisenstein series on the unitary group U(3, 1) used in the Iwasawa theory through the calculation of its Fourier–Jacobi coefficients. For the discussion on this topic, let me assume that  $F = \mathbb{Q}$  and that the imaginary quadratic extension  $K/\mathbb{Q}$  splits at p.

Following the idea of Skinner and Urban [Skinner and Urban 2012; Urban 2006] to show one divisibility of the main conjecture for  $GL_2 \times K^{\times}$  by the method of Eisenstein congruence on U(3, 1), a Hida family of holomorphic Eisenstein series  $\mathcal{E}^{\text{ord}}$  on U(3, 1) is constructed so that its constant terms at all cusps are divisible by the *p*-adic *L*-function of  $GL_2 \times K^{\times}$ . Suppose  $\mathcal{E}^{\text{ord}}$  is defined over a two-variable Iwasawa algebra  $\Lambda$  (refer to Conjecture 1.1), and denote by  $\mathfrak{m}_{\Lambda}$  the maximal ideal of  $\Lambda$ . It is required that  $\mathcal{E}^{\text{ord}} \neq 0 \pmod{\mathfrak{m}_{\Lambda}}$ . Since  $\mathcal{E}^{\text{ord}}$  is obtained by interpolating a *p*-ordinary holomorphic Eisenstein series  $E^{\text{ord}}$ , we have:

**Lemma 5.3.** If  $E^{\text{ord}} \neq 0 \pmod{\mathfrak{m}_p}$ , then  $\mathscr{C}^{\text{ord}} \neq 0 \pmod{\mathfrak{m}_\Lambda}$ , where  $\mathfrak{m}_p$  is the maximal ideal of  $\overline{\mathbb{Z}}_{(p)}$  induced by  $i : \mathbb{C} \to \mathbb{C}_p$ .

So it is enough to show  $E^{\text{ord}}$  does not vanish modulo  $\mathfrak{m}_p$ . For the strict definition and construction of  $E^{\text{ord}}$ , see [Hsieh 2011b]. Although  $E^{\text{ord}}$  and  $E^{\text{int}}$  are both *p*-integral holomorphic Eisenstein series, they are not the same, because we assume  $E^{\text{int}}$  is unramified at *p*, but  $E^{\text{ord}}$  is ramified at *p*. So in order to apply the computation of Fourier–Jacobi coefficients of  $E^{\text{int}}$  in Section 4, two points will be addressed in this section. The first is to relate  $E^{\text{ord}}$  used in the proof of the main conjecture to  $E^{\text{int}}$ . The second is to give a strategy of showing nonvanishing modulo *p* of  $E^{\text{int}}$ .

At p,  $U(\zeta)(\mathbb{Q}_p) \simeq GL_2(\mathbb{Q}_p)$ . In [Hsieh 2011b], the representation  $\Pi$  of  $U(\zeta)$  is chosen so that the local representation  $\Pi_p$  of  $GL_2(\mathbb{Q}_p)$  is ordinary; then  $\Pi_p$  must be of the type  $\pi(\mu_1, \mu_2)$ . Assume that the characters  $\mu_i$  of  $\mathbb{Q}_p$  are unramified. Then an Eisenstein series  $E_0$  is defined with the data of  $\Pi$  and  $\eta$ , so that  $E^{\text{ord}}$  is exactly the ordinary projection of  $E_0$  ( $E^{\text{ord}} = eE_0$ , where e is the map of ordinary projection). The only difference between  $E_0$  considered in [ibid.] and  $E^{\text{int}}$  in (31) is that local sections at p are different. A special section at p is taken to make sure the corresponding  $E_0$  leads to the ordinary *p*-adic Eisenstein series with the optimal constant terms at cusps. But, for  $E^{\text{int}}$ , the local section at *p* is spherical.

Define normalized actions of  $U_p$  and  $T_p$  operators ( $|| U_p$  and  $|| T_p$ ) to a modular form on U( $I_{\zeta}$ ); refer to [Hsieh 2011a; 2011b].

**Lemma 5.4.** Let *E* be a *p*-adic modular form of weight (0, k), k > 2. Suppose that *E* is unramified at *p*. Then we have

 $E \parallel U_p(\alpha_i) \equiv E \parallel T_p(\alpha_i) \pmod{p},$ 

for 
$$\alpha_i = \begin{pmatrix} I_{6-i} & 0\\ 0 & pI_i \end{pmatrix} \in \operatorname{GL}_6(\mathbb{Q}_p) \simeq U(I_{\zeta})(\mathbb{Q}_p), \ i = 1, 2, \dots, 6.$$

The analogous result for modular forms on U(2, 1) was due to Hida's observation. Hsieh summarized it in [2011a, Lemma 7.3], whose proof is essentially the same as that of Lemma 5.4. I want to emphasize that the formulas of the normalized Hecke operators at p used in the proof are especially for holomorphic forms. So, we can safely apply this lemma to  $E^{\text{int}}$  and  $E^{\text{ord}}$ .

## **Lemma 5.5.** $E^{\text{ord}} = eE_0 = C \cdot eE^{\text{int}}$ , where C is a p-adic unit.

Roughly speaking, the proof is as follows. Suppose the local sections at p for the Eisenstein series  $E^{\text{ord}}$  and  $E^{\text{int}}$  are, respectively,  $f_p^{\text{ord}}$  and  $f_p^{\text{int}}$ . There is a unique normalized ordinary local section at p, denoted by  $f_p^{\text{ord},N}$  (for the uniqueness, see [Hida 2004b; Hsieh 2011b, Remark 6.3]). Then

$$f_p^{\text{ord}} = C_1 f_p^{\text{ord},N}$$
 and  $e f_p^{\text{int}} = C_2 f_p^{\text{ord},N}$ .

Moreover it can be shown that  $C = C_1 C_2^{-1}$  is a *p*-adic unit. Combining the two lemmas above, we get

$$E^{\operatorname{ord}} \not\equiv 0 \pmod{\mathfrak{m}_p} \iff E^{\operatorname{int}} \not\equiv 0 \pmod{\mathfrak{m}_p}.$$

We are left to show that  $E^{\text{int}} \not\equiv 0 \pmod{\mathfrak{m}_p}$ .

From the calculation of Section 4, we can construct a linear functional on the space of holomorphic modular forms on U(3, 1), such that

$$l_{\theta}(E^{\text{int}}) = \frac{\langle E_{\psi}^{\text{int}}, \theta \rangle}{\langle \theta, \theta \rangle} \cdot \theta^{\beta}(1).$$

And notice that  $E_{\psi}^{\text{int}}$  and  $\theta$  are both in the space of holomorphic theta functions, and  $\langle \cdot, \cdot \rangle$  is defined to be the inner product in this space. When  $\theta$  is  $\theta_{\varphi}$ , we have  $l_{\theta_{\varphi}}(E^{\text{int}}) \sim L^{\text{alg}}(\eta \chi^{-1}, l - \frac{1}{2})L^{\text{alg}}(\eta \chi^{-1}, l - \frac{3}{2})L^{\text{alg}}(\Pi, \chi, 1)$ . Notice that  $\chi$ is from the Weil representation and we can vary  $\chi$  with only the restrictions that  $\chi_{\infty}(z) = |z|/z$  and  $\chi|_{\mathbb{A}^{\times}} = \epsilon_{K/\mathbb{Q}}$ .  $L^{\text{alg}}(\eta \chi^{-1}, l - \frac{1}{2})$  and  $L^{\text{alg}}(\eta \chi^{-1}, l - \frac{3}{2})$  are both *p*-units for almost all  $\chi$ . These facts are due to Hida [2004a]. The question of whether  $L^{\text{alg}}(\Pi, \chi, 1)$  is a *p*-unit or not for infinitely many  $\chi$  remains an open problem. Many results are known about nonvanishing modulo p of special values of *L*-functions (refer to [Hida 2004a; Vatsal 2003] for results on GL<sub>1</sub> and GL<sub>2</sub> *L*-functions). These facts suggest  $L^{alg}(\Pi, \chi, 1)$  may share the same property. Recent work of M. Brakočević on anticyclotomic *p*-adic *L*-functions of central critical Rankin–Selberg *L*-value may help prove the conjecture. I will address this question in my next paper.

From Appendix B, we see by choosing proper  $\varphi$  that  $\theta_{\varphi}$  is a *p*-integral theta function, and can serve as one element of the basis that spans the space of *p*-integral holomorphic theta functions  $T(m, L, U_f)$ . So  $l_{\theta_{\varphi}}$  maps a *p*-integral modular form in U(3, 1) to a *p*-adic integer.

From the discussion above, we have this:

### **Conjecture 5.6.** $E^{\text{int}} \not\equiv 0 \pmod{\mathfrak{m}_{\Lambda}}$ .

**Remark 5.7.** As mentioned in the introduction, the computations explained in this paper can be generalized to an arbitrary nonquasisplit unitary group U(m, n) for m > n. Let P = MN be the minimal parabolic subgroup of U(m, n), and  $M = \operatorname{GL}_n(K) \times U(m-n)$ . Given  $\Phi \in I_P^{U(m,n)}(\Pi \otimes \eta(\det \cdot), s)$ , define the Eisenstein series  $E(\Phi)$  and consider the Fourier–Jacobi coefficients  $E_{\psi}(\Phi)$  as a theta function on the Jacobi group  $U(m-n) \cdot N$ . Because  $E(\Phi)$  can be written as the pullback of a Siegel–Eisenstein series  $\mathbb{E}(\eta)$  on U(m, m), we can study  $E_{\psi}(\Phi)$  using certain Fourier–Jacobi coefficients of  $\mathbb{E}(\eta)$ . So  $E_{\psi}(\Phi)$  will have similar expressions as in Corollary 4.11. Again using the inner product of the space of theta functions on  $U(m-n) \cdot N$ , we can define a linear functional on  $E(\Phi)$  that leads to special *L*-values.

#### Appendix A: Proof of the Siegel–Weil formula for (U(2, 2), U(1))

The Siegel–Weil formula for the dual reductive pair (U(2, 2), U(1)) is formulated in Theorem 3.6. Let us first fix necessary notation and then give the proof.

Consider the dual reductive pair (U(2, 2), U(V)), where *V* is a Hermitian vector space of dimension 1 over *K*. Now U(2, 2) × *U*(*V*) acts on *S*(*V*<sup>2</sup>(A)) via the Weil representation  $\omega_{\chi}$  determined by an additive character  $\psi$  of A and a character  $\chi$  of  $\mathbb{A}_{K}^{\times}/K^{\times}$  such that  $\chi|_{\mathbb{A}^{\times}} = \epsilon_{K/F}$ . For  $\phi \in S(V^{2}(\mathbb{A}))$ , we have  $f_{\phi,s} \in I_{\mathcal{P}'}^{U(2,2)}(\chi, s)$ , where  $\mathcal{P}'$  is the Siegel parabolic subgroup of U(2, 2) (refer to (17)). Theorem 3.6 states that the Eisenstein series  $E(g, f_{\phi,s})$  on U(2, 2) is holomorphic at  $s = -\frac{1}{2}$  and

$$E(g, f_{\phi,s})|_{s=-\frac{1}{2}} = 2 \int_{[U(V)]} \theta_{\phi}(g, u) \, du$$

I give the proof of this theorem following the idea of Kudla and Rallis [1988], who proved the Siegel–Weil formula for  $(Sp_n, O(V))$  when V is anisotropic. First, we prove that  $E(g, f_{\phi,s})$  is holomorphic at  $s = -\frac{1}{2}$  by studying the analytic properties

#### Bei Zhang

of the constant term. Denote  $\int_{[U(V)]} \theta_{\phi}(g, u) du$  by  $I(g, \phi)$ . Notice that  $I(g, \phi)$  is orthogonal to all cusp forms on U(2, 2)(A) and is moreover concentrated on the Borel subgroup *B* in the sense that the constant term of  $I(g, \phi)$  along any parabolic subgroup *P* strictly containing *B* is orthogonal to all cusp forms on the Levi factor of *P*. Since the same is true for  $E(g, f_{\phi,s})$ , it suffices to show that the constant term of  $E(g, f_{\phi,s})$  along the Siegel parabolic subgroup at  $s = -\frac{1}{2}$  is equal to that of  $I(g, \phi)$ .

Denote the constant term of  $E(g, f_{\phi,s})$  with respect to  $\mathcal{P}'$  by  $E_{\mathcal{P}'}(g, f_{\phi,s})$ . Then we have this:

**Lemma A.1.** For any  $\mathcal{K}'$ -finite section f,  $E(g, f_s)$  and  $E_{\mathcal{P}'}(g, f_s)$  have the same set of poles.

If  $\phi$  is  $\mathscr{K}'$ -finite, so is  $f_{\phi}$ . We can apply the lemma above to find the poles of  $E(g, f_{\phi,s})$ . There are three terms in  $E_{\mathscr{P}'}(g, f_{\phi,s})$ , that is

$$E_{\mathcal{P}'}(g, f_{\phi,s}) = E^0_{\mathcal{P}'}(g, f_{\phi,s}) + E^1_{\mathcal{P}'}(g, f_{\phi,s}) + E^2_{\mathcal{P}'}(g, f_{\phi,s}),$$
(32)

corresponding to the Bruhat decomposition

$$\mathbf{U}(2,2)(F) = \mathcal{P}' \sqcup \mathcal{P}' w_{13} \mathcal{P}' \sqcup \mathcal{P}' w_2 \mathcal{P}',$$

where

$$w_{13} = \begin{pmatrix} 1 & 1 \\ -1 & 1 \end{pmatrix}$$
 and  $w_2 = \begin{pmatrix} I_2 \\ -I_2 \end{pmatrix}$ .

We have  $E^0_{\mathcal{P}'}(g, f_{\phi,s}) = f_{\phi,s}(g)$ . So  $E^0_{\mathcal{P}'}(g, f_{\phi,s})|_{s=-\frac{1}{2}} = \omega_{\chi}(g)\phi(0)$ . The third term has

$$E_{\mathcal{P}'}^2(g, f_{\phi,s}) = \int_{\mathcal{N}'(\mathbb{A})} f_{\phi,s}(w_2 ng) \, dn = M(s) \, f_{\phi,s}(g),$$

where M(s) is the intertwining operator. The second term has

$$E_{\mathcal{P}'}^1(g, f_{\phi,s}) = \sum_{\gamma \in B_1 \setminus \operatorname{GL}_2(K)} f_{\phi,s}^1(\gamma g),$$
(33)

where  $B_1$  is the Borel subgroup of  $GL_2(K)$  and

$$f^{1}_{\phi,s}(\gamma g) = \int_{\mathcal{N}''(\mathbb{A})} f_{\phi,s}(w_{13}ng) \, dn,$$

where  $\mathcal{N}''$  is a subgroup of  $\mathcal{N}'$  such that for  $n \in \mathcal{N}''$ ,

$$n = \begin{pmatrix} 1 & x \\ & 1 & \\ & & 1 \\ & & 1 \end{pmatrix}.$$

Since the Levi part of  $\mathcal{P}'(F)$  is isomorphic to  $\operatorname{GL}_2(K)$ , we see  $E^1_{\mathcal{P}'}(g, f_{\phi,s})|_{\operatorname{GL}_2(\mathbb{A}_K)}$  is an Eisenstein series on  $\operatorname{GL}_2(\mathbb{A}_K)$  associated to the section

$$f_{\phi,s}^{1} \in I_{B^{1}}^{\mathrm{GL}_{2}}(\chi | \cdot |_{K}^{-s-\frac{1}{2}}, \chi | \cdot |_{K}^{s+\frac{3}{2}}),$$

that is,

$$f_{\phi,s}^{1}\left(\begin{pmatrix}a & b\\ & d\end{pmatrix}g\right) = \chi(a)|a|_{K}^{-s-\frac{1}{2}}\chi(b)|b|_{K}^{s+\frac{3}{2}}\left|\frac{a}{b}\right|_{K}^{\frac{1}{2}}f_{\phi,s}^{1}(g).$$
(34)

Let us consider the analytic property of  $M(s) f_{\phi,s}(g)$ . It is well known that M(s) is well-defined for *s* if Re *s* is big enough and it has the meromorphic continuation to the complex plane. Let *S* be a finite set of places of *F* such that  $f_{\phi,s,v} = f_{0,s,v}$  is spherical for all  $v \notin S$ . Then we have

$$M(s)f_{\phi,s} = \frac{a(s)}{b(s)} \prod_{v \in S} \frac{b(s)_v}{a(s)_v} M(s)_v f_{\phi,s,v} \prod_{v \notin S} f_{0,-s,v},$$

where

$$\frac{a(s)}{b(s)} = \frac{\xi(2s-1,1)\xi(2s,\epsilon_{K/F})}{\xi(2s+2,1)\xi(2s+1,\epsilon_{K/F})};$$

specifically,

$$\frac{a(s)_v}{b(s)_v} = \frac{L_v(2s-1,1)L_v(2s,\epsilon_{K/F})}{L_v(2s+2,1)L_v(2s+1,\epsilon_{K/F})}$$

for  $v \neq \infty$ , and

$$\frac{a(s)_{\infty}}{b(s)_{\infty}} = c(s) \frac{\Gamma(2s-1)\Gamma(2s+1)}{\Gamma(2s+2)\Gamma(2s+2)},$$
(35)

where c(s) is some exponential, which will not affect the analytic property of  $M(s) f_{\phi,s}$ .

**Theorem A.2.**  $M(s) f_{\phi,s}|_{s=-\frac{1}{2}} = 0$ 

Proof. Notice that

$$\left.\frac{a(s)}{b(s)}\right|_{s=-\frac{1}{2}}=0.$$

So we only need to show that

$$\frac{b(s)_v}{a(s)_v}M(s)_v f_{\phi,s,v}$$

for  $v \in S$  at  $s = -\frac{1}{2}$  is holomorphic.

When  $v \in S$  and v is a finite place, we always have this:

**Lemma A.3.** 
$$\frac{1}{a(s)_v} M(s)_v f_{s,v}$$
 is holomorphic at  $s = -\frac{1}{2}$  for any  $f_s$ .

Bei Zhang

The general form of the lemma is stated in [Kudla and Rallis 1988]. If  $v \in S$  is an inert place,  $b(s)_v$  is obviously holomorphic at  $s = -\frac{1}{2}$ . So

$$\frac{b(s)_v}{a(s)_v}M(s)_v f_{\phi,s,\iota}$$

is holomorphic at  $s = -\frac{1}{2}$ .

If  $v \in S$  is a splitting place, pay attention that now  $b(s)_v$  has a simple pole. We need a refinement of Lemma A.3 as follows:

Lemma A.4. If v is a splitting place,

$$\frac{1}{a(s)_v}M(s)_v f_{\phi,s,v}$$

vanishes at  $s = -\frac{1}{2}$ .

I have to emphasize that this lemma is only right for the Siegel–Weil section. Then in this case, we still have

$$\frac{b(s)_v}{a(s)_v}M(s)_v f_{\phi,s,v}$$

is holomorphic at  $s = -\frac{1}{2}$ . I omit the proof of Lemma A.4. It is a direct corollary of [Kudla and Sweet 1997, Theorem 1.3].

For  $v = \infty$ , for convenience, we take  $\phi_{\infty}^0(x) = \psi(ixx^*)$  as it is what we use in the paper. For general functions in  $S(V^2)_{\infty}$  in the space of the Fock representation, the holomorphic result still holds. We have that  $\phi_{\infty}^0$  is an eigenfunction under the action of an element *k* in the maximal open compact subgroup of U(2, 2)<sub> $\infty$ </sub>. Then,

**Lemma A.5.** 
$$M(s)_{\infty} f_{\phi_{\infty}^0,s} = c \frac{\Gamma(2s)\Gamma(2s-1)}{\Gamma(s+\frac{3}{2})\Gamma(s+\frac{1}{2})^2\Gamma(s-\frac{1}{2})} f_{\phi_{\infty}^0,-s}.$$

From (35) and Lemma A.5, we can see that

$$\frac{b(s)_{\infty}}{a(s)_{\infty}}M(s)_{\infty}f_{\phi_{\infty}^0,s}$$

is holomorphic at  $s = -\frac{1}{2}$ . Combining the discussions above at each place, we prove Theorem A.2.

Now let us consider (33), which is the second term of the constant term of  $E(g, f_{\phi,s})$ . As we see, restricted to the Levi part of U(2, 2), it is an Eisenstein series associated to the (34). In [Kudla and Rallis 1988, Proposition 6.4], the Eisenstein series like this was discussed and the holomorphic property was confirmed. The idea is to obtain a relation of the form

$$E_{\mathcal{P}'}^{1}(g, f_{\phi,s}) = \sum_{j} \alpha_{j}(s) E^{1}(g, F_{j}(s)),$$
(36)

where  $\alpha_j(s)$  is a meromorphic function holomorphic at  $s = -\frac{1}{2}$ , and  $E^1(g, F_j(s))$  is an Eisenstein series on  $GL_2(\mathbb{A}_K)$  and

$$F_j(g,s) = \mu_1(\det g) |\det g|_K^{\frac{1}{2}} \int_{\mathbb{A}_K^{\times}} \phi_j((0,x)g) \mu_1 \mu_2^{-1}(\det g) |\det g|_K \, dx$$

for  $\phi_j \in S(M_{12}(\mathbb{A}_K))$ , and  $\mu_1 = \chi |\cdot|_K^{-s-\frac{1}{2}}$ ,  $\mu_2 = \chi |\cdot|_K^{s+\frac{3}{2}}$ . The analytic property of this type of Eisenstein series is explicitly worked out. In our case, each  $E^1(\cdot, F_j(s))$  is holomorphic at  $s = -\frac{1}{2}$ . Furthermore, because the sum in (36) is a finite sum,  $E_{\mathcal{P}'}^1(g, f_{\phi,s})$  is also holomorphic at  $s = -\frac{1}{2}$ .

**Theorem A.6.**  $E(g, f_{\phi,s})$  is holomorphic at  $s = -\frac{1}{2}$ .

It is not hard to verify that  $I(g, \phi)$  as an automorphic form on U(2, 2) is orthogonal to all cusp forms on U(2, 2) and is concentrated on the Borel subgroup *B*. Since the constant term of  $I(g, \phi)$  with respect to the Siegel parabolic subgroup  $\mathcal{P}'$  is  $\omega_{\chi}(g)\phi(0)$ , the only thing remaining to prove the Siegel–Weil formula is to confirm the constant term of  $E(g, f_{\phi,s})$  at  $s = -\frac{1}{2}$  is  $2\omega_{\chi}(g)\phi(0)$ .

Let us first calculate the constant term  $E_{\mathcal{P}'_1}(g, f_{\phi,s})$  of  $E(g, f_{\phi,s})$  with respect to the parabolic subgroup  $\mathcal{P}'_1$  whose Levi factor is isomorphic to  $K^{\times} \times U(1, 1)$ . Let

$$i: U(1, 1) \to U(2, 2), \quad \left( \begin{pmatrix} a & b \\ c & d \end{pmatrix} \right) \mapsto \begin{pmatrix} 1 & a & b \\ c & 1 & d \end{pmatrix}.$$

Then:

**Proposition A.7.** For  $g \in U(1, 1)$ ,

$$E_{\mathcal{P}'_{1}}(g, f_{\phi,s}) = E^{0}_{\mathcal{P}'_{1}}(g, s + \frac{1}{2}, i^{*}f_{\phi,s}) + E^{1}_{\mathcal{P}'_{1}}(g, \frac{1}{2} - s, i^{*}M(s)f_{\phi,s}),$$
(37)

where  $E^0_{\mathcal{P}'_1}$  and  $E^1_{\mathcal{P}'_1}$  are both Eisenstein series on U(1, 1).

Notice that  $E^0_{\mathcal{P}'_1}$  is an Eisenstein series associated to the Siegel–Weil section  $i^* f_{\phi,s}$ . Applying the result about the Siegel–Weil formula for the dual reductive pair (U(1, 1), U(1)) [Ichino 2004], we see that the constant term of

$$E^0_{\mathcal{P}'_1}(g,s+\frac{1}{2},i^*f_{\phi,s})$$

with respect to the Siegel parabolic subgroup of U(1, 1) has two terms and they are both equal to  $\omega_{\chi}(i(g))\phi(0)$  for  $g \in U(1, 1)$  when  $s = -\frac{1}{2}$ .

First,  $E_{\mathscr{P}'_1}^1(g, \frac{1}{2} - s, i^*M(s)f_{\phi,s})$  is holomorphic at  $s = -\frac{1}{2}$  by arguments similar to those in [Kudla and Rallis 1988, Section 8]. Then applying Theorem A.2, we have this Eisenstein series is zero at  $s = -\frac{1}{2}$ .

It seems that too many constant terms are involved. The relation can be described by the following diamond graph used in [ibid.]:



From the line

$$U(2, 2) \to GL_1 \times U(1, 1) \to GL_1 \times GL_1,$$

we get  $E_{\mathscr{P}'_1}^{00}$ ,  $E_{\mathscr{P}'_1}^{01}$ ,  $E_{\mathscr{P}'_1}^{10}$ , and  $E_{\mathscr{P}'_1}^{11}$ . They are constant terms of  $E_{\mathscr{P}'_1}^0$  and  $E_{\mathscr{P}'_1}^1$ . From the line

$$U(2, 2) \rightarrow GL_2 \rightarrow GL_1 \times GL_1$$

we get  $E^0_{\mathcal{P}'}$ ,  $E^{10}_{\mathcal{P}'}$ ,  $E^{11}_{\mathcal{P}'}$ , and  $E^2_{\mathcal{P}'}$  (refer to (32)), where  $E^{10}_{\mathcal{P}'}$  and  $E^{11}_{\mathcal{P}'}$  are the constant terms of  $E_{\mathcal{P}'}^1$ . Restricting all these to  $GL_1 \times GL_1$ , we have the following match up relation:

$$\begin{array}{ccccc} E^{00}_{\mathscr{P}'_1} & E^{01}_{\mathscr{P}'_1} & E^{10}_{\mathscr{P}'_1} & E^{11}_{\mathscr{P}'_1} \\ E^{0}_{\mathscr{P}'} & E^{10}_{\mathscr{P}'} & E^{11}_{\mathscr{P}'} & E^{2}_{\mathscr{P}'} \end{array}$$

The top terms match up with the bottom terms. Because  $E_{\mathcal{P}'_1}^1$  is zero at  $s = -\frac{1}{2}$ , then we have that  $E_{\mathcal{P}'}^{11}$  is zero at  $s = -\frac{1}{2}$ .

#### **Proposition A.8.**

 $E^{1}_{\mathcal{P}'}(g, f_{\phi,s})|_{s=-\frac{1}{2}} = \omega_{\chi}(g)\phi(0).$ 

Thus, we obtain that  $E_{\mathcal{P}'}(g, f_{\phi,s})|_{s=-\frac{1}{2}} = 2\omega_{\chi}(g)\phi(0)$ . Then Theorem 3.6 is proved.

#### **Appendix B: Integral theta functions**

First recall classical definitions of theta functions. For applications, let us restrict to the 2-dimensional case here. Most of the results can be generalized to arbitrary dimensions without difficulty. Fix the embeddings

$$\overline{\mathbb{Q}} \xrightarrow{i_{\infty}} \mathbb{C}$$
 and  $\overline{\mathbb{Q}} \xrightarrow{i_p} \mathbb{C}_p$ .

Let V be a 2-dimensional Hermitian vector space over K. Choose an  $\mathbb{O}_K$ -lattice L so that it fixes an abelian variety  $\mathcal{A}_L$  with complex multiplication defined over a number field M, where  $K \subset M \subset \overline{\mathbb{Q}}$ . Under the embedding  $i_{\infty}$ , L can be regarded as a  $\mathbb{Z}$ -module of rank 4, and so a lattice of  $\mathbb{C}^2$ . Then there exists an analytic parametrization over  $\mathbb C$  such that

$$\mathcal{A}_L \otimes_{i_\infty} \mathbb{C} \simeq \mathbb{C}^2 / L.$$

For a Riemann form H on  $\mathbb{C}^2/L$ , and a map  $\epsilon : L \to U$ , where U is the unit circle of  $\mathbb{C}$ , define an analytic line bundle  $\mathfrak{L}_{H,\epsilon}^{an} \simeq \mathbb{C} \times \mathbb{C}^2/L$  with the action of L given by

$$l \cdot (w, x) = \left(w + l, \epsilon(l)e\left(\frac{1}{2i}H\left(l, w + \frac{l}{2}\right)\right)x\right) \quad \text{for } l \in L, (w, x) \in \mathbb{C}^2 \times \mathbb{C},$$

where  $e(x) = e^{2\pi i x}$ . Then the space of global sections  $\Gamma(\mathbb{C}^2/L, \mathfrak{L}_{H,\epsilon}^{an})$  can be identified with the space of holomorphic theta functions  $T(H, \epsilon, L)$  such that for  $f \in T(H, \epsilon, L)$ ,

$$f(w+l) = f(w)\epsilon(l)e\left(\frac{1}{2i}H\left(l,w+\frac{l}{2}\right)\right) \text{ for } w \in \mathbb{C}^2, l \in L.$$

To study arithmetic theta functions and furthermore integral theta functions inside  $T(H, \epsilon, L)$ , let us consider the line bundle  $\mathfrak{L}_{H,\epsilon}$  on  $\mathcal{A}_L$  defined over M, and give  $\mathfrak{L}_{H,\epsilon}$  a rigidification at the origin. At  $\infty$ , fix an isomorphism

$$\mathfrak{L}_{H,\epsilon}\otimes_{i_{\infty}}\mathbb{C}\simeq\mathfrak{L}_{H,\epsilon}^{\mathrm{an}},$$

such that it is consistent with the analytic parametrization of  $\mathcal{A}_L$ , and carries the rigidification of  $\mathfrak{L}_{H,\epsilon}$  into the canonical one of  $\mathfrak{L}_{H,\epsilon}^{an}$ . For a prime p, when  $\mathcal{A}_L \otimes_{i_p} \mathbb{C}_p$  has a good reduction, we can assume it is defined over the ring of integers  $\mathbb{O} = \mathbb{O}(\mathbb{C}_p)$ . Then we require that the rigidification of  $\mathfrak{L}_{H,\epsilon}$  satisfies that the p-integral elements of the stalk of  $\mathfrak{L}_{H,\epsilon}$  over the origin correspond to the p-integral points on the affine line.

Within the context above, we see that  $i_{\infty}(\Gamma(\mathcal{A}_{L}, \mathfrak{L}_{H,\epsilon}))$  is an  $i_{\infty}(M)$ -vector space of theta functions inside  $\Gamma(\mathbb{C}^{2}/L, \mathfrak{L}_{H,\epsilon}^{an}) = T(H, \epsilon, L)$ . Let us denote this space by  $T^{ar}(H, \epsilon, L)$ ; each element inside is an arithmetic theta function. Inside the  $\mathbb{C}_{p}$ -vector space  $\Gamma(\mathcal{A}_{L} \otimes_{i_{p}} \mathbb{C}_{p}, \mathfrak{L}_{H,\epsilon} \otimes_{i_{p}} \mathbb{C}_{p})$ , we have the  $\mathbb{O}$ -module of integral sections. It is in turn the  $i_{\infty}(i_{p}^{-1}(\mathbb{O}) \cap M)$ -module of *p*-integral theta functions in  $i_{\infty}(\Gamma(\mathcal{A}_{L}, \mathfrak{L}_{H,\epsilon})) = T^{ar}(H, \epsilon, L)$ . We give it a notation  $T^{int}(H, \epsilon, L)$ . Thus we have

$$T^{\text{int}}(H,\epsilon,L) \subset T^{\text{ar}}(H,\epsilon,L) \subset T(H,\epsilon,L).$$

For  $f \in T(H, \epsilon, L)$ , define

$$f_*(w) = e\left(\frac{i}{4}H(w,w)\right)f(w).$$

The following lemma gives simple characterizations of arithmetic theta functions and integral theta functions.

**Lemma B.1.**  $T^{\text{ar}}(H, \epsilon, L)$  consists of all functions  $f \in T(H, \epsilon, L)$  such that  $f_*(w)$  is an algebraic number in  $\mathbb{C}$  for any  $w \in KL$ . The module  $T^{\text{int}}(H, \epsilon, L)$  of *p*-integral theta functions consists of  $f \in T(H, \epsilon, L)$  such that  $i_p \cdot i_{\infty}^{-1}(f_*(w))$  is integral in  $\mathbb{C}_p$  for  $w \in KL$ .

For the proof, see the proof [Finis 2006, Lemma 3.1]. Even though, there, only the case of one-dimensional Hermitian space was considered, the idea can be easily generalized here.

Shimura [1976] studied the arithmetic properties of theta functions with complex multiplication, and gave a basis for  $T^{ar}(H, \epsilon, L)$ . Let me mainly discuss the case when V/K is a 2-dimensional anisotropic Hermitian space. Also, V can be regarded as a symplectic vector space with the alternating form

$$\langle v_1, v_2 \rangle = v_1 \zeta v_2^* - v_2 \zeta v_1^*,$$

with  $v_1 \zeta v_2^*$  the anti-Hermitian pairing on *V*. There exist  $w_1, w_2 \in M_2(K)$ , such that  $z = w_2^{-1} w_1$  is a point in the Siegel upper half plane  $\mathfrak{H}_2$ , and

$$L = \left\{ (a \ b) \begin{pmatrix} w_1 \\ w_2 \end{pmatrix} \middle| a \in \mathbb{Z}^2, b \in \mathbb{Z}^2 \begin{pmatrix} 1 & 0 \\ 0 & \alpha \end{pmatrix} \text{ for a fixed } \alpha \in \mathbb{Z} \right\}.$$

Then the Riemann form on V can be written as

$$H(uw_2, vw_2) = 2im \cdot u(z - \bar{z})^{-1}v^* \text{ for } u, v \in K^2,$$
(38)

and  $\epsilon$  is of the form

$$\epsilon(aw_1 + bw_2) = e\left(\frac{m}{2} \cdot ab^t + br^t + as^t\right) \quad \text{for } a \in \mathbb{Z}^2, \ b \in \mathbb{Z}^2 \begin{pmatrix} 1 & 0 \\ 0 & \alpha \end{pmatrix},$$

with a choice of r and s in  $\mathbb{Q}^2$ . For  $u \in \mathbb{C}^2$ ,  $z \in \mathfrak{H}_2$ ,  $r, s \in \mathbb{R}^2$ , and a positive integer  $\mu$ , define

$$\theta_{r,s}(u,z) = e\left(\frac{1}{2}u(z-\bar{z})^{-1}u^t\right)\sum_{x\in\mathbb{Z}^2} e\left(\frac{1}{2}(x+r)z(x+r)^t + (x+r)(u+s)^t\right).$$

Put  $f_{m,r,s}(u) = \theta_{r,s}(muw_2^{-1}, mw_2^{-1}w_1).$ 

**Theorem B.2.** (1) For  $r, s \in \mathbb{Q}^2$ , we have  $f_{m,r,s}(u) \in T^{ar}(H, \epsilon, L)$  for H defined in (38) and  $\epsilon$  defined by

$$\epsilon(aw_1 + bw_2) = e\left(\frac{1}{2}m \cdot ab^t + mbr^t - as^t\right).$$

Moreover, the functions

$$f_{m,r+j,s}$$
 for  $j \in m^{-1} \begin{pmatrix} 1 & 0 \\ 0 & \alpha^{-1} \end{pmatrix} \mathbb{Z}^2 / \mathbb{Z}^2$ 

give a basis of  $T^{ar}(H, \epsilon, L)$  over the field of algebraic numbers in  $\mathbb{C}$ .

(2) Let p be an unramified prime number for  $T(H, \epsilon, L)$ . When z is diagonal,  $f_{m,r,s}$  is a p-integral element in  $T^{\text{int}}(H, \epsilon, L)$ . Thus  $T^{\text{int}}(H, \epsilon, L)$  is spanned by the functions  $f_{m,r+j,s}$  over  $i_{\infty}(i_p^{-1}(\mathbb{O}))$ . *Proof.* See [Shimura 1976, Proposition 2.5] for the proof of (1). For (2), though it is possible to remove the condition that *z* is diagonal, I will skip the discussion of the more general results because the proof would be more technical. When  $z = \begin{pmatrix} z_1 & 0 \\ 0 & z_2 \end{pmatrix}$  is diagonal,

$$\theta_{r,s}(u,z) = \prod_{i=1}^{2} \theta_{r_i,s_i}(u_i,z_i),$$

where

$$\theta_{r_i,s_i}(u_i, z_i) = e\left(\frac{1}{2}u_i(z_i - \bar{z}_i)^{-1}u_i\right) \sum_{x \in \mathbb{Z}} e\left(\frac{1}{2}(x + r_i)z_i(x + r_i) + (x + r_i)(u_i + s_i)\right),$$

for  $r = (r_1, r_2)$  and so on. Notice that these  $\theta_{r_i,s_i}(u_i, z_i)$  are essentially theta functions in the one-dimensional case. The *p*-integrality of them is confirmed by [Finis 2006, Lemma 3.3]. Thus, the *p*-integrality of the  $f_{m,r+j,s}$  follows.

**Remark B.3.** When *L* corresponds to a diagonal element in  $\mathfrak{H}_2$ ,  $\mathscr{A}_L$  is isomorphic to the product of two elliptic curves. In this case, the  $i_{\infty}(i_p^{-1}(\mathbb{O}))$ -basis of  $T^{\text{int}}(H, \epsilon, L)$  is given in Theorem B.2(2), and the number of elements inside is exactly equal to the dimension of  $T^{\text{ar}}(H, \epsilon, L)$ . In general, when  $\mathscr{A}_L$  is an abelian variety with CM, it is always isogenous to  $E_1 \times E_2$ , where  $E_1$  and  $E_2$  are elliptic curves with CM. Such isogeny induces the isomorphism between modules of integral theta functions of  $\mathscr{A}_L$  and  $E_1 \times E_2$  with appropriate choices of Riemann forms accordingly.

Then let me briefly explain the relation between classical theta functions and adelic theta functions. One can find the discussion of one-dimensional case in [Finis 2006]. Define  $T_{\mathbb{A}}(m, L, U_f)$  to be the space of all smooth functions

$$\Theta: N(\mathbb{Q})U(\zeta)(\mathbb{Q}) \setminus N(\mathbb{A})U(\zeta)(\mathbb{A}) / U(\zeta)_{\infty}U_{f}N(L)_{f} \to \mathbb{C},$$

where  $U_f$  is a certain open compact subgroup of  $U(\zeta)$  at finite places such that the level is prime to p, and

$$N(L)_{f} = \{(w, t) \mid w \in \hat{L}, t + \frac{1}{2}w\zeta w^{*} \in \mu(L)\hat{\mathbb{O}}_{K} \},\$$

where  $\mu(L)$  is the ideal generated by  $w\zeta w^*$  for all  $w \in L$ . The function  $\Theta$  satisfies

$$\Theta((0, t)r) = e(mt)\Theta(r) \quad \text{for } r \in N(\mathbb{A})U(\zeta)(\mathbb{A}).$$

Because  $U(\zeta)$  is anisotropic,  $U(\zeta)(\mathbb{Q}) \setminus U(\zeta)(\mathbb{A}) / U(\zeta)_{\infty} U_f$  consists of a finite set of points  $\{u_1, \ldots, u_s\} \subset U(\zeta)(\mathbb{A}_f)$ . So we have

$$T_{\mathbb{A}}(m, L, U_f) = \bigoplus_{i=1}^{s} T(m, u_i L), \quad \Theta \mapsto (\Theta_i)$$

Bei Zhang

such that  $\Theta_i(n) = \Theta(nu_i)$  for  $n \in N(\mathbb{A})$ . Then one may check for each *i*,

$$\theta_i(w_{\infty}) = e\left(-m\frac{w\zeta w^*}{2}\right)\Theta_i((w_{\infty}, 0))$$

is a classical theta function in  $T(H, \epsilon, u_i L)$ , where H and  $\epsilon$  are defined according to the lattice  $u_i L$ , and

$$\langle \Theta, \Theta' \rangle = \sum_{i} \langle \theta_i, \theta'_i \rangle \text{ for } \Theta, \Theta' \in T_{\mathbb{A}}(m, L, U_f),$$

where

$$\langle \Theta, \Theta' \rangle = \int_{N(\mathbb{Q})U(\zeta)(\mathbb{Q}) \setminus N(\mathbb{A})U(\zeta)(\mathbb{A})} \Theta(r) \overline{\Theta'(r)} \, dr$$

and

$$\langle \theta_i, \theta_i' \rangle = \frac{1}{\mu(L)} \int_{\mathbb{C}/L} \theta_i(w) \overline{\theta_i'(w)} \, dw.$$

#### Acknowledgments

This paper is revised from my thesis at Columbia University. I am deeply grateful to my advisor Eric Urban for the suggestion of studying this topic. His classes on Iwasawa theory and numerous discussions with him still serve as guidance for my research. I am also very thankful to many mathematicians from Columbia University, IAS, Princeton University, Northwestern University and others who share ideas with me and give me encouragement. Finally, I would like to thank the referee for suggestions that helped improve the paper.

#### References

- [Finis 2006] T. Finis, "Divisibility of anticyclotomic *L*-functions and theta functions with complex multiplication", *Ann. of Math.* (2) **163**:3 (2006), 767–807. MR 2007m:11151 Zbl 1111.11047
- [Gelbart 1975] S. S. Gelbart, *Automorphic forms on adèle groups*, Annals of Mathematics Studies **83**, Princeton University Press, 1975. MR 52 #280 Zbl 0329.10018
- [Godement and Jacquet 1972] R. Godement and H. Jacquet, *Zeta functions of simple algebras*, Lecture Notes in Mathematics **260**, Springer, Berlin, 1972. MR 49 #7241 Zbl 0244.12011
- [Greenberg 1994] R. Greenberg, "Iwasawa theory and *p*-adic deformations of motives", pp. 193–223 in *Motives* (Seattle, WA, 1991), edited by U. Jannsen et al., Proc. Sympos. Pure Math. **55**, Amer. Math. Soc., Providence, RI, 1994. MR 95i:11053 Zbl 0819.11046
- [Harris et al. 1996] M. Harris, S. S. Kudla, and W. J. Sweet, "Theta dichotomy for unitary groups", *J. Amer. Math. Soc.* **9**:4 (1996), 941–1004. MR 96m:11041 Zbl 0870.11026

[Harris et al. 2006] M. Harris, J.-S. Li, and C. M. Skinner, "*p*-adic *L*-functions for unitary Shimura varieties, I: Construction of the Eisenstein measure", *Doc. Math.* Extra Vol. (2006), 393–464. MR 2008d:11042 Zbl 1143.11019

[Hida 2004a] H. Hida, "Non-vanishing modulo *p* of Hecke *L*-values", pp. 735–784 in *Geometric aspects of Dwork theory*, vol. 2, edited by A. Adolphson et al., de Gruyter, Berlin, 2004. MR 2006b:11055 Zbl 1162.11342

- [Hida 2004b] H. Hida, *p-adic automorphic forms on Shimura varieties*, Springer, New York, 2004. MR 2005e:11054 Zbl 1055.11032
- [Hsieh 2011a] M. Hsieh, "Eisenstein congruence on unitary groups and Iwasawa main conjecture for CM fields", preprint, 2011, available at http://www.math.ntu.edu.tw/~mlhsieh/research.htm.
- [Hsieh 2011b] M.-L. Hsieh, "Ordinary *p*-adic Eisenstein series and *p*-adic *L*-functions for unitary groups", *Ann. Inst. Fourier* (*Grenoble*) **61**:3 (2011), 987–1059. MR 2918724 Zbl 06002990
- [Ichino 2004] A. Ichino, "A regularized Siegel–Weil formula for unitary groups", *Math. Z.* **247**:2 (2004), 241–277. MR 2005g:11082 Zbl 1071.11022
- [Kudla and Rallis 1988] S. S. Kudla and S. Rallis, "On the Weil–Siegel formula", *J. Reine Angew. Math.* **387** (1988), 1–68. MR 90e:11059 Zbl 0644.10021
- [Kudla and Sweet 1997] S. S. Kudla and W. J. Sweet, Jr., "Degenerate principal series representations for U(n, n)", *Israel J. Math.* **98** (1997), 253–306. MR 98h:22021 Zbl 0896.22007
- [Li 1992] J.-S. Li, "Nonvanishing theorems for the cohomology of certain arithmetic quotients", *J. Reine Angew. Math.* **428** (1992), 177–217. MR 93e:11067 Zbl 0749.11032
- [Mainardi 2004] F. Mainardi, *Sur la conjecture principale de Iwasawa pour les corps CM*, thesis, Université de Paris-Nord (Villetaneuse, Seine-Saint-Denis / Bobigny), 2004.
- [Mazur and Wiles 1984] B. Mazur and A. Wiles, "Class fields of abelian extensions of **Q**", *Invent. Math.* **76**:2 (1984), 179–330. MR 85m:11069 Zbl 0545.12005
- [Murase and Sugano 2002] A. Murase and T. Sugano, "Fourier–Jacobi expansion of Eisenstein series on unitary groups of degree three", *J. Mathematical Sciences Univ. Tokyo* **9**:2 (2002), 347–404. MR 2003f:11065 Zbl 1143.11319
- [Ribet 1976] K. A. Ribet, "A modular construction of unramified *p*-extensions of  $Q(\mu_p)$ ", *Invent. Math.* **34**:3 (1976), 151–162. MR 54 #7424 Zbl 0338.12003
- [Shimura 1976] G. Shimura, "Theta functions with complex multiplication", *Duke Math. J.* **43**:4 (1976), 673–696. MR 54 #12664 Zbl 0371.14022
- [Shimura 1997] G. Shimura, *Euler products and Eisenstein series*, CBMS Regional Conference Series in Mathematics **93**, American Mathematical Society, Providence, RI, 1997. MR 98h:11057 Zbl 0906.11020
- [Skinner and Urban 2012] C. Skinner and E. Urban, "The Iwasawa main conjecture for GL(2)", preprint, 2012, available at http://www.math.columbia.edu/~urban/eurp/MC.pdf.
- [Urban 2001] E. Urban, "Selmer groups and the Eisenstein–Klingen ideal", *Duke Math. J.* **106**:3 (2001), 485–525. MR 2002b:11073 Zbl 1061.11027
- [Urban 2006] E. Urban, "Groupes de Selmer et fonctions *L p*-adiques pour les representations modulaires adjointes", preprint, 2006, available at http://www.math.columbia.edu/~urban/EURP.html.
- [Vatsal 2003] V. Vatsal, "Special values of anticyclotomic *L*-functions", *Duke Math. J.* **116**:2 (2003), 219–261. MR 2004e:11069b Zbl 0262.46032
- [Weil 1965] A. Weil, "Sur la formule de Siegel dans la théorie des groupes classiques", *Acta Math.* 113 (1965), 1–87. MR 36 #6421 Zbl 0161.02304
- [Wiles 1990] A. Wiles, "The Iwasawa conjecture for totally real fields", *Ann. of Math.* (2) **131**:3 (1990), 493–540. MR 91i:11163 Zbl 0719.11071
- [Yang 1997] T. Yang, "Theta liftings and Hecke *L*-functions", *J. Reine Angew. Math.* **485** (1997), 25–53. MR 98e:11056 Zbl 0867.11037

Communicated by John H. Coates Received 2011-01-17 Revised 2012-02-03 Accepted 2012-03-03

zbtaishan@gmail.com Northwestern University, Department of Mathematics, 2033 Sheridan Rd., Evanston, IL 60202, United States

mathematical sciences publishers





# The phase limit set of a variety

Mounir Nisse and Frank Sottile

A coamoeba is the image of a subvariety of a complex torus under the argument map to the real torus. We describe the structure of the boundary of the coamoeba of a variety, which we relate to its logarithmic limit set. Detailed examples of lines in three-dimensional space illustrate and motivate these results.

#### 1. Introduction

A coamoeba is the image of a subvariety of a complex torus under the argument map to the real torus. Coamoebae are cousins to amoebae, which are images of subvarieties under the coordinatewise logarithm map  $z \mapsto \log |z|$ . Amoebae were introduced by Gelfand, Kapranov, and Zelevinsky [1994] and have subsequently been widely studied [Kenyon et al. 2006; Mikhalkin 2000; Passare and Rullgård 2004; Purbhoo 2008]. Coamoebae were introduced by Passare in a talk in 2004, and they appear to have many beautiful and interesting properties. For example, coamoebae of  $\mathcal{A}$ -discriminants in dimension 2 are unions of two nonconvex polyhedra [Nilsson and Passare 2010], and a hypersurface coamoeba has an associated arrangement of codimension-1 tori contained in its closure [Nisse 2009].

Bergman [1971] introduced the logarithmic limit set  $\mathscr{L}^{\infty}(X)$  of a subvariety X of the torus as the set of limiting directions of points in its amoeba. Bieri and Groves [1984] showed that  $\mathscr{L}^{\infty}(X)$  is a rational polyhedral complex in the sphere. Logarithmic limit sets are now called tropical algebraic varieties [Speyer and Sturmfels 2004]. For a hypersurface  $\mathscr{V}(f)$ , logarithmic limit set  $\mathscr{L}^{\infty}(\mathscr{V}(f))$  consists of the directions of nonmaximal cones in the outer normal fan of the Newton polytope of f. We introduce a similar object for coamoebae and establish a structure theorem for coamoebae similar to those of Bergman and of Bieri and Groves for amoebae.

Let  $co\mathcal{A}(X)$  be the coamoeba of a subvariety X of  $(\mathbb{C}^*)^n$  with ideal I. The *phase limit set*  $\mathcal{P}^{\infty}(X)$  of X is the set of accumulation points of arguments of sequences in X with unbounded logarithm. For  $w \in \mathbb{R}^n$ , the initial variety  $in_w X \subset (\mathbb{C}^*)^n$  is the variety of the initial ideal of I. The fundamental theorem of tropical geometry

MSC2010: primary 14T05; secondary 32A60.

Keywords: coamoeba, amoeba, initial ideal, toric variety tropical geometry.

asserts that  $in_w X \neq \emptyset$  exactly when the direction of -w lies in  $\mathscr{L}^{\infty}(X)$ . We establish its analog for coamoebae.

**Theorem 1.** The closure of  $co\mathcal{A}$  is  $co\mathcal{A}(X) \cup \mathcal{P}^{\infty}(X)$ , and

$$\mathcal{P}^{\infty}(X) = \bigcup_{w \neq 0} \operatorname{cos}(\operatorname{in}_{w} X).$$

Johansson [2013] used different methods to prove this when X is a complete intersection.

The cone over the logarithmic limit set admits the structure of a rational polyhedral fan  $\Sigma$  in which all weights w in the relative interior of a cone  $\sigma \in \Sigma$  give the same initial scheme in<sub>w</sub>X. Thus, the union in Theorem 1 is finite and indexed by the images of these cones  $\sigma$  in the logarithmic limit set of X. The logarithmic limit set or tropical algebraic variety is a combinatorial shadow of X encoding many properties of X. While the coamoeba of X is typically not purely combinatorial (see the examples of lines in  $(\mathbb{C}^*)^3$  in Section 3), the phase limit set does provide a combinatorial skeleton and that we believe will be useful in the further study of coamoebae.

We give definitions and background in Section 2 and detailed examples of lines in three-dimensional space in Section 3. These examples are reminiscent of the concrete descriptions of amoebae of lines in [Theobald 2002]. We prove Theorem 1 in Section 4.

#### 2. Coamoebae, tropical varieties, and initial ideals

As a real algebraic group, the set  $\mathbb{T} := \mathbb{C}^*$  of invertible complex numbers is isomorphic to  $\mathbb{R} \times \mathbb{U}$  under the map  $(r, \theta) \mapsto e^{r+\sqrt{-1}\theta}$ . Here,  $\mathbb{U}$  is the set of complex numbers of norm 1 that may be identified with  $\mathbb{R}/2\pi\mathbb{Z}$ . The inverse map is  $z \mapsto (\log |z|, \arg(z))$ .

Let *M* be a free abelian group of finite rank and  $N = \text{Hom}(M, \mathbb{Z})$  its dual group. We use  $\langle \cdot, \cdot \rangle$  for the pairing between *M* and *N*. The group ring  $\mathbb{C}[M]$  is the ring of Laurent polynomials with exponents in *M*. It is the coordinate ring of a torus  $\mathbb{T}_N$  that is identified with  $N \otimes_{\mathbb{Z}} \mathbb{T} = \text{Hom}(M, \mathbb{T})$ , the set of group homomorphisms  $M \to \mathbb{T}$ . There are natural maps Log:  $\mathbb{T}_N \to \mathbb{R}_N = N \otimes_{\mathbb{Z}} \mathbb{R}$  and arg:  $\mathbb{T}_N \to \mathbb{U}_N = N \otimes_{\mathbb{Z}} \mathbb{U}$ , which are induced by the maps  $\mathbb{C}^* \ni z \mapsto \log |z|$  and  $z \mapsto \arg(z) \in \mathbb{U}$ . Maps  $N \to N'$ of free abelian groups induce corresponding maps  $\mathbb{T}_N \to \mathbb{T}_{N'}$  of tori and also of  $\mathbb{R}_N$ and  $\mathbb{U}_N$ . If *n* is the rank of *N*, we may identify *N* with  $\mathbb{Z}^n$ , which identifies  $\mathbb{T}_N$ with  $\mathbb{T}^n$ ,  $\mathbb{U}_N$  with  $\mathbb{U}^n$ , and  $\mathbb{R}_N$  with  $\mathbb{R}^n$ .

The *amoeba*  $\mathcal{A}(X)$  of a subvariety  $X \subset \mathbb{T}_N$  is its image under Log:  $\mathbb{T}_N \to \mathbb{R}_N$ , and the *coamoeba*  $co\mathcal{A}(X)$  of X is the image of X under arg:  $\mathbb{T}_N \to \mathbb{U}_N$ . An amoeba has a geometric-combinatorial structure at infinity encoded by the logarithmic limit

set [Bergman 1971; Bieri and Groves 1984]. Coamoebae similarly have phase limit sets that have a related combinatorial structure that we define and study in Section 4.

If we identify  $\mathbb{C}^*$  with  $\mathbb{R}^2 \setminus \{(0, 0)\}$ , then the map arg:  $\mathbb{C}^* \to \mathbb{U}$  given by  $(a, b) \mapsto (a, b)/\sqrt{a^2 + b^2}$  is a real algebraic map. Thus, coamoebae, as they are the image of a real algebraic subset of the real algebraic variety  $\mathbb{T}_N$  under the real algebraic map arg, are semialgebraic subsets of  $\mathbb{U}_N$  [Basu et al. 2006]. It would be very interesting to study them as semialgebraic sets; in particular, what are the equations and inequalities satisfied by a coamoeba? When X is a Grassmannian, such a description would generalize Richter-Gebert's five-point condition for phirotopes from rank 2 to arbitrary rank [Below et al. 2003].

Similarly, we may replace the map  $\mathbb{C}^* \ni z \mapsto \log |z| \in \mathbb{R}$  in the definition of amoebae by the map  $\mathbb{C}^* \ni z \mapsto |z| \in \mathbb{R}^+ := \{r \in \mathbb{R} \mid r > 0\}$  to obtain the *algebraic amoeba* of *X*, which is a subset of  $\mathbb{R}^+_N$ . The algebraic amoeba is a semialgebraic subset of  $\mathbb{R}^+_N$ , and we also ask for its description as a semialgebraic set.

**Example 2.** Let  $\ell \subset \mathbb{T}^2$  be defined by x + y + 1 = 0. The coamoeba  $\cos \mathcal{A}(\ell)$  is the set of points of  $\mathbb{U}^2$  of the form  $(\arg(x), \pi + \arg(x+1))$  for  $x \in \mathbb{C} \setminus \{0, -1\}$ . If *x* is real, then these points are  $(\pm \pi, 0), (\pm \pi, \pm \pi)$ , and  $(0, \pm \pi)$  if *x* lies in the intervals  $(-\infty, -1), (-1, 0)$ , and  $(0, \infty)$ , respectively. For other values, consider the picture below in the complex plane:



For  $\arg(x) \notin \{0, \pi\}$  fixed,  $\pi + \arg(x+1)$  can take on any value strictly between  $\pi + \arg(x)$  (for *w* near  $\infty$ ) and 0 (for *x* near 0), and thus,  $\operatorname{co}\mathcal{A}(\ell)$  consists of the three points  $(\pi, 0), (\pi, \pi)$ , and  $(0, \pi)$  and the interiors of the two triangles displayed below in the fundamental domain  $[-\pi, \pi]^2 \subset \mathbb{R}^2$  of  $\mathbb{U}^2$ . This should be understood modulo  $2\pi$  so that  $\pi = -\pi$ .



The coamoeba is the complement of the region

 $\left\{ (\alpha, \beta) \in [-\pi, \pi]^2 \mid |\alpha - \beta| \le \pi = \arg(-1) \right\}$ 

together with the three images of real points  $(\pm \pi, 0)$ ,  $(\pm \pi, \pm \pi)$ , and  $(0, \pm \pi)$ .

Given a general line ax + by + c = 0 with  $a, b, c \in \mathbb{C}^*$ , we may replace x by cx'/a and y by cy'/b to obtain the line x' + y' + 1 = 0 with coamoeba (1). This transformation rotates the coamoeba (1) by  $\arg(a/c)$  horizontally and  $\arg(b/c)$  vertically.

Let  $f \in \mathbb{C}[M]$  be a polynomial with support  $\mathcal{A} \subset M$ 

$$f := \sum_{\boldsymbol{m} \in \mathcal{A}} c_{\boldsymbol{m}} \cdot \boldsymbol{\xi}^{\boldsymbol{m}}, \quad \text{where } c_{\boldsymbol{m}} \in \mathbb{C}^*,$$
(2)

and we write  $\xi^{\boldsymbol{m}}$  for the element of  $\mathbb{C}[M]$  corresponding to  $\boldsymbol{m} \in M$ . Given  $w \in \mathbb{R}_N$ , let w(f) be the minimum of  $\langle \boldsymbol{m}, w \rangle$  for  $\boldsymbol{m} \in \mathcal{A}$ . Then the initial form  $\operatorname{in}_w f$  of f with respect to  $w \in \mathbb{R}_N$  is the polynomial  $\operatorname{in}_w f \in \mathbb{C}[M]$  defined by

$$\operatorname{in}_w f := \sum_{\langle \boldsymbol{m}, w \rangle = w(f)} c_{\boldsymbol{m}} \cdot \xi^{\boldsymbol{m}}.$$

Given an ideal  $I \subset \mathbb{C}[M]$  and  $w \in \mathbb{R}_N$ , the *initial ideal* with respect to w is

$$\operatorname{in}_w I := \langle \operatorname{in}_w f \mid f \in I \rangle \subset \mathbb{C}[M].$$

Lastly, when *I* is the ideal of a subvariety *X*, the *initial scheme*  $in_w X \subset \mathbb{T}_N$  is defined by the initial ideal  $in_w I$ .

The sphere  $S_N := (\mathbb{R}_N \setminus \{0\})/\mathbb{R}^+$  is the set of directions in  $\mathbb{R}_N$ . Let  $\pi : \mathbb{R}_N \setminus \{0\} \rightarrow S_N$  be the projection. The *logarithmic limit set*  $\mathscr{L}^{\infty}(X)$  of a subvariety X of  $\mathbb{T}_N$  is the set of accumulation points in  $S_N$  of sequences  $\{\pi(\text{Log}(x_n))\}$ , where  $\{x_n\} \subset X$  is an unbounded set. A sequence  $\{x_n\} \subset \mathbb{T}_N$  is unbounded if its sequence of logarithms  $\{\text{Log}(x_n)\}$  is unbounded.

A rational polyhedral cone  $\sigma \subset \mathbb{R}_N$  is the set of points  $w \in \mathbb{R}_N$  that satisfy finitely many inequalities and equations of the form

$$\langle \boldsymbol{m}, w \rangle \geq 0$$
 and  $\langle \boldsymbol{m}', w \rangle = 0$ ,

where  $m, m' \in M$ . The *dimension* of  $\sigma$  is the dimension of its linear span, and *faces* of  $\sigma$  are proper subsets of  $\sigma$  obtained by replacing some inequalities by equations. The relative interior of  $\sigma$  consists of its points not lying in any face. Also,  $\sigma$  is determined by  $\sigma \cap N$ , which is a finitely generated subsemigroup of N.

A rational polyhedral fan  $\Sigma$  is a collection of rational polyhedral cones in  $\mathbb{R}_N$ in which every two cones of  $\Sigma$  meet along a common face.

**Theorem 3.** The cone in  $\mathbb{R}_N$  over the negative  $-\mathscr{L}^{\infty}(X)$  of the logarithmic limit set of X is the set of  $w \in \mathbb{R}_N$  such that  $\operatorname{in}_w X \neq \emptyset$ . Equivalently, it is the set of  $w \in \mathbb{R}_N$  such that for every  $f \in \mathbb{C}[M]$  lying in the ideal I of X,  $\operatorname{in}_w f$  is not a monomial. This cone over  $-\mathscr{L}^{\infty}(X)$  admits the structure of a rational polyhedral

fan  $\Sigma$  with the property that if u and w lie in the relative interior of a cone  $\sigma$  of  $\Sigma$ , then  $\operatorname{in}_{u} I = \operatorname{in}_{w} I$ .

It is important to take  $-\mathscr{L}^{\infty}(X)$ . This is correct as we use the tropical convention of minimum, which is forced by our use of toric varieties to prove Theorem 1 in Section 4.2.

We write  $in_{\sigma}I$  for the initial ideal defined by points in the relative interior of a cone  $\sigma$  of  $\Sigma$ . The fan structure  $\Sigma$  is not canonical for it depends upon an identification  $M \xrightarrow{\sim} \mathbb{Z}^n$ . Moreover, it may be the case that  $\sigma \neq \tau$  but  $in_{\sigma}I = in_{\tau}I$ .

Bergman [1971] defined the logarithmic limit set of a subvariety of the torus  $\mathbb{T}_N$ , and Bieri and Groves [1984] showed it was a finite union of convex polyhedral cones. The connection to initial ideals was made more explicit through work of Kapranov [2006], and the form above is adapted from Speyer and Sturmfels [2004]. The logarithmic limit set of X is now called the tropical algebraic variety of X, and this latter work led to the field of tropical geometry.

#### 3. Lines in space

We consider coamoebae of lines in three-dimensional space. We will work in the torus  $\mathbb{TP}^3$  of  $\mathbb{P}^3$ , which is the quotient of  $\mathbb{T}^4$  by the diagonal torus  $\Delta_{\mathbb{T}}$  and similarly in  $\mathbb{UP}^3$ , the quotient of  $\mathbb{U}^4$  by the diagonal  $\Delta_{\mathbb{U}} := \{(\theta, \theta, \theta, \theta) \mid \theta \in \mathbb{U}\}$ . By *coordinate lines and planes* in  $\mathbb{UP}^3$ , we mean the images in  $\mathbb{UP}^3$  of lines and planes in  $\mathbb{U}^4$  parallel to some coordinate plane.

Let  $\ell$  be a line in  $\mathbb{P}^3$  not lying in any coordinate plane, so  $\ell$  has a parametrization

$$\phi \colon \mathbb{P}^1 \ni [s:t] \mapsto [\ell_0(s,t):\ell_1(s,t):\ell_2(s,t):\ell_3(s,t)], \tag{3}$$

where  $\ell_0$ ,  $\ell_1$ ,  $\ell_2$ , and  $\ell_3$  are nonzero linear forms that do not all vanish at the same point. For i = 0, ..., 3, let  $\zeta_i \in \mathbb{P}^1$  be the zero of  $\ell_i$ . The configuration of these zeroes determine the coamoeba of  $\ell \cap \mathbb{TP}^3$ , which we will simply write as  $co\mathcal{A}(\ell)$ .

Suppose that two zeroes coincide; say  $\zeta_3 = \zeta_2$ . Then  $\ell_3 = a\ell_2$  for some  $a \in \mathbb{C}^*$ , and so  $\ell$  lies in the translated subtorus  $z_3 = az_2$ , and its coamoeba  $\operatorname{coA}(\ell)$  lies in the coordinate subspace of  $U_3$  defined by  $\theta_3 = \arg(a) + \theta_2$ . In fact,  $\operatorname{coA}(\ell)$  is pulled back from the coamoeba of the projection of  $\ell$  to the  $\theta_3 = 0$  plane. It follows that if there are only two distinct roots among  $\zeta_0, \ldots, \zeta_3$ , then  $\operatorname{coA}(\ell)$  is a coordinate line of  $U_3$ . If three of the roots are distinct, then (up to a translation) the projection of the coamoeba  $\operatorname{coA}(\ell)$  to the  $\theta_3 = 0$  plane looks like (1) so that  $\operatorname{coA}(\ell)$  consists of two triangles lying in a coordinate plane.

For each i = 0, ..., 3, define a function depending upon a point  $[s:t] \in \mathbb{P}^1$  and  $\theta \in \mathbb{U}$  by

$$\varphi_i(s, t; \theta) := \begin{cases} \theta & \text{if } \ell_i(s, t) = 0, \\ \arg(\ell_i(s, t)) & \text{otherwise.} \end{cases}$$

For each i = 0, ..., 3, let  $h_i$  be the image in  $\mathbb{UP}^3$  of  $\mathbb{U}$  under the map

$$\theta \mapsto [\varphi_0(\zeta_i, \theta), \varphi_1(\zeta_i, \theta), \varphi_2(\zeta_i, \theta), \varphi_3(\zeta_i, \theta)].$$

**Lemma 4.** For each i = 0, ..., 3,  $h_i$  is a coordinate line in  $\mathbb{UP}^3$  that consists of accumulation points of  $co\mathcal{A}(\ell)$ .

This follows from Theorem 1. For the main idea, note that  $\arg \circ \phi(\zeta_i + \epsilon e^{\theta \sqrt{-1}})$  for  $\theta \in \mathbb{U}$  is a curve in  $\mathbb{UP}^3$  whose Hausdorff distance to the line  $h_i$  approaches 0 as  $\epsilon \to 0$ . The *phase limit set* of  $\ell$  is the union of these four lines.

**Lemma 5.** Suppose that the zeroes  $\zeta_0$ ,  $\zeta_1$ , and  $\zeta_2$  are distinct. Then

$$\mathbb{P}^1 \setminus \{\zeta_0, \zeta_1, \zeta_2\} \ni x \mapsto \arg(\ell_0(x), \ell_1(x), \ell_2(x)) \in \mathbb{U}^3 / \Delta_{\mathbb{U}} = \mathbb{U} \mathbb{P}^2$$

is constant along each arc of the circle in  $\mathbb{P}^1$  through  $\zeta_0$ ,  $\zeta_1$ , and  $\zeta_2$ .

*Proof.* After changing coordinates in  $\mathbb{P}^1$  and translating in  $\mathbb{UP}^2$  (rotating coordinates), we may assume that these roots are  $\infty$ , 0, and -1, and so the circle becomes the real line. Choosing affine coordinates, we may assume that  $\ell_0 = 1$ ,  $\ell_1 = x$ , and  $\ell_2 = x + 1$  so that we are in the situation of Example 2. Then the statement of the lemma is the computation there for *x* real in which we obtained the coordinate points  $(\pi, 0)$ ,  $(\pi, \pi)$ , and  $(0, \pi)$ .

**Lemma 6.** The phase limit lines  $h_0$ ,  $h_1$ ,  $h_2$ , and  $h_3$  are disjoint if and only if the roots  $\zeta_0, \ldots, \zeta_3$  do not all lie on a circle.

*Proof.* Suppose that two of the limit lines meet, say  $h_0$  and  $h_1$ . Without loss of generality, we suppose that we have chosen coordinates on  $\mathbb{U}^4$  and  $\mathbb{P}^1$  so that  $\zeta_i \in \mathbb{C}$  and  $\ell_i(x) = x - \zeta_i$  for i = 0, ..., 3. Then there are points  $\alpha, \beta, \theta \in \mathbb{U}$  such that

 $\begin{aligned} (\varphi_0(\zeta_0,\alpha),\varphi_1(\zeta_0,\alpha),\varphi_2(\zeta_0,\alpha),\varphi_3(\zeta_0,\alpha)) \\ &= (\varphi_0(\zeta_1,\beta),\varphi_1(\zeta_1,\beta),\varphi_2(\zeta_1,\beta),\varphi_3(\zeta_1,\beta)) + (\theta,\theta,\theta,\theta). \end{aligned}$ 

Comparing the last two components, we obtain

$$\arg(\zeta_0 - \zeta_2) = \arg(\zeta_1 - \zeta_2) + \theta$$
 and  $\arg(\zeta_0 - \zeta_3) = \arg(\zeta_1 - \zeta_3) + \theta$ ,

and so the zeroes  $\zeta_0, \ldots, \zeta_3$  have the configuration below:



But then  $\zeta_0, \ldots, \zeta_3$  are cocircular. Conversely, if  $\zeta_0, \ldots, \zeta_3$  lie on a circle *C*, then by Lemma 5, the lines  $h_i$  and  $h_j$  meet only if  $\zeta_i$  and  $\zeta_j$  are the endpoints of an arc of  $C \setminus \{\zeta_0, \ldots, \zeta_3\}$ .

**Lemma 7.** If the roots  $\zeta_0, \ldots, \zeta_3$  do not all lie on a circle, then the map

$$\operatorname{arg} \circ \phi \colon \mathbb{P}^1 \setminus \{\zeta_0, \zeta_1, \zeta_2, \zeta_3\} \to \mathbb{UP}^3$$

is an immersion.

*Proof.* Let  $x \in \mathbb{P}^1 \setminus \{\zeta_0, \zeta_1, \zeta_2, \zeta_3\}$ , which we consider to be a real two-dimensional manifold. After possibly reordering the roots, the circle  $C_1$  containing  $x, \zeta_0$ , and  $\zeta_1$  meets the circle  $C_2$  containing  $x, \zeta_2$ , and  $\zeta_3$  transversally at x. Under the derivative of the map  $\arg \circ \phi$ , tangent vectors at x to  $C_1$  and  $C_2$  are taken to nonzero vectors  $(0, 0, u_1, v_1)$  and  $(u_2, v_2, 0, 0)$  in the tangent space to  $\mathbb{U}^4$ . Furthermore, as the four roots do not all lie on a circle, we cannot have both  $u_1 = v_1$  and  $u_2 = v_2$ , and so this derivative has full rank two at x as a map from  $\mathbb{P}^1 \setminus \{\zeta_0, \zeta_1, \zeta_2, \zeta_3\} \to \mathbb{UP}^3$ , which proves the lemma.

By these lemmas, there is a fundamental difference between the coamoebae of lines when the roots of the linear forms  $\ell_i$  are cocircular and when they are not. We examine each case in detail. First, choose coordinates so that  $\zeta_0 = \infty$ . After dehomogenizing and separately rescaling each affine coordinate (e.g., identifying  $\mathbb{UP}^3$  with  $\mathbb{U}^3$  and applying phase shifts to each coordinate  $\theta_1$ ,  $\theta_2$ ,  $\theta_3$  of  $\mathbb{U}^3$ ), we may assume that the map (3) parametrizing  $\ell$  is

$$\phi \colon \mathbb{C} \ni x \mapsto (x - \zeta_1, x - \zeta_2, x - \zeta_3) \in \mathbb{C}^3.$$
(4)

Suppose first that the four roots are cocircular. As  $z_0 = \infty$ , the other three lie on a real line in  $\mathbb{C}$ , which we may assume is  $\mathbb{R}$ . That is, if the four roots are cocircular, then up to coordinate change, we may assume that the line  $\ell$  is real and the affine parametrization (4) is also real. For this reason, we will call such lines  $\ell$  real lines. We first study the boundary of  $co\mathcal{A}(\ell)$ . Suppose that x lies on a contour C in the upper half plane as in Figure 1 that contains semicircles of radius  $\epsilon$  centered at each root and a semicircle of radius  $1/\epsilon$  centered at 0 but otherwise lies along the real



Figure 1. Contour in upper half plane.

axis for  $\epsilon$ , a sufficiently small positive number. Then  $\arg(\phi(w)) \in \mathbb{U}^3$  is constant on the four segments of *C* lying along  $\mathbb{R}$  with respective values

$$(\pi, \pi, \pi), (0, \pi, \pi), (0, 0, \pi), \text{ and } (0, 0, 0),$$
 (5)

moving from left to right. On the semicircles around  $\zeta_1$ ,  $\zeta_2$ , and  $\zeta_3$ , two of the coordinates are essentially constant (but not quite equal to either 0 or  $\pi$ !) while the third decreases from  $\pi$  to 0. Finally, on the large semicircle, the three coordinates are nearly equal and increase from (0, 0, 0) to  $(\pi, \pi, \pi)$ . The image  $\arg(\phi(C))$  can be made as close as we please to the quadrilateral in  $\mathbb{U}^3$  connecting the points of (5) in cyclic order when  $\epsilon$  is sufficiently small. Thus, the image of the upper half plane under the map  $\arg \circ \phi$  is a relatively open membrane in  $\mathbb{U}^3$  that spans the quadrilateral. It lies within the convex hull of this quadrilateral, which is computed using the affine structure induced from  $\mathbb{R}^3$  by the quotient  $\mathbb{U}^3 = \mathbb{R}^3/(2\pi\mathbb{Z})^3$ .

For this, observe that its projection in any of the four coordinate directions parallel to its edges is one of the triangles of the coamoeba of the projected line in  $\mathbb{CP}^2$  of Example 2, and the convex hull of the quadrilateral is the intersection of the four preimages of these triangles.

Because  $\ell$  is real, the image of the lower half plane is isomorphic to the image of the upper half plane under the map  $(\theta_1, \theta_2, \theta_3) \mapsto (-\theta_1, -\theta_2, -\theta_3)$ , and so the coamoeba is symmetric in the origin of  $\mathbb{U}^3$  and consists of two quadrilateral patches that meet at their vertices. Here are two views of the coamoeba of the line where the roots are  $\infty$ , -1/2, 0, and 3/2:



Now suppose that the roots  $\zeta_0, \ldots, \zeta_3$  do not all lie on a circle. By Lemma 6, the four phase limit lines  $h_1, \ldots, h_3$  are disjoint, and the map from  $\ell$  to the coamoeba is an immersion. Figure 2 shows two views of the coamoeba in a fundamental domain of  $\mathbb{UP}^3$  when the roots are  $\infty$ , 1,  $\zeta$ , and  $\zeta^2$ , where  $\zeta$  is a primitive third root of infinity. This and other pictures of coamoebae of lines are animated on the website [Nisse and Sottile 2010].



Figure 2. Two views of the coamoeba of a symmetric line.

The projection of this coamoeba along a coordinate direction (parallel to one of the phase limit lines  $h_i$ ) gives a coamoeba of a line in  $\mathbb{TP}^2$  as we saw in Example 2. The line  $h_i$  is mapped to the interior of one triangle, and the vertices of the triangles are the images of line segments lying on the coamoeba. These three line segments come from the three arcs of the circle through the three roots other than  $\zeta_i$ , the root corresponding to  $h_i$ .

**Proposition 8.** The interior of the coamoeba of a general line in  $\mathbb{TP}^3$  contains twelve line segments in triples parallel to each of the four coordinate directions.

The symmetric coamoeba we show in Figure 2 has six additional line segments, two each coming from the three longitudinal circles through a third root of unity and 0 and  $\infty$ . Two such segments are visible as pinch points in the leftmost view in Figure 2. We ask, What is the maximal number of line segments on a coamoeba of a line in  $\mathbb{TP}^3$ ?

#### 4. Structure of the phase limit set

The *phase limit set*  $\mathcal{P}^{\infty}(X)$  of a complex subvariety  $X \subset \mathbb{T}_N$  is the set of all accumulation points of sequences  $\{\arg(x_n) \mid n \in \mathbb{N}\} \subset \mathbb{U}_N$ , where  $\{x_n \mid n \in \mathbb{N}\} \subset X$  is an unbounded sequence. For  $w \in N$ ,  $\operatorname{in}_w X \subset \mathbb{T}_N$  is the (possibly empty) initial scheme of X, whose ideal is the initial ideal  $\operatorname{in}_w I$ , where I is the ideal of X. Our main result, Theorem 1, is that the phase limit set of X is the union of the coamoebae of all its initial schemes.

**Remark 9.** The union of Theorem 1 is finite. By Theorem 3,  $in_w X$  is nonempty only when w lies in the cone over the logarithmic set  $\mathscr{L}^{\infty}(X)$ , which can be given the structure of a finite union of rational polyhedral cones such that any two points in the relative interior of the same cone  $\sigma$  have the same initial scheme. If we write  $in_{\sigma} X$  for the initial scheme corresponding to a cone  $\sigma$ , the torus  $\mathbb{T}_{\langle \sigma \rangle} \simeq (\mathbb{C}^*)^{\dim \sigma}$  acts on  $in_{\sigma}X$  by translation (e.g., see Corollary 13). (Here,  $\langle \sigma \rangle \subset N$  is the span of  $\sigma \cap N$ , a free abelian group of rank dim  $\sigma$ .) This implies  $co\mathcal{A}(in_{\sigma}X)$  is a union of orbits of  $co\mathcal{A}(\mathbb{T}_{\langle \sigma \rangle}) = \mathbb{U}_{\langle \sigma \rangle}$  and thus that dim $(co\mathcal{A}(in_{\sigma}X)) \leq 2 \dim(X) - \dim(\sigma)$ .

This discussion implies the following proposition:

**Proposition 10.** Let  $X \subset \mathbb{T}_N$  be a subvariety, and suppose that  $\mathbb{T}_X \subset \mathbb{T}_N$  is the largest subtorus acting on X. Then dim  $co\mathcal{A}(X) \leq min\{\dim \mathbb{T}_N, 2\dim X - \dim \mathbb{T}_X\}$ .

We prove Theorem 1 in the next two subsections.

**4.1.** *Coamoebae of initial schemes.* We review the standard dictionary relating initial ideals to toric degenerations in the context of subvarieties of  $\mathbb{T}_N$  [Gelfand et al. 1994, Chapter 6]. Let  $X \subset \mathbb{T}_N$  be a subvariety with ideal  $I \subset \mathbb{C}[M]$ . We study  $\operatorname{in}_w I$  and the initial schemes  $\operatorname{in}_w X = \mathcal{V}(\operatorname{in}_w I) \subset \mathbb{T}_N$  for  $w \in N$ . Since  $\operatorname{in}_0 I = I$  so that  $\operatorname{in}_0 X = X$ , we may assume that  $w \neq 0$ . As N is the lattice of one-parameter subgroups of  $\mathbb{T}_N$ , w corresponds to a one-parameter subgroup written as  $\mathbb{C}^* \ni t \mapsto t^w \in \mathbb{T}_N$ . Define  $\mathscr{X} \subset \mathbb{C} \times \mathbb{T}_N$  by

$$\mathscr{X} := \{ (t, x) \in \mathbb{C}^* \times \mathbb{T}_N \mid t^w \cdot x \in X \}.$$
(6)

The fiber of  $\mathscr{X}$  over a point  $t \in \mathbb{C}^*$  is  $t^{-w}X$ . Let  $\overline{\mathscr{X}}$  be the closure of  $\mathscr{X}$  in  $\mathbb{C} \times \mathbb{T}_N$ , and set  $X_0$  to be the fiber of  $\overline{\mathscr{X}}$  over  $0 \in \mathbb{C}$ .

#### **Proposition 11.** $X_0 = in_w X$ .

*Proof.* We first describe the ideal  $\mathscr{I}$  of  $\mathscr{X}$ . For  $m \in M$ , the element  $\xi^m \in \mathbb{C}[M]$  takes the value  $t^{\langle m,w \rangle} \in \mathbb{C}^*$  on the element  $t^w \in \mathbb{T}_N$ , and so if  $x \in \mathbb{T}_N$ , then  $\xi^m$  takes the value  $t^{\langle m,w \rangle} \xi^m(x) = t^{\langle m,w \rangle} m(x)$  on  $t^w x$ . Given a polynomial  $f \in \mathbb{C}[M]$  of the form

$$f := \sum_{m \in \mathcal{A}} c_m \xi^m \quad \text{for } c_m \in \mathbb{C}^*,$$

define the polynomial  $f(t) \in \mathbb{C}[t, t^{-1}][M]$  by

$$f(t) := \sum_{\boldsymbol{m} \in \mathcal{A}} c_{\boldsymbol{m}} t^{\langle \boldsymbol{m}, w \rangle} \boldsymbol{\xi}^{\boldsymbol{m}}.$$
(7)

Then  $f(t)(x) = f(t^w x)$ , so  $\mathcal{I}$  is generated by the polynomials f(t) of (7) for  $f \in I$ . A general element of  $\mathcal{I}$  is a linear combination of translates  $t^a f(t)$  of such polynomials for  $a \in \mathbb{Z}$ .

If we set w(f) to be the minimal exponent of t occurring in f(t), then

$$\operatorname{in}_w f = \sum_{\langle \boldsymbol{m}, w \rangle = w(f)} c_{\boldsymbol{m}} \xi^{\boldsymbol{m}},$$

and

$$t^{-w(f)}f(t) = \operatorname{in}_w f + \sum_{\langle \boldsymbol{m}, w \rangle > w(f)} t^{\langle \boldsymbol{m}, w \rangle - w(f)} c_{\boldsymbol{m}} \xi^{\boldsymbol{m}}.$$
This shows that  $\mathcal{I} \cap \mathbb{C}[t][M]$  is generated by polynomials  $t^{-w(f)} f(t)$ , where  $f \in I$ . Since  $\operatorname{in}_w f \in \mathbb{C}[M]$  and the remaining terms are divisible by t, we see that the ideal of  $X_0$  is generated by  $\{\operatorname{in}_w f \mid f \in \mathcal{I}\}$ , which completes the proof.

We use Proposition 11 to prove one inclusion of Theorem 1, namely that

$$\mathcal{P}^{\infty}(X) \supset \bigcup_{w \in N \setminus \{0\}} \operatorname{co}\mathcal{A}(\operatorname{in}_{w} X).$$
(8)

Fix  $0 \neq w \in N$ , and let  $\mathscr{X}, \overline{\mathscr{X}}$ , and  $X_0 = \operatorname{in}_w X$  be as in Proposition 11, and let  $x_0 \in X_0$ . We show that  $\operatorname{arg}(x_0) \in \mathscr{P}^{\infty}(X)$ . Since  $(0, x_0) \in \overline{\mathscr{X}}$ , there is an irreducible curve  $C \subset \mathscr{X}$  with  $(0, x_0) \in \overline{C}$ . The projection of  $C \subset \mathbb{C}^* \times \mathbb{T}_N$  to  $\mathbb{C}^*$  is dominant, so there exists a sequence  $\{(t_n, x_n) \mid n \in \mathbb{N}\} \subset C$  that converges to  $(0, x_0)$  with each  $t_n$  real and positive. Then  $\operatorname{arg}(x_0)$  is the limit of the sequence  $\{\operatorname{arg}(x_n)\}$ .

For each  $n \in \mathbb{N}$ , set  $y_n := t_n^w \cdot x_n \in X$ . Since  $t_n$  is positive and real, every component of  $t_n^w$  is positive and real, and so  $\arg(y_n) = \arg(x_n)$ . Thus,  $\arg(x_0)$  is the limit of the sequence  $\{\arg(y_n)\}$ . Since  $x_n$  converges to  $x_0$  and  $t_n$  converges to 0, the sequence  $\{y_n\} \subset X$  is unbounded, which implies that  $\arg(x_0)$  lies in the phase limit set of X. This proves (8).

**4.2.** *Coamoebae and tropical compactifications.* We finish the proof of Theorem 1 by establishing the other inclusion,

$$\mathcal{P}^{\infty}(X) \subset \bigcup_{w \in N \setminus \{0\}} \operatorname{cost}(\operatorname{in}_w X).$$

Suppose that  $\{x_n \mid n \in \mathbb{N}\} \subset X$  is an unbounded sequence. To study the accumulation points of the sequence  $\{\arg(x_n) \mid n \in \mathbb{N}\}\)$ , we use a compactification of X that is adapted to its inclusion in  $\mathbb{T}_N$ . Suitable compactifications are the tropical compactifications of Tevelev [2007] for in these the boundary of X is composed of initial schemes in<sub>w</sub> X of X in a manner we describe below.

By Theorem 3, the cone over the logarithmic limit set  $\mathscr{L}^{\infty}(X)$  of X is the support of a rational polyhedral fan  $\Sigma$  whose cones  $\sigma$  have the property that all initial ideals in<sub>w</sub> I coincide for w in the relative interior of  $\sigma$ .

Recall the construction of the toric variety  $Y_{\Sigma}$  associated with a fan  $\Sigma$  [Fulton 1993; Gelfand et al. 1994, Chapter 6]. For  $\sigma \in \Sigma$ , set

$$\sigma^{\vee} := \{ \boldsymbol{m} \in M \mid \langle \boldsymbol{m}, w \rangle \ge 0 \text{ for all } w \in \sigma \},\$$
  
$$\sigma^{\perp} := \{ \boldsymbol{m} \in M \mid \langle \boldsymbol{m}, w \rangle = 0 \text{ for all } w \in \sigma \}.$$

Set  $V_{\sigma} := \operatorname{spec} \mathbb{C}[\sigma^{\vee}]$  and  $\mathbb{O}_{\sigma} := \operatorname{spec} \mathbb{C}[\sigma^{\perp}]$ , which is naturally isomorphic to  $\mathbb{T}_N/\mathbb{T}_{\langle \sigma \rangle}$ , where  $\langle \sigma \rangle \subset N$  is the subgroup generated by  $\sigma \cap N$ . The map  $m \mapsto m \otimes m$  determines a comodule map  $\mathbb{C}[\sigma^{\vee}] \to \mathbb{C}[\sigma^{\vee}] \otimes \mathbb{C}[M]$ , which induces the action of

the torus  $\mathbb{T}_N$  on  $V_{\sigma}$ . Its orbits correspond to faces of the cone  $\sigma$ , and the smallest orbit  $\mathbb{O}_{\sigma}$  corresponds to  $\sigma$  itself. The inclusion  $\sigma^{\perp} \subset \sigma^{\vee}$  is split by the semigroup map

$$\sigma^{\vee} \ni \boldsymbol{m} \mapsto \begin{cases} \boldsymbol{m} & \text{if } \boldsymbol{m} \in \sigma^{\perp}, \\ 0 & \text{if } \boldsymbol{m} \notin \sigma^{\perp}, \end{cases}$$
(9)

which induces a map  $\mathbb{C}[M] \twoheadrightarrow \mathbb{C}[\sigma^{\perp}]$ , and thus, we have the  $\mathbb{T}_N$ -equivariant split inclusion

$$\mathbb{O}_{\sigma} \longleftrightarrow V_{\sigma} \xrightarrow{\pi_{\sigma}} \mathbb{O}_{\sigma}.$$
 (10)

On orbits  $\mathbb{O}_{\tau}$  in  $V_{\sigma}$ , the map  $\pi_{\sigma}$  is simply the quotient by  $\mathbb{T}_{\langle \sigma \rangle}$ .

If  $\sigma, \tau \in \Sigma$  with  $\sigma \subset \tau$ , then  $\sigma^{\vee} \supset \tau^{\vee}$ , so  $\mathbb{C}[\sigma^{\vee}] \supset \mathbb{C}[\tau^{\vee}]$ , and so  $V_{\sigma} \subset V_{\tau}$ . Since the quotient fields of  $\mathbb{C}[\sigma^{\vee}]$  and  $\mathbb{C}[M]$  coincide, these are inclusions of open sets, and these varieties  $V_{\sigma}$  for  $\sigma \in \Sigma$  glue together along these natural inclusions to give the toric variety  $Y_{\Sigma}$ . The torus  $\mathbb{T}_N$  acts on  $Y_{\Sigma}$  with an orbit  $\mathbb{O}_{\sigma}$  for each cone  $\sigma$  of  $\Sigma$ .

Since  $V_0 = \mathbb{T}_N$ ,  $Y_{\Sigma}$  contains  $\mathbb{T}_N$  as a dense subset, and thus X is a (nonclosed) subvariety. Let  $\overline{X}$  be the closure of X in  $Y_{\Sigma}$ . As the fan  $\Sigma$  is supported on the cone over  $\mathscr{L}^{\infty}(X)$ ,  $\overline{X}$  will be a tropical compactification of X, and  $\overline{X}$  is complete [Tevelev 2007, Proposition 2.3]. To understand the points of  $\overline{X} \setminus X$ , we study the intersection  $\overline{X} \cap V_{\sigma}$ , which is defined by  $I \cap \mathbb{C}[\sigma^{\vee}]$ , as well as the intersection  $\overline{X} \cap \mathbb{C}_{\sigma}$ , which is defined in  $\mathbb{C}[\sigma^{\perp}]$  by the image  $I(\sigma)$  of  $I \cap \mathbb{C}[\sigma^{\vee}]$  under the map  $\mathbb{C}[\sigma^{\vee}] \twoheadrightarrow \mathbb{C}[\sigma^{\perp}]$  induced by (10).

**Lemma 12.** The initial ideal  $\operatorname{in}_{\sigma} I \subset \mathbb{C}[M]$  of I is generated by  $I(\sigma)$  under the inclusion  $\mathbb{C}[\sigma^{\perp}] \hookrightarrow \mathbb{C}[M]$ .

*Proof.* Let  $f \in I$ . Since  $\sigma$  is a cone in  $\Sigma$ , we have that  $\operatorname{in}_{\sigma} f = \operatorname{in}_{w} f$  for all w in the relative interior of  $\sigma$ . Thus, for  $w \in \sigma$ , the function  $\mathbf{m} \mapsto \langle \mathbf{m}, w \rangle$  on exponents of monomials of f is minimized on (a superset of) the support of  $\operatorname{in}_{\sigma} f$ , and if w lies in the relative interior of  $\sigma$ , then the minimizing set is the support of  $\operatorname{in}_{\sigma} f$ . Multiplying f if necessary by  $\xi^{-\mathbf{m}}$ , where  $\mathbf{m}$  is some monomial of  $\operatorname{in}_{\sigma} f$ , we may assume that for every  $w \in \sigma$ , the linear function  $\mathbf{m} \mapsto \langle \mathbf{m}, w \rangle$  is nonnegative on the support of f so that  $f \in \mathbb{C}[\sigma^{\vee}]$ , and the function is zero on the support of  $\operatorname{in}_{\sigma} f$ . Furthermore, if w lies in the relative interior of  $\sigma$ , then it vanishes exactly on the support of  $\operatorname{in}_{\sigma} f$ . Thus,  $\operatorname{in}_{\sigma} f \in \mathbb{C}[\sigma^{\perp}]$ , which completes the proof.  $\Box$ 

Since  $\mathbb{O}_{\sigma} = \mathbb{T}_N / \mathbb{T}_{\langle \sigma \rangle}$ , Lemma 12 has the following geometric counterpart:

**Corollary 13.** By translation with  $\operatorname{in}_{\sigma} X/\mathbb{T}_{\langle \sigma \rangle} = \overline{X} \cap \mathbb{O}_{\sigma}$ ,  $\mathbb{T}_{\langle \sigma \rangle}$  acts (freely) on  $\operatorname{in}_{\sigma} X$ .

*Proof of Theorem 1.* Let  $\theta \in \mathcal{P}^{\infty}(X)$  be a point in the phase limit set of *X*. Then there exists an unbounded sequence  $\{x_n \mid n \in \mathbb{N}\} \subset X$  with

$$\lim_{n\to\infty}\arg(x_n)=\theta.$$

Since  $\overline{X}$  is compact, the sequence  $\{x_n \mid n \in \mathbb{N}\}$  has an accumulation point x in  $\overline{X}$ . As the sequence is unbounded,  $x \notin \mathbb{O}_0$ , and so  $x \in \overline{X} \setminus X$ . Thus, x is a point of  $\overline{X} \cap \mathbb{O}_\sigma$  for some cone  $\sigma \neq 0$  of  $\Sigma$ . Replacing  $\{x_n\}$  by a subsequence, we may assume that  $\lim_n x_n = x$ .

Because the map  $\pi_{\sigma}$  of (10) is continuous and is the identity on  $\mathbb{O}_{\sigma}$ , we have that  $\{\pi_{\sigma}(x_n)\}$  converges to  $\pi_{\sigma}(x) = x$ , and thus,

$$\pi_{\sigma}(\theta) = \pi_{\sigma} \left( \lim_{n \to \infty} \arg(x_n) \right) = \arg \left( \lim_{n \to \infty} \pi_{\sigma}(x_n) \right) = \arg(x) \in \operatorname{cosl}(\overline{X} \cap \mathbb{O}_{\sigma}).$$
(11)

Corollary 13 implies that  $co\mathcal{A}(\overline{X} \cap \mathbb{O}_{\sigma}) = co\mathcal{A}(in_{\sigma}X)/\mathbb{U}_{\sigma}$  as  $\mathbb{U}_{\langle\sigma\rangle} = arg(\mathbb{T}_{\langle\sigma\rangle})$ . Recall that on  $\mathbb{O}_0$ ,  $\pi_{\sigma}$  is the quotient by  $\mathbb{T}_{\langle\sigma\rangle}$ . Thus, we conclude from (11) that  $\theta \in co\mathcal{A}(in_{\sigma}X)$ , which completes the proof of Theorem 1 as  $in_{\sigma}X = in_w X$  for any w in the relative interior of  $\sigma$ .

**Example 14.** In [Nisse 2009], the closure of a hypersurface coamoeba  $co\mathcal{A}(\mathcal{V}(f))$  for  $f \in \mathbb{C}[M]$  was shown to contain a finite collection of *codual hyperplanes*. These are translates of codimension-1 subtori  $\mathbb{U}_{\sigma}$  for  $\sigma$  a cone in the normal fan of the Newton polytope of f corresponding to an edge. By Theorem 1, these translated tori are that part of the phase limit set of X corresponding to the cones  $\sigma$  dual to the edges, specifically  $co\mathcal{A}(in_{\sigma}X)$ . Since  $\sigma$  has dimension n-1, the torus  $\mathbb{T}_{\sigma}$  acts with finitely many orbits on  $in_{\sigma}X$ , which is therefore a union of finitely many translates of  $\mathbb{T}_{\sigma}$ .

The logarithmic limit set  $\mathscr{L}^{\infty}(C)$  of a curve  $C \subset \mathbb{T}_N$  is a finite collection of points in  $\mathbb{S}_N$ . Each point gives a ray in the cone over  $\mathscr{L}^{\infty}(C)$ , and the components of  $\mathscr{P}^{\infty}(C)$  corresponding to a ray  $\sigma$  are finitely many translations of the dimension-1 subtorus  $\mathbb{U}_{\sigma}$  of  $\mathbb{U}_N$ , which we referred to as lines in Section 3. These were the lines lying in the boundaries of the coamoebae  $\cos \mathscr{A}(\ell)$  of the lines  $\ell$  in  $\mathbb{T}^2$  and  $\mathbb{T}^3$ .

## References

- [Basu et al. 2006] S. Basu, R. Pollack, and M.-F. Roy, *Algorithms in real algebraic geometry*, 2nd ed., Algorithms and Computation in Mathematics **10**, Springer, Berlin, 2006. MR 2007b:14125 Zbl 1102.14041
- [Below et al. 2003] A. Below, V. Krummeck, and J. Richter-Gebert, "Complex matroids phirotopes and their realizations in rank 2", pp. 203–233 in *Discrete and computational geometry*, edited by B. Aronov et al., Algorithms Combin. **25**, Springer, Berlin, 2003. MR 2005d:05041 Zbl 1077.52521
- [Bergman 1971] G. M. Bergman, "The logarithmic limit-set of an algebraic variety", *Trans. Amer. Math. Soc.* **157** (1971), 459–469. MR 43 #6209 Zbl 0197.17102
- [Bieri and Groves 1984] R. Bieri and J. R. J. Groves, "The geometry of the set of characters induced by valuations", *J. Reine Angew. Math.* **347** (1984), 168–195. MR 86c:14001 Zbl 0526.13003
- [Einsiedler et al. 2006] M. Einsiedler, M. Kapranov, and D. Lind, "Non-Archimedean amoebas and tropical varieties", *J. Reine Angew. Math.* **601** (2006), 139–157. MR 2007k:14038 Zbl 1115.14051
- [Fulton 1993] W. Fulton, Introduction to toric varieties, Annals of Mathematics Studies 131, Princeton University Press, 1993. MR 94g:14028 Zbl 0813.14039

- [Gelfand et al. 1994] I. M. Gelfand, M. M. Kapranov, and A. V. Zelevinsky, *Discriminants, resultants, and multidimensional determinants*, Birkhäuser, Boston, MA, 1994. MR 95e:14045 Zbl 0827.14036
- [Johansson 2013] P. Johansson, "The argument cycle and the coamoeba", *Complex Variables and Elliptic Equations* **58**:3 (2013), 373–384.
- [Kenyon et al. 2006] R. Kenyon, A. Okounkov, and S. Sheffield, "Dimers and amoebae", *Ann. of Math.* (2) **163**:3 (2006), 1019–1056. MR 2007f:60014 Zbl 1154.82007
- [Mikhalkin 2000] G. Mikhalkin, "Real algebraic curves, the moment map and amoebas", *Ann. of Math.* (2) **151**:1 (2000), 309–326. MR 2001c:14083 Zbl 1073.14555
- [Nilsson and Passare 2010] L. Nilsson and M. Passare, "Discriminant coamoebas in dimension two", J. Commut. Algebra 2:4 (2010), 447–471. MR 2011k:14033 Zbl 1237.14062
- [Nisse 2009] M. Nisse, "Geometric and combinatorial structure of hypersurface coamoebas", preprint, 2009. arXiv 0906.2729
- [Nisse and Sottile 2010] M. Nisse and F. Sottile, "Coamoebae of lines in 3-space", website, 2010, available at http://www.math.tamu.edu/~sottile/research/stories/coAmoeba/.
- [Passare and Rullgård 2004] M. Passare and H. Rullgård, "Amoebas, Monge–Ampère measures, and triangulations of the Newton polytope", *Duke Math. J.* **121**:3 (2004), 481–507. MR 2005a:32005 Zbl 1043.32001
- [Purbhoo 2008] K. Purbhoo, "A Nullstellensatz for amoebas", *Duke Math. J.* 141:3 (2008), 407–445. MR 2009b:14114 Zbl 1233.14036
- [Speyer and Sturmfels 2004] D. Speyer and B. Sturmfels, "The tropical Grassmannian", *Adv. Geom.* **4**:3 (2004), 389–411. MR 2005d:14089 Zbl 1065.14071
- [Tevelev 2007] J. Tevelev, "Compactifications of subvarieties of tori", Amer. J. Math. 129:4 (2007), 1087–1104. MR 2008f:14068 Zbl 1154.14039
- [Theobald 2002] T. Theobald, "Computing amoebas", *Experiment. Math.* **11**:4 (2002), 513–526. MR 2004b:14100 Zbl 1100.14048

Communicated by Bernd Sturmfels

Received 2011-06-07	Revised 2012-02-14	Accepted 2012-03-16	
nisse@math.tamu.edu	Department of Mathematics, Texas A&M University, College Station, TX 77843, United States http://www.math.tamu.edu/~nisse/		
sottile@math.tamu.edu	Department of College Station http://www.ma	of Mathematics, Texas A&M University, on, TX 77843, United States nath.tamu.edu/~sottile/	







# Base change behavior of the relative canonical sheaf related to higher dimensional moduli

Zsolt Patakfalvi

We show that the compatibility of the relative canonical sheaf with base change fails generally in families of normal varieties. Furthermore, it always fails if the general fiber of a family of pure dimension *n* is Cohen–Macaulay and the special fiber contains a strictly  $S_{n-1}$  point. In particular, in moduli spaces with functorial relative canonical sheaves Cohen–Macaulay schemes can not degenerate to  $S_{n-1}$ schemes. Another, less immediate consequence is that the canonical sheaf of an  $S_{n-1}$ ,  $G_2$  scheme of pure dimension *n* is not  $S_3$ .

# 1. Introduction

The canonical sheaf plays a crucial role in the classification of varieties of characteristic zero. Global sections of its powers define the canonical map, which is birational onto its image for varieties of general type with mild singularities. The image is called the canonical model, and it is a unique representative of the birational equivalence class of the original variety. In particular, the canonical model can be used to construct a moduli space that classifies varieties of general type up to birational equivalence. This moduli space  $\overline{\mathfrak{M}}_h$  of stable schemes is the higher dimensional generalization of the intensely investigated space  $\overline{\mathfrak{M}}_g$  of stable curves. In order to build  $\overline{\mathfrak{M}}_h$ , it is important to understand when the canonical sheaf behaves functorially in families, that is, when it is compatible with base change.

More precisely, to obtain a compact moduli space, in  $\overline{\mathfrak{M}}_h$ , not only canonical models are allowed, but also their generalizations, the semi-log canonical models [Kollár 2010, Definition 15]. By definition these are projective schemes with semi-log canonical singularities [Hacon and Kovács 2010, Definition 3.13.5] and ample canonical bundles. The first naive definition of the moduli functor of stable schemes

MSC2010: primary 14J10; secondary 14D06, 14F10, 14E30.

*Keywords:* canonical sheaf, relative canonical sheaf, dualizing complex, relative dualizing complex, base change, depth, moduli of stable varieties.

#### Zsolt Patakfalvi

with Hilbert function *h* is then as follows. Here  $h : \mathbb{Z} \to \mathbb{Z}$  is an arbitrary function.

$$\overline{\mathfrak{M}}_{h}(B) = \left\{ f: X \to B \middle| \begin{array}{c} f \text{ is flat, proper, } X_{\bar{b}} \text{ is a semi-} \\ \log \text{ canonical model } (\forall b \in B), \\ h(m) = \chi(\omega_{X_{b}}^{[m]}) \ (\forall m \in \mathbb{Z}, b \in B) \end{array} \right\} \middle/ \cong \text{ over } B \quad (1.0.a)$$

As usual, the naive definition works only in the naive cases but not in general. More precisely, (1.0.a) is insufficient to prove the existence of a projective coarse moduli space or a proper Deligne–Mumford stack structure on  $\overline{\mathfrak{M}}_h$ ; see [Kollár 2008; 2010]. In general, (1.0.a) has to be complemented with

$$\omega_{X/B}^{[m]}|_{X_b} \cong \omega_{X_b}^{[m]}$$
 for every integer *m* and  $b \in B$ . (1.0.b)

Usually (1.0.b) is referred to as *Kollár's condition* (for instance, in [Hassett and Kovács 2004, page 238]). Note also that (1.0.b) is not necessary for reduced B, but it does add important extra restrictions when B is nonreduced.

Currently, it is not understood in every aspect why and how deeply this condition is needed. For example it is not known if in characteristic zero it is equivalent or not to the other possible choice, called *Viehweg's condition* (see [Viehweg 1995, Assumption 8.30; Hassett and Kovács 2004, page 238]):

There is an integer *m* such that 
$$\omega_{X/B}^{[m]}$$
 is a line bundle. (1.0.c)

The starting point of this article is the m = 1 case of (1.0.b), that is, the compatibility of the relative canonical sheaf with base change. We will try to understand how restrictive this condition is on flat families. The results will also yield statements about how Serre's  $S_n$  condition behaves in families and for the canonical sheaves of single schemes.

Recently it has been proven in [Kollár and Kovács 2010, Theorem 7.9.3] that the relative canonical sheaf of flat families of projective schemes (over  $\mathbb{C}$ ) with Du Bois fibers is compatible with base change. According to [Kollár and Kovács 2010, Theorem 1.4] this pertains to families with semi-log canonical fibers as well. Furthermore, compatibility holds whenever the fibers are Cohen–Macaulay [Conrad 2000, Theorem 3.6.1].

It is important to note at this point that the m = 1 case of (1.0.b) behaves differently than the rest. For m > 1 there are examples of families of normal surfaces for which (1.0.b) does not hold; see [Hacon and Kovács 2010, Section 14.A]. However, since normal surfaces are Cohen–Macaulay, condition (1.0.b) with m = 1 holds for every flat family of normal surfaces. Hence, any incompatibility can be observed only in higher dimensions. Partly due to this fact, there has been a common misbelief, sometimes even stated in articles, that the relative canonical sheaf is compatible with base change for flat families of normal varieties. The question if this compatibility holds indeed was asked about the same time independently by János Kollár and the author.

**Question 1.1** (Kollár). Is  $\omega_{X/B}|_{X_b} \cong \omega_{X_b}$  for every flat family  $X \to B$  of normal varieties?

Here we construct examples showing that the answer is no. That is, there are flat families of normal varieties over smooth curves such that the relative canonical sheaves are not compatible with base change. The examples also show that the known results are optimal in many senses. That is, the fibers of the given families can be chosen to be  $S_j$  for any  $n > j \ge 2$  and their relative canonical sheaves to be  $\mathbb{Q}$ -line bundles. The precise statement is as follows.

**Theorem 1.2.** For each  $n \ge 3$  and  $n > j \ge 2$  there is a flat family  $\mathcal{H} \to B$  of  $S_j$  (but not  $S_{j+1}$ ) normal varieties of dimension n over some open set  $B \subseteq \mathbb{P}^1$ , with  $\omega_{\mathcal{H}/B} a \mathbb{Q}$ -line bundle, such that

$$\omega_{\mathcal{H}/B}|_{\mathcal{H}_0} \not\cong \omega_{\mathcal{H}_0}, \tag{1.2.a}$$

(Here  $\mathcal{H}_0$  is the central fiber of  $\mathcal{H}$ .)

Moreover, the general fiber of  $\mathcal{H}$  can be chosen to be smooth and the central fiber to have only one singular point.

When j = n - 1 and the general fiber is Cohen–Macaulay, somewhat surprisingly, the incompatibility of (1.2.a) always holds. Furthermore, one can allow  $S_{n-1}$  points also in the general fibers provided the relative  $S_{n-1}$  locus has a component in the central fiber. The precise statement is as follows. (See Section 2 for the assumptions of the article, for instance, scheme is always separated and of finite type over  $k = \bar{k}$ , etc.)

**Theorem 1.3.** If  $f : \mathcal{H} \to B$  is a flat family of schemes of pure dimension *n* over a smooth curve, such that a component of the locus

$$\overline{\{x \in \mathcal{H} \mid x \text{ is closed, depth } \mathbb{O}_{\mathcal{H}_{f(x)}, x} = n - 1\}}$$
(1.3.a)

is contained in the special fiber  $\mathcal{H}_0$ , then the restriction homomorphism  $\omega_{\mathcal{H}/B}|_{\mathcal{H}_0} \rightarrow \omega_{\mathcal{H}_0}$  is not an isomorphism.

In particular, the contrapositive of Theorem 1.3 when the general fiber is Cohen– Macaulay yields the following corollary.

**Corollary 1.4.** If  $f : \mathcal{H} \to B$  is a flat family of schemes of pure dimension n such that  $\omega_{\mathcal{H}/B}$  is compatible with base change and the general fiber of f is Cohen-Macaulay, then the central fiber of f cannot have a closed point x such that depth  $\mathbb{O}_{\mathcal{H}_{f(x)}, x} = n - 1$ .

Corollary 1.4 has many geometric consequences with respect to building moduli spaces with functorial relative canonical sheaves. For example, cone singularities over abelian surfaces cannot be smoothed over irreducible bases. It also generalizes some aspects of theorems by Kollár and Kovács [2010, Theorem 7.12] and Hassett [2001, Theorem 1.1] stating that if all fibers are Du Bois schemes or log canonical surfaces and the general fiber is  $S_k$  or Cohen–Macaulay, respectively, then so is the central fiber.

Interestingly, the nonexistence of a depth n - 1 point is the strongest implication of the compatibility of the relative canonical sheaf with base change.

**Proposition 1.5.** *Corollary 1.4 is sharp in the sense that* n - 1 *cannot be replaced by i for any* i < n - 1.

Summarizing, Corollary 1.4 and Proposition 1.5 state that in moduli spaces satisfying Kollár's condition,  $S_{n-1}$  schemes do not appear in the irreducible components containing Cohen–Macaulay schemes. However,  $S_j$  schemes can possibly show up for some j < n - 1.

If a scheme X is Cohen–Macaulay, which by definition means that  $\mathbb{O}_X$  is Cohen–Macaulay, then  $\omega_X$  is Cohen–Macaulay as well [Kollár and Mori 1998, Corollary 5.70]. One would expect that if  $\mathbb{O}_X$  is only  $S_{n-1}$ , then typically  $\omega_X$  is also  $S_{n-1}$  or at least it can be  $S_{n-1}$ . Surprisingly the truth is quite the opposite. The following application of Theorem 1.3 states that in certain cases an  $S_{n-1}$  scheme cannot have even an  $S_3$  canonical sheaf.

**Theorem 1.6.** If X is an  $S_3$ ,  $G_2$  scheme of pure dimension n, which has a closed point with depth n - 1, then  $\omega_X$  is not  $S_3$ .

The most immediate consequences of Theorem 1.6 deal with compatibility of restriction to subvarieties. For example, one can show that on a cone *X* over a Calabi–Yau threefold *Y* with  $h^2(\mathbb{O}_Y) \neq 0$ , for an effective, normal Cartier divisor *D*,

 $\omega_X(D)|_D \cong \omega_D \iff D$  does not pass through the vertex.

Or more generally, for an  $S_{n-1}$ , normal variety X and an effective, normal Cartier divisor D,

 $\omega_X(D)|_D \cong \omega_D \iff D$  does not pass through any closed point with depth n-1.

Theorem 1.6 can also be related to log canonical centers. If (X, D) is a log canonical pair,  $D \sim_{\mathbb{Q}} -K_X$  and  $\omega_X$  is not  $S_3$  at  $x \in X$ , then x is a log canonical center of the pair (X, D) [Kollár 2011, Theorem 3]. Hence by Theorem 1.6, if X is  $S_{n-1}$  and (X, D) log canonical such that  $D \sim_{\mathbb{Q}} -K_X$ , then (X, D) has a log-canonical center at all closed points with depth n - 1. This statement is of course obvious if we know that the depth n - 1 closed points are already log-canonical centers of X. However, that is not always the case. For example, let X be the

cone, with high enough polarization, over the product *Y* of a K3 surface with the projective line and let *D* be the cone over an anti canonical divisor of *Y*. Then, (X, D) is log-canonical, *X* is  $S_{n-1}$  and the cone point is the only closed point with depth n - 1; see Lemma 4.3. Still, the vertex is not a log-canonical center of *X*, because  $K_X$  is not Q-Cartier.

Theorem 1.6 raises the following question as well.

**Question 1.7.** Is it true that if *X* is a pure *n*-dimensional scheme such that  $\mathbb{O}_X$  is  $S_l$ , but not  $S_{l+1}$ , and  $\omega_X$  is  $S_j$ , but not  $S_{i+1}$ , for some j, l < n, then  $j + l \le n + 1$ ?

**Remark 1.8.** By the methods of Section 4, the answer to Question 1.7 is positive if *X* is a cone over a smooth projective variety.

There are a couple of intuitive reasons for the failure of compatibility in (1.2.a). First, compatibility holds for the relative dualizing complex if the base is smooth by Proposition 3.3.(1). Hence  $\omega_{\mathcal{H}/B}$  is a nonfunctorial component, the -n-th cohomology sheaf, of the functorial object  $\omega_{\mathcal{H}/B}^{\bullet}$ . For example, by the proof of Theorem 1.3, if the general fiber is Cohen–Macaulay and the central fiber is  $S_{n-1}$ , the restriction homomorphism fits into an exact sequence as follows, with a nonzero term on the right.

$$0 \to \omega_{\mathcal{H}/B}|_{X_b} \to \omega_{\mathcal{H}_b} \to \mathcal{T}or^1(h^{-(n-1)}(\omega^{\bullet}_{\mathcal{H}/B}), \mathbb{O}_{\mathcal{H}_0}) \to 0$$
(1.8.a)

This shows in a precise way how the functoriality might be destroyed by passing to the lowest cohomology sheaf of  $\omega_{\mathcal{H}/B}^{\bullet}$ .

Another explanation for the incompatibility (1.2.a) is that  $\mathcal{H}_0$  is too singular. Using stable reduction one may find a replacement for  $\mathcal{H}_0$  with the mildest possible singularities. The reduction steps consist of blow-ups, finite surjective normalized base changes and contractions on the total space of the family. The output is a family, the relative canonical sheaf of which is compatible with base change by [Kollár and Kovács 2010]. At the end of the article, we also present the stable reduction of our construction using a straightforward ad hoc method. The algorithmic, and lengthy, method can be found in the preprint version of the article.

In Section 3, we start with a short background overview on the base-change properties of relative dualizing complexes and relative canonical sheaves. The proofs of the main theorems can be found in Section 6 and Section 7. Some of these results are based on the existence of projective cones with appropriately chosen singularities. In Section 4 we give a cohomological characterization of when certain sheaves on a cone are  $S_d$ . Then in Section 5 we use this characterization to give the desired examples of projective cones. In Section 8 we compute the stable limit of our construction.

#### 2. Notation and assumptions

Unless otherwise stated, *scheme* means a separated scheme of finite type over a fixed field *k* of characteristic zero and every morphism is separated. A *variety* is an integral scheme. A *projective* or *quasiprojective* scheme means a projective or *quasiprojective* scheme means a projective or *quasiprojective*, integral scheme of dimension one. If *Y* is a subscheme of *X*, then  $\mathcal{I}_{Y,X}$  is the ideal sheaf of *Y* in *X*. If  $\mathcal{I}_{Y,X}$  is a line bundle (that is, a locally free sheaf of rank one), then we define  $\mathbb{O}_X(-Y) := \mathcal{I}_{Y,X}$  and  $\mathbb{O}_X(Y) := \mathbb{O}_X(-Y)^{-1}$ . Notice that  $\mathcal{I}_{Y,X}$  being a line bundle is equivalent to *Y* being defined around every point *P* by a single nonzero divisor element of  $\mathbb{O}_{X,P}$ .

A hypersurface of a quasiprojective scheme  $X \subseteq \mathbb{P}^N$  is a subscheme  $H \subseteq X$ defined by a section of  $\mathbb{O}_X(d)$  for some d > 0. If H and H' are hypersurfaces of a quasiprojective scheme  $X \subseteq \mathbb{P}^N$ , defined by  $f_0$  and  $f_\infty \in H^0(\mathbb{P}^N, \mathbb{O}_{\mathbb{P}^N}(d))$ , respectively, then the *pencil* generated by H and H' is the subscheme  $\mathcal{H} \subseteq X \times \mathbb{P}^1$ defined by the section  $f_0t_0 + f_\infty t_1$  of  $H^0(X \times \mathbb{P}^1, \mathbb{O}(d, 1))$ . Here  $t_0$  and  $t_1$  are the usual parameters of  $\mathbb{P}^1$ , and  $f_0$  and  $f_\infty$  are viewed as elements of  $H^0(X, \mathbb{O}_X(d))$ via the natural homomorphism  $H^0(\mathbb{P}^N, \mathbb{O}_{\mathbb{P}^N}(d)) \to H^0(X, \mathbb{O}_X(d))$ .

For a complex  $\mathscr{C}^{\bullet}$  of sheaves,  $h^{i}(\mathscr{C}^{\bullet})$  is the *i*-th cohomology sheaf of  $\mathscr{C}$ . For a morphism  $f: X \to Y, \omega_{X/Y}^{\bullet} := f^{!}\mathbb{O}_{Y}$ , where  $f^{!}$  is the functor obtained in [Hartshorne 1966, Corollary VII.3.4.a]. If f has equidimensional fibers of dimension n, then  $\omega_{X/Y} := h^{-n}(\omega_{X/Y}^{\bullet})$ . Every complex and morphism of complexes is considered in the derived category  $D(qc/\cdot)$  of quasicoherent sheaves up to the equivalences defined there. If Z is a closed subscheme of X, where  $\iota : Z \to X$  is the embedding morphism, then the map  $\Re \iota_* \cong \iota_*$  identifies D(qc/Z) with a full subcategory of D(qc/X). We use this identification at multiple places, equating  $\mathscr{C}^{\bullet}$  and  $\Re \iota_* \mathscr{C}^{\bullet}$  for every  $\mathscr{C}^{\bullet} \in D(qc/Z)$ . If Z is a closed subscheme of a scheme X, then  $(\cdot)|_{Z}^{L}$  denotes the derived restriction functor, which is naturally isomorphic to  $\cdot \otimes^{L} \mathbb{O}_{Z}$  via the identification mentioned above. A *line bundle* is a locally free sheaf of rank one.

If  $X \to B$  is a morphism of schemes, then  $X_b$  is the scheme theoretic fiber of X over B. If a sheaf  $\mathcal{F}$  on X is given, then  $\mathcal{F}_b := \mathcal{F}|_{X_b}$ . The dimension dim<sub>X</sub> P of a point  $P \in X$  is the dimension of its closure in X. The acronym *slc* stands for semi-log canonical [Hacon and Kovács 2010, Definition 3.13.5].

The *depth* of a coherent sheaf  $\mathcal{F}$  at a point  $x \in X$  is by definition the depth of  $\mathcal{F}_x$  with respect to the maximal ideal  $m_{X,x}$  at x and is denoted by depth  $\mathcal{F}_x$ . The depth of a scheme X at x is depth  $\mathbb{O}_{X,x}$ . A coherent sheaf  $\mathcal{F}$  is  $S_d$  on X if for every  $x \in X$ ,

depth 
$$\mathcal{F}_x \ge \min\{d, \dim \mathbb{O}_{X,x}\}.$$
 (2.0.b)

Note that there is an ambiguity in the literature about the definition of  $S_d$  sheaves. Many sources replace  $\mathcal{O}_{X,x}$  in (2.0.b) by  $\mathcal{F}_x$ , thus gaining a weaker notion. Since every sheaf of this article has full support, or equivalently every sheaf is considered over its support, the two definitions are equivalent for all cases considered here. Hence, we decided to include the stronger notion, but the reader should feel free to think about the other one as well. For a morphism  $f : X \to B$ ,  $\mathcal{F}$  is *relative*  $S_d$  if  $\mathcal{F}|_{X_b}$  is  $S_d$  for all  $b \in B$ . The word (relative) *Cohen–Macaulay* is a synonym for (relative)  $S_{\dim X}$ .

A scheme *X* is *G<sub>r</sub>* for some  $r \ge 0$  if it is Gorenstein in codimension *r*. A point  $P \in X$  is an *associated point* of a coherent sheaf  $\mathcal{F}$  if  $m_{X,P}$  is the annihilator of some element of  $\mathcal{F}_P$ . An *associated component* of a coherent sheaf is the closure of an associated point. One can show that if  $Q \in X$ ,  $\mathcal{F}_Q \neq 0$  and  $\mathcal{P}$  is the set of prime ideals of  $\mathbb{O}_{X,Q}$  corresponding to generalizations of *Q* that are associated points of  $\mathcal{F}$ , then

$$\bigcup_{P \in \mathcal{P}} P = \{ x \in \mathbb{O}_{X,Q} \mid \text{there exists } 0 \neq m \in \mathcal{F}_Q \text{ with } xm = 0 \}$$

Consequently, if *s* is a section of a line bundle, then it does not vanish on any associated component of *X* (that is, of  $\mathbb{O}_X$ ) if and only if  $s_P$  is not a zero divisor for every  $P \in X$ . That is, if *H* is the subscheme of *X* cut out by *s*, then  $\mathcal{I}_{H,X}$  is a line bundle if and only if *s* does not vanish on any associated component of *X*.

For an  $S_2$ ,  $G_1$  scheme and an arbitrary coherent sheaf  $\mathcal{F}$ , the *n*-th *reflexive power* is

$$\mathcal{F}^{[n]} := \begin{cases} (\mathcal{F}^{\otimes n})^{**} & \text{if } n \ge 0, \\ (\mathcal{F}^{\otimes (-n)})^{*} & \text{if } n < 0. \end{cases}$$

That is, it is the reflexive hull of the *n*-th tensor power. A coherent sheaf  $\mathcal{F}$  is a  $\mathbb{Q}$ -line bundle if  $\mathcal{F}^{[n]}$  is a line bundle for some n > 0. Note that if  $f : X \to B$  is a family with  $\omega_{X_b}$  a  $\mathbb{Q}$ -line bundle for all  $b \in B$ , then  $\omega_{X/B}$  is not necessarily a  $\mathbb{Q}$ -line bundle [Hacon and Kovács 2010, Section 14.A]. However, if the  $X_b$  are  $S_2$ ,  $G_1$  schemes and  $\omega_{X/B}$  a  $\mathbb{Q}$ -line bundle then  $\omega_{X_b}$  is a  $\mathbb{Q}$ -line bundle for all  $b \in B$ ; see [Hassett and Kovács 2004, Lemma 2.6].

# 3. Background on base change for dualizing complexes

This section contains a general overview on the base change properties of relative dualizing complexes and relative canonical sheaves. For experts, some of the statements might be well known, but they are included here for completeness and easier reference. Readers more interested in geometric arguments and willing to accept the statements of this section without proof should feel free to skip to the next section.

Recall that the relative dualizing complex  $\omega_{X/B}^{\bullet}$  of a quasiprojective family  $f: X \to B$  is defined as  $f^{!}\mathbb{O}_{B}$ . Here  $f^{!}$  is the functor constructed in [Hartshorne 1966, Corollary VII.3.4.a]. The following technical point should be noted here.

**Remark 3.1.** There is also another definition of  $f^{!}$  in [Neeman 1996] as the right adjoint of  $Rf_{*}$ . The two definitions coincide for proper morphisms by [Hartshorne 1966, Theorem VII.3.3; Neeman 1996, Section 6], but not in general. For example, if X is smooth affine variety over  $B = \operatorname{Spec} k$  and f is the structure map, then Hartshorne's definition of  $f^{!}\mathbb{O}_{\operatorname{Spec} k}$  lives in cohomological degree  $-\dim X$  while Neeman's is in cohomological degree zero. See [Lipman and Hashimoto 2009, Part I, Exercise 4.2.3.d] for more details on the differences (Neeman's  $f^{!}$  is denoted  $f^{\times}$  there). We use Hartshorne's definition in this article.

The dualizing complex of a single scheme *Y* is  $\omega_Y^{\bullet} := \omega_{Y/\text{Spec}k}^{\bullet}$ . The following fact is needed in the proof of Proposition 3.3(11). It follows from the invariance of the length of maximal regular sequences [Bruns and Herzog 1993, Theorem 1.2.5].

**Fact 3.2.** Let P be a point of a subscheme H of a scheme X such that  $(\mathcal{G}_{H,X})_P$  is a line bundle, d is an integer, and  $\mathcal{F}$  is a coherent  $S_1$  sheaf with full support (that is,  $\operatorname{supp} \mathcal{F} = X$ ) on X.

- (1) depth  $\mathcal{F}_P \ge d \iff \text{depth}(\mathcal{F}|_H)_P \ge d 1$  (here  $\mathcal{F}|_H$  is regarded as a sheaf on H, not on X),
- (2) depth  $\mathscr{F}_P \ge \min\{d, \dim \mathbb{O}_{X,P}\} \iff \operatorname{depth}(\mathscr{F}|_H)_P \ge \min\{d-1, \dim \mathbb{O}_{H,P}\}.$

**Proposition 3.3.** Suppose we have a flat family  $f : \mathcal{H} \to B$  of schemes of pure dimension *n* over a smooth base, a point  $0 \in B$  and a single quasiprojective scheme *X* of pure dimension *n*.

(1) There is an isomorphism

$$\omega_{\mathcal{H}/B}^{\bullet}|_{\mathcal{H}_{0}}^{L} \cong \omega_{\mathcal{H}_{0}}^{\bullet}. \tag{3.3.a}$$

(2) Fixing any isomorphism in (3.3.a) yields natural homomorphism

$$\omega_{\mathcal{H}/B}|_{\mathcal{H}_0} \to \omega_{\mathcal{H}_0}. \tag{3.3.b}$$

- (3) If B is of pure dimension d with  $\mathbb{O}_B \cong \omega_B$ , then  $\omega_{\mathcal{H}/B}^{\bullet} \cong \omega_{\mathcal{H}}^{\bullet}[-d]$ .
- (4) If  $V \subseteq X$  is any open set, then  $\omega_V^{\bullet} \cong \omega_X^{\bullet}|_V$ .
- (5) If  $U \subseteq \mathcal{H}$  is any open set, then  $\omega_{U/B}^{\bullet} \cong \omega_{\mathcal{H}/B}^{\bullet}|_{U}$ .
- (6) If  $P \in X$  is a point, then depth<sub>P</sub>  $\mathbb{O}_X = d$  if and only if  $h^i(\omega_X^{\bullet})_P$  is zero for  $i > -d \dim_X P$  and nonzero for  $i = -d \dim_X P$ .
- (7) If  $P \in \mathcal{H}$  is a point, then depth<sub>P</sub>  $\mathbb{O}_{\mathcal{H}_{f(P)}} = d$  if and only if  $h^{i}(\omega_{\mathcal{H}/B}^{\bullet})_{P}$  is zero for  $i > -d \dim_{\mathcal{H}_{f(P)}} P$  and nonzero for  $i = -d \dim_{\mathcal{H}_{f(P)}} P$ .

- (8)  $\omega_X$  is  $S_2$ .
- (9)  $\omega_{\mathcal{H}/B}$  is  $S_2$ .
- (10) If the fibers of f are Cohen–Macaulay then  $\omega_{\mathcal{H}/B} \cong \omega_{\mathcal{H}/B}^{\bullet}$  and consequently (3.3.b) is an isomorphism.
- (11) If  $\mathcal{H}_0$  is  $S_2$  and  $G_1$ , then (3.3.b) is isomorphism if and only if

depth 
$$\omega_{\mathcal{H}/B,P} \ge \min\{3, \dim \mathbb{O}_{\mathcal{H},P}\}$$
 for every  $P \in \mathcal{H}_0$ . (3.3.c)

Furthermore if (3.3.c) is not satisfied then not only is (3.3.b) not an isomorphism, but  $\omega_{\mathcal{H}/B}|_{\mathcal{H}_0} \ncong \omega_{\mathcal{H}_0}$ .

*Proof.* First, we prove point (1). It will be an ad hoc proof, since we have not found the exact statement in the literature. The statements we found are either only for flat base change morphisms [Hartshorne 1966, Corollary VII.3.4.a] or for proper f [Lipman and Hashimoto 2009, Part I, Corollary 4.4.3]. Note that, however, it might seem that point (1) follows from base change for proper f, to the best knowledge of the author, it is not clear whether one can compactify a flat morphism to a flat morphism.

First, by [Hartshorne 1966, Corollary VII.3.4.a],  $\omega_{\mathcal{H}/B}^{\bullet}$  is compatible with flat base change. So, since Spec  $\widehat{\mathbb{O}_{B,0}}$  is flat over *B*, we may assume that *B* is the spectrum of a complete local ring of a smooth scheme and 0 is the unique closed point. In particular, then  $B \cong \text{Spec } k[[x_1, \ldots, x_m]]$ . Hence, by induction on *m*, it is enough to prove that

$$\omega_{\mathscr{H}/B}^{\bullet}|_{Y} \cong \omega_{Y/C}^{\bullet}, \tag{3.3.d}$$

where  $C := \operatorname{Spec} k[[x_1, \ldots, x_{m-1}]]$  and  $Y := \mathcal{H} \times_B C$ . To prove (3.3.d), first consider the usual exact triangle

$$\mathbb{O}_{\mathscr{H}} \xrightarrow{\mu} \mathbb{O}_{\mathscr{H}} \to \mathbb{O}_{Y} \xrightarrow{+1} , \qquad (3.3.e)$$

where  $\mu$  is multiplication by  $x_m$ . Tensoring (3.3.e) by  $\omega^{\bullet}_{\mathcal{H}/B}$  yields

$$\omega_{\mathcal{H}/B}^{\bullet} \xrightarrow{\mu \otimes \mathrm{id}_{\omega_{\mathcal{H}/B}^{\bullet}}} \omega_{\mathcal{H}/B}^{\bullet} \to \omega_{\mathcal{H}/B}^{\bullet}|_{Y}^{L} \xrightarrow{+1} .$$
(3.3.f)

On the other hand, applying  $\Re Hom(\cdot, \omega^{\bullet}_{\mathcal{H}/B})$  and a rotation to (3.3.e) yields

$$\omega_{\mathcal{H}/B}^{\bullet} \xrightarrow{\mu \otimes \mathrm{id}_{\omega_{\mathcal{H}/B}^{\bullet}}} \omega_{\mathcal{H}/B}^{\bullet} \to \mathcal{RHom}(\mathbb{O}_Y, \omega_{\mathcal{H}/B}^{\bullet})[1] \xrightarrow{+1} .$$
(3.3.g)

So, (3.3.f) and (3.3.g) together imply that

$$\mathscr{RHom}(\mathbb{O}_Y, \omega^{\bullet}_{\mathscr{H}/B})[1] \cong \omega^{\bullet}_{\mathscr{H}/B}|_Y.$$
(3.3.h)

#### Zsolt Patakfalvi

Denote by  $\iota$  and g the maps  $Y \to X$  and  $Y \to C$ , respectively. The following stream of isomorphisms finishes then the proof of point (1).

$$\omega_{\mathcal{H}/B}^{\bullet}|_{Y} \cong \underbrace{\mathcal{R}\mathcal{H}om_{X}(\mathcal{R}\iota_{*}\mathbb{O}_{Y}, \omega_{\mathcal{H}/B}^{\bullet})[1]}_{\text{by (3.3.h)}} \cong \underbrace{\mathcal{R}\mathcal{H}om_{Y}(\mathbb{O}_{Y}, \iota^{!}\omega_{\mathcal{H}/B}^{\bullet})[1]}_{\text{by Grothendieck duality}}$$

$$\cong \iota^{!}\omega_{\mathcal{H}/B}^{\bullet}[1]! \cong \underbrace{\iota^{!}f^{!}\mathbb{O}_{B}[1]}_{\text{definition of }\omega_{\mathcal{H}/B}^{\bullet}} \cong \underbrace{\iota^{!}f^{!}\omega_{B}^{\bullet}[-(m-1)]}_{\omega_{B}^{\bullet}[-m]\cong\mathbb{O}_{B}}$$

$$\cong \underbrace{\omega_{Y}^{\bullet}[-(m-1)]}_{\omega_{Y}^{\bullet}\cong(f\circ\iota)^{!}\omega_{B}^{\bullet}} \cong \underbrace{g^{!}\omega_{C}^{\bullet}[-(m-1)]}_{\omega_{Y}^{\bullet}\cong g^{!}\omega_{C}^{\bullet}} \cong \underbrace{\omega_{C}^{\bullet}[-(m-1)]}_{\omega_{C}^{\bullet}\cong g^{!}\mathbb{O}_{C}} \cong \omega_{Y/C}^{\bullet}.$$

To prove point (2), notice that since  $\omega_{\mathcal{H}/B} := h^{-n}(\omega_{\mathcal{H}/B}^{\bullet})$  is the lowest cohomology sheaf of  $\omega_{\mathcal{H}/B}^{\bullet}$ , there is a homomorphism

$$\omega_{\mathcal{H}/B}[n] \to \omega^{\bullet}_{\mathcal{H}/B}. \tag{3.3.i}$$

Applying  $(\cdot)|_{\mathcal{H}_0}^L$  to (3.3.i) and then composing with the isomorphism given by (3.3.a) yields a homomorphism

$$\omega_{\mathcal{H}/B}[n]|_{\mathcal{H}_0}^L \to \omega_{\mathcal{H}_0}^{\bullet}.$$
(3.3.j)

Finally taking -n-th cohomology sheaves of (3.3.j) yields the restriction homomorphism of (3.3.b).

Point (3) is shown by the following line of isomorphisms:

$$\omega_{\mathcal{H}/B}^{\bullet} = f^{!} \mathbb{O}_{B} \cong f^{!} \omega_{B} \cong f^{!} \omega_{B}^{\bullet} [-d] \cong \omega_{\mathcal{H}}^{\bullet} [-d].$$

To prove point (4), consider the following commutative diagram.



Since *j* is smooth of relative dimension 0, using the notation of [Hartshorne 1966], we have  $j^! \cong j^{\#} \cong j^{*}$ , and then

$$\omega_V^{\bullet} = \nu^! \mathbb{O}_{\operatorname{Spec} k} \cong j^! \mu^! \mathbb{O}_{\operatorname{Spec} k} \cong j^! \omega_X^{\bullet} \cong j^* \omega_X^{\bullet} = \omega_X^{\bullet}|_V.$$

Point (5) follows from points (4) and (3).

Point (6) is proved in [Kovács 2011, Proposition 3.2] (by taking  $\mathcal{F} := \mathbb{O}_X$ ). To prove point (7), let b := f(P) and consider the following Cartesian square.



By flat base change,

$$(\lambda')^* \omega_{\mathcal{H}/B}^{\bullet} \cong \omega_{\mathcal{H}'/\operatorname{Spec} \mathbb{G}_{B,b}}^{\bullet}.$$
(3.3.k)

That is,

$$h^{i}(\omega_{\mathscr{H}/B}^{\bullet})_{P} \cong \underbrace{h^{i}(\omega_{\mathscr{H}'/\operatorname{Spec}}^{\bullet}\mathbb{O}_{B,b})_{P}}_{\text{by (3.3.k)}} \cong \underbrace{h^{i}(\omega_{\mathscr{H}'}^{\bullet}[-\dim \mathbb{O}_{B,b}])_{P}}_{\text{by point (3)}} \cong h^{i-\dim \mathbb{O}_{B,b}}(\omega_{\mathscr{H}'}^{\bullet})_{P}.$$

Hence,

$$h^{i}(\omega_{\mathscr{H}/B}^{\bullet})_{P} \text{ is } \begin{cases} 0 & \text{if } i > -d - \dim_{\mathscr{H}_{b}} P, \\ \neq 0 & \text{if } i = -d - \dim_{\mathscr{H}_{b}} P. \end{cases}$$

$$h^{i}(\omega_{\mathscr{H}'}^{\bullet})_{P} \text{ is } \begin{cases} 0 & \text{if } i > -d - \dim_{\mathscr{H}_{b}} P - \dim_{\mathscr{D}_{b},b}, \\ \neq 0 & \text{if } i = -d - \dim_{\mathscr{H}_{b}} P - \dim_{\mathscr{D}_{b},b}. \end{cases}$$

$$depth_{P} \mathbb{O}_{\mathscr{H}'} = d + \dim_{\mathscr{D}_{b},b}$$

$$(\text{by Fact } 3.2)$$

$$(depth_{P} \mathbb{O}_{(\mathscr{H}')_{f(P)}} =) depth_{P} \mathbb{O}_{\mathscr{H}_{f(P)}} = d.$$

To prove point (8), by point (4) we may assume that X is affine. Using point (4) again we may also assume that it is projective. Then [Kollár and Mori 1998, Corollary 5.69] concludes the proof of point (8). Point (9) is a consequence of point (8) and point (3). Point (10) is shown in [Conrad 2000, Theorem 3.5.1].

To prove point (11), notice that by point (8),  $\omega_{\mathcal{H}_0}$  is  $S_2$ . Also since  $\mathcal{H}_0$  is  $G_1$ , using point (10), the homomorphism  $\omega_{\mathcal{H}/B}|_{\mathcal{H}_0} \to \omega_{\mathcal{H}_0}$  is isomorphism in codimension one. Then by [Hartshorne 1994, Theorem 1.9 and Theorem 1.12], using that  $\mathcal{H}_0$  is  $S_2$  and  $G_1$ ,  $\omega_{\mathcal{H}/B}|_{\mathcal{H}_0} \to \omega_{\mathcal{H}_0}$  is an isomorphism if and only if  $\omega_{\mathcal{H}/B}|_{\mathcal{H}_0}$  is  $S_2$ . Finally, by Fact 3.2(2), this is equivalent to (3.3.c).

#### Zsolt Patakfalvi

Notice that if (3.3.c) is not satisfied, then  $\omega_{\mathcal{H}/B}|_{\mathcal{H}_0}$  is not  $S_2$  over  $\mathcal{H}_0$ . Hence in this case not only can (3.3.b) not be isomorphism, but any isomorphism between  $\omega_{\mathcal{H}/B}|_{\mathcal{H}_0}$  and  $\omega_{\mathcal{H}_0}$  is impossible.

**Remark 3.4.** A priori, saying that (3.3.b) is an isomorphism is a stronger statement than that  $\omega_{\mathcal{H}/B}|_{\mathcal{H}_0}$  is isomorphic to  $\omega_{\mathcal{H}_0}$ . However, if *B* is smooth,  $\mathcal{H}_0$  is projective,  $S_2$  and  $G_1$ , they are equivalent by the following argument. In this case  $\omega_{\mathcal{H}_0}$  is  $S_2$  and is a line bundle over the Gorenstein locus *U*. Assume that  $\omega_{\mathcal{H}/B}|_{\mathcal{H}_0} \cong \omega_{\mathcal{H}_0}$  via an arbitrary isomorphism  $\alpha$ . Then  $\omega_{\mathcal{H}/B}|_{\mathcal{H}_0}$  is also  $S_2$  and a line bundle over *U*. Since both are  $S_2$ , homomorphisms  $\omega_{\mathcal{H}/B}|_{\mathcal{H}_0} \to \omega_{\mathcal{H}_0}$  are determined in codimension one, e.g., over *U*. Furthermore, any two isomorphisms over *U* between any two line bundles differ by multiplication with an element of  $H^0(U, \mathbb{O}_{\mathcal{H}_0})$ , where  $H^0(U, \mathbb{O}_{\mathcal{H}_0}) \cong k^*$ , by  $\mathcal{H}_0$  being  $S_2$  and projective. Since the restriction of the natural morphism  $\beta : \omega_{\mathcal{H}/B}|_{\mathcal{H}_0} \to \omega_{\mathcal{H}_0}$  over *U* is an isomorphism,  $\alpha$  differs from  $\beta$  over *U* by a multiplication with an element of  $k^*$ . However, then the same is true over entire *X*, by the codimension one determination. Hence  $\beta$  is also an isomorphism.

Finally, we conclude with a statement about restriction behavior of relative dualizing complexes and relative canonical sheaves to hypersurfaces. For that we also need a lemma about flatness of hypersurfaces.

**Lemma 3.5.** If  $f : \mathcal{X} \to B$  is a flat morphism onto a smooth curve and  $\mathcal{H} \subseteq \mathcal{X}$  is a subscheme for which  $\mathcal{I}_{\mathcal{H},\mathcal{X}}$  is a line bundle, then the following are equivalent:

- (1)  $\mathcal{I}_{\mathcal{H}_{b},\mathcal{X}_{b}}$  is a line bundle for every  $b \in B$ .
- (2)  $\mathcal{H}$  is flat over B.

In particular, if  $f : \mathcal{X} \to B$  is flat with fibers of pure dimension n and  $\mathcal{H} \subseteq \mathcal{X}$  is also flat with  $\mathcal{G}_{\mathcal{H},\mathcal{X}}$  a line bundle, then fibers of  $\mathcal{H}$  are of pure dimension n - 1.

*Proof.* We prove only the equivalence statement, since the addendum follows from the ideals  $\mathscr{I}_{\mathscr{H}_b,\mathscr{X}_b}$  being line bundles.

The statement is local on  $\mathcal{H}$ . So, fix  $P \in \mathcal{H}$  and let Q := f(P). By [Hartshorne 1977, Proposition 9.1A.a],  $\mathcal{H}$  and  $\mathcal{X}$  are flat over B at P if and only if the respective homomorphisms  $\mathbb{O}_{\mathcal{H},P} \to \mathbb{O}_{\mathcal{H},P}$  and  $\mathbb{O}_{\mathcal{X},P} \to \mathbb{O}_{\mathcal{X},P}$  induced by multiplication with some power of the local parameter t of  $\mathbb{O}_{B,Q}$  are injective. Furthermore, by induction this is equivalent to the injectivity of multiplication with the first power t.

The assumptions of the lemma state that  $(\mathscr{I}_{\mathscr{H},\mathscr{X}})_P \subseteq \mathbb{O}_{\mathscr{X},P}$  is generated by a nonzero divisor element *s*. Hence there is a commutative diagram with exact rows

and columns as follows.



By the snake lemma applied vertically,  $\ker(\cdot t) = \ker(\cdot s)$ . In particular,  $\ker(\cdot t) = 0$  if and only if  $\ker(\cdot s) = 0$ . The former is equivalent to flatness of  $\mathcal{H} \to B$  at *P* while the latter is equivalent to  $\mathcal{I}_{\mathcal{H}_O, \mathcal{X}_O}$  being a line bundle at *P*.

**Proposition 3.6.** If  $\mathscr{X} \to B$  is a flat family of pure *n*-dimensional schemes, and  $\mathscr{H} \subseteq \mathscr{X}$  a flat subscheme such that  $\mathscr{G}_{\mathscr{H},\mathscr{X}}$  is a line bundle, then

(1) there is an isomorphism

$$\omega_{\mathcal{X}/B}^{\bullet}(\mathcal{H})|_{\mathcal{H}}^{L}[-1] \cong \omega_{\mathcal{H}/B}^{\bullet}, \qquad (3.6.a)$$

(2) there is a homomorphism

$$\omega_{\mathscr{X}/B}(\mathscr{H})|_{\mathscr{H}} \to \omega_{\mathscr{H}/B}, \tag{3.6.b}$$

which is isomorphism over the relative Cohen–Macaulay locus of  $\mathcal{H} \to B$ .

*Proof.* Notice first that by Lemma 3.5,  $\mathcal{H}$  has equidimensional fibers and hence  $\omega_{\mathcal{H}/B}$  is defined indeed. To prove point (1), consider the exact sequence

$$0 \to \mathbb{O}_{\mathscr{X}} \to \mathbb{O}_{\mathscr{X}}(\mathscr{H}) \to \mathbb{O}_{\mathscr{H}}(\mathscr{H}) \to 0.$$
(3.6.c)

Applying  $(\cdot) \otimes^L \omega^{\bullet}_{\mathscr{X}/B}$  to (3.6.c) and then translating yields the exact triangle

$$\omega_{\mathscr{X}/B}^{\bullet}(\mathscr{H})|_{\mathscr{H}}^{L}[-1] \to \omega_{\mathscr{X}/B}^{\bullet} \to \omega_{\mathscr{X}/B}^{\bullet}(\mathscr{H}) \xrightarrow{+1} .$$
(3.6.d)

On the other hand if  $\iota : \mathcal{H} \to \mathcal{H}$  is the embedding morphism, then

$$\omega_{\mathscr{X}/B}^{\bullet} \cong \iota^{!} \omega_{\mathscr{X}/B}^{\bullet} = R \mathcal{H}om_{\mathscr{H}}(\mathbb{O}_{\mathscr{H}}, \iota^{!} \omega_{\mathscr{X}/B}^{\bullet}) \cong \underbrace{R \mathcal{H}om_{\mathscr{H}}(\mathbb{O}_{\mathscr{H}}, \omega_{\mathscr{X}/B}^{\bullet})}_{\text{by Grothendieck duality}}.$$

Now, applying  $R\mathcal{H}om_{\mathscr{X}}(\cdot, \omega_{\mathscr{X}/B}^{\bullet})$  to the twist of (3.6.c) by  $\mathbb{O}_{\mathscr{X}}(-\mathscr{H})$  yields the exact triangle

$$\omega_{\mathscr{H}/B}^{\bullet} \cong R\mathcal{H}om_{\mathscr{X}}(\mathbb{O}_{\mathscr{H}}, \omega_{\mathscr{X}/B}^{\bullet}) \to \omega_{\mathscr{X}/B}^{\bullet} \to \omega_{\mathscr{X}/B}^{\bullet}(\mathscr{H}) \xrightarrow{+1} .$$
(3.6.e)

Putting together (3.6.d) and (3.6.e) finishes the proof of point (1).

To prove (2), take the natural map  $\omega_{\mathscr{X}/B}[n] \to \omega_{\mathscr{X}/B}^{\bullet}$ , twist it with  $\mathbb{O}_{\mathscr{X}}(\mathscr{H})$  and then restrict to  $\mathscr{H}$ . This yields the commutative diagram

$$\omega_{\mathscr{X}/B}[n-1](\mathscr{H})|_{\mathscr{H}}^{L} \longrightarrow \omega_{\mathscr{X}/B}^{\bullet}(\mathscr{H})[-1]|_{\mathscr{H}}^{L} \xrightarrow{\cong} \omega_{\mathscr{H}/B}^{\bullet}$$
(3.6.f)  
by point (1)

Applying then  $h^{-(n-1)}(\cdot)$  to the long composition arrow of (3.6.f) yields the homomorphism (3.6.b).

Let *P* be a point of  $\mathcal{H}$  that is relatively Cohen–Macaulay over *B*, and let *b* be the image of *P* in *B*. By the openness of the relative Cohen–Macaulay locus, there is a neighborhood *U* of *P* where  $X \to B$  is relatively Cohen–Macaulay. In particular, then  $\omega_{\mathcal{H}/B}[n-1] \to \omega_{\mathcal{H}/B}^{\bullet}[-1]$  is an isomorphism over *U* by Proposition 3.3(10) and hence so is the first arrow of (3.6.f). This proves that (3.6.b) is an isomorphism in a neighborhood of *P*, which finishes the proof of point (2) as well.

**Remark 3.7.** The homomorphisms constructed in Propositions 3.3 and 3.6, for example the isomorphisms (3.3.a) and (3.6.a), are not canonical in any sense.

#### 4. Serre's condition on projective cones

In this section we consider sheaves on projective cones that are isomorphic to pullbacks from the base outside the vertex. Lemma 4.3 gives a cohomological description of when such sheaves are  $S_d$ . Before that we also need a short lemma, Lemma 4.2, about how the property  $S_d$  pulls back in flat relatively Cohen–Macaulay families.

We cite the following fact separately here, because it is used at many places throughout the article, including the aforementioned Lemma 4.2.

**Fact 4.1** [Grothendieck 1965, Proposition 6.3.1]. Let X and Y be two noetherian schemes,  $f : X \rightarrow Y$  a flat morphism,  $P \in X$  arbitrary and  $\mathcal{F}$  a coherent Y module. In this situation,

$$\operatorname{depth}_{\mathbb{O}_{X,P}}(f^*\mathcal{F})_P = \operatorname{depth}_{\mathbb{O}_{Y,f(P)}}\mathcal{F}_{f(P)} + \operatorname{depth}_{\mathbb{O}_{X_{f(P)},P}}\mathbb{O}_{X_{f(P)},P}$$

**Lemma 4.2.** If  $\mathscr{G}$  is a full-dimensional coherent  $S_d$  sheaf on the scheme X, and  $f : \mathscr{X} \to X$  is a flat, relatively Cohen–Macaulay family, then  $\mathscr{F} := f^*\mathscr{G}$  is  $S_d$  as well.

*Proof.* For every  $x \in X$ ,

$$depth \mathcal{F}_{x} = \underbrace{\det \mathcal{G}_{f(x)} + \det \mathcal{O}_{\mathscr{X}_{f(x)},x}}_{Fact 4.1} = \underbrace{\det \mathcal{G}_{f(x)} + \dim \mathcal{O}_{\mathscr{X}_{f(x)},x}}_{\mathscr{X}_{f(x)} \text{ is Cohen-Macaulay}}$$

$$\geq \underbrace{\min\{d, \dim \mathcal{O}_{X,f(x)}\} + \dim \mathcal{O}_{\mathscr{X}_{f(x)},x}}_{\mathscr{G} \text{ is } S_{d}}$$

$$\geq \min\{d, \dim \mathcal{O}_{X,f(x)} + \dim \mathcal{O}_{\mathscr{X}_{f(x)},x}\}$$

$$= \underbrace{\min\{d, \dim \mathcal{O}_{\mathscr{X},f(x)} + \dim \mathcal{O}_{\mathscr{X}_{f(x)},x}\}}_{\dim \mathcal{O}_{X,f(x)} + \dim \mathcal{O}_{\mathscr{X},x}}$$
by [Matsumura 1989, Theorem 15.1.ii].

**Lemma 4.3.** Assume that we are in the following situation:

- Y is a projective scheme,
- X is the projectivized cone over Y,
- *P* is the vertex of *X* and  $V := X \setminus P$ ,
- *d* is an integer such that  $2 \le d \le \dim X$ , and
- $\mathcal{F}$  is a coherent sheaf on X, such that  $\mathcal{F}|_V = \pi^* \mathcal{G}$  for some  $S_d$  coherent sheaf  $\mathcal{G}$  on Y, where  $\pi : V \to Y$  is the natural projection.

Then the following conditions are equivalent:

- (1) depth  $\mathcal{F}_P \geq d$ .
- (2) depth  $\mathscr{F}_P \ge \min\{d, \dim \mathbb{O}_{X,P}\}.$
- (3)  $\mathcal{F}$  is  $S_d$ .
- (4)  $\mathscr{F}$  is  $S_2$  and  $H^i(Y, \mathscr{G}(n)) = 0$  for all 0 < i < d 1 and  $n \in \mathbb{Z}$ .

*Proof.* Since  $\mathscr{G}$  is  $S_d$ ,  $\mathscr{F}$  is  $S_d$  everywhere except at the vertex P by Lemma 4.2. Hence, using the assumption  $d \leq \dim X$ ,

 $H_P^i(Z, \mathcal{F}) = 0$  for all i < d and for the affine cone Z,

where the latter equivalence follows from [Hartshorne 1977, Exercises III.3.4.b and III.2.5]. So, we are left to show that the condition  $H_P^i(Z, \mathcal{F}) = 0$  for all i < d is equivalent to point (4). Define  $U := Z \setminus P$ . Then there is a long exact sequence

$$\cdots \to H^i_P(Z, \mathcal{F}) \to H^i(Z, \mathcal{F}) \to H^i(U, \mathcal{F}) \to \cdots$$

Since Z is affine  $H^i(Z, \mathcal{F}) = 0$  for all i > 0. Hence

$$H^{i}(U, \mathcal{F}) \cong H^{i+1}_{P}(Z, \mathcal{F}) \text{ for all } i > 0.$$
(4.3.a)

So, since  $H_P^0(Z, \mathcal{F}) = H_P^1(Z, \mathcal{F}) = 0$  is assumed in point (4), it is enough to show that for all 0 < i < d - 1,

$$H^{i}(U, \mathcal{F}) \cong \bigoplus_{n \in \mathbb{Z}} H^{i}(Y, \mathcal{G}(n)).$$
 (4.3.b)

In fact we will prove this for all *i*. First, notice that  $U \cong \operatorname{Spec}_{Y}(\bigoplus_{n \in \mathbb{Z}} \mathbb{O}_{Y}(n))$  and the natural projection  $\operatorname{Spec}_Y(\bigoplus_{n \in \mathbb{Z}} \mathbb{O}_Y(n)) \to Y$  can be identified with  $\pi|_U$  via this isomorphism. Hence  $(\pi|_U)_* \mathbb{O}_U \cong \bigoplus_{n \in \mathbb{Z}} \mathbb{O}_Y(n)$  and  $R^i(\pi|_U)_* \mathbb{O}_U = 0$  for i > 0. So:

$$H^{i}(U, \mathcal{F}) \cong H^{i}(Y, (\pi|_{U})_{*}\mathcal{F}|_{U}) \cong H^{i}(Y, (\pi|_{U})_{*}(\pi|_{U})^{*}\mathcal{G}) \cong$$
$$\cong H^{i}(Y, \bigoplus_{n \in \mathbb{Z}} \mathcal{G}(n)) \cong \bigoplus_{n \in \mathbb{Z}} H^{i}(Y, \mathcal{G}(n))$$
as claimed in (4.3.b).

as claimed in (4.3.b).

# 5. Construction of varieties with prescribed singularities

In this section, normal  $S_j$  (but not  $S_{j+1}$ ) varieties of dimension  $n \ge 3$  with  $S_l$  (but not  $S_{l+1}$ ), Q-line bundle canonical sheaves are constructed for certain values of j and l. They are going to be used in Section 6 and in Section 7 to build families with prescribed base change behavior for the relative canonical sheaves. First we need some lemmas.

**Lemma 5.1.** If H is a general, high enough degree hypersurface in a projective variety X, then  $H^{i}(H, \mathbb{O}_{H}) \cong H^{i}(X, \mathbb{O}_{X})$  for every *i* such that  $0 < i < \dim H$ .

*Proof.* We start with the usual exact sequence

$$0 \to \mathbb{O}_X(-H) \to \mathbb{O}_X \to \mathbb{O}_H \to 0. \tag{5.1.a}$$

Since deg  $H \gg 0$ ,

$$H^{i}(X, \mathbb{O}_{X}(-H)) = 0 \text{ whenever } i < \dim X.$$
(5.1.b)

Taking the cohomology long exact sequence of (5.1.a) and using (5.1.b) finishes the proof. 

Iterated use of Lemma 5.1 yields the following:

**Lemma 5.2.** If *H* is a general, high enough degree complete intersection (that is, it is the intersection of hypersurfaces, all of which are high enough degree) in a smooth projective variety *X*, then  $H^i(H, \mathbb{O}_H) \cong H^i(X, \mathbb{O}_X)$  for every *i* such that  $0 < i < \dim H$ .

Finally, iterated use of the adjunction formula yields the following:

**Lemma 5.3.** If *H* is a complete intersection in a smooth projective variety *X*, then  $\omega_H \cong \omega_X(m)|_H$  for some m > 0 (here  $\mathbb{O}_X(1)$  is the very ample line bundle given by the projective embedding of *X*).

**Proposition 5.4.** For each  $n \ge 2$  and  $2 \le d$ ,  $l \le n$  such that  $l \le d$  and  $d+l \le n+2$  there is an (n+1-dimensional projective variety  $X_{n+1}$  for which

- $X_{n+1}$  is the projective cone over a smooth projective variety  $Y_n$  with vertex P,
- $X_{n+1}$  is  $S_d$  and depth  $\mathbb{O}_{X_{n+1},P} = d$ ,
- $\omega_{X_{n+1}}$  is  $S_l$  and depth  $\omega_{X_{n+1},P} = l$ , and
- $\omega_{X_{n+1}}$  is a Q-line bundle.

*Proof.* Take first two Calabi–Yau hypersurfaces *Z* and *W* of dimensions d - 1 and n + 1 - l, respectively. Let  $Y := Y_n$  be a general high enough degree complete intersection of codimension d - l in  $Z \times W$ . Notice that  $d - l \ge 0$  by assumption. Finally, let  $X_{n+1}$  be the projective cone over *Y* polarized by  $\mathbb{O}_Y(1) := \mathbb{O}_{Z \times W}(p)|_Y$  for some  $p \gg 0$  (after fixing *Y*). Here  $\mathbb{O}_{Z \times W}(1)$  is the very ample line bundle on  $Z \times W$  coming from its projective embedding.

The Künneth isomorphism yields

$$H^{q}(Z \times W, \mathbb{O}_{Z \times W}) \cong \bigoplus_{r=0}^{q} H^{r}(Z, \mathbb{O}_{Z}) \otimes H^{q-r}(W, \mathbb{O}_{W}).$$

Since Z and W are Calabi–Yau hypersurfaces of dimension d - 1 and n + 1 - l, respectively, the following holds for their cohomology table:

$$H^{q}(Z, \mathbb{O}_{Z}) \neq 0 \iff q = 0 \text{ or } d - 1,$$
  
$$H^{s}(W, \mathbb{O}_{W}) \neq 0 \iff s = 0 \text{ or } n + 1 - l.$$

Hence

$$H^q(Z \times E, \mathbb{O}_{Z \times E}) \neq 0 \iff q = 0, d - 1, n + 1 - l \text{ or } n - l + d.$$

Using Lemma 5.2 yields, for 0 < q < n,

$$H^{q}(Y, \mathbb{O}_{Y}) \neq 0 \iff q = d - 1 \text{ or } n + 1 - l.$$
(5.4.a)

Since  $p \gg 0$ , also,

$$H^q(Y, \mathbb{O}_Y(r)) = 0$$
 for every  $r$  and  $0 < q < n$ .

Then by Lemma 4.3 using that  $d - 1 \le n + 1 - l$  by assumption,  $X_{n+1}$  is  $S_d$  and depth  $\mathbb{O}_{X_{n+1},P} = d$  ( $X_{n+1}$  is  $S_2$  at the vertex, because  $p \gg 0$  and hence Y is projectively normal).

Serre duality implies that

$$H^q(Y, \omega_Y) \cong (H^{n-q}(Y, \mathbb{O}_Y))^*.$$

So, by (5.4.a), for 0 < q < n,

$$H^q(Y, \omega_Y) \neq 0 \iff q = l - 1 \text{ or } n + 1 - d.$$

Since  $X_{n+1}$  is an affine bundle over Y,  $\omega_{X_{n+1}}$  is isomorphic to the pullback of  $\omega_Y$  outside of the vertex. Then by Lemma 4.3 using that  $l - 1 \le n + 1 - d$ ,  $\omega_{X_{n+1}}$  is  $S_l$  and depth  $\omega_{X_{n+1},P} = l$  ( $\omega_{X_{n+1}}$  is always  $S_2$  by Proposition 3.3(8)).

We have left to show that the  $\omega_{X_{n+1}}$  are Q-Cartier. By Lemma 5.3,

$$\omega_Y^{\otimes p} \cong (\omega_{Z \times E}(m)|_Y)^{\otimes p} \cong (\mathbb{O}_{Z \times E}(m)|_Y)^{\otimes p} \cong \mathbb{O}_Y(m).$$

That is,  $\omega_Y^{\otimes p}$  is an integer multiple of the polarization of *Y* used at the construction of  $X_{n+1}$ . Hence, [Hacon and Kovács 2010, Exercise 3.5] concludes the proof.  $\Box$ 

# 6. Construction of families without the base change property

In this section we present the proof of Theorem 1.2. The following lemma contains the key argument. It is also used in the proofs of Proposition 1.5 and Theorem 1.6.

**Lemma 6.1.** Let  $f : \mathcal{H} \to B = \mathbb{P}^1$  be a flat pencil of hypersurfaces of a quasiprojective, equidimensional scheme X, such that  $\mathcal{I}_{\mathcal{H},X\times B}$  is a line bundle and  $\mathcal{H}$ and the closed fibers of f are  $S_2$  and  $G_1$ .

- (1) If  $\omega_X$  is  $S_3$ , the restriction map  $\omega_{\mathcal{H}/B}|_{\mathcal{H}_0} \to \omega_{\mathcal{H}_0}$  is an isomorphism.
- (2) If depth  $\omega_{X,P} \not\geq \min\{3, \dim \mathbb{O}_{X,P}\}$  for some  $P \in X$ , such that  $P \in \mathcal{H}_0$ , but  $P \notin \mathcal{H}_\infty$ , then  $\omega_{\mathcal{H}/B}|_{\mathcal{H}_0} \ncong \omega_{\mathcal{H}_0}$ .

*Proof.* Notice that by flatness of  $\mathcal{H}$  and by Lemma 3.5, it does make sense to talk about  $\omega_{\mathcal{H}/B}$ . Define  $\mathcal{X} := X \times B$ . Then  $\mathcal{H}$  is a hypersurface of  $\mathcal{X}$ . By Proposition 3.6.(2) there is a homomorphism  $\omega_{\mathcal{H}/B}(\mathcal{H})|_{\mathcal{H}} \to \omega_{\mathcal{H}/B}$ , which is an isomorphism in codimension one, over the Gorenstein locus of  $\mathcal{H}$ . Fix this homomorphism for the course of the proof.

Now, we show point (1). If  $\omega_X$  is  $S_3$ , then so is  $\omega_{\mathscr{X}/B} \cong p_1^* \omega_X$  by Lemma 4.2. Hence, by Fact 3.2.(2),  $\omega_{\mathscr{X}/B}(\mathscr{H})|_{\mathscr{H}}$  is  $S_2$ . Then, since  $\omega_{\mathscr{H}/B}$  is  $S_2$  by Proposition 3.3(9),  $\omega_{\mathscr{X}/B}(\mathscr{H})|_{\mathscr{H}} \to \omega_{\mathscr{H}/B}$  is an isomorphism everywhere by [Hartshorne 1994,

Theorems 1.9 and 1.12]. However, for every  $P \in \mathscr{X}_0$ ,

$$depth \, \omega_{\mathscr{X}/B,P} = \underbrace{\det W_{X,p_1(P)} + 1}_{\text{Fact 4.1, applied to } \omega_{\mathscr{X}/B} \cong p_1^* \omega_X} \\ \geq \underbrace{\min\{3, \dim \mathbb{O}_{X,p_1(P)}\} + 1}_{\omega_X \text{ is } S_3} = \min\{4, \dim \mathbb{O}_{\mathscr{X},P}\}. \quad (6.1.a)$$

But then, for every  $P \in \mathcal{H}_0$ ,

$$\operatorname{depth} \omega_{\mathcal{H}/B,P} = \underbrace{\operatorname{depth}(\omega_{\mathcal{H}/B}(\mathcal{H})|_{\mathcal{H}})_{P}}_{\omega_{\mathcal{H}/B} \cong \omega_{\mathcal{H}/B}(\mathcal{H})|_{\mathcal{H}}} \ge \underbrace{\min\{3, \dim \mathbb{O}_{\mathcal{H},P}\}}_{\operatorname{Fact} 3.2.(2) \text{ and } (6.1.a)}$$

which implies point (1) by Proposition 3.3(11).

To prove point (2), denote by *U* the open set  $p_1^{-1}(X \setminus (\mathcal{H}_0 \cap \mathcal{H}_\infty)) \subseteq \mathcal{X}$ . This is the set of points, the first coordinates of which are not contained in every element of the pencil  $\mathcal{H} \to B$ . By Proposition 3.3(4) and 3.3(5), we may replace  $\mathcal{X}$  by *U*, or with other words, *X* by  $X \setminus (\mathcal{H}_0 \cap \mathcal{H}_\infty)$ . In particular, then  $\mathcal{H}_0 \cap \mathcal{H}_\infty = \emptyset$  and *P* is an arbitrary point of  $\mathcal{H}_0$ , such that

depth 
$$\omega_{X,P} \not\geq \min\{3, \dim \mathbb{O}_{X,P}\}.$$
 (6.1.b)

Then all fibers of the projection  $p_1|_{\mathcal{H}} : \mathcal{H} \to X$  have dimension zero. So, for every  $Q \in \mathcal{H}$ ,

depth 
$$\omega_{\mathscr{X}/B,Q} = \underbrace{\operatorname{depth} \omega_{X,p_1(Q)} + 1}_{\operatorname{Fact} 4.1, \text{ applied to } \omega_{\mathscr{X}/B} \cong p_1^* \omega_X} \ge \underbrace{\min\{2, \dim \mathbb{O}_{X,p_1(Q)}\} + 1}_{\operatorname{Proposition} 3.3.(8)} = \min\{3, \dim \mathbb{O}_{\mathscr{X},Q}\}.$$

Then, repeating the argument of the previous paragraph  $\omega_{\mathscr{X}/B}(\mathscr{H})|_{\mathscr{H}} \cong \omega_{\mathscr{H}/B}$ . Also, at the fixed  $P \in \mathscr{H}_0$ , the following computation estimates the depth more precisely.

depth 
$$\omega_{\mathscr{X}/B,P} = \underbrace{\operatorname{depth} \omega_{X,P} + 1}_{\operatorname{Fact} 4.1, \text{ applied to } \omega_{\mathscr{X}/B} \cong p_1^* \omega_X} \not\geq \underbrace{\min\{3, \dim \mathbb{O}_{X,P}\} + 1}_{(6.1.b)} = \min\{4, \dim \mathbb{O}_{\mathscr{X},P}\} \quad (6.1.c)$$

However, then

$$\operatorname{depth} \omega_{\mathcal{H}/B,P} = \underbrace{\operatorname{depth}(\omega_{\mathcal{H}/B}(\mathcal{H})|_{\mathcal{H}})_P}_{\omega_{\mathcal{H}/B} \cong \omega_{\mathcal{H}/B}(\mathcal{H})|_{\mathcal{H}}} \not\geq \underbrace{\min\{3, \dim \mathbb{O}_{\mathcal{H},P}\}}_{\text{by Fact 3.2.(2)}},$$

which concludes the proof by Proposition 3.3(11).

371

**Remark 6.2.** The condition of  $\mathscr{I}_{\mathcal{H},\mathcal{X}}$  being a line bundle in Lemma 6.1 might look superfluous at first sight, since  $\mathscr{H}$  is a hypersurface in  $\mathscr{X}$ . However, according to Section 2, the latter only means that  $\mathscr{H}$  is the zero locus of some special section of a line bundle. That is,  $\mathscr{H}$  or  $\mathscr{H}_b$  for some  $b \in B$  could contain an entire irreducible component of  $\mathscr{X}$  or  $\mathscr{X}_b$ , respectively. Then Proposition 3.6 would not apply. Such situations should definitely be avoided.

The following is the main construction to which Lemma 6.1 is applied in this section.

**Construction 6.3.** Consider a projective cone *X* over a variety *Y*. Let *P* be the vertex of *X*. Take two hypersurfaces in *X*. The first one *H* is a projective cone over a degree *d* generic hypersurface *D* of *Y*. The second one  $\tilde{H}$  is a general degree *d* hypersurface of *X*. Denote by  $\mathcal{H} \to B$  the pencil generated by *H* and  $\tilde{H}$  (for which  $H = \mathcal{H}_0$  and  $\tilde{H} = \mathcal{H}_\infty$ ). Throughout the paper we allow ourselves to replace this family by its restriction to a small enough open neighborhood of  $0 \in B$ . Furthermore, when we compute stable reduction in Section 8, we will assume that  $d \gg 0$ .

**Lemma 6.4.** In the situation of Construction 6.3, if X is  $S_3$  and Y is  $R_1$ , then

- (1)  $\mathcal{H}$  and the closed fibers of f are normal varieties,
- (2)  $\mathcal{I}_{\mathcal{H},X\times B}$  is a line bundle,
- (3) f is flat.

*Proof.* We use the notation  $\mathscr{X} := X \times B$ . Since *Y* is a variety (that is, integral), so are *D*, *X*,  $\mathscr{X}$ ,  $\mathscr{H}_0$  and  $\mathscr{H}_\infty$ . By the definition of a pencil,  $\mathscr{H}$  is defined by a single nonzero equation locally on  $\mathscr{X}$ . So, since  $\mathscr{X}$  is integral, point (2) follows. Similarly, for every  $b \in B$ ,  $\mathscr{H}_b$  is defined locally by a single nonzero equation locally. Hence by integrality of *X*,  $\mathscr{I}_{\mathscr{H}_b, \mathscr{X}_b}$  is also a line bundle for every  $b \in B$ . Thus, Lemma 3.5 yields point (3).

To prove point (1), note that  $\mathscr{X}$  is  $S_3$  by Lemma 4.2 and by the assumption of the lemma. So, by Fact 3.2,  $\mathscr{H}$  and the closed fibers of  $\mathscr{H}$  are  $S_2$ . (Remember, in Construction 6.3 we allowed ourselves to shrink *B* around  $0 \in B$ ). Since *D* is general and *Y* is  $R_1$ , *D* is  $R_1$  as well by Bertini's theorem; see [Harris 1992, Theorem 17.16]. Therefore, so is *H*. Then, by possibly shrinking *B*, each closed fiber of  $\mathscr{H}$  is  $R_1$ . Thus all closed fibers of  $\mathscr{H}$ , and  $\mathscr{H}$  itself, are normal.

**Theorem 6.5.** In the situation of Construction 6.3, if dim  $X \ge 3$ , X is  $S_3$ , Y is  $R_1$ , and depth  $\omega_{X,P} = 2$ , then

$$\omega_{\mathcal{H}/B}|_{\mathcal{H}_0} \cong \omega_{\mathcal{H}_0}. \tag{6.5.a}$$

In addition:

- (1) If  $\omega_X$  is a  $\mathbb{Q}$ -line bundle, then  $\omega_{\mathcal{H}/B}$  is a  $\mathbb{Q}$ -line bundle. In particular then  $\omega_{\mathcal{H}_b}$  is a  $\mathbb{Q}$ -line bundle for all  $b \in B$ .
- (2) If X is  $S_d$  and depth  $\mathbb{O}_{X,P} = d$ , then  $\mathcal{H}_b$  is  $S_{d-1}$  for all  $b \in B$ , and

depth 
$$\mathbb{O}_{\mathcal{H}_0, P} = d - 1$$
.

*Proof.* By Lemma 6.4, we may apply Lemma 6.1(2) to obtain the main statement of the theorem.

To prove addendum (1), note that the normality of  $\mathcal{H}$  and  $\mathcal{H}_b$  for every  $b \in B$ , [Hartshorne 1994, Theorem 1.12] and Proposition 3.6 imply that

$$\omega_{\mathcal{H}_b}^{[n]} \cong (\omega_{\mathcal{H}_b}(\mathcal{H}_b)|_{\mathcal{H}_b})^{[n]} \text{ for any } b \in B, \text{ and } \omega_{\mathcal{H}/B}^{[n]} \cong (\omega_{\mathcal{H}/B}(\mathcal{H})|_{\mathcal{H}})^{[n]}$$
(6.5.b)

for all  $n \in \mathbb{Z}$ . Hence if  $\omega_X$  is a Q-line bundle, then (6.5.b) implies that so is  $\omega_{\mathcal{H}/B}$  and  $\omega_{\mathcal{H}_b}$  for all  $b \in B$ . To prove (2) we use Fact 3.2 once again.

Theorem 1.2 now follows as a corollary:

*Proof of Theorem 1.2.* It follows by combining Proposition 5.4 (setting d = j + 1 and l = 2), Construction 6.3 and Theorem 6.5.

## 7. Degenerations and Serre's condition

We turn to proving the statements relating Serre's condition  $S_d$  to degenerations of flat families. The first half of the section is devoted to Theorem 1.3.

**Remark 7.1.** By the restriction homomorphism  $\omega_{\mathcal{H}/B} \to \omega_{\mathcal{H}_0}$  we mean any homomorphism obtained as in Proposition 3.3(2).

Theorem 1.3 might look technical, but it applies for example to the special case, when the general fiber is Cohen–Macaulay and the central fiber contains at least one closed point with depth n - 1.

*Proof of Corollary 1.4 and Proposition 1.5.* Fix a  $2 \le i < n - 1$ . Consider the projective cone *X* given by Proposition 5.4, setting d = i + 1 and l = 3. Use then Construction 6.3 for *X*. By Lemma 6.4, this yields a flat family  $f : \mathcal{H} \to B$  of normal varieties for which Lemma 6.1(1) applies. That is, the restriction homomorphisms  $\omega_{\mathcal{H}/B}|_{\mathcal{H}_b} \to \omega_{\mathcal{H}_b}$  are isomorphisms for every  $b \in B$ . Finally, since *X* is Cohen–Macaulay outside of *P* and depth  $\mathbb{O}_{X,P} = i + 1$ , by Fact 3.2,  $\mathcal{H}_b$  is Cohen–Macaulay outside of *P*, where depth  $\mathbb{O}_{\mathcal{H}_0,P} = i$ .

We also need the following lemma in the proof of Theorem 1.3.

**Lemma 7.2.** If  $f : \mathcal{H} \to B$  is a flat morphism of schemes onto a smooth curve,  $\mathcal{F}$  is a coherent  $\mathbb{O}_{\mathcal{H}}$ -module on  $\mathcal{H}$ , and  $P \in \mathcal{H}_0$ , then

(1)  $Tor_{\mathcal{H}}^{1}(\mathcal{F}, \mathbb{O}_{\mathcal{H}_{0}})_{P} \neq 0$  if and only if  $\mathcal{F}$  has an associated component W such that  $P \in W \subseteq \mathcal{H}_{0}$ , and

Zsolt Patakfalvi

(2)  $\operatorname{Tor}_{\mathscr{H}}^{i}(\mathscr{F}, \mathbb{O}_{\mathscr{H}_{0}}) = 0$  for i > 1.

*Proof.* By restricting *B*, we may assume that  $\mathcal{I}_{0,B} \cong \mathbb{O}_B$ . Denote by *s* a generator of  $\mathcal{I}_{0,B}$  and consider the exact sequence

$$0 \to \mathbb{O}_{\mathcal{H}} \xrightarrow{\cdot, s} \mathbb{O}_{\mathcal{H}} \to \mathbb{O}_{\mathcal{H}_0} \to 0.$$
(7.2.a)

Then the long exact sequence of  $Tor^{\bullet}_{\mathcal{H}}(\mathcal{F}, \cdot)$  applied to (7.2.a) yields

$$Tor^{1}_{\mathscr{H}}(\mathscr{F}, \mathbb{O}_{\mathscr{H}}) = 0 \to Tor^{1}_{\mathscr{H}}(\mathscr{F}, \mathbb{O}_{\mathscr{H}_{0}}) \to \mathscr{F} \stackrel{\cdot s}{\to} \mathscr{F}.$$

Hence  $Tor_{\mathcal{H}}^{1}(\mathcal{F}, \mathbb{O}_{\mathcal{H}_{0}})_{P} \neq 0$  if and only if *s* annihilates something in  $\mathcal{F}_{P}$ , if and only if  $\mathcal{F}$  has an associated component *W* such that  $P \in W \subseteq \mathcal{H}_{0}$ .

Another part of the long exact sequence of  $Tor_{\mathcal{H}}^{\bullet}(\mathcal{F}, \cdot)$  applied to (7.2.a) yields the following for i > 1:

$$\mathcal{T} \sigma r^{i}_{\mathcal{H}}(\mathcal{F}, \mathbb{O}_{\mathcal{H}}) = 0 \to \mathcal{T} \sigma r^{i}_{\mathcal{H}}(\mathcal{F}, \mathbb{O}_{\mathcal{H}_{0}}) \to \mathcal{T} \sigma r^{i-1}_{\mathcal{H}}(\mathcal{F}, \mathbb{O}_{\mathcal{H}}) = 0.$$

Hence,  $Tor^{i}_{\mathcal{H}}(\mathcal{F}, \mathbb{O}_{\mathcal{H}_{0}}) = 0$  indeed if i > 1.

Proof of Theorem 1.3. Fix a closed point  $x \in \mathcal{H}_0$  with depth  $\mathbb{O}_{\mathcal{H}_0,x} = n - 1$ , contained in a component  $W \subseteq \mathcal{H}_0$  of the locus (1.3.a). The locus (1.3.a) is  $\operatorname{supp}(h^{-(n-1)}(\omega_{\mathcal{H}/B}^{\bullet}))$  by Proposition 3.3(7); hence W is also an associated component of  $h^{-(n-1)}(\omega_{X/B}^{\bullet})$ . Consider an open neighborhood of x, where every closed point has depth at least n - 1. Replacing  $\mathcal{H}$  by this neighborhood, all assumptions of the theorem stay valid, and moreover we may assume that every closed point of  $\mathcal{H}$  has depth at least n - 1. In particular, then

$$h^{i}(\omega_{\mathcal{H}/B}^{\bullet}) \neq 0 \iff i = -n \text{ or } -(n-1).$$
 (7.2.b)

Define  $\mathscr{C} := h^{-(n-1)}(\omega_{\mathscr{H}/B}^{\bullet})$ . By (7.2.b), there is an exact triangle

$$\omega_{\mathcal{H}/B}[n] \to \omega_{\mathcal{H}/B}^{\bullet} \to \mathscr{C}[n-1] \xrightarrow{+1}$$
. (7.2.c)

Applying  $\cdot \otimes_L \mathbb{O}_{\mathcal{H}_0}$  to (7.2.c) and then considering the long exact sequence of cohomology sheaves yields

$$h^{-n-1}(\mathscr{E}[n-1]\otimes_{L}\mathbb{O}_{\mathscr{H}_{0}}) \to h^{-n}(\omega_{\mathscr{H}/B}[n]\otimes_{L}\mathbb{O}_{\mathscr{H}_{0}}) \to h^{-n}(\omega_{\mathscr{H}/B}^{\bullet}\otimes_{L}\mathbb{O}_{\mathscr{H}_{0}})$$
$$\to h^{-n}(\mathscr{E}[n-1]\otimes_{L}\mathbb{O}_{\mathscr{H}_{0}}) \to h^{-n+1}(\omega_{\mathscr{H}/B}[n]\otimes_{L}\mathbb{O}_{\mathscr{H}_{0}}), \quad (7.2.d)$$

where

• 
$$h^{-n-1}(\mathscr{E}[n-1]\otimes_L \mathbb{O}_{\mathscr{H}_0}) \cong Tor^2_{\mathscr{H}}(\mathscr{E}, \mathbb{O}_{\mathscr{H}_0}) = 0$$
 by Lemma 7.2

• 
$$h^{-n}(\omega_{\mathcal{H}/B}[n] \otimes_L \mathbb{O}_{\mathcal{H}_0}) \cong \omega_{\mathcal{H}/B}|_{\mathcal{H}_0},$$

- $h^{-n}(\omega_{\mathcal{H}/B}^{\bullet} \otimes_L \mathbb{O}_{\mathcal{H}_0}) \cong h^{-n}(\omega_{\mathcal{H}_0}^{\bullet}) \cong \omega_{\mathcal{H}_0}$  by Proposition 3.3(1),
- $h^{-n}(\mathscr{E}[n-1]\otimes_L \mathbb{O}_{\mathcal{H}_0})\cong \mathcal{T}or^1_{\mathcal{H}}(\mathscr{E},\mathbb{O}_{\mathcal{H}_0})$  and

374

h<sup>-n+1</sup>(ω<sub>ℋ/B</sub>[n]⊗<sub>L</sub> O<sub>ℋ0</sub>) ≅ h<sup>1</sup>(ω<sub>ℋ/B</sub>⊗<sub>L</sub> O<sub>ℋ0</sub>) = 0 since ·⊗<sub>L</sub> O<sub>ℋ0</sub> is a left derived functor, so ω<sub>ℋ/B</sub> ⊗<sub>L</sub> O<sub>ℋ0</sub> is supported in negative cohomological degrees.

Therefore, (7.2.d) is isomorphic to the exact sequence

$$0 \to \omega_{\mathcal{H}/B}|_{\mathcal{H}_0} \to \omega_{\mathcal{H}_0} \to \mathcal{T}or^1_{\mathcal{H}}(\mathcal{E}, \mathbb{O}_{\mathcal{H}_0}) \to 0.$$

Since  $\mathscr{C}$  has an associated component through *x* contained in  $\mathscr{H}_0$ , we know that  $Tor^1_{\mathscr{H}}(\mathscr{C}, \mathbb{O}_{\mathscr{H}_0})_x \neq 0$  by Lemma 7.2, which concludes our proof.

Having finished the proof of Theorem 1.3, the rest of the section is devoted its consequence, Theorem 1.6. See Section 1 for its motivation.

*Proof of Theorem 1.6.* Since the statement of the theorem is local, we may assume X is affine and hence quasiprojective. Restricting to a sufficiently small neighborhood of a point with depth n - 1, all assumptions of the theorem stay valid and we may assume that all closed points of X have depth at least n - 1. We use the notation  $\mathscr{X} := X \times B$ . Let  $X = \bigcup_{i=1}^{r} X_i$  be the decomposition into irreducible components. Consider a pencil  $f : \mathscr{H} \to B = \mathbb{P}^1$  of hypersurfaces of X such that

- (1)  $\mathcal{H}_0$  contains the entire non-Gorenstein locus,
- (2)  $\emptyset \neq \mathcal{H}_0 \cap X_j \neq X_j$  for every  $1 \le j \le r$ ,
- (3)  $\mathcal{H}_{\infty}$  is a general hypersurface.

In particular then,

$$(\mathcal{H}_0 \setminus \mathcal{H}_\infty) \cap X_j \neq \emptyset$$
 for every  $1 \le j \le r$ . (7.2.e)

By definition of the pencil, if  $P \in \mathcal{H}_0 \setminus \mathcal{H}_\infty$ , then  $P \notin \mathcal{H}_b$  for any  $b \neq 0$ . Hence assumption (2) and (7.2.e) imply that for all  $b \in B$ , there is a point of  $X_j$  not contained in  $\mathcal{H}_b$ . Note now, that since X is  $S_1$ , by Lemma 4.2, so is  $\mathcal{X}$ . In particular, then all associated points of X and  $\mathcal{X}$  are general points of components. So, since none of the  $\mathcal{H}_b$  contains any of the  $X_j$ ,  $\mathcal{I}_{\mathcal{H},\mathcal{X}}$  and  $\mathcal{I}_{\mathcal{H}_b,\mathcal{X}_b}$  for every  $b \in B$  have nonzero divisor local generators and hence are line bundles. Then  $\mathcal{H}$  is flat over B by Lemma 3.5.

Define the loci

$$Z := \overline{\{x \in X \mid x \text{ is closed, depth } \mathbb{O}_{X,x} = n - 1\}},$$
$$W := \overline{\{x \in X \mid x \text{ is closed, depth } \mathbb{O}_{\mathcal{H}_{f(x)},x} = n - 2\}}$$

By construction and by Fact 3.2,  $W_0 = Z$  and  $W = (p^{-1}Z)_{red}$ , where  $p : \mathcal{H} \to X$  is the natural projection. Let Z' be an irreducible component of Z of the highest dimension. By the choice of  $\mathcal{H}_0$  and  $\mathcal{H}_\infty$ , we have  $Z' \subseteq \mathcal{H}_0$ , and  $Z' \not\subseteq \mathcal{H}_\infty$ . Furthermore,  $\mathcal{H}_\infty$  does not contain any of the irreducible components of Z. Hence, the general fiber of the map  $W \to B$  will have dimension at most dim Z' - 1. So,

*W* has dimension dim *Z'*. Hence,  $Z' \subseteq W_0$  is an irreducible component of *W*. In particular, by Theorem 1.3, the restriction morphism  $\omega_{\mathcal{H}/B}|_{\mathcal{H}_0} \to \omega_{\mathcal{H}_0}$  is not an isomorphism.

On the other hand assume that  $\omega_X$  is  $S_3$ . Since X is  $G_2$ ,  $\mathcal{H}$  and  $\mathcal{H}_b$  are  $G_1$  for every  $b \in B$ . In fact,  $\mathcal{H}$ , and  $\mathcal{H}_b$  for a general  $b \in B$ , are  $G_2$  also, but for  $\mathcal{H}_0$  only  $G_1$  can be guaranteed. Also, X is  $S_3$  by assumption and  $\mathcal{H}$  is  $S_3$  because of Lemma 4.2. Then  $\mathcal{H}$  and  $\mathcal{H}_b$  are  $S_2$  for every  $b \in B$  by Fact 3.2. That is, we may apply Lemma 6.1(1), which states that the restriction homomorphism  $\omega_{\mathcal{H}/B}|_{\mathcal{H}_0} \to \omega_{\mathcal{H}_0}$  is an isomorphism. This is a contradiction; hence  $\omega_X$  cannot be  $S_3$ .

#### 8. Stable reduction

In Construction 6.3, although the general fiber of  $\mathcal{H} \to B$  has mild, that is, log canonical, singularities,  $\mathcal{H}_0$  is very singular. The failure of base change for  $\omega_{\mathcal{H}/B}$  implies that by [Kollár and Kovács 2010, Theorem 7.9]  $\mathcal{H}_0$  is not Du Bois. By [ibid., Theorem 1.4], it is also not log canonical. In this section, we compute the stable limit of  $\mathcal{H} \to B$ . It is the limit at 0 of some stable family  $\mathcal{H}' \to \tilde{B}$ . This family has two important properties. First,  $\mathcal{H} \times_B \tilde{B}|_{\tilde{B}\setminus\{0\}} \cong \mathcal{H}'|_{\tilde{B}\setminus\{0\}}$  for a finite cover  $\phi : (\tilde{B}, 0) \to (B, 0)$  totally ramified at 0. Second,  $(\mathcal{H}')_0$  is log canonical, and hence by [ibid., Theorems 1.4 and 7.9],  $\omega_{\mathcal{H}'/\tilde{B}}$  commutes with base change. So,  $(\mathcal{H}')_0$  is the "right" limit of  $\mathcal{H}$ , and the incompatibility of Theorem 1.2 can be thought of as a consequence of using the wrong limit in Construction 6.3.

**Proposition 8.1.** Assuming that Y is smooth, the stable limit of Construction 6.3 is the d-fold cyclic cover of Y ramified exactly over D, with eigen-line bundles  $\mathbb{O}_Y(-i)$  for i = 0, ..., d - 1.

*Proof.* First, shrink *B* if necessary so that  $\infty \notin B$  and that every fiber apart from  $\mathcal{H}_0$  is log canonical. This is possible because the general fiber of  $\mathcal{H}$  is smooth by Bertini's theorem. Also, since we assumed that  $d \gg 0$ , the family  $\mathcal{H} \to B$  has canonically polarized fibers and hence is stable over  $B^* := B \setminus \{0\}$ . Define  $\mathcal{X} := X \times B$ .

The closed embedding  $Y \subseteq \mathbb{P}^{N-1}$  induces a natural closed embedding  $X \subseteq \mathbb{P}^N$ . This yields very ample line bundles  $\mathbb{O}_{\mathbb{P}^N}(1)$  and  $\mathbb{O}_X(1)$ . Then,  $\mathcal{H}$  is the zero locus of a section  $f_0 + tf_\infty$  of  $\mathbb{O}_{\mathcal{H}}(d) := p_1^* \mathbb{O}_X(d)$  for some  $f_0, f_\infty \in H^0(\mathbb{P}^N, \mathbb{O}_{\mathbb{P}^N}(d))$ , as explained in Section 2.

Choose a basis  $z_0, \ldots, z_N$  of  $H^0(\mathbb{P}^N, \mathbb{O}_{\mathbb{P}^N}(1))$ , such that  $z_0, \ldots, z_{N-1}$  form a basis of  $H^0(\mathbb{P}^{N-1}, \mathbb{O}_{\mathbb{P}^{N-1}}(1))$ . Then  $f_0$  and  $f_\infty$  correspond to degree d homogeneous polynomials in variables  $z_0, \ldots, z_{N-1}$  and  $z_0, \ldots, z_N$ , respectively. Also, the fact that  $P \notin \mathcal{H}_\infty$  implies that the coefficient of  $z_N^d$  in  $f_\infty$  is nonzero, say c.

Let  $\phi : (\widetilde{B}, 0) \to (B, 0)$  be the degree *d* cyclic cover branched only at 0, where it is totally ramified, and let *s* be a local parameter of  $\widetilde{B}$  at 0, such that  $s^d = t$ .

Consider the subscheme  $\mathscr{H}' \subseteq \mathscr{X} \times_B \widetilde{B} =: \mathscr{X}_{\phi}$  defined by

$$f_0(z_0,\ldots,z_{N-1})+s^d f_\infty(z_0,\ldots,z_{N-1},\frac{1}{s}z_N) \in H^0(\mathscr{X}_\phi,\mathbb{O}_{\mathscr{X}_\phi}(d)),$$

where  $\mathbb{O}_{\mathscr{X}_{\phi}}(d)$  is the pullback of  $\mathbb{O}_{\mathscr{X}}(d)$  to  $\mathscr{X}_{\phi}$ .

By the uniqueness of stable limit,  $\mathcal{H}'$  is a stable reduction of  $\mathcal{H}$  (that is, a stable family isomorphic generically to the pullback of  $\mathcal{H}$ ), if

(1)  $(\mathcal{H}')_0$  is a canonically polarized manifold, and

(2)  $\mathscr{H}'|_{\widetilde{B}^*} \cong \mathscr{H}_{\phi}|_{\widetilde{B}^*}$ , where  $\mathscr{H}_{\phi} := \mathscr{H} \times_B \widetilde{B}$  and  $\widetilde{B}^* := \widetilde{B} \setminus \{0\}$ .

To prove point (1), notice that  $(\mathcal{H}')_0$  is defined by the zero locus of *s* on  $\mathcal{H}'$  or equivalently by the zero locus of the following section of  $\mathbb{O}_X(d)$  on *X*:

$$f_0(z_0,\ldots,z_{N-1})+cz_N^d$$

Hence it is the cyclic cover of *Y* of degree *d* branched along *D* with eigensheaves  $\mathbb{O}_Y(-i)$  for  $0 \le i \le d-1$ . So first, it is smooth by [Kollár and Mori 1998, Lemma 2.51]. Second, since  $(\mathcal{H}')_0$  is contained in the smooth part of *X*, we have  $\omega_{(\mathcal{H}')_0} \cong \omega_X(d)|_{(\mathcal{H}')_0}$  by Proposition 3.6 and it is a line bundle. So, since  $d \gg 0$ ,  $(\mathcal{H}')_0$  is a canonically polarized manifold indeed.

To prove point (2), notice that the equation of  $\mathcal{H}_{\phi}$  in  $\mathcal{X}_{\phi}$  is

$$f_0(z_0, \ldots, z_{N-1}) + s^d f_\infty(z_0, \ldots, z_{N-1}, z_N) \in H^0(\mathscr{X}_\phi, \mathbb{O}_{\mathscr{X}_\phi}(d)).$$

Hence,  $\mathcal{H}_{\phi}|_{\widetilde{B}^*} \cong \mathcal{H}'|_{\widetilde{B}^*}$  via the isomorphism induced by the following automorphism of  $\mathbb{P}^N \times \widetilde{B}^*$ :

$$(z_0,\ldots,z_{N+1},z_N)\mapsto(z_0,\ldots,z_{N-1},s_{Z_N}).$$

We proved both points (1) and (2). Consequently,  $\mathcal{H}'$  is a stable reduction of  $\mathcal{H}$  indeed. Through the course of the proof of point (1), we also proved that  $(\mathcal{H}')_0$  is indeed the cyclic cover described in the statement of the proposition.

# Acknowledgements

The discussion contains ideas that originated from János Kollár and my advisor Sándor Kovács. I would like to thank both of them for their help. I would also like to thank Joseph Lipman for useful comments.

#### References

- [Bruns and Herzog 1993] W. Bruns and J. Herzog, *Cohen–Macaulay rings*, Cambridge Studies in Advanced Mathematics **39**, Cambridge University Press, 1993. MR 95h:13020 Zbl 0788.13005
- [Conrad 2000] B. Conrad, *Grothendieck duality and base change*, Lecture Notes in Mathematics **1750**, Springer, Berlin, 2000. MR 2002d:14025 Zbl 0992.14001

- [Grothendieck 1965] A. Grothendieck, "Éléments de géométrie algébrique, IV: Étude locale des schémas et des morphismes de schémas, II", *Inst. Hautes Études Sci. Publ. Math.* **24** (1965), 1–231. MR 33 #7330 Zbl 0135.39701
- [Hacon and Kovács 2010] C. D. Hacon and S. J. Kovács, *Classification of higher dimensional algebraic varieties*, Oberwolfach Seminars **41**, Birkhäuser, Basel, 2010. MR 2011f:14025 Zbl 1204. 14001
- [Harris 1992] J. Harris, *Algebraic geometry: A first course*, Graduate Texts in Mathematics **133**, Springer, New York, 1992. MR 93j:14001 Zbl 0779.14001
- [Hartshorne 1966] R. Hartshorne, *Residues and duality*, Lecture Notes in Mathematics **20**, Springer, Berlin, 1966. MR 36 #5145 Zbl 0212.26101
- [Hartshorne 1977] R. Hartshorne, *Algebraic geometry*, Graduate Texts in Mathematics **52**, Springer, New York, 1977. MR 57 #3116 Zbl 0367.14001
- [Hartshorne 1994] R. Hartshorne, "Generalized divisors on Gorenstein schemes", *K-Theory* **8**:3 (1994), 287–339. MR 95k:14008 Zbl 0826.14005
- [Hassett 2001] B. Hassett, "Stable limits of log surfaces and Cohen–Macaulay singularities", J. Algebra 242:1 (2001), 225–235. MR 2002f:14051 Zbl 1064.14037
- [Hassett and Kovács 2004] B. Hassett and S. J. Kovács, "Reflexive pull-backs and base extension", *J. Algebraic Geom.* **13**:2 (2004), 233–247. MR 2005b:14028 Zbl 1081.14017
- [Kollár 2008] J. Kollár, "Hulls and husks", preprint, 2008. arXiv 0805.0576
- [Kollár 2010] J. Kollár, "Moduli of varieties of general type", preprint, 2010. arXiv 1008.0621
- [Kollár 2011] J. Kollár, "A local version of the Kawamata–Viehweg vanishing theorem", *Pure Appl. Math. Q.* **7**:4 (2011), 1477–1494. MR 2918170 Zbl 06107785
- [Kollár and Kovács 2010] J. Kollár and S. J. Kovács, "Log canonical singularities are Du Bois", *J. Amer. Math. Soc.* **23**:3 (2010), 791–813. MR 2011m:14061 Zbl 1202.14003
- [Kollár and Mori 1998] J. Kollár and S. Mori, *Birational geometry of algebraic varieties*, Cambridge Tracts in Mathematics **134**, Cambridge University Press, 1998. MR 2000b:14018 Zbl 0926.14003
- [Kovács 2011] S. Kovács, "Irrational centers", *Pure Appl. Math. Q.* **7**:4 (2011), 1495–1515. MR 2918171 Zbl 06107786
- [Lipman and Hashimoto 2009] J. Lipman and M. Hashimoto, Foundations of Grothendieck duality for diagrams of schemes, Lecture Notes in Mathematics 1960, Springer, Berlin, 2009. MR 2010b:18001 Zbl 1163.14001
- [Matsumura 1989] H. Matsumura, *Commutative ring theory*, 2nd ed., Cambridge Studies in Advanced Mathematics **8**, Cambridge University Press, 1989. MR 90i:13001 Zbl 0666.13002
- [Neeman 1996] A. Neeman, "The Grothendieck duality theorem via Bousfield's techniques and Brown representability", *J. Amer. Math. Soc.* **9**:1 (1996), 205–236. MR 96c:18006 Zbl 0864.14008
- [Viehweg 1995] E. Viehweg, *Quasi-projective moduli for polarized manifolds*, Ergebnisse der Mathematik und ihrer Grenzgebiete (3) **30**, Springer, Berlin, 1995. MR 97j:14001 Zbl 0844.14004

Communicated by Ravi Vakil Received 2011-06-09 Revised 2012-01-26 Accepted 2012-03-09 pzs@princeton.edu Department of Mathematics, Princeton University, Fine Hall, Washington Road, Princeton, NJ 08540, United States www.math.princeton.edu/~pzs







# Two ways to degenerate the Jacobian are the same

Jesse Leo Kass

We provide sufficient conditions for the line bundle locus in a family of compact moduli spaces of pure sheaves to be isomorphic to the Néron model. The result applies to moduli spaces constructed by Eduardo Esteves and Carlos Simpson, extending results of Busonero, Caporaso, Melo, Oda, Seshadri, and Viviani.

# 1. Introduction

**1.1.** *Background.* This paper relates two different approaches to extending families of Jacobian varieties. Recall that if  $X_0$  is a smooth projective curve of genus g, then the associated Jacobian variety is a g-dimensional smooth projective variety  $J_0$  that can be described in two different ways: as the universal abelian variety that contains  $X_0$  (the Albanese variety), and as the moduli space of degree 0 line bundles on  $X_0$  (the Picard variety). If  $X_U \rightarrow U$  is a family of smooth, projective curves, then the Jacobians of the fibers fit together to form a family  $J_U \rightarrow U$ . In this paper, U will be an open subset of a smooth curve B (or, more generally, a Dedekind scheme), and we will be interested in extending  $J_U$  to a family over B. Corresponding to the two different ways of describing the Jacobian (Albanese vs. Picard) are two different approaches to extending the family  $J_U \rightarrow U$ .

Viewing the Jacobian as the Albanese variety, it is natural to try to extend  $J_U \rightarrow U$  by extending it to a family of group varieties over *B*. Néron [1964] showed that this can be done in a canonical way. He worked with an arbitrary family of abelian varieties  $A_U \rightarrow U$  and proved that there is a unique *B*-smooth group scheme N := N( $A_U$ )  $\rightarrow$  *B* extending  $A_U \rightarrow U$  which is universal with respect to a natural mapping property. This scheme is called the Néron model. Arithmetic geometry has seen the use of the Néron model in a number of important results, e.g., [Mazur 1972; 1977; Mazur and Wiles 1984; Gross 1990]. The Néron model of a Jacobian variety plays a particularly prominent role, and an alternative description

The author was partially supported by NSF grant DMS-0502170.

MSC2010: primary 14H40; secondary 14L15, 14D22, 14H10.

Keywords: Néron model, stable sheaf, compactified Jacobian, relative Picard functor.

of this scheme in terms of the relative Picard functor was given by Raynaud [1970]. We primarily work with Raynaud's description, which is recalled in Section 2.

An alternative approach, suggested by viewing the Jacobian as the Picard variety, is to extend  $J_U \rightarrow U$  as a family of moduli spaces of sheaves. This approach was first proposed by Mayer and Mumford [1964]. Typically, one first extends  $X_U \rightarrow U$  to a family of curves  $X \rightarrow B$  and then extends  $J_U$  to a family  $\bar{J} \rightarrow B$ with the property that the fiber over a point  $b \in B$  is a moduli space of sheaves on  $X_b$  parametrizing certain line bundles, together with their degenerations. In this paper, we show that the line bundle locus J in  $\bar{J}$  is canonically isomorphic to the Néron model for some schemes  $\bar{J}$  constructed in the literature.

To state this more precisely, we need to specify which schemes  $\overline{J}$  we consider. The problem of constructing such a family of moduli spaces has been studied by many mathematicians, and they have constructed a number of different compactifications; see for example [Ishida 1978; D'Souza 1979; Oda and Seshadri 1979; Altman and Kleiman 1980; Caporaso 1994; Simpson 1994; Pandharipande 1996; Jarvis 2000; Esteves 2001]. Many of the difficulties to performing such a construction arise from the fact that, when  $X_b$  is reducible, the degree 0 line bundles on a fiber  $X_b$  do not form a bounded family.

For simplicity, assume the residue field k(b) is algebraically closed and  $X_b$  is reduced with components labeled  $X_1, \ldots, X_n$ . Given a line bundle  $\mathcal{M}$  of degree 0 on  $X_b$ , the sequence  $(\deg(\mathcal{M}|_{X_1}), \ldots, \deg(\mathcal{M}|_{X_n}))$  is called the multidegree of  $\mathcal{M}$ . This sequence must sum to 0, but may otherwise be arbitrary, which implies unboundedness. A bounded family can be obtained by fixing the multidegree, and typically the scheme  $\overline{J}$  is defined so that it parametrizes (possibly coarsely) line bundles (and their degenerations) that satisfy a numerical condition on the multidegree. This paper focuses on constructions given by Simpson [1994] and by Esteves [2001], which we now describe in more detail.

For the Simpson moduli space, the numerical condition imposed on line bundles is slope semistability with respect to an auxiliary ample line bundle. This condition arises from the method of construction: the moduli space is constructed using geometric invariant theory (GIT), and slope stability corresponds to GIT stability. In general, the Simpson moduli space is a coarse moduli space in the sense that nonisomorphic sheaves may correspond to the same point of the space, but it contains an open subscheme (the stable locus) that is a fine moduli space, and we will work exclusively with this locus. Families of moduli spaces of semistable sheaves have been constructed for arbitrary families of projective schemes, but we will only be concerned with the case of families of curves.

The moduli spaces of Esteves parametrize sheaves that are  $\sigma$ -quasistable. Like slope stability,  $\sigma$ -quasistability is a numerical conditions on the multidegree, but it is defined in terms of an auxiliary vector bundle  $\mathscr{C}$  and a section  $\sigma$ , rather than an

ample line bundle. The moduli spaces are constructed for families over an arbitrary locally noetherian base, but strong conditions are required of the fibers: They must be geometrically reduced. The space is constructed as a closed subspace of a (nonnoetherian, nonseparated) algebraic space that was constructed in [Altman and Kleiman 1980]. For nodal curves, Melo and Viviani [2012] describe the relation between the Esteves moduli spaces and the Simpson moduli spaces. However, here we treat these moduli spaces separately.

For a discussion of the relation between these moduli spaces and other moduli spaces constructed in the literature, the reader is directed to [Alexeev 2004; Casalaina-Martin et al. 2011, Section 2]. The reader familiar with the work of Caporaso is warned of one potential point of confusion. In [Caporaso 1994], the compactified Jacobian associated to a stable curve X parametrizes pairs (Y, L)consisting of a line bundle L on a nodal curve Y stably equivalent to X that satisfies certain conditions. The line bundle locus J that we study corresponds to the locus parametrizing pairs (Y, L) with X = Y.

**1.2.** *Main result.* The main result of this paper relates the line bundle locus in a proper family of moduli spaces of sheaves to the Néron model of the Jacobian:

**Theorem 1.** Fix a Dedekind scheme B. Let  $f : X \to B$  be a family of geometrically reduced curves with regular total space X and smooth generic fiber  $X_{\eta}$ . Let  $J \subset \overline{J}$  the locus of line bundles in one of the following moduli spaces:

- the Esteves compactified Jacobian  $\bar{J}_{\mathscr{C}}^{\sigma}$ ;
- the Simpson compactified Jacobian  $\bar{J}^0_{\mathcal{L}}$  associated to an f-ample line bundle  $\mathcal{L}$  such that slope semistability coincides with slope stability.

# Then J is the Néron model of its generic fiber.

Theorem 1 is the combination of Corollaries 4.2 and 4.5, which themselves are consequences of Theorem 3.9. Theorem 3.9 is quite general, and we expect that it applies to many other fine moduli spaces of sheaves (but *not* coarse ones). In particular, Theorem 3.9 applies to families of curves with possibly nonreduced fibers, though then general results asserting the existence of a suitable moduli space are unknown (but see Section 4.3 for some simple examples).

The arithmetically inclined reader should note Theorem 1 and the results later in this paper do not place any hypotheses on the base Dedekind scheme B. In particular, we do not assume that the residue fields are perfect. This surprised the author initially as there is a body of work (e.g., [Liu et al. 2004; Raynaud 1970]) showing that various pathologies can arise when k(b) fails to be perfect.

Theorem 1 has interesting consequences for both the Néron model and the compactified Jacobian. One consequence of the theorem is that Néron models of Jacobians can often be constructed over high-dimensional bases. The Néron

model of an abelian variety is only defined over a (regular) 1-dimensional base *B*, but no such dimensional hypotheses are needed to apply the existence results from [Simpson 1994; Esteves 2001]. At the end of Section 4.3, we examine a family  $J \rightarrow \mathbb{P}^2$  over the plane with the property that a dense, open subset of  $\mathbb{P}^2$  is covered by lines *C* such that the restriction  $J_C$  of *J* is the Néron model of its generic fiber. Surprisingly, while the Néron models fit into a 2-dimensional family, their group structure does not.

Theorem 1 also has interesting consequences for the moduli spaces of Esteves and Simpson. Indeed, if  $f: X \to B$  is a family of curves satisfying the hypotheses of the theorem, then both the Esteves Jacobians  $J_{\mathscr{E}}^{\sigma}$  and the Simpson Jacobians  $J_{\mathscr{L}}^{0}$ (for  $\mathscr{L}$  as in the hypothesis) are independent of the particular polarizations, and every such Simpson Jacobian is isomorphic to every Esteves Jacobian. This is not immediate from the definitions. At the end of Section 4.1, we discuss this fact in greater detail and pose a related question.

**1.3.** *Past results.* Certain cases of Theorem 1 were already known. In his (unpublished) thesis, Simone Busonero [2008] established Theorem 1 for certain Esteves Jacobians. A different proof using similar techniques that extends the result to the Simpson moduli spaces is due to Melo and Viviani [2012, Theorem 3.1]. They prove Theorem 1 when the fibers of f are nodal and X is regular. We do not discuss the Caporaso universal compactified Jacobian here, but the relation between that scheme and the Néron model was described by Caporaso [2008a; 2008b; 2012, especially Theorem 2.9]. Earlier still, Oda and Seshadri related their  $\phi$ -semistable compactified Jacobians, also not discussed here, to Néron models [Oda and Seshadri 1979, Corollary 14.4]. In each of those papers, an important step in the proof is a combinatorial argument establishing that, for example, the natural map from the set of  $\sigma$ -quasistable multidegrees to the degree class group is a bijection.

The proof given here does not use any combinatorics, and the idea can be described succinctly. Consider the special case where  $B := \operatorname{Spec}(\mathbb{C}[[t]])$ , which is a strict henselian discrete valuation ring with algebraically closed residue field. There is a natural map  $J \to N$  to the Néron model coming from the universal property of N, and an application of Zariski's main theorem shows that this morphism is an open immersion. Thus the only issue is set-theoretic surjectivity. Because *B* is henselian, every point on the special fiber of N is the specialization of a section, so surjectivity is equivalent to the surjectivity of the map  $J(\mathbb{C}[[t]]) \to J(\operatorname{Frac} \mathbb{C}[[t]])$  that sends a section to its restriction to the generic fiber. A given point  $p \in J(\operatorname{Frac} \mathbb{C}[[t]])$  may be extended to a section  $\sigma \in \overline{J}(\mathbb{C}[[t]])$  of  $\overline{J}$  by the valuative criteria. As  $\overline{J}$  is a fine moduli space,  $\sigma$  corresponds to a family of rank 1, torsion-free sheaves, which in fact must be a family of line bundles because X is factorial. We may conclude that  $\sigma \in J(\mathbb{C}[[t]])$ , yielding the result.

**1.4.** *Questions.* It would be interesting to know when a Simpson Jacobian  $J_{\mathscr{L}}^0$  satisfying the hypotheses of Theorem 1 exists; that is, given a family  $f: X \to B$ , does there exist an ample line bundle  $\mathscr{L}$  such that every  $\mathscr{L}$ -slope semistable sheaf of degree 0 is stable? We briefly survey the literature on this question at the end of Section 4.2.

More generally, given a family  $f : X \rightarrow B$ , it would be desirable to have a description of the maximal subfunctors of the degree 0 relative Picard functor P<sup>0</sup> representable by a separated *B*-scheme. We discuss this question in Section 4.3, where we analyze the simple case of genus 1 curves.

**1.5.** *Organization.* We end this introduction with a few technical remarks about the paper. The moduli spaces of sheaves that we consider are moduli spaces of pure sheaves. On a curve, a coherent sheaf is pure if and only if it is Cohen–Macaulay. This condition is also equivalent to the condition of being torsion-free in the sense of elementary algebra when the curve is integral, and the term "torsion-free" is sometimes used in place of "pure".

The term "family of curves" only to refers to families with geometrically irreducible generic fibers. This is done to avoid notational complications concerning multidegrees. Families of curves are required to be proper, but not projective. A family of curves over a Dedekind scheme can fail to be projective (e.g., [Altman and Kleiman 1980, 8.10]), but projectivity is automatic if the local rings of the total space are factorial, which is the main case of interest. (See Proposition 4.1.)

We prove the main results for families over a base scheme *S* that is the spectrum of a strict henselian discrete valuation ring rather than a more general Dedekind scheme. Doing so lets us make sectionwise arguments because a smooth family of a henselian base admits many sections. Furthermore, this is not a real restriction: Results over a general Dedekind base can be deduced by passing to the strict henselization.

The body of the paper is organized as follows. In Section 2, we review Rayanud's construction of the maximal separated quotient. We then relate this scheme to a general moduli space of line bundles satisfying some axioms in Section 3. Finally, we describe some schemes that satisfy these axioms in the final section, Section 4.

# Conventions

**1.1.** The symbol  $X_T$  denotes the fiber product  $X \times_S T$ .

**1.2.** The letter *S* denotes the spectrum of a strict henselian discrete valuation ring with special point 0 and generic point  $\eta$ .

**1.3.** A curve over a field k is a proper k-scheme  $f_0: X_0 \rightarrow \text{Spec}(k)$  that is geometrically connected and of pure dimension 1.

**1.4.** If *B* is a scheme, then a family of curves over *B* is a proper, flat morphism  $f : X \to B$  whose fibers are curves and whose geometric generic fibers are irreducible.

**1.5.** If  $f: Y \to B$  is a finitely presented morphism, then we write  $Y^{sm} \subset Y$  for the smooth locus of f.

**1.6.** A coherent module  $I_0$  on a noetherian scheme  $X_0$  is rank 1 if the localization of  $I_0$  at x is isomorphic to  $\mathbb{C}_{X_0,x}$  for every generic point x.

**1.7.** A coherent module  $I_0$  on a noetherian scheme  $X_0$  is pure if the dimension of Supp( $I_0$ ) equals the dimension of Supp( $J_0$ ) for every nonzero subsheaf  $J_0$  of  $I_0$ .

**1.8.** If  $X_0 \to \text{Spec}(k)$  is proper, then the degree of a coherent  $\mathbb{O}_{X_0}$ -module  $\mathcal{F}$  is defined by  $\text{deg}(\mathcal{F}) := \chi(\mathcal{F}) - \chi(\mathbb{O}_X)$ .

#### 2. Raynaud's maximal separated quotient

We begin by reviewing Raynaud's construction of the Néron model of a Jacobian and, more generally, the maximal separated quotient of the relative Picard functor [Raynaud 1970]. Much of this material is also treated in [Bosch et al. 1990, Chapter 9].

Let *S* be a strict henselian discrete valuation ring with generic point  $\eta$  and special point 0. Given a family of curves  $f : X \to S$ , the *relative Picard functor* P of *f* is defined to be the étale sheaf P : *S*-Sch  $\to$  Grp associated to the functor

$$T \mapsto \operatorname{Pic}(X_T).$$
 (2-1)

Here  $Pic(X_T)$  is the set of isomorphism classes of line bundles on  $X_T$ . Rayanud actually defines P to be the associated fppf sheaf, but then observes that this is the same as the associated étale sheaf ([Raynaud 1970, 1.2]; see also [Kleiman 2005, Remark 9.2.11]). The fibers of P are representable by group schemes locally of finite type, and P itself is representable by an algebraic space if and only if *f* is cohomologically flat [Raynaud 1970, Theorem 5.2]. Regardless of its representability properties, P is locally finitely presented and formally *S*-smooth.

Inside of P, we may consider the functor E : *S*-Sch  $\rightarrow$  Grp that is defined to be the scheme-theoretic closure of the identity section. This is the fppf subsheaf of P generated by the elements  $g \in P(T)$ , where  $T \rightarrow S$  is flat and  $g_{\eta} \in P(T_{\eta})$  is the identity element. When P is a scheme, this coincides with the usual notion of closure. The representability properties of E are similar to those of P: The fibers of E are group schemes locally of finite type, and E is representable by an algebraic space precisely when *f* is cohomologically flat [Raynaud 1970, Proposition 5.2]. When representable,  $E \rightarrow S$  is an étale *S*-group space; in general, the generic fiber of E is the trivial group scheme, and the special fiber is a group scheme of dimension equal to  $h^0(\mathbb{O}_{X_0}) - h^0(\mathbb{O}_{X_{\eta}})$ .
When E is not the trivial *S*-group scheme, P does not satisfy the valuative criteria of separatedness. We can, however, form the *maximal separated quotient* Q : *S*-Sch  $\rightarrow$  Grp of P. By definition, this is the fppf quotient sheaf Q := P / E. The maximal separated quotient Q is always representable by a scheme that is *S*-smooth, separated, and locally of finite type [Raynaud 1970, Theorem 4.1.1, Proposition 8.0.1]. Rather than working directly with Q, we shall primarily work with the slightly smaller subfunctor Q<sup> $\tau$ </sup> : *S*-Sch  $\rightarrow$  Grp, which we now define.

Suppose generally that *B* is a scheme and G: S-Sch  $\rightarrow$  Grp is a *B*-group functor whose fibers are representable by group schemes locally of finite type. For every point  $b \in B$ , we may form the identity component  $G_b^o \subset G_b$  and the component group  $G_b/G_b^o$ . The subgroup functor  $G^\tau \subset G$  is defined to the subfunctor whose *T*valued points are elements  $g \in G(T)$  with the property that, for every  $t \in T$  mapping to  $b \in B$ , the element  $g_t \in G_b(k(t))$  maps to a torsion element of  $G_b/G_b^o(k(t))$ . If we instead require that  $g_t$  maps to the identity element, then we obtain the subgroup functor  $G^o \subset G$ . Let us examine these constructions when *B* equals *S* and *G* equals P or Q.

The functors  $P^o$  and  $P^\tau$  coincide, and this common functor is the étale sheaf associated to the assignment sending *T* to the set of isomorphism classes of line bundles on  $X_T$  that fiberwise have multidegree 0. From this description, it is easy to see that  $P^o = P^\tau \subset P$  is an open subfunctor. Another open subfunctor of P is the subfunctor parametrizing line bundles on  $X_T$  with fiberwise degree 0, which we denote by  $P^0$ . It is typographically difficult to distinguish between  $P^0$  and  $P^o$ , but we will not make use of  $P^o$  in this paper, so this should not cause confusion.

The functors  $Q^o$  and  $Q^\tau$  are different in general. They are, however, both open subfunctors of Q [Grothendieck 1966b, Theorem 1.1(i.i), Corollary 1.7]. In particular, they are both representable by smooth and separated *S*-group schemes that are locally of finite type. In fact, both schemes are of finite type over *S* as their fibers are easily seen to have a finite number of connected components. The condition that  $Q^\tau \subset Q$  is a closed subscheme is important, but slightly subtle. A characterization of this condition is given by [Raynaud 1970, Proposition 8.1.2(iii)]; one sufficient (but not necessary) condition for  $Q^\tau \subset Q$  to be closed is that the local rings of *X* are factorial.

The factoriality condition is also almost sufficient to ensure that  $Q^{\tau}$  is the Néron model of its generic fiber. Suppose that the generic fiber of f is smooth, so the generic fiber of  $Q^{\tau} \rightarrow S$  is an abelian variety, and thus it makes sense to speak of the Néron model  $N := N(Q_{\eta}^{\tau})$ . By the universal property, there is a unique morphism  $Q^{\tau} \rightarrow N$  that is the identity on the generic fiber. Theorem 8.1.4 of [Raynaud 1970] states that if the local rings of X are factorial, then  $Q^{\tau} \rightarrow N$  is an isomorphism in the cases that k(0) is perfect and that a certain invariant  $\delta$  is coprime to the residual characteristic. The proof uses the characterization of the Néron model in terms of the weak Néron mapping property. Recall that a *S*-scheme  $Y \rightarrow S$  is said to be a weak Néron model of its generic fiber if the natural map  $Y(S) \rightarrow Y(\eta)$  is bijective. If  $G \rightarrow S$  is a finite type *S*-group scheme whose generic fiber is an abelian variety, then *G* is the Néron model of its generic fiber if and only if it satisfies the weak Néron mapping property [Bosch et al. 1990, Section 7.1, Theorem 1].

#### 3. The main theorem

Here we derive the main results for families over a strict henselian discrete valuation ring *S* with generic point  $\eta$  and special point 0. Specifically, we provide sufficient conditions for the maximal separated quotient  $Q^{\tau}$  of the Picard functor to be the Néron model and we relate  $Q^{\tau}$  to a fine moduli space of line bundles that satisfies certain axioms. These moduli spaces are, by definition, subfunctors of a (large) functor that we now define.

**Definition 3.1.** If *T* is a *S*-scheme, then we define  $\text{Sheaf}(X_T)$  to be the set of isomorphism classes of  $\mathbb{O}_T$ -flat, finitely presented  $\mathbb{O}_{X_T}$ -modules  $\mathscr{I}$  on  $X_T$  that are fiberwise pure, rank 1, and of degree 0.

The functor  $Sh = Sh_{X/S} : S$ -Sch  $\rightarrow$  Sets is defined to be the étale sheaf associated to the functor

$$T \mapsto \operatorname{Sheaf}(X_T).$$
 (3-1)

There is a tautological transformation  $P^0 \rightarrow Sh$  that realizes  $P^0$  as a subfunctor of Sh.

# **Lemma 3.2.** The subfunctor $P^0 \subset Sh$ is open.

*Proof.* Given a *S*-scheme *T* and a morphism  $g: T \to Sh$ , we must show that  $T \times_{Sh} P^0$  is representable by a scheme and that  $T \times_{Sh} P^0 \to T$  is an open immersion. Thus, let *g* be given.

By definition, there exists an étale surjection  $p : T' \to T$  and a sheaf  $\mathscr{I}' \in$ Sheaf( $X_{T'}$ ) that represents  $g \circ p : T' \to Sh$ . Consider the subset  $U' \subset T'$  of points  $t \in T'$  with the property that the restriction of  $\mathscr{I}$  to the fiber  $X_t$  is a line bundle. This locus is open by [Altman and Kleiman 1980, Lemma 5.12(a)], and one may easily show that U' represents  $T' \times_{Sh} P^0$ . A descent argument establishes the analogous property for the image U of U' under  $T' \to T$ . This completes the proof.

A remark about topologies: We work with the étale sheaf associated to (3-1), but one could instead work with the associated fppf sheaf. When f is projective, it is a theorem of Altman and Kleiman [1980, Theorem 7.4] that the subfunctor of Sh parametrizing simple sheaves can be represented by a quasiseparated, locally finitely presented *S*-algebraic space, and hence is an fppf sheaf. We do not know if Sh is an fppf sheaf in general. Here Sh is just used as a tool for keeping track of

representable functors, and certainly any representable subfunctor of Sh is an fppf sheaf.

One reason for working with the étale topology instead of the fppf topology is that it makes the following fact easy to prove.

**Fact 3.3.** The natural map  $\text{Sheaf}(X) \rightarrow \text{Sh}(S)$  is surjective.

*Proof.* Let  $g \in Sh(S)$  be given. By definition, there is an étale morphism  $S' \to S$  and an element  $\mathscr{I}' \in Sheaf(X_{S'})$  that maps to  $g_{S'} \in Sh(S')$ . But *S* is strict henselian, so  $S' \to S$  may be taken to be an isomorphism  $S \to S$  [Grothendieck 1967, Proposition 18.8.1(c)], in which case the result is obvious.

The following two facts about separably closed fields are standard, but they will be used so frequently that it is convenient to record them.

**Fact 3.4.** If k(0) is a separably closed field and  $f_0: Y_0 \rightarrow \text{Spec}(k(0))$  is smooth of relative dimension *n*, then the closed points of  $Y_0$  with residue field k(0) are dense.

*Proof.* This is [Bosch et al. 1990, Corollary 13]. The scheme  $Y_0$  can be covered by affine opens  $U_0$  that admit an étale morphism  $p: U_0 \to \mathbb{A}_{k(0)}^n$ . Certainly, the closed points with residue field k(0) are dense in the image of p. If  $v_0 \in \mathbb{A}_{k(0)}^n$  is one such point, then  $p^{-1}(v_0)$  is a finite, étale k(0)-algebra, and hence a disjoint union of closed points defined over k(0). Density follows.

Fact 3.4 is typically used in conjunction with the following fact to assert that a smooth morphism has many sections.

**Fact 3.5.** Let  $Y \to S$  be a smooth morphism over strict henselian discrete valuation ring. Then  $Y(S) \to Y(k(0))$  is surjective.

*Proof.* This is [Grothendieck 1967, Corollary 17.17.3], or [Bosch et al. 1990, Proposition 14]. If U and X' are as in the statement of the former, then we must have U = S and  $X' \rightarrow U$  may be taken to be an isomorphism (again, by [Grothendieck 1967, Proposition 18.8.1(c)]).

We now prove the main results of the paper.

**Proposition 3.6.** Let  $f : X \to S$  be a family of curves and  $J \subset P^0$  a subfunctor such that the generic fibers  $J_\eta = P_\eta^0$  coincide. Assume J is represented by a smooth, finitely presented S-scheme.

If J is S-separated, then  $J \to Q$  is an open immersion. Furthermore, the image is contained in  $Q^{\tau}$  provided  $Q^{\tau} \subset Q$  is closed (e.g., the local rings of X are factorial).

*Proof.* This is an application of Zariski's main theorem. We begin by showing that the induced map  $J \rightarrow Q$  is injective on closed points. It is enough to verify this after extending base *S* so that k(0) is algebraically closed. Thus, we will temporarily assume k := k(0) is algebraically closed and work with *k*-valued points instead of closed points. Given  $q \in Q(k)$ , there is nothing to show when the fiber over *q* is empty. If nonempty, pick  $p \in J(k)$  mapping to *q*. We may invoke Fact 3.5 to assert that there exists a section  $\sigma \in J(S)$  with  $\sigma(0) = p$ .

The fiber of  $P \rightarrow Q$  over q is the set of elements of the form p + e with  $e \in E(k)$ or, equivalently, the elements of  $(\sigma + E)(k)$  [Raynaud 1970, Corollary 4.1.2]. Restricting to J, we see that the fiber of q under  $J \rightarrow Q$  is the set of k-valued points of  $(\sigma + E) \cap J$ . But  $(\sigma + E) \cap J$  is the scheme-theoretic closure of  $\sigma$  in J (by [Grothendieck 1965, 2.8.5]), which is just the image of  $\sigma$  by separatedness. In particular, the preimage of q under  $J \rightarrow Q$  must be the singleton set  $\{p\}$ . This proves that the map is injective on closed points. We now return to the case where S is a henselian discrete valuation ring (so k(0) is no longer assumed to be algebraically closed).

It follows that the set-theoretic fibers of  $J \rightarrow Q$  are finite sets. Indeed, if  $Z \subset J$  is the locus of points  $x \in J$  with the property that x lies in a positive dimensional fiber, then Z is closed by Chevalley's theorem [Grothendieck 1965, 13.1.3]. Furthermore, Z is contained in the special fiber  $J_0$  and contains no closed points. This is only possible if Z is the empty scheme. In other words, the set-theoretic fibers of  $J \rightarrow Q$  are 0-dimensional, and hence finite (by [Grothendieck 1964, 14.1.9]).

It follows immediately from Zariski's main theorem [Grothendieck 1961, 4.4.9] that  $J \rightarrow Q$  is an open immersion. This proves the first part of the theorem. To complete the proof, observe that flatness implies that the generic fiber of  $J_{\eta}$  is dense in J [Grothendieck 1965, 2.8.5]. In particular, J is contained in the closure of  $J_{\eta}$  in Q. The generic fiber of J coincides with the generic fiber of  $Q^{\tau}$ , so the closure of this common scheme is contained in  $Q^{\tau}$  when  $Q^{\tau} \subset Q$  is closed. This completes the proof.

**Remark 3.7.** In Proposition 3.6, we do *not* assume that  $J \subset P^0$  is an open subfunctor, but this condition holds in most cases of interest. When open, J is automatically formally smooth and locally of finite presentation. Thus, the key hypothesis in the proposition is that J is represented by a *S*-separated scheme. A similar remark holds for Theorem 3.9; there the key hypotheses are that  $\overline{J}$  satisfies the valuative criteria of properness and that J is representable. Indeed, we do not even need to assume that  $\overline{J}$  is representable.

Under stronger assumptions, we can actually show that the natural map  $J \to Q^{\tau}$  is an isomorphism. The essential point is to prove that J satisfies the weak Néron mapping property. When J can be embedded in a S-proper moduli space  $\overline{J}$ , this

property holds provided that the local rings of X are factorial. The content of this claim is that a line bundle on the generic fiber can specialize only to a line bundle on the special fiber. By localizing, the claim is equivalent to the following lemma, which is based on a proof from [Altman and Kleiman 1979, p. 27 after Step XII].

**Lemma 3.8.** Suppose that  $(R, \pi)$  is a discrete valuation ring and  $R \to \mathbb{O}$  a local, flat algebra extension with  $\mathbb{O}$  noetherian. Let M be a R-flat, finite  $\mathbb{O}$ -module with the property that  $M[\pi^{-1}]$  is free of rank 1 and  $M/\pi M$  is a rank 1, pure module. If  $\mathbb{O}$  is factorial, then M is free of rank 1.

*Proof.* We can certainly assume  $\mathbb{O}$  is not the zero ring. To ease notation, we write  $\overline{M} := M/\pi M$  and  $\overline{\mathbb{O}} := \mathbb{O}/\pi \mathbb{O}$ . It is enough to prove that M is isomorphic to a height 1 ideal. Indeed, such an ideal is principal by the factoriality assumption.

We argue by first showing that M is isomorphic to an ideal of  $\mathbb{O}$ . Let  $\bar{\mathfrak{p}}_1, \ldots, \bar{\mathfrak{p}}_n$  be the minimal primes of  $\overline{\mathbb{O}}$  and  $\mathfrak{p}_1, \ldots, \mathfrak{p}_n$  the corresponding primes of  $\mathbb{O}$ . We have assumed that the stalk  $\overline{M} \otimes k(\bar{\mathfrak{p}}_i)$  is 1-dimensional. This stalk coincides with the stalk  $M \otimes k(\mathfrak{p}_i)$ , so we can conclude that the localization  $M_{\mathfrak{p}_i}$  is free of rank 1 for  $i = 1, \ldots, n$ .

We can also conclude that the same holds for the localizations of the dual module  $M^{\vee} := \text{Hom}(M, \mathbb{O})$ . An application of the prime avoidance lemma shows that there exists an element  $\phi \in M^{\vee}$  that maps to a generator of  $M_{\mathfrak{p}_i}^{\vee}$  for all *i*. We will show that  $\phi : M \to \mathbb{O}$  realizes *M* as a *R*-flat family of ideals (i.e.,  $\phi$  is injective with *R*-flat cokernel).

It is enough to show that the reduction  $\overline{\phi}: \overline{M} \to \overline{\mathbb{O}}$  is injective. An element of the kernel of this map is also in the kernel of the composition

$$\overline{M} \to \bigoplus \overline{M}_{\bar{\mathfrak{p}}_i} \to \bigoplus \overline{\mathbb{O}}_{\bar{\mathfrak{p}}_i}.$$

The kernel of the leftmost map is a submodule whose support does not contain any of the primes  $\bar{p}_i$ , and thus must be zero by pureness. Furthermore, the rightmost map is an isomorphism by construction. This proves injectivity.

Consider the ideal  $I[\pi^{-1}]$  given by the image of  $\phi[\pi^{-1}]: M[\pi^{-1}] \to \mathbb{O}[\pi^{-1}]$ . This is a principal ideal, and hence is either the unit ideal or an ideal of height at most 1 (Hauptidealsatz!). By flatness, the same is true of the image I of  $\phi$ . In fact, I cannot be a height zero ideal: The only such prime is the zero ideal, which does not satisfy the hypotheses. Thus, I is either the unit ideal or a height 1 ideal. In either case, I must be principal, and the proof is complete.

We record the factorial condition as a hypothesis.

**Hypothesis 1.** We say a family of curves  $f : X \to B$  over a Dedekind scheme satisfies Hypothesis 1 if the generic fiber  $X_{\eta}$  is smooth and the local rings of  $X_S$  are factorial for every strict henselization  $S \to B$ .

Hypothesis 1 is satisfied when X is regular and  $X_{\eta}$  is smooth. We now prove the main theorem of this paper.

**Theorem 3.9.** Let  $f : X \to S$  be a family of curves and  $\overline{J}$  a subfunctor of Sh such that the generic fibers  $\overline{J}_{\eta} = \operatorname{Sh}_{\eta}$  coincide. Assume the line bundle locus  $J \subset \overline{J}$  is represented by a smooth and finitely presented S-scheme.

If  $\overline{J}$  satisfies the valuative criteria of properness and f satisfies Hypothesis 1, then  $Q^{\tau}$  is the Néron model and

$$J \subset Q^{\tau} = N$$

is an open subscheme that contains all the k(0)-valued points of  $Q^{\tau}$ . Furthermore,

$$J = Q^{\tau} = N$$

provided one of the following conditions hold:

- (1) k(0) is algebraically closed;
- (2) J is stabilized by the identity component  $Q^{0}$ .

*Proof.* By Proposition 3.6, the natural map  $J \to Q$  is an open immersion with image contained in  $Q^{\tau}$ . Using this fact, we can prove that  $Q^{\tau}$  is the Néron model of its generic fiber. Indeed, it is enough to prove that  $Q^{\tau}$  satisfies the weak Néron mapping property. The open subscheme  $J \subset Q^{\tau}$ , in fact, already satisfies this property. Let  $\sigma_{\eta} \in Q^{\tau}(\eta) = J(\eta)$  be given. By properness, we can extend  $\sigma_{\eta}$  to a section  $\sigma \in \overline{J}(S)$ , and this element can be represented by a family  $\mathcal{I}$  of pure, rank 1 sheaves (by Fact 3.3). But every such family is a family of line bundles (Lemma 3.8), and hence  $\sigma$  lies in  $J(S) \subset \overline{J}(S)$ . In other words, J satisfies the weak Néron mapping property.

The weak Néron mapping property of J also implies that the image of J contains all the k(0)-valued points of  $Q^{\tau}$ . Indeed, every k(0)-valued point of  $Q^{\tau}$  is the specialization of a section by Fact 3.5. If k(0) is algebraically closed, then we have shown that J contains every k(0)-valued point of  $Q^{\tau}$ , hence every closed point. Thus,  $J = Q^{\tau}$ , and there is nothing more to show.

Let us now turn our attention to the case where k(0) is only separably closed, but J is stabilized by  $Q^0$ . Our goal is to show  $J = Q^{\tau}$ , and to show this, we pass to the special fiber  $J_0 \rightarrow Q_0^{\tau}$  and argue with points. Let x be a  $\bar{k}(0)$ -valued point of  $Q^{\tau}$ , where  $\bar{k}(0)$  is the algebraic closure of the residue field. By density (Fact 3.4), there exists a  $\bar{k}(0)$ -valued point y in the image of  $J_0 \rightarrow Q_0^{\tau}$  that lies in the same connected component as x. We have x = y + (x - y), which expresses x as the sum of a point of  $Q_0^0$  and a point of  $J_0$ . The point x must lie in  $J_0$  by assumption. This shows that the image of J contains all of  $Q^{\tau}$ , completing the proof.

390

**Remark 3.10.** The hypothesis that J is stabilized by the identity component  $Q^o$  is perhaps unexpected, but it is often satisfied in practice. The moduli space  $\overline{J}$  is typically constructed by imposing numerical conditions on the multidegree of a sheaf, and the multidegree is invariant under the action of  $Q^o$  (because the action is given by tensoring with a multidegree 0 line bundle).

In the next section, we will show that certain moduli spaces constructed in the literature satisfy the hypotheses of Theorem 3.9. There are, however, families of curves  $f: X \to S$  with factorial local rings  $\mathbb{O}_{X,x}$  such that there does not exist a *S*-scheme  $\overline{J}$  satisfying the conditions of the theorem. Indeed, the family  $f: X \to S$  in [Raynaud 1970, Example 9.2.3] is a family of genus 1 curves such that the local rings of *X* are factorial (even regular), but the natural map  $Q^{\tau} \to N$  is not an isomorphism. In particular, no  $\overline{J}$  satisfying the hypotheses of Theorem 3.9 can exist.

## 4. Applications

Here we apply Theorem 3.9 to some families of moduli spaces from the literature and then deduce consequences. The two moduli spaces that we are interested in are the Esteves moduli space of quasistable sheaves (Section 4.1) and the Simpson moduli space of slope stable sheaves (Section 4.2). In Section 4.3, we discuss the special case of families of genus 1 curves, where suitable moduli spaces can be constructed explicitly.

The moduli spaces we study are associated to a relatively projective family of curves. We are primarily interested in families over a Dedekind scheme with locally factorial total space, in which case projectivity is automatic. This fact is a consequence of the generalized Chevalley Conjecture when the Dedekind scheme is defined over a field, but we do not know a reference. For completeness, we prove:

**Proposition 4.1.** Let  $f : X \to B$  be a family of curves over a Dedekind scheme. If the local rings of X are factorial, then f is projective.

*Proof.* This proof was explained to the author by Steven Kleiman. Fix a closed point  $b_0 \in B$ . Given any component  $F \subset X_{b_0}$ , I claim that we can find a line bundle  $\mathscr{L}$  on X that has nonnegative degree on every component of every fiber and strictly positive degree on F.

Pick a closed point  $x \in F$  and an open affine neighborhood  $U \subset X$  of that point. By the prime avoidance lemma, we can find a regular function  $r \in H^0(U, \mathbb{O}_X)$  that does not vanish on any component of  $X_{b_0}$  but does vanish at x. Pick a component D of the closure of  $\{r = 0\} \subset U$  in X. Then D is a Cartier divisor (by the factoriality assumption) that does not contain any component of any fiber  $X_b$  (by construction). Furthermore, *D* has nontrivial intersection with *F*. The associated line bundle  $\mathscr{L} := \mathbb{O}_X(D)$  has the desired positivity property.

Now construct one such line bundle for every irreducible component F of  $X_{b_0}$  and define  $\mathcal{M}$  to be their tensor product. The line bundle  $\mathcal{M}$  is nef on every fiber and ample on  $X_{b_0}$ . Ampleness is an open condition, so  $\mathcal{M}$  is in fact ample on all but finitely many fibers of f. After repeating the construction for each such fiber and forming the tensor product, we have constructed a f-relatively ample line bundle on X. This completes the proof.

We now turn our attention to the moduli spaces.

**4.1.** *Esteves Jacobians.* We first discuss the Esteves moduli space of quasistable sheaves. This moduli space fits very naturally into the framework of the previous section.

Suppose *B* be a locally noetherian scheme and  $f: X \to B$  a projective family of curves whose fibers are geometrically reduced. Quasistability is defined in terms of a section  $\sigma: B \to X^{\text{sm}}$  and a vector bundle  $\mathscr{C}$  on *X* with fiberwise integral slope  $\deg(\mathscr{C}_b)/\operatorname{rank}(\mathscr{C}_b)$ , which we assume is constant as a function of  $b \in B$ . Given  $\sigma$  and  $\mathscr{C}$ ,  $\sigma$ -quasistability is a numerical condition on the multidegree of a rank 1, torsion-free sheaf of degree

$$d(\mathscr{E}) = d := -\chi(\mathbb{O}_{X_b}) - \deg(\mathscr{E}_b) / \operatorname{rank}(\mathscr{E}_b).$$

For the definitions (which we will not use), we direct the reader to [Esteves 2001, p. 3051] (for a single sheaf) and [ibid., p. 3054] (for a family). The basic existence theorem is [ibid., Theorem A on p. 3047], which is proved in [ibid., Section 4]). It states that if Sheaf<sup> $\sigma$ </sup><sub> $\mathcal{E}$ </sub> : *S*-Sch  $\rightarrow$  Sets is the functor defined by setting Sheaf<sup> $\sigma$ </sup><sub> $\mathcal{E}$ </sub>(*T*) equal to the set of isomorphism classes of  $\mathbb{O}_T$ -flat, finitely presented  $\mathbb{O}_{X_T}$ -modules that are fiberwise  $\sigma$ -quasistable, then there is a *B*-proper algebraic space  $\bar{J}^{\sigma}_{\mathcal{E}} \rightarrow B$  of finite type that represents the étale sheaf associated to Sheaf<sup> $\sigma$ </sup><sub> $\mathcal{E}$ </sub>.

Strictly speaking, our definition differs from the one given in [ibid.] in two ways. First, Esteves does not work with isomorphism classes of sheaves but rather with equivalence classes under the relation given by identifying two sheaves  $\mathscr{I}_1$  and  $\mathscr{I}_2$  on  $X_T$  when  $\mathscr{I}_1$  is isomorphic to  $\mathscr{I}_2 \otimes f^*(\mathscr{L})$  for some line bundle  $\mathscr{L}$  on T. Zariski locally on T, the sheaves  $\mathscr{I}_1$  and  $\mathscr{I}_2$  are isomorphic, and it follows that the étale sheaf associated to Sheaf $\mathscr{G}_{\mathscr{E}}$  is canonically isomorphic to the sheaf considered by Esteves. Second, Esteves only defines his functor as a functor from locally noetherian schemes to sets. However, the functor Sheaf $\mathscr{G}_{\mathscr{E}}$  and its associated étale sheaf are easily seen to be locally finitely presented. It follows that  $\overline{J}_{\mathscr{E}}^{\sigma}$  represents the étale sheaf associated to Sheaf $\mathscr{G}_{\varepsilon}$ , rather than just the restriction of this sheaf to locally noetherian schemes. If f satisfies stronger conditions, then the space  $\bar{J}_{\mathscr{C}}^{\sigma}$  is actually a scheme. This is the content of [Esteves 2001, Theorem B, p. 3048], proved on [ibid., p. 3086]. The theorem states that if there exist sections  $\sigma_1, \ldots, \sigma_n : B \to X^{\text{sm}}$  of f with the property that every irreducible component of a fiber  $X_b$  is geometrically integral and contains one of the points  $\sigma_1(b), \ldots, \sigma_n(b)$ , then  $\bar{J}_{\mathscr{C}}^{\sigma}$  is a scheme.

In the special case where B = S is a strict henselian discrete valuation ring with generic point  $\eta$  and special point 0, the hypotheses of Theorem B are automatically satisfied. Indeed, the locus of k(0)-valued points is dense in the smooth locus  $X_0^{\text{sm}}$  (Fact 3.4), which in turn is dense in  $X_0$  as  $X_0$  is geometrically reduced. We may conclude that the irreducible components of  $X_0$  are geometrically integral. Finally, every k(0)-valued point of  $X_0$  extends to a section  $\sigma : S \to X$  (Fact 3.5), so the hypotheses of Theorem B are certainly satisfied.

We call  $\bar{J}^{\sigma}_{\&}$  the *Esteves compactified Jacobian*. Inside of the Esteves compactified Jacobian, we can consider the open subscheme parametrizing line bundles. This scheme is called the *Esteves Jacobian* and denoted by  $J^{\sigma}_{\&}$ . While the scheme  $\bar{J}^{\sigma}_{\&}$  parametrizes sheaves, it is not naturally a subfunctor of Sh because it does not parametrize degree 0 sheaves. We can, however, define a natural transformation  $\bar{J}^{\sigma}_{\&} \rightarrow$  Sh by the rule

$$\mathscr{I} \mapsto \mathscr{I}(-d \cdot \sigma)$$

Both Proposition 3.6 and Theorem 3.9 apply to  $\bar{J}_{\&}^{\sigma}$ .

**Corollary 4.2.** Fix a Dedekind scheme B. Let  $f : X \to B$  be a projective family of geometrically reduced curves. Let  $\sigma : B \to X^{sm}$  be a section and  $\mathscr{E}$  a vector bundle on X with fiberwise integral slope.

Then the natural map  $J_{\mathscr{C}}^{\sigma} \to Q$  is an open immersion.

Assume further that f satisfies Hypothesis 1. Then  $J_{\mathcal{C}}^{\sigma} = Q^{\tau}$ , and this scheme is the Néron model.

*Proof.* By localizing, we can assume that B = S is a strict henselian discrete valuation ring, in which case we are reduced to proving that the hypotheses of Proposition 3.6 and Theorem 3.9 hold. The scheme  $J_{\mathscr{C}}^{\sigma}$  is easily seen to be formally *S*-smooth. Indeed,  $\sigma$ -quasistability is a condition on fibers, so the formal smoothness of P<sup>0</sup> implies the formal smoothness of  $J_{\mathscr{C}}^{\sigma}$ . The remaining hypotheses of Proposition 3.6 are explicitly assumed, so we can deduce the first part of the theorem. To complete the proof, it is enough to show that  $J_{\mathscr{C}}^{\sigma}$  is stabilized by Q<sup>0</sup>. But the action of Q<sup>0</sup> on  $J_{\mathscr{C}}^{\sigma}$  is given by the tensor product against a multidegree 0 line bundle, so this action preserves multidegree and hence  $\sigma$ -quasistability.

Corollary 4.2 implies that  $J_{\mathscr{C}}^{\sigma}$  is a scheme with (unique) *B*-group scheme structure that extends the group scheme structure of the generic fiber. It is not immediate from the definition that  $J_{\mathscr{C}}^{\sigma}$  admits such structure, and Example 4.9 shows that the

group structure is special to the case of families over a 1-dimensional base. The result also implies uniqueness results for the Esteves Jacobian; if  $\sigma' : B \to X^{\text{sm}}$  is a second section and  $\mathscr{C}'$  a second vector bundle on X, then  $J_{\mathscr{C}'}^{\sigma'}$  is canonically isomorphic to  $J_{\mathscr{C}}^{\sigma}$ . In the next section, we will define the Simpson stable Jacobian  $J_{\mathscr{L}}^0(X)$ , and this scheme is also isomorphic to  $J_{\mathscr{C}}^{\sigma}$  provided every slope semistable sheaf is stable. It would be interesting to know if these isomorphisms extend to the compactifications. Important results along these lines can be found in [Melo and Viviani 2012; Esteves 2009], but many basic question remain unanswered. Currently, there is no example of a curve  $X_0 \to \text{Spec}(k)$  such that two Esteves compactified Jacobians associated to  $X_0$  are nonisomorphic.

**4.2.** *Simpson Jacobians.* The hypotheses to Proposition 3.6 and Theorem 3.9 are satisfied by certain moduli spaces of stable sheaves, which we call Simpson Jacobians. Here we recall Simpson's construction, along with later work of Langer and Maruyama, and then apply results from Section 3. We restrict our attention to families of reduced curves (but see Remark 4.4, and the discussion preceding Example 4.9).

We work over a scheme *B* that is finitely generated over a universally Japanese ring *R* (e.g.,  $R = \mathbb{C}, \mathbb{F}_p, \mathbb{Z}, ...$ ). Let  $f: X \to B$  a family of curves with *f*-relatively ample line bundle  $\mathcal{L}$ , and assume the Euler–Poincaré characteristics  $\chi(\mathbb{O}_{X_b})$  and  $\chi(\mathcal{L}_b)$  are constant as functions of the base *B*. Set  $P_d$  equal to the polynomial

$$P_d(t) := \deg(\mathscr{L}_b) \cdot t + d + \chi, \tag{4-1}$$

where  $\chi$  is the Euler–Poincaré characteristic of a fiber of f and deg( $\mathscr{L}_b$ ) is the degree of the restriction of  $\mathscr{L}$  to a fiber. This is the Hilbert polynomial of a degree d line bundle.

Given this data, Simpson constructed an associated moduli space in the case that  $R = \mathbb{C}$ . The Simpson moduli space  $M(\mathbb{O}_X, P_d)$  parametrizes slope semistable sheaves with Hilbert polynomial  $P_d$ . (See [Simpson 1994, pp. 54–56] for the definition of semistability). To be precise, define  $M^{\sharp}(\mathbb{O}_X, P_d)$  to be the functor whose *T*-valued points are isomorphism classes of  $\mathbb{O}_T$ -flat, finitely presented  $\mathbb{O}_{X_T}$ modules whose fibers are  $\mathscr{L}$ -slope semistable sheaves with Hilbert polynomial  $P_d$ . The main existence result [Simpson 1994, Theorem 1.21] asserts that there is a projective scheme  $M(\mathbb{O}_X, P_d)$  that corepresents  $M^{\sharp}(\mathbb{O}_X, P_d)$ . Inside of  $M(\mathbb{O}_X, P_d)$ , we may consider the open subscheme  $M^{st}(\mathbb{O}_X, P_d)$  parametrizing  $\mathscr{L}$ -slope stable sheaves. The stable locus is a fine moduli space: Its  $\mathbb{C}$ -valued points are in natural bijection with the isomorphism classes of  $\mathscr{L}$ -slope stable sheaves with Hilbert polynomial  $P_d$ , and étale locally on  $M^{st}(\mathbb{O}_X, P_d)$ , the product  $X \times_B M^{st}(\mathbb{O}_X, P_d)$ admits a universal family of sheaves. The reader may check that these conditions are equivalent to the condition that  $M^{st}(\mathbb{O}_X, P_d)$  represents the étale sheaf associated the functor parametrizing stable sheaves. While Simpson only considers the case  $R = \mathbb{C}$ , later work of Langer [2004a, Theorem 4.1; 2004b, Theorem 0.2] and Maruyama [1996] extends these results to the case where R is an arbitrary universally Japanese ring.

Let us now specialize to the case where *B* is a Dedekind scheme. When *f* has reducible fibers,  $M^{st}(\mathbb{O}_X, P_d)$  may contain points corresponding to sheaves that are not rank 1; see [López-Martín 2005, Example 2.2]. This is potentially a major source of confusion: The term "rank" is used in a different way in [Simpson 1994], and the sheaves parametrized by  $M^{st}(\mathbb{O}_X, P_d)$  are rank 1 in Simpson's sense but not necessary in the sense used here.

We avoid these sheaves. Define the Simpson stable Jacobian  $J_{\mathcal{L}}^d$  of degree d to be the locus of stable line bundles in  $M^{st}(\mathbb{O}_X, P_d)$  (which is an open subscheme by [Altman and Kleiman 1980, Lemma 5.12(a)]). We define the Simpson stable compactified Jacobian  $\overline{J}_{\mathcal{L}}^d$  to be the subset of the stable locus  $M^{st}(\mathbb{O}_X, P_d)$  that corresponds to pure, rank 1 sheaves. (Warning: The compactified Jacobian is a *B*-proper scheme when every semistable pure sheaf with Hilbert polynomial  $P_d$  is stable but not in general!)

When the fibers of  $X \to B$  are geometrically reduced, a minor modification of the proof of [Pandharipande 1996, Lemma 8.1.1] shows that the subset  $\bar{J}_{\mathcal{L}}^d \subset M^{\text{st}}(\mathbb{O}_X, P_d)$  is closed and open, and hence has a natural scheme structure:

**Lemma 4.3.** Assume the fibers of  $f : X \to B$  are geometrically reduced. Then the subset  $\overline{J}_{\mathcal{L}}^d$  is closed and open in  $M^{st}(\mathbb{O}_X, P_d)$ .

*Proof.* The main point to prove is that a 1-parameter family of line bundles cannot specialize to a pure sheaf that fails to have rank 1, and this is shown by examining the leading term of the Hilbert polynomial. To begin, we may cover  $M^{st}(\mathbb{O}_X, P_d)$  by étale morphisms  $M \to M^{st}(\mathbb{O}_X, P_d)$  with the property that a universal family  $\mathscr{I}_{uni.}$  on  $M \times_B X$  exists. It is enough to verify the claim after passing from  $M^{st}(\mathbb{O}_X, P_d)$  to an arbitrary such scheme, and so for the remainder of the proof we work with M in place of  $M^{st}(\mathbb{O}_X, P_d)$ . We will also abuse notation by denoting the pullback of  $\overline{J}_{\mathscr{L}}^d$  under  $M \to M^{st}(\mathbb{O}_X, P_d)$  by the same symbol  $\overline{J}_{\mathscr{L}}^d$ .

We first need to check that  $\overline{J}_{\mathscr{L}}^d \subset M$  is constructible, so that we can make use of the valuative criteria. Given  $m \in M$  mapping to  $b \in B$ , the condition that the fiber  $I_m$  is rank 1 is just the condition that the restriction of  $I_m$  to  $X_b^{\text{sm}}$  is a line bundle. Constructibility thus follows from [Grothendieck 1966a, 9.4.7] applied to  $M \times_B X^{\text{sm}} \to M$ .

To finish, it is enough to prove that  $\bar{J}_{\mathcal{L}}^d$  is closed under specialization and generalization. Thus, we pass from M to a discrete valuation ring T mapping to M. If

 $\mathcal{I}$  is the sheaf on  $X_T$  given by the pullback of the universal family, then we need to show that the generic fiber of  $I_\eta$  is rank 1 if and only if the special fiber  $I_0$  is.

To prove this, we turn our attention to the Hilbert polynomial  $P_d$  of a fiber of I. This polynomial is defined so that the leading term is  $\deg(\mathscr{L}_b)$ , and we can express this number in terms of components of a fiber of  $X_T \to T$  as follows. If x is generic point of the special fiber  $X_0$ , then we define  $\deg_x(\mathscr{L}_0)$  to be the degree of the restriction of  $\mathscr{L}_0$  to the irreducible component corresponding to x. (Give the component the reduced subscheme structure.) For any generic point y of  $X_\eta$ , we define  $\deg_y(\mathscr{L}_\eta)$  in the analogous manner. If  $x_1, \ldots, x_n$  are all the generic points of  $X_0$  and  $y_1, \ldots, y_m$  all the generic points of  $X_\eta$ , then we have

$$deg(\mathscr{L}_b) = deg_{\chi_1}(\mathscr{L}_0) + \dots + deg_{\chi_n}(\mathscr{L}_0)$$
$$= deg_{\chi_1}(\mathscr{L}_\eta) + \dots + deg_{\chi_n}(\mathscr{L}_\eta)$$

by, say, [Altman and Kleiman 1979, 2.5.1]. The terms  $\deg_{x_i}(\mathcal{L}_0)$  and  $\deg_{y_j}(\mathcal{L}_\eta)$  in the equation above are each strictly positive as  $\mathcal{L}$  is relatively ample.

We can also express deg( $\mathcal{L}_b$ ) in terms of the generic rank of a fiber of *I*. Using [Altman and Kleiman 1979, 2.5.1] again, we have

$$\deg(\mathscr{L}_b) = \deg_{x_1}(\mathscr{L}_0) \cdot \ell_{x_1}(I_0) + \dots + \deg_{x_n}(\mathscr{L}_0) \cdot \ell_{x_n}(I_0)$$
  
= 
$$\deg_{y_1}(\mathscr{L}_\eta) \cdot \ell_{y_1}(I_\eta) + \dots + \deg_{y_m}(\mathscr{L}_\eta) \cdot \ell_{y_m}(I_\eta).$$

Here  $\ell_{x_i}(I_0)$  denotes the length of the localization of  $I_0$  at  $x_i$  and similarly for  $\ell_{y_j}(I_\eta)$ . The fibers of  $X_T \to T$  are reduced, so such a length is equal to the minimal number of generators. In particular, these numbers are upper semicontinuous. In other words, if  $y_j$  specializes to  $x_i$ , then we have  $\ell_{y_j}(I_\eta) \le \ell_{x_i}(I_0)$  (by Nakayama's lemma).

The desired result now follows. Suppose first that  $I_0$  is rank 1. Then we have  $\ell_{y_i}(I_\eta) \le 1$  for all *i* by semicontinuity. If some inequality was strict, say  $\ell_{y_1}(I_\eta) = 0$ , then we would have

$$deg(\mathscr{L}_b) = deg_{y_1}(\mathscr{L}_\eta) + deg_{y_2}(\mathscr{L}_\eta) + \dots + deg_{y_m}(\mathscr{L}_\eta)$$

$$> deg_{y_2}(\mathscr{L}_\eta) + \dots + deg_{y_m}(\mathscr{L}_\eta)$$

$$\geq deg_{y_1}(\mathscr{L}_\eta) \cdot \ell_{y_1}(I_\eta) + \dots + deg_{y_m}(\mathscr{L}_\eta) \cdot \ell_{y_m}(I_\eta)$$

$$= deg(\mathscr{L}_b).$$

This is absurd! Thus, we must have  $\ell_{y_i}(I_\eta) = 1$  for all  $y_i$  and  $I_\eta$  is rank 1. Similar reasoning shows that if  $I_\eta$  is rank 1, then  $I_0$  is rank 1.

**Remark 4.4.** The hypothesis that the fibers of f are geometrically reduced is necessary. Indeed, the moduli space  $M^{st}(\mathbb{O}_X, P_d)$  was described in [Chen and Kass 2011] in the case that X is a nonreduced curve whose reduced subscheme  $X_{red}$ 

is smooth and whose nilradical  $\mathcal{N}$  is square-zero (i.e., X is a ribbon). Using that description it is easy to produce examples where  $\bar{J}_{\mathcal{L}}^d \subset M^{\text{st}}(\mathbb{O}_X, P_d)$  is not closed (e.g., take d equal to 0, X to have even genus, and  $X_{\text{red}}$  to have genus 1). The points of the complement in the closure correspond to stable rank 2 vector bundles on  $X_{\text{red}}$ .

We now apply Proposition 3.6 and Theorem 3.9 to the Simpson Jacobians.

**Corollary 4.5.** Fix a Dedekind scheme B that is finitely generated over a universally Japanese ring. Let  $f : X \to B$  be a family of geometrically reduced curves. Let  $\mathcal{L}$  be f-relatively ample line bundle.

Then the natural map  $J^0_{\mathcal{L}}(X) \to Q^{\tau}$  is an open immersion. Assume further that both of the following conditions hold:

- Every *L*-slope semistable rank 1, torsion free sheaf of degree 0 is *L*-slope stable.
- f satisfies Hypothesis 1.

Then  $J^0_{\varphi}(X) = Q^{\tau}$ , and this scheme is the Néron model.

*Proof.* The local existence of a universal family [Simpson 1994, Theorem 2.1(4)] implies that there is a natural transformation  $\overline{J}_{\mathcal{L}}(X) \to Sh$  with the property that  $J_{\mathcal{L}}(X)$  is the preimage of  $P^0 \subset Sh$ . Furthermore, the slope stability condition is a fiberwise condition, so a modification of the argument given in Corollary 4.2 completes the proof.

**Remark 4.6.** A minor generalization of Corollary 4.5 can be obtained by allowing for moduli spaces of degree *d* lines bundles, with  $d \neq 0$ . If we are given a line bundle  $\mathcal{M}$  on *X* with fiberwise degree *d*, then there is an associated map  $J_{\mathcal{L}}^d(X) \to \mathbb{Q}$  that extends the map on the generic fiber given by tensoring with  $\mathcal{M}^{-1}$ . With only notational changes the previous corollary generalizes to a statement about this map.

Corollary 4.5 is, of course, only of interest when there exists an  $\mathcal{L}$  such that  $\mathcal{L}$ -slope stability coincides with  $\mathcal{L}$ -slope semistability. Thus, we ask, When does such an  $\mathcal{L}$  exist? A comprehensive discussion of this question would require a digression on stability conditions, so we limit ourselves to reviewing known results about a single curve  $X_0$  over an algebraically closed field. When  $X_0$  is integral, the stability condition is vacuous, so every ample  $\mathcal{L}_0$  has the desired property. If  $X_0$  is reducible of genus  $g \neq 1$  with only nodes as singularities, then Melo and Viviani have proven the existence of a suitable  $\mathcal{L}_0$  [2012, Proposition 6.4]. Stability conditions on reduced, genus 1 curves were analyzed by López-Martín [2005]. She exhibits curves  $X_0$  with the property that there is no  $\mathcal{L}_0$  such that every  $\mathcal{L}_0$ -slope semistable, pure, rank 1 sheaf degree 0 is stable, but a suitable  $\mathcal{L}_0$  always exists if one considers sheaves of fixed degree  $d \neq 0$ . Finally, stability conditions for a

ribbon were analyzed in [Chen and Kass 2011]. On a ribbon, the stability condition is independent of  $\mathcal{L}_0$ , and for rational ribbons, slope stability coincides with slope semistability precisely when the genus g is even. It would be desirable to have a general result asserting (non)existence of a suitable  $\mathcal{L}_0$ .

**4.3.** *Genus* 1 *curves.* The Néron model of the Jacobian of a genus 1 curve can be quite complicated (see for example [Liu et al. 2004]), but these complications do not arise if the family admits a section. Suppose *B* is a Dedekind scheme and  $f : X \to B$  is a family of curves such that the total space *X* is regular and the generic fiber  $X_{\eta}$  is smooth. If  $\sigma : B \to X^{\text{sm}}$  is a section contained in the smooth locus, then there is a canonical identification of the smooth locus  $X^{\text{sm}}$  with the Néron model N of the Jacobian of  $X_{\eta}$ . Here we examine how this fact fits into the preceding framework.

**Definition 4.7.** Let  $f : X \to B$  be a family of genus 1 curves over a Dedekind scheme and  $\sigma : B \to X^{sm}$  a section contained in the smooth locus. We define a sheaf  $\mathcal{I}_{uni.}$  on  $X \times_B X$  by the formula

$$\mathscr{I}_{\text{uni.}} := \mathscr{I}_{\Delta}(\pi_1^*(\sigma) + \pi_2^*(\sigma)). \tag{4-2}$$

Here  $\mathcal{I}_{\Delta}$  is the ideal sheaf of the diagonal, and  $\pi_1, \pi_2 : X \times_B X \to X$  are the projection maps.

The sheaf  $\mathcal{I}_{uni.}$  determines a transformation  $X \to Sh$  that realizes X as a moduli space of sheaves over itself. Proposition 3.6 and Theorem 3.9 apply to this moduli space.

**Corollary 4.8.** Fix a Dedekind scheme B. Let  $f : X \to B$  be a family of genus 1 curves. Let  $\sigma : B \to X^{sm}$  be a section.

Then the natural map  $X^{sm} \rightarrow Q$  is an open immersion.

Assume further that f satisfies Hypothesis 1. Then  $X^{sm} = Q^{\tau}$ , and this scheme is the Néron model.

Let us consider the special case where *B* is a discrete valuation ring, *X* is a minimal regular surface, and the residue field k(0) is algebraically closed. The possibilities for the special fiber  $X_0$  are given by the Kodaria–Néron classification ([Kodaira 1960; Néron 1964]; see [Silverman 1994, pp. 353–354] for a recent exposition). The reduced curves appearing in the classification are the reduction types  $I_n$ , II, III, and IV. In these cases, one may show that the induced morphism  $X \rightarrow$  Sh identifies X with the Esteves compactified Jacobian  $\bar{J}_0^{\sigma}$ .

In every remaining case (reduction type  $I_n^*$ ,  $II^*$ ,  $III^*$ , or  $IV^*$ ) the morphism  $X \to$ Sh is not a special case of the fine moduli spaces discussed in the previous two sections. Indeed, the special fiber  $X_0$  is nonreduced, so the Esteves Jacobian of Xis not defined. In Section 4.2, we reviewed Simpson's moduli space  $M^{st}(\mathbb{O}_X, P_d)$  of stable sheaves, but the image of  $X \to Sh$  cannot be described as a closed subscheme of that space. The reason is that slope stable sheaves are simple, but some fibers of  $\mathscr{I}_{uni.}$  are not simple. Specifically, if  $p_0 \in X_0$  lies on the intersection of two components, then the fiber of  $\mathscr{I}_{uni}$  of  $p_0$  fails to be simple. This can be seen as follows. This fiber is the sheaf  $\mathscr{I}_{p_0}(+\sigma(0))$ , where  $\mathscr{I}_{p_0}$  is the ideal of  $p_0$ . If  $\nu : X'_0 \to X_0$  is the blow-up of  $X_0$  at  $p_0$ , then one may show that  $H^0(X'_0, \mathbb{O}_{X'_0})$  is canonically isomorphic to the endomorphism ring of  $\mathscr{I}_{p_0}(+\sigma(0))$ . An inspection of the Kodaria–Néron table shows that  $X'_0$  is disconnected, so  $H^0(X'_0, \mathbb{O}_{X'_0})$  does not equal k(0) and  $\mathscr{I}_{p_0}(+\sigma(0))$  is not simple.

Corollary 4.8 provides a partial answer to a question posed in the introduction: What are the maximal subfunctors J of  $P^0$  represented by a separated *B*-scheme? When X is, say, regular, a strong result one could hope for is that there is always a subfunctor  $\overline{J}$  of Sh satisfying the hypotheses of Theorem 3.9. The line bundle locus  $J \subset \overline{J}$  in such a functor has the property that  $J \to Q^r$  is an isomorphism, and hence J is maximal. Corollary 4.8 shows that such a  $\overline{J}$  exists when  $f : X \to S$  is a family of genus 1 curves that f admits a section. Similarly, the Esteves compactified Jacobian represents a suitable subfunctor when f has geometrically reduced fibers and admits a section. In general, however, the hope is too optimistic: Raynaud's family, mentioned at the end of Section 3, has that property that no such  $\overline{J}$  can exist.

The question of describing maximal subfunctors J is most interesting when f has nonreduced fibers. The slope stable line bundles form a subfunctor  $J \subset P^0$  represented by a S-separated scheme, but our discussion of genus 1 families together with Remark 4.4 suggest that we should consider other methods for constructing a suitable J when f has nonreduced fibers.

In a different direction, one nice property of the moduli spaces described by Corollary 4.8 is that their geometry is very simple. We use these spaces to provide an example showing that a family  $J \rightarrow B$  of Esteves Jacobians over a regular 2-dimensional base may not have group scheme structure.

**Example 4.9** (Néron models in 2-dimensional families). We will construct a 2-dimensional family  $f: X \to B$  of plane cubics and an associated Esteves Jacobian  $J \to B$  with the property that the group law on the locus  $J_U \to U$  parametrizing nonsingular cubics does not extend over all of B. Furthermore, the family is constructed so that a dense open subset of B is covered by nonsingular curves C with the property that the restriction  $X_C$  of X to C is regular, so  $J_C$  is the Néron model of its generic fiber (and in particular admits group scheme structure that extends the group scheme structure over  $C \cap U$ ). Thus, the Néron models fit into a 2-dimensional family, but their group scheme structure does not.

The idea is as follows. The family we construct has a reducible element  $X_{b_0} \rightarrow b_0$  with the property that, for every nonsingular curve  $C \subset B$  passing through  $b_0$ 

such that  $X_C$  is regular, the restriction of the Esteves Jacobian  $J_C$  is the Néron model of its generic fiber. The fiber  $J_{b_0}$  inherits a group law from this Néron model, and we show explicitly that this group law depends on the particular choice of *C*. But, if the group law on  $J_U$  extended to *J*, then all the different group laws on  $J_{b_0}$  coming from the different curves *C* would be the restriction of one common group law on *J*, which is absurd. We now construct the family.

We work over an algebraically closed field k. The family  $X \to B$  will be a net of plane cubics. Let  $X_0 \subset \mathbb{P}^2_k$  be a reducible plane cubics that is the union of a smooth quadric  $Q_0$  and a line  $L_0$  that meet in two distinct points. (See Figure 1.) Fix two general points  $p_1, p_2 \in L_0(k)$  on the line and one general point  $q_1 \in Q_0(k)$  on the quadric. Say that  $F \in H^0(\mathbb{P}^2_k, \mathbb{O}(3))$  is an equation for  $X_0$  and  $G, H \in H^0(\mathbb{P}^2_k, \mathbb{O}(3))$ are two general cubic equations that vanish on all of the points  $p_1, p_2, q_1$ . We will work with the net  $V := \langle F, G, H \rangle \subset H^0(\mathbb{P}^2_k, \mathbb{O}(3))$  and the associated family of curves

$$X := \{ (p, [r, s, t]) : r \cdot F(p) + s \cdot G(p) + t \cdot H(p) = 0 \}$$

$$\subset \mathbb{P}_k^2 \times \mathbb{P}_k^2.$$
(4-3)

There are two obvious morphisms  $e, f: X \to \mathbb{P}_k^2$  given by the two projections. If we set  $B := \mathbb{P}_k^2$  equal to the plane, then the second morphism  $f: X \to B$  realizes X as a family of genus 1 curves with  $X_0 = f^{-1}(b_0)$ , where  $b_0 := [1, 0, 0]$ . Corresponding to the points  $p_1, p_2, q_1 \in X_0(k)$  are three section  $\sigma_1, \sigma_2, \tau_1 : B \to X^{\text{sm}}$ , which lie in the smooth locus by the generality assumption.



**Figure 1.** The pencil  $X_C$ .

Another application of the generality assumption shows that the fibers of f are reduced, so we can form the Esteves Jacobian  $J := J_{\mathscr{C}}^{\sigma_1}$ , where  $\mathscr{C} = \mathbb{O}_X$ . The quasistability condition on a line bundle  $\mathscr{L}_0$  on  $X_0$  is the condition that the bidegree  $(\deg(\mathscr{L}_{L_0}), \deg(\mathscr{L}_{Q_0}))$  equals (0, 0) or (1, -1). Now we assume  $J \to B$  is a group scheme and derive a contradiction.

Suppose that we are given a general line  $C \subset B$  in the plane that contains  $b_0$ . Such a line corresponds to a 2-dimensional linear subspace of the form  $W := \langle F, G_C \rangle \subset V$  for some  $G_C \in V$ . Invoking generality again, the base locus

$$\{p \in \mathbb{P}_k^2 : F(p) = G_C(p) = 0\}$$
(4-4)

consists of 9 distinct points. The first projection map  $e: X \to \mathbb{P}_k^2$  realizes  $X_C$  as the blow-up of the plane at these points, so  $X_C$  is regular, and thus  $J_C$  is the Néron model of its generic fiber. We now study the group of sections of  $J_C \to C$ .

The base locus (4-4) includes the points  $p_1$ ,  $p_2$ ,  $q_1$ . In addition to the points  $p_1$ ,  $p_2$ , a unique third point of the base locus must lie on the line  $L_0$ . Let us label that point  $p_C$  and write  $\sigma_C : C \to X_C$  for the corresponding section.

Now consider the following line bundles on  $X_C$ :

$$\begin{aligned} \mathscr{L}_1 &:= \mathbb{O}(\sigma_1 - \tau_1), \qquad \mathscr{L}_C &:= \mathbb{O}(\sigma_C - \tau_1), \\ \mathscr{L}_2 &:= \mathbb{O}(\sigma_2 - \tau_1), \qquad \mathcal{M} &:= \mathbb{O}(1) \otimes \mathbb{O}(-3 \cdot \tau_1). \end{aligned}$$

These lines bundles are all  $\sigma_1$ -quasistable. If we let  $g_1, g_2, g_C, h \in J_C(C)$  respectively correspond to  $\mathcal{L}_1, \mathcal{L}_2, \mathcal{L}_C, \mathcal{M}$ , then I claim we have

$$g_1 + g_2 + g_C = h. (4-5)$$

Indeed, it is enough to verify the claim after passing to the generic fiber of  $J_C \to C$ , where the equation is just the statement that the points  $p_1$ ,  $p_2$ ,  $p_C$  all lie on a line (the line  $L_0$ ). Now suppose that  $J \to \mathbb{P}_k^2$  admits a group law extending the group law of the generic fiber. Then the specialization of (4-5) to  $J_{b_0}$  holds for all C simultaneously. In particular, the isomorphism class of the line bundle  $\mathbb{O}_{X_{b_0}}(p_C - q_1)$ is independent of the particular line  $C \subset \mathbb{P}_k^2$  chosen. But this is absurd: For distinct general lines  $C_1$ ,  $C_2$ , the points  $p_{C_1}$  and  $p_{C_2}$  (and hence the associated line bundles) are distinct! This completes our discussion of this example.

This example is particularly interesting in light of [Oda and Seshadri 1979]. The authors of that paper consider the case of a family of nodal curves  $f : X \rightarrow B$  over a suitable Dedekind scheme with the property that X is regular. Let  $J_{\eta}$  be the Jacobian of the generic fiber. Given a closed point  $0 \in B$ , they prove that the special fiber N<sub>0</sub> of the Néron model of  $J_{\eta}$  depends only on the curve  $X_0$  and not the particular family f [ibid., Corollary 14.4]. This result must be interpreted with

care: In our example, the group law depends on a particular choice of family, but any two such group laws define isomorphic group schemes.

#### 5. Acknowledgements

The results of this paper are a part of my thesis. I would like to thank my advisor Joe Harris for his invaluable help. The thesis was carefully read by Filippo Viviani and Margarida Melo, who where very helpful in writing this paper, and I thank them. I thank Steven Kleiman for explaining the proof of Proposition 4.1, and Valery Alexeev for a useful discussion of stability conditions. I also thank the anonymous referee, Bryden Cais, Lucia Caporaso, Eduardo Esteves, Robert Lazarsfeld, and Dino Lorenzini for feedback about early drafts of this paper. Finally, I thank Bhargav Bhatt, Matthew Satriano, and Karen Smith for conversations that were helpful in clarifying technical aspects of this paper.

#### References

- [Alexeev 2004] V. Alexeev, "Compactified Jacobians and Torelli map", *Publ. Res. Inst. Math. Sci.* **40**:4 (2004), 1241–1265. MR 2006a:14016 Zbl 1079.14019
- [Altman and Kleiman 1979] A. B. Altman and S. L. Kleiman, "Compactifying the Picard scheme, II", *Amer. J. Math.* **101**:1 (1979), 10–41. MR 81f:14025b Zbl 0427.14016
- [Altman and Kleiman 1980] A. B. Altman and S. L. Kleiman, "Compactifying the Picard scheme", *Adv. in Math.* **35**:1 (1980), 50–112. MR 81f:14025a Zbl 0427.14015
- [Bosch et al. 1990] S. Bosch, W. Lütkebohmert, and M. Raynaud, *Néron models*, Ergeb. Math. Grenzgeb. (3) **21**, Springer, Berlin, 1990. MR 91i:14034 Zbl 0705.14001
- [Busonero 2008] S. Busonero, *Compactified Picard schemes and Abel maps for singular curves*, thesis, Università di Roma La Sapienz, 2008.
- [Caporaso 1994] L. Caporaso, "A compactification of the universal Picard variety over the moduli space of stable curves", J. Amer. Math. Soc. 7:3 (1994), 589–660. MR 95d:14014 ZbI 0827.14014
- [Caporaso 2008a] L. Caporaso, "Compactified Jacobians, Abel maps and theta divisors", pp. 1–23 in *Curves and abelian varieties* (Athens, GA, 2007), edited by V. Alexeev et al., Contemp. Math. **465**, Amer. Math. Soc., Providence, RI, 2008. MR 2010b:14088 Zbl 1152.14027
- [Caporaso 2008b] L. Caporaso, "Néron models and compactified Picard schemes over the moduli stack of stable curves", *Amer. J. Math.* **130**:1 (2008), 1–47. MR 2009j:14030 Zbl 1155.14023
- [Caporaso 2012] L. Caporaso, "Compactified Jacobians of Néron type", *Rend. Lincei Mat. Appl.* **23**:2 (2012), 213–227. Zbl 1244.14031
- [Casalaina-Martin et al. 2011] S. Casalaina-Martin, J. L. Kass, and F. Viviani, "The local structure of compactified Jacobians: Deformation theory", preprint, 2011. arXiv 1107.4166v1
- [Chen and Kass 2011] D. Chen and J. L. Kass, "Moduli of generalized line bundles on a ribbon", preprint, 2011. arXiv 1106.5441v1
- [D'Souza 1979] C. D'Souza, "Compactification of generalised Jacobians", *Proc. Indian Acad. Sci. Sect. A Math. Sci.* **88**:5 (1979), 419–457. MR 81h:14004 Zbl 0442.14016
- [Esteves 2001] E. Esteves, "Compactifying the relative Jacobian over families of reduced curves", *Trans. Amer. Math. Soc.* **353**:8 (2001), 3045–3095. MR 2003b:14036 Zbl 0974.14009

- [Esteves 2009] E. Esteves, "Compactified Jacobians of curves with spine decompositions", *Geom. Dedicata* **139** (2009), 167–181. MR 2010d:14041 Zbl 1171.14018
- [Gross 1990] B. H. Gross, "A tameness criterion for Galois representations associated to modular forms (mod *p*)", *Duke Math. J.* **61**:2 (1990), 445–517. MR 91i:11060 Zbl 0743.11030
- [Grothendieck 1961] A. Grothendieck, "Éléments de géométrie algébrique, III: Étude cohomologique des faisceaux cohérents, I", *Inst. Hautes Études Sci. Publ. Math.* **11** (1961), 5–167. MR 36 #177c Zbl 0118.36206
- [Grothendieck 1964] A. Grothendieck, "Éléments de géométrie algébrique, IV: Étude locale des schémas et des morphismes de schémas, I", *Inst. Hautes Études Sci. Publ. Math.* **20** (1964), 5–259. MR 30 #3885 Zbl 0136.15901
- [Grothendieck 1965] A. Grothendieck, "Éléments de géométrie algébrique, IV: Étude locale des schémas et des morphismes de schémas, II", *Inst. Hautes Études Sci. Publ. Math.* **24** (1965), 5–231. MR 33 #7330 Zbl 0135.39701
- [Grothendieck 1966a] A. Grothendieck, "Éléments de géométrie algébrique, IV: Étude locale des schémas et des morphismes de schémas, III", *Inst. Hautes Études Sci. Publ. Math.* **28** (1966), 5–255. MR 36 #178 Zbl 0144.19904
- [Grothendieck 1966b] A. Grothendieck, "Technique de descente et théorèmes d'existence en géométrie algébrique, VI, Les schémas de Picard: propriétés générales", exposé 236 in Séminaire Bourbaki 1961–1962, W. A. Benjamin, Amsterdam, 1966. Reprinted as pp. 221–243 in Séminaire Bourbaki 7, Soc. Math. France, Paris, 1995. MR 1610872 Zbl 0238.14015
- [Grothendieck 1967] A. Grothendieck, "Éléments de géométrie algébrique, IV: Étude locale des schémas et des morphismes de schémas, IV", *Inst. Hautes Études Sci. Publ. Math.* **32** (1967), 5–361. MR 39 #220 Zbl 0153.22301
- [Ishida 1978] M.-N. Ishida, "Compactifications of a family of generalized Jacobian varieties", pp. 503–524 in *Proceedings of the International Symposium on Algebraic Geometry* (Kyoto, 1977), edited by M. Nagata, Kinokuniya Book Store, Tokyo, 1978. MR 81h:14025 Zbl 0415.14015
- [Jarvis 2000] T. J. Jarvis, "Compactification of the universal Picard over the moduli of stable curves", *Math. Z.* 235:1 (2000), 123–149. MR 2001m:14040 Zbl 0980.14020
- [Kleiman 2005] S. L. Kleiman, "The Picard scheme", pp. 235–321 in Fundamental algebraic geometry: Grothendieck's FGA explained (Trieste, 2003), Math. Surveys Monogr. 123, Amer. Math. Soc., Providence, RI, 2005. MR 2223410 Zbl 1085.14001
- [Kodaira 1960] K. Kodaira, "On compact analytic surfaces", pp. 121–135 in *Analytic functions* (Princeton, NJ, 1958), Princeton Mathematical Series **24**, Princeton University Press, 1960. MR 25 #3939 Zbl 0137.17401
- [Langer 2004a] A. Langer, "Moduli spaces of sheaves in mixed characteristic", *Duke Math. J.* **124**:3 (2004), 571–586. MR 2005g:14082 Zbl 1086.14036
- [Langer 2004b] A. Langer, "Semistable sheaves in positive characteristic", *Ann. of Math.* (2) **159**:1 (2004), 251–276. MR 2005c:14021 Zbl 1080.14014
- [Liu et al. 2004] Q. Liu, D. Lorenzini, and M. Raynaud, "Néron models, Lie algebras, and reduction of curves of genus one", *Invent. Math.* 157:3 (2004), 455–518. MR 2005m:14039 Zbl 1060.14037
- [López-Martín 2005] A. C. López-Martín, "Simpson Jacobians of reducible curves", *J. Reine Angew. Math.* **582** (2005), 1–39. MR 2006d:14032 Zbl 1078.14033
- [Maruyama 1996] M. Maruyama, "Construction of moduli spaces of stable sheaves via Simpson's idea", pp. 147–187 in *Moduli of vector bundles* (Kyoto, 1994), edited by M. Maruyama, Lecture Notes in Pure and Appl. Math. **179**, Dekker, New York, 1996. MR 97h:14020 Zbl 0885.14005

- [Mayer and Mumford 1964] A. Mayer and D. Mumford, "Further comments on boundary points", Lecture notes, Summer Institute on Algebraic Geometry (Woods Hole, MA), Amer. Math. Soc. Summer Institute, 1964, Available at http://www.jmilne.org/math/Documents/index.html.
- [Mazur 1972] B. Mazur, "Rational points of abelian varieties with values in towers of number fields", *Invent. Math.* **18** (1972), 183–266. MR 56 #3020 Zbl 0245.14015
- [Mazur 1977] B. Mazur, "Modular curves and the Eisenstein ideal", *Inst. Hautes Études Sci. Publ. Math.* **47** (1977), 33–186. MR 80c:14015 Zbl 0394.14008
- [Mazur and Wiles 1984] B. Mazur and A. Wiles, "Class fields of abelian extensions of Q", *Invent. Math.* **76**:2 (1984), 179–330. MR 85m:11069 Zbl 0545.12005
- [Melo and Viviani 2012] M. Melo and F. Viviani, "Fine compactified Jacobians", *Math. Nachr.* **285**:8–9 (2012), 997–1031. Zbl 06049557
- [Néron 1964] A. Néron, "Modèles minimaux des variétés abéliennes sur les corps locaux et globaux", *Inst. Hautes Études Sci. Publ.Math.* **21** (1964), 5–128. MR 31 #3423 Zbl 0132.41403
- [Oda and Seshadri 1979] T. Oda and C. S. Seshadri, "Compactifications of the generalized Jacobian variety", *Trans. Amer. Math. Soc.* **253** (1979), 1–90. MR 82e:14054 Zbl 0418.14019
- [Pandharipande 1996] R. Pandharipande, "A compactification over  $\overline{M}_g$  of the universal moduli space of slope-semistable vector bundles", *J. Amer. Math. Soc.* **9**:2 (1996), 425–471. MR 96f:14014 Zbl 0886.14002
- [Raynaud 1970] M. Raynaud, "Spécialisation du foncteur de Picard", Inst. Hautes Études Sci. Publ. Math. 38 (1970), 27–76. MR 44 #227 Zbl 0207.51602
- [Silverman 1994] J. H. Silverman, *Advanced topics in the arithmetic of elliptic curves*, Graduate Texts in Mathematics **151**, Springer, New York, 1994. MR 96b:11074 Zbl 0911.14015
- [Simpson 1994] C. T. Simpson, "Moduli of representations of the fundamental group of a smooth projective variety. I", *Inst. Hautes Études Sci. Publ. Math.* **79** (1994), 47–129. MR 96e:14012 Zbl 0891.14005

Communicated by Brian Conrad Received 2011-09-23 Revised 2012-01-13 Accepted 2012-02-20

jkass@umich.edu

Institut für Algebraische Geometrie, Leibniz Universität Hannover, Welfengarten 1, D-30167 Hannover, Germany http://www2.iag.uni-hannover.de/~kass/

404



# Arithmetic motivic Poincaré series of toric varieties

Helena Cobo Pablos and Pedro Daniel González Pérez

The *arithmetic motivic Poincaré series* of a variety V defined over a field of characteristic zero is an invariant of singularities that was introduced by Denef and Loeser by analogy with the Serre–Oesterlé series in arithmetic geometry. They proved that this motivic series has a rational form that specializes to the Serre–Oesterlé series when V is defined over the integers. This invariant, which is known explicitly for a few classes of singularities, remains quite mysterious. In this paper, we study this motivic series when V is an affine toric variety. We obtain a formula for the rational form of this series in terms of the Newton polyhedra of the *ideals of sums of combinations* associated to the minimal system of generators of the semigroup of the toric variety. In particular, we explicitly deduce a finite set of candidate poles for this invariant.

### Introduction

Let *S* denote an irreducible and reduced algebraic variety defined over a field *k* of characteristic zero. The set H(S) of formal arcs of the form Spec  $k[[t]] \rightarrow S$  can be given the structure of scheme over *k* (not necessarily of finite type). If  $0 \in S$ , we denote by  $H(S)_0$  the subscheme of the arc space consisting of arcs in H(S) with origin at 0. The set  $H_m(S)$  of *m*-jets of *S* of the form Spec  $k[t]/(t^{m+1}) \rightarrow S$  has the structure of algebraic variety over *k*. By a theorem of Greenberg, the image of the space of arcs H(S) by the natural morphism of schemes  $j_m : H(S) \rightarrow H_m(S)$  that maps any arc to its *m*-jet is a constructible subset of  $H_m(S)$ .

It follows that  $j_m(H(S))$  defines a class  $[j_m(H(S))]$  in the Grothendieck ring of varieties  $K_0(\text{Var}_k)$  and also a class  $\chi_c([H_m(S)]) \in K_0(\text{CHMot}_k)$  in the Grothendieck ring of Chow motives, where  $\chi_c : K_0(\text{Var}_k) \to K_0(\text{CHMot}_k)$  is the unique ring homomorphism that maps the class of a smooth projective variety to its Chow motive (see [Gillet and Soulé 1996; Guillén and Navarro Aznar 2002]). We denote

Cobo Pablos is supported by FWO-Flanders project G031806N. González Pérez is supported by Programa Ramón y Cajal of the Ministerio de Ciencia e Innovación (MCI), Spain. Both authors are supported by MCI grant MTM2010-21740-C02-01.

MSC2010: primary 14M25; secondary 14J17, 14B05.

Keywords: arithmetic motivic Poincaré series, toric geometry, singularities, arc spaces.

by  $K_0^{\text{mot}}(\text{Var}_k)$  the image of  $K_0(\text{Var}_k)$  by the homomorphism  $\chi_c$ . We use the same symbol **L** to denote the class  $[\mathbf{A}_k^1] \in K_0(\text{Var}_k)$  and the class  $\chi_c([\mathbf{A}_k^1]) \in K_0^{\text{mot}}(\text{Var}_k)$ .

Denef and Loeser [DL 2004] have defined various notions of motivic Poincaré series, motivated by some generating series in arithmetic geometry. Assume for simplicity that the variety *S* is defined over the integers. We denote by *p* a prime number and by  $\mathbb{Z}_p$  the *p*-adic integers. For every positive integer *m*, the symbol  $N_{p,m}(S)$  denotes the number of rational points of *S* over  $\mathbb{Z}/p^{m+1}\mathbb{Z}$  that can be lifted to rational points of *S* over  $\mathbb{Z}_p$  by the projection induced by the natural map  $\mathbb{Z}_p \to \mathbb{Z}/p^{m+1}\mathbb{Z}$ . The *Serre–Oesterlé* series of *S* at the prime *p* is

$$P_p^S(T) = \sum_{m \ge 0} N_{p,m}(S) T^m \in \mathbb{Z}\llbracket T \rrbracket.$$

The definition of the geometric motivic Poincaré series,

$$P_{\text{geom}}^{S}(T) = \sum_{m \ge 0} \chi_{c}([j_{m}(H)])T^{m} \in K_{0}^{\text{mot}}(\text{Var}_{k}) \otimes \mathbb{Q}[[T]],$$

is inspired by that of the Serre–Oesterlé series. However, there is no specialization of the series  $P_{\text{geom}}^S(T)$  into  $P_p^S(T)$  in general [DL 2004].

Denef and Loeser studied the "motivic nature" of the series  $P_p^S(T)$ , passing through the Grothendieck ring  $K_0(\text{Field}_k)$  of ring formulas over k. By Greenberg's theorem, for every m there exists a formula  $\psi_m$  over k, such that for any field extension  $k \subset K$ , the m-jets over k that can be lifted to arcs defined over Kcorrespond to the tuples satisfying  $\psi_m$  in K. It follows that  $\psi_m$  defines an element  $[\psi_m] \in K_0(\text{Field}_k)$ . Then, Denef and Loeser defined a ring homomorphism  $\chi_f$ :  $K_0(\text{Field}_k) \to K_0^{\text{mot}}(\text{Var}_k) \otimes \mathbb{Q}$ . This homomorphism can be seen as a generalization of  $\chi_c$ , since the image by  $\chi_f$  of the class of the ring formula defining a variety V coincides with the class  $\chi_c([V])$  in  $K_0^{\text{mot}}(\text{Var}_k) \otimes \mathbb{Q}$ . The arithmetic motivic Poincaré series of S is defined as

$$P_{\mathrm{ar}}^{S}(T) = \sum_{m \ge 0} \chi_{f}([\psi_{m}]) T^{m} \in K_{0}^{\mathrm{mot}}(\mathrm{Var}_{k}) \otimes \mathbb{Q}[[T]].$$

Denef [1984] proved the rationality of the series  $P_p^S(T)$  using quantifier elimination results. Denef and Loeser [1999; 2001] proved the rationality of the series  $P_{geom}^S(T)$  and  $P_{ar}^S(T)$  by using quantifier eliminations theorems, various forms of motivic integration, and the existence of resolution of singularities.

If V is a variety defined over the integers and p is a prime number, the symbol  $N_p(V)$  denotes the number of rational points of V over the field of p elements. Denef and Loeser proved that the result of applying the operator  $N_p$  to the motivic arithmetic series  $P_{ar}^{S}(T)$  provides the Serre–Oesterlé series  $P_p^{S}(T)$  for almost all primes p.

406

If we fix the origin of the arcs at a fixed point  $0 \in S$ , we obtain local versions of the series  $P_{ar}^{(S,0)}(T)$  and  $P_{geom}^{(S,0)}(T)$ , which are also rational; see [DL 1999; 2001]. The rationality proofs therein are qualitative in nature; in particular, there is no conjecture on the significance of the terms appearing in the denominator of the rational form of the series  $P_{ar}^{(S,0)}(T)$  or in  $P_{geom}^{(S,0)}(T)$ . The rational form of the series  $P_{ar}^{(S,0)}(T)$  is known explicitly for a few classes

The rational form of the series  $P_{ar}^{(S,0)}(T)$  is known explicitly for a few classes of singularities. If (S, 0) is an analytically irreducible germ of plane curve, the information provided by the series  $P_{ar}^{(S,0)}(T)$  is equivalent to the data of the Puiseux pairs [DL 2001]. Nicaise [2005a] proved the equality of the geometric and arithmetic motivic Poincaré series in the case of varieties that admit a very special resolution of singularities, in particular for normal toric surfaces; see also [Nicaise 2005b; Lejeune-Jalabert and Reguera 2003]. He gave a criterion for the equality  $P_{ar}^{(S,0)}(T) = P_{geom}^{(S,0)}(T)$  for various classes of singularities and also an example of a normal toric threefold  $(S_0, 0)$  such that the series  $P_{ar}^{(S_0,0)}(T)$  and  $P_{geom}^{(S_0,0)}(T)$  are different. Some features of the motivic arithmetic series are studied for quasi-ordinary singularities in [Rond 2009].

In this paper, we describe the arithmetic motivic Poincaré series of an affine toric variety  $Z^{\Lambda} = \operatorname{Spec} k[\Lambda]$ , in terms of the semigroup  $\Lambda$ . We assume that  $\Lambda$  is a semigroup of finite type of a rank *d* lattice *M* (lattice of characters), which generates *M* as a group, and such that the cone  $\mathbb{R}_{\geq 0}\Lambda$  contains no lines. In this situation, there is a unique minimal system of generators  $e_1, \ldots, e_n$  of the semigroup  $\Lambda$ . The monomial ideal  $(X^{e_i})_{i=1,\ldots,n} \subset k[\Lambda]$  is maximal, and defines the distinguished point  $0 \in Z^{\Lambda}$ . In this paper we consider other monomial ideals as the *logarithmic jacobian ideals*  $\mathcal{F}_l$ , generated by monomials of the form  $X^u$  for *u* in the set

$$\{e_{i_1} + \dots + e_{i_l} \mid e_{i_1} \wedge \dots \wedge e_{i_l} \neq 0\}$$

for l = 1, ..., d (see [Cobo Pablos and González Pérez 2012]), and the *ideals of sums of combinations*  $\mathscr{C}_j$ , defined by monomials  $X^w$ , with w in the set

$$\left\{e_{i_1}+\cdots+e_{i_j}\mid \{i_1,\ldots,i_j\}\in \binom{\{1,\ldots,n\}}{j}\right\}$$

where  $\binom{\{1,\ldots,n\}}{j}$  denotes the set of combinations of *j* elements of  $\{1,\ldots,n\}$  for  $j = 1, \ldots, n$ .

We study the motivic arithmetic series  $P_{ar}^{(Z^{\Lambda},0)}(T)$  by extending the approach we used in [CoGP 2010; 2012] to describe the geometric motivic Poincaré series of toric and quasi-ordinary singularities.

For convenience, we explain the methods and results first when the variety  $Z^{\Lambda}$  is normal. The set  $j_m(H(Z^{\Lambda})_0)$  of *m*-jets of arcs through  $(Z^{\Lambda}, 0)$  is constructible; it is a finite disjoint union of locally closed subsets of the form  $j_m(H_v^*)$  [Cobo Pablos and González Pérez 2012]. Here  $H_v^*$  denotes the set of arcs through  $(Z^{\Lambda}, 0)$  that have generic point in the torus and a given order  $\nu \in M^*$ . The set  $H_{\nu}^*$  is an orbit of the natural action of the arc space of the torus on the arc space of the toric variety  $Z^{\Lambda}$  [Ishii 2004; 2005].

We describe the class, denoted by  $\chi_f([j_m(H_v^*)]_f)$ , of the formula defining the locally closed subset  $j_m(H_v^*)$  in terms of the Newton polyhedra of the logarithmic jacobian ideals and the degree of a certain Galois cover. This Galois cover reflects the relation between the initial coefficients of the arcs in  $H_v^*$  and the initial coefficients of the *m*-jets in  $j_m(H_v^*)$ ; see Section 5.

A key point in the description of the rational form of the series  $P_{\rm ar}^{(Z^{\Lambda},0)}(T)$ is that using the ideals  $\mathscr{C}_j$ , we can refine a finite partition of the set of possible pairs  $\{(v, m)\}$ , which was defined in [Cobo Pablos and González Pérez 2012] to describe the sum of  $P_{\text{geom}}^{(Z^{\Lambda},0)}(T)$ . If  $(\nu, m)$  and  $(\nu', m')$  belong to the same subset of this refinement, then the degrees of the Galois covers associated to  $j_m(H_v^*)$  and  $j_{m'}(H^*_{\nu'})$  coincide (see Sections 6 and 7). Using these partitions, we decompose the series  $P_{ar}^{(Z^{\Lambda},0)}(T)$  as a sum of a finite number of contributions. The main result is a formula for the rational form of  $P_{ar}^{(Z^{\Lambda},0)}(T)$  (see Theorem 11.4 and Corollary 10.4). The proofs pass by the results on the generating function of the projection of the set of integral points in the interior of a rational polyhedral cone; see [Cobo Pablos and González Pérez 2012]. The denominator of  $P_{ar}^{(Z^{\Lambda},0)}(T)$  is a finite product of terms of the form  $1 - \mathbf{L}^{a} T^{b}$  with a > 0 and b > 0, which are determined explicitly in terms of the ideals of sums of combinations  $\mathscr{C}_i$ . The integers a and b can be described in terms of the orders of vanishing of the ideals  $\mathscr{C}_i$  and  $\mathscr{G}_l$  at the codimension-one orbits of various toric modifications given by the Newton polyhedra of the ideals  $\mathscr{C}_i$ (see Remark 10.8). In the normal toric case, we obtain a formula for  $P_{\rm ar}^{Z^{\Lambda}}(T)$  in terms of arithmetic motivic series at the distinguished points of the orbits.

In the nonnormal case, we obtain in a similar way a formula for the rational form of  $P_{ar}^{(Z^{\Lambda},0)}(T)$  and the factors of its denominator. The main difference is that we have to consider contributions of jets of arcs with generic point in the various orbits of  $Z^{\Gamma}$ . We deduce a formula for the difference  $P_{geom}^{(Z^{\Lambda},0)}(T) - P_{ar}^{(Z^{\Lambda},0)}(T)$  and we give a criterion for the equality of these two series that generalizes the one given by Nicaise [2005b] (see Proposition 10.5 and Corollary 10.6).

The paper is organized as follows. In Sections 1 and 2 we introduce the Grothendieck rings, the arc and jet spaces, and the motivic Poincaré series. The notations on toric varieties, their monomial ideals, and their arcs are introduced in Sections 3 and 4. The computation of the class  $\chi_f([j_m(H_v^*)]_f)$  is given in Section 5. Sections 6 and 7 deal with the partitions associated to sequences of monomial ideals. The main results are stated and proved in Sections 8, 9 and 10. In the case of normal toric varieties, some features of the computation can be simplified (see Section 11). We discuss some examples in Section 12.

408

#### 1. Grothendieck rings of varieties and of ring formulas

The *Grothendieck ring*  $K_0(\operatorname{Var}_k)$  of *k*-varieties is the free abelian group of isomorphism classes [X] of *k*-varieties *X* modulo the relations  $[X] = [X'] + [X \setminus X']$  if *X'* is closed in *X*, and where the product is defined by  $[X][X'] = [X \times X']$ . We denote by  $\mathbf{L} := [\mathbf{A}_k^1]$  the class of the affine line. If *C* is a constructible subset of some variety *X*, that is, a disjoint union of finitely many locally closed subvarieties  $A_i$  of *X*, then  $[C] \in K_0(\operatorname{Var}_k)$  is well-defined as  $[C] := \sum_i [A_i]$  independently of the representation. Bittner [2004] proved, using the weak factorization theorem, that the ring  $K_0(\operatorname{Var}_k)$  is generated by classes of smooth projective *k*-varieties, modulo relations of the form [W] - [E] = [X] - [Y], where  $Y \subset X$  is a closed subvariety, and *W* is the blowing up of *X* along *Y* with exceptional divisor *E*.

We refer to [Scholl 1994; Gillet and Soulé 1996; Guillén and Navarro Aznar 2002] for the definition of the category of *Chow motives*. Roughly speaking, its definition involves replacing the category of smooth projective algebraic varieties over k by a category with basically the same objects, and whose morphisms are suitable correspondences modulo rational equivalence. There exists a unique ring homomorphism

$$\chi_c: K_0(\operatorname{Var}_k) \to K_0(\operatorname{CHMot}_k) \tag{1}$$

that maps the class of a smooth projective variety over k to its Chow motive, where  $K_0(\text{CHMot}_k)$  denotes the *Grothendieck ring of the category of Chow motives* over k (with coefficients in Q). This fundamental theorem, which is due to Gillet and Soulé [1996] and Guillén and Navarro Aznar [2002], can be seen also in terms of Bittner's result. We refer to [Gillet and Soulé 1996; Guillén and Navarro Aznar 2002; Bittner 2004] for precise definitions and proofs and to [Scholl 1994] for a survey on the notion of motives. We denote by  $K_0^{\text{mot}}(\text{Var}_k)$  the image of  $K_0(\text{Var}_k)$  in  $K_0(\text{CHMot}_k)$  under this homomorphism.

Notice that the image of  $\mathbf{L}$  in  $K_0^{\text{mot}}(\text{Var}_k)$ , which we denote with the same symbol, is not a zero divisor in  $K_0^{\text{mot}}(\text{Var}_k)$  since it is a unit in  $K_0(\text{CHMot}_k)$ . However, it seems that it is not known if  $\mathbf{L}$  is a zero divisor in  $K_0(\text{Var}_k)$ .

A *ring formula*  $\psi$  over *k* is a first-order formula in the language of *k*-algebras and free variables  $x_1, \ldots, x_n$ , that is, the formula  $\psi$  is built from boolean combinations ("and", "or", "not") of polynomial equations over *k* and existential and universal quantifiers. The *Grothendieck ring*  $K_0(\text{Field}_k)$  of ring formulas over *k* is generated by symbols  $[\psi]$ , where  $\psi$  is a ring formula over *k*, subject to the relations

$$[\psi_1 \lor \psi_2] = [\psi_1] + [\psi_2] - [\psi_1 \land \psi_2]$$

if  $\psi_1$  and  $\psi_2$  have the same free variables, and  $[\psi_1] = [\psi_2]$  if there exists a ring formula  $\Psi$  over *k* such that when interpreted in any field,  $K \supseteq k$  provides the graph of a bijection between the tuples of elements of *K* satisfying  $\psi_1$  and those satisfying

 $\psi_2$ . The ring multiplication is induced by the conjunction of formulas in disjoint sets of variables [DL 2001]. Denef and Loeser defined a ring homomorphism

$$\chi_f: K_0(\operatorname{Field}_k) \to K_0^{\operatorname{mot}}(\operatorname{Var}_k) \otimes \mathbb{Q}.$$
 (2)

They proved that this homomorphism is characterized by two conditions. The first one is that for any ring formula  $\psi$  that is a conjunction of polynomial equations over k, the element  $\chi_f([\psi])$  is equal to the class  $\chi_c([V])$  in  $K_0^{\text{mot}}(\text{Var}_k) \otimes \mathbb{Q}$  of the variety V defined by  $\psi$ . The second condition, which is more technical, expresses that certain relations should hold in terms of unramified Galois coverings over k. We refer to [DL 2001; 2004] for the precise statement. In the simplest case it implies the following:

**Example 1.1** [DL 2004, Example 6.4.3]. If  $n \ge 1$  is a fixed integer, k is a field containing all *n*-th roots of unity, and  $\psi$  is the ring formula

 $\psi$ : there exists y such that  $x = y^n$  and  $x \neq 0$ ,

then we have that  $\chi_f([\psi]) = (1/n)(\mathbf{L} - 1)$ .

**Lemma 1.2.** Let  $\psi$  be the ring formula whose interpretation in any field  $K \supseteq k$  provides the set of K-rational points of T that lift to K-rational points of T' by a Galois covering  $T' \to T$  of degree n of d-dimensional algebraic k-tori. If the field k contains all the n-th roots of unity, then we have that  $\chi_f([\psi]) = (1/n)(\mathbf{L}-1)^d$ .

*Proof.* The morphism  $T' \to T$  induces a finite index inclusion of the corresponding character group  $M \subseteq M'$ , and hence a map of *k*-algebras  $k[M] \hookrightarrow k[M']$ . By the classification theorem of finitely generated abelian groups applied to M'/M, there exists a basis  $\{v_1, \ldots, v_d\}$  of M' and unique integers  $b_1 | b_2 | \cdots | b_d$ , where | denotes division, such that  $\{b_1v_1, \ldots, b_dv_d\}$  is a basis of M and  $n = b_1 \ldots b_d$ . It follows that the map of coordinate rings  $K[M] \hookrightarrow K[M']$  expresses in coordinates as  $K[z_1^{\pm b_1}, \ldots, z_d^{\pm b_d}] \hookrightarrow K[z_1^{\pm 1}, \ldots, z_d^{\pm 1}]$ . We deduce that the ring formula  $\psi$  is the conjunction of formulas  $\psi_i$ : there exists  $y_i$  such that  $x_i = y_i^{b_i}$  and  $x_i \neq 0$  for  $i = 1, \ldots, d$ , where the variables  $x_1, \ldots, x_d$  are independent. Then we get that

$$\chi_f([\psi]) = \frac{1}{b_1 \dots b_d} (\mathbf{L} - 1)^d.$$

**Remark 1.3.** Denef and Loeser defined the map  $\chi_f$  by factoring it through the Grothendieck ring  $K_0(\text{PFF}_k)$  of ring formulas for the category of *pseudofinite fields* containing *k*. See [DL 2001; 2004; 2002].

#### 2. Arcs, jets and motivic Poincaré series

We start this section by recalling the definition of the space of arcs of a variety S. We assume for simplicity that S is an affine irreducible and reduced algebraic variety defined over a field k of characteristic zero.

For any integer  $m \ge 0$ , the functor from the category of *k*-algebras to the category of sets, sending a *k*-algebra *R* to the set of  $R[t]/(t^{m+1})$ -rational points of *S*, is representable by a *k*-scheme  $H_m(S)$  of finite type over *k*, called the *m*-jet scheme of *S*. The natural maps induced by truncation  $j_m^{m+1}: H_{m+1}(S) \to H_m(S)$  are affine, and hence the projective limit  $H(S) := \lim_{m \to \infty} H_m(S)$  is a *k*-scheme, not necessarily of finite type, called the *arc space* of *S*.

In what follows, we consider the schemes  $H_m(S)$  and H(S) with their reduced structure. We have natural morphisms  $j_m : H(S) \to H_m(S)$ . By an *arc* we mean a *k*-rational point of H(S), that is, a morphism Spec  $k[[t]] \to S$ . By an *m*-jet we mean a *k*-rational point of  $H_m(S)$ , that is, a morphism Spec  $k[[t]]/(t^{m+1}) \to S$ . The origin of the arc (respectively of the *m*-jet) is the image of the closed point 0 of Spec k[[t]] (respectively of Spec  $k[t]/(t^{m+1})$ ).

If  $Z \subset S$  is a closed subvariety, then  $H(S)_Z := j_0^{-1}(Z)$  (respectively  $H_m(S)_Z := (j_0^m)^{-1}(Z)$ ) denotes the subscheme of H(S) (respectively of  $H_m(S)$ ) formed by arcs (respectively *m*-jets) in *S* with origin in *Z*.

By a theorem of Greenberg [1966],  $j_m(H(S))$  is a constructible subset of the *k*-variety  $H_m(S)$  for any integer  $m \ge 0$ . We can then consider the class

$$[j_m(H(S))] \in K_0(\operatorname{Var}_k).$$

Greenberg's result implies also that there is a ring formula  $\psi_m$  over k, such that for any field K containing k, the k-rational points of  $H_m(S)$  that can be lifted to K-rational points of H(S) correspond to the tuples satisfying  $\psi_m$  in K. If  $\psi'_m$  is another ring formula over k with the same property, then  $[\psi_m] = [\psi'_m]$  in  $K_0(\text{Field}_k)$ . The same applies for  $j_m(H(S)_Z)$  if  $Z \subset S$  is a closed subvariety.

**Notation 2.1.** We denote the class  $[\psi_m]$  by  $[j_m(H(S))]_f$  to avoid confusion with the class  $[j_m(H(S))] \in K_0(\text{Var}_k)$ .

Definition 2.2 [DL 1999; 2001].

(1) The geometric motivic Poincaré series of (S, Z) is

$$P_{\text{geom}}^{(S,Z)}(T) := \sum_{m \ge 0} \chi_c([j_m(H(S)_Z)]) T^m \in K_0^{\text{mot}}(\text{Var}_k) \otimes \mathbb{Q}[\![T]\!].$$

(2) The arithmetic motivic Poincaré series of (S, Z) is

$$P_{\mathrm{ar}}^{(S,Z)}(T) := \sum_{m \ge 0} \chi_f([j_m(H(S)_Z)]_f) T^m \in K_0^{\mathrm{mot}}(\mathrm{Var}_k) \otimes \mathbb{Q}[\![T]\!].$$

**Remark 2.3.** In order to have the geometric and arithmetic setting in the same ring, we have slightly modified the original definition of the geometric motivic Poincaré series, since  $\sum_{m\geq 0} [j_m(H(S)_Z)]T^m \in K_0(\text{Var}_k)[[T]]$ ; see [DL 1999]. This does not affect the rationality results below.

Denef and Loeser proved that these series have a rational form:

**Theorem 2.4** [DL 1999, Theorem 1.1; 2001, Theorem 9.2.1]. *The series*  $P_{\text{geom}}^{(S,Z)}(T)$  and  $P_{\text{ar}}^{(S,Z)}(T)$  belong to the subring of  $K_0^{\text{mot}}(\text{Var}_k) \otimes \mathbb{Q}[\![T]\!]$  generated by

$$K_0^{\mathrm{mot}}(\mathrm{Var}_k)\otimes \mathbb{Q}[T]$$

and the series  $(1 - \mathbf{L}^a T^b)^{-1}$ , with  $a \in \mathbb{Z}$  and b > 0.

The arithmetic motivic Poincaré series has interesting properties of specialization to classical arithmetic series. Let *p* be a prime number. The operators  $N_p$  and  $N_{p,m}$  are applied to a variety *V* defined over the integers by  $N_p(V) := \#V(\mathbb{Z}/p\mathbb{Z})$  and  $N_{p,m}(V) := \#\{\pi_m(V(\mathbb{Z}_p))\}$ , where  $\mathbb{Z}_p$  denotes the *p*-adic integers,

$$\pi_m(V(\mathbb{Z}_p)) \subset V(\mathbb{Z}/p^{m+1}\mathbb{Z})$$

is the image of  $V(\mathbb{Z}_p)$ ) by the natural projection induced by  $\mathbb{Z}_p \to \mathbb{Z}/p^{m+1}\mathbb{Z}$ , and # denotes the cardinality. Suppose that the variety *S* is defined over the integers. The *Serre–Oesterlé* series  $P_p^S(T) := \sum_{m \ge 0} N_{p,m}T^m \in \mathbb{Z}[[T]]$  of *S* at the prime *p* is a rational function of *T* [Denef 1984]. Denef and Loeser proved that for  $p \gg 0$ , the series  $P_p^S(T)$  is obtained from  $P_{ar}^S(T)$  by applying to each coefficient the operator  $N_p$  [DL 2001; 2002; 2004].

**Remark 2.5.** These results hold in a more general setting, in particular when *S* is not affine as assumed here [DL 1999; 2001]. The proof of the rationality of  $P_p^S(T)$  involves the use of quantifier elimination results and *p*-adic integration [Denef 1984]. The proof of the rationality of  $P_{ar}^S(T)$  requires also quantifier elimination results and arithmetic motivic integration [DL 2001; 2004; 2002].

#### 3. Affine toric varieties and monomial ideals

In this section we introduce the basic notions and notations from toric geometry; see [Ewald 1996; Oda 1988; Fulton 1993; Gel'fand et al. 1994] for proofs.

If  $N \cong \mathbb{Z}^d$  is a lattice, we denote by  $N_{\mathbb{R}} := N \otimes \mathbb{R}$  the vector space spanned by N over the field  $\mathbb{R}$ , and by  $N_{\mathbb{Q}} := N \otimes \mathbb{Q}$  the vector space spanned by N over  $\mathbb{Q}$ . In what follows, a *cone* in  $N_{\mathbb{R}}$  means a *rational convex polyhedral cone*: the set of nonnegative linear combinations of vectors  $a_1, \ldots, a_r \in N$ . The cone  $\tau$  is *strictly convex* if it contains no line through the origin, in which case we denote by 0 the 0-dimensional face of  $\tau$ ; the cone  $\tau$  is *simplicial* if the primitive vectors of the 1-dimensional faces are linearly independent over  $\mathbb{R}$ . We denote by  $\mathring{\tau}$  or by  $int(\tau)$  the relative interior of the cone  $\tau$ .

We denote by *M* the dual lattice. The *dual* cone  $\tau^{\vee} \subset M_{\mathbb{R}}$  of  $\tau$  is the set  $\{w \in M_{\mathbb{R}} \mid \langle w, u \rangle \ge 0 \text{ for all } u \in \tau\}$ . The *orthogonal* cone  $\tau^{\perp}$  has the condition  $\langle w, u \rangle = 0$  instead of  $\langle w, u \rangle \ge 0$ .

A fan  $\Sigma$  is a family of strictly convex cones in  $N_{\mathbb{R}}$  such that any face of such a cone is in the family and the intersection of any two of them is a face of each. The relation  $\theta \leq \tau$  denotes that  $\theta$  is a face of  $\tau$ . By  $\theta < \tau$ , we mean  $\theta \neq \tau$  is a face of  $\tau$ . The *support* of the fan  $\Sigma$  is the set  $|\Sigma| := \bigcup_{\tau \in \Sigma} \tau \subset N_{\mathbb{R}}$ . The *k*-skeleton of the fan  $\Sigma$  is  $\Sigma^{(k)} = \{\tau \in \Sigma \mid \dim \tau = k\}$ . We say that a fan  $\Sigma'$  is a *subdivision* of the fan  $\Sigma$  if both fans have the same support and if every cone of  $\Sigma'$  is contained in a cone of  $\Sigma$ . If  $\Sigma_i$  for i = 1, ..., n are fans with the same support, their intersection  $\bigcap_{i=1}^n \Sigma_i := \{\bigcap_{i=1}^n \tau_i \mid \tau_i \in \Sigma_i\}$  is also a fan.

**Notation 3.1.** In this paper,  $\Lambda$  is a subsemigroup of finite type of a lattice M, which generates M as a group and such that the cone  $\sigma^{\vee} = \mathbb{R}_{\geq 0}\Lambda$  is strictly convex and of dimension d. We denote by N the dual lattice of M and by  $\sigma \subset N_{\mathbb{R}}$  the dual cone of  $\sigma^{\vee}$ . We denote by  $Z^{\Lambda}$  the *affine toric variety*  $Z^{\Lambda} = \operatorname{Spec} k[\Lambda]$ , where  $k[\Lambda] = \{\sum_{\text{finite}} a_{\lambda} X^{\lambda} \mid a_{\lambda} \in k\}$  denotes the semigroup algebra of the semigroup  $\Lambda$  with coefficients in the field k. The semigroup  $\Lambda$  has a unique minimal set of generators  $e_1, \ldots, e_n$ ; see the proof of [Ewald 1996, Lemma V.3.5, page 155]. We have an *embedding* of  $Z^{\Lambda} \subset \mathbf{A}_k^n$  given by  $x_i := X^{e_i}$  for  $i = 1, \ldots, n$ .

If  $\Lambda = \sigma^{\vee} \cap M$ , then the variety  $Z^{\Lambda}$ , which we denote also by  $Z_{\sigma,N}$  or by  $Z_{\sigma}$  when the lattice is clear from the context, is normal. If  $\Lambda \neq \sigma^{\vee} \cap M$ , the inclusion of semigroups  $\Lambda \to \overline{\Lambda} := \sigma^{\vee} \cap M$  defines a toric modification  $Z^{\overline{\Lambda}} \to Z^{\Lambda}$ , which is the *normalization map*.

The torus  $T_N := Z^M$  is an open dense subset of  $Z^{\Lambda}$  that acts on  $Z^{\Lambda}$ , and the action extends the action of the torus on itself by multiplication. The *origin* 0 of the affine toric variety  $Z^{\Lambda}$  is the 0-dimensional orbit, defined by the maximal ideal  $(X^{\lambda})_{0 \neq \lambda \in \Lambda}$  of  $k[\Lambda]$ . There is a one-to-one inclusion reversing correspondence between the faces of  $\sigma$  and the orbit closures of the torus action on  $Z^{\Lambda}$ . If  $\theta \leq \sigma$ , we denote by  $\operatorname{orb}_{\theta}^{\Lambda}$  the orbit corresponding to the face  $\theta$  of  $\sigma$ . The orbit closures are of the form  $Z^{\Lambda \cap \theta^{\perp}}$  for  $\theta \leq \sigma$ .

**Notation 3.2.** The *Newton polyhedron* of a monomial ideal corresponding to a nonempty set of lattice vectors  $\mathscr{I} \subset \Lambda$  is defined as the convex hull of the Minkowski sum of sets  $\mathscr{I} + \sigma^{\vee}$ . We denote this polyhedron by  $\mathscr{N}(\mathscr{I})$ . Notice that the vertices of  $\mathscr{N}(\mathscr{I})$  are elements of  $\mathscr{I}$ . We denote by  $\operatorname{ord}_{\mathscr{I}}$  the *support function* of the polyhedron  $\mathscr{N}(\mathscr{I})$ , which is defined by  $\operatorname{ord}_{\mathscr{I}} : \sigma \to \mathbb{R}$ ,  $\nu \mapsto \inf_{\omega \in \mathscr{N}(\mathscr{I})} \langle \nu, \omega \rangle$ . A vector  $\nu \in \sigma$  defines the face  $\mathscr{F}_{\nu} := \{\omega \in \mathscr{N}(\mathscr{I}) \mid \langle \nu, \omega \rangle = \operatorname{ord}_{\mathscr{I}}(\nu)\}$  of the polyhedron  $\mathscr{N}(\mathscr{I})$ . All faces of  $\mathscr{N}(\mathscr{I})$  are of this form, and the compact faces are defined by vectors  $\nu \in \mathring{\sigma}$ .

The fan  $\Sigma(\mathcal{I})$  of the polyhedron  $\mathcal{N}(\mathcal{I})$  is the set of cones

$$\sigma(\mathcal{F}) := \{ \nu \in \sigma \mid \langle \nu, \omega \rangle = \operatorname{ord}_{\mathcal{F}}(\nu) \text{ for all } \omega \in \mathcal{F} \}$$

for  $\mathscr{F}$  running through the faces of  $\mathcal{N}(\mathscr{I})$ . The fan  $\Sigma(\mathscr{I})$  is supported on  $\sigma$ . By definition, it is easy to check that if  $\tau \in \Sigma(\mathscr{I})$  and if  $\nu, \nu' \in \mathring{\tau}$ , then the faces of the polyhedron  $\mathcal{N}(\mathscr{I})$  defined by  $\nu$  and  $\nu'$  coincide, that is,  $\mathscr{F}_{\nu} = \mathscr{F}_{\nu'}$ ; we denote this face also by  $\mathscr{F}_{\tau}$ .

The affine varieties  $Z_{\tau}$  corresponding to cones  $\tau$  in a fan  $\Sigma$  glue up to define a *toric variety*  $Z_{\Sigma}$ . A fan  $\Sigma$  subdividing the cone  $\sigma$  defines a *toric modification*  $\pi_{\Sigma}: Z_{\Sigma} \to Z_{\sigma}$ .

If  $\mathscr{I} \subset \Lambda$  defines a monomial ideal, the composite  $Z_{\Sigma(\mathscr{I})} \xrightarrow{\pi_{\Sigma(\mathscr{I})}} Z_{\sigma} \longrightarrow Z^{\Lambda}$  is equal to the *normalized blowing up* of  $Z^{\Lambda}$  centered at  $\mathscr{I}$ ; see for instance [Lejeune-Jalabert and Reguera 2003].

**Definition 3.3.** For  $1 \le j \le n$ , the *j*-th ideal of sums of combinations of  $Z^{\Lambda}$  is the monomial ideal  $\mathscr{C}_j$  of  $k[\Lambda]$  generated by  $X^{\alpha}$ , where  $\alpha$  runs through

$$\left\{e_{i_1} + \dots + e_{i_j} \mid \{i_1, \dots, i_j\} \in \binom{\{1, \dots, n\}}{j}\right\},\tag{3}$$

where  $\binom{\{1,\ldots,n\}}{j}$  denotes the set of combinations of j elements of  $\{1,\ldots,n\}$  for  $j = 1, \ldots, n$ . We denote by  $\Theta_j$  the fan of the polyhedron  $\mathcal{N}(\mathcal{C}_j)$ , and by  $\operatorname{ord}_{\mathcal{C}_j}$  its support function; see Notation 3.2. The maps

$$\varphi_1 := \operatorname{ord}_{\mathscr{C}_1}$$
 and  $\varphi_j := \operatorname{ord}_{\mathscr{C}_j} - \operatorname{ord}_{\mathscr{C}_{j-1}}$  for  $j = 2, \ldots, n$ ,

are piecewise linear functions defined on the cone  $\sigma$ . If  $\nu \in \sigma$ , we write  $\varphi_0(\nu) := 0$ and  $\varphi_{n+1}(\nu) := +\infty$  for convenience.

**Definition 3.4.** For  $1 \le l \le d$ , the *l*-th logarithmic jacobian ideal of  $Z^{\Lambda}$  is the monomial ideal  $\mathcal{J}_l$  of  $k[\Lambda]$  generated by  $X^{\alpha}$ , where  $\alpha$  runs through

$$\{e_{i_1} + \dots + e_{i_l} \mid e_{i_1} \wedge \dots \wedge e_{i_l} \neq 0 \text{ for } 1 \le i_1, \dots, i_l \le n\}.$$
(4)

We denote by  $\Sigma_l$  the fan of the polyhedron  $\mathcal{N}(\mathcal{J}_l)$ , and by  $\operatorname{ord}_{\mathcal{J}_l}$  its support function; see Notation 3.2. The maps

$$\phi_1 := \operatorname{ord}_{\mathscr{Y}_1}$$
 and  $\phi_l := \operatorname{ord}_{\mathscr{Y}_l} - \operatorname{ord}_{\mathscr{Y}_{l-1}}$  for  $l = 2, \ldots, d$ ,

are piecewise linear functions defined on the cone  $\sigma$ . If  $\nu \in \sigma$ , we write  $\phi_0(\nu) := 0$ and  $\phi_{d+1}(\nu) := +\infty$  for convenience.

We also use the notations  $\mathcal{J}_l$  and  $\mathcal{C}_j$  for the sets (4) and (3), respectively.

**Lemma 3.5.** If  $v \in \mathring{\sigma}$  and if  $(p_1, \ldots, p_n)$  is a permutation of  $(1, \ldots, n)$  such that

$$\langle \nu, e_{p_1} \rangle \leq \cdots \leq \langle \nu, e_{p_n} \rangle,$$

then 
$$\operatorname{ord}_{\mathscr{C}_j}(v) = \langle v, \sum_{r=1}^j e_{p_r} \rangle$$
 and  $\varphi_j(v) = \langle v, e_{p_j} \rangle$  for  $1 \le j \le n$ . Moreover,  
 $0 = \varphi_0(v) \le \varphi_1(v) \le \cdots \le \varphi_n(v)$  and  $0 = \phi_0(v) \le \phi_1(v) \le \cdots \le \phi_d(v)$ .

*Proof.* The first assertion follows by induction on  $j \in \{1, ..., n\}$ .

See [Cobo Pablos and González Pérez 2012, Lemma 5.3] for the second sequence of inequalities.

**Proposition 3.6.** The Newton polyhedra of the ideals  $\mathscr{C}_j$  for j = 1, ..., n determine and are determined by the minimal system of generators of the semigroup  $\Lambda$ .

*Proof.* The Newton polyhedron  $\mathcal{N}(\mathcal{C}_j)$  determines and is determined by its support function  $\operatorname{ord}_{\mathcal{C}_j}$  for j = 1, ..., n. By Lemma 3.5 and the definitions if  $\theta$  is a *d*-dimensional cone of the fan  $\bigcap_{r=1}^n \Theta_r$ , there exists a permutation  $i_1, ..., i_n$  of 1, ..., n such that  $\varphi_j(v) = \langle v, e_{i_j} \rangle$  for j = 1, ..., n and all  $v \in \mathring{\theta}$ . Thus, the functions  $\varphi_j$  for j = 1, ..., n or equivalently,  $\operatorname{ord}_{\mathcal{C}_j}$  for j = 1, ..., n determine the vectors  $e_1, ..., e_n$ .

# 4. Arcs and jets on a toric singularity

Let  $\Lambda$  be a semigroup as in Notation 3.1. If R is a k-algebra, an R-rational point of  $Z^{\Lambda}$  is a homomorphism of semigroups  $(\Lambda, +) \rightarrow (R, \cdot)$ , where  $(R, \cdot)$ denotes the semigroup R for the multiplication. In particular, the closed points are obtained for R = k. An arc h on the affine toric variety  $Z^{\Lambda}$  is given by a semigroup homomorphism  $(\Lambda, +) \rightarrow (k[[t]], \cdot)$ . An arc in the torus  $T_N$  is defined by a semigroup homomorphism  $\Lambda \rightarrow k[[t]]^*$ , where  $k[[t]]^*$  denotes the group of units of the ring k[[t]].

**Notation 4.1.** We denote the set of arcs  $H(Z^{\Lambda})_0$  of  $Z^{\Lambda}$  with origin at the distinguished point 0 of  $Z^{\Lambda}$  simply by  $H_{\Lambda}$ , and by  $H^*_{\Lambda}$  the set consisting of those arcs of  $H_{\Lambda}$  with generic point in the torus  $T_N$ .

Notice that  $h \in H^*_{\Lambda}$  if and only if the formal power series  $X^u \circ h \in k[[t]]$  is nonzero for all  $u \in \Lambda$ . Any arc  $h \in H^*_{\Lambda}$  defines two group homomorphisms  $v_h : M \to \mathbb{Z}$  and  $\omega_h : M \to k[[t]]^*$  by  $X^m \circ h = t^{v_h(m)} \omega_h(m)$ . If  $m \in \Lambda$ , then  $v_h(m) > 0$ , and hence  $v_h$ belongs to  $\mathring{\sigma} \cap N$ . Notice that  $\omega_h$  defines an arc in the torus, that is,  $\omega_h \in H(T_N)$ .

**Remark 4.2.** The space of arcs in the torus acts on the arc space of a toric variety; [Ishii 2004; 2005].

**Lemma 4.3** [Ishii 2004, Theorem 4.1; 2005, Lemma 5.6; Lejeune-Jalabert and Reguera 2003, Proposition 3.3]. The map  $\mathring{\sigma} \cap N \times H(T_N) \to H^*_{\Lambda}$ , which applies a pair  $(v, \omega)$  to the arc h defined by  $X^u \circ h = t^{\langle v, u \rangle} \omega(u)$  for  $u \in \Lambda$  is a bijection. The sets  $H^*_{\Lambda, v} := \{h \in H^*_{\Lambda} \mid v_h = v\}$  for  $v \in \mathring{\sigma} \cap N$  are orbits for the action of  $H(T_N)$  on  $H^*_{\Lambda}$ , and we have  $H^*_{\Lambda} = \bigsqcup_{v \in \mathring{\sigma} \cap N} H^*_{\Lambda, v}$ . **Remark 4.4.** We often denote the set  $H^*_{\Lambda}$  and orbit  $H^*_{\Lambda,\nu}$  by  $H^*$  and  $H^*_{\nu}$ , respectively, if  $\Lambda$  is clear from the context.

An arc  $h \in H_{\Lambda}$  has its generic point  $\eta$  contained in exactly one orbit of the torus action on  $Z^{\Lambda}$ . If  $h(\eta) \in \operatorname{orb}_{\theta}^{\Lambda}$  for some  $\theta \leq \sigma$ , then h factors through the orbit closure  $Z^{\Lambda \cap \theta^{\perp}}$  and  $h \in H^*_{\Lambda \cap \theta^{\perp}}$ , that is, h is an arc through  $(Z^{\Lambda \cap \theta^{\perp}}, 0)$  with generic point in the torus  $\operatorname{orb}_{\theta}^{\Lambda}$ . We can apply Lemma 4.3 to describe the set  $H^*_{\Lambda \cap \theta^{\perp}}$ , just replacing the semigroup  $\Lambda$  by  $\Lambda \cap \theta^{\perp}$ ; see [Cobo Pablos and González Pérez 2012]. In particular, if  $\theta = 0$ , then  $h \in H^*_{\Lambda}$ ; if  $\theta = \sigma$ , then  $\Lambda \cap \theta^{\perp} = 0$  and h is the constant arc at the distinguished point  $0 \in Z^{\Lambda}$ . We have a partition  $H_{\Lambda} = \bigsqcup_{\theta < \sigma} H^*_{\Lambda \cap \theta^{\perp}}$ .

#### 5. The image of the class of the formula defining $j_s(H_v^*)$

**Definition 5.1.** We associate to  $(\nu, s) \in (\mathring{\sigma} \cap N) \times \mathbb{Z}_{>0}$  the sets

$$M_{\nu}^{s} := \operatorname{span}_{\mathbb{Z}} \{ e_{i} \mid \langle \nu, e_{i} \rangle \leq s, i = 1, \dots, n \},$$
  
$$\ell_{\nu}^{s} := \operatorname{span}_{\mathbb{Q}} \{ e_{i} \mid \langle \nu, e_{i} \rangle \leq s, i = 1, \dots, n \}.$$

We denote by l(v, s) the dimension of the Q-vector space  $\ell_v^s$ . The integer l(v, s) is also the rank of the lattice  $M_v^s$ . We denote by q(v, s) the index of the lattice extension  $M_v^s \subset \ell_v^s \cap M$ .

**Proposition 5.2.** If  $(v, s) \in \mathring{\sigma} \times \mathbb{Z}_{>0}$ , with l(v, s) > 0, and if the field k contains all the q(v, s)-th roots of unity, then we have

$$\chi_f([j_s(H_{\nu}^*)]_f) = \frac{1}{q(\nu,s)} (\mathbf{L}-1)^{l(\nu,s)} \times \mathbf{L}^{sl(\nu,s) - \operatorname{ord}_{\mathcal{J}_{l(\nu,s)}}(\nu)}.$$

*If* l(v, s) = 0, then we have  $\chi_f([j_s(H_v^*)]_f) = 1$ .

*Proof.* If  $h \in H_{\nu}^*$ , the equality  $\operatorname{ord}_t(X^{e_i} \circ h) = \langle \nu, e_i \rangle$  holds for  $1 \le i \le n$ . By Definition 5.1, those vectors  $e_i$  such that  $j_s(X^{e_i} \circ h) \ne 0$  span the Q-vector space  $\ell_{\nu}^s$ , since  $\langle \nu, e_i \rangle \le s$ . If  $l(\nu, s) = 0$ , this vector space is empty, the jet space  $j_s(H_{\nu}^*)$  consists of the constant 0-jet, and the conclusion follows easily from the definitions.

Suppose then that l := l(v, s) > 0. If  $h \in H_v^*$ , then it is given by *n* series of the form

$$X^{e_i} \circ h = t^{\langle v, e_i \rangle} c(e_i) \left( 1 + \sum_{m \ge 1} u_m(e_i) t^m \right) \quad \text{for } i = 1, \dots, n.$$

The *s*-jet  $j_s(X^{e_i} \circ h)$  is different from zero if and only if  $\langle v, e_i \rangle \leq s$ .

By [Cobo Pablos and González Pérez 2012, Lemma 5.7], there exist integers  $1 \le k_1, \ldots, k_l \le n$  such that  $\phi_i(v) = \langle v, e_{k_i} \rangle \le s$  for  $i = 1, \ldots, l$ ,

$$\ell_{\nu}^{s} = \operatorname{span}_{\mathbb{Q}} \{ e_{k_{1}}, \dots, e_{k_{l}} \}, \text{ and } \operatorname{ord}_{\mathcal{J}_{l}}(\nu) = \sum_{i=1}^{l} \langle \nu, e_{k_{i}} \rangle.$$

By [Cobo Pablos and González Pérez 2012, Section 6], if *h* is the universal family of arcs parametrizing  $H_{\nu}^*$ , the terms  $\{u_m(e_{k_i}) \mid i = 1, ..., l, m \ge 1\}$  are algebraically independent over  $\mathbb{Q}$  and the terms  $\{c(e_i)^{\pm 1} \mid i = 1, ..., n\}$  generate a *k*-algebra isomorphic to k[M] by the isomorphism that maps  $c(e_i) \mapsto X^{e_i}$ .

By the proof of [Cobo Pablos and González Pérez 2012, Theorem 7.1], a formula defining  $j_s(H_v^*)$  is the conjunction of two formulas,  $\psi_1$  and  $\psi_2$ , with independent sets of variables. The first formula,  $\psi_1$ , is a finite sequence of polynomial equalities with rational coefficients expressing the terms  $u_r(e_i)$  appearing in  $j_s(X^{e_i} \circ h)$ , for  $1 \le r \le s - \langle v, e_i \rangle$ , in terms of the variables

$$\{u_r(e_{k_i}) \mid 1 \le i \le l, \ 1 \le r \le s - \langle v, e_{k_i} \rangle\}.$$

We deduce that  $\chi_f([\psi_1]) = \mathbf{L}^{sl - \operatorname{ord}_{\mathcal{J}_l}(\nu)}$ . The second formula comes from the effect on the initial coefficients  $c(e_i)$  for  $e_i \in \ell_{\nu}^s$  of the operation taking the *s*-jet of an arc. This operation is described by taking the image by the map

$$\Psi: T' := \operatorname{Spec} k[c(e_i)^{\pm 1}]_{e_i \in \ell_v^s} \to T := \operatorname{Spec} k[c(e_i)^{\pm 1}]_{\langle v, e_i \rangle \leq s}$$

of the point determined by  $h \in H_{\nu}^*$ . The map  $\Psi$  is the unramified covering of *l*-dimensional algebraic tori determined by the inclusion  $M_{\nu}^s \subset \ell_{\nu}^s \cap M$  of index  $q(\nu, s)$  of rank  $l(\nu, s)$  lattices. Thus the second formula is equivalent to  $\psi_2$ : there exists  $y \in T'$  such that  $\Psi(y) = x$  for  $x \in T$ , and hence by Lemma 1.2 we get that  $\chi_f([\psi_2]) = (1/q(\nu, s))(\mathbf{L} - 1)^l$ .

#### 6. Sequences of convex piecewise linear functions and fans

Let  $\sigma \subset N_{\mathbb{R}}$  be a rational convex polyhedral cone of dimension  $d = \dim N_{\mathbb{R}}$ . Consider a sequence of piecewise linear continuous functions

$$h_p: \sigma \to \mathbb{R} \text{ for } 1 \leq p \leq m_p$$

such that  $h_p(\sigma \cap N) \subset \mathbb{Z}$ , and

$$0 \le h_1(\nu) \le \dots \le h_m(\nu) \quad \text{for all } \nu \in \sigma.$$
(5)

By convenience we set  $h_0(v) = 0$  and  $h_{m+1}(v) = +\infty$ . We denote by  $\Xi_0$  the fan consisting on the faces of  $\sigma$  and by  $\Xi_p$  the coarser fan such that the restriction of  $h_p$  to  $\eta$  is linear for any cone  $\eta \in \Xi_p$  for  $1 \le p \le m$ . In addition we assume that for any cone  $\eta \in \Xi_{p-1}$  the restriction  $h_{p|\eta}$  is *upper convex*, that is,  $h_p(v) + h_p(v') \le h_p(v + v')$  for all  $v, v' \in \eta$ .

**Notation 6.1.** For  $0 \le p \le m$  and for  $\eta \in \bigcap_{r=1}^{p} \Xi_r$  we set

$$\eta(\underline{h}, p) := \{ (\nu, s) \in N_{\mathbb{R}} \times \mathbb{R}_{\geq 0} \mid \nu \in \mathring{\sigma} \cap \mathring{\eta}, h_p(\nu) \le s < h_{p+1}(\nu) \}.$$

**Lemma 6.2.** The closure  $\bar{\eta}(\underline{h}, p)$  of the set  $\eta(\underline{h}, p)$  is a convex polyhedral cone that is rational for the lattice  $N \times \mathbb{Z}$ .

*Proof.* If  $\eta \in \bigcap_{r=0}^{p} \Xi_r$ , then the restriction  $h_{j|\eta} \colon \eta \to \mathbb{R}$  is linear if j = p, and upper convex if j = p + 1. It follows that  $\overline{\eta}(\underline{h}, p)$  is a convex polyhedral cone, rational for the lattice  $N \times \mathbb{Z}$ , since  $h_p$  and  $h_{p+1}$  take integral values on N.

**Notation 6.3.** For  $0 \le p \le m$  and  $\eta \in \Xi_p$  we define the following sets:

- (i)  $A(\underline{h}, p) := \{(v, s) \in N \times \mathbb{Z} \mid v \in \mathring{\sigma}, h_p(v) \le s < h_{p+1}(v)\}.$
- (ii)  $A(\underline{h}, p, \eta) := \{(v, s) \in N \times \mathbb{Z} \mid v \in \mathring{\sigma} \cap \mathring{\eta}, h_p(v) \le s < h_{p+1}(v)\}.$

**Remark 6.4.** We have partitions

418

$$(\mathring{\sigma} \cap N) \times \mathbb{Z}_{\geq 0} = \bigsqcup_{p=0}^{m} A(\underline{h}, p) \text{ and } A(\underline{h}, p) = \bigsqcup_{\eta \in \bigcap_{r=0}^{p} \Xi_{r}} A(\underline{h}, p, \eta).$$

#### 7. Refinements of partitions

We apply the procedure of Section 6 to both sequences  $\underline{\phi} = (\phi_1, \dots, \phi_d)$  and  $\varphi = (\varphi_1, \dots, \varphi_n)$  (see Lemma 3.5).

**Remark 7.1.** The sequence of fans associated to  $\underline{\phi}$  (respectively  $\underline{\phi}$ ) is  $\bigcap_{r=0}^{i} \Sigma_r$  for i = 0, ..., d (respectively  $\bigcap_{r=0}^{i} \Theta_r$  for i = 0, ..., n), where for convenience we denote by  $\Sigma_0$  or by  $\Theta_0$  the fan consisting of the faces of the cone  $\sigma$ .

**Lemma 7.2.** If  $A(\underline{\varphi}, j, \theta) \neq \emptyset$  for some  $1 \leq j \leq n$  and  $\theta \in \bigcap_{r=1}^{J} \Theta_r$  (using Notation 6.3), then the restriction of the functions  $(\mathring{\sigma} \cap N) \times \mathbb{Z}_{>0} \to \mathbb{Z}_{\geq 0}$  given by

$$(v, s) \mapsto l(v, s)$$
 and  $(v, s) \mapsto q(v, s)$ 

to the set  $A(\varphi, j, \theta)$  are constant functions. We denote their values on the set  $A(\varphi, j, \theta)$  by  $\overline{l}(j, \theta)$  and  $q(j, \theta)$ , respectively.

*Proof.* If  $v \in \mathring{\theta}$  for  $\theta \in \bigcap_{r=1}^{j} \Theta_r$ , then there exists a unique cone  $\theta_r \in \Theta_r$  such that  $v \in \mathring{\theta}_r$  for r = 1, ..., j. We denote by  $\mathcal{F}_{r,\theta}$  the face of the polyhedron  $\mathcal{N}(\mathscr{C}_r)$  defined by any such vector  $v \in \mathring{\theta}$  for r = 1, ..., j (see Notation 3.2).

Suppose that  $\nu, \nu' \in \hat{\theta}$  and  $(p_1, \ldots, p_n)$  and  $(p'_1, \ldots, p'_n)$  are two permutations of  $(1, \ldots, n)$  such that the inequalities

$$\langle \nu, e_{p_1} \rangle \leq \cdots \leq \langle \nu, e_{p_n} \rangle$$
 and  $\langle \nu', e_{p'_1} \rangle \leq \cdots \leq \langle \nu', e_{p'_n} \rangle$  (6)

hold. We prove first that

$$\langle v, e_{p'_1} \rangle \leq \cdots \leq \langle v, e_{p'_j} \rangle.$$
 (7)

By definition, for any  $1 \le r \le j$ , we have that  $\operatorname{ord}_{\mathscr{C}_r}(v) = \langle v, u_r \rangle$  for any  $u_r \in \mathscr{F}_{r,\theta}$ . We get from Lemma 3.5 that the vectors  $u_r := e_{p_1} + \cdots + e_{p_r}$  and  $u'_r := e_{p'_1} + \cdots + e_{p'_r}$  belong to  $\mathscr{F}_{r,\theta}$  for  $1 \le r \le j$ . This implies (7).

If  $(v, s) \in A(\underline{\varphi}, j, \theta)$ , then by Lemma 3.5, we obtain  $\varphi_j(v) = \langle v, e_{p_j} \rangle \le s < \varphi_{j+1}(v)$ . We deduce that if (v, s) and (v', s') belong to  $A(\varphi, j, \theta)$ , then

$$\{e_i \mid 1 \le i \le n, \ \langle v, e_i \rangle \le s\} = \{e_i \mid 1 \le i \le n, \ \langle v', e_i \rangle \le s'\} = \{e_{p_1}, \dots, e_{p_j}\}.$$
(8)

Since (8) spans the lattice  $M_{\nu}^{s}$  and the vector space  $\ell_{\nu}^{s}$ , the sublattices  $\ell_{\nu}^{s} \cap M$  and  $M_{\nu}^{s}$  are independent of the choice of  $(\nu, s)$  in  $A(\underline{\varphi}, j, \theta)$ . This implies the constancy of the functions l and q on  $A(\varphi, j, \theta)$ .

**Remark 7.3.** If  $1 \le l \le d$  and if  $\tau \in \bigcap_{r=1}^{l} \Sigma_r$ , we denoted in [Cobo Pablos and González Pérez 2012] the sets  $A(\underline{\phi}, l)$  and  $A(\underline{\phi}, l, \tau)$  by  $A_l$  and  $A_{l,\tau}$ , respectively. The map  $l(\nu, s)$  is also constant on the sets of the form  $A(\underline{\phi}, l, \tau)$  for  $\tau \in \bigcap_{i=0}^{l} \Sigma_i$ ; see [Cobo Pablos and González Pérez 2012, Lemma 5.7].

By Remark 6.4, we have two partitions

$$(\mathring{\sigma} \cap N) \times \mathbb{Z}_{>0} = \bigsqcup_{j=0}^{n} \bigsqcup_{\theta \in \bigcap_{r=0}^{j} \Theta_{r}} A(\underline{\varphi}, j, \theta),$$

$$(\mathring{\sigma} \cap N) \times \mathbb{Z}_{>0} = \bigsqcup_{l=0}^{d} \bigsqcup_{\eta \in \bigcap_{i=0}^{l} \Sigma_{i}} A(\underline{\phi}, l, \eta)$$

$$(9)$$

associated to the sequences  $\varphi$  and  $\phi$ .

**Proposition 7.4.** If  $\theta(\underline{\varphi}, j) \neq \emptyset$  for some  $1 \leq j \leq n$  and  $\theta \in \bigcap_{r=1}^{j} \Theta_r$ , then there exists a unique cone  $\tau \in \bigcap_{r=1}^{l(j,\theta)} \Sigma_r$  such that  $\theta \subset \tau$  and

$$A(\varphi, j, \theta) \subset A(\phi, l(j, \theta), \tau).$$
(10)

*Proof.* Given (v, s) and (v', s') in  $A(\underline{\varphi}, j, \theta) \subset (\overset{\circ}{\sigma} \cap N) \times \mathbb{Z}_{>0}$ , we deduce from (9) that there exist cones  $\tau \in \bigcap_{i=0}^{l} \Sigma_i$  and  $\tau' \in \bigcap_{i=0}^{l'} \Sigma_i$  for integers  $0 \le l, l' \le d$  such that  $(v, s) \in A(\underline{\phi}, l, \tau)$  and  $(v', s') \in A(\underline{\phi}, l', \tau')$ . By [Cobo Pablos and González Pérez 2012, Lemma 5.7], we have l = l(v, s) and l' = l(v', s'), and then l = l' by (8). Then  $l = l(j, \theta)$  by definition in Lemma 7.2.

Let  $(p_1, \ldots, p_n)$  and  $(p'_1, \ldots, p'_n)$  be two permutations of  $(1, \ldots, n)$  such that (6) holds. Then we can apply the method given in [Cobo Pablos and González Pérez 2012, Proposition 5.1] to determine the value of  $\operatorname{ord}_{\mathcal{F}_i}(v)$  for  $1 \le i \le l(v, s)$ . Moreover, it is enough to apply this on the set (8) instead of  $\{e_1, \ldots, e_n\}$ . We deduce from (7) that v and v' define the same face of  $\mathcal{N}(\mathcal{F}_i)$  for  $1 \le i \le l(v, s)$ . This is equivalent to the equality  $\tau = \tau'$ . We have proven (10), and as a consequence, the inclusion  $\theta \subset \tau$  holds.

**Definition 7.5** [Cobo Pablos and González Pérez 2012, Definition 8.1 and Remark 8.6]. We consider the equivalence relation ~ defined on the set  $(\mathring{\sigma} \cap N) \times \mathbb{Z}_{>0}$  by

$$(\nu, s) \sim (\nu', s') \quad \iff \quad s = s', \ \ell_{\nu}^s = \ell_{\nu'}^s \text{ and } \nu_{\mid \ell_{\nu}^s} = \nu'_{\mid \ell_{\nu'}^s}.$$

**Lemma 7.6.** The set  $A(\underline{\varphi}, j, \theta)$  is a union of equivalence classes by the relation  $\sim$  of Definition 7.5 for  $1 \leq j \leq n$  and  $\theta \in \bigcap_{r=1}^{j} \Theta_r$ . Moreover, we have

$$A(\underline{\phi}, l, \tau)/_{\sim} = \bigsqcup_{\theta \in \bigcap_{r=1}^{j} \Theta_{r}, \, l(j,\theta) = l}^{\theta \subset \tau} A(\underline{\phi}, j, \theta)/_{\sim}.$$
 (11)

*Proof.* By (9) and Proposition 7.4, it follows that

$$A(\underline{\phi}, l, \tau) = \bigsqcup_{\theta \in \bigcap_{r=1}^{j} \Theta_{r}, \, l(j,\theta) = l}^{\theta \subset \tau} A(\underline{\phi}, j, \theta).$$

If (v, s) belongs to  $A(\underline{\varphi}, j, \theta)$  and  $(v, s) \sim (v', s)$ , then (8) holds. The vectors v and v' define the same face of  $\mathcal{N}(\mathscr{C}_r)$  for  $1 \leq r \leq j$ , and therefore  $v' \in \operatorname{int} \theta$ . Since  $\varphi_j(v') \leq s < \varphi_{j+1}(v')$ , we conclude that  $(v', s) \in A(\underline{\varphi}, j, \theta)$ .

8. The structure of the series  $P_{ar}^{(Z^{\Lambda},0)}(T)$ 

We consider the auxiliary Poincaré series

$$P_{\mathrm{ar}}(\Lambda) := \sum_{s \ge 0} \chi_f \left( \left[ j_s(H_\Lambda) \setminus \bigcup_{0 \neq \theta \le \sigma} j_s(H_{\Lambda \cap \theta^\perp}) \right]_f \right) T^s \in K_0^{\mathrm{mot}}(\mathrm{Var}_k) \otimes \mathbb{Q}[\![T]\!].$$
(12)

Notice that the Poincaré series  $P_{ar}(\Lambda)$  measures the class of the formula defining the set of jets of arcs with origin in 0 that are not jets of arcs factoring through proper orbit closures of the toric variety  $Z^{\Lambda}$ .

**Proposition 8.1.** We have 
$$P_{ar}^{(Z^{\Lambda},0)}(T) = \sum_{\theta \leq \sigma} P_{ar}(\Lambda \cap \theta^{\perp}).$$

It follows from Proposition 8.1 that in order to describe the motivic series  $P_{ar}^{(Z^{\Lambda},0)}(T)$ , it is enough to describe the form of the auxiliary series  $P_{ar}(\Lambda)$  for any semigroup  $\Lambda$ .

**Remark 8.2.** In the normal case, the equality  $j_m(H_\Lambda) = j_m(H_\Lambda^*)$  holds for all  $m \ge 0$  [Nicaise 2005b], but this property fails in general.

**Definition 8.3** [Cobo Pablos and González Pérez 2012, Definition 8.9]. If  $1 \le l \le d$ ,  $\tau \in \bigcap_{i=1}^{l} \Sigma_{i}$ , and  $\nu \in \mathring{\tau}$ , then  $\nu$  defines a face  $\mathscr{F}_{l,\nu}$  of the polyhedron  $\mathscr{N}(\mathscr{F}_{l})$ . Since the face  $\mathscr{F}_{l,\nu}$  is independent of the choice of  $\nu \in \mathring{\tau}$ , we denote it by  $\mathscr{F}_{l,\tau}$ . If  $1 \le l \le d$ ,
the set  $\mathfrak{D}_l$  is the subset of cones  $\tau \in \bigcap_{i=1}^l \Sigma_i$  such that the face  $\mathcal{F}_{l,\tau}$  of  $\mathcal{N}(\mathcal{G}_l)$  is contained in the interior of  $\sigma^{\vee}$ . We denote by  $|\mathfrak{D}_l|$  the set  $\bigcup_{\tau \in \mathfrak{D}_l} \tau$ .

**Proposition 8.4.** Let us fix an integer  $s_0 \ge 1$ . The set  $j_{s_0}(H^*_{\Lambda}) \setminus \bigcup_{0 \neq \theta \le \sigma} j_{s_0}(H_{\Lambda \cap \theta^{\perp}})$  expresses as a finite disjoint union of locally closed subsets, as follows:

$$j_{s_0}(H^*_{\Lambda}) \setminus \bigcup_{0 \neq \theta \le \sigma} j_{s_0}(H_{\Lambda \cap \theta^{\perp}}) = \bigsqcup_{j=1}^n \bigsqcup_{\theta \in \bigcap_{r=1}^j \Theta_r} \bigsqcup_{[(\nu, s_0)] \in A(\underline{\varphi}, j, \theta)/\sim} j_{s_0}(H^*_{\Lambda, \nu}).$$
(13)

*Proof.* This partition follows from the partition in [Cobo Pablos and González Pérez 2012, Proposition 8.11] by using formula (11) (see Remark 7.3).  $\Box$ 

If  $s_0 \ge 1$ , the coefficient of  $T^{s_0}$  in the auxiliary series  $P(\Lambda)$  is obtained by applying the map  $\chi_f$  to the class of the formula defining (13). Then we determine this class by using Proposition 5.2.

We introduce the following auxiliary series for  $\theta \in \bigcap_{r=1}^{j} \Theta_r$ :

$$P_{\underline{\varphi},j,\theta}(\Lambda) := (\mathbf{L}-1)^{l(j,\theta)} \sum_{s \ge 1} \sum_{[(\nu,s)] \in A(\underline{\varphi},j,\theta)/\sim} \mathbf{L}^{l(j,\theta)s - \operatorname{ord}_{\mathcal{J}_{l(j,\theta)}}(\nu)} T^{s}.$$
(14)

We deduce the next proposition from Propositions 8.4 and 5.2 and formula (14).

**Proposition 8.5.** 
$$P_{\mathrm{ar}}(\Lambda) = \sum_{j=1}^{n} \sum_{\theta \in \bigcap_{r=1}^{j} \Theta_r}^{\theta \subset |\mathfrak{D}_{l(j,\theta)}|} \frac{1}{q(j,\theta)} P_{\underline{\varphi},j,\theta}(\Lambda).$$

#### 9. The rational form of some generating series

In this section, we fix an integer  $1 \le j \le n$  and a cone  $\theta \in \bigcap_{r=1}^{j} \Theta_r$  such that  $A(\underline{\varphi}, j, \theta) \ne \emptyset$ . For simplicity, we denote by *l* the integer  $l(j, \theta)$  defined in Lemma 7.2 and by  $\tau$  the unique element of the fan  $\bigcap_{r=1}^{l} \Sigma_r$  such that (10) holds.

Since  $\theta \subset \tau \subset \bigcap_{r=1}^{l} \Sigma_r$ , the restriction of  $\phi_r$  to  $\theta$  is a linear function of the form

$$(\phi_r)_{|\theta}(\nu) = \langle \nu, e_{i_r} \rangle$$
 for  $r = 1, \dots, l$ ,

where  $\{i_1, ..., i_l\} \subset \{1, ..., n\}$ .

Consider the lattice homomorphisms

$$\mu: N \times \mathbb{Z} \to \mathbb{Z}^{l+1}, \quad (\nu, s) \mapsto (\langle \nu, e_{i_1} \rangle, \dots, \langle \nu, e_{i_l} \rangle, s)$$

and

$$\pi = (\pi_1, \pi_2) : \mathbb{Z}^{l+1} \to \mathbb{Z}^2, \quad (a_1, \dots, a_{l+1}) \mapsto (la_{l+1} - a_1 - \dots - a_l, a_{l+1}).$$

We set  $\xi = \pi \circ \mu$ .

**Remark 9.1.** The homomorphisms  $\pi$ ,  $\mu$ , and  $\xi$  were considered in [Cobo Pablos and González Pérez 2012]. Since  $\theta$  is contained in  $\tau$ , the kernels of  $\mu$  and  $\xi$  intersect the cone  $\theta$  only at the origin. Similarly, by formula (10), the inclusion  $\xi(A(\underline{\varphi}, j, \theta)) \subset \mathbb{Z}^2_{\geq 0} \setminus \{(0, 0)\}$  holds. See [Cobo Pablos and González Pérez 2012, Section 9].

If  $j \neq n$ , the *lower boundary* of the cone  $\theta$  is the set

$$\partial_{-}\theta(\varphi, j) := \{(\nu, s) \mid \nu \in \theta, \ s = \varphi_{j}(\nu)\}.$$

Notice that  $\partial_{-}\theta(\underline{\varphi}, j)$  is a cone since  $\theta \in \bigcap_{r=1}^{j} \Theta_{j}$ , and then the function  $\varphi_{j}$  is linear on  $\theta$ . The *upper boundary* is the set

$$\partial_{+}\theta(\varphi, j) := \{(\nu, s) \mid \nu \in \theta, \ s = \varphi_{j+1}(\nu) \neq \varphi_{j}(\nu)\}.$$

If j = n, then l = d and  $\varphi_{n+1}(\nu) = +\infty$ , and the upper boundary is the union of cones spanned by  $(0, 1) \in N_{\mathbb{R}} \times \mathbb{R}$  and the proper faces of the cone  $\partial_{-}\theta(\underline{\varphi}, j)$ . The edges of the cone  $\theta(\varphi, j)$  are edges of  $\partial_{-}\theta(\varphi, j) \cup \partial_{+}\theta(\varphi, j)$ .

**Notation 9.2.** If  $\rho \subset \tau$  is a one-dimensional cone rational for the lattice *N*, we denote by  $\nu_{\rho}$  the primitive integral vector on  $\rho$ , that is, the generator of the semigroup  $\rho \cap N$ .

**Remark 9.3.** The primitive integral vectors for the lattice  $N \times \mathbb{Z}$  on the edges of the cone  $\theta$  are

 $(\nu_{\rho}, \varphi_{i}(\nu_{\rho}))$  for  $\rho \leq \theta$ , dim  $\rho = 1$ 

together with

$$\begin{cases} (0,1) & \text{if } j = n, \\ (\nu_{\rho}, \varphi_{j+1}(\nu_{\rho})) & \text{if } j \neq n \end{cases}$$

for  $\rho \in \Theta_{j+1}$ ,  $\rho \subset \theta$ , dim  $\rho = 1$ , and  $\varphi_j(\nu) \neq \varphi_{j+1}(\nu)$ . Then notice that

$$\xi(\nu, s) = \begin{cases} (l\varphi_{j}(\nu_{\rho}) - \operatorname{ord}_{\mathcal{G}_{l}}(\nu_{\rho}), \varphi_{j}(\nu_{\rho})) & \text{if } (\nu, s) = (\nu_{\rho}, \varphi_{j}(\nu_{\rho})), \\ (l\varphi_{j+1}(\nu_{\rho}) - \operatorname{ord}_{\mathcal{G}_{l}}(\nu_{\rho}), \varphi_{j+1}(\nu_{\rho})) & \text{if } (\nu, s) = (\nu_{\rho}, \varphi_{j+1}(\nu_{\rho})), \\ (d, 1) & \text{if } (\nu, s) = (0, 1). \end{cases}$$
(15)

**Definition 9.4.** Suppose that  $A(\underline{\varphi}, j, \theta) \neq \emptyset$ . We denote by  $B_{\underline{\varphi}, j, \theta}(\Lambda)$  the finite set

$$\left\{ (l\varphi_j(v_\rho) - \operatorname{ord}_{\mathcal{J}_l}(v_\rho), \varphi_j(v_\rho)) \middle| \rho \le \theta, \dim \rho = 1 \right\}$$

$$\cup \begin{cases} \left\{ (l\varphi_{j+1}(v_\rho) - \operatorname{ord}_{\mathcal{J}_l}(v_\rho), \varphi_{j+1}(v_\rho)) \middle| \rho \in \Theta_{j+1}^{(1)}, \rho \subset \theta \right\} & \text{if } j \neq n, \\ \\ \left\{ (d, 1) \right\} & \text{if } j = n. \end{cases}$$

**Definition 9.5.** If  $A \subset \mathbb{Z}^{l+1}$  is a set, we denote by  $F_A(x) := \sum_{a \in A} x^a$  the generating *function* of *A*; see [Cobo Pablos and González Pérez 2012, Section 12].

**Proposition 9.6.** We have the following equality:

$$P_{\underline{\varphi},j,\theta}(\Lambda) = (\mathbf{L}-1)^{l(j,\theta)} \sum_{a \in \mu(A(\underline{\varphi},j,\theta))} \mathbf{L}^{\pi_1(a)} T^{\pi_2(a)} \in \mathbb{Z}[\mathbf{L}][[T]].$$
(16)

There exists a polynomial  $R_{\underline{\varphi},j,\theta} \in \mathbb{Z}[\mathbf{L},T]$  such that  $P_{\underline{\varphi},j,\theta}(\Lambda)$  has the rational form

$$P_{\underline{\varphi},j,\theta}(\Lambda) = R_{\underline{\varphi},j,\theta} \prod_{(a,b)\in B_{\underline{\varphi},j,\theta}(\Lambda)} (1 - \mathbf{L}^a T^b)^{-1}.$$

*Proof.* The map  $\mu$  defines a bijection

$$A(\varphi, j, \theta)/_{\sim} \to \mu(A(\varphi, j, \theta)), \quad [(\nu, s)] \mapsto \mu(\nu, s);$$

see [Cobo Pablos and González Pérez 2012, Lemma 9.3] and Lemma 7.6. Then the equality (16) follows from the definitions.

We denote by  $\pi_*: k[[\mathbb{Z}^{l+1}]] \to k[[\mathbb{L}, T]]$  the monomial transformation defined by  $\pi_*(x^a) := \mathbf{L}^{\pi_1(a)} T^{\pi_2(a)}$  for  $a \in \mathbb{Z}^{l+1}$ . Then we get that

$$P_{\varphi,j,\theta}(\Lambda) = (\mathbf{L} - 1)^{l(j,\theta)} \pi_*(F_{\mu(A_{\varphi,j,\theta})}(x)).$$

We apply [Cobo Pablos and González Pérez 2012, Theorem 12.4]. We obtain that the denominator of a rational form of  $F_{\mu(A_{\underline{\varphi},j,\theta})}(x)$  consists of products of terms  $1 - x^{\mu(b)}$  for *b* running through the primitive integral vectors in the edges of the closure of the cone  $\theta(\underline{\varphi}, j)$ . Then the result follows by Remark 9.3 and Definition 9.4.

#### 10. Main results

**Definition 10.1.** (i) For a semigroup  $\Lambda$  generating a rank  $d \ge 1$  lattice, we define a finite subset  $B_{ar}(\Lambda)$  of  $\mathbb{Z}_{>0}^2$  as (see Definition 9.4):

$$B_{\rm ar}(\Lambda) := \bigcup_{\theta \in \bigcap_{r=1}^{j} \Theta_r, \, \theta \subset |\mathfrak{D}_{l(j,\theta)}|}^{1 \le j \le n} B_{\underline{\varphi}, j, \theta}(\Lambda).$$
(17)

If  $0 < \eta < \sigma$ , then  $\Lambda \cap \eta^{\perp}$  is a semigroup generating lattice of rank  $d - \dim \eta$ . We use formula (17) and Definition 9.4 to define in this case a finite subset  $B_{\rm ar}(\Lambda \cap \eta^{\perp})$  of  $\mathbb{Z}^2_{\geq 0}$ . We set

$$B_{\mathrm{ar}}(\Lambda \cap \sigma^{\perp}) := \{(0, 1)\} \text{ and } B_{\mathrm{ar},\Lambda} := \bigcup_{\tau \leq \sigma} B_{\mathrm{ar}}(\Lambda \cap \sigma^{\perp}).$$

(ii) We define the integer

$$q(\Lambda) := \operatorname{lcm} \left\{ q(j,\theta) \mid \theta \in \bigcap_{r=1}^{J} \Theta_r, \ \theta \subseteq |D_{l(j,\theta)}|, \ 1 \le j \le n \right\}.$$
(18)

If  $0 < \eta < \sigma$ , then  $q(\Lambda \cap \eta^{\perp})$  is the number obtained by replacing  $\Lambda$  by  $\Lambda \cap \eta^{\perp}$  in (18). We set  $q(\Lambda \cap \sigma^{\perp}) := 1$ . We define also the integer

$$q_{\Lambda} := \operatorname{lcm}\{q(\Lambda \cap \eta^{\perp}) \mid \eta \leq \sigma\}.$$

**Theorem 10.2.** Suppose that the field k contains all  $q(\Lambda)$ -th roots of unity. Then there exists a polynomial  $Q_{ar}(\Lambda) \in \mathbb{Z}[\mathbf{L}, T]$  such that

$$P_{\mathrm{ar}}(\Lambda) = \frac{1}{q(\Lambda)} Q_{\mathrm{ar}}(\Lambda) \prod_{(a,b)\in B_{\mathrm{ar}}(\Lambda)} (1 - \mathbf{L}^a T^b)^{-1}.$$

*Proof.* This follows from Propositions 8.5 and 9.6.

**Notation 10.3.** If  $\eta < \sigma$ , then the polynomial  $Q_{ar}(\Lambda \cap \eta^{\perp})$  is obtained from Theorem 10.2 by replacing  $\Lambda$  by the semigroup  $\Lambda \cap \eta^{\perp}$ . We set  $Q_{ar}(\Lambda \cap \sigma^{\perp}) := 1$ .

**Corollary 10.4.** *If the field k contains all*  $q_{\Lambda}$ *-th roots of unity, then there exists a polynomial*  $Q_{ar,\Lambda} \in \mathbb{Z}[\mathbf{L}, T]$  *such that* 

$$P_{\mathrm{ar}}^{(Z^{\Lambda},0)}(T) = \frac{1}{q_{\Lambda}} Q_{\mathrm{ar},\Lambda} \prod_{(a,b)\in B_{\mathrm{ar},\Lambda}} (1 - \mathbf{L}^{a} T^{b})^{-1}.$$

Moreover, we have the equality

$$P_{\mathrm{ar}}^{(Z^{\Lambda},0)}(T) = \sum_{\eta \le \sigma} \frac{1}{q(\Lambda \cap \eta^{\perp})} Q_{\mathrm{ar}}(\Lambda \cap \eta^{\perp}) \prod_{(a,b) \in B_{\mathrm{ar}}(\Lambda \cap \eta^{\perp})} (1 - \mathbf{L}^{a} T^{b})^{-1}.$$
(19)

*Proof.* The result follows by Theorem 10.2 and Proposition 8.1.

We can now compare the series  $P_{\text{geom}}^{(Z,0)}(T)$  and  $P_{\text{ar}}^{(Z,0)}(T)$  (see Definition 2.2). In [Cobo Pablos and González Pérez 2012] we introduced the series

$$P_{\text{geom}}(\Lambda) := \sum_{s \ge 0} \chi_c \left( \left[ j_s(H^*_{\Lambda}) \setminus \bigcup_{0 \neq \theta \le \sigma} j_s(H_{\Lambda \cap \theta^{\perp}}) \right] \right) T^s \in K_0^{\text{mot}}(\text{Var}_k) \otimes \mathbb{Q}[\![T]\!],$$
(20)

and we proved that

$$P_{\text{geom}}^{(Z^{\Lambda},0)}(T) = \sum_{\theta \le \sigma} P_{\text{geom}}(\Lambda \cap \theta^{\perp}).$$

**Proposition 10.5.** *If the field k contains all*  $q(\Lambda)$ *-th roots of unity, then* 

$$P_{\rm ar}(\Lambda) - P_{\rm geom}(\Lambda) = \sum_{j=1}^{n} \sum_{\theta \in \bigcap_{r=1}^{j} \Theta_r}^{\theta \subset |\mathcal{G}_{l(j,\theta)}|} \left(1 - \frac{1}{q(j,\theta)}\right) R_{\underline{\varphi},j,\theta} \prod_{(a,b) \in B_{\underline{\varphi},j,\theta}(\Lambda)} (1 - \mathbf{L}^a T^b)^{-1}.$$

*Proof.* This follows from Proposition 9.6, formula (20), Theorem 10.2, and the results in [Cobo Pablos and González Pérez 2012] for  $P_{\text{geom}}(\Lambda)$ .

424

**Corollary 10.6.** If for every integer  $1 \le l \le d$  and any vertex v of the Newton polyhedra  $\mathcal{N}(\mathcal{F}_l)$ , there exists a subset  $I_v \subset \{1, \ldots, n\}$  of l elements such that

$$v = \sum_{i \in I_v} e_i$$

and the vectors  $e_i$  for  $i \in I_v$  form part of a basis of M, then the series  $P_{ar}^{(Z^{\Lambda},0)}(T)$ and  $P_{geom}^{(Z^{\Lambda},0)}(T)$  coincide.

*Proof.* This condition implies that q(v, s) = 1 for every  $(v, s) \in (\mathring{\sigma} \cap N) \times \mathbb{Z}_{>0}$ . By Proposition 10.5, we get that  $P_{ar}(\Lambda) = P_{geom}(\Lambda)$ . Now for any face  $\eta \leq \sigma$ , the vertices of the Newton polyhedra of the logarithmic jacobian ideals of  $\Lambda \cap \eta^{\perp}$  are also vertices of the logarithmic jacobian ideals of  $\Lambda$ . The hypothesis implies that  $\Lambda \cap \eta^{\perp}$  spans the lattice  $M \cap \eta^{\perp}$  and then also that  $P_{ar}(\Lambda \cap \eta^{\perp}) = P_{geom}(\Lambda \cap \eta^{\perp})$ .  $\Box$ 

**Remark 10.7.** Corollary 10.6 is a generalization of the Nicaise condition [2005b, Theorem 1] in the case of normal toric varieties.

**Remark 10.8.** The coordinates of the vectors in the set  $B_{\underline{\varphi},j,\theta}(\Lambda)$  can be described geometrically in terms of the ideals  $\mathscr{C}_j$  and  $\mathscr{J}_l$  for  $l = l(j, \overline{\theta})$ . Let  $\pi_j : Z_j \to Z^{\Lambda}$  be the composite of the normalization of  $Z^{\Lambda}$ , with the toric modification defined by the subdivision  $\bigcap_{r=1}^{j} \Theta_r$  of  $\sigma$ . The modification  $\pi_j$  is the minimal toric modification that factors through the normalized blowing up with center  $\mathscr{C}_r$  for  $r = 1, \ldots, j$ . If  $\rho$  is an edge of  $\theta$ , the orbit closure  $E_{\rho}$  of the orbit associated to  $\rho$  on  $Z_j$  has codimension one. We denote by  $v_{\rho}$  the divisorial valuation defined by  $E_{\rho}$ . It satisfies  $v_{\rho}(X^m) = \langle v_{\rho}, m \rangle$  for  $m \in M$ . The pullback  $\pi_j^*(\mathscr{G}_r)$  of a monomial ideal  $\mathscr{I}$ of  $Z^{\Lambda}$  is a sheaf of monomial ideals on  $Z_j$ . The ideals  $\pi_j^*(\mathscr{C}_r)$  for  $r = 1, \ldots, j$  are locally principal on  $Z_j$ . Then we get the following identities:

$$\varphi_j(v_\rho) = v_\rho(\pi_j^*(\mathscr{C}_j)) - v_\rho(\pi_j^*(\mathscr{C}_{j-1})),$$
  

$$\varphi_{j+1}(v_\rho) = v_\rho(\pi_j^*(\mathscr{C}_{j+1})) - v_\rho(\pi_j^*(\mathscr{C}_j)),$$
  

$$\operatorname{ord}_{\mathscr{G}_l}(v_\rho) = v_\rho(\pi_j^*(\mathscr{G}_l)).$$

Compare this with the geometrical description of the set of candidate poles of  $P_{\text{geom}}^{(Z^{\Lambda},0)}(T)$ ; see [Cobo Pablos and González Pérez 2012].

#### 11. The normal case

In the normal case, when the semigroup  $\Lambda$  is saturated, that is,  $\Lambda = \sigma^{\vee} \cap M$ , we describe the motivic arithmetic series in a simpler way by using  $j^s(H_{\Lambda}) = j_s(H_{\Lambda}^*)$ ; see [Nicaise 2005b].

Notation 11.1. (i)  $\mathcal{A} = \bigsqcup_{l=1}^{d} \bigsqcup_{\tau \in \bigcap_{r=1}^{l} \Sigma_{r}} A(\underline{\phi}, l, \tau)/_{\sim}.$ (ii) For  $s_{0} \ge 0$  we set  $\mathcal{A}_{s_{0}} = \{[(\nu, s)] \in \mathcal{A} \mid s = s_{0}\}.$  **Remark 11.2.** The set  $\mathcal{A}_s$  is finite; see [Cobo Pablos and González Pérez 2012, Remark 8.2]. By (9) and Lemma 7.6, we deduce that

$$\mathscr{A} = \bigsqcup_{j=1}^{n} \bigsqcup_{\theta \in \bigcap_{r=1}^{j} \Theta_{r}} A(\underline{\varphi}, j, \theta) /_{\sim}.$$

**Proposition 11.3.** Let us fix an integer  $s_0 \ge 1$ . The set  $j_{s_0}(H^*)$  expresses as a finite disjoint union of locally closed subsets as  $j_{s_0}(H^*) = \bigsqcup_{[(v,s)] \in \mathcal{A}_{s_0}} j_{s_0}(H^*_v)$ . We deduce that  $\chi_f([j_s(H^*)]_f) = \sum_{[(v,s)] \in \mathcal{A}_s} \chi_f([j_s(H^*_v)]_f)$ .

*Proof.* The first claim follows by the method of [Cobo Pablos and González Pérez 2012, Proposition 8.11]. The second is a consequence of the first and Proposition 5.2.

**Theorem 11.4.** If  $Z^{\Lambda}$  is normal, then we have

$$P_{\mathrm{ar}}^{(Z^{\Lambda},0)} = \sum_{j=1}^{n} \sum_{\theta \in \bigcap_{r=1}^{j} \Theta_{r}} \frac{1}{q(j,\theta)} R_{\underline{\varphi},j,\theta} \prod_{(a,b)\in B_{\underline{\varphi},j,\theta}(\Lambda)} (1 - \mathbf{L}^{a} T^{b})^{-1}.$$

*Proof.* It is a consequence of Propositions 11.3 and 9.6 and Remark 11.2.  $\Box$ 

**Corollary 11.5.** Suppose that the affine toric variety  $Z^{\Lambda}$  is normal. If  $\theta \leq \sigma$ , we denote by  $\sigma_{\theta}^{\vee}$  the image of the cone  $\sigma^{\vee}$  in  $(M_{\theta})_{\mathbb{R}}$ , where  $M_{\theta} := M/\theta^{\perp} \cap M$ , and by  $\Lambda(\theta)$  the saturated semigroup  $\Lambda(\theta) := (\sigma_{\theta}^{\vee} \cap M_{\theta}) \times \mathbb{Z}_{\geq 0}^{\operatorname{codim}\theta}$ . With this notation, we have

$$P_{\mathrm{ar}}^{Z^{\Lambda}}(T) = \sum_{\theta \leq \sigma} (\mathbf{L} - 1)^{\operatorname{codim}\theta} P_{\mathrm{ar}}^{(Z^{\Lambda(\theta)}, 0)}(T).$$

*Proof.* The proof follows by the arguments of [Cobo Pablos and González Pérez 2012, Corollary 4.11].

#### 12. Examples

**12a.** *The case of monomial curves.* Let  $\Lambda \subset \mathbb{Z}_{\geq 0}$  be a semigroup with minimal system of generators  $e_1 < e_2 < \cdots < e_n$  such that  $gcd\{e_1, \ldots, e_n\} = 1$ . In this case, we have  $\mathring{\sigma} \cap N = \mathbb{Z}_{>0}$ . If  $q_i := gcd\{e_1, \ldots, e_i\}$ , then we obtain

$$P_{\rm ar}^{(Z^{\Lambda},0)}(T) = \frac{1}{1-T} + \frac{\mathbf{L}-1}{1-\mathbf{L}T} \Big( \frac{1}{q_1} \frac{T^{e_1}}{1-T^{e_1}} + \sum_{i=2}^n \frac{q_{i-1}-q_i}{q_{i-1}q_i} \frac{\mathbf{L}^{e_i-e_1}T^{e_i}}{1-\mathbf{L}^{e_i-e_1}T^{e_i}} \Big).$$
(21)

This follows from the results of this paper, taking the following observations into account:

- We have the equality  $j_s(H) = j_s(H^*)$ .
- If  $\nu, \nu' \in \mathbb{Z}_{>0}$  satisfy  $j_s(H_{\nu}^*)$  and  $j_s(H_{\nu'}^*) \neq \{0\}$ , then the equality  $j_s(H_{\nu}^*) = j_s(H_{\nu'}^*)$  implies  $\nu = \nu'$ .

• If  $\nu \in \mathbb{Z}_{>0}$  satisfies  $\nu e_i \leq s < \nu e_{i+1}$ , then  $q(\nu, s) = q_i$ .

Then, setting  $e_{d+1} := \infty$ , we get the following equality, which implies (21):

$$P_{\rm ar}^{(Z^{\Lambda},0)}(T) = \frac{1}{1-T} + \sum_{i=1}^{n} \sum_{\nu=1}^{\infty} \sum_{s=\nu e_i}^{\nu e_{i+1}-1} (\mathbf{L}-1) \frac{1}{q_i} \mathbf{L}^{s-\nu e_1} T^s$$

**Remark 12.1.** The inequalities  $q_1 \ge q_2 \ge \cdots \ge q_n = 1$  are not always strict. For instance, if  $\Lambda$  is generated by  $e_1 = 8$ ,  $e_2 = 18$ ,  $e_3 = 20$  and  $e_4 = 21$ , then we get  $q_1 = 8$ ,  $q_2 = q_3 = 2$  and  $q_4 = 1$ . It follows from (21) that the term  $1 - \mathbf{L}^{12}T^{20}$  is not a factor of the denominator of the series  $P_{\text{ar}}^{(Z^{\Lambda},0)}(T)$ . If  $\Lambda'$  is the semigroup generated by  $e_1$ ,  $e_2$  and  $e_4$ , then we obtain from (21) that

$$P_{\rm ar}^{(Z^{\Lambda},0)}(T) = P_{\rm ar}^{(Z^{\Lambda'},0)}(T),$$

while the semigroups  $\Lambda$  and  $\Lambda'$  are not isomorphic. In contrast with this behavior, the motivic series  $P_{ar}^{(C,0)}(T)$  of a plane branch (C, 0) determines the semigroup of the branch (C, 0); see [DL 2001].

**12b.** An example of non-normal toric surface. Consider the semigroup  $\Lambda$  generated by the vectors  $e_1 = (5, 0)$ ,  $e_2 = (0, 2)$ ,  $e_3 = (0, 3)$  and  $e_4 = (6, 2)$ . The cone  $\sigma$  is  $\mathbb{R}^2_{\geq 0}$  and the lattice *M* is equal to  $\mathbb{Z}^2$ . We have the semigroups  $\Lambda \cap \eta_1^{\perp} = (5, 0)\mathbb{Z}_{>0}$  and  $\Lambda \cap \eta_2^{\perp} = (0, 2)\mathbb{Z}_{>0} + (0, 3)\mathbb{Z}_{>0}$ , where  $\eta_1$  and  $\eta_2$  are the one-dimensional faces of  $\sigma$ . By the case of monomial curves, we get

$$P_{\rm ar}(\Lambda \cap \eta_1^{\perp}) = \frac{\mathbf{L} - 1}{1 - \mathbf{L}T} \frac{T}{1 - T} \quad \text{and} \quad P_{\rm ar}(\Lambda \cap \eta_2^{\perp}) = \frac{\mathbf{L} - 1}{2(1 - \mathbf{L}T)} \Big( \frac{T^2}{1 - T^2} + \frac{\mathbf{L}T^3}{1 - \mathbf{L}T^3} \Big).$$

Figure 1 shows the subdivisions associated with the ideals  $\mathscr{C}_r$  for r = 1, 2, 3.

In the following table, we give the different values of  $q(j, \theta)$  and  $l(j, \theta)$  for  $\theta$  in the subdivisions of Figure 1 and j such that  $A(\varphi, j, \theta) \neq \emptyset$ . We exclude from



**Figure 1.** The subdivisions  $\Theta_1$ ,  $\Theta_1 \cap \Theta_2$  and  $\Theta_1 \cap \Theta_2 \cap \Theta_3$ .

this table the cones in  $\theta \in \bigcap_{r=1}^{4} \Theta_r$  for j = 4, since in this case  $q(4, \theta) = 1$  and  $l(4, \theta) = 2$ .

$q(1, \theta_{11}) = 2 l(1, \theta_{11}) = 1$	$q(1, \theta_{12}) = 5 l(1, \theta_{12}) = 1$			
$q(2,\theta_{21}) = 1$	$q(2,\theta_{22}) = 10$	$q(2,\theta_{23}) = 10$	$q(2, \rho_1) = 10$	
$l(2,\theta_{21})=1$	$l(2,\theta_{22})=2$	$l(2,\theta_{23})=2$	$l(2, \rho_1) = 2$	
$q(3,\theta_{31})=5$	$q(3,\theta_{32})=5$	$q(3,\theta_{33})=5$	$q(3,\theta_{34})=2$	$q(3, \rho_1) = 5$
				$q(3, \rho_2) = 5$
$l(3,\theta_{31})=2$	$l(3,\theta_{32}) = 5$	$l(3,\theta_{33})=2$	$l(3,\theta_{34})=2$	$l(3, \rho_1) = 2$
				$l(3, \rho_2) = 2$

Notice that  $A(\underline{\varphi}, 1, \rho_1) = A(\underline{\varphi}, 2, \rho_2) = A(\underline{\varphi}, 3, \rho_3) = \emptyset$ . In the following table, we have filled in the cases corresponding to the pairs  $(a, b) \in B_{ar}(\Lambda)$ ,  $(a, b) \neq (2, 1)$  in terms of the rays appearing in the subdivisions of Figure 1:

$(a, b) \in B_{\mathrm{ar}}(\Lambda)$	$\nu_{\rho_1} = (2,5)$	$v_{\rho_2} = (3, 5)$	$v_{\rho_3} = (1, 6)$	$\nu_{\sigma^{\vee}\cap\eta_1^{\perp}}=(1,0)$	) $\nu_{\sigma^{\vee} \cap \eta_2^{\perp}} = (0, 1)$
$(2\varphi_2 - \operatorname{ord}_{\mathcal{F}_2}, \varphi_2)$	(0, 10)	(5, 15)			(2, 2)
$(2\varphi_3 - \operatorname{ord}_{\mathcal{J}_2}, \varphi_3)$	(10, 15)	(5, 15)	(19, 18)	(5, 5)	(2, 2)
$(2\varphi_4 - \operatorname{ord}_{\mathcal{J}_2}, \varphi_4)$	(24, 22)	(31, 28)	(19, 18)	(7, 6)	(4, 3)

It follows that  $B_{ar,\Lambda} = B_{ar}(\Lambda) \cup \{(1,3), (0,2), (1,1), (0,1)\}$ . We have computed the sum of the series  $P_{ar}^{(Z^{\Lambda},0)}(T)$  with the methods of [Cobo Pablos and González Pérez 2012]. We have obtained an irredundant representation of the form

$$P_{\mathrm{ar}}^{(Z^{\Lambda},0)}(T) = R(\mathbf{L},T) \prod_{(a,b)\in B} (1 - \mathbf{L}^{a}T^{b})^{-1}$$

with  $R(\mathbf{L}, T) \in \mathbb{Q}[\mathbf{L}, T]$ , and where  $B = B_{ar, \Lambda} \setminus \{(24, 22), (31, 28)\}$ .

#### References

- [Bittner 2004] F. Bittner, "The universal Euler characteristic for varieties of characteristic zero", *Compos. Math.* **140**:4 (2004), 1011–1032. MR 2005d:14031 Zbl 1086.14016
- [Cobo Pablos and González Pérez 2012] H. Cobo Pablos and P. D. González Pérez, "Motivic Poincaré series, toric singularities and logarithmic Jacobian ideals", J. Algebraic Geom. 21:3 (2012), 495–529. MR 2914802 Zbl 06051369
- [CoGP 2010] H. Cobo Pablos and P. D. González Pérez, "Geometric motivic Poincaré series of quasiordinary sigularities", *Math. Proc. Cambridge Philos. Soc.* 149:1 (2010), 49–74. MR 2011j:14032 Zbl 1198.14015
- [Denef 1984] J. Denef, "The rationality of the Poincaré series associated to the *p*-adic points on a variety", *Invent. Math.* **77**:1 (1984), 1–23. MR 86c:11043 Zbl 0537.12011

- [DL 1999] J. Denef and F. Loeser, "Germs of arcs on singular algebraic varieties and motivic integration", *Invent. Math.* **135**:1 (1999), 201–232. MR 99k:14002 Zbl 0928.14004
- [DL 2001] J. Denef and F. Loeser, "Definable sets, motives and *p*-adic integrals", *J. Amer. Math. Soc.* **14**:2 (2001), 429–469. MR 2002k:14033 Zbl 1040.14010
- [DL 2002] J. Denef and F. Loeser, "Motivic integration and the Grothendieck group of pseudo-finite fields", pp. 13–23 in *Proceedings of the International Congress of Mathematicians* (Beijing, 2002), vol. 2, edited by T. Li, Higher Education Press, Beijing, 2002. MR 2004f:14040 Zbl 1101.14029
- [DL 2004] J. Denef and F. Loeser, "On some rational generating series occurring in arithmetic geometry", pp. 509–526 in *Geometric aspects of Dwork theory* (Padova, 2001), vol. 1, edited by A. Adolphson et al., Walter de Gruyter, Berlin, 2004. MR 2005h:11267 Zbl 1061.11067
- [Ewald 1996] G. Ewald, *Combinatorial convexity and algebraic geometry*, Graduate Texts in Mathematics **168**, Springer, New York, 1996. MR 97i:52012 Zbl 0869.52001
- [Fulton 1993] W. Fulton, *Introduction to toric varieties*, Annals of Mathematics Studies **131**, Princeton University Press, 1993. MR 94g:14028 Zbl 0813.14039
- [Gel'fand et al. 1994] I. M. Gel'fand, M. M. Kapranov, and A. V. Zelevinsky, *Discriminants, resultants, and multidimensional determinants*, Birkhäuser, Boston, MA, 1994. MR 95e:14045 Zbl 0827.14036
- [Gillet and Soulé 1996] H. Gillet and C. Soulé, "Descent, motives and *K*-theory", *J. Reine Angew. Math.* **478** (1996), 127–176. MR 98d:14012 Zbl 0863.19002
- [Greenberg 1966] M. J. Greenberg, "Rational points in Henselian discrete valuation rings", *Inst. Hautes Études Sci. Publ. Math.* **31** (1966), 59–64. MR 34 #7515 Zbl 0146.42201
- [Guillén and Navarro Aznar 2002] F. Guillén and V. Navarro Aznar, "Un critère d'extension des foncteurs définis sur les schémas lisses", *Publ. Math. Inst. Hautes Études Sci.* 95 (2002), 1–91. MR 2004i:14020 Zbl 1075.14012
- [Ishii 2004] S. Ishii, "The arc space of a toric variety", *J. Algebra* **278**:2 (2004), 666–683. MR 2005f: 14100 Zbl 1073.14066
- [Ishii 2005] S. Ishii, "Arcs, valuations and the Nash map", J. Reine Angew. Math. 588 (2005), 71–92.
   MR 2006k:14005 Zbl 1082.14007
- [Lejeune-Jalabert and Reguera 2003] M. Lejeune-Jalabert and A. J. Reguera, "The Denef–Loeser series for toric surface singularities", *Rev. Mat. Iberoamericana* **19**:2 (2003), 581–612. MR 2004m: 14001 Zbl 1058.14006
- [Nicaise 2005a] J. Nicaise, "Arcs and resolution of singularities", *Manuscripta Math.* **116**:3 (2005), 297–322. MR 2006h:14019 Zbl 1073.14050
- [Nicaise 2005b] J. Nicaise, "Motivic generating series for toric surface singularities", *Math. Proc. Cambridge Philos. Soc.* **138**:3 (2005), 383–400. MR 2006d:14012 Zbl 1075.14020
- [Oda 1988] T. Oda, Convex bodies and algebraic geometry: An introduction to the theory of toric varieties, Ergebnisse der Mathematik und ihrer Grenzgebiete (3) 15, Springer, Berlin, 1988. MR 88m:14038 Zbl 0628.52002
- [Rond 2009] G. Rond, "Séries de Poincaré motiviques d'un germe d'hypersurface irréductible quasiordinaire", pp. 371–396 in *Présentation [Équations différentielles et singularités: En l'honneur de J. M. Aroca*], edited by F. Cano et al., Astérisque **323**, Société Mathématique de France, Paris, 2009. MR 2011c:14039 Zbl 1190.14008
- [Scholl 1994] A. J. Scholl, "Classical motives", pp. 163–187 in *Motives* (Seattle, WA, 1991), edited by U. Jannsen et al., Proc. Sympos. Pure Math. 55, American Mathematical Society, Providence, RI, 1994. MR 95b:11060 Zbl 0814.14001

# 430 Helena Cobo Pablos and Pedro Daniel González Pérez

Communicated by Joseph Received 2011-11-07	Gubelad Revised	lze 2012-01-31	Accepted 2012-03-03	
helena.cobo@wis.kuleuver	1.ac.be	Department of Celestijnenlaan	<sup>7</sup> Mathematics, Univers 200B, B-3001 Heverle	ity of Leuven, e, Belgium
pgonzalez@mat.ucm.es		Instituto de Ciencias Matemáticas, Departamento de Álgebra, Facultad de Ciencias Matemáticas, Universidad Complutense de Madrid, Plaza de las Ciencias 3, 28040 Madrid, Spain		





# Maximal ideals and representations of twisted forms of algebras

Michael Lau and Arturo Pianzola

Given a central simple algebra  $\mathfrak{g}$  and a Galois extension of base rings S/R, we show that the maximal ideals of twisted S/R-forms of the algebra of currents  $\mathfrak{g}(R)$  are in natural bijection with the maximal ideals of R. When  $\mathfrak{g}$  is a Lie algebra, we use this to give a complete classification of the finite-dimensional simple modules over twisted forms of  $\mathfrak{g}(R)$ .

#### 1. Introduction

Let S/R be a (finite) Galois extension of commutative, associative, and unital algebras over a field k, and let  $\mathfrak{g}$  be a finite-dimensional central simple k-algebra. Let  $\mathscr{L}$  be an S/R-form of  $\mathfrak{g} \otimes_k R$ , that is, an R-algebra  $\mathscr{L}$  such that

$$\mathscr{L} \otimes_R S \simeq \mathfrak{g} \otimes_k S \tag{1.1}$$

as algebras over S.

In this paper we accomplish two tasks:

(1) We establish a natural correspondence between the maximal ideals of  $\mathcal{L}$  and those of the base ring *R*.

(2) If  $\mathfrak{g}$  is a Lie algebra, *k* is algebraically closed of characteristic 0, and *R* is of finite type, we describe all the finite-dimensional irreducible modules of  $\mathcal{L}$  and classify them up to isomorphism.

In what follows, we will denote  $\mathfrak{g} \otimes_k S$  as  $\mathfrak{g}(S)$ . Recall that if  $\Gamma$  is the Galois group of S/R, then there is a natural correspondence between the set of isomorphism classes of S/R-forms of  $\mathfrak{g}(R) = \mathfrak{g} \otimes_k R$  and the pointed set of nonabelian Galois cohomology  $\mathrm{H}^1(\Gamma, \mathrm{Aut}_{S-\mathrm{alg}} \mathfrak{g}(S))$ . See [Knus and Ojanguren 1974], for example.

Funding from the Natural Sciences and Engineering Research Council of Canada is gratefully acknowledged.

MSC2010: primary 17B10; secondary 17B67, 12G05, 17A60.

*Keywords:* Galois descent, maximal ideals, finite-dimensional modules, multiloop algebras, twisted forms.

For example, consider the multiloop algebra  $\mathscr{L}(\mathfrak{g}, \sigma)$ , where  $\mathfrak{g}$  is a finite-dimensional Lie algebra over an algebraically closed field *k* of characteristic 0, and  $\sigma$  is an *N*-tuple of commuting automorphisms

$$\sigma_1,\ldots,\sigma_N: \mathfrak{g} \to \mathfrak{g}$$

of finite orders  $m_1, \ldots, m_N$ , respectively. This is a  $\mathbb{Z}^N$ -graded Lie subalgebra of the Lie algebra  $\mathfrak{g}(S)$ , where  $S = k[t_1^{\pm 1}, \ldots, t_N^{\pm 1}]$ :

$$\mathscr{L}(\mathfrak{g};\sigma) = \bigoplus_{j \in \mathbb{Z}^N} \mathfrak{g}_j \otimes t_1^{j_1} t_2^{j_2} \cdots t_N^{j_N},$$

where  $\mathfrak{g}_j = \{x \in \mathfrak{g} \mid \sigma_i(x) = \xi_i^{j_i} x \text{ for all } i\}$ , for fixed primitive  $m_i$ -th roots of unity  $\xi_i \in k$ . Then  $\mathscr{L}(\mathfrak{g}, \sigma)$  is an S/R-form of  $\mathfrak{g}(R)$ , where  $R = k[t_1^{\pm m_1}, \ldots, t_N^{\pm m_N}]$ . The Galois group  $\Gamma$  of S/R is  $\mathbb{Z}_{m_1} \times \cdots \times \mathbb{Z}_{m_N}$ , and the corresponding (constant) 1-cocycle in  $\mathrm{H}^1(\Gamma, \mathrm{Aut}_{S-\mathrm{alg}} \mathfrak{g}(S))$  is the group homomorphism taking a fixed generator  $\alpha_i$  of  $\mathbb{Z}_{m_i}$  to  $\sigma_i^{-1} \otimes 1$ . Such algebras play an important role in affine Kac–Moody, toroidal, and extended affine Lie theory.<sup>1</sup>

We open the paper with a detailed investigation of the maximal ideals of twisted forms  $\mathcal{L}^2$  Given any ideal  $\mathscr{I}$  of the *R*-algebra  $\mathscr{L}$ , we show that there is a unique  $\Gamma$ -stable ideal  $J(\mathscr{I}) \subseteq S$  for which  $\mathscr{I} \otimes_R S$  maps to  $\mathfrak{g} \otimes_k J(\mathscr{I})$  under the isomorphism  $\mathscr{L} \otimes_R S \to \mathfrak{g} \otimes_k S$ . As all maximal ideals  $\mathscr{I}$  of the *k*-algebra  $\mathscr{L}$  are *R*-stable, this produces a bijection  $\psi : \mathscr{I} \mapsto J(\mathscr{I}) \cap R$  between maximal ideals of the *k*-algebra  $\mathscr{L}$ and the set  $\operatorname{Max}(R)$  of maximal ideals of *R*. Explicitly,  $\psi^{-1} : I \mapsto I\mathscr{L}$  for maximal ideals  $I \subseteq R$ .

To have access to the attractive results of classical representation theory, we then assume that  $\mathfrak{g}$  is a finite-dimensional simple Lie algebra and R is of finite type over an algebraically closed field k of characteristic 0. The classification of finite-dimensional simple  $\mathscr{L}$ -modules V proceeds by observing that the kernel of the representation  $\phi : \mathscr{L} \to \operatorname{End}_k(V)$  is an intersection of a finite collection of distinct maximal ideals  $\mathscr{I}_1, \ldots, \mathscr{I}_n \subseteq \mathscr{L}$ . Given any maximal ideals  $M_1, \ldots, M_n \in \operatorname{Max}(S)$  lying over the maximal ideals  $\psi(\mathscr{I}_1), \ldots, \psi(\mathscr{I}_n) \in \operatorname{Max}(R)$ , respectively, we obtain evaluation maps

$$\operatorname{ev}_M : \mathscr{L} \hookrightarrow \mathfrak{g} \otimes_k S \to (\mathfrak{g} \otimes_k S/M_1) \oplus \cdots \oplus (\mathfrak{g} \otimes_k S/M_n) \simeq \mathfrak{g}^{\oplus n}.$$

We then use properties of forms to show that  $ev_M$  is surjective and descends to an isomorphism  $ev_M : \mathcal{L} / \ker \phi \xrightarrow{\sim} \mathfrak{g}^{\oplus n}$ . The finite-dimensional simple  $\mathcal{L}$ -modules *V* 

<sup>&</sup>lt;sup>1</sup>For simplicity of notation, we use integral powers of the variables  $t_i$ , though fractional exponents are sometimes used to work with the absolute Galois group of the base ring *R* or with twisted modules for vertex algebras.

<sup>&</sup>lt;sup>2</sup>Throughout this paper, all ideals are assumed to be two-sided unless there is an explicit mention to the contrary.

are thus pullbacks of tensor products of  $\mathfrak{g}$ -modules along  $ev_M$ :

$$V \simeq V(\lambda, M) = V_{\lambda_1}(M_1) \otimes_k \cdots \otimes_k V_{\lambda_n}(M_n)$$

for some nonzero dominant integral highest weights  $\lambda_1, \ldots, \lambda_n$  of  $\mathfrak{g}$  (relative to a triangular decomposition  $\mathfrak{g} = \mathfrak{n}_- \oplus \mathfrak{h} \oplus \mathfrak{n}_+$ ) and maximal ideals  $M_1, \ldots, M_n \in \operatorname{Max}(S)$ , where  $V_{\lambda_i}(M_i)$  is the simple  $\mathfrak{g}$ -module of highest weight  $\lambda_i$ , viewed as an  $\mathscr{L}$ -module via the composition of maps

$$\mathscr{L} \xrightarrow{\operatorname{CV}_{M_i}} \mathfrak{g} \otimes_k S/M_i \simeq \mathfrak{g} \to \operatorname{End}(V_{\lambda_i}).$$

Two such representations  $V(\lambda, M) = V_{\lambda_1}(M_1) \otimes_k \cdots \otimes_k V_{\lambda_m}(M_m)$  and  $V(\mu, N) = V_{\mu_1}(N_1) \otimes_k \cdots \otimes_k V_{\mu_n}(N_n)$  are isomorphic  $(\mathcal{L}/\ker \phi)$ -modules, and thus isomorphic  $\mathcal{L}$ -modules, if and only if their highest weights are equal, relative to the induced triangular decomposition

$$\mathscr{L}/\ker\phi = \operatorname{ev}_M^{-1}(\mathfrak{n}_-^{\oplus n}) \oplus \operatorname{ev}_M^{-1}(\mathfrak{h}^{\oplus n}) \oplus \operatorname{ev}_M^{-1}(\mathfrak{n}_+^{\oplus n}).$$

The cohomological interpretation of forms leads to an action of the group  $\Gamma$  on  $P_+ \times \text{Max}(S)$ , for which  $V(\lambda, M) \simeq V(\mu, N)$  if and only if m = n and

$$(\lambda_i, M_i) = {}^{\gamma_i}(\mu_i, N_i)$$

for some  $\gamma_1, \ldots, \gamma_n \in \Gamma$ . This classification (Proposition 3.7) is then described in terms of  $\Gamma$ -invariant functions from the maximal spectrum Max(*S*) to the set  $P_+$  of dominant integral weights. This gives a constructive description (Theorem 3.9) of the moduli space of finite-dimensional simple  $\mathscr{L}$ -modules in terms of finitely supported  $\Gamma$ -invariant functions Max(*S*)  $\rightarrow P_+$ .

One of our main motivations in the present paper was to generalize and provide more intuitive proofs of previous work on (twisted) loop and multiloop algebras. See [Lau 2010; Senesi 2010] for a summary of past work on this problem. However, the interpretation of isomorphism classes as spaces of  $\Gamma$ -equivariant maps used in past work does not generalize to our context of twisted forms. Instead, the  $\Gamma$ -equivariant functions had to be reinterpreted as  $\Gamma$ -invariant functions Max $(S) \rightarrow P_+$ . This turned out to be the correct perspective to include cases where there is no natural action of  $\Gamma$  on the space  $P_+^{\times}$  of nonzero dominant integral weights. More significantly, with new proofs, we have eliminated all dependence on the  $\mathbb{Z}^N$ -grading of  $\mathcal{L}(\mathfrak{g}, \sigma)$ , a point that was crucial in the arguments of [Lau 2010]. This lets us apply our work to nongraded contexts, including a classification of modules for the mysterious Margaux algebras explained in Section 4.

Perhaps the most striking feature of the present work is its nearly complete independence from the particular S/R-form under consideration. The maximal

ideals of any S/R-form  $\mathscr{L}$  of  $\mathfrak{g}(R)$  are in bijection with Max(R), and the finitedimensional simple  $\mathscr{L}$ -modules are evaluation modules enumerated by finitely supported  $\Gamma$ -invariant maps  $Max(S) \rightarrow P_+$ . Indeed, the only place where the Galois cocycle (and hence the isomorphism class) of the S/R-form plays an explicit role is in the isomorphism criterion for  $\mathscr{L}$ -modules (Proposition 3.7). But in many interesting examples, even this condition vanishes, as we illustrate in Section 4.

*Notation.* Throughout this paper, k will denote a field. We let  $k^{\times} = k \setminus \{0\}$  and denote the set of nonnegative integers by  $\mathbb{Z}_+$ . The category of finitely generated unital commutative associative k-algebras will be denoted by k-alg, and we will write Max(S) for the maximal spectrum of each  $S \in k$ -alg.

#### 2. Twisted forms and their maximal ideals

In this section, k will denote an arbitrary field and S/R will be a finite Galois extension in k-alg with Galois group  $\Gamma$ . Let  $\mathfrak{g}$  be a finite-dimensional central simple algebra over k, and let  $R \in k$ -alg. We may view  $\mathfrak{g}(R) \cong \mathfrak{g} \otimes_k R$  as an algebra over R by base change, the multiplication given by  $(x \otimes r)(y \otimes s) = xy \otimes rs$  (for each  $x, y \in \mathfrak{g}$  and  $r, s \in R$ ). As before,  $\mathscr{L}$  will denote an S/R-form of  $\mathfrak{g}(R)$ . Any such  $\mathscr{L}$  is obviously an algebra over k by restriction of scalars, and we may thus speak of k-ideals and R-ideals of  $\mathscr{L}$ , namely the ideals of  $\mathscr{L}$  viewed as an algebra over k and over R, respectively.<sup>3</sup> The goal of this section is to classify the maximal k-ideals of  $\mathscr{L}$ .

Since Galois extensions are faithfully flat, we have the following general facts. See [Matsumura 1989, Theorem 7.5], for instance.

**Lemma 2.1.** Let I be an ideal of R, and let M be an R-module.

(1) The canonical map

$$M \to M \otimes_R S, \quad x \mapsto x \otimes 1$$

is injective. In particular, R can be identified with a k-subalgebra of S.

(2) After viewing R inside of S via (1), IS is an ideal of S and  $R \cap IS = I$ .

Up to coboundary, we can associate a Galois 1-cocycle

$$u = (u_{\gamma})_{\gamma \in \Gamma} \in Z^{1}(\Gamma, \operatorname{Aut}_{S-\operatorname{alg}}(\mathfrak{g}(S)))$$

to  $\mathcal{L}$ , such that  $\mathcal{L} \simeq \mathcal{L}_u = \{z \in \mathfrak{g} \otimes_k S \mid u_{\gamma}^{\gamma} z = z \text{ for all } \gamma \in \Gamma\}$ . We therefore can (and henceforth will) view  $\mathcal{L}$  as an *R*-subalgebra of  $\mathfrak{g}(S) = \mathfrak{g} \otimes_k S$ . Note that the *S*-algebra isomorphism

$$\mathscr{L} \otimes_R S \simeq \mathfrak{g}(R) \otimes_R S = \mathfrak{g}(S)$$

<sup>&</sup>lt;sup>3</sup>We remind the reader that the word *ideal* means two-sided ideal.

may be realized as the multiplication map

$$\mu: \mathscr{L} \otimes_R S \to \mathfrak{g}(S), \quad \left(\sum_i x_i \otimes s_i\right) \otimes s \mapsto \sum_i x_i \otimes s_i s \tag{2.2}$$

for all  $\sum_i x_i \otimes s_i \in \mathcal{L}$  and  $s \in S$ . This will allow us to associate an ideal of *S* to every *R*-ideal of  $\mathcal{L}$ .

**Lemma 2.3.** Let  $\mathscr{G}$  be an *R*-ideal of  $\mathscr{L}$ . Then  $\mathscr{G} \otimes_R S$  is an *S*-ideal of  $\mathscr{L} \otimes_R S$ , and there is a unique ideal  $J = J(\mathscr{G}) \subseteq S$  such that  $\mathfrak{g} \otimes_k J = \mu(\mathscr{G} \otimes_R S)$ .

*Proof.* Fix a *k*-basis  $\{x_1, \ldots, x_m\}$  of  $\mathfrak{g}$ . Let  $J = J(\mathfrak{I})$  be the set of all  $s \in S$  for which there exists  $\sum_{i=1}^m x_i \otimes s_i \in \mu(\mathfrak{I} \otimes_R S)$  such that  $s = s_i$  for some *i*. By the definition of *J*, it is clear that  $\mu(\mathfrak{I} \otimes_R S) \subseteq \mathfrak{g} \otimes_k J$ . Moreover, since  $\mathfrak{g} \otimes 1 \subseteq \mathfrak{g} \otimes_k S$  is a finite-dimensional central simple *k*-algebra, it follows from the Jacobson density theorem that  $x_i \otimes s \in \mu(\mathfrak{I} \otimes_R S)$  for all  $s \in J$  and for all  $i \leq m$ . Thus  $\mathfrak{g} \otimes_k J \subseteq \mu(\mathfrak{I} \otimes_R S)$ . The uniqueness of *J* is clear since the tensor product  $\mathfrak{g} \otimes_k J$  is being taken over a field *k*.

**Proposition 2.4.** Let  $\mathscr{I}_1$  and  $\mathscr{I}_2$  be *R*-ideals of  $\mathscr{L}$ . Then  $J(\mathscr{I}_1) \subseteq J(\mathscr{I}_2)$  if and only if  $\mathscr{I}_1 \subseteq \mathscr{I}_2$ . In particular, the map  $J : \{R\text{-ideals of } \mathscr{L}\} \to \{\text{ideals of } S\}$  is injective.

*Proof.* Let  $\mathcal{I} = \mathcal{I}_1 + \mathcal{I}_2$ . The restriction of the multiplication map

$$\mu: \mathcal{L} \otimes_R S \to \mathfrak{g}(S)$$

to  $\mathscr{I} \otimes_R S$  gives an isomorphism  $\mu_{\mathscr{I}} : \mathscr{I} \otimes_R S \to \mathfrak{g} \otimes_k J(\mathscr{I})$  with  $J(\mathscr{I}) = J(\mathscr{I}_1) + J(\mathscr{I}_2)$ . By flatness of S/R,

$$(\mathscr{I}/\mathscr{I}_2) \otimes_R S \simeq \frac{\mathscr{I} \otimes_R S}{\mathscr{I}_2 \otimes_R S}$$

as S-modules. The injection  $\mu_{\mathcal{I}}$  restricts to an isomorphism

$$\mathcal{I}_2 \otimes_R S \to \mathfrak{g} \otimes_k J(\mathfrak{I}_2),$$

so we see that

$$\frac{\mathscr{G}\otimes_R S}{\mathscr{G}_2\otimes_R S} \simeq \frac{\mathfrak{g}\otimes_k J(\mathscr{G})}{\mathfrak{g}\otimes_k J(\mathscr{G}_2)} = \mathfrak{g}\otimes_k (J(\mathscr{G})/J(\mathscr{G}_2)).$$

Thus  $(\mathscr{I}/\mathscr{I}_2) \otimes_R S = 0$  if and only if  $\mathfrak{g} \otimes_k (J(\mathscr{I})/J(\mathscr{I}_2)) = 0$ ; then by faithful flatness,  $\mathscr{I}/\mathscr{I}_2 = 0$  if and only if  $J(\mathscr{I})/J(\mathscr{I}_2) = 0$ . That is,  $\mathscr{I}_1 \subseteq \mathscr{I}_2$  if and only if  $J(\mathscr{I}_1) \subseteq J(\mathscr{I}_2)$ .

**Proposition 2.5.** Let  $\mathfrak{I} \subseteq \mathfrak{L}$  be an *R*-ideal. Then  $J(\mathfrak{I})$  is stable under the action of the Galois group  $\Gamma = \mathfrak{Gal}(S/R)$ .

*Proof.* As in the proof of Lemma 2.3, we fix a *k*-basis  $\beta = \{x_1, \ldots, x_m\}$  of  $\mathfrak{g}$ . From the definition of  $J = J(\mathfrak{F})$ , it is easy to see that *J* is the ideal of *S* generated by the set  $E_{\beta}(\mathfrak{F})$  of those elements  $s \in S$  for which there is an element  $\sum_i x_i \otimes s_i \in \mathfrak{F}$  for which  $s_i = s$  for some *i*. It is thus enough to show  $\gamma s \in J$  for all  $\gamma \in \Gamma$  and  $s \in E_{\beta}(\mathfrak{F})$ .

Let  $u \in Z^1(\Gamma, \operatorname{Aut}_{S-\operatorname{alg}}(\mathfrak{g}(S)))$  be a cocycle corresponding to the S/R-form  $\mathscr{L}$ . Fix  $\gamma \in \Gamma$ , and write  $u_{\gamma}(x_i \otimes 1) = \sum_{j=1}^m x_j \otimes a_{ij}$ . Since  $u_{\gamma}$  is an automorphism of  $\mathfrak{g}(S)$ , the matrix  $A = (a_{ij})$  is invertible in  $M_m(S)$ . Let  $z = \sum x_i \otimes s_i \in \mathscr{I}$ . It suffices to show that  $\gamma s_i \in J$  for  $i = 1, \ldots, m$ . We have

$$\sum x_i \otimes s_i = \mu(z \otimes 1) = \mu(u_{\gamma}^{\gamma} z \otimes 1) = \mu\left(\sum_i u_{\gamma}(x_i \otimes \gamma s_i) \otimes 1\right)$$
$$= \mu\left(\sum_i \gamma s_i u_{\gamma}(x_i \otimes 1) \otimes 1\right) = \mu\left(\sum_i u_{\gamma}(x_i \otimes 1) \otimes \gamma s_i\right)$$
$$= \mu\left(\sum_{i,j} x_j \otimes a_{ij} \otimes \gamma s_i\right) = \sum_j x_j \otimes \left(\sum_i a_{ij}^{\gamma} s_i\right).$$

In matrix form, we see that

$$\begin{pmatrix} \gamma_{s_1} \\ \vdots \\ \gamma_{s_m} \end{pmatrix} = (A^t)^{-1} \begin{pmatrix} s_1 \\ \vdots \\ s_m \end{pmatrix}$$

By definition,  $s_i \in E_\beta(\mathcal{I}) \subseteq J$  for all i, and  $(A^t)^{-1} \in M_m(S)$ . Hence  $\gamma s_i \in J$  for all i.

**Lemma 2.6.** Let I be an ideal of R. Then  $I\mathcal{L}$  is an ideal of  $\mathcal{L}$ , and  $J(I\mathcal{L}) = IS$ .

*Proof.* It is obvious that  $I\mathscr{L}$  is an ideal of  $\mathscr{L}$ . As S-modules (in fact, as S-algebras),

$$I\mathscr{L} \otimes_R S = \mathscr{L} \otimes_R IS \simeq \mathscr{L} \otimes_R S \otimes_S IS \simeq \mathfrak{g} \otimes_k S \otimes_S IS \simeq \mathfrak{g} \otimes_k IS$$

so  $J(I\mathcal{L}) = IS$ .

We now turn to the classification of maximal k-ideals  $\mathcal{I}$  of the S/R-form  $\mathcal{L}$ .

#### **Lemma 2.7.** The sets of maximal k-ideals and maximal R-ideals of $\mathcal{L}$ coincide.

*Proof.* Let  $\mathscr{I}$  be a maximal k-ideal of  $\mathscr{L}$ . We claim that  $\mathscr{I}$  is stable under the action of R. For any  $r \in R$ , the space  $r\mathscr{I}$  is clearly a k-ideal of  $\mathscr{L}$ , and if  $r\mathscr{I} \not\subseteq \mathscr{I}$ , then  $\mathscr{I} + r\mathscr{I} = \mathscr{L}$  by the maximality of  $\mathscr{I}$ . The algebra  $\mathscr{L}$  is perfect by descent considerations, as has already been noted in [Gille and Pianzola 2007], for instance. Thus

$$\mathcal{L} = \mathcal{L}\mathcal{L} = (\mathcal{I} + r\mathcal{I})\mathcal{L} = \mathcal{I}\mathcal{L} + \mathcal{I}(r\mathcal{L}) \subseteq \mathcal{I}\mathcal{L} \subseteq \mathcal{I},$$

since  $\mathcal{L}$  is an *R*-algebra. But this contradicts the proper inclusion  $\mathscr{I} \subsetneq \mathscr{L}$ , so  $r\mathscr{I} \subseteq \mathscr{I}$  as claimed. From this, it follows that every maximal *k*-ideal of  $\mathscr{L}$  is also a maximal *R*-ideal of  $\mathscr{L}$  and conversely.

#### **Lemma 2.8.** Let M be a maximal ideal of R.

- (1) There exist prime ideals of S lying over M, and any such ideal is maximal. The group  $\Gamma$  acts transitively on the set of such maximal ideals. In particular, this set is finite.
- (2)  $MS = \bigcap_i M_i$ , where the intersection is taken over the (finite) set of maximal ideals of S lying over M.

*Proof.* (1) This is well-known, but we recall the main ideas for completeness. From basic properties of Galois extensions, we know that  $R = S^{\Gamma}$ , and hence S/R is integral. From this it follows that the set of prime ideals of *S* lying over *M* is not empty, that any such ideal is maximal, and that the action of  $\Gamma$  on this set is transitive. (See [Bourbaki 1964, §2.1 Proposition 1 and §2.2 Théorème 2].)

(2) Any maximal ideal  $\mathfrak{m}$  of S containing MS will lie over M, since the intersection  $\mathfrak{m} \cap R$  is a proper ideal of R containing  $MS \cap R$ , which is equal to the maximal ideal M by Lemma 2.1(2). Thus  $\mathfrak{m} = M_i$  for some i, and  $\bigcap_i M_i$  is the radical of MS. Since S/R is flat,

$$S/MS \simeq (R \otimes_R S)/(M \otimes_R S) \simeq (R/M) \otimes_R S.$$

Let L = R/M, a field extension of k. Since the extension S is Galois over R, general facts about base change guarantee that the extension  $(R/M) \otimes_R S$  is Galois over  $(R/M) \otimes_R R \simeq L$ . (See [Milne 1980, §I.5], for instance.) That is, S/MSis a Galois extension of L. Galois extensions are finite étale and the only such extensions of L are products  $L_1 \times \cdots \times L_m$ , where the  $L_i$  are finite separable field extensions of L. We see from this that S/MS has trivial Jacobson radical. Hence MS is a radical ideal of S, and  $MS = \bigcap_i M_i$ .

**Theorem 2.9.** The map  $\psi : I \mapsto I \mathcal{L}$  defines a bijection between the set of maximal ideals of *R* and the set of maximal ideals of  $\mathcal{L}$ .

*Proof.* Let  $\mathscr{I}$  be a maximal ideal of  $\mathscr{L}$ , and let  $J = J(\mathscr{I}) \subseteq S$  be the ideal corresponding to  $\mathscr{I}$ . Let  $P \subseteq S$  be a maximal ideal containing J, and let  $M = P \cap R$ . Since S/R is integral, M is a maximal ideal of R [Bourbaki 1964, §2.1 Proposition 1].

As explained in Lemma 2.8(1), the Galois group  $\Gamma$  acts transitively on the finite set  $M_1, \ldots, M_N$  of maximal ideals *S* lying over *M*. Since *J* is  $\Gamma$ -stable (Proposition 2.5) and contained in a maximal ideal *P* lying over *M*, we see that  $J \subseteq \bigcap_{i=1}^{N} M_i$ . By Lemma 2.8(2),  $MS = \bigcap_{i=1}^{N} M_i$ . Hence  $J \subseteq MS$ .

Note that  $M\mathcal{L}$  is an ideal of  $\mathcal{L}$  whose corresponding ideal is MS, by Lemma 2.6. By Proposition 2.4,  $\mathcal{I} \subseteq M\mathcal{L}$ . Since  $MS = \bigcap_{i=1}^{N} M_i$  is a proper ideal of S, Lemma 2.3 guarantees that  $M\mathcal{L}$  is a proper ideal of  $\mathcal{L}$ . Hence  $\mathcal{I} = M\mathcal{L}$  by the maximality of  $\mathcal{I}$ , so the image of the map  $\psi$  includes all maximal ideals of  $\mathcal{L}$ .

Let  $I_1$  and  $I_2$  be maximal ideals of R. If  $I_1 \mathcal{L} = I_2 \mathcal{L}$ , then  $I_1 S = I_2 S$  by Proposition 2.4 and Lemma 2.6. Now Lemma 2.1(2) yields that  $I_1 = I_2$ , hence that  $\psi$  is injective. It remains only to check that  $I \mathcal{L} \subseteq \mathcal{L}$  is maximal whenever  $I \subseteq R$ is maximal. Suppose that  $I \subseteq R$  is a maximal ideal, and let  $\mathcal{I} \subseteq \mathcal{L}$  be a maximal ideal containing  $I \mathcal{L}$ . We have already shown that there is a maximal ideal  $M \subseteq R$ for which  $\mathcal{I} = M \mathcal{L}$ . By Lemma 2.1(2) and Lemma 2.6,

$$M = MS \cap R = J(M\mathcal{L}) \cap R = J(\mathcal{I}) \cap R.$$

By Proposition 2.4,  $J(I\mathcal{L}) \subseteq J(\mathcal{I})$ , so

$$I = IS \cap R = J(I\mathscr{L}) \cap R \subseteq J(\mathscr{I}) \cap R = M.$$

By the maximality of *I*, we see that I = M. Hence  $I \mathscr{L} = M \mathscr{L} = \mathscr{I}$  is a maximal ideal of  $\mathscr{L}$ .

As an application, we recover the following well-known fact; see [Knus and Ojanguren 1974, Corollary III.5.2].

**Corollary 2.10.** *Let A be an Azumaya algebra over R. Every (two-sided) maximal ideal of A is of the form I A for some maximal ideal I of R.* 

#### 3. Classification of simple modules

We maintain the notation of the previous section but now assume that  $\mathfrak{g}$  is a finitedimensional simple Lie algebra over an algebraically closed field k of characteristic zero. The base ring R will be of finite type in k-alg, and all modules (representations) will be of finite dimension over k. Unless explicitly indicated otherwise,  $\otimes$  will denote a tensor product  $\otimes_k$  taken over the base field k.

Let  $\mathscr{L} \subset \mathfrak{g} \otimes S$  be an S/R-form of  $\mathfrak{g}(R)$  as before, and let  $\phi : \mathscr{L} \to \operatorname{End}_k(V)$ be a finite-dimensional irreducible representation of  $\mathscr{L}$ . We fix a cocycle  $u \in Z^1(\Gamma, \operatorname{Aut}_{S-\operatorname{Lie}}(\mathfrak{g}(S)))$  so that  $\mathscr{L} = \mathscr{L}_u$ .

**3a.** *Evaluation maps and simple modules.* Since  $\mathcal{L}$  is perfect,  $\mathcal{L}/\ker \phi$  is a finitedimensional semisimple Lie algebra over k [Lau 2010, Proposition 2.1]. Hence there is an isomorphism

$$f: \mathscr{L}/\ker \phi \to \mathfrak{g}_1 \oplus \cdots \oplus \mathfrak{g}_n$$

for some collection of finite-dimensional simple *k*-Lie algebras  $\mathfrak{g}_1, \ldots, \mathfrak{g}_n$ . Let  $\pi : \mathcal{L} \to \mathcal{L}/\ker \phi$  be the natural projection. Then

$$\mathscr{L}/\ker\phi\simeq \mathscr{L}/\mathscr{M}_1\oplus\cdots\oplus \mathscr{L}/\mathscr{M}_n,$$

where  $\mathcal{M}_1, \ldots, \mathcal{M}_n$  are pairwise distinct maximal ideals of  $\mathcal{L}$  whose intersection is ker  $\phi$ . More precisely, we can take

$$\mathcal{M}_i = \pi^{-1} \circ f^{-1}(\mathfrak{g}_1 \oplus \cdots \oplus \widehat{\mathfrak{g}_i} \oplus \cdots \oplus \mathfrak{g}_n)$$

for i = 1, ..., n, where  $\widehat{\mathfrak{g}}_i$  indicates that the *i*-th summand is omitted. To classify the simple modules of  $\mathscr{L}$ , it thus suffices to consider quotients of  $\mathscr{L}$  by maximal ideals.<sup>4</sup>

Let  $\mathscr{I} \subseteq \mathscr{L}$  be a maximal ideal. By Theorem 2.9,  $\mathscr{I} = I\mathscr{L}$  for some maximal ideal  $I \subseteq R$ . Let  $P \subseteq S$  be a maximal ideal lying over I, and let

$$\epsilon: S \to S/P \simeq k \tag{3.1}$$

be the natural evaluation map.<sup>5</sup> Then the composition

$$\operatorname{ev}_P: \mathscr{L} \hookrightarrow \mathfrak{g} \otimes S \xrightarrow{1 \otimes \epsilon} \mathfrak{g} \otimes k \simeq \mathfrak{g}$$
 (3.2)

is a homomorphism of k-Lie algebras.

**Proposition 3.3.** The map  $ev_P : \mathcal{L} \to \mathfrak{g}$  is surjective and has kernel  $\mathcal{I} = (P \cap R)\mathcal{L}$ .

*Proof.* The multiplication map  $\mu : \mathcal{L} \otimes_R S \to \mathfrak{g}(S)$  is an isomorphism (2.2), so given any element  $x \in \mathfrak{g}$ , there exist elements  $z_i \in \mathcal{L}$  and  $t_i \in S$  such that

$$\mu\Big(\sum_i z_i \otimes t_i\Big) = x \otimes 1.$$

That is, if  $z_i = \sum_j x_j \otimes s_{ij}$  for some k-basis  $\{x_j\}$  of  $\mathfrak{g}$  and  $s_{ij} \in S$ , then

$$\sum_{i,j} x_j \otimes s_{ij} t_i = x \otimes 1.$$

Applying the map  $1 \otimes \epsilon$  introduced in (3.1), we get  $\sum_{i,j} x_j \otimes \epsilon(s_{ij}) \epsilon(t_i) = x \otimes 1$ . But  $\mathcal{L}$  is closed under multiplication by elements of k, so  $\sum_i \epsilon(t_i) z_i \in \mathcal{L}$ , and

$$\operatorname{ev}_P\left(\sum_i \epsilon(t_i) z_i\right) = \sum_{i,j} x_j \epsilon(s_{ij}) \epsilon(t_i) = x.$$

Hence  $ev_P$  is surjective.

Let  $z = \sum_i x_i \otimes s_i \in \mathcal{L}$  and  $r \in I$ . Then  $\epsilon(r) = 0$ , since  $I = P \cap R \subseteq P = \ker \epsilon$ . Hence

$$\operatorname{ev}_P(rz) = \sum x_i \epsilon(rs_i) = \sum x_i \epsilon(r) \epsilon(s_i) = 0,$$

<sup>&</sup>lt;sup>4</sup>Recall that there is no difference in the concept of maximal ideal if we view  $\mathcal{L}$  as an *R*- or *k*-Lie algebra.

<sup>&</sup>lt;sup>5</sup>*S* is of finite type over *R* and *R* is assumed to be of finite type over *k*. Thus *S* is of finite type over *k* and therefore  $S/P \simeq k$  by the Nullstellensatz.

so  $I\mathscr{L} \subseteq \ker \operatorname{ev}_P$ . Since  $\mathscr{I} = I\mathscr{L}$  is a maximal ideal and  $\operatorname{ev}_P$  is nonzero, the kernel of  $\operatorname{ev}_P$  is precisely  $\mathscr{I}$ .

We have now shown that  $\mathcal{L}/\ker \phi$  is isomorphic to a direct sum of finitely many copies of  $\mathfrak{g}$ . Explicitly,  $\ker \phi$  is the intersection of a (finite) family of distinct maximal ideals  $\mathcal{M}_1, \ldots, \mathcal{M}_n$  in  $\mathcal{L}$ . Let  $I_1, \ldots, I_n$  be the (distinct) maximal ideals of R given by Theorem 2.9. For any collection  $\underline{M}$  of maximal ideals  $M_1, \ldots, M_n$  of S lying over  $I_1, \ldots, I_n$ , respectively, the map

$$\operatorname{ev}_{\underline{M}} = (\operatorname{ev}_{M_1}, \dots, \operatorname{ev}_{M_n}) : \mathcal{L} \to \mathfrak{g} \oplus \dots \oplus \mathfrak{g},$$
  
 $z \mapsto (\operatorname{ev}_{M_1}(z), \dots, \operatorname{ev}_{M_n}(z))$ 

descends to an isomorphism  $ev_M : \mathscr{L} / \ker \phi \to \mathfrak{g} \oplus \cdots \oplus \mathfrak{g}$ .

Since the irreducible representations of  $\mathfrak{g}^{\oplus n} = \mathfrak{g} \oplus \cdots \oplus \mathfrak{g}$  are precisely the tensor products

$$\rho = (\rho_1, \dots, \rho_n) : \mathfrak{g} \oplus \dots \oplus \mathfrak{g} \to \operatorname{End}_k(V_1 \otimes \dots \otimes V_n),$$
$$(x_1, \dots, x_n) \mapsto \sum_{i=1}^n \operatorname{id} \otimes \dots \otimes \rho_i(x_i) \otimes \dots \otimes \operatorname{id}$$

of simple g-modules  $(\rho_i, V_i)$ , we now have a complete list of the simple  $\mathscr{L}$ -modules.

**Theorem 3.4.** Let  $\phi : \mathcal{L} \to End_k(V)$  be a finite-dimensional irreducible representation of  $\mathcal{L}$ . Then there exists a finite collection  $\underline{P} = (P_1, \ldots, P_n)$  of maximal ideals of S with  $P_i \cap R \neq P_j \cap R$  for  $i \neq j$ , and a simple  $\mathfrak{g}^{\oplus n}$ -module  $(\rho, V_1 \otimes \cdots \otimes V_n)$ such that  $V \simeq V_1 \otimes \cdots \otimes V_n$  and  $\phi = \rho \circ ev_P$ .

**Remark 3.5.** The converse of Theorem 3.4 is obvious. Given a collection of maximal ideals  $P_1, \ldots, P_n$  of S for which the ideals  $P_i \cap R$  of R are pairwise distinct, the Chinese remainder theorem gives an isomorphism

$$\mathscr{L}/\mathscr{M}_1 \oplus \cdots \oplus \mathscr{L}/\mathscr{M}_n \simeq \mathscr{L}/\cap_i \mathscr{M}_i,$$

where  $\mathcal{M}_i = (P_i \cap R)\mathcal{L}$ . (This uses the fact that the  $P_i \cap R$  are maximal, as shown in the proof of Theorem 2.9.) Thus the map

$$\mathscr{L} \to \mathscr{L}/\mathscr{M}_1 \oplus \cdots \oplus \mathscr{L}/\mathscr{M}_n \simeq \mathfrak{g}^{\oplus n}$$

is surjective, so the pullback of any simple  $\mathfrak{g}^{\oplus n}$ -module  $V = V_1 \otimes \cdots \otimes V_n$  will be a simple  $\mathscr{L}$ -module.

**3b.** *Isomorphism classes of simple modules.* Fix a Cartan subalgebra  $\mathfrak{h}$  of  $\mathfrak{g}$  and an épinglage of  $(\mathfrak{g}, \mathfrak{h})$ ; see [Bourbaki 1975, VIII, §4.1]. Given a maximal ideal  $M \in Max(S)$  and a finite-dimensional representation  $\rho : \mathfrak{g} \to End_k(W)$ , we write

W(M) for the vector space W, viewed as an  $\mathcal{L}$ -module with action given by the composition of maps

$$\mathscr{L} \hookrightarrow \mathfrak{g} \otimes S \xrightarrow{\operatorname{ev}_M} \mathfrak{g} \xrightarrow{\rho} \operatorname{End}_k(W),$$

where  $ev_M$  is the quotient map

$$\operatorname{ev}_M : \mathfrak{g} \otimes S \to (\mathfrak{g} \otimes S)/(\mathfrak{g} \otimes M) = \mathfrak{g} \otimes (S/M) \simeq \mathfrak{g},$$
  
 $x \otimes s \mapsto (x \otimes s)(M) = s(M)x$ 

for all  $x \in \mathfrak{g}$  and  $s \in S$ . For each automorphism  $\alpha \in \operatorname{Aut}_{S-\operatorname{Lie}}(\mathfrak{g}(S))$  and  $M \in \operatorname{Max}(S)$ , we write  $\alpha(M) \in \operatorname{Aut}(\mathfrak{g})$  for the automorphism defined by

$$(\alpha(M))(x) = (\alpha(x \otimes 1))(M) = \operatorname{ev}_M(\alpha(x \otimes 1))$$

for each  $x \in \mathfrak{g}$ . It is straightforward to verify that the map

$$\operatorname{Aut}_{S-\operatorname{Lie}}(\mathfrak{g}(S)) \to \operatorname{Aut}(\mathfrak{g}), \quad \alpha \mapsto \alpha(M)$$

is a group homomorphism for each  $M \in Max(S)$ . We write  $Out \alpha(M)$  and  $Int \alpha(M)$  for the outer and inner parts, respectively, of the automorphism

$$\alpha(M) = \operatorname{Int} \alpha(M) \circ \operatorname{Out} \alpha(M).$$

See [Bourbaki 1975, VIII, §5.3 corollaire 1] for details.

By Theorem 3.4, the (finite-dimensional) simple  $\mathcal{L}$ -modules are those of the form  $V(\lambda, M) = V_{\lambda_1}(M_1) \otimes \cdots \otimes V_{\lambda_n}(M_n)$ , where each  $\lambda_i$  is in the set  $P_+^{\times}$  of nonzero dominant integral weights,  $V_{\lambda_i}$  is the simple g-module of highest weight  $\lambda_i$ , and  $M = (M_1, \ldots, M_n)$  is an *n*-tuple of maximal ideals of *S* lying over distinct (closed) points of Spec(*R*).

**Lemma 3.6.** Suppose that the  $\mathscr{L}$ -modules  $V(\lambda, M) = V_{\lambda_1}(M_1) \otimes \cdots \otimes V_{\lambda_m}(M_m)$ and  $V(\mu, N) = V_{\mu_1}(N_1) \otimes \cdots \otimes V_{\mu_n}(N_n)$  are isomorphic for certain  $\lambda_1, \ldots, \lambda_m$ ,  $\mu_1, \ldots, \mu_n \in P_+^{\times}$  and  $M_1, \ldots, M_m, N_1, \ldots, N_n \in Max(S)$ . Then m = n, and up to reordering,  $M_i \cap R = N_i \cap R$  for all i.

*Proof.* Let  $\phi_{\lambda,M} : \mathcal{L} \to \text{End}_k(V(\lambda, M))$  and  $\phi_{\mu,N} : \mathcal{L} \to \text{End}_k(V(\mu, N))$  be the homomorphisms determining the module actions. Since  $V(\lambda, M) \simeq V(\mu, N)$ , their kernels are equal, so

$$\bigcap_{i=1}^{m} (M_i \cap R) \mathscr{L} = \ker \phi_{\lambda,M} = \ker \phi_{\mu,N} = \bigcap_{j=1}^{n} (N_j \cap R) \mathscr{L}.$$

By Lemma 2.1(2) and Lemma 2.6,

$$\bigcap_{i=1}^{m} (M_i \cap R) = \left(\bigcap_{i=1}^{m} (M_i \cap R)S\right) \cap R = J\left(\bigcap_{i=1}^{m} (M_i \cap R)\mathscr{L}\right) \cap R$$
$$= J\left(\bigcap_{j=1}^{n} (N_j \cap R)\mathscr{L}\right) \cap R = \bigcap_{j=1}^{n} (N_j \cap R).$$

For  $I \subseteq R$ , let Var I be the set of  $\mathfrak{m} \in \operatorname{Spec} R$  with  $I \subseteq \mathfrak{m}$ . Then

$$\bigcup_{i=1}^{m} \{M_i \cap R\} = \bigcup_{i=1}^{m} \operatorname{Var} (M_i \cap R) = \operatorname{Var} \left( \bigcap_{i=1}^{m} (M_i \cap R) \right)$$
$$= \operatorname{Var} \left( \bigcap_{j=1}^{n} (N_j \cap R) \right) = \bigcup_{j=1}^{n} \{N_j \cap R\}.$$

Thus m = n, and after reordering,  $M_i \cap R = N_i \cap R$  for all *i*.

Recall that  $u_{\gamma}$  is the image of  $\gamma \in \Gamma = \mathcal{G}al(S/R)$  under the Galois cocycle  $u: \Gamma \to \operatorname{Aut}_{S-\operatorname{Lie}}(\mathfrak{g}(S))$ . The group  $\Gamma$  acts on the set of pairs  $(\lambda, M) \in P_+^{\times} \times \operatorname{Max}(S)$  by  $\gamma(\mu, N) = (\mu \circ \operatorname{Out} u_{\gamma}^{-1}(\gamma N), \gamma N)$ .

**Proposition 3.7.** Suppose

 $V(\lambda, M) = V_{\lambda_1}(M_1) \otimes \cdots \otimes V_{\lambda_n}(M_n) \quad and \quad V(\mu, N) = V_{\mu_1}(N_1) \otimes \cdots \otimes V_{\mu_n}(N_n)$ 

are irreducible  $\mathcal{L}$ -modules with  $\lambda, \mu \in (P_+^{\times})^n$  and  $M_i \cap R = N_i \cap R$  for all *i*. Then  $V(\lambda, M) \simeq V(\mu, N)$  if and only if there exist  $\gamma_1, \ldots, \gamma_n \in \Gamma$  such that

$$(\lambda_i, M_i) = {}^{\gamma_i}(\mu_i, N_i) \quad for \ i = 1, \dots, n.$$

*Proof.* Let  $\phi_{\lambda,M} : \mathcal{L} \to \operatorname{End}_k(V(\lambda, M))$  and  $\phi_{\mu,N} : \mathcal{L} \to \operatorname{End}_k(V(\mu, N))$  be the homomorphisms defining the module actions. Since each  $\lambda_i$  is nonzero, the kernel of the action of  $\mathfrak{g}^{\oplus n}$  on  $V(\lambda, M)$  is trivial, and the evaluation maps  $\operatorname{ev}_{M_i}$  induce an automorphism

$$\operatorname{ev}_M = \operatorname{ev}_{M_1} \oplus \cdots \oplus \operatorname{ev}_{M_n} \colon \mathscr{L} / \ker \phi_{\lambda, M} \xrightarrow{\sim} \mathfrak{g}^{\oplus n}$$

Similarly,  $ev_N : \mathcal{L} / \ker \phi_{\mu,N} \to \mathfrak{g}^{\oplus n}$  is a Lie algebra isomorphism.

Let  $\mathfrak{g} = \mathfrak{n}_- \oplus \mathfrak{h} \oplus \mathfrak{n}_+$  be the triangular decomposition of  $\mathfrak{g}$  relative to the épinglage of  $(\mathfrak{g}, \mathfrak{h})$ . We pull back the corresponding triangular decomposition of  $\mathfrak{g}^{\oplus n}$  to obtain the triangular decomposition

$$\mathscr{L}/\ker\phi_{\lambda,M} = \operatorname{ev}_{M}^{-1}(\mathfrak{n}_{-}^{\oplus n}) \oplus \operatorname{ev}_{M}^{-1}(\mathfrak{h}^{\oplus n}) \oplus \operatorname{ev}_{M}^{-1}(\mathfrak{n}_{+}^{\oplus n}).$$
(3.8)

The representations  $V(\lambda, M)$  and  $V(\mu, N)$  will be isomorphic precisely when they have the same highest weights relative to the decomposition (3.8).

The Galois group  $\Gamma = \mathcal{G}al(S/R)$  acts transitively on the fibers of the pullback map  $\operatorname{Spec}(S) \to \operatorname{Spec}(R)$  over maximal ideals of *R*. Choose  $\gamma_i \in \Gamma$  so that  $M_i = \gamma_i N_i$  for all *i*.

Let  $\mathfrak{g}^i = 0 \oplus \cdots \oplus \mathfrak{g} \oplus \cdots \oplus 0$  be the *i*-th component of  $\mathfrak{g}^{\oplus n}$ . Note that

$$\operatorname{ev}_{M}^{-1}(\mathfrak{g}^{i}) = \bigcap_{r \neq i} \ker \operatorname{ev}_{M_{r}} = \bigcap_{r \neq i} (M_{r} \cap R) \mathscr{L} = \bigcap_{r \neq i} (N_{r} \cap R) \mathscr{L} = \bigcap_{r \neq i} \ker \operatorname{ev}_{N_{r}}.$$

Therefore,  $ev_{N_j} \circ ev_M^{-1}(\mathfrak{g}^i) = 0$  for all  $i \neq j$ , and

$$\operatorname{ev}_N \circ \operatorname{ev}_M^{-1}(x^i) = \iota_i \circ \operatorname{ev}_{N_i} \circ \operatorname{ev}_M^{-1}(x^i) = \iota_i \circ \operatorname{ev}_{N_i} \circ \operatorname{ev}_{M_i}^{-1}(x)$$

for all  $x^i \in \mathfrak{g}^i$ , where  $\iota_i$  is the inclusion of  $\mathfrak{g}$  as the *i*-th component of  $\mathfrak{g}^{\oplus n}$ :

$$\iota_i:\mathfrak{g}\hookrightarrow 0\oplus\cdots\oplus\mathfrak{g}\oplus\cdots\oplus 0\subseteq\mathfrak{g}^{\oplus n}.$$

Relative to the decomposition (3.8), the highest weight of  $V(\lambda, M)$  is thus  $\sum_{i=1}^{n} \lambda_i \circ ev_{M_i}$  and the highest weight of  $V(\mu, N)$  is  $\sum_{i=1}^{n} v_i \circ ev_{N_i}$ , where  $v_i \in (ev_{N_i} \circ ev_{M_i}^{-1}(\mathfrak{h}))^*$  is the highest weight of  $V_{\mu_i}$ , relative to the new triangular decomposition

$$\mathfrak{g} = \operatorname{ev}_{N_i} \circ \operatorname{ev}_{M_i}^{-1}(\mathfrak{n}_-) \oplus \operatorname{ev}_{N_i} \circ \operatorname{ev}_{M_i}^{-1}(\mathfrak{h}) \oplus \operatorname{ev}_{N_i} \circ \operatorname{ev}_{M_i}^{-1}(\mathfrak{n}_+).$$

By [Lau 2010, Lemma 5.2],  $v_i = \mu_i \circ \tau_i^{-1}$ , where  $\tau_i = \text{Int}(\text{ev}_{N_i} \circ \text{ev}_{M_i}^{-1})$ . That is,  $V(\lambda, M) \simeq V(\mu, N)$  if and only if

$$\sum_{i=1}^{n} \lambda_i \circ \operatorname{ev}_{M_i} = \sum_{i=1}^{n} \mu_i \circ \tau_i^{-1} \circ \operatorname{ev}_{N_i}$$

on  $ev_M^{-1}(\mathfrak{h}^{\oplus n})$ . For the *i*-th component  $\mathfrak{h}^i = 0 \oplus \cdots \oplus \mathfrak{h} \oplus \cdots \oplus 0$ , we have

$$\operatorname{ev}_M^{-1}(\mathfrak{h}^i) \subseteq \operatorname{ev}_M^{-1}(\mathfrak{g}^i) = \bigcap_{j \neq i} (M_j \cap R) \mathscr{L},$$

so  $\lambda_j \circ \operatorname{ev}_{M_j}(\operatorname{ev}_M^{-1}(\mathfrak{h}^i)) = 0$  for  $i \neq j$ . Therefore,  $V(\lambda, M) \simeq V(\mu, N)$  if and only if  $\lambda_i \circ \operatorname{ev}_{M_i} = \mu_i \circ \tau_i^{-1} \circ \operatorname{ev}_{N_i}$  for all *i*; that is, if and only if  $\lambda_i = \mu_i \circ \operatorname{Out}(\operatorname{ev}_{N_i} \circ \operatorname{ev}_{M_i}^{-1})$ .

We now simplify the expression for the automorphism  $ev_{N_i} \circ ev_{M_i}^{-1} : \mathfrak{g} \to \mathfrak{g}$ . For  $x \in \mathfrak{g}$ , write  $ev_{M_i}^{-1}(x) = \sum_j x_j \otimes s_j + \ker ev_{M_i} \in \mathscr{L} / \ker ev_{M_i} = \mathscr{L} / \ker ev_{N_i}$ , where  $x_j \in \mathfrak{g}$  and  $s_j \in S$  for all j. Then  $ev_{N_i} \circ ev_{M_i}^{-1}(x) = \sum_j s_j(N_i)x_j$ . By definition,

$$s_j(N_i) + N_i = s_j + N_i \in S/N_i,$$

and  $s_j(N_i) \in k \subseteq R$  is clearly fixed by  $\gamma_i \in \Gamma$ . Hence

$$s_i(N_i) + {}^{\gamma_i}N_i = {}^{\gamma_i}s_j + {}^{\gamma_i}N_i \in S/{}^{\gamma_i}N_i = S/M_i,$$

and  $s_j(N_i) = {}^{\gamma_i}s_j(M_i)$ . Therefore,

$$\operatorname{ev}_{N_i} \circ \operatorname{ev}_{M_i}^{-1}(x) = \sum_j^{\gamma_i} s_j(M_i) x_j$$

Moreover,  $\sum_{j} x_j \otimes s_j \in \mathcal{L} = \{ z \in \mathfrak{g} \otimes S \mid u_{\gamma} \gamma z = z \text{ for all } \gamma \in \Gamma \}$ , so

$$\operatorname{ev}_{N_{i}} \circ \operatorname{ev}_{M_{i}}^{-1}(x) = {}^{\gamma_{i}} \left(\sum_{j} x_{j} \otimes s_{j}\right) (M_{i}) = (u_{\gamma_{i}})^{-1} \left(\sum_{j} x_{j} \otimes s_{j}\right) (M_{i})$$
$$= u_{\gamma_{i}}^{-1}(M_{i}) \sum_{j} s_{j}(M_{i}) x_{j} = u_{\gamma_{i}}^{-1}(M_{i})(x),$$

and  $\operatorname{ev}_{N_i} \circ \operatorname{ev}_{M_i}^{-1} = u_{\gamma_i}^{-1}(M_i)$ . Hence  $V(\lambda, M) \simeq V(\mu, N)$  if and only if there exist  $\gamma_1, \ldots, \gamma_n \in \Gamma$  such that  $\gamma_i(\mu_i, N_i) = (\lambda_i, M_i)$  for all *i*.

We identify the  $\mathscr{L}$ -module  $V(\lambda, M) = V_{\lambda_1}(M_1) \otimes \cdots \otimes V_{\lambda_n}(M_n)$  with the map

$$\chi_{[\lambda,M]}$$
: Max $(S) \rightarrow P_+,$ 

where  $\chi_{[\lambda,M]} = \sum_{\gamma \in \Gamma} \sum_{i=1}^{n} \chi_{\gamma(\lambda_i,M_i)}$  and

$$\chi_{(\mu_i,N_i)} : \operatorname{Max}(S) \to P_+, \quad I \mapsto \begin{cases} \mu_i & \text{if } I = N_i, \\ 0 & \text{otherwise.} \end{cases}$$

The Galois group  $\Gamma$  acts on the set  $\mathcal{F}$  of finitely supported functions  $Max(S) \to P_+$ , by identifying each function f with the set of ordered pairs  $\{(f(M), M) \mid M \in Max(S)\}$  and defining  $\gamma f = \{\gamma(f(M), M) \mid M \in Max(S)\}$ . The function  $\chi_{[\gamma,M]}$  is then  $\Gamma$ -invariant, and the set  $\mathcal{F}^{\Gamma}$  of  $\Gamma$ -invariant functions in  $\mathcal{F}$  is in bijection with the set  $\mathscr{C}$  of isomorphism classes [V] of (finite-dimensional) simple  $\mathscr{L}$ -modules V:

**Theorem 3.9.** The map  $\psi : [V(\lambda, M)] \mapsto \chi_{[\lambda,M]}$  is a well-defined natural bijection between  $\mathscr{C}$  and  $\mathscr{F}^{\Gamma}$ .

*Proof.* By Theorem 3.4, Lemma 3.6, and Proposition 3.7, two simple  $\mathscr{L}$ -modules  $W_1$  and  $W_2$  are isomorphic if and only if there exist  $n \ge 0$ , ordered pairs

$$(M, \lambda), (N, \mu) \in (\operatorname{Max}(S))^n \times (P_+^{\times})^n$$

with  $M_i \cap R = N_i \cap R \neq N_j \cap R = M_j \cap R$  for  $i \neq j$ , and  $\gamma_1, \ldots, \gamma_n \in \Gamma$ such that  $W_1 \simeq V(\lambda, M)$ ,  $W_2 \simeq V(\mu, N)$ , and  $(M_i, \lambda_i) = {}^{\gamma_i}(N_i, \mu_i)$  for all *i*. Thus  $V(\lambda, M) \simeq V(\mu, N)$  if and only if  $\chi_{[\lambda, M]} = \chi_{[\mu, N]}$ . In particular, the map  $\psi : \mathscr{C} \to \mathscr{F}^{\Gamma}$  is well-defined and injective. It is also surjective, as the support of any  $f \in \mathscr{F}^{\Gamma}$  decomposes into a disjoint union of  $\Gamma$ -orbits. Therefore,

$$f = \sum_{\gamma \in \Gamma} \sum_{i=1}^{m} \chi_{\gamma(\lambda_i, M_i)}$$

444

Maximal ideals and representations of twisted forms of algebras 445

for some collection of orbit representatives  $M_1, \ldots, M_m \in Max(S)$ .

#### 4. Applications

In this section, k will denote an algebraically closed field of characteristic zero.

**4a.** *Multiloop algebras.* Multiloop algebras are multivariable generalizations of the loop algebras in affine Kac–Moody theory. The study of these algebras and their extensions includes a substantial body of work on (twisted and untwisted) multiloop, toroidal, and extended affine Lie algebras. The representation theory of multiloop algebras has also been adapted to include generalized current algebras and equivariant map algebras [Chari et al. 2010; Neher et al. 2012]. When *R* and *S* are Laurent polynomial rings, the intersection of the class of algebras with the class of twisted forms discussed in the present paper includes multiloop algebras (Section 4a), but not Margaux algebras (Section 4b), for instance.

Let  $\mathfrak{g}$  be a finite-dimensional simple Lie algebra over k, with commuting automorphisms  $\sigma_1, \ldots, \sigma_N : \mathfrak{g} \to \mathfrak{g}$  of finite orders  $m_1, \ldots, m_N$ , respectively. Fix a primitive  $m_j$ -th root of unity  $\xi_j \in k$  for each j, and let  $R = k[t_1^{\pm m_1}, \ldots, t_N^{\pm m_N}] \subseteq S = k[t_1^{\pm 1}, \ldots, t_N^{\pm 1}].$ 

The (twisted) multiloop algebra  $\mathscr{L} = \mathscr{L}(\mathfrak{g}, \sigma)$  is a  $\mathbb{Z}^N$ -graded subalgebra of  $\mathfrak{g}(S) = \mathfrak{g} \otimes S$ :

$$\mathscr{L}(\mathfrak{g},\sigma) = \bigoplus_{j \in \mathbb{Z}^N} \mathfrak{g}_j \otimes t^j,$$

where  $j = (j_1, \ldots, j_N)$ ,  $\mathfrak{g}_j = \{x \in \mathfrak{g} \mid \sigma_i(x) = \xi_i^{j_i} x \text{ for } i = 1, \ldots, N\}$ , and  $t^j = t_1^{j_1} t_2^{j_2} \cdots t_N^{j_N}$ . It is easy to see that  $\mathscr{L}$  is a Lie algebra over R and an S/R-form of  $\mathfrak{g}(R)$ .

Specializing our main theorems to the case of multiloop algebras, we recover the results of [Lau 2010]. Maximal ideals  $M_i = M_{a_i} = (t_1 - a_{i1}, \dots, t_N - a_{iN})$  of *S* correspond to points  $a_i = (a_{i1}, \dots, a_{iN})$  on the algebraic *n*-torus  $(k^{\times})^N = k^{\times} \times \dots \times k^{\times}$ . Note that  $M_i \cap R$  is the ideal (of *R*) of polynomials vanishing at  $a_i$ . Thus  $M_i \cap R \in \text{Max } R$  is generated by  $\{t_1^{m_1} - a_{i1}^{m_1}, \dots, t_N^{m_N} - a_{iN}^{m_N}\}$ . Therefore,  $M_i \cap R = M_j \cap R$  if and only if  $m(a_i) = m(a_j)$ , where we write  $m(a_\ell) = (a_{\ell 1}^{m_1}, \dots, a_{\ell N}^{m_N})$  for all  $a_\ell \in (k^{\times})^N$ .

The Galois group  $\Gamma = \mathcal{G}al(S/R)$  is  $\mathbb{Z}_{m_1} \times \cdots \times \mathbb{Z}_{m_N}$ , where each  $\mathbb{Z}_{m_i}$  is generated by an element

$$\alpha_i: t_j \mapsto \begin{cases} \xi_i t_i & \text{if } i = j, \\ t_j & \text{otherwise} \end{cases}$$

The 1-cocycle  $u: \Gamma \to \operatorname{Aut}_{S-\operatorname{Lie}}(\mathfrak{g}(S))$  corresponding to  $\mathscr{L}$  is given by

$$u_{\gamma} = \sigma_1^{-r_1} \cdots \sigma_N^{-r_N} \otimes 1,$$

for each  $\gamma = (\alpha_1^{r_1}, \dots, \alpha_N^{r_N}) \in \Gamma$ . Then  $u_{\gamma}(M) = \sigma_1^{-r_1} \cdots \sigma_N^{-r_N}$  for all  $M \in Max(S)$ . The fact that

$$u_{\gamma} : \operatorname{Max}(S) \to \operatorname{Aut} \mathfrak{g}, \quad M \mapsto u_{\gamma}(M)$$

is constant means that the action of  $\Gamma$  on  $P_+^{\times} \times Max(S)$  splits into separate actions of  $\Gamma$  on Max(S) and on  $P_+^{\times}$  by

$$\psi: \Gamma \times P_+^{\times} \to P_+^{\times}, \quad (\gamma, \lambda) \mapsto \lambda \circ \operatorname{Out} \sigma_1^{-r_1} \cdots \sigma_N^{-r_N}$$

In this language,  $\Gamma$  acts on  $P_+^{\times} \times \text{Max}(S)$  as  $\gamma(\lambda, M) = (\psi(\gamma^{-1}, \lambda), \gamma M)$ . The  $\Gamma$ -invariant functions  $\chi_{[\lambda, M]} : \text{Max}(S) \to P_+$  become  $\Gamma$ -equivariant functions under the new action  $\psi$  on  $P_+^{\times}$ . We thus recover the following theorem [Lau 2010, Corollary 4.4, Theorem 4.5, and Corollary 5.10]:

- **Theorem 4.1.** (1) The finite-dimensional simple modules of  $\mathscr{L}(\mathfrak{g}; \sigma)$  are those of the form  $V(\lambda, a) = V_{\lambda_1}(M_{a_1}) \otimes \cdots \otimes V_{\lambda_n}(M_{a_n})$  for  $n \ge 0$ ,  $a_i \in (k^{\times})^N$ , and  $m(a_i) \ne m(a_j)$  whenever  $i \ne j$ .
- (2) The isomorphism classes of finite-dimensional simple  $\mathscr{L}(\mathfrak{g}; \sigma)$ -modules are in bijection with the finitely supported  $\Gamma$ -equivariant maps  $(k^{\times})^N \to P_+$ .
- 4b. Azumaya and Margaux algebras. Fix Laurent polynomial rings

$$R = k[t_1^{\pm 2}, t_2^{\pm 2}]$$
 and  $S = k[t_1^{\pm 1}, t_2^{\pm 1}].$ 

Let A = A(1, 2) be the standard Azumaya algebra, the unital associative *R*-algebra generated by  $\{T_1^{\pm 1}, T_2^{\pm 1}\}$  with relations  $T_2T_1 = -T_1T_2$  and  $T_i^2 = t_i^2$  for i = 1, 2. Then *A* is an *S*/*R*-form of the associative algebra  $M_2(R)$  of  $2 \times 2$  matrices over *R*, as can be readily verified using one of the well-known representations of the quaternions as matrices in  $M_2(\mathbb{C})$ .

Since **PGL**<sub>2</sub> is the automorphism group (scheme) of both  $M_2(k)$  and  $\mathfrak{sl}_2(k)$ , there is a natural correspondence between S/R-forms of  $M_2(R)$  and  $\mathfrak{sl}_2(R)$ . Namely, given any S/R-form B of the matrix algebra  $M_2(R)$ , view B as a Lie algebra Lie Bwith bracket [a, b] = ab - ba. Its derived subalgebra (Lie B)' = Span{ $[a, b] \mid a, b \in B$ } is then an S/R-form of  $\mathfrak{sl}_2(R)$ .

Applying this construction to  $\mathcal{L}_1 = (\text{Lie } A)'$  and computing explicitly, it follows that  $\mathcal{L}_1 \simeq \mathcal{L}(\mathfrak{sl}_2(k), \sigma_1, \sigma_2)$  where  $\sigma_1$  and  $\sigma_2$  are conjugation by  $\begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix}$  and  $\begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}$ , respectively [Gille and Pianzola 2007]. Therefore, we obtain the representations of  $\mathcal{L}_1$  as in the previous section.

Surprisingly, not every twisted form of  $\mathfrak{g}(k[t_1^{\pm 1}, t_2^{\pm 1}])$  is a multiloop algebra. This can be seen using loop torsors. The only known *S*/*R*-forms of  $\mathfrak{g}(R)$  that are not isomorphic to multiloop algebras are called *Margaux algebras*. The simplest of these can be constructed concretely as follows. See [Gille and Pianzola 2007] for details.

446

Let A, R, and S be as in Section 4a. The right A-module

$$M = \{ (\lambda, \mu) \in A \oplus A \mid (1 + T_1)\lambda = (1 + T_2)\mu \}$$

is projective but not free. This can be used to show that its endomorphism ring  $\mathcal{M} = \operatorname{End}_A(\mathcal{M})$ , while also an S/R-form of  $M_2(R)$ , is not isomorphic to A as an A-algebra. It follows that  $\mathcal{L}_1$  and  $\mathcal{L}_2 = (\operatorname{Lie} \mathcal{M})'$  are nonisomorphic S/R-forms of  $\mathfrak{sl}_2(R)$ . By the classification of involutions in PGL<sub>2</sub>(k) and a study of loop torsors, it can be shown that  $\mathcal{L}_2$  is *not* a (twisted) multiloop algebra.

By Theorems 3.4 and 3.9, the irreducible representations of  $\mathcal{L}_2$  are the tensor products  $V(\lambda, M) = V_{\lambda_1}(M_1) \otimes \cdots \otimes V_{\lambda_n}(M_n)$ , where  $\lambda_1, \ldots, \lambda_n \in \mathbb{Z}_+ \setminus \{0\}$  are highest weights of  $\mathfrak{sl}_2(k)$  and  $M_i = \langle t_1 - a_{i1}, t_2 - a_{i2} \rangle$  are maximal ideals of  $S = k[t_1^{\pm 1}, t_2^{\pm 1}]$  corresponding to points in distinct fibers over Spec *R*. That is,  $(a_{i1}^2, a_{i2}^2) \neq (a_{j1}^2, a_{j2}^2)$  for  $i \neq j$ .

Two such representations

$$V(\lambda, M) = V_{\lambda_1}(M_1) \otimes \cdots \otimes V_{\lambda_m}(M_m)$$
 and  $V(\mu, N) = V_{\mu_1}(N_1) \otimes \cdots \otimes V_{\mu_n}(N_n)$ 

are isomorphic precisely when the corresponding Gal(S/R)-invariant functions  $\chi_{[\lambda,M]}$  and  $\chi_{[\mu,N]}$  are equal. But the action

$$\gamma(\lambda_i, M_i) = (\lambda_i \circ \operatorname{Out} u_{\gamma}^{-1}(\gamma M_i), \gamma M_i)$$

is simply an action on Max(S),

$$^{\gamma}(\lambda_i, M_i) = (\lambda_i, {}^{\gamma}M_i),$$

since  $u_{\gamma}^{-1}({}^{\gamma}M) \in \text{Aut } \mathfrak{sl}_2(k)$ , and every automorphism of  $\mathfrak{sl}_2(k)$  is inner! Thus  $V(\lambda, M) \simeq V(\mu, N)$  if and only if (after reordering the tensor factors) m = n,  $\lambda_i = \mu_i$ , and the  $a_i, b_i \in k^{\times} \times k^{\times}$  corresponding to  $M_i$  and  $N_i$  satisfy  $a_{ij} = \pm b_{ij}$  for all *i* and *j*.

As for any Galois extension S/R, the isomorphism classes of the (finite-dimensional) simple modules of any S/R-form of  $\mathfrak{sl}_2(R)$  are given by restrictions of the same evaluation modules of  $\mathfrak{sl}_2(S)$ . In particular, the irreducible  $\mathcal{L}_1$ - and  $\mathcal{L}_2$ -modules come from the same  $\mathfrak{sl}_2(S)$ -modules.

#### Acknowledgements

We would like to thank Jean Auger and Zhihua Chang for their careful reading of the manuscript.

#### References

<sup>[</sup>Bourbaki 1964] N. Bourbaki, *Éléments de mathématique: Algèbre commutative, Chapitre 5*, Actualités Scientifiques et Industrielles **1308**, Hermann, Paris, 1964. MR 33 #2660 Zbl 0205.34302

- [Bourbaki 1975] N. Bourbaki, *Éléments de mathématique: Groupes et algèbres de Lie, Chapitre 7 et 8*, Actualités Scientifiques et Industrielles **1364**, Hermann, Paris, 1975. MR 56 #12077 Zbl 0329.17002
- [Chari et al. 2010] V. Chari, G. Fourier, and T. Khandai, "A categorical approach to Weyl modules", *Transform. Groups* **15**:3 (2010), 517–549. MR 2011m:17014 Zbl 1245.17004
- [Gille and Pianzola 2007] P. Gille and A. Pianzola, "Galois cohomology and forms of algebras over Laurent polynomial rings", *Math. Ann.* **338**:2 (2007), 497–543. MR 2008b:20055 Zbl 1131.11070
- [Knus and Ojanguren 1974] M.-A. Knus and M. Ojanguren, *Théorie de la descente et algèbres d'Azumaya*, Lecture Notes in Math. **389**, Springer, Berlin, 1974. MR 54 #5209 Zbl 0284.13002
- [Lau 2010] M. Lau, "Representations of multiloop algebras", *Pacific J. Math.* 245:1 (2010), 167–184.
   MR 2011f:17041 Zbl 1237.17012
- [Matsumura 1989] H. Matsumura, *Commutative ring theory*, 2nd ed., Cambridge Studies in Advanced Mathematics **8**, Cambridge University Press, 1989. MR 90i:13001 Zbl 0666.13002
- [Milne 1980] J. S. Milne, *Étale cohomology*, Princeton Mathematical Series **33**, Princeton University Press, 1980. MR 81j:14002 Zbl 0433.14012
- [Neher et al. 2012] E. Neher, A. Savage, and P. Senesi, "Irreducible finite-dimensional representations of equivariant map algebras", *Trans. Amer. Math. Soc.* 364:5 (2012), 2619–2646. MR 2888222 Zbl 06028848
- [Senesi 2010] P. Senesi, "Finite-dimensional representation theory of loop algebras: a survey", pp. 263–283 in *Quantum affine algebras, extended affine Lie algebras, and their applications*, edited by Y. Gao et al., Contemp. Math. **506**, Amer. Math. Soc., Providence, RI, 2010. MR 2011d:17040 Zbl 05901955

Communicated by Georgia Benkart Received 2011-11-07 Accepted 2012-03-03

michael.lau@mat.ulaval.ca	Département de mathématiques et de statistique, Université Laval, Pavillon Vachon (Local 1056), 1045 av. de la Médecine, Québec, QC G1V0A6, Canada
a.pianzola@gmail.com	Department of Mathematical and Statistical Sciences, University of Alberta, 632 Central Academic Building, Edmonton, AB T6G2G1, Canada
	Centro de Altos Estudios en Ciencias Exactas, Avenida de Mayo 866, (1084), Buenos Aires, Argentina



# Higher Chow groups of varieties with group action

Amalendu Krishna

We give explicit descriptions of the higher Chow groups of toric bundles and flag bundles over schemes. We derive several consequences of these descriptions for the equivariant and ordinary higher Chow groups of schemes with group action.

We prove a decomposition theorem for the equivariant higher Chow groups of a smooth scheme with action of a diagonalizable group. This theorem is applied to compute the equivariant and ordinary higher Chow groups of smooth toric varieties. The results of this paper play fundamental roles in the proof of the Riemann–Roch theorems for equivariant higher K-theory.

#### 1. Introduction

A *scheme* in this paper will mean a separated and reduced scheme of finite type over a perfect field k, which admits an ample line bundle. This base field k will be fixed throughout this paper. A *linear algebraic group* G over k will mean a smooth and affine group scheme over k. By a closed subgroup H of an algebraic group G, we shall mean a morphism  $H \rightarrow G$  of algebraic groups over k that is a closed immersion of k-schemes. In particular, a closed subgroup of a linear algebraic group will be of the same type and hence smooth. Recall from [Borel 1991, Proposition 1.10] that a linear algebraic group over k is a closed subgroup of a general linear group, defined over k.

Recall that an action of a linear algebraic group G on a k-scheme X is said to be *linear* if X admits a G-equivariant ample line bundle, a condition that is always satisfied if X is normal (see [Sumihiro 1975, Theorem 2.5] for G connected and [Thomason 1988, 5.7] for G general). All G-actions in this paper will be assumed to be linear. Let  $\mathcal{V}_k$  denote the category of quasiprojective k-schemes and let  $\mathcal{V}_k^S$  denote the full subcategory of smooth k-schemes. We shall denote the category of quasiprojective G-schemes with G-equivariant maps by  $\mathcal{V}_G$ , and the full subcategory of smooth G-schemes will be denoted by  $\mathcal{V}_G^S$ .

MSC2010: primary 14C40, 14C35; secondary 14C25.

Keywords: algebraic cycles, group action.

The *G*-equivariant higher Chow groups  $CH^G_*(X, i)$  of  $X \in \mathcal{V}_G$  were defined by Edidin and Graham [1998] in terms of the ordinary higher Chow groups (motivic Borel–Moore homology) of the quotient space  $X \times^G U$ . Here, *U* is a *G*-invariant open subscheme of a finite-dimensional representation *V* of *G* such that it acts freely on *U* and  $V \setminus U$  is of sufficiently high codimension. This definition of Edidin and Graham is based on an earlier construction of Totaro [1999], who invented the idea above to define the Chow groups of the classifying spaces of linear algebraic groups.

In this paper, we develop further the Edidin–Graham theory of equivariant higher Chow groups and establish many important properties of this theory. We also prove some decomposition theorems for the equivariant higher Chow groups of smooth schemes with torus action. These results turn out to have many applications.

Brion [1997] proved many results about the equivariant Chow groups of the form  $CH^G_*(X, 0)$ . Many of the structural results in this paper can be described as the generalization of the results of [Brion 1997] to the case of equivariant *higher* Chow groups. In Section 2, we recall the definition of equivariant higher Chow groups from [Edidin and Graham 1998] and prove its basic properties, which are all well known for the ordinary higher Chow groups; see [Bloch 1986]. As a consequence, one finds that the equivariant higher Chow groups form a Borel–Moore oriented bigraded homology theory in the category of schemes with the action of a given linear algebraic group. Other important results about these groups such as the Morita isomorphism are proven in Section 3. We also prove a structure theorem (see Theorem 3.5) for the equivariant higher Chow groups of schemes with action of tori.

Section 4 contains the proof of the self-intersection formula for the higher Chow groups. This formula plays a very important role in the proofs of the main results of this paper. In Section 5, we construct Demazure operators on equivariant higher Chow groups and give some consequences of these operators.

In Section 6, we prove the Leray–Hirsch theorem for the higher Chow groups. As a consequence of this theorem, we compute the higher Chow groups of toric bundles in Section 7. In Section 8, we turn to the description of the higher Chow groups of principal bundles and flag bundles over schemes. We give several applications of these descriptions in the study of equivariant higher Chow groups.

In Sections 9 and 10, we prove a decomposition theorem (see Theorem 10.3) for the equivariant higher Chow groups of smooth schemes with action of a diagonalizable group G. This result describes the equivariant higher Chow group of a G-scheme in terms of the equivariant higher Chow groups of the loci where the stabilizers have a fixed dimension. This result is an analogue of a similar result of Vezzosi and Vistoli [2003] in equivariant K-theory and has many important

450

applications in the study of equivariant and ordinary higher Chow groups of smooth schemes.

Theorem 10.3 is the basic step in the proof of the equivariant Riemann–Roch theorem in [Krishna 2009b]. This theorem presents an explicit relation between the equivariant K-theory and the equivariant higher Chow groups. Like in the ordinary case, this Riemann–Roch is a fundamental result in equivariant algebraic geometry. This theorem was in fact one of the main motivations for the author to work on this paper. We expect Theorem 10.3 to have many more applications in the computation of equivariant and ordinary higher Chow groups. In Section 11, we apply this theorem to compute the equivariant and ordinary higher Chow groups of smooth toric varieties. We shall follow the following convention while studying the equivariant and ordinary higher Chow groups with the rational coefficients.

*Convention.* In this paper, all the results and statements up to Section 7 do not make any assumption on the coefficient ring of the higher Chow groups. On the other hand, all the results and statements from Section 8 onwards assume rational coefficients. In order to simplify the notation, the following convention will be followed.

From Section 8 onwards, an abelian group *A* will actually mean its extension  $A \otimes_{\mathbb{Z}} \mathbb{Q}$ . In particular, all higher Chow groups and other cohomology groups will be considered with the rational coefficients. For  $\mathbb{Q}$ -vector spaces *A* and *B*, the tensor product  $A \otimes_{\mathbb{Q}} B$  will be simply written as  $A \otimes B$ . We shall however, indicate the appropriate coefficients in the statements of the all results.

# 2. Equivariant higher Chow groups

In this section, we recall the definition of the equivariant higher Chow groups from [Edidin and Graham 1998] and review their main functorial properties. It turns out in particular that the equivariant higher Chow groups have all the properties of an oriented bigraded Borel–Moore homology theory.

Let *G* be a linear algebraic group and let *X* be a scheme over *k* with a *G*-action. We shall denote the dimension of the underlying group *G* usually by the letter *g*. All representations of *G* in this paper will be finite-dimensional. The definition of equivariant higher Chow groups of *X* needs one to consider certain kind of mixed spaces which in general may not be schemes even if the original spaces are schemes. The following well-known (see [Edidin and Graham 1998, Proposition 23]) lemma shows that this problem does not occur in our context and all the mixed spaces in this paper are schemes with ample line bundles.

**Lemma 2.1.** Let *H* be a linear algebraic group acting freely and linearly on a *k*-scheme *U* such that the quotient U/H exists as a quasiprojective scheme. Let *X* be a *k*-scheme with a linear action of *H*. Then the mixed quotient  $X \times^H U$  exists for

the diagonal action of H on  $X \times U$  and is quasiprojective. Moreover, this quotient is smooth if both U and X are so. In particular, if H is a closed subgroup of a linear algebraic group G and X is a k-scheme with a linear action of H, then the quotient  $G \times^H X$  is a quasiprojective scheme.

*Proof.* It is already shown in [Edidin and Graham 1998, Proposition 23] using [Mumford et al. 1994, Proposition 7.1] that the quotient  $X \times^H U$  is a scheme. Moreover, as U/H is quasiprojective, [Mumford et al. 1994, Proposition 7.1] in fact shows that  $X \times^H U$  is also quasiprojective. The similar conclusion about  $G \times^H X$  follows from the first case by taking U = G and by observing that G/H is a smooth quasiprojective scheme; see [Borel 1991, Theorem 6.8]. The assertion about the smoothness is clear since  $X \times U \to X \times^H U$  is a principal *H*-bundle.

**2a.** *Good pairs and equivariant higher Chow groups.* For any integer  $j \ge 0$ , let V be an l-dimensional representation of G and let U be a G-invariant open subset of V such that the codimension of the complement  $V \setminus U$  in V is sufficiently larger than j, and G acts freely on U such that the quotient U/G is a quasiprojective scheme. Such a pair (V, U) will be called a *good* pair for the G-action corresponding to j. It is easy to see that a good pair always exists; see [Edidin and Graham 1998, Lemma 9].

For an equidimensional *G*-scheme *X*, let  $X_G$  denote the quotient  $X \times^G U$  of the product  $X \times U$  by the diagonal action of *G*, which is free. We define the equivariant higher Chow group  $\operatorname{CH}_G^j(X, i)$  as the homology group  $H_i(\mathfrak{Z}^j(X_G, \bullet))$ , where  $\mathfrak{Z}^j(X_G, \bullet)$  is the Bloch cycle complex of the scheme  $X_G$ . It is known [Edidin and Graham 1998, Section 2] that this definition of  $\operatorname{CH}_G^j(X, i)$  is independent of the choice of a good pair (V, U) for the *G*-action up to unique isomorphisms. One should also observe that  $\operatorname{CH}_G^j(X, i)$  may be nonzero for infinitely many values of *j*, a crucial change from the case of nonequivariant (ordinary) higher Chow groups.

If X is of dimension d, which is not necessarily equidimensional, one defines the equivariant higher Chow groups as

$$\operatorname{CH}_{i}^{G}(X, i) := H_{i}(\mathscr{X}_{j+l-g}(X_{G}, \bullet)), \qquad (2-1)$$

where (V, U) is an *l*-dimensional good pair for the *G*-action corresponding to d - j, and  $\mathscr{Z}_p(X_G, \bullet)$  is the homological cycle complex of Bloch such that  $\mathscr{Z}_p(X_G, i)$  is the group of admissible algebraic cycles on  $X_G \times \Delta^i$  of dimension p + i. We write

$$\operatorname{CH}^{G}_{*}(X,i) = \bigoplus_{-\infty < j \le d} \operatorname{CH}^{G}_{j}(X,i) \quad \text{and} \quad \operatorname{CH}^{G}_{*}(X) = \bigoplus_{i \ge 0} \operatorname{CH}^{G}_{*}(X,i).$$
(2-2)

It is easy to see that  $CH_j^G(X, i) = CH_G^{d-j}(X, i)$  if X is equidimensional of dimension d. For most of this paper, we shall use the cohomological indexing for the equivariant

higher Chow groups while dealing with smooth schemes. In particular,  $CH^*_G(X)$  will denote the sum  $\bigoplus_{i>0} CH^*_G(X, i)$ .

For a commutative ring *R*, the equivariant higher Chow groups  $\operatorname{CH}_{G}^{j}(X, i; R)$  are defined as the homology groups of the complex  $\mathscr{L}^{j}(X_{G}, \bullet) \otimes_{\mathbb{Z}} R$ . The symbol  $\operatorname{CH}_{G}^{*}(X; R)$  will denote the direct sum of all  $\operatorname{CH}_{G}^{j}(X, i; R)$ . We shall denote the rings  $\operatorname{CH}_{G}^{*}(k, 0)$  and  $\operatorname{CH}_{G}^{*}(k, 0; R)$  by S(G) and S(G; R), respectively.

# **2b.** *Equivariant operational Chow groups.* For $X \in \mathcal{V}_G$ , we define

$$OPCH_G^j(X, i) = \lim_{K \to G} CH_G^j(Y, i), \qquad (2-3)$$

where the limit is taken over the category of arrows  $X \to Y$  in  $\mathcal{V}_G$  with  $Y \in \mathcal{V}_G^S$ . Notice that the natural map  $\operatorname{OPCH}_G^j(X, i) \to \operatorname{CH}_G^j(X, i)$  is an isomorphism if X is smooth. We shall write the sum  $\bigoplus_{i,j>0} \operatorname{OPCH}_G^j(X, i)$  as  $\operatorname{OPCH}_G^*(X)$ .

It follows from [Bloch 1986, Proposition 5.5, Corollary 5.6] that  $OPCH_G^*(X)$  has a ring structure and  $OPCH_G^*(X, 0)$  is a subring of  $OPCH_G^*(X)$ . Moreover,  $X \mapsto OPCH_G^*(X)$  is a contravariant functor on  $\mathcal{V}_G$  which acts on the higher Chow groups of X. In particular,  $OPCH_G^1(X, 0) \xrightarrow{\sim} Pic_G(X)$  acts on  $CH_*^G(X, i)$ . This action is same as the action of the Chern classes of equivariant line bundles on the homology theory  $CH_*^G(X, i)$ .

**2c.** *Main properties of equivariant higher Chow groups.* The following result summarizes most of the essential properties of the equivariant higher Chow groups that will be used in this paper.

**Proposition 2.2.** The equivariant higher Chow groups as defined above satisfy the following properties.

(1) Functoriality: Covariance for proper maps, contravariance for flat maps and their compatibility. That is, for a fiber diagram

$$\begin{array}{c|c} X' \xrightarrow{g'} X \\ f' & & & \\ f' & & & \\ Y' \xrightarrow{g} Y \end{array}$$

in  $\mathcal{V}_G$  with f proper and g flat, one has

$$g^* \circ f_* = f'_* \circ g'^* : \operatorname{CH}^G_*(X, i) \to \operatorname{CH}^G_*(Y', i).$$

Moreover, if  $f : X \to Y$  is a morphism in  $\mathcal{V}_G$  with Y in  $\mathcal{V}_G^S$ , then there is a pull-back map  $f^* : CH^G_*(Y, i) \to CH^G_*(X, i)$ .

(2) Homotopy: If  $f: X \to Y$  is an equivariant vector bundle, then

 $f^*: \mathrm{CH}^G_*(Y, i) \xrightarrow{\sim} \mathrm{CH}^G_*(X, i).$ 

(3) Exterior product: There is a natural product map

$$\operatorname{CH}^G_*(X, i) \otimes \operatorname{CH}^G_*(Y, i') \to \operatorname{CH}^G_*(X \times Y, i+i').$$

Moreover, if  $f : X \to Y$  is such that  $Y \in \mathcal{V}_G^S$ , then there is a pull-back via the graph map  $\Gamma_f : X \to X \times Y$ , which makes  $\operatorname{CH}_G^*(Y)$  a bigraded ring and  $\operatorname{CH}_*^G(X)$  a module over this ring. In particular,  $\operatorname{CH}_*^G(X, i)$  an S(G)-module for  $X \in \mathcal{V}_G$  and  $i \ge 0$ .

(4) Localization: If  $Y \subset X$  is a G-invariant closed subscheme with complement U, then there is a long exact localization sequence of S(G)-modules

$$\cdots \to \operatorname{CH}^G_*(Y,i) \to \operatorname{CH}^G_*(X,i) \to \operatorname{CH}^G_*(U,i) \to \operatorname{CH}^G_*(Y,i-1) \to \cdots$$

This sequence is compatible with the push-forward and flat pull-back maps of higher Chow groups.

- (5) Chern classes: For any *G*-equivariant vector bundle of rank *r*, there are equivariant Chern classes  $c_l^G(E)$ :  $\operatorname{CH}_j^G(X, i) \to \operatorname{CH}_{j-l}^G(X, i)$  for  $0 \le l \le r$ , having the same functoriality properties as in the nonequivariant case and  $c_0^G(E) = 1$ .
- (6) Projection formula: For a proper map  $f : X \to Y$  in  $\mathcal{V}_G$  and for  $x \in CH^G_*(X)$ ,  $y \in OPCH^*_G(Y)$ , one has  $f_*(f^*(y) \cdot x) = y \cdot f_*(x)$ . Here, the action of  $OPCH^*_G(Y)$  on  $CH^G_*(X)$  is given by (3) above.
- (7) *Free action: If G acts freely on X with quotient Y, then there is a canonical isomorphism*  $CH^G_*(X, i) \xrightarrow{\sim} CH_*(Y, i)$ .

*Proof.* Since the equivariant higher Chow groups of X are defined in terms of the higher Chow groups of  $X_G$ , the proposition (except possibly the last property) can be easily deduced from the similar results for the nonequivariant higher Chow groups as in [Bloch 1986] and the techniques of [Edidin and Graham 1998]. We therefore skip the proof. To see that the maps in the localization sequence are S(G)-linear, it suffices to know that for a good pair (V, U), the long exact sequence

$$\cdots \rightarrow \operatorname{CH}_*(Y_G, i) \rightarrow \operatorname{CH}_*(X_G, i) \rightarrow \operatorname{CH}_*(U_G, i) \rightarrow \operatorname{CH}_*(Y_G, i-1) \rightarrow \cdots$$

is a sequence of  $CH_*(U/G)$ -modules. But this is a well-known fact as U/G is smooth and the above is a sequence of higher Chow groups of schemes over it; see [Bloch 1986, Exercise 5.8(ii)].

To prove (6), we need to show that if  $Y \stackrel{g}{\to} Z$  is a *G*-equivariant map with  $Z \in \mathcal{V}_G^S$ , then the map  $\operatorname{CH}^G_*(X) \stackrel{f_*}{\to} \operatorname{CH}^G_*(Y)$  is  $\operatorname{CH}^*_G(Z)$ -linear. Since the push-forward and the pull-back maps of equivariant Chow groups are nothing but the maps of ordinary higher Chow groups of suitable mixed quotients, it suffices to prove the statement above for the push-forward map of the higher Chow groups corresponding to the

454

maps of mixed quotients  $X_G \rightarrow Y_G \rightarrow Z_G$ . Since  $Z_G$  is smooth, this nonequivariant version is well-known [Bloch 1986, §5.5, Exercise 5.8].

For the last property, fix  $j \leq d$  and choose a good pair (V, U) of dimension l for the *G*-action corresponding to  $d - j \geq 0$ . Since *G* acts freely on *X*, it acts likewise also on  $X \times V$  with quotient, say  $X_V$ . Then  $X_G$  is an open subset of  $X_V$  and  $X_V \rightarrow Y$ is a vector bundle, which implies that the map  $\operatorname{CH}_j(Y, i) \rightarrow \operatorname{CH}_{j+l}(X_V, i)$  is an isomorphism by the homotopy invariance. On the other hand, the restriction map  $\operatorname{CH}_{j+l}(X_V, i) \rightarrow \operatorname{CH}_{j+l}(X_G, i) = \operatorname{CH}_{j-g}^G(X, i)$  is an isomorphism by the property (4) as d - j is sufficiently small.

**Remark 2.3.** The reader should be warned that the various isomorphisms between the (equivariant) higher Chow groups in the proposition above are true only up to some obvious shift in the dimension of cycles, which we have chosen not to write.

We next recall from [Edidin and Graham 1998] that the Chern classes  $c_l^G(E)$  of an equivariant vector bundle E, as described in Proposition 2.2 above, live in the operational Chow groups  $OPCH^l(X_G)$ . If X is in  $\mathcal{V}_G^S$  however, this operational Chow group is isomorphic to the equivariant Chow group  $CH_G^l(X, 0)$  and the action of  $c_l^G(E)$  on  $CH_G^*(X)$  then coincides with the intersection product in the ring  $CH_G^*(X)$ .

Finally, we recall from [ibid.] that if  $H \subset G$  is a closed subgroup and if (V, U) is a good pair, then for  $X \in \mathcal{V}_G$ , the natural map of quotients  $X \times^H U \to X \times^G U$  is an étale locally trivial G/H-fibration and hence there is a natural restriction map

$$r_{H,X}^G: \operatorname{CH}^G_*(X, i) \to \operatorname{CH}^H_*(X, i).$$
(2-4)

Taking  $H = \{1\}$ , one obtains the *forgetful* map

$$r_X^G : \operatorname{CH}^G_*(X, i) \to \operatorname{CH}_*(X, i).$$
(2-5)

Moreover, as  $r_{H,X}^G$  is the pull-back under a flat (in fact, a smooth) map, it commutes (see Proposition 2.2) with the pull-back for any flat map, and with the push-forward for any proper map in  $\mathcal{V}_G$ . We remark here that although the definition of  $r_{H,X}^G$  uses a good pair (V, U) for any given  $j \leq \dim(X)$ , it is easy to check from the homotopy invariance that  $r_{H,X}^G$  is independent of the choice of the good pair (V, U).

### 3. Morita isomorphisms

In this section, we prove some Morita-type isomorphisms that address the question of comparison between the equivariant higher Chow groups for the action of two different algebraic groups. We also prove a structure theorem for these equivariant higher Chow groups under the trivial action of split tori. These results are analogues of the similar results of Thomason in equivariant *K*-theory; see [Thomason 1986, Lemma 5.6; 1988, Section 1].

**Proposition 3.1** (Morita isomorphism). Let H be a normal subgroup of a linear algebraic group G and let F = G/H. Let  $f : X \to Y$  be a G-equivariant morphism of G-varieties that is an H-torsor for the restricted action. Then the map induced on the equivariant higher Chow groups

$$\operatorname{CH}^{F}_{*}(Y, i) \xrightarrow{f^{*}} \operatorname{CH}^{G}_{*}(X, i).$$

is an isomorphism.

*Proof.* We first observe from [Springer 1998, Corollary 12.2.2] that *F* is also a linear algebraic group over the given ground field *k*. Now, since *f* is an *H*-torsor, it is clear that *G* acts on *Y* via *F*. Fix  $j \leq \dim(X)$  and choose a good pair (V, U) of dimension *l* for the *F*-action corresponding to  $\dim(Y) - j$ . Then *V* is also a representation of *G* in which *U* is *G*-invariant. In particular, *G* acts on  $X \times U$  via the diagonal action, which is easily seen to be free since *H* acts freely on *X* and *F* acts freely on *U*. By the same reason, we see that  $X \times U \to Y \times U$ , which is a principal *H*-bundle, is *G*-equivariant. This in turn implies that the map  $(X \times U)/G \to Y_F$  is an isomorphism and hence we get

$$\operatorname{CH}_{j}^{F}(Y,i) \cong \operatorname{CH}_{j+l-g+h}(Y_{F},i) \xrightarrow{\sim}_{f^{*}} \operatorname{CH}_{j+l-g+h}(X \times^{G} U,i), \qquad (3-1)$$

where  $\dim(H) = h$ . On the other hand, we have

$$\mathrm{CH}_{j+h}^G(X,i) \cong \mathrm{CH}_{j+h+l}^G(X \times V,i) \cong \mathrm{CH}_{j+h+l}^G(X \times U,i) \cong \mathrm{CH}_{j+h+l-g}(X \times^G U,i),$$

where the first isomorphism is due to the homotopy invariance, the second follows from the localization property (see Proposition 2.2(4)) as j is sufficiently small, and the third isomorphism follows from Proposition 2.2(7). The proof of the proposition now follows by combining this with (3-1).

**Corollary 3.2** (see [Edidin and Graham 2000]). Let  $H \subset G$  be a closed subgroup and let  $X \in \mathcal{V}_H$ . Then for any  $i \ge 0$ , there is a natural isomorphism

$$\operatorname{CH}^{G}_{*}(G \times^{H} X, i) \xrightarrow{\sim} \operatorname{CH}^{H}_{*}(X, i).$$
 (3-2)

*Proof.* Define an action of  $H \times G$  on  $G \times X$  by

$$(h,g) \cdot (g',x) = (gg'h^{-1},hx)$$
(3-3)

and an action of  $H \times G$  on X by  $(h, g) \cdot x = hx$ . Then the projection map  $G \times X \xrightarrow{p} X$  is  $(H \times G)$ -equivariant and is a G-torsor. Hence by Proposition 3.1, the natural map

$$\operatorname{CH}^{H}_{*}(X,i) \xrightarrow{p^{*}} \operatorname{CH}^{H \times G}_{*}(G \times X,i)$$
is an isomorphism. On the other hand, the projection map  $G \times X \to G \times^H X$  is  $(H \times G)$ -equivariant and is an *H*-torsor. Hence we get an isomorphism

$$\operatorname{CH}^G_*(G \times^H X, i) \xrightarrow{\sim} \operatorname{CH}^{H \times G}_*(G \times X, i).$$

The corollary follows by combining these two isomorphisms.

**Proposition 3.3.** Let G be a connected reductive group over k. Let B be a Borel subgroup of G containing a maximal torus T over k. Then for any  $i \ge 0$ , the restriction map

$$\operatorname{CH}^{B}_{*}(X,i) \xrightarrow{r^{B}_{T,X}} \operatorname{CH}^{T}_{*}(X,i)$$
 (3-4)

is an isomorphism for any  $X \in \mathcal{V}_B$ .

*Proof.* By Corollary 3.2, we only need to show that

$$\operatorname{CH}^{B}_{*}(X,i) \xrightarrow{\sim} \operatorname{CH}^{B}_{*}(B \times^{T} X,i).$$
 (3-5)

By [M. Demazure 1970, XXII, 5.9.5], there exists a characteristic filtration

$$B^u = U_0 \supseteq U_1 \supseteq \cdots \supseteq U_n = \{1\}$$

of the unipotent radical  $B^u$  of B such that  $U_{j-1}/U_j$  is a vector group, each  $U_j$  is normal in B and  $TU_j = T \ltimes U_j$ . Moreover, this filtration also implies that for each j, the natural map  $B/TU_j \rightarrow B/TU_{j-1}$  is a torsor under the vector bundle  $U_{j-1}/U_j \times B/TU_{j-1}$  on  $B/TU_{j-1}$ . Hence, the homotopy invariance gives an isomorphism

$$\operatorname{CH}^B_*(B/TU_{j-1} \times X, i) \xrightarrow{\sim} \operatorname{CH}^B_*(B/TU_j \times X, i).$$

Composing these isomorphisms successively for j = 1, ..., n, we get

$$\operatorname{CH}^{B}_{*}(X,i) \xrightarrow{\sim} \operatorname{CH}^{B}_{*}(B/T \times X,i).$$

The canonical isomorphism of *B*-varieties  $B \times^T X \cong B/T \times X$  (see Corollary 3.2) now proves (3-5) and hence (3-4).

Recall that a linear algebraic group *G* over *k* of dimension *g* is *diagonalizable* if  $G_{\bar{k}} \cong H \times (\mathbb{G}_m)^g$ , where *H* is a finite abelian group. The group *G* is called split diagonalizable, if such an isomorphism is defined over *k*. A connected reductive group *G* over *k* is said to be *split* if it contains a split maximal torus *T* over *k* such that *G* is given by a root system relative to *T*. Every connected and reductive group containing a split maximal torus is split; see [M. Demazure 1970, Chapter XXII, Proposition 2.1].

Recall from [Springer 1998, Lemma 14.1.1] that every solvable group *G* over *k* has a filtration  $\{e\} = G_0 \subseteq G_1 \subseteq \cdots \subseteq G_n = G$  by closed normal *k*-subgroups such that each quotient group  $G_j/G_{j-1}$  is either diagonalizable or an elementary

unipotent group; see [Springer 1998, §3.4]. The group *G* is called *split* over *k* if each  $G_j/G_{j-1}$  is either split diagonalizable or  $\mathbb{G}_a$ . It is known [Springer 1998, Corollary 14.3.10] that every unipotent group over a perfect field is split.

**Proposition 3.4.** Let H be a possibly nonreductive group over k. Assume that H has a Levi decomposition  $H = L \ltimes H^u$  such that  $H^u$  is split over k (for example, if k is of characteristic zero). Then for each  $i \ge 0$ , the map

$$\operatorname{CH}^{H}_{*}(X,i) \xrightarrow{r_{L,X}^{H}} \operatorname{CH}^{L}_{*}(X,i)$$
 (3-6)

is an isomorphism.

*Proof.* Since the unipotent radical of H is split over k, the proof is exactly same as in the proof of (3-4), where we just have to replace B and T by H and L, respectively.

**3a.** A structure theorem for  $CH_*^T(X)$ . We end this section with the following structure theorem for the equivariant higher Chow groups of a scheme with the action of a split diagonalizable group on which certain subgroup acts trivially. This theorem is the initial step in the proof of its far reaching generalization in Theorem 10.3.

**Theorem 3.5.** Let T be a split diagonalizable group and let  $X \in \mathcal{V}_T$ . Let H be a connected closed subgroup of T that acts trivially on X. Then there is a natural isomorphism

$$\operatorname{CH}^{T/H}_*(X) \otimes_{\mathbb{Z}} S(H) \xrightarrow{i^T_{H,X}} \operatorname{CH}^T_*(X).$$

This is a bigraded ring isomorphism if X is smooth.

*Proof.* Put T' = T/H. Since *H* is a split torus, we can choose a decomposition (not necessarily canonical)  $T = H \times T'$ . Fix an integer  $j \leq \dim(X)$  and let (V, U) and (V', U') be good pairs for the actions of *H* and *T'*, respectively, corresponding to  $\dim(X) - j$  as in [Edidin and Graham 1998, Example 3.1]. Thus *U* is a product of punctured affine spaces and  $U/H = (\mathbb{P}^n)^r$  for some  $n \gg 0$ , where  $r = \operatorname{rank}(H)$ . Then  $(V_T, U_T)$ , with  $V_T = V \times V'$  and  $U_T = U \times U'$ , is a good pair for the action of *T*. We now have

$$X_T = (X \times U \times U')/(H \times T') = (X \times U') \times^{T'} U/H = X_{T'} \times (\mathbb{P}^n)^r,$$

where the second equality holds since H acts trivially on  $X \times U'$  and the third equality holds because T' acts trivially on U. It follows from the projective bundle formula (see also Lemma 6.2) for the ordinary higher Chow groups that the map

$$\operatorname{CH}_{*}(X_{T'}) \otimes_{\mathbb{Z}} \operatorname{CH}_{*}((\mathbb{P}^{n})^{r}, 0) \to \operatorname{CH}_{*}(X_{T})$$
 (3-7)

is an isomorphism. We conclude the proof by noting that  $\operatorname{CH}_p(X_{T'}) \cong \operatorname{CH}_p^{T'}(X)$ and  $\operatorname{CH}_p(X_T) \cong \operatorname{CH}_p^T(X)$  for all  $p \leq j$ . If X is smooth, the assertion about the ring isomorphism of  $i_{H,X}^T$  now follows because (3-7) is known to be a bigraded ring isomorphism in that case.

## 4. Self-intersection and projection formulas

Our aim in this section is to prove the following two results for the ordinary and equivariant higher Chow groups. The first result is the self-intersection formula for the higher Chow groups of smooth schemes. The analogue of this formula for the higher *K*-theory was proven by Thomason [1993, Theorem 3.1]. Surprisingly, this formula for the higher Chow groups has remained unnoticed. Its equivariant version will play a very crucial role in the decomposition Theorem 10.3 for the equivariant higher Chow groups of smooth schemes with an action of a diagonalizable group.

The second result of this section is a version of projection formula for the higher Chow groups of singular schemes. Such a formula for the smooth schemes was proven by Bloch [1986]. We shall need this version of the projection formula in our construction of Demazure operators on the equivariant higher Chow groups.

**4a.** *Self-intersection formula.* We shall use the method of *deformation to the normal cone* as the main technical tool to prove the self-intersection formula. Since this technique will be used several times in this paper, we briefly recall the construction from [Fulton 1984, Chapter 5] for our, as well as reader's, convenience. Let *X* be a smooth scheme over *k* and let  $f : Y \hookrightarrow X$  be a smooth closed subscheme of codimension  $d \ge 1$ . Let  $\widetilde{M}$  be the blow-up of  $X \times \mathbb{P}^1$  along  $Y \times \infty$ . Then  $Bl_Y(X)$  is a closed subscheme of  $\widetilde{M}$  and one denotes its complement by *M*. There is a natural map  $\pi : M \to \mathbb{P}^1$  such that  $\pi^{-1}(\mathbb{A}^1) \cong X \times \mathbb{A}^1$  with  $\pi$  the projection map and  $\pi^{-1}(\infty) \cong X'$ , where X' is the total space of the normal bundle  $N_{Y/X}$  of *Y* in *X*. One also gets the following diagram, where all the squares and the triangles commute.



In this diagram, all the vertical arrows are the closed embeddings,  $i_0$  and  $i_\infty$  are the obvious inclusions of Y in  $Y \times \mathbb{P}^1$  along the specified points, i and j are inclusions of the inverse images of  $\infty$  and  $\mathbb{A}^1$ , respectively, under the map  $\pi$ , u and f' are zero section embeddings and  $p_Y$  is the projection map. In particular, one has  $p_Y \circ i_0 = p_Y \circ i_\infty = \mathrm{id}_Y$ .

In case X is a G-scheme and Y is G-invariant, then by letting G act trivially on  $\mathbb{P}^1$  and diagonally on  $X \times \mathbb{P}^1$ , one gets a natural action of G on M, and all the spaces in the diagram above become G-spaces and all the morphisms become G-equivariant. This observation will be used later in this paper.

We shall need the following result about the higher Chow groups, which is an easy consequence of Bloch's moving lemma.

#### Lemma 4.1. Let



be a fiber diagram of closed immersions of schemes such that X and Y are smooth and Y and Z intersect properly in X. Then one has for each  $i \ge 0$ ,

$$f^* \circ g_* = g'_* \circ f'^* : \operatorname{CH}_*(Y, i) \to \operatorname{CH}_*(Z, i)$$

*Proof.* Since X and Y are smooth, we can assume them to be equidimensional. Let

$$\mathscr{Z}^p_{ZW}(Y, \bullet) \stackrel{i_Y}{\hookrightarrow} \mathscr{Z}^p(Y, \bullet)$$

be the subcomplex that is generated by cycles on  $Y \times \Delta^{\bullet}$  which intersect all faces of  $Z \times \Delta^{\bullet}$  and  $W \times \Delta^{\bullet}$  properly. Similarly, let

$$\mathcal{Z}^p_Z(X,\,\bullet) \stackrel{i_X}{\hookrightarrow} \mathcal{Z}^p(X,\,\bullet)$$

be the subcomplex generated by the cycles on  $X \times \Delta^{\bullet}$  that intersect all faces of  $Z \times \Delta^{\bullet}$  properly. Then  $i_X$  and  $i_Y$  are quasi-isomorphisms by the moving lemma; see [Krishna and Levine 2008, Theorem 1.10]. However, if  $V \in \mathscr{Z}^p_{ZW}(Y, \bullet)$  is an irreducible cycle in  $Y \times \Delta^n$ , then the conclusion of the lemma is checked easily.  $\Box$ 

Corollary 4.2. Let G be a linear algebraic group and let



be a fiber diagram of closed immersions of smooth *G*-schemes such that *Y* and *Z* intersect properly in *X*. Then one has  $f^* \circ g_* = g'_* \circ f'^* : CH^*_G(Y, i) \to CH^*_G(Z, i)$ . *Proof.* By choosing a good pair (V, U) for the *G*-action and then considering the appropriate mixed quotients, we can reduce to proving the corollary for the nonequivariant higher Chow groups. But this is shown in Lemma 4.1.

**Lemma 4.3.** Consider the diagram (4-1) and let  $y \in CH^*(Y, m)$ . Then there exists  $z \in CH^*(M, m)$  such that  $f_*(y) = h^*(z)$  and  $f'_*(y) = i^*(z)$ .

*Proof.* Put  $\tilde{y} = p_Y^*(y)$  and  $z = F_*(\tilde{y})$ . Then

$$f_{*}(y) = f_{*}((p_{Y} \circ i_{0})^{*}(x)) = f_{*} \circ i_{0}^{*} \circ p_{Y}^{*}(y) = f_{*} \circ i_{0}^{*}(\tilde{y}) = f_{*} \circ u'^{*} \circ j'^{*}(\tilde{y})$$
  
=  $u^{*} \circ F'_{*}(j'^{*}(\tilde{y}))$  (by Lemma 4.1)  
=  $u^{*} \circ j^{*} \circ F_{*}(\tilde{y})$  (since *j* is an open immersion)  
=  $h^{*} \circ F_{*}(\tilde{y}) = h^{*}(z)$ .

Similarly,

$$f'_{*}(y) = f'_{*}((p_{Y} \circ i_{\infty})^{*}(x)) = f'_{*} \circ i_{\infty}^{*} \circ p_{Y}^{*}(y) = f'_{*} \circ i_{\infty}^{*}(\tilde{y})$$
  
=  $i^{*} \circ F_{*}(\tilde{y})$  (by Lemma 4.1)  
=  $i^{*}(z)$ .

**Theorem 4.4** (self-intersection formula). Let  $Y \xrightarrow{f} X$  be a closed immersion of smooth varieties of codimension  $d \ge 0$ , and let  $N_{Y/X}$  be the normal bundle of Y in X. Then for every  $y \in CH^*(Y, i)$ , one has  $f^* \circ f_*(y) = c_d(N_{Y/X}) \cdot y$ .

*Proof.* There is nothing to prove when d = 0 and so we assume  $d \ge 1$ . We first consider the case when  $X \xrightarrow{p} Y$  is a vector bundle of rank d and f is the zero section embedding so that  $p \circ f = id_Y$ . In that case, we have

$$f^* \circ f_*(y) = f^* \circ f_*(f^* \circ p^*(y)) = f^*(f_*(1) \cdot p^*(y)) \quad \text{(by Proposition 2.2(6))}$$
$$= f^*(f_*(1)) \cdot (f^* \circ p^*(y)) = f^*(f_*(1)) \cdot y = c_d(N_{Y/X}) \cdot y,$$

where the last equality follows from the self-intersection formula for Fulton's Chow groups; see [Fulton 1984, Corollary 6.3]. This proves the theorem in the case of zero section embedding.

Now let  $Y \hookrightarrow X$  be as in the theorem. We consider the deformation to the normal cone diagram (4-1) and choose  $z \in CH^*(M, i)$  as in Lemma 4.3. Then we have

$$f^* \circ f_*(y) = f^* \circ h^*(z) = i_0^* \circ F^*(z) = i_\infty^* \circ F^*(z) = f'^* \circ i^*(z)$$
$$= f'^* \circ f'_*(y) \qquad \text{(by Lemma 4.3)}$$
$$= c_d(N_{Y/X'}) \cdot y \qquad \text{(by the case of vector bundle above)}$$
$$= c_d(N_{Y/X}) \cdot y.$$

This completes the proof of the theorem.

**Corollary 4.5.** Let G be a linear algebraic group over k and let  $Y \xrightarrow{f} X$  be a closed immersion of codimension  $d \ge 0$  in  $\mathcal{V}_G^S$ . Then for every  $i \ge 0$  and  $y \in CH_G^*(Y, i)$ , one has  $f^* \circ f_*(y) = c_d^G(N_{Y/X}) \cdot y$ .

*Proof.* There is nothing to prove if d = 0 and so we assume  $d \ge 1$ . Fix  $i, j \ge 0$  and choose a good pair (V, U) for  $n \gg j + d$ . We can then identify  $\operatorname{CH}_G^P(X, i)$  with  $\operatorname{CH}^P(X_G, i)$  (and same for Y) for  $p \le n$ . We can also identify  $c_d^G(E)$  with  $c_d(E_G)$  for any equivariant vector bundle E on Y; see [Edidin and Graham 1998, Section 2.4]. Now, the proof of the corollary would follow straightaway from Theorem 4.4, once we show that  $(N_{Y/X})_G$  is the normal bundle of  $Y_G$  in  $X_G$ . But this follows immediately from the elementary fact that if G acts freely on a smooth scheme Z and W is a smooth closed and G-invariant subscheme of Z with normal bundle N, then G acts freely on N, and moreover, N/G is the normal bundle of W/G in Z/G. We leave the proof of this fact to the reader.

**4b.** A projection formula for singular schemes. Recall from Section 2b (see also [Bloch 1986, Corollary 5.6]) that the operational Chow groups  $X \mapsto OPCH^*(X)$  form a ring-valued contravariant functor on  $\mathcal{V}_k$  that acts on the higher Chow groups. The action of  $OPCH^1(X, 0) \xrightarrow{\sim} Pic(X)$  on  $CH_*(X, j)$  coincides with the action of the Chern classes of line bundles.

**Proposition 4.6.** Let  $X \in \mathcal{V}_k$  and let  $f : Y = \mathbb{P}(E) \to X$  be the projective bundle associated to a vector bundle E of rank n + 1 on X and let

$$\xi = c_1(\mathbb{O}_Y(1)) \in \operatorname{OPCH}^1(Y, 0)$$

be the first Chern class of the relative tautological line bundle on Y. Then for any  $x \in CH_*(X, j)$ , one has

$$f_*(\xi^i \cdot f^*(x)) = \begin{cases} 0 & \text{if } i < n, \\ x & \text{if } i = n. \end{cases}$$

*Proof.* If *X* is smooth, the proposition is an easy consequence of the projection formula [Bloch 1986, Exercise 5.8], as this formula implies that

$$f_*(\xi^i \cdot f^*(x)) = f_*(\xi^i) \cdot x.$$

Moreover, it follows from [Fulton 1984, Proposition 3.1] that  $f_*(\xi^i) = 1$  if i = n and zero otherwise. The case of singular schemes is the hard part of the theorem because we cannot directly apply the projection formula of [Bloch 1986]. We obtain a proof by an indirect approach of reduction to the smooth case and by unravelling the action of Chern classes on the higher Chow cycles on singular schemes.

462

By [Fulton 1984, Lemma 18.2], we can find a closed embedding  $\iota : X \to X'$  and a vector bundle E' of rank n + 1 on X' such that  $E \cong \iota^*(E')$  and X' is smooth. We set  $Y' = \mathbb{P}(E')$ ,  $\xi' = c_1(\mathbb{O}_{Y'}(1))$  and consider the Cartesian diagram

Recall from the construction of the action of  $\xi'^i$  on  $CH_*(Y, j)$  in [Bloch 1986] that for any irreducible cycle [V] on  $Y \times \Delta^j$ , the support of  $\xi^i \cdot [V] = \xi'^i \cdot [V]$  is supp $(V \cap \alpha)$ , where  $\alpha$  is cycle on Y' representing  $\xi'^i$  and such that each component of  $\alpha$  intersects V properly. This is achieved by using the moving lemma on Y', a smooth scheme. Since  $\xi'$  reduces the dimension of a cycle on  $Y \times \Delta^j$  by exactly one, we see that dim $(\xi^i \cdot f^*([W])) = \dim(W) + n - i$  and the support of  $f_*(\xi^i \cdot f^*([W]))$ is contained in W, whenever W is an irreducible admissible cycle on  $X \times \Delta^j$ . We conclude from the definition of the push-forward map that  $f_*(\xi^i \cdot f^*([W]))$  must be zero if n - i > 0. Since any admissible cycle on  $X \times \Delta^j$  is a sum of irreducible admissible cycles, this proves the first case.

We prove the case of i = n by induction on n. If E is of rank one, then f is an isomorphism and  $\mathbb{O}_Y(1)$  is trivial and hence  $c_0(\mathbb{O}_Y(1)) = 1$ . So we assume that  $n \ge 1$ . We let  $\iota : X \hookrightarrow X'$  be a closed embedding into a smooth scheme as in (4-2).

By [Panin 2003, Lemma 3.24], there is a morphism  $\phi': T' \to X'$ , which is a composite of projective and affine bundles on X' such that  $\phi'^*(E') = F' \oplus L'$ , where L' is a line bundle on T'. Moreover, if  $\phi: T \to X$  is the restriction of  $\phi'$  on X, then the pull-back map  $\phi^*: CH_*(X, j) \to CH_*(T, j)$  is a split injection and the same holds for  $\phi'^*$ . Notice here that X is a closed subscheme of X' and hence  $CH_*(X) \xrightarrow{\sim} CH^X_*(X')$  in the notation of [ibid., Definition 2.1]. We denote the restrictions of F' and L' on X by F and L respectively.

Consider the Cartesian diagram

$$Z \xrightarrow{\psi} Y$$

$$g \downarrow \qquad \qquad \downarrow f$$

$$T \xrightarrow{\phi} X$$

$$(4-3)$$

and suppose the given assertion holds for the projective bundle g. We then have

$$\phi^*(f_*(\xi^n \cdot f^*(x))) = g_*(\psi^*(\xi^n \cdot f^*(x))) = g_*((\psi^*(\xi))^n \cdot \psi^* \circ f^*(x))$$
  
=  $g_*((\psi^*(\xi))^n \cdot g^* \circ \phi^*(x)) = \phi^*(x),$ 

where the first equality holds by Proposition 2.2 and the last equality holds by our assumption. Since  $\phi^*$  is injective, we see that the conclusion holds for f as well. Thus we have reduced the problem to the case when  $E' = F' \oplus L'$  and  $E = F \oplus L$ .

If we set  $\widetilde{E} = E \otimes L^{-1}$  and  $\widetilde{Y} = \mathbb{P}(\widetilde{E})$ , there is a commutative diagram



such that *h* is an isomorphism and  $\mathbb{O}_{\widetilde{Y}}(1) = h^*(\mathbb{O}_Y(1)) \otimes \widetilde{f}^*(L^{-1})$ . Set  $\eta = c_1(L)$ in OPCH<sup>1</sup>(*X*, 0) and  $\widetilde{\eta} = \widetilde{f}^*(\eta)$  and  $\widetilde{\xi} = c_1(\mathbb{O}_{\widetilde{Y}}(1))$  in OPCH<sup>1</sup>( $\widetilde{Y}, 0$ ).

Suppose that our assertion holds for the projective bundle  $\tilde{f}$ . In this case, we get

$$f_*(\xi^n \cdot f^*(x)) = \tilde{f}_*((\tilde{\eta} + \tilde{\xi})^n \cdot \tilde{f}^*(x)) = \tilde{f}_*((\tilde{\xi})^n \cdot \tilde{f}^*(x)) = x$$

where the last equality holds by our assertion about  $\tilde{f}$ . The second equality holds because of the fact that

$$\tilde{f}_*((\eta+\tilde{\xi})^n\cdot\tilde{f}^*(x)) = \sum_{i=0}^n \binom{n}{i} \tilde{f}_*((\tilde{\eta})^i(\tilde{\xi})^{n-i}\cdot\tilde{f}^*(x))$$
$$= \sum_{i=0}^n \binom{n}{i} \eta^i \cdot \tilde{f}_*((\tilde{\xi})^{n-i}\cdot\tilde{f}^*(x))$$

(see [Bloch 1986, Exercise 5.8]) and that the term  $\tilde{f}_*((\tilde{\xi})^{n-i} \cdot \tilde{f}^*(x))$  vanishes for  $0 \le n-i < n$  by the first assertion of proposition. Hence, we are further reduced to the case when  $E' = F' \oplus \mathbb{O}_{X'}$  and  $E = F \oplus \mathbb{O}_X$ .

Let  $Z = \mathbb{P}(F)$  and let p and q be the closed and the open inclusions of  $Z \hookrightarrow Y$ and  $F \hookrightarrow Y$ , respectively. Let  $g : Z \hookrightarrow Y \to X$  be the composite map and set  $\zeta = c_1(\mathbb{O}_Z(1))$ . We observe that Z is a Cartier divisor on Y such that  $\mathbb{O}_Y(1) = \mathcal{L}(Z)$ . In particular, the pull-back  $p^*$  is defined and we have  $\xi \cdot f^*(a) = p_*(p^* \circ f^*(a))$ . Since F is of rank n, the desired assertion holds for  $g : Z \to X$  by induction. That is,  $g_*(\zeta^{n-1} \cdot g^*(x)) = x$ . We now have

$$f_*(\xi^n \cdot f^*(x)) = f_*(\xi^{n-1} p_*(p^* \circ f^*(x)))$$
  
=  $f_*(p_*(\zeta^{n-1} \cdot p^* \circ f^*(x))) = g_*(\zeta^{n-1} \cdot g^*(x)) = x.$ 

This proves the desired assertion and the proof of the proposition is complete.  $\Box$ 

#### 5. Demazure operators on equivariant higher Chow groups

In this section, we introduce Demazure (divided difference) operators on the equivariant higher Chow groups of schemes. Such operators were constructed by Brion

[1997] on the equivariant Chow groups  $CH^G_*(X, 0)$  and by Holm and Sjamaar [2008] on the equivariant singular cohomology  $H^*_G(X)$ . We extend these operators to the higher Chow groups and discuss some consequences.

Let *G* be a connected reductive group with a split maximal torus *T* of rank *n*. Let  $\Phi = \Phi(G, T) \subset \mathcal{X}(T)$  be the root system of *G* with respect to *T*, where  $\mathcal{X}(T)$  is the character group of *T*. For any  $\alpha \in \Phi$ , let  $P_{\alpha}$  be the minimal parabolic subgroup of *G* corresponding to  $\alpha$  and let  $B = TU_{\alpha}U$  and  $B' = TU_{-\alpha}U$  be the opposite Borel subgroups of *G* containing *T* in  $P_{\alpha}$ . Let  $W_{\alpha} = \{s_{\alpha}, s_{-\alpha}\}$  denote the Weyl group of  $P_{\alpha}$ , where  $s_{\alpha}$  is the reflection in  $\mathcal{X}(T)$  given by

$$s_{\alpha}(\lambda) = \lambda - \langle \alpha^{\vee}, \lambda \rangle \alpha$$
 for  $\lambda \in \mathscr{X}(T)$ .

Let X be a k-scheme with a free G-action and let B and B' act on  $X \times P_{\alpha}$  by  $b \cdot (x, g) = (x, bg)$  and  $b' \cdot (x, g) = (b'x, gb'^{-1})$ . It is easy to check that the two actions are free and they commute with each other. Hence we get a free action of  $B \times B'$  on  $X \times G$  by  $(b, b') \cdot (x, g) = (b'x, bgb'^{-1})$ . One checks that B acts freely on  $X' = X \times^{B'} P_{\alpha}$ .  $P_{\alpha}$  acts freely on X' by acting trivially on its X-factor and by left multiplication on the  $P_{\alpha}$ -factor. In particular, we have a B-equivariant map  $X' \to X$  given by  $[x, g] \mapsto x$ , which yields the projection map on quotients  $f_1 : X'/B \to X/B \cong X'/P_{\alpha}$ . One also has the  $P_{\alpha}$ -equivariant map  $X' \to X$  given by its the map  $f_2 : X'/B \to X/B$  on quotients by the action of B. It is also easy to check that the data above yield the commutative diagram



which is Cartesian. The section  $\sigma_2$  of  $f_2$  is defined by  $\sigma_2([x]) = [x, 1]$  and the section  $\sigma_1$  of  $f_1$  is defined by  $\sigma_1([x]) = [x, n_\alpha]$ , where  $n_\alpha$  is a representative of  $s_\alpha$  in  $N_{P_\alpha}(T)$ . Since  $f_2$  is induced by a  $P_\alpha$ -equivariant map, we see that it is  $W_\alpha$ -equivariant with respect to the natural action of  $W_\alpha$  on X'/B and X/B. In particular, we get

$$f_2^* \circ s_\alpha = s_\alpha \circ f_2^*,$$
  

$$f_2^*(ax) = af_2^*(x) \quad \text{for all } a \in S \text{ and } x \in \operatorname{CH}_*(X/B).$$
(5-2)

Note that the all the maps (except the sections) in (5-1) are  $P_{\alpha}/B \cong \mathbb{P}^1$ -bundles and hence they are all smooth and projective. Let  $\mathbb{O}_{X/B}(1)$  denote the universal quotient line bundle on X/B relative to  $p_X$ . Amalendu Krishna

Let *D* denote the image of  $\sigma_1$  and let  $\mathscr{L}(D)$  denote the associated line bundle on X'/B. Let  $\mathscr{L}_{\lambda}$  and  $\mathscr{L}'_{\lambda}$  denote the line bundles  $X \times^B L_{\lambda}$  and  $X' \times^B L_{\lambda}$  on X/B and X'/B respectively, where  $L_{\lambda}$  is the *B*-equivariant line bundle on Spec(*k*) corresponding to the character  $\lambda$  of *T*. It follows from [Demazure 1974, Lemme 2, Proposition 2] that

$$\mathcal{L}'_{\alpha} = f_1^*(\mathcal{L}_{\alpha}) \otimes (\mathcal{L}(D))^{\otimes 2},$$
  

$$s_{\alpha}(\mathcal{L}(D)) = (\mathcal{L}(D) \otimes f_1^*(\mathcal{L}_{\alpha}))^{\otimes (-1)}.$$
(5-3)

**5a.** *Demazure operators on*  $CH_*(X)$ . Let *G* be a connected reductive group with split maximal torus *T* as above and let *X* be a *G*-scheme of dimension *d*. Let  $j \le d$  be an integer and let (V, U) be a good pair for the *G*-action corresponding to d - j. The smooth and projective morphism  $p_{X \times U} : X_B \to X_{P_\alpha}$  yields the maps

$$p_{X \times U}^*: \operatorname{CH}_j^{P_{\alpha}}(X, i) \to \operatorname{CH}_j^B(X, i) \text{ and } p_{X \times U_*}: \operatorname{CH}_j^B(X, i) \to \operatorname{CH}_{j+1}^{P_{\alpha}}(X, i).$$
 (5-4)

For the rest of this text, the ring S will denote the equivariant Chow ring

$$S(T) = \operatorname{CH}_{T}^{*}(k, 0).$$

**Lemma 5.1.** The maps  $p_{X \times U}^*$  and  $p_{X \times U*}$  do not depend on the choice of the good pair (V, U).

*Proof.* We prove the lemma for the push-forward map and a very similar proof works also for the pull-back map; see [Edidin and Graham 1998, Section 1].

Let g and b denote the dimensions of  $P_{\alpha}$  and B, respectively. Let (V, U) and (V', U') be good pairs of dimensions l and l', respectively, corresponding to d - j. We let  $\mathcal{V} = V \oplus V'$  and  $\mathcal{U} = (U \oplus V') \cup (V \oplus U')$ . Let G act diagonally on  $\mathcal{V}$ . Then it is easy to see that the dimension of the complement of the open subset  $X \times^B (U \oplus V')$  in  $X \times^B \mathcal{U}$  is sufficiently smaller than l + l' - b + j. Similarly, the dimension of the complement of the open subset  $X \times^{P_{\alpha}} (U \oplus V')$  in  $X \times^{P_{\alpha}} \mathcal{U}$  is sufficiently smaller than l + l' - b + j. It follows from the localization sequence for the higher Chow groups and Lemma 5.2 that the there is a commutative diagram

where the vertical maps are the push-forward maps and the horizontal maps are isomorphisms.

On the other hand, the maps

$$X \times^G (U \oplus V') \to X \times^G U$$
 and  $X \times^B (U \oplus V') \to X \times^B U$ 

are vector bundles of rank l' and hence we get another commutative diagram

where the vertical arrows are the push-forward maps and the horizontal arrows are isomorphisms by the homotopy invariance.

Combining the two isomorphisms above, we get the commutative diagram

where the horizontal maps are isomorphisms. By repeating the same argument with U', we get the diagram above with U replaced by U' and V' replaced by V on the right column. This proves the lemma.

**Lemma 5.2.** Let  $p: X \to Y$  be a morphism in  $\mathcal{V}_{P_{\alpha}}$  such that  $P_{\alpha}$  acts freely on X and Y. Then the diagram of quotients



is Cartesian such that the vertical maps are smooth and projective.

*Proof.* This is an easy exercise.

**Proposition 5.3.** For any  $X \in \mathcal{V}_G$ , one has the restriction and the push-forward maps

$$r_{B,X}^{P_{\alpha}}: \operatorname{CH}^{P_{\alpha}}_{*}(X,i) \to \operatorname{CH}^{B}_{*}(X,i) \quad and \quad p_{B,X}^{P_{\alpha}}: \operatorname{CH}^{B}_{*}(X,i) \to \operatorname{CH}^{P_{\alpha}}_{*+1}(X,i).$$

These maps are contravariant with respect to the flat maps and covariant with respect to the proper morphisms of schemes in  $\mathcal{V}_G$ .

*Proof.* Let  $j \leq d$  be an integer and let (V, U) be a good pair for the *G*-action corresponding to d - j. We define  $r_{B,X}^{P_{\alpha}}$  and  $p_{B,X}^{P_{\alpha}}$  to be  $p_{X \times U}^*$  and  $p_{X \times U*}$  respectively. It follows from Lemma 5.1 that these maps are well-defined.

The functoriality properties of  $r_{B,X}^{P_{\alpha}}$  is already known; see [Edidin and Graham 1998, Section 1]. To prove these properties for  $p_{B,X}^{P_{\alpha}}$ , it suffices to prove the same

for  $p_{X \times U_*}$ . But this follows easily from Lemma 5.2 and the similar properties of the ordinary higher Chow groups.

The following result generalizes [Brion 1997, Theorem 6.3] to equivariant higher Chow groups.

**Theorem 5.4.** Let  $\alpha$  be a simple root of G. For any  $X \in \mathcal{V}_G$  and  $i \geq 0$ , there is a unique operator  $\delta^X_{\alpha}$  on  $\operatorname{CH}^T_*(X, i)$  such that for all  $u \in S$  and  $v \in \operatorname{CH}^T_*(X, i)$ , we have

(1)  $\alpha \delta^X_{\alpha}(v) = v - s_{\alpha}(v)$  and

(2)  $\delta^X_{\alpha}(uv) = u\delta^X_{\alpha}(v) + \delta^k_{\alpha}(u)s_{\alpha}(v)$  if X is smooth.

Moreover,  $\delta_{\alpha}^{X}$  commutes with the *G*-equivariant flat pull-back and proper pushforward maps between *T*-equivariant higher Chow groups.

*Proof.* Let *B* and *B'* be the opposite Borel subgroups of  $P_{\alpha}$  containing *T*. Using Proposition 3.3, we can replace *T* by *B* to define  $\delta_{\alpha}^{X}$ . We let

$$\delta^X_{\alpha} := r^{P_{\alpha}}_{B,X} \circ p^{P_{\alpha}}_{B,X} : \operatorname{CH}^B_j(X,i) \to \operatorname{CH}^B_{j+1}(X,i).$$
(5-5)

The co- and contravariant functoriality of  $\delta_{\alpha}^{X}$  follows from Proposition 5.3. The uniqueness of  $\delta_{\alpha}^{X}$  follows from [Brion 1997, Theorem 6.3] since this definition of  $\delta_{\alpha}^{X}$  coincides with the one defined for  $CH_{*}^{B}(X, 0)$  by Brion. We only need to show the first and the second assertions.

Let  $j \leq \dim(X)$  and (V, U) be a good pair the *G*-action corresponding to d - j. There is a *G*-equivariant projection  $X \times U \to X$  such that

$$\operatorname{CH}_{i}^{G}(X,i) \xrightarrow{\sim} \operatorname{CH}_{i}^{G}(X \times U,i)$$

and  $\delta_{\alpha}^{X}$  on  $CH_{j}^{G}(X, i)$  coincides with the operator  $\delta_{\alpha}^{X \times U}$  on  $CH_{j}^{G}(X \times U, i)$  by its construction. Hence we can assume that *G* acts freely on *X*.

We now consider the diagram (5-1). Since  $f_2$  is a  $\mathbb{P}^1$ -bundle, the map  $f_2^*$  is split injective by Proposition 4.6. Hence it suffices to show that the two assertions of the theorem hold after applying  $f_2^*$ .

On the other hand,  $f_2^*(s_\alpha(v)) = s_\alpha(f_2^*(v))$  and  $f_2^*(\alpha \delta_\alpha^X(v)) = \alpha f_2^*(\delta_\alpha^X(v))$ by (5-2). Since  $f_2$  is induced by a *G*-equivariant map  $X' \to X$ , we also have  $f_2^*(\delta_\alpha^X(v)) = \delta_\alpha^{X'}(f_2^*(v))$  by the functoriality of  $\delta_\alpha$ . Thus we need to show for  $u \in S$  and  $v' \in CH_*(X'/B, i)$  that

$$\alpha \delta_{\alpha}^{X'}(v') = v' - s_{\alpha}(v') \quad \text{and} \quad \delta_{\alpha}^{X'}(uv') = u \delta_{\alpha}^{X'}(v') + \delta_{\alpha}^{k}(u) s_{\alpha}(v') \quad \text{if} \ X \in \mathcal{V}_G.$$
(5-6)

Let  $c : \operatorname{CH}_T^*(k, 0) \cong \operatorname{OPCH}_T^*(k, 0) \to \operatorname{OPCH}^*(X/B, 0)$  be the ring homomorphism on the operational Chow groups induced by the map on the Picard groups  $L_{\alpha} \mapsto \mathscr{L}_{\alpha}$ . We denote the corresponding map  $\operatorname{OPCH}_T^*(k, 0) \to \operatorname{OPCH}^*(X'/B, 0)$  by c'. We set  $\xi = c_1(\mathscr{L}(D))$  and  $\zeta = c_1(\mathbb{O}_{X'/B}(1))$ , where  $\mathbb{O}_{X'/B}(1)$  is the universal

quotient line bundle associated to the  $\mathbb{P}^1$ -bundle  $f_1$ . We shall write  $\delta_{\alpha}^k$  simply as  $\delta_{\alpha}$  in what follows.

Since  $\mathscr{L}(D)$  and  $\mathbb{O}_{X'/B}(1)$  have same degree on every fiber of  $f_1$ , there is a line bundle  $\mathscr{M}$  on X/B such that  $\mathbb{O}_{X'/B}(1) \cong \mathscr{L}(D) \otimes f_1^*(\mathscr{M})$ . In particular, there exists  $\eta \in \operatorname{OPCH}^1(X/B, 0)$  such that  $\zeta = \xi + f_1^*(\eta)$ . Since  $f_1^*$  commutes with the action of  $\operatorname{OPCH}^*(X/B, 0)$  on the higher Chow groups, it follows from Proposition 4.6 that we can write  $v' \in \operatorname{CH}_*(X'/B, i)$  as

$$v' = f_1^*(a) + \xi f_1^*(b), \quad \text{with } a, b \in CH_*(X/B, i).$$
 (5-7)

Furthermore, it also follows that for any  $b \in CH_*(X/B, i)$ ,

$$f_{1*}(\xi f_1^*(b)) = f_{1*}(\zeta f_1^*(b)) - f_{1*}(f_1^*(\eta b)) = b + 0 = b.$$
(5-8)

Since  $s_{\alpha}$  keeps the elements of the form  $f^*(a)$  invariant, we get

$$s_{\alpha}(v') = f_1^*(a) + s_{\alpha}(\xi) f_1^*(b)$$
  
=  $f_1^*(a) - [\xi + f_1^*(c(\alpha))] f_1^*(b),$  (5-9)

where the second equality follows from (5-3).

On the other hand, we have seen in (5-1) that  $f_1$  is same as the quotient map  $X'/B \to X'/P_{\alpha}$  and hence  $\delta_{\alpha}^{X'} = f_1^* f_{1*}$ . This yields

$$v' - c'(\alpha)\delta_{\alpha}^{X'}(v') = f_{1}^{*}(a) + \xi f_{1}^{*}(b) - c'(\alpha)f_{1}^{*}[f_{1*}f_{1}^{*}(a) + f_{1*}(\xi f_{1}^{*}(b))]$$
  

$$=^{\dagger} f_{1}^{*}(a) + \xi f_{1}^{*}(b) - c'(\alpha)f_{1}^{*}(b)$$
  

$$=^{\ddagger} f_{1}^{*}(a) + \xi f_{1}^{*}(b) - [f_{1}^{*}(c(\alpha)) + 2\xi]f_{1}^{*}(b)$$
  

$$= f_{1}^{*}(a) - [\xi + f_{1}^{*}(c(\alpha))]f_{1}^{*}(b), \qquad (5-10)$$

where  $=^{\dagger}$  follows from Proposition 4.6 and (5-8), and  $=^{\ddagger}$  follows from (5-3). The first equality of (5-6) follows at once by comparing (5-9) and (5-10).

To prove the second equality of (5-6) for  $u \in S$  and  $v' \in CH_*(X'/B, i)$  with X smooth, we can assume using (5-7) that v' is either  $f_1^*(a)$  or  $\xi f_1^*(a)$ . We now have

$$\delta_{\alpha}^{X}(uf_{1}^{*}(a)) = f_{1}^{*} \circ f_{1*}(uf_{1}^{*}(a)) =^{1} f_{1}^{*}(f_{1*}(u) \cdot a) = f_{1}^{*}f_{1*}(u) \cdot f_{1}^{*}(a)$$
$$=^{2} \delta_{\alpha}^{k}(u) \cdot f_{1}^{*}(a) = \delta_{\alpha}^{k}(u) \cdot s_{\alpha}(f_{1}^{*}(a)) + u\delta_{\alpha}^{X}(f_{1}^{*}(a)).$$

The equality  $=^1$  holds by the projection formula for smooth schemes (see [Bloch 1986, Exercise 5.8]) and  $=^2$  holds by [Brion 1997, Theorem 6.3]. The last equality holds because  $f_1^*(a)$  is invariant under  $s_\alpha$  and  $f_1^*f_{1*}(a)$  vanishes, again by the projection formula. The required formula for  $v' = \xi f_1^*(a)$  is proved exactly in the similar way using the observation that  $f_{1*}(\xi) = 1$  and that the equality  $=^2$  holds even if we replace u by  $\xi u$ .

**Proposition 5.5.** For  $X \in \mathcal{V}_G$ , let  $r_{T,X}^G : CH^G_*(X, i) \to CH^T_*(X, i)$  be the restriction map.

- (1)  $\delta_{\alpha} \circ r_T^G = 0.$
- (2) If X is smooth, then  $\delta_{\alpha}$  is  $CH^*_G(X)$ -linear.
- (3)  $\delta_{\alpha}^2 = 0.$
- (4)  $s_{\alpha}\delta_{\alpha} = \delta_{\alpha} = -\delta_{-\alpha}, \ \delta_{\alpha}s_{\alpha} = -\delta_{\alpha}.$

*Proof.* Since  $r_{T,X}^G = r_{T,X}^{P_{\alpha}} \circ r_{P_{\alpha},X}^G$ , we can replace G by  $P_{\alpha}$ . It suffices then to show that

$$p_{B,X}^{P_{\alpha}} \circ r_{B,X}^{P_{\alpha}} = 0.$$

But this follows immediately from Proposition 4.6. The second point follows from the observation that  $f_1^*$  and  $f_{1*}$  in (5-1) are  $CH^*(X'/G)$ -linear. The third point follows directly from the first and the fourth point is an immediate consequence of the other assertions of the proposition.

**5b.** *Ring of Demazure operators.* Let  $\{\alpha_1, \ldots, \alpha_m\}$  be the set of all simple roots of *G*. For any sequence  $I = \{i_1, \ldots, i_l\}$  of integers in the interval [1, m], we define the operator  $\delta_I^X$  on  $CH_*^T(X, i)$  by

$$\delta_I^X = \delta_{\alpha_{i_l}} \circ \dots \circ \delta_{\alpha_{i_1}}. \tag{5-11}$$

Following the notation of [Brion 1997, §6.4], we let **D** denote the subring of  $\operatorname{End}_{\mathbb{Z}}(S)$  generated by the elements  $\delta_{\alpha}^{k}$  and the endomorphisms given by the multiplication in *S*. It is clear from the definition of **D** and Theorem 5.4 that **D** contains the twisted group algebra *S*[*W*] and there are inclusions of rings  $S \subsetneq S[W] \subsetneq \mathbf{D}$ . It is known that **D** is a free *S*-module with basis  $\{\partial_w\}_{w\in W}$ , where  $\partial_w$  is same as  $\delta_I^k$  above whenever  $w = s_{\alpha_{i_1}} \cdots s_{\alpha_{i_l}}$ . As an immediate consequence of Theorem 5.4, Proposition 5.5 and (5-11), we get:

**Corollary 5.6.** For any  $X \in \mathcal{V}_G$  and  $i \geq 0$ , there is a unique **D**-module structure on  $\operatorname{CH}^T_*(X, i)$ , which extends the action of S[W]. Moreover, the flat pull-back and proper push-forward maps between the *T*-equivariant higher Chow groups are **D**-linear. For  $X \in \mathcal{V}^S_G$ , the **D**-module structure on  $\operatorname{CH}^*_T(X)$  commutes with its  $\operatorname{CH}^*_G(X)$ -module structure.

Let  $I(\mathbf{D})$  be the subset of **D** consisting of those operators  $\delta$  such that  $\delta(1) = 0$ . It is easy to check that  $I(\mathbf{D})$  is a left ideal of **D** generated by  $\{\partial_w\}_{w\neq 1}$ . For any  $X \in \mathcal{V}_G$ , let

$$(\mathrm{CH}^{T}_{*}(X,i))^{I(\mathbf{D})} = \left\{ x \in \mathrm{CH}^{T}_{*}(X,i) \mid \delta(x) = 0 \; \forall \delta \in I(\mathbf{D}) \right\}.$$
(5-12)

Since the Weyl group is generated by simple reflections, it follows from Theorem 5.4 and Proposition 5.5 that

$$r_{T,X}^G(\mathrm{CH}^G_*(X,i)) \subseteq (\mathrm{CH}^T_*(X,i))^{I(D)} \subseteq (\mathrm{CH}^T_*(X,i))^W.$$
 (5-13)

Recall that the torsion index  $t_G$  of the group G is the order of the cokernel of the map  $S_N \to \operatorname{CH}^N(G/B)$ , where  $N = \dim(G/B) = |\Phi^+|$ . Let R denote the localized ring  $\mathbb{Z}[t_G^{-1}]$ .

**Theorem 5.7.** Let  $X \in \mathcal{V}_G^S$  and  $i \ge 0$  be such that  $\operatorname{CH}_T^*(X, i)$  is torsion-free. Then the map  $\operatorname{CH}_*^G(X, i) \to (\operatorname{CH}_*^T(X, i))^W$  is an isomorphism over R.

*Proof.* Let *B* be a Borel subgroup of *G* containing *T*. Let (V, U) be a good pair for the *G*-action and consider the Cartesian diagram

$$\begin{array}{c|c}
X_B & \xrightarrow{q_B} & U/B \\
\xrightarrow{p_X} & & \downarrow^p \\
X_G & \xrightarrow{q_G} & U/G.
\end{array}$$
(5-14)

By the definition of  $t_G$ , it follows that there is  $a \in CH^N(U/B, 0)$  such that  $p_*(a) = t_G \in CH^0(U/G, 0)$ . Using the projection formula, we get for any  $x \in CH^*(X_G, i)$ ,

$$p_{X*}(q_B^*(a)p_X^*(x)) = p_{X*}(q_B^*(a))x = q_G^*(p_*(a))x = t_G x$$

In particular,  $r_{T,X}^G$  is split injective over *R*.

To show the surjectivity, we see from the above that for any  $x \in CH^*(X_G, i)$ ,

$$p_X^* f p_X^*(x) = p_X^*(t_G x) = t_G p_X^*(x),$$

where  $f(y) = p_{X_*}(ay)$ . This in particular implies that  $p_X^* f(y) = t_G y$  for all y in the image of  $p_X^*$ . It follows from Corollary 8.7 that

$$p_X^* f(y) = t_G y$$
 for all  $y \in (CH^*(X_B, i)_{\mathbb{Q}})^W$ .

Since  $CH_T^*(X, i)$  is torsion-free, we must have

$$p_X^* f(y) = t_G y$$
 for all  $y \in (CH^*(X_B, i))^W$ .

Hence the map  $r_{T,X}^G$  is surjective onto the *W*-invariants over *R*.

**Remark 5.8.** The proof of the theorem above in fact shows that the map  $r_{T,X}^G$  is split injective over *R* for any  $X \in \mathcal{V}_G^S$ .

**Corollary 5.9.** Let X be a smooth projective scheme with a G-action such that the fixed point locus  $X^T$  for the T-action is isolated. Then the map

$$\operatorname{CH}^G_*(X,0) \to (\operatorname{CH}^T_*(X,0))^W$$

is an isomorphism over R.

*Proof.* This is an immediate consequence of [Krishna 2009a, Theorem 4.2] and Theorem 5.7.  $\Box$ 

#### 6. The Leray–Hirsch Theorem

In algebraic topology, the Leray–Hirsch theorem is a very important tool for describing the cohomology of the total space of a fiber bundle. Since the arguments in this theorem are mostly topological, one cannot always expect such results for the cohomology theories of algebraic varieties. A version of the Leray–Hirsch theorem was proven for the Chow groups of the total space of a Zariski-locally trivial fibration in [Ellingsrud and Strømme 1989, Lemma 2.8; Edidin and Graham 1997, Lemma 6]. In this section, we prove the general form of the Leray–Hirsch theorem for the higher Chow groups of schemes. We shall give several important applications of this theorem in the next few sections.

**6a.** *A Künneth formula.* In [Fulton 1984, Example 1.9.1], a *k*-scheme *L* is called *cellular* if it has a filtration  $\emptyset = L_{n+1} \subsetneq L_n \subsetneq \cdots \subsetneq L_1 \subsetneq L_0 = L$  by closed subschemes such that each  $L_i \setminus L_{i+1}$  is an affine space  $\mathbb{A}_k^{r_i, 1}$  It follows from the Bruhat decomposition that schemes of the type *G*/*B* are cellular, where *B* is a Borel subgroup of a split reductive group *G*.

**Lemma 6.1.** Let L be a cellular scheme with the cellular decomposition

 $\emptyset = L_{n+1} \subsetneq L_n \subsetneq \cdots \subsetneq L_1 \subsetneq L_0 = L$ 

and let  $U_i = L \setminus L_i$  for  $0 \le i \le n+1$ . Then for any  $0 \le i \le n$  and  $p \ge 0$ , the sequence

$$0 \rightarrow \operatorname{CH}_*(U_{i+1} \setminus U_i, p) \rightarrow \operatorname{CH}_*(U_{i+1}, p) \rightarrow \operatorname{CH}_*(U_i, p) \rightarrow 0$$

is exact.

*Proof.* The proof is very similar to the arguments of [Kahn 1999, Lemma 3.3] using an induction on the number of cells.  $\Box$ 

**Lemma 6.2.** *Let L be a cellular scheme and let X be a any k-scheme. Then the exterior product map* 

$$\operatorname{CH}_{*}(X) \otimes_{\mathbb{Z}} \operatorname{CH}_{*}(L, 0) \to \operatorname{CH}_{*}(X \times L)$$
 (6-1)

<sup>&</sup>lt;sup>1</sup>Some authors allow  $L_i \setminus L_{i+1}$  to be a disjoint union of affine spaces over k. But both definitions are equivalent.

is an isomorphism. In particular, the natural map  $CH^*(k) \otimes_{\mathbb{Z}} CH_*(L, 0) \rightarrow CH_*(L)$ is an isomorphism.

*Proof.* Consider the cellular decomposition of L as in Lemma 6.1. Then each  $U_i = L \setminus L_i$  is also a cellular scheme. It suffices to show by induction that (6-1) holds when L is any of these  $U_i$ . There is nothing to prove for i = 0 and the case i = 1 follows by the homotopy invariance since  $U_1$  is an affine space. In general, we have the short exact sequence

$$0 \to \operatorname{CH}_*(U_{i+1} \setminus U_i, 0) \to \operatorname{CH}_*(U_{i+1}, 0) \to \operatorname{CH}_*(U_i, 0) \to 0$$
(6-2)

by applying Lemma 6.1 with p = 0. Since each  $U_{i+1} \setminus U_i$  is an affine space, it also follows from Lemma 6.1 and by induction on the number of affine cells that each  $CH_*(U_i, 0)$  is a free abelian group of finite rank. Tensoring this with  $CH_*(X)$  over  $CH^*(k, 0) = \mathbb{Z}$ , we get a commutative diagram



where the left column is exact by the freeness of each  $CH_*(U_i, 0)$  and the right column is the localization exact sequence. The top horizontal arcolumn is an isomorphism by the homotopy invariance and the bottom horizontal arcolumn is an isomorphism by the induction. In particular,  $j^*$  is surjective in all indices. We conclude that  $i_*$  is injective in all indices and the middle horizontal arcolumn is also an isomorphism.

**6b.** *Leray–Hirsch with integral coefficients.* Let *F* be a cellular scheme over *k*. For any field extension  $k \hookrightarrow l$ , the scheme  $F_l$  is also cellular, for which the cellular decomposition and the affine cells are the base extensions of the cellular decomposition and affine cells of *F*. It follows from Lemma 6.1 that the map  $CH_*(F, 0) \to CH_*(F_l, 0)$  is an isomorphism. The following is the integral version of the Leray–Hirsch theorem for the Zariski-locally trivial fibrations.

**Theorem 6.3.** Let B be a smooth k-scheme and let  $F \xrightarrow{i} E \xrightarrow{p} B$  be a Zariskilocally trivial fibration such that the fiber F is a smooth cellular scheme. Assume that there are elements  $\{e_1, \ldots, e_r\}$  in CH<sup>\*</sup>(E, 0) such that

$$\{f_1 = i^*(e_1), \ldots, f_r = i^*(e_r)\}$$

forms a  $\mathbb{Z}$ -basis of CH<sup>\*</sup>( $F_v$ , 0) for each fiber  $F_v$  of the fibration. Then the map

$$\Phi: \mathrm{CH}^*(F, 0) \otimes_{\mathbb{Z}} \mathrm{CH}^*(B) \to \mathrm{CH}^*(E), \quad \sum_{1 \le i \le r} f_i \otimes b_i \mapsto \sum_{1 \le i \le r} p^*(b_i) e_i$$

is an isomorphism. In particular,  $CH^*(E)$  is a free  $CH^*(B)$ -module with basis  $\{e_1, \ldots, e_r\}$ .

*Proof.* Since k is perfect, we can find a filtration

$$\emptyset = B_{n+1} \subsetneq B_n \subsetneq \cdots \subsetneq B_1 \subsetneq B_0 = B$$

of *B* by closed subschemes such that for each  $0 \le i \le n$ , the scheme  $B_i \setminus B_{i+1}$ is smooth and the given fibration is trivial over it. We set  $U_i = B \setminus B_i$  and  $V_i = U_i \setminus U_{i-1} = B_{i-1} \setminus B_i$ . Observe then that each of the  $U_i$  and  $V_i$  is smooth. Set  $E_i = p^{-1}(U_i)$  and  $W_i = p^{-1}(V_i)$ . We prove by induction on *i* that the map  $CH^*(F, 0) \otimes_{\mathbb{Z}} CH^*(U_i) \to CH^*(E_i)$  is an isomorphism, which will prove the theorem. Since  $U_0 = \emptyset$  and  $E_1 = U_1 \times F$ , the desired isomorphism for i = 1follows from Lemma 6.2. We now consider the commutative diagram

$$CH^{*}(U_{i}) \otimes CH^{*}(F, 0) \longrightarrow CH^{*}(E_{i})$$

$$\downarrow \qquad \qquad \downarrow$$

$$CH^{*}(V_{i+1}) \otimes CH^{*}(F, 0) \longrightarrow CH^{*}(W_{i+1})$$

$$\downarrow \qquad \qquad \downarrow$$

$$CH^{*}(U_{i+1}) \otimes CH^{*}(F, 0) \longrightarrow CH^{*}(E_{i+1})$$

$$\downarrow \qquad \qquad \downarrow$$

$$CH^{*}(U_{i}) \otimes CH^{*}(F, 0) \longrightarrow CH^{*}(E_{i})$$

$$\downarrow \qquad \qquad \downarrow$$

$$CH^{*}(V_{i+1}) \otimes CH^{*}(F, 0) \longrightarrow CH^{*}(W_{i+1}).$$

$$(6-3)$$

The left column is obtained by tensoring the long exact localization sequence for higher Chow groups with  $CH^*(F, 0)$  over  $\mathbb{Z}$ , and the right column is just the localization exact sequence. Since  $CH^*(F, 0)$  is a free abelian group, the left column is also exact.

It is easily checked that the second and the third squares commute using the commutativity property of the push-forward and pull-back maps of higher Chow groups in a fiber diagram. We show that the other squares also commute. It is enough to show that the first square commutes as the fourth one is same as the first.

Let  $\delta$  denote the connecting homomorphism in a long exact localization sequence for the higher Chow groups.

Before we show the required commutativity, let us recall that if (X, Y) is pair of k-schemes where  $i : Y \hookrightarrow X$  is a closed subscheme with complement  $j : U \hookrightarrow X$ , then the localization exact sequence is the long exact homology sequence associated to the short exact sequence of cycle complexes

$$0 \to \mathscr{Z}_n(Y, \bullet) \xrightarrow{i_*} \mathscr{Z}_n(X, \bullet) \xrightarrow{j^*} \frac{\mathscr{Z}_n(X, \bullet)}{\mathscr{Z}_n(Y, \bullet)} \to 0,$$

where the natural map

$$\frac{\mathscr{L}_n(X, \bullet)}{\mathscr{L}_n(Y, \bullet)} \to \mathscr{L}_n(U, \bullet)$$

is a quasi-isomorphism. So we identify the last term with  $\mathscr{L}_n(U, \bullet)$ . The formalism of the homological algebra now shows that the connecting homomorphism  $\delta : \operatorname{CH}_n(U, i) \to \operatorname{CH}_n(Y, i - 1)$  is obtained as one obtains the connecting homomorphism in the snake lemma. In particular, this is same as the differential map  $\partial : \mathscr{L}_n(X, i) \to \mathscr{L}_n(X, i - 1)$ , evaluated on the homology groups. The Leibniz rule for this differential now implies that the connecting homomorphism  $\delta$  also satisfies the Leibniz rule; see [Panin 2003, §2.4].

If we now start with an element  $b \otimes i^*(e_j) \in CH^*(U_i) \otimes CH^*(F, 0)$  and map this vertically, we get  $\delta b \otimes i^*(e_j)$ , which maps horizontally down to  $p^*(\delta b) \cdot e_j$ . On the other hand, if we first map horizontally, we get  $p^*(b) \cdot e_j$  which maps vertically to  $\delta(p^*(b) \cdot e_j)$ . Using the Leibniz rule above, this last term is same as  $\delta p^*(b) \cdot e_j = p^*(\delta b) \cdot e_j$  since  $\delta e_j = 0$ . We have shown that the diagram above commutes.

The first and the fourth horizontal arrows in (6-3) are isomorphisms by induction. The second and the fifth horizontal arrows are isomorphisms by Lemma 6.2. Hence the middle horizontal arrow is also an isomorphism by the 5-lemma.

**6c.** *Leray–Hirsch with rational coefficients.* We need the following step to prove the rational version of the Leray–Hirsch theorem for the étale locally trivial fibrations of smooth schemes.

Let *F* be a smooth cellular scheme over *k*. We have seen before that for any field extension  $k \hookrightarrow l$ , the natural map  $CH^*(F, 0) \to CH^*(F_l, 0)$  is an isomorphism. Moreover, each of these is a free abelian group with the basis vectors given by the closures of the affine cells in the cellular decomposition. We fix this basis  $\{f_1, \ldots, f_r\}$  of  $CH^*(F_l, 0)$  in what follows. For a complete flag variety G/B, this set is same as the set of Schubert cycles  $\{\zeta_w\}_{w \in W}$ .

**Lemma 6.4.** Let B be a smooth k-scheme and let  $F \xrightarrow{i} E \xrightarrow{p} B$  be an étale locally trivial fibration such that the fiber F is a smooth cellular scheme. Assume that this fibration becomes trivial after a finite étale cover of B of degree d. Assume furthermore that there are elements  $\{e_1, \ldots, e_r\}$  in CH<sup>\*</sup>(E, 0) such that  $\{f_1 = i^*(e_1), \ldots, f_r = i^*(e_r)\}$  is the basis of CH<sup>\*</sup>(F<sub>y</sub>, 0) for each geometric fiber  $F_y$  of the fibration. Then the map

$$\Phi: \mathrm{CH}^*(F, 0) \otimes_{\mathbb{Z}} \mathrm{CH}^*(B) \to \mathrm{CH}^*(E), \quad \sum_{1 \le i \le r} f_i \otimes b_i \mapsto \sum_{1 \le i \le r} p^*(b_i) e_i$$

is an isomorphism over  $\mathbb{Z}[d^{-1}]$ . In particular,  $CH^*(E)[d^{-1}]$  is a free  $CH^*(B)[d^{-1}]$ -module with basis  $\{e_1, \ldots, e_r\}$ .

*Proof.* Let  $B' \xrightarrow{q} B$  be a finite étale cover such that  $E' = E \times_B B' \xrightarrow{p'} B'$  is a trivial fibration and let  $q' : E' \to E$  be the other projection. It follows from our assumption and the isomorphism of  $CH^*(F, 0)$  under the field extensions that the set  $\{e'_i = q'^*(e_i)\}$  restricts to the basis  $\{f_i\}$  of  $CH^*(F_y, 0)$  for every fiber  $F_y$  of the fibration p'.

Setting  $\Phi'(b' \otimes f_j) = p'^*(b')e'_j$  and using the fact that  $q'^* \circ p^* = p'^* \circ q^*$ ,  $p^* \circ q_* = q'_* \circ p'^*$  and  $q_* \circ q^* = d = q'_* \circ q'^*$  (see [Bloch 1986, Exercise 5.8(i)]), one checks that the diagram

commutes. The middle vertical arrow is an isomorphism by Lemma 6.2. A diagram chase shows that  $\Phi$  is an isomorphism over  $\mathbb{Z}[d^{-1}]$ .

**Theorem 6.5.** Let B be a smooth k-scheme and let  $F \xrightarrow{i} E \xrightarrow{p} B$  be an étale locally trivial fibration such that the fiber F is a smooth cellular scheme. Assume that there are elements  $\{e_1, \ldots, e_r\}$  in CH<sup>\*</sup>(E, 0) such that  $\{f_1 = i^*(e_1), \ldots, f_r = i^*(e_r)\}$  is the basis of CH<sup>\*</sup>(F<sub>y</sub>, 0) for each geometric fiber F<sub>y</sub> of the fibration. Then the map

$$\Phi: \mathrm{CH}^*(F, 0) \otimes_{\mathbb{Q}} \mathrm{CH}^*(B) \to \mathrm{CH}^*(E), \quad \sum_{1 \le i \le r} f_i \otimes b_i \mapsto \sum_{1 \le i \le r} p^*(b_i) e_i$$

is an isomorphism over the rationals. In particular,  $CH^*(E)$  is a free  $CH^*(B)$ -module with basis  $\{e_1, \ldots, e_r\}$  over the rationals.

*Proof.* We assume all abelian groups to be tensored with  $\mathbb{Q}$  in this proof. Since k is perfect and since every étale cover is generically finite, we can find a filtration

$$\emptyset = B_{n+1} \subsetneq B_n \subsetneq \cdots \subsetneq B_1 \subsetneq B_0 = B$$

of *B* by closed subschemes such that for each  $0 \le i \le n$ , the scheme  $V_i = B_{i-1} \setminus B_i$  is smooth and there is a finite étale cover  $V'_i \to V_i$  such that the given fibration is trivial over  $V'_i$ . We set  $U_i = B \setminus B_i$  as before, which implies that  $V_i = U_i \setminus U_{i-1}$ . Observe then that each of the  $U_i$  and  $V_i$  is smooth. Set  $E_i = p^{-1}(U_i)$  and  $W_i = p^{-1}(V_i)$ . We prove by induction on *i* that the map

$$\operatorname{CH}^*(F, 0) \otimes_{\mathbb{Q}} \operatorname{CH}^*(U_i) \to \operatorname{CH}^*(E_i)$$

is an isomorphism, which will imply the proposition. Since  $U_0 = \emptyset$  and since the map  $E_1 \rightarrow U_1$  is a smooth fibration which becomes trivial over the finite étale cover  $V'_1 \rightarrow V_1 = U_1$ , the desired isomorphism for i = 1 follows from Lemma 6.4. We now consider the diagram

The top row is obtained by tensoring the long exact localization sequence for higher Chow groups with  $CH^*(F, 0)$  over  $\mathbb{Q}$  and hence is exact. The bottom row is just the localization exact sequence.

One checks as in the proof of Theorem 6.3 that the diagram above is commutative. The first and the fourth vertical arrows are isomorphisms by induction. The second and the fifth vertical arrows are isomorphisms by Lemma 6.4. Hence the middle vertical arrow is also an isomorphism by 5-lemma.

### 7. Higher Chow groups of toric bundles and applications

As an application of Theorem 6.3, we describe the ordinary higher Chow groups of toric bundles with integral coefficients. Let *T* be a split torus of rank *n* over *k* and let  $M = \text{Hom}(\mathbb{G}_m, T)$  be the group of its one-parameter subgroups. Let  $X = X(\Delta)$  be a smooth projective toric variety associated to a fan  $\Delta$  in  $M_{\mathbb{R}}$  (see Section 11). Let *B* be a smooth *k*-scheme and let  $p : E \to B$  be a principal *T*-bundle. Setting  $E(X) = E \times^T X$ , we see that

$$\pi: E(X) \to B, \qquad \pi((e, x)) = p(e)$$

is a Zariski-locally trivial smooth fibration with all fibers isomorphic to X. Since X is projective, it follows that  $\pi$  is a projective morphism.

We fix an ordering  $\{\sigma_1, \ldots, \sigma_m\}$  of  $\Delta_{\max}$  and let  $\tau_i \subset \sigma_i$  be the cone that is the intersection of  $\sigma_i$  with all those  $\sigma_j$  such that  $j \ge i$  and that intersect  $\sigma_i$  in dimension n-1. Let  $\tau'_i \subset \sigma_i$  be the cone such that  $\tau_i \cap \tau'_i = \{0\}$  and  $\dim(\tau_i) + \dim(\tau'_i) = n$  for  $1 \le i \le m$ . It is easy to see that  $\tau'_i$  is the intersection of  $\sigma_i$  with all those  $\sigma_j$  such that  $j \le i$  and that intersect  $\sigma_i$  in dimension n-1. Since *X* is smooth and projective, it is well-known that we can choose the ordering above of  $\Delta_{\max}$  such that

 $\tau_i \subset \sigma_j$  implies  $i \leq j$  and  $\tau'_i \subset \sigma_j$  implies  $j \leq i$ . (7-1)

Let  $\Delta_1 = \{\rho_1, \dots, \rho_d\}$  be the set of one-dimensional cones in  $\Delta$  and let  $\{v_1, \dots, v_d\}$  be the associated primitive elements of M. We choose  $\{\rho_1, \dots, \rho_n\}$  to be a set of one-dimensional faces of  $\sigma_m$  such that  $\{v_1, \dots, v_n\}$  is a basis of M. Let  $\{\chi_1, \dots, \chi_n\}$  be the dual basis of  $M^{\vee}$ .

**Definition 7.1.** Let *A* be a commutative ring with unit and let  $\{r_1, \ldots, r_n\}$  be a subset of *A*. Let  $I_A$  denote the ideal of the polynomial algebra  $A[t_1, \ldots, t_d]$  generated by the elements

$$t_{j_1} \cdots t_{j_l} \quad \text{for } 1 \le j_p \le d \tag{7-2}$$

such that  $\rho_{j_1}, \ldots, \rho_{j_l}$  do not span a cone of  $\Delta$ . Let  $\overline{I}_{\Delta}$  denote the ideal of  $A[t_1, \ldots, t_d]$  generated by  $I_{\Delta}$  and the relations

$$s_i := \left(\sum_{j=1}^d \langle \chi_i, v_j \rangle t_j\right) - r_i \quad \text{for } 1 \le i \le n.$$
(7-3)

We define the A-algebras  $R_{eq}(A, \Delta)$  and  $R(A, \Delta)$  to be quotients of  $A[t_1, \ldots, t_d]$  by the ideals  $I_{\Delta}$  and  $\overline{I}_{\Delta}$ , respectively.

The ring  $R_{eq}(A, \Delta)$  is also known in the literature as the *Stanley–Reisner* algebra over A associated to the fan  $\Delta$ ; see [Sankaran and Uma 2003, Definition 2.1]. Notice that any character  $\chi$  of T acts on  $R_{eq}(A, \Delta)$  through the multiplication by the element  $\sum_{j=1}^{d} \langle \chi, v_j \rangle t_j$ . This makes  $R_{eq}(A, \Delta)$  into an S = S(T)-algebra.

Any *T*-equivariant line bundle  $L \rightarrow X$  uniquely defines a line bundle

$$E(L) = E \times^T L$$

on E(X). Every  $\rho \in \Delta_1$  defines a unique *T*-equivariant line bundle  $L_\rho$  on *X* with a *T*-equivariant section  $s_\rho : X \to L_\rho$  that is transverse to the zero section and whose zero locus is the orbit closure  $V_\rho = \overline{O_\rho}$ . For any  $\sigma \in \Delta$ , let  $u_\sigma$  denote the fundamental class of the *T*-invariant cycle  $[V_\sigma]$  in CH<sup>\*</sup>(*X*, 0) and let  $y_\sigma$  denote the cycle  $[E(V_\sigma)]$  in CH<sup>\*</sup>(E(X), 0). Notice that  $\pi_\sigma : E(V_\sigma) \to B$  is a smooth projective toric subbundle of  $\pi : E(X) \to B$  with fiber  $V_\sigma$ . Suppose that  $\rho_{j_1}, \ldots, \rho_{j_l}$  do not span a cone in  $\Delta$ . Then  $s = (s_{j_1}, \ldots, s_{j_l})$  is a nowhere vanishing section of  $L_{\rho_{j_1}} \oplus \cdots \oplus L_{\rho_{j_l}}$  and hence

$$c_1(E(L_{\rho_{j_1}})) \cdots c_1(E(L_{\rho_{j_l}})) = 0$$
 in  $CH^*(E(X))$ .

In particular, we get

$$y_{\rho_{j_1}} \cdots y_{\rho_{j_l}} = 0$$
 in  $CH^*(E(X))$ . (7-4)

We now consider the commutative diagram

$$X_{l} \xrightarrow{\iota} E(X) \xleftarrow{\overline{p}} E \times X \xrightarrow{p_{X}} X$$

$$\pi_{l} \downarrow \qquad \pi \downarrow \qquad \downarrow p_{E} \qquad \downarrow \pi_{X}$$

$$Spec(l) \longrightarrow B \xleftarrow{p} E \xrightarrow{\pi_{E}} Spec(k),$$

$$(7-5)$$

where Spec(l) is any point of *B*. It is clear that all squares are Cartesian and all the maps in the right square are *T*-equivariant. Let  $L_{\chi}$  denote the *T*-equivariant line bundle on Spec(k) associated to a character  $\chi$  of *T*. Since *p* and  $\overline{p}$  are principal *T*-bundles, we see that there is a unique line bundle  $\zeta$  on *B* such that

$$\pi_E^*(L_\chi) = p^*(\zeta) \quad \text{and} \quad p_X^* \circ \pi_X^*(L_\chi) = \overline{p}^* \circ \pi^*(\zeta). \tag{7-6}$$

Using the identity

$$\pi_X^*(c_1^T(L_\chi)) = \sum_{\rho \in \Delta_1} \langle \chi, v_\rho \rangle \, u_\rho \quad \text{in } \operatorname{CH}^*_T(X)$$

and the isomorphisms  $CH^*(B) \cong CH^*_T(E)$ ,  $CH^*(E(X)) \cong CH^*_T(E \times X)$ , we see that  $\pi^*(c_1(\zeta)) = \sum_{\rho \in \Delta_1} \langle \chi, v_\rho \rangle y_\rho$ . Let  $\zeta_i \in Pic(B)$  be such that  $\pi^*_E(L_{\chi_i}) = p^*(\zeta_i)$ , where  $\{\chi_1, \ldots, \chi_n\}$  is a chosen basis of  $M^{\vee}$  as above. Setting  $r_i = c_1(\zeta_i) \in CH^*(B)$ , we conclude that

$$\pi^*(r_i) = \sum_{j=1}^d \langle \chi_i, v_j \rangle y_{\rho_j} \quad \text{in } CH^*(E(X)), \quad 1 \le i \le n.$$
(7-7)

We define a homomorphism of  $CH^*(B)$ -algebras

$$\operatorname{CH}^*(B)[t_1,\ldots,t_d] \to \operatorname{CH}^*(E(X))$$

by the assignment  $t_i \mapsto y_{\rho_i}$  for  $1 \le i \le d$ . It follows from (7-4) and (7-7) that this homomorphism descends to a CH<sup>\*</sup>(*B*)-algebra homomorphism

$$\psi: R(\mathrm{CH}^*(B), \Delta) \to \mathrm{CH}^*(E(X)).$$
(7-8)

The following result describes the higher Chow groups of the projective toric bundle  $\pi : E(X) \to B$  and generalizes [Sankaran and Uma 2003, Theorem 1.2(iii)] to higher Chow groups.

## **Theorem 7.2.** The homomorphism $\psi$ is an isomorphism.

*Proof.* To prove this theorem, we first observe that for any  $\sigma \in \Delta$  and any point Spec $(l) \rightarrow B$ , one has  $\iota^*(E(V_{\sigma})) = (V_{\sigma})_l$ . Also, it is well-known [Sankaran and Uma 2003, Lemma 3.1] that  $\{\iota^*(y_{\tau_1}), \ldots, \iota^*(y_{\tau_m})\}$  forms a  $\mathbb{Z}$ -basis of CH<sup>\*</sup>( $X_l$ , 0).

Since  $y_{\tau_i} = y_{\rho_{i_1}} \cdots y_{\rho_{i_p}}$  for every  $1 \le i \le m$ , where  $\{\rho_{i_1}, \ldots, \rho_{i_p}\}$  is the set of edges of  $\tau_i$ , it follows from the proof of Theorem 6.3 that  $CH^*(E(X))$  is generated by  $\{y_{\rho_1}, \ldots, y_{\rho_d}\}$  as a  $CH^*(B)$ -algebra. In particular, the map  $\psi$  is surjective.

To prove injectivity, let  $x(\sigma)$  denote the monomial  $t_{i_1} \cdots t_{i_p}$  in  $R(CH^*(B), \Delta)$ such that  $\{\rho_{i_1}, \ldots, \rho_{i_p}\}$  is the set of edges of  $\sigma \in \Delta$ . Then by [Sankaran and Uma 2003, Lemma 2.1(ii)], the set  $\{x(\tau_1), \ldots, x(\tau_m)\}$  spans  $R(CH^*(B), \Delta)$  as a  $CH^*(B)$ -module. Since  $\psi(x(\tau_i)) = y_{\tau_i}$  for  $1 \le i \le m$  and since  $CH^*(E(X))$  is a free  $CH^*(B)$ -module with basis  $\{y_{\tau_1}, \ldots, y_{\tau_m}\}$  by Theorem 6.3, we conclude that  $\psi$  must be injective.

**7a.** Equivariant and ordinary higher Chow groups of smooth projective toric varieties. As a consequence of Theorem 7.2, we derive some explicit formulas for the equivariant and ordinary higher Chow groups of smooth projective toric varieties with integral coefficients. We shall later show in Section 11 that such formulas hold for all smooth toric varieties with rational coefficients. Recall that if  $X = X(\Delta)$  is a toric variety, then for every  $\sigma \in \Delta$ , the orbit closure  $V_{\sigma}$  is a *T*-invariant closed toric subvariety of *X* and hence uniquely defines a class  $y_{\sigma} = [V_{\sigma}]$  in  $CH_T^*(X, 0)$ ; see [Edidin and Graham 1998, Section 2.2]. This is called the fundamental equivariant class of  $V_{\sigma}$ .

We consider  $A = CH^*(k) \otimes_{\mathbb{Z}} S \cong CH^*(k)[t_1, \dots, t_n]$  as a graded  $CH^*(k)$ -algebra whose degree zero part is  $CH^*(k)$ . We have seen above that  $R_{eq}(CH^*(k), \Delta)$  has an action of *S* that makes it a graded *A*-algebra. Moreover,  $R(CH^*(k), \Delta)$  is just the quotient  $R_{eq}(CH^*(k), \Delta) \otimes_S \mathbb{Z} \cong R_{eq}(CH^*(k), \Delta) \otimes_A CH^*(k)$ .

**Corollary 7.3.** Let  $X = X(\Delta)$  be a smooth projective toric variety as above. Then the assignment  $t_i \mapsto y_{\rho_i}$  induces  $CH^*(k)$ -algebra isomorphisms

$$\Psi_X : R_{\rm eq}({\rm CH}^*(k), \Delta) \xrightarrow{\sim} {\rm CH}^*_T(X), \tag{7-9}$$

$$\overline{\Psi}_X : R(\mathrm{CH}^*(k), \Delta)) \xrightarrow{\sim} \mathrm{CH}^*(X).$$
(7-10)

*Proof.* The second isomorphism is just a special case of Theorem 7.2 when B = Spec(k).

To prove the isomorphism of (7-9), we first observe that  $\Psi_X$  is a graded A-linear homomorphism; see Section 11. Let M denote the kernel of this map. It follows

from [Krishna 2009a, Theorem 4.2] that  $CH_T^*(X)$  is a free *A*-module with basis  $\{y_{\tau_i}\}$ . It follows from this that  $\Psi_X$  is surjective and we get an exact sequence of graded *A*-modules

$$0 \to M \to R_{eq}(CH^*(k), \Delta) \xrightarrow{\Psi_X} CH^*_T(X) \to 0.$$

The freeness of  $\operatorname{CH}_T^*(X)$  as an *A*-module ensures that this sequence remains short exact after tensoring with  $\operatorname{CH}^*(k)$  via the augmentation  $A \twoheadrightarrow \operatorname{CH}^*(k)$ .

It follows from [Krishna 2009a, Theorem 1.1] that

$$\operatorname{CH}^*_T(X) \otimes_A \operatorname{CH}^*(k) \xrightarrow{\sim} \operatorname{CH}^*(X).$$

In particular,  $\Psi_X$  becomes an isomorphism after tensoring with  $CH^*(k)$ . We conclude that  $M \otimes_A CH^*(k) = 0$ . Since M is a nonnegatively graded A-module, it must be zero.

## 8. Higher Chow groups of flag bundles and applications

We remind the readers of our convention that all the higher Chow groups for the rest of this text will be considered with rational coefficients. We shall however, indicate the coefficients in the statement of all results.

In this section, we describe a formula for the higher Chow groups of complete flag bundles with rational coefficients. Such a formula for general flag bundles is an immediate consequence of the case of complete flag bundles. We also give some applications of this formula to the theory of equivariant higher Chow groups.

**8a.** Complete flag bundles. Let G be a connected reductive group over k and let B be a Borel subgroup of G containing a split maximal torus T. Let X be a k-scheme and let  $p : E \to X$  be a principal G-bundle and let  $\pi : E/B \to X$  be the associated complete flag bundle. Vistoli [1989] described the classical Chow groups  $CH_*(E, 0)$  and  $CH_*(E/B, 0)$  in terms of the Chow groups of X. In this section we generalize Vistoli's results to the case of higher Chow groups. The proof below is completely different from Vistoli's; it is much shorter and relies more on the equivariant techniques.

The restriction map  $r_{T,X}^G$  induces for every  $i \ge 0$ , a natural map of S(T)-modules

$$\operatorname{CH}^{G}_{*}(E,i) \otimes_{S(G)} S(T) \to \operatorname{CH}^{T}_{*}(E,i), \quad w \otimes \alpha \mapsto \alpha \cdot r^{G}_{T,E}(w).$$

Since *G* acts freely on *E*, one identifies  $CH^G_*(E, i)$  with  $CH_*(X, i)$  by Proposition 2.2. The group  $CH^B_*(E, i)$  is canonically identified with  $CH_*(E/B, i)$  for the same reason. The map above then translates into a natural map of *S*(*T*)-modules

$$\lambda_X : \operatorname{CH}_*(X, i) \otimes_{S(G)} S(T) \to \operatorname{CH}_*(E/B, i).$$
(8-1)

Taking the direct sum over  $\{CH_*(X, i)\}_{i \ge 0}$ , we get a natural map of S(T)-modules

$$\lambda_X : \operatorname{CH}_*(X) \otimes_{S(G)} S(T) \to \operatorname{CH}_*(E/B).$$
(8-2)

This map is a ring homomorphism if X is smooth. One can easily check that  $\lambda_X$  commutes with the flat pull-back and the proper push-forward maps between the higher Chow groups of the base schemes of the bundle. We wish to show that  $\lambda_X$  is an isomorphism. We begin with the following special case.

**Lemma 8.1.** Let X be a smooth (not necessarily connected) scheme over k and let  $f : X' \to X$  be a finite étale morphism such that the principal bundle  $p : E \to X$  is trivialized over X'. Then the map  $\lambda_X$  is an isomorphism with rational coefficients. In particular,  $\lambda_X$  is an isomorphism with rational coefficients if X is zero-dimensional.

*Proof.* Since X is a disjoint union of connected smooth schemes, it is enough to prove the lemma when X is smooth and connected. If  $E/B = G/B \rightarrow \text{Spec}(k)$  is the flag variety, then we have

$$CH_*(k) \otimes_{S(G)} S(T) \cong CH_*(k) \otimes_{CH^*(k,0)} (CH^*(k,0) \otimes_{S(G)} S(T))$$
$$\cong^{\dagger} CH_*(k) \otimes_{CH^*(k,0)} CH_*(G/B,0)$$
$$\cong CH_*(G/B) \qquad (by Lemma 6.2),$$

where  $\cong^{\dagger}$  follows from [Demazure 1973, théorème 2].

If  $G/B \times X \xrightarrow{\pi} X$  is the trivial bundle, then we get

$$CH_*(X) \otimes_{S(G)} S(T) \cong CH_*(X) \otimes_{CH^*(k,0)} (CH^*(k,0) \otimes_{S(G)} S(T))$$
$$\cong CH_*(X) \otimes_{CH^*(k,0)} CH_*(G/B,0)$$
$$\cong CH_*(G/B \times X)$$
(by Lemma 6.2).

In general, we consider the diagram

where the tensor product in the top row is over S(G). We have just shown that  $\lambda_{X'}$  is an isomorphism. It follows from the projection formula (see [Bloch 1986, Exercise 5.8]) that the composite horizontal maps in both rows are multiplication by [k(X'):k(X)]. Hence,  $\lambda_X$  must be an isomorphism too.

**Theorem 8.2.** For any k-scheme X, the map  $\lambda_X$  is an isomorphism of S(T)-modules with rational coefficients. This is a ring isomorphism if X is smooth.

*Proof.* We only need to prove the first assertion, for which we use induction on the dimension of X. The zero-dimensional case follows from Lemma 8.1. In general, we can find an étale cover of X over which the bundle  $p : E \to X$  becomes trivial. Since any such cover is generically finite and since the base field k is perfect, we can find a dense open subset  $j : U \hookrightarrow X$  and a finite étale cover  $f : U' \to U$  such that U is a disjoint union of connected smooth schemes and the given bundle is trivial over U'. Let  $\iota : Z \hookrightarrow X$  be the complement of U with its reduced induced closed subscheme structure.

We now consider the diagram

of localization exact sequences, where the tensor product in the top row is over the ring S(G). In particular, this row is exact by the flatness of S(T) over S(G). The second and the third squares commute by the compatibility of  $\lambda_X$  with the push-forward and the pull-back maps as remarked above.

To see that the first square commutes, let us consider an element

$$a \otimes b \in CH_*(U) \otimes S(T).$$

If we map this horizontally, we get  $\partial(a) \otimes b$ , which is mapped vertically down to  $b \cdot \pi_Z^* \circ \partial(a)$ . Since the localization sequence of higher Chow groups is compatible with respect to the flat pull-back, this last term is same as  $b \cdot \partial \circ \pi_U^*(a)$ . On the other hand, mapping  $a \otimes b$  vertically down gives  $b \cdot \pi_U^*(a)$  and if we map this horizontally, we get  $\partial(b \cdot \pi_U^*(a))$ . Since the horizontal maps in the bottom row are S(T)-linear (see Proposition 2.2), we conclude that the first (hence the fourth) square commutes.

The first and the fourth vertical arrows in the diagram above are isomorphisms by Lemma 8.1. Since U is a dense open in X, the dimension of Z is strictly smaller than that of X. Hence the second and the fifth vertical arrows are isomorphisms by induction on the dimension and Lemma 8.1. We conclude from the 5-lemma that  $\lambda_X$  is an isomorphism.

**8b.** *Principal bundles and flag bundles.* The following extension of Theorem 8.2 to all flag bundles is a direct generalization of the projective bundle formula for higher Chow groups.

**Corollary 8.3.** Let  $p: E \to X$  be a principal *G*-bundle over a *k*-scheme *X* and let  $\pi: E/P \to X$  denote the flag bundle associated to a parabolic subgroup *P*. Then

the natural map of S(P)-modules

$$\lambda_X : \mathrm{CH}_*(X; \mathbb{Q}) \underset{S(G; \mathbb{Q})}{\otimes} S(P; \mathbb{Q}) \to \mathrm{CH}_*(E/P; \mathbb{Q}), \quad w \otimes \alpha \mapsto \alpha \cdot r_{P, E}^G(w).$$

is an isomorphism. This is a ring isomorphism if X is smooth.

*Proof.* We only need to prove the first assertion. Let  $B \subseteq P$  be a Borel subgroup of G containing a split maximal torus T. Let  $\Phi(G, T)$  be the root system of G with respect to T such that B corresponds to the base  $\Delta$  of  $\Phi(G, T)$  and P corresponds to a subset  $I \subset \Delta$ . Let  $P = M \ltimes N$  be the Levi decomposition and let  $B_M = B \cap M$  be the Borel subgroup of M containing T. Let  $W_P \subset W$  be the Weyl group of P with respect to T. It follows from Propositions 3.3, 3.4 and Corollary 8.7 that the natural map

$$\operatorname{CH}_{*}(E/P) \to (\operatorname{CH}_{*}(E/B))^{W_{P}}$$
(8-3)

is an isomorphism.

On the other hand, following the proof of Corollary 8.7, we get

$$(CH_*(X) \otimes_{S(G)} S(T))^{W_P} \cong CH_*(X) \otimes_{S(G)} (S(T))^{W_P}$$
$$\cong CH_*(X) \otimes_{S(G)} S(M)$$
$$\cong CH_*(X) \otimes_{S(G)} S(P),$$
(8-4)

where the last isomorphism follows again from Proposition 3.4. The corollary now follows from Theorem 8.2 by combining (8-3) and (8-4).  $\Box$ 

The following result generalizes [Vistoli 1989, Corollary 3.2] to higher Chow groups.

**Corollary 8.4.** Let G be connected and split reductive group over k and let  $p: E \rightarrow X$  be a principal G-bundle over a k-scheme X. Then there is a strongly convergent spectral sequence

$$E_2^{p,q} = \operatorname{Tor}_p^{S(G;\mathbb{Q})}(\mathbb{Q}, \operatorname{CH}_*(X, q; \mathbb{Q})) \Rightarrow \operatorname{CH}_*(E, p+q; \mathbb{Q}).$$

The edge homomorphism yields an isomorphism

$$\operatorname{CH}_*(X, 0; \mathbb{Q}) \underset{S(G; \mathbb{Q})}{\otimes} \mathbb{Q} \xrightarrow{\sim} \operatorname{CH}_*(E, 0; \mathbb{Q}).$$

*Proof.* This is an immediate consequence of Theorem 8.2, [Krishna 2009a, Theorem 1.1] and flatness of S(T) over S(G).

**8c.** *A change of groups isomorphism.* The following theorem is an analogue of a similar result in equivariant *K*-theory by Merkurjev [2005, Proposition 8]. However, this result for the equivariant higher Chow groups has an advantage over Merkurjev's theorem in that it holds for the action of any split reductive group (though with

rational coefficients) whereas [ibid., Proposition 8] is known only for the groups whose derived subgroups are simply connected, for example,  $GL_n$ . The special case  $CH^G_*(X, 0)$  of the result below was proven by Brion [1997, Theorem 6.7].

**Theorem 8.5.** Let G be a connected reductive group and let T be a split maximal torus of G. Then for any  $X \in \mathcal{V}_G$ , the natural map of S(T)-modules

$$\lambda_X : \mathrm{CH}^G_*(X; \mathbb{Q}) \underset{S(G; \mathbb{Q})}{\otimes} S(T; \mathbb{Q}) \to \mathrm{CH}^T_*(X; \mathbb{Q})$$
(8-5)

is an isomorphism. This is a ring isomorphism if X is smooth.

*Proof.* We only need to show the first assertion. If (V, U) is a good pair for the *G*-action, then  $CH^G_*(X, i)$  and  $CH^T_*(X, i)$  in suitable degrees are the same as  $CH_*(X_G, i)$  and  $CH_*(X_B, i)$ , respectively, where *B* is a Borel subgroup of *G* containing *T*. Hence, it suffices to show that for any *k*-scheme *Z* with a free action of *G* with quotients G/B and Z/G, the natural map

$$\lambda_{Z/G}$$
: CH<sub>\*</sub>(Z/G)  $\otimes_{S(G)} S(T) \rightarrow$  CH<sub>\*</sub>(Z/B)

is an isomorphism. But this follows immediately by applying Theorem 8.2 to the principal bundle  $Z \rightarrow Z/G$ .

**Remark 8.6.** The first remark is that Theorem 8.5 holds if G is any connected linear algebraic group (not necessarily reductive) if the base field is of characteristic zero. This is an immediate consequence of Proposition 3.4. The second remark is that the theorem above can also be proven as a simple consequence of the Leray–Hirsch theorem 6.5. The case of smooth schemes is a direct consequence of Theorem 6.5 and the general case can be proven using noetherian induction and the localization sequence. We leave it an exercise to fill in the details.

**8d.** Some consequences of Theorem 8.5. Recall that if *G* is a connected reductive group with a split maximal torus *T*, then the normalizer *N* of *T* in *G* and all its connected components are defined over *k* and the quotient N/T is the Weyl group *W* of the corresponding root system. In particular,  $W \subset G/T$ . If *G* acts on a variety *X* and if (V, U) is a good pair for the *G*-action, then  $X \times^T U \to X \times^G U$  is an étale-locally trivial smooth fibration with fiber G/T. In particular, *W* acts on each  $\operatorname{CH}_j^T(X, i)$  and the map  $\operatorname{CH}_j^G(X, i) \to \operatorname{CH}_j^T(X, i)$  factors through the *W*-invariants. We get the following consequence of Theorem 8.5.

**Corollary 8.7** (see Theorem 5.7). Let G be a connected reductive group and let T be a split maximal torus of G with the Weyl group W. Then for any  $X \in \mathcal{V}_G$ , the restriction map  $r_T^G_X$  induces an isomorphism

$$\operatorname{CH}^G_*(X; \mathbb{Q}) \xrightarrow{\sim} (\operatorname{CH}^T_*(X; \mathbb{Q}))^W.$$

*Proof.* Since W is a finite group, the trivial  $\mathbb{Q}[W]$ -module  $\mathbb{Q}$  is a projective  $\mathbb{Q}[W]$ -module of finite rank. In particular, it follows from Theorem 8.5 that

$$(\operatorname{CH}_{T}^{*}(X))^{W} = \operatorname{Hom}_{\mathbb{Q}[W]}(\mathbb{Q}, \operatorname{CH}_{*}^{G}(X) \otimes_{S(G)} S(T))$$
$$= \operatorname{CH}_{*}^{G}(X) \otimes_{S(G)} \operatorname{Hom}_{\mathbb{Q}[W]}(\mathbb{Q}, S(T))$$
$$= \operatorname{CH}_{*}^{G}(X) \otimes_{S(G)} (S(T))^{W}$$
$$=^{\dagger} \operatorname{CH}_{*}^{G}(X) \otimes_{S(G)} S(G)$$
$$= \operatorname{CH}_{*}^{G}(X),$$

where  $=^{\dagger}$  holds by [Edidin and Graham 1998, Proposition 6].

As an important consequence of the result above, we get the following analogue of a similar result of Thomason [1988, Theorem 1.13] in equivariant *K*-theory.

**Corollary 8.8** (see Remark 5.8). Let G be a connected and reductive group over k and let T be a split maximal torus in G. Then the restriction map

$$\operatorname{CH}^{G}_{*}(X; \mathbb{Q}) \xrightarrow{r_{T,X}^{G}} \operatorname{CH}^{T}_{*}(X; \mathbb{Q})$$
 (8-6)

is a split monomorphism. Moreover, this splitting is natural for morphisms in  $\mathcal{V}_G$ . In particular, if H is any closed subgroup of G, then there is a split injective map

$$\operatorname{CH}^{H}_{*}(X; \mathbb{Q}) \xrightarrow{r_{T,X}^{G}} \operatorname{CH}^{T}_{*}(G \times^{H} X; \mathbb{Q}).$$
(8-7)

*Proof.* The first statement follows directly from Corollary 8.7, where the splitting is given by the trace map into the subgroup of *W*-invariants. The last statement follows from (8-6) and Corollary 3.2.  $\Box$ 

**Remark 8.9.** Let  $\rho \in S(T) \cong CH^*_G(G/B, 0)$  be such that the forgetful map takes  $\rho$  to the class of the zero-dimensional Schubert cycle in  $CH^*(G/B, 0)$ . For a flag bundle  $\pi : E/B \to X$  over a scheme X, if we define  $\psi_X : CH_*(X) \to CH_*(X)$  by  $\psi_X(\alpha) = \pi_*(\rho \cdot \pi^*(\alpha))$ , then it can be shown that  $\psi_X$  is an isomorphism. In particular,  $\pi^*$  is split injective. This gives another (and more conceptual) proof of Corollary 8.8. In fact, this proof shows that  $r^G_{T,X}$  is split injective with integer coefficients if G is *special*.

Let *G* be a connected reductive group and let *B* be a Borel subgroup of *G* containing a split maximal torus *T*. It follows from [Demazure 1973, théorème 2] that the forgetful map  $S(T) \rightarrow CH^*(G/B, 0)$  is surjective. Moreover, if  $\{\varrho_w\}_{w \in W}$  are polynomials in S(T) which map to the classes of Schubert cycles  $\{\zeta_w\}_{w \in W}$  in  $CH^*(G/B, 0)$ , then S(T) is a free S(G)-module with basis  $\{\varrho_w\}_{w \in W}$ . The following is a direct generalization of Demazure's theorem to the case of all smooth schemes and all higher Chow groups. This also strengthens Corollary 8.7 for smooth schemes.

**Corollary 8.10.** Let  $X \in \mathcal{V}_G^S$  and let  $X \xrightarrow{p_X}$  Spec(k) be the structure map. Set  $\varrho_{w,X} = p_X^*(\varrho_w)$ . Then  $\operatorname{CH}_T^*(X; \mathbb{Q})$  is a free  $\operatorname{CH}_G^*(X; \mathbb{Q})$ -module with basis  $\{\varrho_{w,X}\}_{w \in W}$ .

*Proof.* It follows from the construction that the map

$$\lambda_X : \operatorname{CH}^*_G(X) \otimes_{S(G)} S(T) \to \operatorname{CH}^*_T(X)$$

takes  $1 \otimes \varrho_w$  onto  $\varrho_{w,X}$ . The corollary is now an immediate consequence of Theorem 8.5.

## 9. Cohomological rigidity and specializations

Let *G* be a split diagonalizable group over *k* acting on a smooth scheme *X*. Recall [Springer 1998, 13.2.5] that all the diagonalizable subgroups of *G* are defined and split over *k*. The equivariant *K*-theory of *X* for the *G*-action was studied by Vezzosi and Vistoli [2003]. Their main result (Theorem 1) is to reconstruct the equivariant *K*-theory ring of *X* in terms of the equivariant *K*-theory of the loci where the stabilizers have constant dimension. In the next two sections, we use the ideas of Vezzosi and Vistoli to prove an analogous decomposition theorem (see Theorem 10.3) for the equivariant higher Chow groups of *X* for the *G*-action. As mentioned in the introduction, this theorem and its compatibility with the corresponding result for the equivariant *K*-theory play fundamental roles in the proof of the equivariant Riemann–Roch theorems in [Krishna 2009b]. This theorem is very useful in computing the equivariant and ordinary higher Chow groups of smooth schemes with torus action. Some applications of this kind are given in Section 11.

This section is concerned with the study of the notion of cohomological rigidity and the construction of certain specialization maps in equivariant higher Chow groups. In this and the next section, the group G will denote a split diagonalizable group and the all schemes will be assumed to be smooth with G-action. We have seen (Proposition 2.2) that for such a scheme X,  $CH_G^*(X)$  is a bigraded ring, which is an algebra over the ring  $CH_G^*(k)$ .

## 9a. Cohomological rigidity.

**Definition 9.1.** Let  $Y \subset X$  be a smooth and *G*-invariant closed subscheme of codimension  $d \ge 0$  and let  $N_{Y/X}$  denote the normal bundle of *Y* in *X*. We say that *Y* is *cohomologically rigid* inside *X* if  $c_d^G(N_{Y/X})$  is a not a zero-divisor in the ring  $CH_G^*(Y)$ .

As one observes, this definition has reasonable meaning only in the equivariant setting, since every element of positive degree in the nonequivariant Chow ring is nilpotent. The importance of cohomological rigidity for the equivariant higher Chow groups comes from the following analogue of the *K*-theory splitting theorem (Proposition 4.3) of [Vezzosi and Vistoli 2003].

**Proposition 9.2.** Let Y be a smooth and G-invariant closed subscheme of X of codimension  $d \ge 0$ . Assume that Y is cohomologically rigid inside X, and put  $U = X \setminus Y$ . Let  $i: Y \hookrightarrow X$  and  $j: U \hookrightarrow X$  be the inclusion maps.

(i) The localization sequence

$$0 \to \operatorname{CH}^*_G(Y; \mathbb{Q}) \xrightarrow{i_*} \operatorname{CH}^*_G(X; \mathbb{Q}) \xrightarrow{j^*} \operatorname{CH}^*_G(U; \mathbb{Q}) \to 0$$

is exact.

(ii) The restriction ring homomorphisms

$$\operatorname{CH}^*_G(X; \mathbb{Q}) \xrightarrow{(i^*, j^*)} \operatorname{CH}^*_G(Y; \mathbb{Q}) \times \operatorname{CH}^*_G(U; \mathbb{Q})$$

give an isomorphism of rings

$$\operatorname{CH}^*_G(X; \mathbb{Q}) \xrightarrow{\sim} \operatorname{CH}^*_G(Y; \mathbb{Q}) \underset{\operatorname{CH}^*_G(Y; \mathbb{Q})}{\times} \operatorname{CH}^*_G(U; \mathbb{Q}),$$

where  $\widetilde{CH}_{G}^{*}(Y; \mathbb{Q}) = CH_{G}^{*}(Y; \mathbb{Q})/(c_{d}^{G}(N_{Y/X}))$ , and the maps

$$\operatorname{CH}^*_G(Y; \mathbb{Q}) \to \operatorname{CH}^*_G(Y; \mathbb{Q}), \qquad \operatorname{CH}^*_G(U; \mathbb{Q}) \to \operatorname{CH}^*_G(Y; \mathbb{Q})$$

are, respectively, the natural surjection and the map

$$\operatorname{CH}^*_G(U; \mathbb{Q}) = \frac{\operatorname{CH}^*_G(X; \mathbb{Q})}{i_*(\operatorname{CH}^*_G(Y; \mathbb{Q}))} \xrightarrow{i^*} \frac{\operatorname{CH}^*_G(Y; \mathbb{Q})}{c_d^G(N_{Y/X})} = \widetilde{\operatorname{CH}^*_G(Y; \mathbb{Q})},$$

which is well-defined by Corollary 4.5.

*Proof.* Part (i) follows directly from Corollary 4.5 and the definition of cohomological rigidity. Since  $i^*$  and  $j^*$  are ring homomorphisms, the proof of the second part follows directly from the first part and [Vezzosi and Vistoli 2003, Lemma 4.4].  $\Box$ 

To apply the result above in our context, we need to find some sufficient conditions for checking the cohomological rigidity in specific examples. We begin with the following elementary result.

**Lemma 9.3.** Let A be a ring which is a Q-algebra. Then an element of the form  $t^d$ , where  $t = \sum_{j=1}^{r} a_j t_{i_j} \in A[t_1, \ldots, t_n]$ , is not a zero-divisor for any  $d \ge 0$  whenever  $a_j \in \mathbb{Q}$  for all j and  $a_j \ne 0$  for some j.

*Proof.* Since all  $a_j \in \mathbb{Q}$  and some  $a_j \neq 0$ , we see that  $t^d$  is a nonzero element of  $\mathbb{Q}[t_1, \ldots, t_n]$  and hence a nonzero divisor in this ring. Since tensoring with A over  $\mathbb{Q}$  is exact, we see that the multiplication by  $t^d$  is injective in  $A[t_1, \ldots, t_n]$  too.  $\Box$ 

**Proposition 9.4.** Let G be a split diagonalizable group acting on a smooth scheme X and let E be a G-equivariant vector bundle of rank d on X. Assume that there is a subtorus  $T \subset G$  of positive rank which acts trivially on X, such that in the eigenspace decomposition of E with respect to T, the submodule corresponding to the trivial character is zero. Then  $c_d^G(E)$  is not a zero-divisor in  $CH_G^*(X; \mathbb{Q})$ .

Proof. By [Thomason 1986, Lemma 5.6], E has a unique direct sum decomposition

$$E=\bigoplus_{i=1}^r E_{\chi_i}\otimes L_{\chi_i},$$

where we choose a splitting  $G = D \times T$ ,  $E_{\chi_i}$  are *D*-bundles and  $\chi_i$  are characters of *T* with associated line bundles  $L_{\chi_i} \in \text{Pic}_T(k)$ . This decomposition is via the functor

$$\operatorname{Bun}_X^D \times \operatorname{Rep}(T) \to \operatorname{Bun}_X^G, \quad (F, \rho) \mapsto p_1^*(F) \otimes p_2^*(\rho),$$

where  $p_1: D \times T \to D$  and  $p_2: D \times T \to T$  are the projections.

Since rank(*E*) = *d*, the Whitney sum formula yields  $c_d^G(E) = \prod_{i=1}^r c_{d_i}^G(E_{\chi_i} \otimes L_{\chi_i})$ , where  $d_i = \operatorname{rank}(E_{\chi_i})$ . We can thus assume that  $E = E_{\chi} \otimes L_{\chi}$ , where  $\chi$  is not a trivial character by our assumption. In particular, we can write

$$c_1^T(L_{\chi}) = t = \sum_{i=1}^p n_i t_i \in \mathbb{Q}[t_1, \dots, t_n]$$
 (9-1)

with  $n_i \neq 0$  for some *i*. By neglecting those *i* for which the coefficients  $n_i$  are zero, we can assume that  $n_i \neq 0$  for all *i*. Now we have

$$c_d^G(E) = c_d^G(p_1^*(E_{\chi}) \otimes p_2^*(L_{\chi})) =^{\dagger} \sum_{i=0}^d c_{d-i}^G(p_1^*(E_{\chi})) \cdot (c_1^G(p_2^*(L_{\chi})))^i$$
$$= \sum_{i=0}^d p_1^*(c_{d-i}^D(E_{\chi})) \cdot p_2^*((c_1^T(L_{\chi}))^i) = \sum_{i=0}^d \alpha_i t^i,$$

where  $\alpha_i \in CH_D^*(X)$  and  $c_d^G(E) \in CH_G^*(X) \cong CH_D^*(X) \otimes S(T)$  by Theorem 3.5 and  $=^{\dagger}$  holds by [Fulton 1984, Remark 3.2.3]. Furthermore,  $\alpha_d = p_1^*(c_0^D(E_{\chi})) = 1$ . Thus we get  $c_d^G(E) = t^d + \alpha_{d-1}t^{d-1} + \cdots + \alpha_1t + \alpha_0 = g(t)$ .

We need to show that g(t) is not a zero divisor in  $\operatorname{CH}^*_D(X)[t_1, \ldots, t_n]$ . So suppose  $f(\underline{t})$  is a nonzero polynomial such that  $g(t)f(\underline{t}) = 0$ , and let  $f'(\underline{t})$  be the homogeneous part of  $f(\underline{t})$  of largest degree which is not zero. By comparing the homogeneous parts, it is easy to see that  $g(t)f(\underline{t}) = 0$  only if  $t^d f'(\underline{t}) = 0$ . But this is a contradiction since t satisfies the condition of Lemma 9.3 by (9-1), and hence is not a zero-divisor. Let *G* be a split diagonalizable group as above and let  $X \in \mathcal{V}_G^S$ . Following the notation of [Vezzosi and Vistoli 2003], for any  $s \ge 0$ , we let  $X_{\le s} \subset X$  be the open subset of points whose stabilizers have dimension at most *s*. We shall often write  $X_{\le s-1}$  also as  $X_{\le s}$ . Let  $X_s = X_{\le s} \setminus X_{\le s}$  denote the locally closed subset of *X*, where the stabilizers have dimension exactly *s*. We think of  $X_s$  as a subspace of *X* with the reduced induced structure. It is clear that  $X_{\le s}$  and  $X_s$  are *G*-invariant subspaces of *X*. Let  $N_s$  denote the normal bundle of  $X_s$  in  $X_{\le s}$ , and let  $N_s^0$  denote the complement of the 0-section in  $N_s$ . Then *G* clearly acts on  $N_s$ . The following result describes some very useful properties of these subspaces.

# **Proposition 9.5.** *Let* $s \ge 0$ *be an integer.*

- (i) There exists a finite number of s-dimensional subtori T<sub>1</sub>,..., T<sub>r</sub> in G such that X<sub>s</sub> is the disjoint union of the fixed point spaces X<sup>T<sub>j</sub></sup><sub><s</sub>.
- (ii)  $X_s$  is smooth locally closed subscheme of X.
- (iii)  $N_s^0 = (N_s)_{<s}$ .

*Proof.* Since the base field k is perfect, this is a special case of [Vezzosi and Vistoli 2003, Proposition 2.2], which holds for regular *G*-schemes over any connected and separated Noetherian base scheme.

**Remark 9.6.** We mention here that although the proposition above has been stated for the smooth schemes, part (i) of the proposition holds also when X is not necessarily smooth, since the proof given in [loc. cit.] only uses Thomason's generic étale slice theorem, which holds very generally.

**Corollary 9.7.** For  $s \ge 1$ ,  $X_s$  is cohomologically rigid inside  $X_{\le s}$ .

*Proof.* Let  $d_s$  be the codimension of  $X_s$  in  $X_{\leq s}$ . We need to show that  $c_{d_s}^G(N_s)$  is not a zero-divisor in  $CH_G^*(X_s)$ . By Proposition 9.4, it suffices to show that there exists a subtorus T in G of positive rank that acts trivially on  $X_s$ , such that in the eigenspace decomposition of  $N_s$  with respect to T, the submodule corresponding to the trivial character is zero. But this follows directly from parts (i) and (iii) of Proposition 9.5 and the fact that  $s \geq 1$ ; see [Vezzosi and Vistoli 2003, Proposition 4.6].

**9b.** Specialization maps. Let G and X be as above and let n be the dimension of G. As seen above, there is a filtration of X by G-invariant open subsets

$$\emptyset = X_{\leq -1} \subset X_{\leq 0} \subset \cdots \subset X_{\leq n} = X.$$

In particular, *G* acts on  $X_{\leq 0}$  with finite stabilizers, and the toral component of *G* acts trivially on  $X_n$ . We fix  $1 \leq s \leq n$  and let  $f_s : X_s \hookrightarrow X_{\leq s}$  and  $g_s : X_{< s} \hookrightarrow X_{\leq s}$  denote the closed and the open embeddings, respectively. Let  $\pi : M_s \to \mathbb{P}^1$  be the deformation to the normal cone for the embedding  $f_s$  as in Section 4. We have already observed there that for the trivial action of *G* on  $\mathbb{P}^1$ ,  $M_s$  has a

natural *G*-action. Moreover, the deformation diagram (4-1) is a diagram of smooth *G*-spaces. For  $0 \le t \le s$ , we shall often denote the open subspace  $(M_s)_{\le t}$  of  $M_s$  by  $M_{s,\le t}$ . The terms like  $M_{s,t}$  and  $M_{s,< t}$  (and also for  $N_s$ ) will have similar meaning in what follows. Since *G* acts trivially on  $\mathbb{P}^1$ , it acts on  $M_s$  fiberwise with  $N_s = \pi^{-1}(\infty)$  and

$$M_{s,\leq t} \cap \pi^{-1}(\mathbb{A}^1) = X_{\leq t} \times \mathbb{A}^1, \qquad M_{s,t} \cap \pi^{-1}(\mathbb{A}^1) = X_t \times \mathbb{A}^1.$$
(9-2)

Let  $i_{s,\leq t}: N_{s,\leq t} \hookrightarrow M_{s,\leq t}$  and  $j_{s,\leq t}: X_{\leq t} \times \mathbb{A}^1 \hookrightarrow M_{s,\leq t}$  denote the obvious closed and open embeddings. We define  $i_{s,t}$  and  $j_{s,t}$  similarly. Let  $\eta_{s,t}: N_{s,t} \hookrightarrow N_{s,\leq t}$  and  $\delta_{s,t}: M_{s,t} \hookrightarrow M_{s,\leq t}$  denote the other closed embeddings. One has a commutative diagram

where  $g_{\leq t}$  is the 0-section embedding, and the composite of all the maps in the bottom row is identity. This gives us the diagram

$$\begin{array}{c} \operatorname{CH}^*_G(N_{s,t}) \xrightarrow{i_{s,t_*}} \operatorname{CH}^*_G(M_{s,t}) \xrightarrow{j_{s,t_*}^*} \operatorname{CH}^*_G(X_t \times \mathbb{A}^1) \xrightarrow{g_t^*} \operatorname{CH}^*_G(X_t) \\ \eta_{s,t_*} \downarrow & \downarrow \delta_{s,t_*} & \downarrow f_{t_*} & \downarrow f_{t_*} \\ \operatorname{CH}^*_G(N_{s,\leq t}) \xrightarrow{i_{s,\leq t_*}} \operatorname{CH}^*_G(M_{s,\leq t}) \xrightarrow{j_{s,\leq t}^*} \operatorname{CH}^*_G(X_{\leq t} \times \mathbb{A}^1) \xrightarrow{g_{\leq t}^*} \operatorname{CH}^*_G(X_{\leq t}) \end{array}$$

of equivariant higher Chow groups, where the left square commutes by the covariance of the push-forward map, the middle commutes by Proposition 2.2(1) and the right commutes by Corollary 4.2. Since the last horizontal maps in both rows are natural isomorphisms by the homotopy invariance, we shall often identify the last two terms in both rows and use  $j_{s, \le t}^*$  and  $(j_{s, \le t} \circ g_{\le t})^*$  interchangeably.

**Theorem 9.8.** The maps  $j_{s,\leq t}^*$  and  $j_{s,t}^*$  are surjective and there are ring homomorphisms

$$\overline{\operatorname{Sp}}_{X,s}^{\leq t}: \operatorname{CH}_{G}^{*}(X_{\leq t}; \mathbb{Q}) \to \operatorname{CH}_{G}^{*}(N_{s, \leq t}; \mathbb{Q}) \quad and$$
  
$$\overline{\operatorname{Sp}}_{X,s}^{t}: \operatorname{CH}_{G}^{*}(X_{t}; \mathbb{Q}) \to \operatorname{CH}_{G}^{*}(N_{s,t}; \mathbb{Q})$$

such that  $i_{s,\leq t}^* = \overline{\operatorname{Sp}}_{X,s}^{\leq t} \circ j_{s,\leq t}^*$  and  $i_{s,t}^* = \overline{\operatorname{Sp}}_{X,s}^t \circ j_{s,t}^*$ . Moreover, both the squares in the following diagram commute:

$$\begin{array}{cccc}
\operatorname{CH}_{G}^{*}(X_{\leq t}; \mathbb{Q}) & \xrightarrow{f_{t}^{*}} \operatorname{CH}_{G}^{*}(X_{t}; \mathbb{Q}) & \xrightarrow{f_{t}_{*}} \operatorname{CH}_{G}^{*}(X_{\leq t}; \mathbb{Q}) \\
& \overline{\operatorname{Sp}}_{X,s}^{\leq t} & & & & & & & \\
& \overline{\operatorname{Sp}}_{X,s}^{\leq t} & & & & & & & & \\
\operatorname{CH}_{G}^{*}(N_{s,\leq t}; \mathbb{Q}) & \xrightarrow{\eta_{s,t}^{*}} \operatorname{CH}_{G}^{*}(N_{s,t}; \mathbb{Q}) & \xrightarrow{\eta_{s,t}_{*}} \operatorname{CH}_{G}^{*}(N_{s,\leq t}; \mathbb{Q})
\end{array} \tag{9-4}$$

*Proof.* Using the results obtained so far in this section, one can define the specialization maps along the lines of the construction of such maps for *K*-theory in [Vezzosi and Vistoli 2003, Theorem 3.2; 2005]. However, it is not at all clear from the construction of the specialization maps in [Vezzosi and Vistoli 2005] that these maps have good functorial properties, and, in particular, if they are ring homomorphisms. Moreover, it is not clear if these maps will have the compatibility properties with the Chern character and Riemann–Roch maps (see [Krishna 2009b]) from the equivariant *K*-groups to higher Chow groups.

We give here a more direct and functorial construction of the specialization maps, which works both for the K-theory as well as the higher Chow groups, and the proof of various compatibilities of these maps then becomes essentially obvious. We give here the construction of these maps for the higher Chow groups. The same construction works also for the K-theory without any change.

First of all, using Corollary 9.7 and Proposition 9.2, we see that for  $1 \le s \le n$ and  $0 \le t \le s$ , the map  $CH^*_G(X_{\le s}) \to CH^*_G(X_{\le t})$  is surjective. We now consider the commutative diagram

Since the composite map in the bottom row of (9-3) is identity, we see by the homotopy invariance that  $j_{\leq s}^*$  is surjective. Thus  $j_{s,\leq t}^*$  is also surjective. Applying this surjectivity for  $j_{s,\leq t}^*$  and  $j_{s,\leq t-1}^*$ , we obtain the commutative diagram of Figure 1, which is such that the second and the third rows are exact. All the columns are exact by Corollary 9.7 and Proposition 9.2. We conclude that the localization sequence of the top row is also exact. This proves the surjectivity part of the theorem.

Next, we apply the self-intersection formula (Corollary 4.5) to the inclusions  $i_{s,\leq t}$  and  $i_{s,t}$  to see that the composites  $i_{s,\leq t}^* \circ i_{s,\leq t*}$  and  $i_{s,t}^* \circ i_{s,t*}$  are multiplication by the first Chern class  $c_1^G$  of the corresponding normal bundles. But these normal bundles are the inverse images of a line bundle on  $\mathbb{P}^1$ . It follows that these normal bundles are trivial, because the restriction of any line bundle on  $\mathbb{P}^1$  to  $\infty \in \mathbb{P}^1$  and hence on the fiber over  $\infty$  is clearly trivial. We conclude that the composites  $i_{s,\leq t}^* \circ i_{s,\leq t*}$  and  $i_{s,t}^* \circ i_{s,t*}$  are zero.

The diagram above now automatically defines the specializations  $\overline{\text{Sp}}_{X,s}^{\leq t}$  and  $\overline{\text{Sp}}_{X,s}^{t}$  and gives the desired factorization of  $i_{s,\leq t}^{*}$  and  $i_{s,t}^{*}$ . Since  $i_{s,t}^{*}$  and  $j_{s,t}^{*}$  are ring homomorphisms, and since the latter is surjective as shown in 1, we deduce that  $\overline{\text{Sp}}_{X,s}^{t}$  is also a ring homomorphism. The map  $\overline{\text{Sp}}_{X,s}^{\leq t}$  is a ring homomorphism for the same reason.


#### Figure 1

We are now left with the proof of the commutativity of (9-4). To prove that the right square commutes, we consider the following diagram.



It is easy to check that  $N_{s,\leq t}$  and  $M_{s,t}$  are Tor-independent over  $M_{s,\leq t}$  (see [Vezzosi and Vistoli 2005, Lemma 1]) and hence the back face of the diagram above commutes by Corollary 4.2. The upper face commutes by diagram 1. Since  $j_{s,t}^*$  is surjective, a diagram chase shows that the lower face also commutes, which is what we needed to prove.

Finally, since we have shown that  $\eta_{s,t_*}$  is injective, and the right square commutes, it now suffices to show that the composite square in (9-4) commutes in order to show that the left square commutes.

By the projection formula, the composite maps  $f_{t*} \circ f_t^*$  and  $\eta_{s,t*} \circ \eta_{s,t}^*$  are multiplication by  $f_{t*}(1)$  and  $\eta_{s,t*}(1)$  respectively. Since

$$\overline{\operatorname{Sp}}_{X,s}^{\leq t}$$
 and  $\overline{\operatorname{Sp}}_{X,s}^{t}$ 

are ring homomorphisms, it suffices to show that

$$\overline{\operatorname{Sp}}_{X,s}^{\leq t}(f_{t*} \circ j_{s,t}^{*}(1)) = \overline{\operatorname{Sp}}_{X,s}^{\leq t}(f_{t*}(1)) = \eta_{s,t*}(1).$$

But this follows directly from the commutativity of the right square.

10. Decomposition theorem for equivariant higher Chow groups

We use the specialization maps to prove the main decomposition theorem for the equivariant higher Chow groups of  $X \in \mathcal{V}_G^S$ , where *G* is a split diagonalizable group. We continue with the notation of the previous section.

Proposition 10.1. The restriction maps

$$\mathrm{CH}^*_G(X_{\leq s};\mathbb{Q}) \xrightarrow{(f^*_s,g^*_s)} \mathrm{CH}^*_G(X_s;\mathbb{Q}) \times \mathrm{CH}^*_G(X_{< s};\mathbb{Q})$$

define an isomorphism of rings

$$\mathrm{CH}^*_G(X_{\leq s};\mathbb{Q})\xrightarrow{\sim}\mathrm{CH}^*_G(X_s;\mathbb{Q})\underset{\mathrm{CH}^*_G(N^0_s;\mathbb{Q})}{\times}\mathrm{CH}^*_G(X_{< s};\mathbb{Q}),$$

where  $\operatorname{CH}^*_G(X_s; \mathbb{Q}) \xrightarrow{\eta^*_{s,\leq s-1}} \operatorname{CH}^*_G(N^0_s; \mathbb{Q})$  is the pull-back

$$\operatorname{CH}^*_G(X_s; \mathbb{Q}) \xrightarrow{\sim} \operatorname{CH}^*_G(N_s; \mathbb{Q}) \to \operatorname{CH}^*_G(N_s^0; \mathbb{Q})$$

and

$$\mathrm{CH}^*_G(X_{< s}; \mathbb{Q}) \xrightarrow{\overline{\mathrm{Sp}}_{X,s}^{\leq s-1}} \mathrm{CH}^*_G(N_{s, \leq s-1}; \mathbb{Q}) = \mathrm{CH}^*_G(N_s^0; \mathbb{Q})$$

is the specialization map of Theorem 9.8.

*Proof.* We only need to identify the pull-back and the specialization maps with the appropriate maps of Proposition 9.2. In the diagram

$$0 \longrightarrow \operatorname{CH}^*_G(X_s) \xrightarrow{f_{s,\infty_*}} \operatorname{CH}^*_G(N_s) \xrightarrow{\eta^*_{s,\leq s-1}} \operatorname{CH}^*_G(N^0_s) \longrightarrow 0$$

$$\downarrow f^*_{s,\infty}$$

$$\operatorname{CH}^*_G(X_s),$$

where  $f_{s,\infty}: X_s \to N_s$  is the 0-section embedding, the top sequence is exact, and the lower triangle commutes by Corollary 4.5. Since  $f_{s,\infty}^*$  is an isomorphism, this immediately identifies the pull-back map of the proposition with the quotient map

$$\operatorname{CH}^*_G(X_s) \to \frac{\operatorname{CH}^*_G(X_s)}{(c^G_{d_s}(N_s))}.$$

Next, we consider the diagram

$$\begin{array}{c|c}
\operatorname{CH}^{*}_{G}(X_{\leq s}) \xrightarrow{\overline{\operatorname{Sp}}_{X,s}^{\leq s}} \operatorname{CH}^{*}_{G}(N_{s}) \xrightarrow{f_{s,\infty}^{*}} \operatorname{CH}^{*}_{G}(X_{s}) \\
f_{s,\leq s-1}^{*} & & \downarrow & & \downarrow \\
\operatorname{CH}^{*}_{G}(X_{< s}) \xrightarrow{\overline{\operatorname{Sp}}_{X,s}^{\leq s-1}} \operatorname{CH}^{*}_{G}(N_{s}^{0}). \\
\end{array}$$
(10-1)

Since the left vertical arrow in the diagram above is surjective, we only need to show that

$$\overline{\operatorname{Sp}}_{X,s}^{\leq s-1} \circ f_{s,\leq s-1}^* = \eta_{s,\leq s-1}^* \circ f_s^*$$

in order to identify  $\overline{\text{Sp}}_{X,s}^{\leq s-1}$  with the map  $j^*$  of Proposition 9.2. It is clear from the diagram 1 and the definition of the specialization maps that the left square in the diagram (10-1) commutes. We have just shown above that the right side triangle also commutes. This reduces us to showing that

$$f_{s,\infty}^* \circ \overline{\mathrm{Sp}}_{X,s}^{\leq s} = f_s^*.$$
(10-2)

If  $X_s \times \mathbb{P}^1 \xrightarrow{F_s} M_s$  denotes the embedding (see (4-1)), then for  $x \in CH^*_G(X_{\leq s})$ , we can write  $x = j^*_{\leq s}(y)$  by Theorem 9.8. Then

$$f_{s,\infty}^* \circ \overline{\mathrm{Sp}}_{X,s}^{\leq s} \circ j_{\leq s}^*(y) = f_{s,\infty}^* \circ i_{s,\leq s}^*(y) = {}^{\dagger}g_{\infty,\leq s}^* \circ F_s^*(y) = g_{0,\leq s}^* \circ F_s^*(y) = f_s^* \circ j_{\leq s}^*(y) = f_s^*(x),$$

where  $=^{\dagger}$  follows from Corollary 4.2. This proves (10-2) and the proposition.  $\Box$ 

We need the following algebraic result before we prove the main theorem. Let A be a  $\mathbb{Q}$ -algebra (not necessarily commutative) and let Z(A) denote the center of A. For any linear form  $f(\underline{t}) = \sum_{i=1}^{n} a_i t_i$  in  $A[t_1, \ldots, t_n]$  such that  $a_i \in \mathbb{Q}$  for each i, let c(f) denote the vector  $(a_1, \ldots, a_n) \in \mathbb{Q}^n$  consisting of the coefficients of the form f.

**Lemma 10.2.** Let A be as above and let  $S = \{f_1, \ldots, f_s\}$  be a set of linear forms in  $A[t_1, \ldots, t_n]$  such that the vectors  $\{c(f_1), \ldots, c(f_s)\}$  are pairwise nonproportional in  $\mathbb{Q}^n$ . Let

$$\gamma_j = \sum_{i=0}^{d_j} m_i^j f_j^i$$

such that  $m_{d_i}^j \in \mathbb{Q}^*$  for  $1 \le j \le s$ , and  $m_{j'}^j \in Z(A)$  for all j, j'. Then one has

$$(\gamma_1 \cdots \gamma_s) = \bigcap_{j=1}^s (\gamma_j)$$

as ideals in  $A[t_1, \ldots, t_n]$ .

*Proof.* Using a simple induction, it suffices to show that for  $j \neq j'$ , the relation  $\gamma_j | q \gamma_{j'}$  implies that  $\gamma_j | q$ . So we can assume  $S = \{f_1, f_2\}$ . Since  $c(f_1)$  and  $c(f_2)$  are nonproportional, we can extend the set  $\{c(f_1), c(f_2)\}$  to a basis *B* of  $\mathbb{Q}^n$ . Applying the linear automorphism of  $A[t_1, \ldots, t_n]$  given by the invertible matrix *B*, we can assume that  $f_j = t_j$  for j = 1, 2. Now the proof follows along the same lines as the proof of [Vezzosi and Vistoli 2003, Lemma 4.9].

**Theorem 10.3.** Let G be a split diagonalizable group of dimension n and let  $X \in \mathcal{V}_G^S$ . The ring homomorphism

$$\operatorname{CH}^*_G(X; \mathbb{Q}) \to \prod_{s=0}^n \operatorname{CH}^*_G(X_s; \mathbb{Q})$$

is injective. Moreover, its image consists of the n-tuples  $(\alpha_s)$  in the product with the property that for each s = 1, ..., n, the pull-back of  $\alpha_s \in CH^*_G(X_s; \mathbb{Q})$  in  $CH^*_G(N_{s,s-1}; \mathbb{Q})$  is the same as  $\overline{Sp}^{s-1}_{X,s}(\alpha_{s-1}) \in CH^*_G(N_{s,s-1}; \mathbb{Q})$ . In other words, there is a ring isomorphism

$$\mathrm{CH}^*_G(X;\mathbb{Q})\xrightarrow{\sim}\mathrm{CH}^*_G(X_n;\mathbb{Q})\underset{\mathrm{CH}^*_G(N_{n,n-1};\mathbb{Q})}{\times}\cdots\underset{\mathrm{CH}^*_G(N_{1,0};\mathbb{Q})}{\times}\mathrm{CH}^*_G(X_0;\mathbb{Q}).$$

*Proof.* We prove by induction on the largest integer s such that  $X_s \neq \emptyset$ .

If s = 0, there is nothing to prove. If s > 0, we have by induction

$$\operatorname{CH}^*_G(X_{< s}) \xrightarrow{\cong} \operatorname{CH}^*_G(X_{s-1}) \underset{\operatorname{CH}^*_G(N_{s-1,s-2})}{\times} \cdots \underset{\operatorname{CH}^*_G(N_{1,0})}{\times} \operatorname{CH}^*_G(X_0).$$
(10-3)

Using (10-3) and Proposition 10.1, it suffices to show that if  $\alpha_s \in CH^*_G(X_s)$ and if  $\alpha_{<s} \in CH^*_G(X_{<s})$  with the restriction  $\alpha_{s-1} \in CH^*_G(X_{s-1})$  are such that  $\alpha_s \mapsto \alpha_s^0 \in CH^*_G(N_s^0)$  and  $\alpha_s \mapsto \alpha_{s,s-1} \in CH^*_G(N_{s,s-1})$ , then

$$\overline{\mathrm{Sp}}_{X,s}^{\leq s-1}(\alpha_{< s}) = \alpha_s^0$$
 if and only if  $\overline{\mathrm{Sp}}_{X,s}^{s-1}(\alpha_{s-1}) = \alpha_{s,s-1}$ .

Using the commutativity of the left square in Theorem 9.8, this is reduced to showing that the restriction map

$$\operatorname{CH}^*_G(N^0_s) \to \operatorname{CH}^*_G(N_{s,s-1}) \tag{10-4}$$

is injective.

To prove this, we first use Proposition 9.5 to assume that the toral component T of the isotropy groups of the points of  $X_s$  is fixed, and choose a splitting  $G = D \times T$ .

Now, following the proof of the analogous result for *K*-theory [Vezzosi and Vistoli 2003, Theorem 4.5], we can write

$$N_s = E = \bigoplus_{i=1}^q E_i$$
 and  $N_{s,s-1} = \coprod_i E_i^0$ ,

where each  $E_i$  is of the form  $\bigoplus E_{m_j} \otimes \chi_i^{m_j}$  such that for  $i \neq i'$ ,  $\chi_i$  and  $\chi_{i'}$  are nonproportional characters of T, and  $E_i^0$  is embedded in E by setting all the other components equal to zero. Let  $d_i = \operatorname{rank}(E_i)$ .

Now we see from Proposition 10.1 that

$$\operatorname{Ker}(\operatorname{CH}^*_G(X_s) \to \operatorname{CH}^*_G(N_{s,s-1})) = \bigcap_i (c^G_{d_i}(E_i)),$$
$$\operatorname{Ker}(\operatorname{CH}^*_G(X_s) \to \operatorname{CH}^*_G(N^0_s)) = (c^G_{d_s}(N_s)) \quad \text{with } d_s = \sum d_i$$

Put  $\gamma_i = c_{d_i}^G(E_i)$  and  $\gamma = c_{d_s}^G(N_s)$ . Since the map  $CH_G^*(X_s) \to CH_G^*(N_s^0)$  is surjective, showing the injectivity of the map in (10-4) is equivalent to showing that

$$(\gamma) = \left(\prod_{i} \gamma_{i}\right) = \bigcap_{i} (\gamma_{i}) \tag{10-5}$$

in  $CH_D^*(X_s)[t_1, ..., t_s].$ 

However, we have seen in the proof of Proposition 9.4 that each  $\gamma_i$  is of the form

$$\gamma_i = u_i^{d_i} + \alpha_{d_i-1}^i u_i^{d_i-1} + \dots + \alpha_1^i u_i + \alpha_0^i,$$

where  $\alpha_j^i \in CH_D^*(X_s, 0) \subseteq Z(CH_D^*(X_s))$  and  $u_i = c_1^T(L_{\chi_i}) = \sum_{j=1}^s b_j^i t_j \neq 0$  in  $\mathbb{Q}[t_1, \ldots, t_s]$ . Moreover, the pairwise nonproportionality of  $\chi_i$  implies the same for the vectors  $\{c(u_1), \ldots, c(u_q)\}$  in  $\mathbb{Q}^s$ . We now apply Lemma 10.2 to conclude the proof of (10-5) and hence the theorem.

#### 11. Equivariant higher Chow groups of toric varieties

In this section, we apply our decomposition theorem to give explicit descriptions of the equivariant higher Chow groups of smooth toric varieties. An analogous description of the equivariant cohomology of such varieties was earlier given by Bifet, De Concini and Procesi in [Bifet et al. 1990] and such a description of the equivariant *K*-theory was given by Vezzosi and Vistoli [2003]. Brion [1997, Theorem 5.4] had proven similar results for the classical equivariant Chow groups of toric varieties. As a consequence of our descriptions of the equivariant higher Chow groups, we shall obtain formulas for the ordinary higher Chow groups (or motivic cohomology) of smooth toric varieties.

Let *T* be a split torus of rank *n* over *k*. Let  $M = \text{Hom}(\mathbb{G}_m, T)$  be the lattice of the one-parameter subgroups of *T* and let  $M^{\vee}$  be the character lattice of *T*. Let  $\Delta$  be a fan in  $M_{\mathbb{R}}$  and let  $\Delta_1$  and  $\Delta_{\max}$  denote the subsets of the one-dimensional cones and the maximal cones in  $\Delta$ , respectively.

Let  $X = X(\Delta)$  be the smooth toric variety associated to the fan  $\Delta$ . The smoothness of X is equivalent to the condition that every positive dimensional cone of  $\Delta$  is generated by it edges such that the primitive vectors along these edges form a

subset of a basis of M. In this case, there is an one-to-one correspondence between the *T*-orbits in X and the cones in  $\Delta$ . For every cone  $\sigma \in \Delta$ , the corresponding orbit  $O_{\sigma}$  is isomorphic to the torus  $T/T_{\sigma}$ , where  $T_{\sigma}$  is associated to the sublattice  $M_{\sigma}$  of M generated by  $\sigma \cap M$ . Under this isomorphism, the origin (identity point) of  $T/T_{\sigma}$  corresponds to the distinguished k-rational point  $x_{\sigma}$  of  $O_{\sigma}$ . In particular, for every  $0 \le s \le n$ ,  $X_s$  is of the form

$$X_s = \coprod_{\dim(\sigma)=s} O_{\sigma} \cong \coprod_{\dim(\sigma)=s} T/T_{\sigma}.$$
 (11-1)

We shall write  $\tau \leq \sigma$  if  $\tau$  is a face of  $\sigma$  as cones in  $\Delta$ . The orbit closure  $V_{\sigma}$  of  $O_{\sigma}$  is the toric variety associated to the fan  $*(\sigma) = \{\tau \in \Delta \mid \sigma \leq \tau\}$ , called the star of  $\sigma$ . Moreover, it is clear from the characterization of the smoothness of toric varieties that  $V_{\sigma}$  is also smooth and is the disjoint union of all orbits  $O_{\tau}$  such that  $\sigma$  is a face of  $\tau$ . In particular,  $O_{\sigma}$  is closed in X if and only if  $\sigma \in \Delta_{\text{max}}$ . The following is our first description of the equivariant higher Chow groups of smooth toric varieties.

**Theorem 11.1.** Let  $X = X(\Delta)$  be a smooth toric variety associated to a fan  $\Delta$  in  $M_{\mathbb{R}}$ . There is an injective homomorphism of S-algebras

$$\Phi_X: \mathrm{CH}^*_T(X; \mathbb{Q}) \to \prod_{\sigma \in \Delta_{\max}} \mathrm{CH}^*(k; \mathbb{Q}) \otimes S(T_\sigma; \mathbb{Q}).$$

An element

$$(a_{\sigma}) \in \prod_{\sigma \in \Delta_{\max}} \operatorname{CH}^{*}(k; \mathbb{Q}) \otimes S(T_{\sigma}; \mathbb{Q})$$

is in the image of this homomorphism if and only if for any two maximal cones  $\sigma_1$ and  $\sigma_2$ , the restrictions of  $a_{\sigma_1}$  and  $a_{\sigma_2}$  to  $CH^*(k; \mathbb{Q}) \otimes S(T_{\sigma_1 \cap \sigma_2}; \mathbb{Q})$  coincide.

*Proof.* We only need to appropriately identify the various terms and the maps in the statement of Theorem 10.3. We follow the notation of Section 9 and Section 10. It follows from [Vezzosi and Vistoli 2003, Lemma 6.1] that for every  $s \ge 1$ , there is a canonical isomorphism

$$N_{s,s-1} = \coprod_{\substack{\sigma \in \Delta \\ \dim(\sigma) = s}} \coprod_{\tau \in \partial \sigma} O_{\tau}.$$
 (11-2)

Furthermore, for each *s*-dimensional cone  $\sigma$  and  $\tau \in \partial \sigma$ , the composition of the map

$$\operatorname{Sp}_{X,s}^{s-1}:\operatorname{CH}_{T}^{*}(X_{s-1}) = \prod_{\substack{\tau \in \Delta \\ \dim(\tau) = s-1}} \operatorname{CH}_{T}^{*}(O_{\tau}) \to \operatorname{CH}_{T}^{*}(N_{s,s-1}) = \prod_{\substack{\sigma \in \Delta \\ \dim(\sigma) = s}} \prod_{\tau \in \partial \sigma} \operatorname{CH}_{T}^{*}(O_{\tau})$$

with the projection

$$\Pr_{\sigma,\tau}:\prod_{\substack{\sigma\in\Delta\\\dim(\sigma)=s}}\prod_{\tau\in\partial\sigma}\operatorname{CH}^*_T(O_{\tau})\to\operatorname{CH}^*_T(O_{\tau})$$

is the projection map

$$\prod_{\substack{\tau \in \Delta \\ \dim(\tau) = s - 1}} \operatorname{CH}^*_T(O_\tau) \to \operatorname{CH}^*_T(O_\tau)$$

The identification above of the specialization maps in [Vezzosi and Vistoli 2003, Lemma 6.1] was shown for the equivariant K-theory, but the same holds in the present case as well without any modification in view of the construction of these specializations for higher Chow groups in Section 9.

It also follows from Corollary 3.2 and Theorem 3.5 that

$$\operatorname{CH}_{T}^{*}(O_{\sigma}) \cong \operatorname{CH}_{T}^{*}(T/T_{\sigma}) \cong \operatorname{CH}_{T_{\sigma}}^{*}(k) \cong \operatorname{CH}^{*}(k) \otimes S(T_{\sigma}).$$
(11-3)

We conclude now from (11-1) and Theorem 10.3 that  $\operatorname{CH}_T^*(X)$  is a subring of  $\prod_{\sigma \in \Delta} \operatorname{CH}^*(k) \otimes S(T_{\sigma})$ , consisting of elements  $(a_{\sigma})$  with the property that the restriction of  $a_{\sigma} \in \operatorname{CH}^*(k) \otimes S(T_{\sigma})$  to  $\operatorname{CH}^*(k) \otimes S(T_{\tau})$  coincides with  $a_{\tau}$  whenever  $\tau \leq \sigma$ . The theorem now follows from the fact that every cone in  $\Delta$  is contained in a maximal cone in  $\Delta$ .

As an immediate consequence of Theorem 11.1, we obtain the following localization theorem for the equivariant higher Chow groups of smooth projective toric varieties. This was earlier proven for  $CH_T^*(X, 0)$  by Brion [1997, Theorem 3.4].

**Corollary 11.2.** Let  $X = X(\Delta)$  be a smooth projective toric variety and  $\{x_1, \ldots, x_r\}$  be the fixed point locus of X. Then the map

$$\operatorname{CH}^*_T(X; \mathbb{Q}) \to \operatorname{CH}^*_T(X^T; \mathbb{Q}) \xrightarrow{\sim} (\operatorname{CH}^*(k; \mathbb{Q})[t_1, \dots, t_n])^r$$

is injective and its image is the set of all n-tuples  $(f_1, \ldots, f_n)$  such that  $f_i \equiv f_j \pmod{\chi}$  whenever  $x_i$  and  $x_j$  lie on a T-invariant smooth irreducible curve on which T acts through its character  $\chi$ .

*Proof.* This follows directly from Theorem 11.1 once we observe that the fixed points of X for the torus action are same as the *T*-orbits corresponding to the maximal cones in  $\Delta$  that are *n*-dimensional. Moreover, the orbit closures corresponding to the codimension one cones in  $\Delta$  are the smooth *T*-invariant curves.

**11a.** *Stanley–Reisner presentation.* Using Theorem 11.1, we now give another explicit presentation of the equivariant higher Chow groups of smooth toric varieties. This presentation is analogous to the Stanley–Reisner presentation of the equivariant cohomology in [Bifet et al. 1990, Theorem 8] and equivariant *K*-theory in [Vezzosi

and Vistoli 2003, Theorem 6.1]. This presentation has the advantage that it can often be used to describe the ordinary higher Chow groups of smooth toric varieties.

Let *T* be a split torus of rank *n* and let *M* denote the lattice of the one-parameter subgroups of *T*. Let  $X = X(\Delta)$  be a smooth toric variety associated to a fan  $\Delta$  in  $M_{\mathbb{R}}$ . For  $r \ge 1$ , let  $\Delta_r$  denote the set of *r*-dimensional cones in  $\Delta$ . For  $\sigma \in \Delta_{\max}$ , let  $M_{\sigma}$  denote the sublattice of one-parameter subgroups of  $T_{\sigma}$  so that  $\widehat{T}_{\sigma} = M_{\sigma}^{\vee}$  as an abelian group. For any  $\rho \in \Delta_1$ , let  $v_{\rho}$  denote the generator of the monoid  $\rho \cap M$ . Note that if  $\{\rho_1, \ldots, \rho_s\}$  is the set of one-dimensional faces of  $\sigma \in \Delta_{\max}$ , then the smoothness of *X* implies that  $\{v_{\rho_1}, \ldots, v_{\rho_s}\}$  is a basis of  $M_{\sigma}$ . Let  $\{v_{\rho_1}^{\vee}, \ldots, v_{\rho_s}^{\vee}\}$ denote the dual basis of  $M_{\sigma}^{\vee}$ .

We recall that for  $\sigma \in \Delta$ , there is a canonical isomorphism of abelian groups  $\widehat{T}_{\sigma} \hookrightarrow S(T_{\sigma})_1$  given by  $\chi \mapsto c_1^{T_{\sigma}}(L_{\chi})$ . For each  $\rho \in \Delta_1$ , we define an element  $u_{\rho} = (u_{\rho}^{\sigma}) \in \prod_{\sigma \in \Delta_{\max} \prod} S(T_{\sigma})$  such that

$$u_{\rho}^{\sigma} = \begin{cases} v_{\rho}^{\vee} & \text{if } \rho \leq \sigma, \\ 0 & \text{otherwise.} \end{cases}$$
(11-4)

Then  $u_{\rho}$  has the property that for all  $\sigma_1, \sigma_2 \in \Delta_{\max}$ , the restrictions of  $u_{\rho}^{\sigma_1} \in \widehat{T}_{\sigma_1}$  and  $u_{\rho}^{\sigma_2} \in \widehat{T}_{\sigma_2}$  in  $\widehat{T}_{\sigma_1 \cap \sigma_2}$  coincide.

We have the obvious inclusion

$$\prod_{\sigma \in \Delta_{\max}} S(T_{\sigma}) \subseteq \prod_{\sigma \in \Delta_{\max}} CH^*(k) \otimes S(T_{\sigma})$$
(11-5)

and using the description of  $\operatorname{CH}_T^*(X)$  in Theorem 11.1 and the description of  $u_\rho$  above, we can consider these  $u_\rho$  as elements of the ring  $\operatorname{CH}_T^*(X)$ . In other words, we get a bigraded  $\operatorname{CH}^*(k)$ -algebra homomorphism

$$\operatorname{CH}^*(k)[t_{\rho}] \to \operatorname{CH}^*_T(X), \quad t_{\rho} \mapsto u_{\rho},$$

where  $CH^*(k)[t_{\rho}]$  is the polynomial algebra  $CH^*(k)[t_{\rho} | \rho \in \Delta_1]$ .

If *S* is a subset of  $\Delta_1$  that is not contained in any maximal cone of  $\Delta$ , then for any given  $\sigma \in \Delta_{\text{max}}$ , there is one  $\rho \in S$  such that  $\rho \nleq \sigma$ . This implies in particular that  $u_{\rho}^{\sigma} = 0$ . We conclude from this that the elements  $u_{\rho}$  satisfy the relation

$$\prod_{\rho \in S} u_{\rho} = 0 \quad \text{in } \operatorname{CH}^*_T(X) \tag{11-6}$$

whenever  $S \subseteq \Delta_1$  is such that it is not contained in any maximal cone of  $\Delta$ . We shall denote the collection of all such subsets of  $\Delta_1$  by  $\Delta_1^0$ . We conclude that if  $I_{\Delta}$  denotes the graded ideal of CH<sup>\*</sup>(*k*)[ $t_{\rho}$ ] generated by the set of monomials  $\{\prod_{\rho \in S} t_{\rho} \mid S \in \Delta_1^0\}$ , then there is a CH<sup>\*</sup>(*k*)-algebra homomorphism

$$\Psi_X : \frac{\operatorname{CH}^*(k)[t_\rho]}{I_\Delta} \to \operatorname{CH}^*_T(X), \quad t_\rho \mapsto u_\rho.$$
(11-7)

Note also that any character  $\chi \in M^{\vee}$  defines multiplication by the element  $t_{\chi} = \sum_{\rho \in \Delta_1} \langle \chi, v_{\rho} \rangle t_{\rho}$  in CH<sup>\*</sup>(k)[ $t_{\rho}$ ], and this makes the term on the left hand side of (11-7) an *S*-algebra and  $\Psi_X$  is also an *S*-algebra homomorphism. Furthermore, it is easy to check from the definition of  $u_{\rho}$  that it is the fundamental class (see [Edidin and Graham 1998, Section 2]) of the *T*-equivariant Chow cycle

$$[V_{\rho} \to X] \in CH^1_T(X, 0) \subsetneq CH^*_T(X),$$

where  $V_{\sigma}$  is the orbit closure in X associated to a cone  $\sigma \in \Delta$ .

**Theorem 11.3.** For a smooth toric variety  $X = X(\Delta)$  associated to a fan  $\Delta$  in  $M_{\mathbb{R}}$ , the homomorphism  $\Psi_X$  is an isomorphism with rational coefficients.

*Proof.* We prove the theorem by induction on the number of maximal cones in  $\Delta$ . Suppose  $\Delta_{\max} = \{\sigma\}$  is a singleton set. In that case,  $\sigma$  is the only maximal cone and  $X = U_{\sigma}$  is a *T*-equivariant vector bundle over  $O_{\sigma}$  such that the inclusion

$$O_{\sigma} \stackrel{i_{\sigma}}{\hookrightarrow} X$$

is the zero-section embedding. Hence, we conclude from (11-3) that there are isomorphisms

$$\operatorname{CH}_{T}^{*}(X) \xrightarrow[i_{\sigma}^{*}]{\sim} \operatorname{CH}_{T}^{*}(O_{\sigma}) \cong \operatorname{CH}^{*}(k) \otimes S(T_{\sigma}) = \operatorname{CH}^{*}(k)[t_{1}, \ldots, t_{s}]$$

where s is the dimension of  $\sigma$ . It is also clear in this case that the ideal  $I_{\Delta}$  in (11-7) is zero. Hence, we have isomorphisms

$$\operatorname{CH}^*(k)[t_1,\ldots,t_s] \xrightarrow{\simeq} \operatorname{CH}^*_T(X) \xrightarrow{\simeq} \operatorname{CH}^*(k) \otimes S(T_{\sigma}).$$

We consider now the general case. We assume that  $|\Delta_{\max}| \ge 2$  and choose a maximal cone  $\sigma$  of dimension  $s \ge 1$  in  $\Delta$ . Let  $X' = X'(\Delta')$  be the toric variety associated to the fan  $\Delta' = \Delta \setminus \{\sigma\}$ . Note that  $O_{\sigma}$  is a closed *T*-orbit in *X* and *X'* is the complement of  $O_{\sigma}$  in *X*. Let  $U_{\sigma} \subset X$  be the principal open set associated to the fan consisting of all faces of  $\sigma$  and let *U'* be the complement of  $O_{\sigma}$  in  $U_{\sigma}$ . Then *U'* is nothing but the complement of the zero-section in the *T*-equivariant vector bundle  $U_{\sigma} \rightarrow O_{\sigma}$ . Let  $i_{\sigma} : O_{\sigma} \hookrightarrow X$  and  $j_{\sigma} : X' \hookrightarrow X$  denote the *T*-invariant closed and open embeddings respectively. Let  $S_{\sigma} = \{\rho_1, \ldots, \rho_s\}$  be the set of one-dimensional faces of  $\sigma$  and set

$$x_{\sigma} = \prod_{j=1}^{s} t_{\rho_j} \in \frac{\operatorname{CH}^*(k)[t_{\rho}]}{I_{\Delta}} \quad \text{and} \quad y_{\sigma} = \prod_{j=1}^{s} u_{\rho_j} \in \operatorname{CH}^*_T(X).$$

Since  $N_{O_{\sigma}/X} = N_{O_{\sigma}/U_{\sigma}}$  and since the latter is of the form  $\bigoplus_{j=1}^{s} L_{\chi_{j}}$ , where  $\{\chi_{1}, \ldots, \chi_{s}\}$  is a basis of  $\widehat{T}_{\sigma}$ , it follows from the definition of the elements  $u_{\rho}$  (see

Proposition 9.4) that

$$c_s^T(N_{O_\sigma/X}) = y_\sigma \in \operatorname{CH}^*_T(X).$$
(11-8)

We consider the diagram

where the horizontal maps on the top are the obvious isomorphisms taking  $t_{\rho_j}$  to  $u_{\rho_j}$ . The left and the right vertical maps are the multiplication by the indicated elements in the target rings. We claim that all the vertical arrows are injective and the left square in this diagram commutes.

To prove the claim, notice that the composite outer square clearly commutes by the definition of  $x_{\sigma}$  and  $y_{\sigma}$  and the map  $\Psi_X$ . Since  $\Phi_X$  is injective by Theorem 11.1, we only need to show that the right square commutes and the right vertical arrow is injective to prove the claim.

We first observe that the right vertical arrow is the multiplication by  $y_{\sigma}$  on the factor  $CH^*(k) \otimes S(T_{\sigma})$  and is zero on the other factors of  $\prod_{\tau \in \Delta_{max}} CH^*(k) \otimes S(T_{\tau})$ . Thus the required injectivity is equivalent to showing that the multiplication by  $y_{\sigma}$  is injective in  $CH^*(k) \otimes S(T_{\sigma})$ . We can thus assume that  $X = U_{\sigma}$  and then  $CH^*_T(X) \cong CH^*(k)[t_1, \ldots, t_s]$ . In this case, y is just the element  $t_1 \cdots t_s$  and hence is a nonzero divisor in  $CH^*(k)[t_1, \ldots, t_s]$ .

To show the commutativity of the right square, we observe from the proof of Theorem 11.1 that  $\Phi_X$  is simply the product of the pull-back maps

$$i_{\tau}^* : \operatorname{CH}^*_T(X) \to \operatorname{CH}^*_T(O_{\tau}) \text{ for } \tau \in \Delta_{\max}.$$

Hence the composite  $\Phi_X \circ i_{\sigma*}$  is  $i_{\sigma}^* \circ i_{\sigma*}$  on the factor  $\operatorname{CH}_T^*(O_{\sigma})$  and zero on the other factors of  $\prod_{\tau \in \Delta_{\max}} \operatorname{CH}_T^*(O_{\tau})$ . Since we have just seen that the composite

$$\operatorname{CH}^*_T(O_{\sigma}) \xrightarrow{y_{\sigma}} \prod_{\tau \in \Delta_{\max}} \operatorname{CH}^*_T(O_{\tau})$$

is of similar type, we are reduced to showing that the triangle



502

commutes. But this follows immediately from Corollary 4.5 and (11-8). This proves the claim.

To complete the proof of the theorem, we now consider the diagram

(11-10) where  $\overline{j}_{\sigma}^*$  is the natural quotient map by the ideal  $(x_{\sigma})$  in CH<sup>\*</sup> $(k)[t_{\rho}]/I_{\Delta}$ . Note that the image of the first map in the top row is the ideal  $(x_{\sigma})$  because the product of  $x_{\sigma}$  with any  $t_{\rho}$  for  $\rho \notin \{\rho_1, \ldots, \rho_s\}$  is zero.

The left square in this diagram commutes and the first maps in both the rows are injective by the claim above. The bottom row is exact by Proposition 2.2. Since  $\sigma$  is not a cone of  $\Delta'$ , the element  $x_{\sigma}$  is zero in  $CH^*(k)[t_{\rho}, \rho \in \Delta'_1]/I_{\Delta'}$  and hence the map  $j_{\sigma}^* \circ \Psi_X$  has a factorization:

$$\frac{\operatorname{CH}^*(k)[t_{\rho}]}{I_{\Delta}} \twoheadrightarrow \frac{\operatorname{CH}^*(k)[t_{\rho}]}{(I_{\Delta}, x_{\sigma})} \to \frac{\operatorname{CH}^*(k)[t_{\rho}, \ \rho \in \Delta'_1]}{I_{\Delta'}} \xrightarrow{\Psi_{X'}} \operatorname{CH}^*_T(X'),$$

where the middle arrow is the natural map of the Stanley–Reisner rings induced by the inclusion of the fans  $\Delta' \subset \Delta$ . Letting  $\overline{\Psi}_{X'}$  denote the composite

$$\frac{\operatorname{CH}^*(k)[t_{\rho}]}{(I_{\Delta}, x_{\sigma})} \to \frac{\operatorname{CH}^*(k)[t_{\rho}, \ \rho \in \Delta'_1]}{I_{\Delta'}} \xrightarrow{\Psi_{X'}} \operatorname{CH}^*_T(X'),$$

we see that the right square in the diagram (11-10) also commutes.

If all the cones of  $\Delta$  are at most one-dimensional, then  $x_{\sigma} = t_{\rho}$ , where  $\rho = \sigma$ and it is obvious that  $CH^*(k)[t_{\rho}]/(I_{\Delta}, x_{\sigma})$  is the Stanley–Reisner ring associated to the fan  $\Delta'$ . If  $\Delta$  has a cone of dimension at least two, we can assume that  $\sigma$  is of dimension at least two. In that case, we have  $\Delta'_1 = \Delta_1$  and the natural inclusion  $\Delta_1^0 \subseteq \Delta'_1^0$  gives the equality  $\Delta'_1^0 = \Delta_1^0 \coprod \{S_{\sigma}\}$ . In particular, we have

$$\frac{\operatorname{CH}^*(k)[t_{\rho}, \rho \in \Delta_1]}{(I_{\Delta}, x_{\sigma})} \xrightarrow{\sim} \frac{\operatorname{CH}^*(k)[t_{\rho}, \rho \in \Delta'_1]}{I_{\Delta'}}.$$

On the other hand,  $\Delta'$  is a fan with smaller number of maximal cones than in  $\Delta$  and  $X' = X'(\Delta')$ . Hence the map

$$\frac{\operatorname{CH}^*(k)[t_{\rho},\,\rho\in\Delta'_1]}{I_{\Delta'}}\xrightarrow{\Psi_{X'}}\operatorname{CH}^*_T(X')$$

is an isomorphism by induction. We conclude that the map  $\overline{\Psi}_{X'}$  in the diagram (11-10) is an isomorphism. A diagram chase in (11-10) now shows that  $\Psi_X$  is also an isomorphism.

As an important application of Theorem 11.3, we obtain the following presentation of the ordinary higher Chow groups (motivic cohomology groups) of smooth toric varieties. An explicit description of  $CH^*(X, 0)$  for a smooth projective toric variety X was given in [Fulton 1993, Proposition 5.2]. The following result extends this to all smooth toric varieties, not necessarily projective. In fact, such a description extends to all higher Chow groups of smooth projective toric varieties. In particular, we obtain another proof of Corollary 7.3 with rational coefficients. Recall that for every  $\sigma \in \Delta$ , the orbit closure  $V_{\sigma} = \overline{O_{\sigma}}$  in X is a T-invariant Weil divisor and defines a unique element  $[V_{\sigma}] \in CH^1(X, 0)$ .

**Corollary 11.4.** Let  $X = X(\Delta)$  be a smooth projective toric variety. Then the assignment  $t_{\rho} \mapsto [V_{\sigma}]$  defines a CH<sup>\*</sup>(k; Q)-algebra isomorphism

$$\overline{\Psi}_{X}: \frac{\operatorname{CH}^{*}(k; \mathbb{Q})[t_{\rho}]}{\left(I_{\Delta}, \sum_{\rho \in \Delta_{1}} \langle \chi, v_{\rho} \rangle t_{\rho}\right)} \to \operatorname{CH}^{*}(X; \mathbb{Q}),$$
(11-11)

where  $\chi$  runs over  $M^{\vee}$ .

If X is not necessarily projective, the map

$$\frac{\mathbb{Q}[t_{\rho}]}{\left(I_{\Delta}, \sum_{\rho \in \Delta_{1}} \langle \chi, v_{\rho} \rangle t_{\rho}\right)} \to \mathrm{CH}^{*}(X, 0; \mathbb{Q})$$

is a ring isomorphism.

*Proof.* We have already seen before that every character  $\chi \in M^{\vee}$  acts on CH<sup>\*</sup>(k)[ $t_{\rho}$ ] by multiplication with the element  $\sum_{\rho \in \Delta_1} \langle \chi, v_{\rho} \rangle t_{\rho}$  which makes the left hand side of (11-11) an *S*-algebra. The corollary now follows directly from Theorem 11.3 and [Krishna 2009a, Theorem 1.3]. The second isomorphism follows in the same way from Theorem 11.3 and [Brion 1997, Corollary 2.3].

**Corollary 11.5.** Let  $X = X(\Delta)$  be a smooth toric variety. Then there are canonical ring isomorphisms

$$\operatorname{CH}^{*}(k; \mathbb{Q}) \otimes \operatorname{CH}^{*}_{T}(X, 0; \mathbb{Q}) \xrightarrow{\sim} \operatorname{CH}^{*}_{T}(X; \mathbb{Q}),$$
  
$$\operatorname{CH}^{*}(k; \mathbb{Q}) \otimes \operatorname{CH}^{*}(X, 0; \mathbb{Q}) \xrightarrow{\sim} \operatorname{CH}^{*}(X; \mathbb{Q}).$$

*Proof.* It follows from (11-7) that  $CH^*(k) \otimes (\mathbb{Q}[t_\rho]/I_\Delta) \xrightarrow{\sim} CH^*(k)[t_\rho]/I_\Delta$ . The first part of the corollary now follows directly from Theorem 11.3. The second part follows from the first and [Krishna 2009a, Theorem 1.1], which says that there is a convergent spectral sequence

$$\operatorname{Tor}_{p}^{S}(\mathbb{Q}, \operatorname{CH}_{T}^{*}(X, q)) \Rightarrow \operatorname{CH}^{*}(X, p+q).$$

**Remark 11.6.** All the results in this section about the (equivariant) higher Chow groups of smooth toric varieties have been stated over the rationals. However, an attentive reader can check that these results (and the proofs) for the subrings  $CH_T^*(X, 0)$  and  $CH^*(X, 0)$  hold true with the integral coefficients. The basic reason is that  $CH_T^*(k, 0)$  and  $CH^*(k, 0)$  are torsion-free abelian groups. But this is false for the higher Chow groups of *k*.

### Acknowledgments

I wish to thank Angelo Vistoli, who drew my attention to the reconstruction theorem for equivariant K-theory in [Vezzosi and Vistoli 2003] and motivated me to think about such questions for equivariant higher Chow groups. I would also like to thank the referee for carefully reading this paper and for giving valuable input to improve its presentation.

#### References

- [Bifet et al. 1990] E. Bifet, C. De Concini, and C. Procesi, "Cohomology of regular embeddings", *Adv. Math.* **82**:1 (1990), 1–34. MR 91h:14052 Zbl 0743.14018
- [Bloch 1986] S. Bloch, "Algebraic cycles and higher *K*-theory", *Adv. in Math.* **61**:3 (1986), 267–304. MR 88f:18010 Zbl 0608.14004
- [Borel 1991] A. Borel, *Linear algebraic groups*, 2nd ed., Graduate Texts in Mathematics **126**, Springer, New York, 1991. MR 92d:20001 Zbl 0726.20030
- [Brion 1997] M. Brion, "Equivariant Chow groups for torus actions", *Transform. Groups* **2**:3 (1997), 225–267. MR 99c:14005 Zbl 0916.14003
- [Demazure 1973] M. Demazure, "Invariants symétriques entiers des groupes de Weyl et torsion", *Invent. Math.* **21** (1973), 287–301. MR 49 #7268 Zbl 0269.22010
- [Demazure 1974] M. Demazure, "Désingularisation des variétés de Schubert généralisées", *Ann. Sci. École Norm. Sup.* (4) **7** (1974), 53–88. MR 50 #7174 Zbl 0312.14009
- [Edidin and Graham 1997] D. Edidin and W. Graham, "Characteristic classes in the Chow ring", *J. Algebraic Geom.* **6**:3 (1997), 431–443. MR 98m:14007 Zbl 0922.14003
- [Edidin and Graham 1998] D. Edidin and W. Graham, "Equivariant intersection theory", *Invent. Math.* 131:3 (1998), 595–634. MR 99j:14003a Zbl 0940.14003
- [Edidin and Graham 2000] D. Edidin and W. Graham, "Riemann–Roch for equivariant Chow groups", *Duke Math. J.* **102**:3 (2000), 567–594. MR 2001f:14018 Zbl 0997.14002
- [Ellingsrud and Strømme 1989] G. Ellingsrud and S. A. Strømme, "On the Chow ring of a geometric quotient", *Ann. of Math.* (2) **130**:1 (1989), 159–187. MR 90h:14019 Zbl 0716.14002
- [Fulton 1984] W. Fulton, *Intersection theory*, Ergebnisse der Mathematik und ihrer Grenzgebiete (3) **2**, Springer, Berlin, 1984. MR 85k:14004 Zbl 0541.14005
- [Fulton 1993] W. Fulton, *Introduction to toric varieties*, Annals of Mathematics Studies **131**, Princeton University Press, 1993. MR 94g:14028 Zbl 0813.14039
- [Holm and Sjamaar 2008] T. S. Holm and R. Sjamaar, "Torsion and abelianization in equivariant cohomology", *Transform. Groups* **13**:3-4 (2008), 585–615. MR 2009j:57041 Zbl 1221.55007

- [Kahn 1999] B. Kahn, "Motivic cohomology of smooth geometrically cellular varieties", pp. 149–174 in *Algebraic K-theory* (Seattle, WA, 1997), edited by W. Raskind and C. Weibel, Proc. Sympos. Pure Math. **67**, Amer. Math. Soc., Providence, RI, 1999. MR 2001b:14031 Zbl 0954.19004
- [Krishna 2009a] A. Krishna, "Equivariant *K*-theory and higher Chow groups of smooth varieties", preprint, 2009. arXiv 0906.3109
- [Krishna 2009b] A. Krishna, "Riemann–Roch for equivariant *K*-theory", preprint, 2009. arXiv 0906. 1696
- [Krishna and Levine 2008] A. Krishna and M. Levine, "Additive higher Chow groups of schemes", J. *Reine Angew. Math.* **619** (2008), 75–140. MR 2009d:14005 Zbl 1158.14009
- [M. Demazure 1970] A. G. M. Demazure (editor), *Schémas en groupes, III*, Lecture Notes in Math. **153**, Springer, Berlin, 1970. Zbl 0212.52810
- [Merkurjev 2005] A. S. Merkurjev, "Equivariant *K*-theory", pp. 925–954 in *Handbook of K-theory*, vol. 2, edited by E. M. Friedlander and D. R. Grayson, Springer, Berlin, 2005. MR 2006m:14011 Zbl 1108.19002
- [Mumford et al. 1994] D. Mumford, J. Fogarty, and F. Kirwan, *Geometric invariant theory*, 3rd ed., Ergebnisse der Mathematik und ihrer Grenzgebiete (2) **34**, Springer, Berlin, 1994. MR 95m:14012 Zbl 0797.14004
- [Panin 2003] I. Panin, "Oriented cohomology theories of algebraic varieties", *K*-*Theory* **30**:3 (2003), 265–314. MR 2005f:14043 Zbl 1047.19001
- [Sankaran and Uma 2003] P. Sankaran and V. Uma, "Cohomology of toric bundles", *Comment. Math. Helv.* **78**:3 (2003), 540–554. MR 2004f:14076 Zbl 1050.14047
- [Springer 1998] T. A. Springer, *Linear algebraic groups*, 2nd ed., Progress in Mathematics 9, Birkhäuser, Boston, MA, 1998. MR 99h:20075 Zbl 0927.20024
- [Sumihiro 1975] H. Sumihiro, "Equivariant completion, II", J. Math. Kyoto Univ. 15:3 (1975), 573–605. MR 52 #8137 Zbl 0331.14008
- [Thomason 1986] R. W. Thomason, "Lefschetz–Riemann–Roch theorem and coherent trace formula", *Invent. Math.* **85**:3 (1986), 515–543. MR 87j:14028 Zbl 0653.14005
- [Thomason 1988] R. W. Thomason, "Equivariant algebraic vs. topological *K*-homology Atiyah–Segal-style", *Duke Math. J.* **56**:3 (1988), 589–636. MR 89f:14015 Zbl 0655.55002
- [Thomason 1993] R. W. Thomason, "Les *K*-groupes d'un schéma éclaté et une formule d'intersection excédentaire", *Invent. Math.* **112**:1 (1993), 195–215. MR 93k:19005 Zbl 0816.19004
- [Totaro 1999] B. Totaro, "The Chow ring of a classifying space", pp. 249–281 in *Algebraic K-theory* (Seattle, WA, 1997), edited by W. Raskind and C. Weibel, Proc. Sympos. Pure Math. **67**, Amer. Math. Soc., Providence, RI, 1999. MR 2001f:14011 Zbl 0967.14005
- [Vezzosi and Vistoli 2003] G. Vezzosi and A. Vistoli, "Higher algebraic *K*-theory for actions of diagonalizable groups", *Invent. Math.* **153**:1 (2003), 1–44. MR 2006c:19003 Zbl 1032.19001
- [Vezzosi and Vistoli 2005] G. Vezzosi and A. Vistoli, "Erratum to [Vezzosi and Vistoli 2003]", *Invent. Math.* **161**:1 (2005), 219–224. MR 2178662 Zbl 1032.19001
- [Vistoli 1989] A. Vistoli, "Characteristic classes of principal bundles in algebraic intersection theory", *Duke Math. J.* **58**:2 (1989), 299–315. MR 90k:14052 Zbl 0685.14006

Communicated by Hélène Esnault

Received 2011-11-10 Revised 2012-02-28 Accepted 2012-03-28

amal@math.tifr.res.in School of Mathematics, Tata Institute of Fundamental Research, 1 Homi Bhabha Road, Mumbai 400005, India

mathematical sciences publishers



## **Guidelines for Authors**

Authors may submit manuscripts in PDF format on-line at the Submission page at the ANT website.

**Originality**. Submission of a manuscript acknowledges that the manuscript is original and and is not, in whole or in part, published or under consideration for publication elsewhere. It is understood also that the manuscript will not be submitted elsewhere while under consideration for publication in this journal.

Language. Articles in ANT are usually in English, but articles written in other languages are welcome.

**Required items**. A brief abstract of about 150 words or less must be included. It should be self-contained and not make any reference to the bibliography. If the article is not in English, two versions of the abstract must be included, one in the language of the article and one in English. Also required are keywords and subject classifications for the article, and, for each author, postal address, affiliation (if appropriate), and email address.

**Format**. Authors are encouraged to use LATEX but submissions in other varieties of TEX, and exceptionally in other formats, are acceptable. Initial uploads should be in PDF format; after the refereeing process we will ask you to submit all source material.

**References**. Bibliographical references should be complete, including article titles and page ranges. All references in the bibliography should be cited in the text. The use of BibT<sub>E</sub>X is preferred but not required. Tags will be converted to the house format, however, for submission you may use the format of your choice. Links will be provided to all literature with known web locations and authors are encouraged to provide their own links in addition to those supplied in the editorial process.

**Figures**. Figures must be of publication quality. After acceptance, you will need to submit the original source files in vector graphics format for all diagrams in your manuscript: vector EPS or vector PDF files are the most useful.

Most drawing and graphing packages (Mathematica, Adobe Illustrator, Corel Draw, MATLAB, etc.) allow the user to save files in one of these formats. Make sure that what you are saving is vector graphics and not a bitmap. If you need help, please write to graphics@msp.org with details about how your graphics were generated.

White space. Forced line breaks or page breaks should not be inserted in the document. There is no point in your trying to optimize line and page breaks in the original manuscript. The manuscript will be reformatted to use the journal's preferred fonts and layout.

**Proofs**. Page proofs will be made available to authors (or to the designated corresponding author) at a Web site in PDF format. Failure to acknowledge the receipt of proofs or to return corrections within the requested deadline may cause publication to be postponed.

# Algebra & Number Theory

Volume 7 No. 2 2013

The system of representations of the Weil–Deligne group associated to an abelian variety RUTGER NOOT	243
Fourier–Jacobi coefficients of Eisenstein series on unitary groups BEI ZHANG	283
The phase limit set of a variety MOUNIR NISSE and FRANK SOTTILE	339
Base change behavior of the relative canonical sheaf related to higher dimensional moduli ZSOLT PATAKFALVI	353
Two ways to degenerate the Jacobian are the same JESSE LEO KASS	379
Arithmetic motivic Poincaré series of toric varieties HELENA COBO PABLOS and PEDRO DANIEL GONZÁLEZ PÉREZ	405
Maximal ideals and representations of twisted forms of algebras MICHAEL LAU and ARTURO PIANZOLA	431
Higher Chow groups of varieties with group action AMALENDU KRISHNA	449

